The Nexus Between Energy Systems and Public Health

An investigation into the co-impacts of energy sector technology transitions on outdoor air pollution and public health in the United Kingdom and Greater London

Melissa Christenberry Lott UCL Institute for Sustainable Resources University College London

Supervisors: Paul Ekins & Michael Davies

A thesis submitted for the degree of **Doctor of Philosophy**

July 11, 2017

This page was intentionally left blank.

Declaration

I, Melissa Christenberry Lott, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Melissa Christenberry Lott July 11, 2017 This page was intentionally left blank.

Abstract

There is significant value to be gained from insights on the trade-offs and synergies between proposed air quality and climate interventions. But, the models used in support of decarbonisation and air quality policies have not holistically considered these co-impacts.

This thesis documents the use of an energy systems model to quantify the co-impacts of decarbonisation pathways on air pollution and vice versa in the United Kingdom. This manuscript further documents the soft-linking of this model to a public health tool in order to quantify the public health implications of these pathways.

This research made a number of unique contributions to its field of research, including:

- incorporating air pollution emissions for particulate matter and nitrogen oxides, in the United Kingdom TIMES model (UKTM-UCL) to create the U.K. TIMES model with air quality (UKTM-UCL-AQ)¹
- the creation of the PollutION Emissions from EneRgy (PIONEER) model, an air pollution and public health tool²
- 3. soft-linking UKTM-UCL-AQ to PIONEER to quantify the air pollution and public health co-impacts of U.K. energy technology transitions for Greater London

The results suggest that there are numerous opportunities for climate and air quality policies to be mutually supportive. However, without considering their co-impacts, individual policies can undermine the others' progress and create tension between policy efforts. The results also show the increasing importance of modal shifting in the transport sector in order to avoid future air pollution challenges.

¹ The author of this thesis manuscript completed all TIMES model update work related to the transport sector (including development and implementation) as well as a significant portion of the electricity sector. She provided input to work for all other sectors, but did not directly implement these model updates.

² The PollutION Emissions from EneRgy (PIONEER) model was created specifically for this research project exclusively by the author of this thesis.

This page was intentionally left blank.

Table of Contents

Table of Contents7List of Figures12List of Tables16Publications Based on this PhD Thesis18Acknowledgements20Chapter 1 – Introduction211.1 Overview211.2 Global Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview392.1 Overview392.2.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Particulate Matter Exposure412.2.1.2 Health Impacts of Outdoor Air Pollution in the United Kingdom442.2.2.1 Public Health Organization Guidelines for Air Pollution442.2.2.1 Public Health Impact of Outdoor Air Pollution in the United Kingdom452.2.2.1.3 Life Tables572.3 Overview of existing studies58	Table of Contents7List of Figures12List of Tables16Publications Based on this PhD Thesis18Acknowledgements20Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom281.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview392.1 Overview392.2 Background and Context392.2.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Particulate Matter Exposure432.2.1.3 World Health Organization Guidelines for Air Pollution442.2.2.1.4 Linking Climates of Outdoor Air Pollution in the United Kingdom452.2.1.1 Impact Pathway Approach55
List of Figures12List of Tables16Publications Based on this PhD Thesis18Acknowledgements20Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview38Chapter 2 - Literature Review392.1 Overview392.2.1.1 Health Impacts of Particulate Matter Exposure412.2.1.2 Health Impacts of Nitrogen Dioxide Exposure432.2.1.3 World Health Organization Guidelines for Air Pollution442.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom442.2.2.1.1 Impact Pathway Approach552.2.2.1.1 Damage Cost Approach562.2.2.1.3 Life Tables572.3 Overview of existing studies58	List of Figures12List of Tables16Publications Based on this PhD Thesis18Acknowledgements20Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview392.1 Overview392.2 Background and Context392.2.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Particulate Matter Exposure412.2.1.2 World Health Organization Guidelines for Air Pollution442.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom452.2.1.1 Impact Pathway Approach55
List of Tables16Publications Based on this PhD Thesis18Acknowledgements20Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context.251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview38Chapter 2 - Literature Review392.1 Overview392.2.1.1 Health Impacts of Particulate Matter Exposure412.2.1.2 Health Impacts of Nitrogen Dioxide Exposure432.2.1.3 World Health Organization Guidelines for Air Pollution442.2.2 The Health Impact of Outdoor Air Pollution in the United Kingdom452.2.2.1.1 Impact Pathway Approach552.2.2.1.3 Life Tables572.3 Overview of existing studies58	List of Tables16Publications Based on this PhD Thesis18Acknowledgements20Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview392.1 Overview392.2.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Particulate Matter Exposure432.2.1.3 World Health Organization Guidelines for Air Pollution442.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom452.2.1.1 Impact Pathway Approach55
Publications Based on this PhD Thesis18Acknowledgements20Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview392.1 Overview392.2.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Particulate Matter Exposure412.2.2.1.2 World Health Organization Guidelines for Air Pollution442.2.2.1.1 Impact of Outdoor Air Pollution in the United Kingdom452.2.2.1.1 Impact Pathway Approach552.2.2.1.3 Life Tables572.3 Overview of existing studies58	Publications Based on this PhD Thesis 18 Acknowledgements 20 Chapter 1 - Introduction 21 1.1 Overview 21 1.2 Global Context 22 1.3 UK Context 25 1.3.1 Air Quality Policy in the United Kingdom 25 1.3.2 Air Quality Monitoring in the United Kingdom 28 1.3.3 Greenhouse Gas Policy in the United Kingdom 29 1.3.4 Linking Climate and Air Quality Policies 31 1.4 Scope and objectives 34 1.5 Research Questions 37 1.6 Thesis Overview 39 2.1 Overview 39 2.2 Background and Context 39 2.1.1 World Health Organization Definitions and Guidelines for Air Pollution 40 2.2.1.1 Health Impacts of Nitrogen Dioxide Exposure 41 2.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom 44 2.2.2.1 Public Health Impact Calculations 54 2.2.2.1.1 Impact Pathway Approach 55
Acknowledgements20Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview38Chapter 2 - Literature Review392.1 Overview392.2 Background and Context392.2.1.1 Health Impacts of Particulate Matter Exposure412.2.1.2 Health Impacts of Outdoor Air Pollution in the United Kingdom442.2.2 The Health Impact of Outdoor Air Pollution in the United Kingdom452.2.2.1.1 Impact Pathway Approach552.2.2.1.1 Damage Cost Approach562.2.2.1.3 Life Tables572.3 Overview of existing studies58	Acknowledgements20Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview38Chapter 2 - Literature Review392.1 Overview392.2 Background and Context392.2.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Nitrogen Dioxide Exposure432.2.2.1 Public Health Impact Sof Outdoor Air Pollution in the United Kingdom452.2.2.1 Public Health Impact Calculations542.2.2.1 Limpact Pathway Approach55
Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview38Chapter 2 - Literature Review392.1 Overview392.2 Background and Context392.1.1 Health Organization Definitions and Guidelines for Air Pollution402.2.1.2 Health Impacts of Particulate Matter Exposure432.2.1.3 World Health Organization Guidelines for Air Pollution442.2.2.1.4 Linking Calculations542.2.2.1.5 Life Tables572.3 Overview of existing studies58	Chapter 1 - Introduction211.1 Overview211.2 Global Context221.3 UK Context.251.3.1 Air Quality Policy in the United Kingdom251.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview38Chapter 2 - Literature Review392.1 Overview392.2.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Nitrogen Dioxide Exposure432.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom442.2.2.1 Public Health Impact Calculations542.2.1.1 Impact Pathway Approach55
1.3.1 Air Quality Policy in the Onited Kingdom231.3.2 Air Quality Monitoring in the United Kingdom281.3.3 Greenhouse Gas Policy in the United Kingdom291.3.4 Linking Climate and Air Quality Policies311.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview392.1 Overview392.2 Background and Context392.1.1 Health Impacts of Particulate Matter Exposure412.2.1.2 Health Impacts of Nitrogen Dioxide Exposure432.2.1.3 World Health Organization Guidelines for Air Pollution442.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom452.2.1.1 Impact Pathway Approach552.2.2.1.1 Damage Cost Approach562.2.2.1.3 Life Tables572.3 Overview of existing studies58	1.3.1 All Quality Policy in the Onited Kingdom 23 1.3.2 Air Quality Monitoring in the United Kingdom 28 1.3.3 Greenhouse Gas Policy in the United Kingdom 29 1.3.4 Linking Climate and Air Quality Policies 31 1.4 Scope and objectives 34 1.5 Research Questions 37 1.6 Thesis Overview 38 Chapter 2 - Literature Review 39 2.1 Overview 39 2.2.1 World Health Organization Definitions and Guidelines for Air Pollution 40 2.2.1.1 Health Impacts of Particulate Matter Exposure 41 2.2.1.2 Health Impacts of Nitrogen Dioxide Exposure 43 2.2.1.3 World Health Organization Guidelines for Air Pollution 44 2.2.2.1 health Impacts of Outdoor Air Pollution in the United Kingdom 45 2.2.2.1 public Health Impact Calculations 54 2.2.2.1.1 Impact Pathway Approach 55
1.3.2 All Quality Mollicening in the Onited Kingdom 29 1.3.3 Greenhouse Gas Policy in the United Kingdom 29 1.3.4 Linking Climate and Air Quality Policies 31 1.4 Scope and objectives 34 1.5 Research Questions 37 1.6 Thesis Overview 38 Chapter 2 - Literature Review 39 2.1 Overview 39 2.2 Background and Context 39 2.2.1 World Health Organization Definitions and Guidelines for Air Pollution 40 2.2.1.1 Health Impacts of Particulate Matter Exposure 41 2.2.1.2 Health Impacts of Nitrogen Dioxide Exposure 43 2.2.1.3 World Health Organization Guidelines for Air Pollution 44 2.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom 45 2.2.1.1 Impact Pathway Approach 55 2.2.2.1.1 Damage Cost Approach 55 2.2.2.1.3 Life Tables 57 2.3 Overview of existing studies 58	1.3.2 All Quality Mollitoring in the Onlited Kingdom 20 1.3.3 Greenhouse Gas Policy in the United Kingdom 29 1.3.4 Linking Climate and Air Quality Policies 31 1.4 Scope and objectives 34 1.5 Research Questions 37 1.6 Thesis Overview 38 Chapter 2 - Literature Review 39 2.1 Overview 39 2.2.1 World Health Organization Definitions and Guidelines for Air Pollution 40 2.2.1.1 Health Impacts of Particulate Matter Exposure 41 2.2.1.2 Health Impacts of Nitrogen Dioxide Exposure 43 2.2.1.3 World Health Organization Guidelines for Air Pollution 44 2.2.2.1 Health Impacts of Outdoor Air Pollution in the United Kingdom 45 2.2.2.1 Public Health Impact Calculations 54 2.2.2.1.1 Impact Pathway Approach 55
1.3.4 Linking Climate and Air Quality Policies 31 1.4 Scope and objectives 34 1.5 Research Questions 37 1.6 Thesis Overview 38 Chapter 2 - Literature Review 39 2.1 Overview 39 2.2 Background and Context 39 2.1.1 World Health Organization Definitions and Guidelines for Air Pollution 40 2.2.1.1 Health Impacts of Particulate Matter Exposure 41 2.2.1.2 Health Impacts of Nitrogen Dioxide Exposure 43 2.2.1.3 World Health Organization Guidelines for Air Pollution 44 2.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom 45 2.2.1.1 Impact Pathway Approach 55 2.2.2.1.1 Damage Cost Approach 56 2.2.2.1.3 Life Tables 57 2.3 Overview of existing studies 58	1.3.4 Linking Climate and Air Quality Policies 31 1.4 Scope and objectives 34 1.5 Research Questions 37 1.6 Thesis Overview 38 Chapter 2 - Literature Review 39 2.1 Overview 39 2.2 Background and Context 39 2.2.1 World Health Organization Definitions and Guidelines for Air Pollution 40 2.2.1.1 Health Impacts of Particulate Matter Exposure 41 2.2.1.2 Health Impacts of Nitrogen Dioxide Exposure 43 2.2.1.3 World Health Organization Guidelines for Air Pollution 44 2.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom 45 2.2.1.1 Jublic Health Impact Calculations 54 2.2.2.1.1 Impact Pathway Approach 55
1.4 Scope and objectives341.5 Research Questions371.6 Thesis Overview38Chapter 2 - Literature Review392.1 Overview392.2 Background and Context392.1.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Particulate Matter Exposure412.2.1.2 Health Impacts of Nitrogen Dioxide Exposure432.2.1.3 World Health Organization Guidelines for Air Pollution442.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom452.2.2.1.1 Impact Pathway Approach552.2.2.1.1 Damage Cost Approach562.2.2.1.3 Life Tables572.3 Overview of existing studies58	1.4 Scope and objectives 34 1.5 Research Questions 37 1.6 Thesis Overview 38 Chapter 2 - Literature Review 39 2.1 Overview 39 2.2 Background and Context 39 2.2.1 World Health Organization Definitions and Guidelines for Air Pollution 40 2.2.1.1 Health Impacts of Particulate Matter Exposure 41 2.2.1.2 Health Impacts of Nitrogen Dioxide Exposure 43 2.2.1.3 World Health Organization Guidelines for Air Pollution 44 2.2.2.1 Health Impacts of Outdoor Air Pollution in the United Kingdom 45 2.2.2.1 Public Health Impact Calculations 54 2.2.2.1.1 Impact Pathway Approach 55
Chapter 2 - Literature Review392.1 Overview392.2 Background and Context392.2 Background and Context392.2.1 World Health Organization Definitions and Guidelines for Air Pollution402.2.1.1 Health Impacts of Particulate Matter Exposure412.2.1.2 Health Impacts of Nitrogen Dioxide Exposure432.2.1.3 World Health Organization Guidelines for Air Pollution442.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom452.2.2.1 Public Health Impact Calculations542.2.2.1.1 Impact Pathway Approach552.2.2.1.2 Life Tables572.3 Overview of existing studies58	Chapter 2 - Literature Review 39 2.1 Overview 39 2.2 Background and Context 39 2.2.1 World Health Organization Definitions and Guidelines for Air Pollution 40 2.2.1.1 Health Impacts of Particulate Matter Exposure 41 2.2.1.2 Health Impacts of Nitrogen Dioxide Exposure 43 2.2.1.3 World Health Organization Guidelines for Air Pollution 44 2.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom 45 2.2.2.1 Public Health Impact Calculations 54 2.2.2.1.1 Impact Pathway Approach 55
 2.3.1 Literature Identification Process	2.2.2.1.1 Damage Cost Approach

2.3.3.4 Dessens, et. al (2014)	70
2.3.4 National Studies	71
2.3.4.1 Barker, et. al (2010)	72
2.3.4.2 Jensen, et. al. (2013)	73
2.3.4.3 Pye, et. al (2008a and 2008b)	76
2.3.4.4 Wadud and Waitz (2011)	77
2.3.4.5 U.S. EPA (2009)	78
2.3.5 Urban Studies	79
2.3.5.1 Jack and Kinney (2010)	79
2.3.5.2 Woodcock, et. al. (2009)	80
2.3.5.3 Jarrett et. al (2012)	81
2.4 Modelling Air Quality Co-Impacts	82
2.5 Articulation of the research gap	87
Chapter 3 - Methodology	
3.1 Overview	
3.2 Sustainability Gap Methodology	97 00
3.3.1 A Brief History of Energy Systems Modelling	
3.3.1.1 Choice Of The Energy Systems Model Used For This Research	104
3.3.2 TIMES Methodology Overview	
3.3.2.1 The TIMES Model Generator	109
3.3.2.2 TIMES Model Inputs	110
3.3.2.3 The TIMES Objective Function	113
3.3.3 Energy System Modelling in the United Kingdom	116
3.4 Background – Energy and Air Pollution in the United Kingdom	118
3.4.1 U.K. Total Primary Energy Supply & Final Energy Consumption	118
3.4.2 Greenhouse Gas Emissions in the United Kingdom	121
3.4.3 Air Pollution Emissions in the United Kingdom	122
3.4.4 Climate Change Legislation	129
3.4.5 Air Quality Legislation	130
3.5 Sustainability Gap (SGAP) Calculations	131
3.6 UKTM-UCL-AQ	134 13/
3.6.1.1 Emission Factors	127
3.6.2 Analysis of Emissions Accounting Coverage	1/10
3.6.3 Damage Cost Database	1/10
3.7 Damand Drivers & Technology Cost Assumptions	142 1 <i>1</i> E
3.8 Scenarios for the United Kingdom	145

3.9 The PollutION Emissions from EneRgy (PIONEER) Model	146	
3.9.1 PIONEER Methodology Overview	146	
3.9.1.1 Disaggregation of Demand in PIONEER	150	
3.9.1.2 Linking UKTM-UCL-AQ Outputs to PIONEER	152	
3.9.1.4 Incorporating Scenario Constraints in PIONEER	153	
3.9.1.4.1 Scenarios with Changes to the Technology Transition Pathway 3.9.1.4.2 Scenarios with Behavioural Change	457	153 156
3.9.1.5 Calculating Public Health Impacts in PIONEER	157	
3.9.2 Mode Shifting Potential in Greater London	162	
3.10 Ongoing Related Projects	164	
 Chapter 4 – UK Analysis	165 167 168 	165
4.3.2 Scenarios that include damage costs	173	
4.4 Limiting New Nuclear Capacity 4.5 Discussion & Conclusions	185 190	
 5.1 Overview 5.2 Key Outcomes & Insights 5.3 Scenario Development 5.3.1 Emission Factors 	195 196 197 202	192
5.3.2 Demand Disaggregation Factors	209	
5.4 Outputs for Scenarios Without Modal Shift 5.4.1 Results	210 211	
5.4.2 Sensitivity Analysis for Scenarios without Behavioural Change	214	
5.5 Scenarios with Behavioural Change 5.5.1 Results	223 226	
5.5.2 Sensitivity Analysis for Scenarios with Behavioural Change	230	
5.5.3 Public Health Impacts	234	
5.6 Discussion & Conclusions	239	
Chanter 6 – Discussion and Conclusions		241
6.1 Overview and Key Contributions	241	671
6.2 Key Insights and Comparisons with Previous Work	243	
6.2.1 Key Insights	243	
6.2.2 Comparison of United Kingdom Results with Previous Work	247	
6.2.2.1 Comparison with IEA (2016)	247	
6.2.2.1 Comparison with Pye, et. al. (2008)	249	
6.2.3 Comparison of Greater London Results with Previous Work	251	
6.2.3.1 Comparison with Barker et. al (2010)	251	

6.2.3.2 Comparison with Woodcock, et. al (2009)	252
6.2.3.3 Comparison with Jarrett, et. al. (2012)	252
 6.3 Significance for the Modelling Community & Policy Development 6.4 Limitations of the Research Approach 6.4 1 Cost Optimisation Approach 	253 254 255
6.4.1.1 Energy Service Demand Assumptions	256
6.4.1.2 Resource Availability and Technology Costs Assumptions	256
6.4.1.3 Emission Eactor and Damage Cost Assumptions	257
6.4.2 Spatial Resolution & System Boundaries	259
6.4.3 Temporal Resolution	260
6.5 Opportunities for Future Work	200 261
Ribliography	201 262
Appendix A.1: Documentation for the TIMES Model - PART II (July 2016) A.2: Emission Factors – National Scale (United Kingdom) Analysis A.3: Graphs of Air Pollution by Sector for All Scenarios (National Scale, United F	287 287 647 Xingdom) 709
A.3.1: Graphs of Air Pollution by Sector – Base Scenario	709
A.3.2: Graphs of Air Pollution by Sector – Ref Scenario	711
A.2.3: Graphs of Air Pollution by Sector – lowGHG Scenario	714
A.2.4: Graphs of Air Pollution by Sector – Base_DAMC Scenario	716
A.2.5: Graphs of Air Pollution by Sector – Ref_DAMC Scenario	719
A.2.6: Graphs of Air Pollution by Sector – lowGHG_DAMC Scenario	721
A.3: Demand Values – Urban Scale (Greater London) Analysis A.3.1: For Scenarios without Behaviour Change	725 725
A.3.2: For Scenarios with Behaviour Change	725
A.3.3: Zero-Tailpipe Emission Cars Availability versus Demand For Scenarios Behaviour Change	with 726
A.4: Emission Factors – Urban Scale (Greater London) Analysis for Scenarios wi Behaviour Change (i.e. no mode shift) for all Demand Groups A.4.1: PM ₁₀	thout 729 729
A.4.2: PM _{2.5}	730
A.4.3: NO _x	731
A.5: Emission Factors – Urban Scale (Greater London) Analysis for Scenarios wi Mode Shift Away from Cars A.5.1: PM ₁₀	th 20% 732 732
A.5.1.1: Group 1	732
A.5.1.2: Group 2	733
A.5.1.3: Group 3	734

A.5.2: PM _{2.5}	735
A.5.1.1: Group 1	735
A.5.1.2: Group 2	736
A.5.1.3: Group 3	737
A.5.3: NO _x	738
A.5.1.1: Group 1	738
A.5.1.2: Group 2	739
A.5.1.3: Group 3	740
A.6: Emission Factors – Urban Scale (Greater London) Analysis for Scenarios wi Mode Shift Away from Cars	ith 40% 741
A. 5. 1. 1: Croup 1	7/1
A.5.1.1: Group 1	741
A.5.1.2. Group 2	742
A.5.1.3: Group 3 A.6.2: PM _{2.5}	743 744
A.5.1.1: Group 1	744
A.5.1.2: Group 2	745
A.5.1.3: Group 3	746
A.6.3: NO _x	747
A.5.1.1: Group 1	747
A.5.1.2: Group 2	748
A.5.1.3: Group 3	749

List of Figures

Figure 1.1: Average annual outdoor PM2.5 concentrations in selected urban areas (International Energy Agency (IEA), 2016; World Health Organization, 2016b)
Figure 2.1: Particulate Matter (PM _{2.5}) Concentrations across the United Kingdom from Yim and Barrett (Yim and Barrett, 2012)
Figure 2.2: Average annual outdoor PM _{2.5} concentrations in selected urban areas (International Energy Agency (IEA), 2016; World Health Organization, 2016a, 2016c)
Figure 3.1: The Topical Coverage of the Core Models Used in This Research Project
Figure 3.2: The Sustainability Gap (SGAP) methodology (Lott, Ekins and Davies, 2014)
Figure 3.3: Three-Dimensional Assessment of Energy-Economy Models (Hourcade, Jaccard and Bataille, 2006)
Figure 3.4: Schematic of a generalised TIMES model (Loulou et al., 2005) 112
Figure 3.5: Final Energy Consumption by Country in the United Kingdom (2010) (Department of Energy and Climate Change (DECC), 2015a)
Figure 3.6 Final Energy Consumption by Country in the United Kingdom with Greater London Break-Out (2010) (Department of Energy and Climate Change (DECC), 2015a) 119
Figure 3.7 Final Energy Consumption by Sector in the United Kingdom (2010) (Department of Energy and Climate Change (DECC), 2015a)
Figure 3.8: Final Energy Consumption in Greater London by Sector (2010) (Department of Energy and Climate Change (DECC), 2015a)
Figure 3.9: UK greenhouse gas emissions (1990-2010), weighted by global warming potential (Department of Energy and Climate Change (DECC), 2016a)
Figure 3.10: Annual emissions of particulate matter (PM ₁₀) by sector, 1970-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)
Figure 3.11: Annual emissions of particulate matter (PM _{2.5}) by sector, 1970-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)
Figure 3.12: Map of annual emissions of particulate matter (PM _{2.5}) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016d)
Figure 3.13: Annual Emissions of NMVOCs by sector, 1970-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)
Figure 3.14: Annual Emissions of SO _x by sector, 1970-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)
Figure 3.15: Annual Emissions of NO _x by sector, 1970-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)
Figure 3.16: Annual Emissions of NH ₃ by sector, 1980-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)
Figure 3.17: Map of annual emissions of non-methane volatile organic compounds (NMVOCs) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016c)
Figure 3.18: Map of annual emissions of nitrogen oxides (NO _x as NO ₂) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016b)

Figure 3.19: Map of annual emissions of sulphur oxides (SO _x as SO ₂) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016e)
Figure 3.20: Map of annual emissions of ammonia (NH ₃) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016a)
Figure 3.21: Overview of the VEDA system for TIMES modelling (Energy Technology Systems Analysis Program (ETSAP), 2016)
Figure 3.22: Analysis Flow Diagram UKTM-UCL-AQ
Figure 3.23: Scenarios Map for United Kingdom Analysis
Figure 3.24: Analysis Flow Diagram for Soft-linked PIONEER and UKTM-UCL-AQ Models 149
Figure 3.25 Scenario Map for Greater London Analysis
Figure 3.26: Geographies Included in the LAEI 2013 Publication, from the GLA (Martin, 2016) 158
Figure 4.1: Topical Coverage of the UKTM-UCL-AQ Model in Relation to PIONEER and CMAQ
Figure 4.2: Scenarios Map for United Kingdom Analysis
Figure 4.3: Total Air Pollution Emissions by Type in the United Kingdom for Scenarios Without Damage Costs, 2010-2050
Figure 4.4 Total Particulate Matter (PM ₁₀) Emissions in the United Kingdom for lowGHG_lessconstrainedbio" and "lowGHG", 2010-2050
Figure 4.5: Total Particulate Matter (PM _{2.5}) Emissions in the United Kingdom for lowGHG_lessconstrainedbio" and "lowGHG", 2010-2050
Figure 4.6: Primary energy consumption (PJ) in 2050 by scenario
Figure 4.7: Total PM ₁₀ Emissions from Energy by Sub-Sector, 2010-2050
Figure 4.8: Air Pollution Levels in 2050 by Scenario
Figure 4.9: Air Pollution Emissions from Road Transport by Technology for the lowGHG_DAMC scenario, 2010-2050
Figure 4.10: Transport demand by engine type (bvkm) for the base and based scenarios
Figure 4.11: Total PM ₁₀ emissions from the agriculture sector from 2010-2050 across six scenarios
Figure 4.12 Total PM ₁₀ emissions from the industrial sector from 2010-2050 across six scenarios 182
Figure 4.13: Total PM _{2.5} emissions from the industrial sector from 2010-2050 across six scenarios 183
Figure 4.14: Total SO _x emissions from the industrial sector from 2010-2050 across six scenarios 183
Figure 4.15: Total annual carbon dioxide equivalent emissions in the UK across six scenarios, 2010- 2050
Figure 4.16: Overall System Costs (annual, undiscounted), including CO ₂ or air pollution tax levels
Figure 4.17: Electricity Generation by Technology Type, 2010-2050 for lowGHG Scenario
Figure 4.18: Electricity Generation by Technology Type, 2010-2050 for lowGHG_NuclearLimit Scenario
Figure 4.19: Total annual carbon dioxide equivalent emissions in the U.K. for lowGHG and lowGHG_NuclearLimit scenarios, 2010-2050
Figure 4.20: Total annual PM ₁₀ emissions in the U.K. for lowGHG and lowGHG_NuclearLimit scenarios, 2010-2050

Figure 4.21: Total annual PM _{2.5} emissions in the U.K. for lowGHG and lowGHG_NuclearLimit scenarios, 2010-2050
Figure 4.22: Total annual nitrogen oxide (NO _x) emissions in the U.K. for lowGHG and lowGHG_NuclearLimit scenarios, 2010-2050
Figure 5.1: Analysis Flow Diagram for Soft-linked PIONEER and UKTM-UCL-AQ Models 198
Figure 5.2: Topical Coverage Map of the Core Models Used in This Research Project
Figure 5.3: Scenario Map for Greater London Analysis
Figure 5.4 Scenario Dimension Mapping
Figure 5.5 Car demand by engine type (bvkm) in the United Kingdom for the lowGHG_DAMC scenario
Figure 5.6: Total Greater London Car Demand (Group 1) Versus Availability of Zero-Tailpipe Emission Vehicles in the lowGHG-DAMC scenario (2010-2050)
Figure 5.7: Four Scenarios for Zero-Emission Car Deployment in Greater London versus Total Availability in the United Kingdom and Total Demand for Cars in Greater London (2010-2050) 206
Figure 5.8: London Emission Factors, Cars (2010 – 2050) for PM_{10} , $PM_{2.5}$ and NO_x
Figure 5.9: London Emission Factors, 2 Wheelers (2010 – 2050) for PM_{10} , $PM_{2.5}$ and NO_x 207
Figure 5.10: London Emission Factors, Buses (2010 – 2050) for PM_{10} , $PM_{2.5}$ and NO_x 208
Figure 5.11: London Emission Factors, LGVs (2010 – 2050) for PM_{10} , $PM_{2.5}$ and NO_x 208
Figure 5.12: London Emission Factors, HGVs ($2010 - 2050$) for PM_{10} , $PM_{2.5}$ and NO_x 209
Figure 5.13: Scenario Dimensions Covered in this Discussion
Figure 5.14: Total PM ₁₀ Emissions in Greater London area for scenarios without behavioural shifts for Group 1 demand values (2010-2050)
Figure 5.15: Total PM _{2.5} Emissions in Greater London area for scenarios without modal shifting for Group 1 demand values (2010-2050)
Figure 5.16: Total NO _x Emissions in Greater London area for scenarios without modal shifting for Group 1 demand values (2010-2050)
Figure 5.17 Factors Contributing to Modal Shifts from Cars to Public Transport in Greater London (Transport for London (TfL), 2014)
Figure 5.18: Demand in Greater London by Road Transport Type (2010-2050) for Group 1, 2, 3 219
Figure 5.19: Total PM ₁₀ Emissions in Greater London area for scenarios without behavior change for a range of demand groups, 2010-2050
Figure 5.20: Total PM _{2.5} Emissions in Greater London area for scenarios without behavior change for a range of demand groups, 2010-2050
Figure 5.21: Total NO _x Emissions in Greater London area for scenarios without behavior change for a range of demand groups, 2010-2050
Figure 5.22: Demand for Scenarios that Include Behavioural Change, 2010-2050 (Group 1)
Figure 5.23: Total PM ₁₀ Emissions in Greater London area for scenarios both with and without behavioural change, 2010 – 2050 (Group 1)
Figure 5.24: Total PM _{2.5} Emissions Greater London area for scenarios both with and without behavioural change, 2010 – 2050 (Group 1)

Figure 5.25: Total NO _x Emissions Greater London area for scenarios both with and without behavioural change, 2010 – 2050 (Group 1)	9
Figure 5.26: Demand for Scenarios that Include Behavioural Change, 2010-2050 (Group 2)	0
Figure 5.27: Demand for Scenarios that Include Behavioural Change, 2010-2050 (Group 3)	1
Figure 5.28 Updated Car Demand Inputs for UKTM-UCL-AQ for All Groups in Scenarios with Behavioral Shift	1
Figure 5.29: Total PM ₁₀ Emissions in Greater London area for all scenarios and demand groups, 2010-2050	3
Figure 5.30: Total PM _{2.5} Emissions in Greater London area for all scenarios and demand groups, 2010-2050	3
Figure 5.31: Total NO _x Emissions in Greater London area for all scenarios and demand groups, 2010-2050	- 4
Figure 5.32: Annual Premature Deaths by Scenario for PM2.5 and NOx (as NO2) in 2050 for Groups 1, 2 and 3	; 7
Figure 5.33: Life Years Lost by Scenario for PM _{2.5} and NO _x (as NO ₂) in 2050 for Groups 1, 2 and 3	8

List of Tables

Table 1.1: Primary health impacts and main sources of pollution in the UK (Committee on the Medical Effects of Air Pollution (COMEAP), 2011)	27
Table 1.2: U.K. Carbon Budgets (Committee on Climate Change, 2010, 2015b)	30
Table 2.1: WHO Air Quality Guidelines	45
Table 2.2: London Mortality Burden Estimates Resulting from Long-Term Air Pollution Exposure (Miller, 2010; Yim and Barrett, 2012; Gowers, Miller and Stedman, 2014; Walton <i>et al.</i> , 2015))50
Table 2.3: Number of Journal Articles Identified Using UCL Library Search Engine for Journal Articles Only in April 2014	60
Table 2.4: Number of Journal Articles Identified Using Elsevier Science Direct Search Engine in April 2014	61
Table 2.5: Comparison of Key Related Studies of Interest	63
Table 2.6: The treatment of air quality co-benefits by integrated assessment models of climate chan policy (Stern and Taylor, 2006; Intergovernmental Panel on Climate Change (IPCC), 2007; Energy Information Administration, 2008; Department of Energy and Climate Change (DECC 2009; Nemet, Holloway and Meier, 2010)	ge), 83
Table 3.1 Classification of energy-economy models (Bhattacharyya and Timilsina, 2010) from (Pandey, 2002) 1	03
Table 3.2: Air pollution standards and targets for SGAP calculations (World Health Organization, 2005; Committee on the Medical Effects of Air Pollution (COMEAP), 2011; Department for Environment Food and Rural Affairs (DEFRA), 2013c; Lott, Ekins and Davies, 2014) [,]	132
Table 3.3: SGAP and YtS indicator for key air pollutants (total annual emissions level targets) (Lott Ekins and Davies, 2014)	t, 33
Table 3.4: Air pollution inventory mapping between NAEI and UKTM-UCL-AQ (Lott <i>et al.</i> , 2016; Lott, Pye and Dodds, 2017)	41
Table 3.5: Particulate matter emissions that were excluded from UKTM-UCL-AQ in the 2010 emissions calibration, by sector (Lott <i>et al.</i> , 2016; Lott, Pye and Dodds, 2017)	41
Table 3.6: Damage costs by sector and subsector (modified from (Department for Environment Foo and Rural Affairs (DEFRA), 2011)) 1	od 44
Table 4.1: Scenario Overview 1	69
Table 4.2: Total Air Pollution by Scenario in 2050	76
Table 4.3: Total Transport Emissions by Type and Scenario in 2050	77
Table 4.4: Total Transport PM ₁₀ Emissions by Scenario, 2010 - 2050 1	77
Table 4.5: Total Transport PM2.5 Emissions by Scenario, 2010 - 2050	78
Table 4.6: Total Transport NO _x Emissions by Scenario, 2010 – 2050 1	78
Table 4.7: Total Transport SO _x Emissions by Scenario, 2010 – 2050 1	79
Table 4.8: Annual undiscounted energy systems costs for lowGHG and lowGHG_NuclearLimit scenarios	90
Table 5.1: Scenario Names and Definitions with regards to the Analysis Dimensions 2	201
Table 5.2 Car demand by engine type (bvkm) for electric and hydrogen vehicles in the United Kingdom for the lowGHG_DAMC scenario 2	204

Table 5.3 Demand for Transport in Greater London, 2010-2050 (bvkm) for Group 1	. 210
Table 5.4: Scenarios Isolating Impacts of Technological Change Dimension	. 210
Table 5.5 Demand in Greater London by Road Transport Type (2010-2050) for Group 1, 2, 3	. 218
Table 5.6: Scenarios Focusing on Behavioural Change Dimension	. 224
Table 5.7 Updated Road Transport Demand Projections from UKTM-UCL-AQ lowGHG_DAMC the 80:20 Scenarios (Group 1)	for 225
Table 5.8: Updated Road Transport Demand Projections from UKTM-UCL-AQ lowGHG_DAMO the 60:40 Scenarios (Group 1)	C for . 225

Publications Based on this PhD Thesis

Peer-Reviewed Journal Publications

Lott, M.C., Pye, S. and Dodds, P. (2017) *Quantifying the co-impacts of energy sector decarbonisation on outdoor air pollution in the United Kingdom*. Energy Policy. DOI: 10.1016/j.enpol.2016.11.028

Nick Watts, W Neil Adger, Sonja Ayeb-Karlsson, Yuqi Bai, Peter Byass, Diarmid Campbell-Lendrum, Tim Colbourn, Peter Cox, Michael Davies, Michael Depledge, Anneliese Depoux, Paula Dominguez-Salas, Paul Drummond, Paul Ekins, Antoine Flahault, Delia Grace, Hilary Graham, Andy Haines, Ian Hamilton, Anne Johnson, Ilan Kelman, Sari Kovats, Lu Liang, **Melissa Lott**, Robert Lowe, Yong Luo, Georgina Mace, Mark Maslin, Karyn Morrissey, Kris Murray, Tara Neville, Maria Nilsson, Tadj Oreszczyn, Christine Parthemore, David Pencheon, Elizabeth Robinson, Stefanie Schütte, Joy Shumake-Guillemot, Paolo Vineis, Paul Wilkinson, Nicola Wheeler, Bing Xu, Jun Yang, Yongyuan Yin, Chaoqing Yu, Peng Gong, Hugh Montgomery, Anthony Costello. (2016) *The Lancet Countdown: tracking progress on health and climate change*. The Lancet, Volume 389, Issue 10074, 1151–1164. DOI: 10.1016/S0140-6736(16)32124-9

Nick Watts, W Neil Adger, Paolo Agnolucci, Jason Blackstock, Peter Byass, Wenjia Cai, Sarah Chaytor, Tim Colbourn, Mat Collins, Adam Cooper, Peter M Cox, Joanna Depledge, Paul Drummond, Paul Ekins, Victor Galas, Delia Grace, Hilary Graham, Michael Grubb, Andy Haines, Ian Hamilton, Alasdair Hunter, Xujia Jiang, Moxuan Li, Ilan Kelman, Lu Liang, **Melissa Lott**, Robert Lowe, Yong Luo, Georgina Mace, Mark Maslin, Maria Nilsson, Tadj Oreszczyn, Steve Pye, Tara Quinn, My Svensdotter, Sergey Venevsky, Koko Warner, Bing Xu, Jun Yang, Yongyuan Yin, Chaoqing Yu, Qiang Zhang, Peng Gong, Hugh Montgomery, Anthony Costello. (2015) *Health and Climate Change: policy responses to protect public health*. The Lancet, Volume 386, Issue 10006, 1861–1914. DOI: 10.1016/S0140-6736(15)60854-6

Conference publications

Lott, M.C., Pye, S., Fais, B. and Dodds, P. (2016) *Up In The Air: A framework for quantifying the co-impacts of energy sector decarbonisation on outdoor air pollution*. International Energy Workshop. Cork, Ireland.

Williams, M., Beevers, S., Lott, M.C., and Kitwiroon, N. (2016) *The Impacts of Policies to Meet the UK Climate Change Act Target on Air Quality – an explicit modelling study.* International Energy Workshop. Cork, Ireland.

Lott, M.C. and Daly, H. (2015) *The Impacts of Transport Sector Decarbonisation Pathways on Air Quality and Public Health in the United Kingdom*. Journal of Transport and Health. DOI: 10.1016/j.jth.2015.04.550

Lott, Melissa C. and Hannah Daly. (2015) *The Impacts of Transport Sector Decarbonisation Pathways on Air Quality and Public Health in the United Kingdom*. 2015 Transport and Public Health Conference. London, United Kingdom.

Lott, Melissa C., Paul Ekins and Michael Davies. (2014) Using the Air Emissions Component of the "Years to Sustainability" Indicator to Determine Sustainability Trends in the United Kingdom – Health Impacts. ISEE Conference 2014. Reykjavik, Iceland.

Other Publications

Williams, M., Beevers, S., Kitwiroon, N., Dajnak, D., Walton, H., Lott, M.C., Fecht, Daniela, Holland, Mike. *Public health air pollution impacts of pathway options to meet the UK Climate Change Act target of 80% reduction of UK CO2 equivalent emissions by 2050*. National Institute for Health Research. [report submitted to NIHR in January 2017 – expected publication in mid-2017]

Lott, M.C. and Jon Cartwright. (2015) *Mobility in the city can have an effect on health.* Scientific American. Supplement of the French edition (Pour la Science) n°447 pp 54. January 2015. Edition M0770447-01. ISSN 0 153-4092.

Lott, Melissa C. (2015). *Why EVs can reduce (but not eliminate) urban air pollution*. Scientific American – Plugged In. November 30, 2015. Retrieved from https://blogs.scientificamerican.com/plugged-in/why-evs-can-reduce-but-not-eliminate-urban-air-pollution/

Lott, Melissa C. (2015). *Climate Change Mitigation Could Be the Greatest Global Health Opportunity of the 21st Century*. Scientific American – Plugged In. July 17, 2015. Retrieved from https://blogs.scientificamerican.com/plugged-in/climate-change-mitigation-could-bethe-greatest-global-health-opportunity-of-the-21st-century/

Lott, Melissa C. (2014). *The Cyclic Nature of "Crazy Bad" Air Pollution and Fuel Transitions*. Scientific American – Plugged In. August 11, 2014. Retrieved from https://blogs.scientificamerican.com/plugged-in/the-cyclic-nature-of-8220-crazy-bad-8221-air-pollution-and-fuel-transitions/

Acknowledgements

I would first like to thank my research supervisors, Professors Paul Ekins and Michael Davies for their input throughout the PhD process. I would also like to acknowledge and thank a number of my UCL colleagues who provided invaluable help and feedback during my research process – particularly Steve Pye, Andrew Smith, Paul Dodds, Birgit Fais, and Hannah Daly. Thank you to my PhD viva examiners, Professor Neil Strachan and Christian Brand. I am also appreciative of input from colleagues at Kings College London including Professor Frank Kelly, Sean Beevers, Martin Williams, David Dajnak and Nutthida Kitwiroon related to the air pollution and air quality aspects of this work.

This thesis is dedicated to my family - I am a better person for knowing each of you and am incredibly grateful for your constant support, encouragement and love.

To my mom and dad – I am so lucky to be your daughter. To my family, including incredible brother – thank you for teaching and inspiring me throughout my life, helping me to become a better person and to continuously learn and grow.

To my amazing husband, Francis – thank you for your support, understanding and encouragement throughout this process. Your ability to make me laugh is truly incredible. You are a wonderful person and I am lucky to have you as my partner.

Chapter 1 – Introduction

1.1 Overview

Access to energy is a foundation of modern life and one of the key differentiators between healthy, wealthy societies and sick, poor ones (International Energy Agency (IEA), 2004; Gaye, 2008). As populations grow and countries develop, energy demand has historically risen.

In many instances, increasing energy use will be beneficial to public health. Indeed, over the past century fossil fuels have contributed to huge improvements in global public health and development (Costello *et al.*, 2009). However, energy production and consumption have also led to some negative environmental and human health impacts, including those resulting from increased pollution levels. Rising energy demand will increase these pressures. As a result, the reduction of pollution from the energy sector has emerged as a key priority in energy and environmental policies around the globe, including in the United Kingdom (Sokhi and Kitwiroon, 2011).

With regards to the energy system, multiple air pollutants are often produced by the same individual technologies (e.g. fossil fuel power plants, gasoline and diesel vehicles) (Pye *et al.*, 2008; Lott, Pye and Dodds, 2017). Therefore, one can surmise that actions to reduce a subset of these emissions would have impacts on other pollutants. The question is "by how much?" and, furthermore, "in what direction?". For example, how could action to reduce greenhouse gas emissions impact urban air pollution?

While it is tempting to believe that actions to reduce a subset of air pollutants produced by the energy sector will lead to reductions in others, the true relationships are more nuanced. Indeed, technologies and policies that are designed to reduce a sub-set of pollutants can have a wide

range of impacts on others – both positive and negative. For example, fuel switching from fossil fuels to biomass can result in increasing levels of local air pollution and the introduction of carbon capture and storage (CCS) can increase non- CO_2 air pollution because of the parasitic load created by the capture and storage processes that can reduce net power output by 20-30% (Cohen, 2012; Lott, Pye and Dodds, 2017). In turn, it is vital to holistically consider a range of key pollutants in the design of a future energy systems and policies that impact their evolution.

Energy systems models that incorporate a range of air pollutants can provide insights on these co-impacts, as this research demonstrates. This chapter gives an overview of the context of this study on both a global and UK basis, including information on efforts to reduce air pollution from the UK energy sector as well as facilitate its decarbonisation (Section 1.2). This overview is followed by a discussion of the scope and overarching objectives of this study (Section 1.3) and a list of specific research questions that were explored (Section 1.4). The chapter concludes with details of the structure of this thesis, including the five chapters that follow this introduction (Section 1.5).

1.2 Global Context

There exists widespread agreement in the scientific community that outdoor air pollution is detrimental to the environment and human health, both through its contribution to global climate change and local air quality challenges (World Health Organization, 2013b; Watts *et al.*, 2015). Each year, air pollution kills more people than HIV, malaria, and tuberculosis combined (Carrington, 2015; Lelieveld *et al.*, 2015). Outdoor particulate air pollution (PM_{2.5}) alone causes an estimated 800,000 early deaths, corresponding to 6.4 million years of life lost each and every year (Cohen *et al.*, 2006). This public health threat is particularly increasing in urban areas, corresponding to rising energy demands in these population centres (Sokhi and Kitwiroon, 2011).

Current epidemiological and toxicological evidence supports the causal linkage between air pollution and public health effects (Beevers *et al.*, 2013; World Health Organization, 2013b; Williams *et al.*, 2014). Of particular significance to this research project are studies documenting a causal relationship between traffic-generated nitrogen oxides (NO_x) and particulate matter (PM) air pollution and the onset of childhood asthma, non-asthma respiratory symptoms, impaired lung function, total and cardiovascular mortality, and cardiovascular morbidity (Committee on the Medical Effects of Air Pollution (COMEAP), 2006; Health Effects Institute, 2010).

According to the World Health Organization (WHO) in their Review of Evidence on Health Aspects of Air Pollution (REVIHAAP) project published in 2013, the health impacts of particulate matter (PM) have been particularly well documented (World Health Organization, 2013b). Overall, this evidence shows that there exists "no evidence of a safe level of exposure or a threshold below which no adverse health effects occur" (Pope *et al.*, 2002; World Health Organization, 2013b). Indeed, the adverse health impacts from both short- and long-term particulate matter exposure have been documented in urban populations in both developed and developing countries (Pope *et al.*, 2002; World Health Organization, 2005).

Despite this existing body of evidence on the negative health impacts of air pollution, the vast majority (~90%) of Europeans living in urban areas and almost all (98%) of those living in cities in low and middle income countries are exposed to air pollution levels in excess of World Health Organization standards. As an illustration of this fact, current average annual outdoor particulate matter ($PM_{2.5}$) concentrations are shown in Figure 1.1 with the dashed line indicating the current World Health Organization guideline level for this type of air pollution

(International Energy Agency (IEA), 2016). This figure was published by the International Energy Agency in their 2016 report "Energy and Air Pollution", which is discussed in more detail elsewhere in this thesis.



Figure 1.1: Average annual outdoor PM2.5 concentrations in selected urban areas³ (International Energy Agency (IEA), 2016; World Health Organization, 2016b)

Sources: WHO (2016) Global Urban Ambient Air Pollution Database; Demographia (2015) for population; country groups per income based on World Bank (2016).

On the global scale, greenhouse gas emissions (CO₂-eq) and their contributions to climate change are also recognized as a significant threat to human health, particularly in developing economies (Haines *et al.*, 2009). In 2015, the multi-disciplinary and international Lancet Commission on Health and Climate Change concluded that "that tackling climate change could be the greatest global health opportunity of the 21st century" (Watts *et al.*, 2015).

According to the IPCC, global anthropogenic CO_2 emissions were approximately 38 Gigatonnes (Gt) in 2010 (Intergovernmental Panel on Climate Change (IPCC), 2014). Of this total, around two-thirds of these emissions come from the energy sector, with 17% (6.5 Gt)

³ graphic by the International Energy Agency (International Energy Agency (IEA), 2016)

being produced directly by road transport (Intergovernmental Panel on Climate Change (IPCC), 2014; International Energy Agency (IEA), 2015a). Indeed, air pollution from human activities is overwhelmingly produced by the energy sector including almost all nitrogen oxide (NO_x) and the majority (85%) of particulate matter (International Energy Agency (IEA), 2016). In the IPCC's Fifth Assessment Report, Working Group 3 highlights these interlinkages in the report's Chapter 7 on "Energy Systems" and go on to state that (Bruckner *et al.*, 2014):

"To avoid creating new environmental and health problems, assessments of mitigation technologies need to address a wide range of issues, such as land and water use, as well as air, water, and soil pollution, which are often location-specific."

1.3 UK Context

This section includes discussion of the air quality and greenhouse gas policies currently found in the United Kingdom. It also includes a section that is focused on current air quality monitoring practices in the United Kingdom.

1.3.1 Air Quality Policy in the United Kingdom

As mentioned by London Mayor Sadiq Khan in 2016, the United Kingdom's largest city was plagued by "pea soupers" (smog) in the first 60 years of the 20th century (Mayor of London. London Assembly, 2016b). Perhaps most famous was the "Great Smog" of 1952, which lasted five days and shrouded London in "a fog so thick and polluted it left thousands dead...The smoke-like pollution was so toxic it was even reported to have choked cows to death in the fields. It was so thick it brought road, air and rail transport to a virtual standstill" (Met Office, 2015). According to reports, about 4,000 people were known to have died prematurely as a result of this smog (Met Office, 2015). The subsequent year, a Committee on Air Pollution (sometimes referred to as the "Beaver Committee") was established under the chairmanship of

Sir Hugh Beaver. This committee would go on to recommend a Clean Air Act for the United Kingdom (Brimblecombe, 1987).

In 1956, the U.K. Parliament passed the initial Clean Air Act, which included a number of air pollution control measures (United Kingdom Parliament, 1956). Perhaps most significant was its requirement that only smokeless fuels be burned in London. Subsequent Acts – including a revision in 1968 and then a consolidated Clean Air Act in 1993 - would further restrict air pollution emissions from the energy sector, including transport, power production, and residential heating with a particular focus on small combustion processes (Abbott *et al.*, 2012). As a result, the type of smog seen in 1952 has become a distant memory for Londoners though challenges remain as discussed elsewhere in this manuscript (Met Office, 2015).

In addition to the Clean Air Act of 1993, the United Kingdom is also subject to a number of directives at the European (EU) level⁴, including the National Emissions Ceilings Directive (2001/81/EC) and the EU Air Quality Directive (2008/50/EC) and its legally binding limits on outdoor air pollution levels. The former requires that Member States develop and maintain national programmes to meet emissions ceilings and required reporting of emissions inventories for sulphur dioxide (SO₂), nitrogen oxides (NO_x), non-methane volatile organic compounds (NMVOCs), and ammonia (NH₃). The latter includes limits for particulate matter (both PM₁₀ and PM_{2.5}) and nitrogen dioxide (NO₂). Due to its lack of compliance with European Union guidelines, the United Kingdom's Supreme Court ruled in 2015 that the government must take action to reduce air pollution levels to meet EU Air Quality Directive

⁴ Though this could change significantly should the United Kingdom invoke article 50 of the Lisbon treaty, and withdraw from the European Union as is currently under discussion.

limits for outdoor air pollution, which it currently violates. The result of this decision has already led to the proposal of an Ultra-Low Emission Zone (ULEZ) in central London as a mechanism for reducing the high air pollution concentrations currently found in this urban area (Transport for London (TfL), 2015a, 2015b). The ULEZ is currently expected to become fully active in September 2020, though the current Mayor of London Sadig Khan has proposed to accelerate this timeline (Vaughan, 2016).

Due to their prevalence in the United Kingdom, both NO₂ and particulate matter (PM₁₀ and PM_{2.5}) are included by the Committee on the Medical Effects of Air Pollution (COMEAP) in their Review of the UK Daily Air Quality Index (DAQI), which "covers the [air pollutants] that are most likely to affect health on a day-to-day basis" (Committee on the Medical Effects of Air Pollution (COMEAP), 2011). The primary health impacts and main sources of particulate matter and nitrogen dioxide (NO₂) air pollution in the United Kingdom are outlined in Table 1.1 (Committee on the Medical Effects of Air Pollution (COMEAP), 2011). Noted here is that the main sources for both types of air pollution are combustion – in particular, combustion processes in the energy sector.

Air Pollutant	Primary Health Impacts	Main sources in the UK
Particulate	Respiratory and cardiovascular	Combustion - stationary (power
matter (PM ₁₀	illness, in particular for finer	plants, quarries, other industry)
and PM _{2.5})	particles (PM2.5 and smaller)	and mobile (transport)
Nitrogen	Airway inflammation, asthma,	Combustion (transportation and
Oxides (NO _x)	respiratory stress.	electricity production)
including NO ₂		
	Also, contributes to secondary	
	particle and ground level ozone	
	formation	

Table 1.1: Primary health impacts and main sources of pollution in the UK (Committee on the Medical Effects of Air Pollution (COMEAP), 2011)

Under the Environment Act 1995, the U.K. Government and devolved administrations in England, Scotland, Wales and Northern Ireland are responsible for producing a national air

quality strategy to address air quality challenges. This strategy was last reviewed and published in 2007 and set out a plan for meeting the United Kingdom's air quality objectives via action at national, regional and local levels for a number of pollutants including nitrogen dioxide, particulate matter, and sulphur dioxide. Under Part IV of this Act, along with Order 2002, local authorities in the UK are required to measure their local air quality and establish air quality management areas for locations requiring improvement (Department for Environment Food and Rural Affairs (DEFRA), 2013a).

While outdoor air pollution levels have improved considerably in the United Kingdom since the famous "pea soupers" (smog) seen in the first half of the 20th century and the decade after the Second World War – due in large part to the nation's Clean Air Act - an estimated 40,000 people still prematurely die each year due to exposure to outdoor air pollution with the overall cost for the UK economy estimated as £20 billion annually due to negative mortality impacts resulting from air pollution exposure (Royal College of Physicians, 2016). In London, ~9,400 people die prematurely each year due to anthropogenic particulate matter ($PM_{2.5}$) and nitrogen dioxide (NO_2) pollution exposure alone, with an estimated annual monetized cost of £1.4–3.7 billion (Walton *et al.*, 2015). The bottom of this range includes the estimated economic costs associated with long-term exposure to $PM_{2.5}$ and mortality, short-term exposure to $PM_{2.5}$ and hospital admissions, and short-term exposure to NO_2 and both premature deaths and hospital admissions. The top of this range replaces values for short-term exposure to NO_2 with longterm exposure to NO_2 and its impact on mortality.

1.3.2 Air Quality Monitoring in the United Kingdom

In order to monitor air quality in the United Kingdom, air pollution data for the United Kingdom are collected and published via the United Kingdom's Department for Environment

Food & Rural Affairs (DEFRA) in their Air Information Resource (UK-AIR) and National Atmospheric Emissions Inventory (NAEI), which consists of the Greenhouse Gas Inventory (GHGI) and the Air Quality Pollutant Inventory (AQPI). These databases are hosted and maintained by Ricardo-AEA on behalf of DEFRA.

The UK-AIR database includes data from approximately 300 monitoring sites across the UK, which automatically measure NO_2 and PM_{10} concentrations on an hourly basis (Department for Environment Food and Rural Affairs (DEFRA), 2013c). The National Atmospheric Emissions Inventory (NAEI) database is compiled by gathering activity data and background data, which is then used to calculate overall emissions levels and corresponding emissions factors (Department for Environment Food and Rural Affairs (DEFRA), 2013b).

At an urban level, the London Atmospheric Emissions Inventory (LAEI) provides an emissions inventory including sources and location for the Greater London area. In particular for the transport sector, this inventory includes vehicle speeds and flows for each road link and uses automatic number plate recognition data to build vehicle stock information. The LAEI is maintained by the Environmental Research Group at Kings College London⁵.

1.3.3 Greenhouse Gas Policy in the United Kingdom

In addition to the targeted clean air regulations discussed previously, the United Kingdom has set a long-term national greenhouse reduction target of 80% by 2050 compared to 1990 levels through the national Climate Change Act. Passed into law in 2008, this Act includes the requirement that the Government set a series of legally binding "carbon budgets" for each five-

⁵ http://www.kcl.ac.uk/lsm/research/divisions/aes/research/ERG/modelling/Emissions-Inventory.aspx

year period starting in 2008, which are shown in Table 1.2 (Committee on Climate Change, 2010, 2015b).

The Climate Change Act 2008 also established the Committee on Climate Change (CCC), an independent, statutory body that provides advice to the Government on how to set and meet the carbon budgets and tracks the country's progress toward meeting these budgets (United Kingdom Parliament, 2008). Advice from the Committee on Climate Change is published via formal reports that are open to the public. The United Kingdom is currently in its 2^{nd} Carbon Budget period and has passed a total of five (5) carbon budgets into law. The 2^{nd} Carbon Budget spans from 2013-2015 and requires a 29% reduction in total equivalent carbon dioxide (CO₂e) emissions compared to the 1990 base year. This reduction equates to a final carbon budget level of 2,782 million metric tons of CO₂e. Most recently, the U.K. Government passed the 5th carbon budget (2028 – 2032) into law in July 2016, based on guidance published by the Committee on Climate Change in November 2015 (Committee on Climate Change, 2015b).

Carbon Budget	Carbon Budget Level (MtCO ₂ e)	% emissions reduction below 1990 base year	Has been enacted into law?
1 st (2008 – 2012)	3,018	23%	yes
$2^{nd}(2013-2017)$	2,782	29%	yes
$3^{rd}(2018-2022)$	2,544	35% by 2020	yes
4 th (2023 - 2027)	1,950	50% by 2025	yes
$5^{\text{th}}(2028-2032)$	1,765	57%	yes

Table 1.2: U.K. Carbon Budgets (Committee on Climate Change, 2010, 2015b)

In its June 2016 report, the Committee on Climate Change (CCC) states that carbon dioxide equivalent (CO_2e) emissions in the United Kingdom have fallen by an average of 4.5% per year since 2012 (Committee on Climate Change, 2016). According to the CCC, these drops were almost entirely due to rapid decarbonisation in the power sector, particularly through the

rapid decline in coal for power generation in favour of renewables. Indeed, the report highlights that:

"There has been almost no progress in the rest of the [United Kingdom's] economy, where emissions have fallen less than 1% a year since 2012 on a temperature-adjusted basis. That is because there has been slow uptake of low-carbon technologies and behaviours in the buildings sector (i.e. low rates of insulation improvement, low takeup of low-carbon heat) and improved vehicle efficiency has been offset by increased demand for travel as the economy has grown and fuel prices have fallen...Progress will need to be broader to meet the recommended fifth carbon budget and to prepare sufficiently for 2050. For example, while the complete replacement of coal-fired generation with low-carbon generation in the power sector is an important part of our scenarios, this would provide less than half of the total emissions reduction required by 2030."

1.3.4 Linking Climate and Air Quality Policies

Both greenhouse gas and other key types of air pollution overwhelmingly arise from the energy sector, particularly via the combustion of fossil fuels (e.g. oil, natural gas, coal) and biomass (International Energy Agency (IEA), 2016). The energy sector represents the largest single source of greenhouse gas emissions globally according to the International Energy Agency (IEA), producing an estimated two-thirds of all greenhouse-gas emissions resulting from human activities (Intergovernmental Panel on Climate Change (IPCC), 2014; International Energy Agency (IEA), 2015a). Furthermore, energy sector technologies are responsible for almost all sulphur dioxide (SO₂) and nitrogen oxide (NO_x) emissions as well as around 85% of

particulate matter emissions produced around the world each year according to the IEA in their 2016 publication *Energy and Air Pollution* (International Energy Agency (IEA), 2016).

Given that multiple air pollutants are often produced by the same energy sector technologies, one can surmise that actions to reduce a subset of these emissions would have impacts on other air pollutants. This fact indicates the potential for tensions between efforts to reduce subsets of these pollutants. It also highlights the potential opportunities for substantial co-benefits if all types of air pollution are factored into the decision-making process (Watts *et al.*, 2015).

In Section 7.9 of the International Panel on Climate Change 5th Assessment Report, the authors highlight this potential for both opportunity and tensions, stating that (Bruckner *et al.*, 2014):

"Energy supply options differ with regard to their overall environmental and health impacts, not only their GHG emissions...Renewable energies are often seen as environmentally benign by nature; however, no technology—particularly in large scale application—comes without environmental impacts."

The United Kingdom's Department for Environment, Food and Rural Affairs (DEFRA) highlighted this opportunity in their 2010 report on the relationship between climate action and air pollution (Department for Environment Food and Rural Affairs (DEFRA), 2010). In this report, the authors state that (Department for Environment Food and Rural Affairs (DEFRA), 2010):

"Our commitments to building a low carbon economy as set out in the UK and Scottish Climate Change Acts will reduce air pollution, but choices about the route we take to 2050 will affect the scale of improvements to air quality. Factoring air quality into decisions about how to reach climate change targets results in policy solutions with even greater benefits to society. Optimising climate change policies for air pollution can yield additional benefits of some £24 billion (net present value) by 2050."

Existing studies have explored these potential co-impacts at a variety of scales as discussed in Chapter 2 of this thesis. However, none of these studies have conducted in-depth analysis of the nexus between climate change mitigation and its co-impacts on outdoor air pollution (and vice versa).

This fact is significant for UK energy policy makers, as highlighted in 2013, Jensen et. al. where these authors stated their belief that (Jensen *et al.*, 2013a):

"UK policy makers will, most likely, have to adopt elements which involve the initial net societal costs in order to achieve future emission targets and longer-term benefits from GHG reduction. Cost-effectiveness of GHG strategies is likely to require technological mitigation interventions and/or demand-constraining interventions with important health co-benefits and other efficient-enhancing policies that promote internalization of externalities."

These researchers also identified the need to develop holistic assessment methodologies that include the total co-impacts (both positive and negative) of these technological and demand-constraining interventions, stating (Jensen *et al.*, 2013a):

"Health co-benefits can play a crucial role in bringing down net costs, but our results also suggest the need for adopting holistic assessment methodologies which give proper consideration to welfare-improving health co-benefits with potentially negative economic repercussions (such as increased longevity)."

In their work, Jensen et. al. highlight that increased longevity comes with a combination of positive and negative economic impacts. For instance, in their study these authors discuss the impacts of increased longevity on total social benefit pay-outs to elderly individuals (Jensen *et al.*, 2013a).

1.4 Scope and objectives

The scope of this research is broadly defined under three overarching goals, which are to better understand the following with respect to the energy sector:

- The co-impacts of the United Kingdom's efforts to decarbonize the energy sector on other types of outdoor air pollution (including particulate matter and nitrogen oxides) at both the national and urban scale.
- The ways in which considering the costs of other types of outdoor air pollution (including particulate matter and nitrogen oxides) might alter the "optimum" decarbonisation pathway for the national energy sector.
- The extent to which local action in the urban transport sector could potentially address local air pollution challenges and contribute to national progress toward the decarbonisation of the energy sector.

As a result, this research included the outdoor air pollution generated by local energy systems in the United Kingdom and the Greater London area (e.g. pollution from cars travelling through the city or region of interest). In turn, it did not include a full accounting of emissions produced in the production of energy technologies (e.g. pollution produced in the construction of solar panels or vehicles in overseas manufacturing centres). It also did not include shifts in sources of cross-boundary pollution originating outside of the United Kingdom, though these aspects were considered in a related collaborative project between this researcher and colleagues at University College London and Kings College London (Williams *et al.*, 2016). Previous work by the IPCC's Working Group 3 as presented in Chapter 7 of this report discussed the lifecycle impacts of renewable and coal power generation technologies, concluding that (Bruckner *et al.*, 2014):

"Reducing fossil fuel combustion, especially coal combustion, can reduce many forms of pollution and may thus yield co-benefits for health and ecosystems...most renewable power projects offer a reduction of emissions contributing to particulate matter exposure even compared to modern fossil fuel-fired power plants with state-of-the-art pollution control equipment."

This limitation in scope is appropriate given the three overarching goals of this research. However, any future work related to the impacts of the UK energy system on global air pollution levels should consider the importance of life-cycle emissions.

This research primarily focused on the road transport sector because of both global trends and project resource considerations. With the former, the global trend toward increasing levels of urbanization, the current predominance of road transport as a local air polluter in developed countries, and the growing demand for access to transportation in developing economies make this sector of high interest (Haines *et al.*, 2009, 2014; Woodcock *et al.*, 2009; Sokhi and

Kitwiroon, 2011; Jensen *et al.*, 2013a; Pascal *et al.*, 2013). As stated by Woodcock, et. al. "the adverse health effects resulting from climate change, road-traffic crashes, physical inactivity, urban air pollution, energy insecurity, and environmental degradation are linked via their common antecedent of fossil-fuel energy use in transport" (Woodcock *et al.*, 2007).

With regard to project resource considerations, given the overall PhD timeframe for a single primary researcher – albeit with significant input and feedback from members of the research community as well as a number of collaborative side projects that provided the opportunity for intensive learning and practical experience gathering - urban transportation provides a good focus area given the plethora of data available related to urban transportation in the United Kingdom (in particular, within London) for a focused and successful Ph.D. project. Furthermore, this focus will create a strong platform for future research.

This research focused predominately on the health impacts resulting from changes in outdoor air pollution levels for particulate matter (PM_{10} and $PM_{2.5}$) as well as nitrogen oxides (NO_x). While some road transport methods (e.g. active transport with biking and walking) can have significant positive health effects that are not air quality related (e.g. increased activity levels contributing to reduced instances of obesity), these health impacts are not studied directly here (Woodcock *et al.*, 2009; Jensen *et al.*, 2013a). Rather, the focus of this research is solely on the health effects resulting from changes in outdoor air pollution levels from pollution produced within the geographical boundaries that are studied (i.e. the United Kingdom, Greater London). This is an appropriate focus for this research project given its primary goal of developing a better understanding of the co-impacts of and interactions between local air pollution and local decarbonisation efforts on public health.
There are certainly a quite large number of interesting questions to be asked that fall within the identified research gap, but exist outside of the identified scope. These questions offer valuable opportunities for later work and/or collaborative projects to be completed outside of the core Ph.D. research presented in this thesis and are part of a longer-term strategic vision for research at the intersection of energy systems and public health.

1.5 Research Questions

The key research questions that were explored in this research were as follows with regards to energy technology transitions in the United Kingdom and Greater London urban area:

- What are the co-impacts (both positive and negative) on particulate matter and nitrogen oxide air pollution levels for energy sector decarbonisation pathways that are optimised with regards to reducing total greenhouse gas emissions on both a national and urban scale?
- 2. How does considering the impact of these other types of outdoor air pollution (i.e. particulate matter and nitrogen oxides) impact the decarbonisation pathway on both a national and urban scale?

The intention of the first research question is to better understand the extent to which decarbonisation pathways could impact other types of air pollution – namely particulate matter and nitrogen oxides. These two air pollutants are of particular interest due to their prevalence in the United Kingdom and Greater London urban area, as well as their public health impact as described elsewhere in this thesis. The driver of change in research question 1 is decarbonisation, while changes in particulate matter and nitrogen oxide air pollution are viewed as co-impacts.

The aim of the second research question is to explore how the "optimum" decarbonisation pathway could change if the impacts of other types of air pollution are considered. The answers to this research question could provide insight on potential areas for tension between climate and air quality interventions. In turn, this information could be quite useful in the design of climate change and air quality policies that would be mutually supportive.

These research questions are explored in the context of a transition between present day and 2050 as this is the timeline currently used by the United Kingdom government in setting its long-term decarbonisation goals.

1.6 Thesis Overview

The remaining chapters of this thesis are structured as follows. First, a literature review provides an overview and discussion of the existing body of scientific literature as it relates to this research project (Chapter 2). This discussion is followed by a presentation of the methodology that was applied in the answering of the posed research questions (Chapter 3). The next two chapters present the results from the application of this methodology for both the United Kingdom (Chapter 4) and London (Chapter 5) regions. These results are followed by a summary of the main insights and conclusions that can be drawn from this work as well as its limitations and opportunities for future investigations and research (Chapter 6).

Chapter 2 - Literature Review

2.1 Overview

This chapter includes a discussion of the existing body of scientific literature as it relates to this Ph.D. research project and is structured into three sections. First, pertinent background information is presented on the health impacts of air pollution with emphasis being placed on energy sector pollution sources in order to reinforce the motivations behind this research project (Section 2.2). This section is followed by an overview of key existing co-impact studies, grouped by spatial focus (global, national, and urban) (Section 2.3) and a discussion of prominent models that have been used in assessing the co-impacts of climate change mitigation efforts (Section 2.4). The chapter concludes with a clear articulation of the existing research gap (Section 2.5). As discussed elsewhere in this manuscript, the research presented in this thesis partially fills a portion of this gap.

2.2 Background and Context

This section provides an overview of the health impacts of outdoor air pollution, the recommendations included in the World Health Organization's guidance for air quality, and specific information relating to the health impacts of air pollution in the United Kingdom. This section is meant as an overview to provide context for this research as opposed to presenting a detailed systematic review of the epidemiological evidence related to the health impacts of air pollution. Such reviews can be found in the studies referenced in this thesis, in particular by the World Health Organization in their recent Review of the Evidence on the Health Impacts of Air Pollution (REVIHAAP) project (World Health Organization, 2013a, 2013b).

2.2.1 World Health Organization Definitions and Guidelines for Air Pollution

Air pollution is defined by the World Health Organization as the "contamination of the indoor or outdoor environment by any chemical, physical or biological agent that modifies the natural characteristics of the atmosphere" (World Health Organization, 2007, 2013b). Current epidemiological and toxicological evidence supports the causal linkage between air pollution and public health (Beevers *et al.*, 2013; World Health Organization, 2013b; Williams *et al.*, 2014; Walton *et al.*, 2015). Furthermore, according to the World Health Organization "[air] pollutants of major public health concern include particulate matter, carbon monoxide, ozone, nitrogen dioxide and sulphur dioxide. Outdoor and indoor air pollution causes respiratory and other diseases, which can be fatal" (World Health Organization, 2007, 2013b).

Of particular relevance to this research are existing studies that document a causal relationship between traffic-generated nitrogen oxides (NO_x) and particulate matter⁶ (PM) air pollution with the onset of childhood asthma, non-asthma respiratory symptoms, impaired lung function, total and cardiovascular mortality, and cardiovascular morbidity (Hoek *et al.*, 2002, 2013; Committee on the Medical Effects of Air Pollution (COMEAP), 2006; Health Effects Institute, 2010; Ashmore *et al.*, 2011; Beevers *et al.*, 2012; World Health Organization, 2013b; Favarato *et al.*, 2014). These two types of pollutants are both prevalent in the United Kingdom and London, which are the geographic focus areas examined within this research. These pollutants are also largely produced by the energy sector, and have been linked to tens of thousands of premature deaths nationwide, including around 9,400 deaths in the Greater London area (Miller

⁶ Particulate matter (PM) air pollution is described in terms of the diameter of individual particles. Particulate matter that consists of particles of 10 micrometres or less is referred to as PM_{10} . A subset of PM_{10} includes $PM_{2.5}$, which refers to particles with a diameter of 2.5 micrometres or less.

and Hurley, 2010; Committee on the Medical Effects of Air Pollution (COMEAP), 2011; Yim and Barrett, 2012; Walton *et al.*, 2015).

2.2.1.1 Health Impacts of Particulate Matter Exposure

The negative health impacts of both long- and short-term exposure to particulate matter have been documented in the literature through studies of populations in both developed and developing countries (Pope *et al.*, 2002; World Health Organization, 2005, 2013b). This section includes discussion of four key studies that review the existing epidemiological evidence base and have been published since 2000.

In their study published in 2002, Pope, et. al. assess the relationships between long-term exposure to fine particulate matter air pollution (PM_{2.5}) and all-cause, lung cancer and cardiopulmonary mortality. In their analysis, Pope. et. al. use statistics from the American Cancer Society's Cancer Prevention II study, which began in 1982 and included 1.2 million adults in the United States (Pope *et al.*, 2002). According to the analysis produced by these researchers, the American Cancer Society's statistics show that each 10 μ g/m³ increase in PM_{2.5} concentrations in the ambient air is associated with a 4%, 6% and 8% increased risk in all-cause, cardiopulmonary, and lung cancer mortality (respectively) (Pope *et al.*, 2002).

In 2013, Hoek et. al. summarised their review of evidence from epidemiological studies published through January of that year in their paper published in the journal *Environmental Health* (Hoek *et al.*, 2013). According to their analysis, the studies that they identified in their search process support the previous associations made between long-term exposure to $PM_{2.5}$ and increased risk of all-cause and cardiovascular mortality identified in previous reviews of the epidemiological evidence. Furthermore, Hoek et. al. found a 6% overall increase in the risk of all-cause mortality for each 10 µg/m³ increase in PM_{2.5} (i.e. higher than the 4% observed by

Pope, et. al. in their 2002 review) as well as an 11% increase in risk for cardiovascular mortality.

Also in 2013, the World Health Organization published a review of the current body of evidence on the health effects of air pollution gathered in their Review of EVIdence on Health Aspects of Air Pollution (REVIHAAP) project. Overall, the authors of this publication conclude that there exists "no evidence of a safe level of exposure [to particulate matter pollution] or a threshold below which no adverse health effects occur" (World Health Organization, 2013b). Furthermore, and of particulate relevance to the portion of this research dedicated to the United Kingdom, "more than 80% of the population in the WHO European Region (including the European Union, EU) lives in cities with levels of PM exceeding WHO Air Quality Guidelines" (World Health Organization, 2013b). On average, exposure to particulate matter air pollution reduces life expectancy by almost 9 months in Europe (Pascal *et al.*, 2013).

In 2014, Atkinson, et. al. conducted a systematic review and meta-analysis of 110 peerreviewed time series studies to assess the associations between particulate matter ($PM_{2.5}$), daily mortality, and hospital admissions (Atkinson *et al.*, 2014). According to their analysis, increases in $PM_{2.5}$ exposure levels are positively associated with mortality and hospital admissions related to cardiovascular and respiratory illnesses. Across the studies, an incremental change in $PM_{2.5}$ concentrations of 10 µg/m³ was associated with a 1.04% (95% CI 0.52% to 1.56%) increase in the risk of death with hospital admissions data revealing that respiratory causes of deaths were larger than cardiovascular (1.51% vs. 0.84%). However, overall mortality rate increases varied substantially by region, from 0.25% to 2.08% indicating that caution would be wise when applying these values (Atkinson *et al.*, 2014; Walton *et al.*, 2015).

2.2.1.2 Health Impacts of Nitrogen Dioxide Exposure

Nitrogen dioxide (NO₂) has been associated with morbidity and early mortality, both from its associated toxicants and the pollutant itself (Hoek *et al.*, 2013; World Health Organization, 2013b; Favarato *et al.*, 2014). While the attributional evidence for the health effects is not currently as strong as that for particulate pollution, oxides of nitrogen have been linked to increases in total mortality from both short- and long-term exposure. In particular, NO₂ has been linked to respiratory and cardiovascular illness leading to mortality in both cases (Hoek *et al.*, 2013; U.S. Environmental Protection Agency, 2013; World Health Organization, 2013b). Long-term NO₂ exposure has also been linked with reproductive and developmental effects as well as higher instances of cancer (U.S. Environmental Protection Agency, 2013).

With regards to the transportation sector, nitrogen dioxide pollution has been identified as a potential primary indicator of other traffic-related air pollution impacts because of the uncertainty relating to co-pollutant confounding (U.S. Environmental Protection Agency, 2013). The effect refers to the extent to which observed health impacts in people with exposure to NO₂ can be attributed directly to NO₂ versus other co-pollutants (e.g. PM, volatile organic compounds, SO₂, and O₃) (Tétreault, Perron and Smargiassi, 2013; U.S. Environmental Protection Agency, 2013). The WHO has recommended that up to one-third of the long-term effects of NO₂ exposure may overlap with effects from long-term PM_{2.5} exposure based on evidence from cohort studies published to date (World Health Organization, 2013b). In addition to co-pollutants, other confounding factors (e.g. time, space, dietary habits, smoking) can distort the estimated health impacts of air pollution exposure (Jerrett *et al.*, 2009; Zanobetti and Schwartz, 2011; Zanobetti *et al.*, 2012; U.S. Environmental Protection Agency, 2013).

2.2.1.3 World Health Organization Guidelines for Air Pollution

Due to the potential for air pollutants to negatively impact human health and given their current prevalence around the globe, the World Health Organization has developed air quality guidelines for air pollutants including particulate matter (PM) and nitrogen dioxide (NO₂), with a focused group of guidelines for Europe (World Health Organization, 2000, 2005). In their subsequent 2013 review of the current scientific evidence relating to the health impacts of air pollution, the World Health Organization primarily focused on particulate matter (PM₁₀ and PM_{2.5}), nitrogen oxides (NO and NO₂), and tropospheric ozone⁷ (O₃) (World Health Organization, 2013b).

The World Health Organization air quality guidelines are based on both scientific evidence regarding indicators of health effects (e.g. physiological measures like changes in lung function, inflammation markers) and "the most critical population health indicators, such as mortality and unscheduled hospitalizations" (World Health Organization, 2005). In their review of the scientific literature, the World Health Organization found that only the complete removal of air pollution could eliminate the threat to human health (World Health Organization Media Center, 2011; World Health Organization, 2013b). Therefore, compliance with the WHO air quality guidelines does not eliminate the public health impacts of air pollution. Rather, meeting these targets will only help in reducing or eliminating the worst of the potential negative health impacts caused by these pollutants.

Generally speaking, World Health Organization guideline values represent concentrations that have been shown to be achievable in large urban areas in developed countries and also expected to significantly reduce health risks. Furthermore, the guideline values acknowledge that

⁷ A secondary pollutant formed in the presence of nitrogen oxides (NO_x), carbon monoxide (CO), and volatile organic compounds (VOCs) when they are exposed to sunlight.

"national standards will vary according to the approach adopted for balancing health risks, technological feasibility, economic considerations and various other political and social factors, which in turn will depend on, among other things, the level of development and national capability in air quality management" (World Health Organization, 2005). These guidelines are meant to provide countries with a basis for reducing the negative health impacts of air pollution. They are listed in Table 2.1 (World Health Organization, 2000, 2005).

Table 2.1: WHO Air Quality Guidelines

Air pollutant	WHO Air Quality Guidelines (2005)			
PM ₁₀	$20 \ \mu g/m^3$ annual mean, $50 \ \mu g/m^3 24$ -hour mean			
PM _{2.5} (including black carbon)	$10 \ \mu g/m^3$ annual mean, $25 \ \mu g/m^3$ 24-hour mean			
NO _X (NO and NO ₂)	NO ₂ : 40 μ g/m ³ annual mean, 200 μ g/m ³ 1-hour mean			

2.2.2 The Health Impacts of Outdoor Air Pollution in the United Kingdom

Despite significant improvements in air quality in the United Kingdom since the passing of the Clean Air Act in 1956, studies estimate that tens of thousands of people in the United Kingdom still die prematurely due to exposure to particulate matter and nitrogen dioxide air pollution. According to the Committee on the Medical Effects of Air Pollutants (COMEAP) in the United Kingdom, an estimated 29,000 premature deaths occurred in the U.K. in 2008 due to anthropogenic particulate matter pollution, the equivalent of around 340,000 years of life lost (i.e. 12 years of life per person) (Miller and Hurley, 2010; Committee on the Medical Effects of Air Pollution (COMEAP), 2011).

In 2012, Yim and Barrett published their analysis on the impacts of particulate matter ($PM_{2.5}$) air pollution from combustion processes on human health in the United Kingdom (Yim and Barrett, 2012). According to their analysis, $PM_{2.5}$ air pollution emissions from combustion cause ~19,000 premature deaths in the United Kingdom each year. Of these deaths, approximately 13,000 are caused by emissions produced in the United Kingdomwhile the

remaining ~6,000 are linked to non-U.K. European Union combustion emissions. Furthermore, the leading domestic contributor to these premature deaths is $PM_{2.5}$ air pollution produced by transport sector, which leads to around 7,500 premature deaths per year in the United Kingdom. According to Yim and Barrett, power generation and industrial emissions of $PM_{2.5}$ air pollution result in ~2,500 and ~830 early deaths per year, respectively (Yim and Barrett, 2012). Overall, $PM_{2.5}$ concentrations and their corresponding health impacts were highest in the Greater London area (Yim and Barrett, 2012). This observation makes sense given the population density of this area, which leads to higher total exposure levels.

Figure 2.1: Particulate Matter (PM_{2.5}) Concentrations across the United Kingdom from Yim and Barrett (Yim and Barrett, 2012)



Annual avg. PM_{2.5} concentrations from:

- (a) Power generation
- (b) Commercial, institutional, residential, and agricultural sources
- (c) Industry
- (d) Road transport
- (e) Other transport
- (f) All UK combustion sources

Reference: Yim, Steve H.L. and Steve R.H. Barrett. Public Health Impacts of Combustion Emissions in the United Kingdom. Environmental Science & Technology 2012, 46, 4291-4296

In May 2016, newly elected Mayor of London Sadiq Khan clearly stated his "mandate to clean up London's air – [the city's] biggest environmental challenge" (Mayor of London. London Assembly, 2016b). According to the Mayor (Mayor of London. London Assembly, 2016a): "In the past, London has only responded after an emergency, like with the Clean Air Act, which followed the Great London Smogs of the 1950s. But I want to act before an emergency, which is why we need big, bold and sometimes difficult policies if London is to match the scale of the challenge."

In his full remarks, Khan refers to scientific evidence published by Walton, et. al with Kings College London in 2015 (Mayor of London. London Assembly, 2016a). In this study, researchers estimate that around 9,400 Londoners died prematurely in 2010 due to exposure in the city to two types of air pollution (Walton *et al.*, 2015). More specifically, this figure includes the estimated health burden of human-produced particulate matter ($PM_{2.5}$) and – for the first time – nitrogen dioxide (NO_2) air pollution. These two pollutants are of particular concern in London due to their currently unhealthy (and, in the case of nitrogen dioxide, illegally high⁸) concentration levels in the city (Vidal, 2013; World Health Organization, 2016c).

According to Walton, et. al. the total mortality burden of long-term exposure to particulate matter ($PM_{2.5}$) air pollution in 2010 is estimated to be more than 52,000 life-years lost – the equivalent of around 3,500 deaths at typical ages. The health consequences of long-term exposure to nitrogen dioxide (NO_2) was estimated at significantly higher levels than $PM_{2.5}$ air pollution, with around 88,000 life-years lost (the equivalent of almost 5,900 premature deaths) in 2010. Moreover, the estimate for the health impacts of NO_2 is potentially a conservative one,

⁸ In 2015, the UK Supreme Court ruled that the government must take further action to reduce air pollution levels in order to meet European Untion Air Quality Directive limits for outdoor air pollution, with which it is currently in violation.

as it assumes a 30% overlap⁹ between the health effects of $PM_{2.5}$ and NO_2 as suggested by the World Health Organization in their 2013 Review of Evidence on Health Aspects of Air Pollution (REVIHAAP) project technical report (World Health Organization, 2013a; Walton *et al.*, 2015).

Walton, et. al. also attribute the health burden from $PM_{2.5}$ and NO_2 air pollution in 2010 to the pollution source, finding that exposure to anthropogenic $PM_{2.5}$ air pollution from:

- London road transport pollution led to 346 premature deaths (5,147 life-years lost)
- other (i.e. non-road transport) London sources led to 666 premature deaths (9,913 lifeyears lost)
- non-London pollution sources led to 2525 premature deaths (37,570 life-years lost)

Furthermore, this research estimate that the mortality burden of NO₂ from:

- London sources (road transport + other sources) led to between 3,892 and 5,302 premature deaths (58,332 to 79,441 life-years lost)
- Non-London sources led to between 1,987 and 2,707 premature deaths (29,781 to 40,558 and 29,781 life-years lost)

Both of these source apportionment breakdowns highlight the limits to local policy action, including the limits of the Mayor of London's ability to eliminate the city's air quality challenges. Indeed, according to Walton, et. al. the majority of premature deaths resulting from

⁹ This recommendation by the World Health Organization resulted from a review by Hoek. et. al. of all cohort studies published before 2013 on the long-term health effects of ambient air pollution – a total of eleven (11) studies (Hoek *et al.*, 2013; World Health Organization, 2013b).

particulate matter ($PM_{2.5}$) pollution are caused by non-London sources of air pollution, which highlights the necessity of coordinated action to address these challenges. The numbers for NO₂ pollution source apportionment are slightly more encouraging with regards to the impact of local action to reduce premature mortality due to air pollution exposure. Indeed, of the estimated range of 5,879 to 8,009 total premature deaths occurring in London each year due to NO₂ air pollution exposure, around two-thirds of them could be off-set by eliminating pollution sources that are within London.

Of note here is that the higher of the values presented by Walton, et. al. represents the total estimated attributable NO₂ mortality burden while the smaller value assumes the 30% overlap with the health effects of particulate matter ($PM_{2.5}$) previously mentioned. Furthermore, the London sources of NO₂ pollution could not be apportioned into road traffic and other sources by Walton, et. al. in order to comply with guidelines from the U.K. Department for Environment, Food and Rural Affairs (Defra) (Walton *et al.*, 2015).

This 2015 report by Walton et. al. represents the first publication of estimates assessing the mortality burden of nitrogen dioxide (NO₂) air pollution in London using the findings and recommendations from the World Health Organization's REVIHAAP and Health Risks of Air Pollution in Europe (HRAPIE) projects (World Health Organization, 2013a, 2013b; Walton *et al.*, 2015). It also presents updated estimates of the health burden of particulate matter (PM_{2.5}) in London, which had been previously evaluated by Public Health England and the Institute of Occupational Medicine predominately using methods recommended by the Committee on the Medical Effects of Air Pollution (COMEAP) (Committee on the Medical Effects of Air Pollution (COMEAP), 2010, 2011; Miller, 2010; Miller and Hurley, 2010; Gowers, Miller and Stedman, 2014). Another study by Yim and Barrett estimates the mortality burden of air

pollution specifically from combustion processes (Yim and Barrett, 2012). Key differences between the estimates presented by Walton, et. al. and its predecessors include refinements to the previously used methodology as well as somewhat more significant updates to input data. Walton et. al. also accounted for air pollution produced by human activities (i.e. anthropogenic) as opposed to total PM_{2.5} and/or combustion-specific emission sources as was done by Miller, et. al. in their 2010 study (Miller, 2010; Walton *et al.*, 2015). A comparison of the results from Walton, et. al. and these preceding studies is shown in Table 2.2. As one can clearly see in this table, the estimated mortality burden in London of exposure to particulate matter (PM_{2.5}) is similar between all four of these studies though there does exist a range due in part to the fact that these studies analysed different years in addition to the previously differences in the applied methodologies discussed elsewhere in this section.

Table 2.2: London Mortality Burden Estimates Resulting from Long-Term Air Pollution Exposure (Miller, 2010; Yim and Barrett, 2012; Gowers, Miller and Stedman, 2014; Walton *et al.*, 2015)

Study	Estimated Mortality	Difference compared	Estimated
	Burden in London –	to Walton, et. al.	Mortality Burden
	PM _{2.5}	(2015)	in London (2010)
			$-NO_2$
Kings College London	3,537 premature deaths		5,879 premature
(Walton, et. al. 2015)	(53,630 life-years lost)		deaths (88,113 life-
	in 2010		years lost)
Public Health England	3,389 premature deaths	Premature deaths:	n/a
(Gowers, et. al 2014)	(41,404 life-years lost)	-148 (4%)	
	in 2010	Life-years lost:	
		12,226 (22%)	
Massachusetts	~3,200 air quality-	Premature deaths:	n/a
Institute of Technology	related deaths per year	-337 (9.5%)	
(Yim & Barrett, 2012)	(based on 2007 data)		
Institute of	4,267 premature deaths	Not calculated due to	n/a
Occupational Medicine	in 2008	the difference in base	
(Miller, et. al 2010)		years	

In the study published by Public Health England in 2014, Gowers, et. al. present mortality risk increases associated with long-term exposure to particulate air pollution ($PM_{2.5}$). In their work, the authors of this study model the health impacts of particulate matter ($PM_{2.5}$) using annual

average concentrations of this pollutant resulting from human activities with spatial resolution down to the local authority level. Furthermore, the authors used central estimates of the mortality burden that could be attributed to long-term $PM_{2.5}$ pollution. Overall, Gowers et. al. estimate that $PM_{2.5}$ air pollution resulted in 3,389 premature deaths (41,404 life-years lost) in 2010 in Greater London. However, the authors highlight that, due to uncertainties in the mortality risk associated with outdoor $PM_{2.5}$, the actual health burdens "could range from approximately one-sixth to about double" of these figures (Gowers, Miller and Stedman, 2014).

In the 2010 study authored by Brian G. Miller at the Institute of Occupational Medicine, the author calculates the health burden of air pollution in London using the relationships between air pollution concentration and mortality rates as recommended by the Committee on the Medical Effects of Air Pollution (COMEAP) in their national-level study (Committee on the Medical Effects of Air Pollution (COMEAP), 2010; Miller, 2010; Miller and Hurley, 2010). In turn, Miller calculates the mortality impacts for Greater London with spatial resolution down to the ward level for 2008. Overall, Miller estimates that PM_{2.5} air pollution led to the mortality equivalent of 4,267 deaths in Greater London in 2008 with a range of 756 to 7,965 when uncertainties in the direct health impacts of PM_{2.5} are included. Miller also estimates the potential health impacts that would result from a permanent reduction in PM2.5 concentrations by 1 μ g/m³. Miller's calculations show that this decrease in air pollution concentrations would lead an overall gain of 400,000 years of life for the current population – an average of 3 weeks per person (Miller, 2010). Noted here is that the air pollution concentrations used by Miller in his modelling work for this study include both human-produced (i.e. anthropogenic) pollution and that coming from natural sources. The latter would be more difficult to significantly reduce than the pollution resulting from identified human activities.

London is not alone in its air pollution challenges and resulting public health burdens (Jack and Kinney, 2010; Kan *et al.*, 2010; Pascal *et al.*, 2013). According to the World Health Organization, "more than 80% of people living in urban areas that monitor air pollution are exposed to air quality levels that exceed the World Health Organization (WHO) limits" including 98% of those living in cities in low- and middle-income countries (World Health Organization, 2016a, 2016b). Furthermore, according to the 2016 release of the World Health Organization's urban air quality database, global urban air pollution levels increased by 8% from 2008-2013 despite improvements in urban air quality in some regions (World Health Organization, 2016a, 2016c).





Sources: WHO (2016) Global Urban Ambient Air Pollution Database; Demographia (2015) for population; country groups per income based on World Bank (2016).

While low-income cities are the most impacted by these air pollution challenges, all regions of the world are currently affected (World Health Organization, 2016c). Globally, air pollution is

¹⁰ graphic by the International Energy Agency (International Energy Agency (IEA), 2016)

the fourth greatest overall risk to human health with around 7 million premature deaths, including both indoor and outdoor air pollution (World Health Organization, 2014) Of these premature deaths, 3.7 million are linked to outdoor air pollution while the remaining 3.3 million result from outdoor air pollution exposure (World Health Organization, 2014a).

In the same assessment, the World Health Organisation published a breakdown of premature deaths attributed to specific diseases (World Health Organization, 2014a). These data underlining that the vast majority of premature deaths resulting from outdoor air pollution exposure are due to the following diseases:

- 40% ischaemic heart disease
- 40% stroke
- 11% chronic obstructive pulmonary disease (COPD)
- 6% lung cancer
- 3% acute lower respiratory infections in children

Furthermore, the vast majority of premature deaths resulting from indoor air pollution exposure were attributed to the following diseases:

- 34% stroke
- 26% ischaemic heart disease
- 22% chronic obstructive pulmonary disease (COPD)
- 12% acute lower respiratory infections in children
- 6% lung cancer

As one can see in these data, a set of five diseases lead to the vast majority of all premature deaths resulting from air pollution exposure. However, indoor air pollution exposure leads to a higher percentage of deaths due to chronic obstructive pulmonary disorder disease (COPD) and acute lower respiratory infections in children. Outdoor air pollution exposure lead to a higher percentage of premature deaths due to ischaemic heart disease and strokes than for indoor air pollution exposure.

The World Health Organization's estimates were based on mortality data from 2012 and the evidence base linking health outcomes with air pollution exposures as it was available prior to their estimates as published in March 2014 (World Health Organization, 2014a). Exposure level estimates for outdoor air pollution exposure incorporated satellite data, ground-level monitoring measurements and data on pollution emissions from key sources, as well as modelling of how pollution drifts in the air.

2.2.2.1 Public Health Impact Calculations

Mortality and morbidity are two key indicators in measuring the overall effects of air pollution on public health (Miller and Hurley, 2003; World Health Organization, 2014b). Furthermore, changes in these indicators resulting from shifts in air quality can be calculated using an impact pathway or a more generalized damage function approach (Department for Environment Food and Rural Affairs (DEFRA), 2013d). Both of these methodologies have been widely used in the literature and are discussed in more detail here (Miller and Hurley, 2003; Department for Environment Food and Rural Affairs (DEFRA), 2011; Walton *et al.*, 2015). Also discussed is the life tables approach, which is related to the impact pathway methodology.

2.2.2.1.1 Impact Pathway Approach

The impact pathway approach (IPA) methodology traces the sources of air pollutants and their movement through the location/population that it impacts, effectively mapping the cause and effect of air pollutants on mortality (Department for Environment Food and Rural Affairs (DEFRA), 2013d). The impact pathway approach methodology is currently used by the United Kingdom Department for Environment, Food and Rural Affairs (DEFRA) to place a value on changes in air quality in the United Kingdom to measure impacts across four categories: health, amenity, productivity and ecosystem impacts (Department for Environment Food and Rural Affairs (DEFRA), 2013d).

In the implementation of the impact pathway approach, one begins with quantifying baseline emissions and the likely emissions levels under a proposed scenario. Subsequently, air pollution dispersion modelling is used to convert these emissions levels into population-weighted concentrations. These values are then used to quantify exposure levels and then health and non-health impacts are calculated. When sufficient data exist, these values are monetized in order to measure the economic impacts (Department for Environment Food and Rural Affairs (DEFRA), 2013d).

Challenges with the impact pathway approach include large data requirements and significant computational intensity. Furthermore, uncertainties exist throughout the process, including those related to emissions and the dispersion of air pollution as well as those uncertainties related to health impact and valuation estimations for changes in public health outcomes (Department for Environment Food and Rural Affairs (DEFRA), 2013d; Walton *et al.*, 2015). These uncertainties can be minimised through detailed air quality and exposure modelling (Department for Environment Food and Rural Affairs (DEFRA), 2013d). A related project

focusing on this type of detailed air quality modelling is being undertaken by the author of this thesis in partnership with researchers at Kings College London (Williams *et al.*, 2016).

For both the impact pathway approach and damage cost approach, the timeframe considered is of prime importance. Many health benefits, including the development of cancer and cardiovascular illnesses, present with a significant time lag to the environmental change of interest. As this lag period can be a decade or more in many cases, care must be taken to prevent understating of public health benefits. Conversely, given the multiple factors that influence public health and uncertainty about future air pollution scenarios, consideration should also be given to prevent the overstating of any benefits (Jarrett *et al.*, 2012; Jensen *et al.*, 2013b).

2.2.2.1.1 Damage Cost Approach

An alternative to the impact pathway approach relies on damage costs. This approach uses the results of impact pathway analyses, but is less resource intensive.

The damage function approach estimates the direct monetary impacts of changes in air pollution levels. First, one estimates a coefficient to represent the physical damage of changes in air quality on non-fatal human health effects (morbidity) or fatal effects (mortality) based on existing epidemiological evidence in the scientific literature. Second, the quantity of the health effects resulting from changes in air quality are estimated by examining the population that are exposed to these changes in air quality (i.e. exposure levels) as well as the environmental change itself. This calculation therefore takes into account how many individuals are exposed to a change in air quality. Third, the number of health effects are multiplied by the associated costs, potentially including lost wages, medical treatment expenses, and other costs of interest (e.g. non-financial welfare costs) depending on the purpose behind the calculation (U.S. Environmental Protection Agency National Center for Environmental Economics, 2014a). The results are then overlaid with the associated change in air pollution levels, resulting in damage cost values (e.g. in units of \$ per tonne of pollution emitted).

This approach to estimating health benefits presents two challenges in valuing morbidity decreases. First, the potential impacts of behavioural changes are not included (e.g. individuals choosing to stay indoors on polluted days or buying and using an air filter in one's home to reduce exposure levels or walking a different route to avoid a pollution hot spot that has developed) nor is willingness to pay. This limitation to the damage cost approach is also present in mortality estimates. Second, as morbidity effects are measured in terms of directly avoided costs, the amount that individuals would be willing to pay to avoid illnesses is not accounted for. Consequently, the total estimated benefits could be incomplete and therefore not representative of the total benefit that could be realised (U.S. Environmental Protection Agency National Center for Environmental Economics, 2014a, 2014b).

2.2.2.1.3 Life Tables

A life table is a technique that is frequently used to summarise mortality patterns across populations (Miller and Hurley, 2010; Department for Environment Food and Rural Affairs (DEFRA), 2013d). Life tables frequently compute survival rates for different age groups, either from birth or from the previous year of life. In turn, average life expectancies are calculated using age-specific death rates to provide additional insights (Miller and Hurley, 2010).

In order to complete a life-table mortality impact estimation, one uses age-specific all-cause mortality rates to calculate survival curves, which plot the number of survivors by age over time. Changes in air pollution levels are then converted to changes in hazard rates and applied to each age group. Changes in death rates and the corresponding life-years lost are then summed over the combination of age groups and timeframe desired for the analysis (Miller and Hurley, 2010).

The World Health Organization began producing annual life tables for all of its Member States in 1999. In 2009, these tables shifted from an annual to a biannual publication schedule. These tables are used in all of the World Health Organization's calculations of all-cause and causespecific mortality. Civil registration and vital statistics information (e.g. records of births and deaths) is the primary data resource used for the production of these life tables. Additional information is gathered from the United Nations Population Division (World Health Organization, 2014b).

2.3 Overview of existing studies

This section provides an overview of the process that was undertaken to identify existing studies that 1) focused on the co-impacts of decarbonisation of energy systems on air quality and vice versa and 2) were of interest to the research project presented in this thesis. This process overview is followed by discussion of twelve (12) key studies that were identified in this process. These studies are grouped by geographical focus (global, national, urban).

2.3.1 Literature Identification Process

The construction of this section of the literature review consisted of three main steps:

- 1. a formal search process using keywords
- 2. discussions with experts in this research field
- 3. reading and analysis of identified studies

The formal search process was conducted using the University College London's online library and Elsevier Science Direct search engines. A series of keywords were identified using literature that has been previously discovered in the development of the initial research concept. The keywords that were used in this search included:

- 1. air pollution
- 2. outdoor air quality
- 3. health
- 4. energy
- 5. transportation
- 6. climate change

In the search process, these identified keywords could be found in any portion of a journal publication, including subject, keywords, author, title, or the article text itself. The potential pool of peer-reviewed literature was narrowed by requiring that the publication include at least three of these keywords, or related permutations (e.g. "air pollution" and "air pollutants"). The abstracts of the publications identified in these search processes were then reviewed to determine the paper's potential relevance to this research project. Tables 2.3 and 2.4 present the results of this formal search process.

Primary keyword		Secondary keyword		Tertiary keyword	
Air pollution	423,211	Outdoor Air Quality	1,930	Health	580
Air pollution	423,211	Outdoor Air Quality	1,930	Energy	114
Outdoor Air Quality	3,107	Health	882	Energy	109
Air Pollution	423,211	Energy	2,885	Health	246
Health	1,615,760	Energy	68,516	Transportation	1,407
Energy	2,457,838	Climate Change	30,090	Health	1,422

Table 2.3: Number of Journal Articles Identified Using UCL Library Search Engine for Journal Articles Only in April 2014

For the search using Elsevier Science Direct, keywords were identified and applied to a search through the entire paper including title, author, keywords, and the article text. The results of this search are displayed in Table 2.4 The first column represents the number of articles containing the first keyword only. The second column shows the number of articles containing both the first and second keywords. The third column displays the number of articles that both contained the first and second keyword and also had been classified as being part of the listed topic. Of note here is that topics related to the previously identified keywords were also reviewed (e.g. "public health" and "health" were both viewed as acceptable topics for "health"). A zero in the far-right column indicates that no published journal articles were found as being listed under the indicated topic (or reasonably related topics) for the indicated combination of keywords.

Elsevier Science Direct – journal articles only					
First Keyword		Second Keyword		Торіс	
Air pollution	210,117	Health	101,464	Climate Change	624
Air pollution	210,117	Outdoor Air Quality	11,816	Climate Change	77
Outdoor Air Quality	25,857	Health	15,439	Climate Change	85
Health	2,647,815	Energy	470,382	Climate Change	1,187
Energy	3,039,001	Climate Change	161,259	Health	0
Energy	3,039,001	Health	470,382	Air pollution	0
Health	2,647,815	Climate Change	106,063	Air Quality	412
Transportation	281,424	Health	92,195	Air pollution	305
Transportation	281,424	Air pollution	32,438	Climate change	279

Table 2.4: Number of Journal Articles Identified Using Elsevier Science Direct Search Engine in April 2014

These search engines certainly do not represent a comprehensive compilation of all existing publications and do not include publications that are in progress or under review. Nor should the keywords used in this search process be considered as the only applicable search terms. However, this process still provided a first approach for systematically identifying literature of potential interest and narrowing the field of potential literature to be reviewed further.

This formal search process was complemented by discussions with researchers in the field. In this process, experts were consulted in London, the United Kingdom and the United States and asked to suggest additional literature of potential interest. As with the previous step in this process, the abstracts of the publications recommended by experts in the field were reviewed to determine the paper's potential relevance to this research project. Unsurprisingly, there was a significantly higher rate of success – defined in terms of the number of directly relevant papers discovered compared to the total number of papers initially identified - in this portion of the literature review process compared to the broader use of search engines.

In the third step in the literature review process, identified papers of potentially high interest were read in full. Those that were found to be of particularly high interest and applicability were then grouped according to geographic focus (global, national, local) for presentation in this thesis in order to provide an overview of the existing literature and how it relates to the research presented here. Publications were also monitored on an on-going basis to ensure that the most recent literature was captured.

2.3.2 Key Related Studies

The previously discussed process of identifying key related publications resulted in the identification of a group of particularly high interest publications, including ten (10) key related studies in the peer-reviewed literature in the last decade. An additional two (2) consulting reports by Pye, et. al. were also identified, due to their particular relevance to this work given that they incorporated other air pollutants into the United Kingdom MARKet ALlocation (MARKAL) model, which is the predecessor to the UK TIMES Model (UKTM-UCL) used in the research presented in this thesis. An overview of these twelve (12) studies is included in Table 2.5.

Study	Spatial focus	Sector/ Scenarios	Energy systems model?	Multiple scenarios included?	Co-impacts considered in optimisation?
Intergovernmental Panel on Climate Change (2014)	Global	Energy Sector	Yes	Yes	No
International Energy Agency (2016)	Global	Clean Air Scenario	Yes	Yes	Yes/No ¹¹
Anenberg, et. al.	Global	Climate change	No	Yes	No
(2012)	(focus on Africa, Asia)	mitigation via methane and black carbon emissions controls			
Dessens, et. al. (2014)	Global	Transport – shipping and aviation	No	No	Yes/No
Barker, et. al	Global	Rapid global	Yes/No	Yes	Yes
(2010)	(focus on Mexico - with urban conclusions)	decarbonisation			
Jensen, et. al. (2013)	National (UK)	Healthy diet, active travel, household energy efficiency, cleaner cars	No	No	Yes
Pye, et. al (2008a & 2008b)	National (UK)	Low GHG emissions, BAU – accounting for SO ₂ , PM_{10} , NO_x	Yes	No	Yes
Wadud and Waitz (2011)	National (United States)	Transport (road, ocean, rail, aviation)	No	No	Yes
U.S. EPA (2009)	National (United States)	Impact of climate change on O ₃	No	Yes	Yes
Jack and Kinney (2010)	Urban	A range of policy scenarios – a review paper of existing literature	No	No	No
Woodcock, et. al. (2009)	Urban (London, Delhi) with other	BAU and low-GHG using WHO comparative risk assessment	No	Indirectly	No
Jarrett, et. al. (2012)	Urban (England and Wales)	Travel mode shifting (active)	No	No	No

Table 2.5: Comparison of Key Related Studies of Interest

¹¹ The "Clean Air Scenario" published in this report includes the "implementation of additional measures intended to achieves a significant reduction in air pollutant emissions" including energy efficiency, targeted actions to reduce in coal-fired power plant use with a complete ban of new coal power plant construction, emission limits for all combustion power plants, higher vehicle emission standards, increasing renewable energy investment and the phasing out fossil fuel subsidies (International Energy Agency (IEA), 2016).

As discussed in further detail in the following sections, each of these studies included the use models to understand aspects of the co-impacts of climate change mitigation on air pollution and/or vice versa. It is noted that the global study by the International Energy Agency and the reports by Pye, et. al. were of particular importance in this research project (Pye and Palmer, 2008; Pye *et al.*, 2008; International Energy Agency (IEA), 2016). The former utilised the International Energy Agency's World Energy Model (WEM) in conjunction with the Greenhouse Gas and Air Pollution Interactions and Synergies (GAINS) model from the International Institute for Applied System Analysis (IIASA) and represents the current state of the art in joint modelling on the co-impacts of energy system transitions on air pollution. The latter included the most advanced work done in the United Kingdom to analyse the co-impacts of changes to the energy system on air pollution levels using an energy systems model that is at the core of UK government decision making.

2.3.3 Global Studies

This section contains discussion of four key global scale studies related to this research that were identified as previously discussed in this thesis. The first, by the Intergovernmental Panel on Climate Change, can be considered the landmark review of this area and pays particular attention to the potential co-impacts of energy system transitions in its Chapter 7 (Bruckner *et al.*, 2014). The second, by the International Energy Agency, uses their World Energy Model in conjunction with the GAINS model from the International Institute for Applied System Analysis (IIASA) to construct a Clean Air Scenario for the global energy system and represents arguably the most closely related and state-of-the-art in joint modelling in this area. The third, Anenburg, et. al. examines the impacts on air quality and health of a group of specific black carbon and methane emission control measures that are expected to have climate benefits, showing an alternative point of view of climate change mitigation as a co-benefit of air

pollution controls (as opposed to vice versa). The fourth, Dessens, et. al. focuses on the potential impacts of an explicit global greenhouse emissions trading scheme as a method for reducing both greenhouse gases and other atmospheric emissions that lead to air pollution and provides insights drawn from the examination of a specific scheme for reducing emissions rather than policy-agnostic approach. Their focus is on international transport, including both air and shipping.

2.3.3.1 Intergovernmental Panel on Climate Change (2014)

In Section 7.9 of the Fifth Assessment Report by the Intergovernmental Panel on Climate Change (IPCC), its authors state that (Bruckner *et al.*, 2014):

"Besides economic cost aspects, the final deployment of [climate change] mitigation measures will depend on a variety of additional factors, including synergies and tradeoffs across mitigation and other policy objectives. The implementation of mitigation policies and measures can have positive or negative effects on these other objectives – and vice versa. To the extent these side-effects are positive, they can be deemed 'co-benefits'; if adverse and uncertain, they imply risks."

In their analysis, the authors focus on the co-impacts of a set of mitigation measures in the energy supply sector. Most relevant to the research presented in this thesis is their consideration of the replacement of coal with nuclear for power generation as well as the increased use of renewable energy resources (e.g. solar, wind, geothermal, hydro) and the corresponding potential impact on air pollution. In their report, the authors highlight that (Bruckner *et al.*, 2014):

"To avoid creating new environmental and health problems, assessments of mitigation technologies need to address a wide range of issues, such as land and water use, as well as air, water, and soil pollution, which are often location-specific."

In the Fifth Assessment Report, the authors highlight the fact that the "stabilization of GHG concentrations [will require] fundamental changes in the global energy system relative to a baseline scenario" and illustrate the potential pathways for this transition using three models. These models include MESSAGE, REMIND and GCAM which are applied in order to explore the changes to the global primary energy supply that would be required to stabilise global CO₂-equivalent emissions (Bruckner *et al.*, 2014). Of these tools, the MESSAGE model is the most closely related integrated assessment model to the TIMES model that is used to support energy policy development in the United Kingdom as described elsewhere in this thesis. Furthermore, it is utilised by the International Institute of Applied Systems Analysis (IIASA), which also houses the Greenhouse Gas and Air Pollution Interactions and Synergies (GAINS) model that is perhaps the most advanced tool for understanding the interactions between greenhouse gas and air pollution mitigation options (International Institute for Applied Systems Analysis (IIASA), 2016). The GAINS model was used in a 2016 report by the International Energy Agency that is described in more detail elsewhere in this chapter (International Energy Agency (IEA), 2016).

Each of the scenarios presented in the IPCC report include changes in the primary energy supply that could have significant impacts on air pollution and public health. However, none of the models are applied to directly quantify the co-impact of these changes to the energy system on air pollution levels around the globe (Bruckner *et al.*, 2014). For example, each model shows increasing use of renewable energy technologies including solar and wind as well

as higher levels of energy efficiency that might result in significant decreased levels of air pollution emissions compared to a baseline scenario. At the same time, these models show increasing use of coal, natural gas oil, and biomass in systems that include carbon capture and storage, which could have mixed implications for air pollution and its corresponding impact on human health.

2.3.3.2 International Energy Agency (2016)

In 2016, the International Energy Agency (IEA) published a World Energy Outlook special report titled "Energy and Air Pollution" dedicated to the connections between energy, air pollution and health. In this report, scenarios are constructed in this report using the IEA's World Energy Model (WEM) in conjunction with the Greenhouse Gas and Air Pollution Interactions and Synergies (GAINS¹²) model from the International Institute for Applied System Analysis (IIASA). The former produces projections of energy-related greenhouse gas emissions but "does not, in isolation, generate projections for energy-related air pollution" according to the IEA (International Energy Agency (IEA), 2015b, 2016). The GAINS model has been used to estimate historic air pollution emissions by country and was used in this application to project future emission levels, its effects on ambient air quality, and the resulting impacts on human health and ecosystems (Amann *et al.*, 2009; International Institute for Applied Systems Analysis (IIASA), 2009, 2016; Kiesewetter *et al.*, 2014; International Energy Agency (IEA), 2016).

¹² http://www.iiasa.ac.at/web/home/research/researchPrograms/air/GAINS.en.html

In this report, the IEA presents two scenarios (International Energy Agency (IEA), 2016):

- New Policies includes the energy-related components of the Intended Nationally Determined Contributions (INDCs) pledged at the COP21 meeting in Paris
- Clean Air includes "proven energy policies and technologies" that are "tailored to national circumstances" in order to achieve significant additional reductions in other air pollution emissions.

This work by the IEA is the first of its kind to analyse scenarios for achieving a set of climate change mitigation targets as set under the Paris Agreement in the Intended Nationally Determined Contributions (INDCs) both with and without the integration of air pollution mitigation policies and technologies. In turn, it currently sets the standard for global analysis of the linkages between changes in the energy systems and its resulting co-impacts on air pollution and public health around the globe with country-level spatial resolution through the year 2040.

Beyond the differences in the time horizon (2040 in the IEA report versus 2050 in the research presented in this thesis) and geographic focus (global, with some regional and country-level work versus a country and urban geographic focus) that were examined, the work by the IEA differs from the work presented in this research in that, broadly speaking, the WEM is designed to project future pathways based on current trajectories in the energy system brought forward while incorporating the expected impact of specific policies like the INDCs.

Conversely, tools like the TIMES model used in this research are designed to create potential energy system development pathways to achieving a defined set of future goals (e.g. climate change mitigation) at the least cost. The reasons for using the TIMES model in this research

are discussed in Chapter 3. Additional discussion on the GAINS model, focuses on the implementation of specific approaches to air pollution abatement, is included elsewhere in this chapter.

2.3.3.3 Anenberg, et. al. (2012)

In 2012 an international group of researchers from the United States, United Kingdom, Austria, and Kenya published findings from their investigation of the global air quality and health cobenefits of mitigating near-term climate change through methane and black carbon emission controls (Anenberg *et al.*, 2012). The objective of this study was to examine the air quality and health benefits of 14 specific air pollution emission control measures that targeted black carbon and methane. In this work, Anenberg *et.* al. explore a relevant related research thread on the potential for targeting action to reduce air pollution to result in co-benefits in the form of climate change mitigation.

The measures considered in this study included technical measures that would target methane or black carbon emissions as well as non-technical measures for reducing black carbon and methane emissions (Anenberg *et al.*, 2012). These specific control measures "were selected because of their potential to reduce the rate of climate change over the next 20-40 years" in addition to reducing black carbon and methane emissions (Anenberg *et al.*, 2012). The latter is a precursor to tropospheric ozone (O_3) formation.

This study used global composition-climate models (GISS-PUCCINI and ECHAM-HAMMOZ) for their analysis (Anenberg *et al.*, 2012). The National Aeronautics and Space Administration (NASA) Goddard Institute for Space Studies (GISS) model for Physical

Understanding of Composition-Climate Interactions and Impacts (GISS-PUCCINI¹³) is designed to simulate the Earth's climate system with a major focus on studying human impacts on climate. EECHAM-HAMMOZ¹⁴ is an atmospheric general circulation model that was developed at the Max Planck Institute for Meteorology.

Critically, this work included mitigation control measures for portions of the energy system, including some specific to transport that are particularly interesting with regards to the transport-focused analysis presented in Chapter 5 of this thesis. It also provided significant insights on the impacts of air pollution mitigation on health and the uncertainty in the concentration response functions currently used to estimate human health impacts. These uncertainties are highlighted in Chapter 5 and 6 in discussions on the health impacts of changes to the London transport sector. Unlike this research, it did not calculate the expected carbon emissions impacts of these control measures nor holistically consider the entire transport or energy system.

2.3.3.4 Dessens, et. al (2014)

Whereas Anenberg, et. al. examined the potential air pollution control measures that would likely also benefit climate, Dessens, et. al. examined the potential impacts of one specific method for reducing both greenhouse gases and other atmospheric emissions that lead to air pollution, namely an explicit global emissions trading scheme (GETS). Their focus was on the co-benefits of climate change mitigation in international shipping and aviation on air pollution and radiative forcing.

¹³ http://www.giss.nasa.gov/projects/gcm/

¹⁴ http://www.mpimet.mpg.de/en/science/models/echam/

This research utilised the E3MG global energy-environment-economy model and the p-TOMCAT atmospheric model to evaluate changes in global CO_2 , NO_x , SO_2 , VOC, CH_4 and CO emissions over the period from 2000 to 2050 (Dessens *et al.*, 2014). The p-TOMCAT model version used in this research was the same as that used in the QUANTIFY project, a joint research initiative led by P. Hoor at the Max Planck Institute in Germany with researchers from the Netherlands, United Kingdom, USA, Norway, Germany, France, Italy, and Switzerland to quantify the impact of emissions by road, aircraft and ship traffic air pollution concentrations in the air, including ozone (Hoor and Borken-Kleefeld, 2009)

This study revealed significant insights on the potential impacts of a theoretical global emissions trading scheme with a specific focus on international transport (shipping and air) and explores the linkages between changes in transport demand and final air pollution emissions levels, which is particularly relevant to the analysis presented in Chapter 5 of this thesis. It did not evaluate each impact on a national or urban scale, though the former could have been presented using these tools and a related study for Mexico is presented elsewhere in this chapter.

2.3.4 National Studies

This section discusses four key national-level studies that informed the work presented in this thesis. The first, Barker, et. al., investigated the potential impacts of climate change mitigation on air quality in Mexico under two scenarios - one with national decarbonisation goals and the other with global targets to 2050 (Barker *et al.*, 2010). The second, Jensen, et. al. focused on the importance of health co-benefits in macroeconomic assessments of climate change policy impacts using a single-country computational general equilibrium (CGE) model (Jensen *et al.*,

2013a). The third, by Wadud and Waitz further investigated these types of health co-impacts through a review of the literature related to the health impacts of the transport sector air pollution in the United States (Wadud and Waitz, 2011). The fourth, by the U.S. Environmental Protection Agency looks at the impacts of climate change on air quality in the United States, which is an interesting related topic to that explored in this research (U.S. Environmental Protection Agency, 2009).

2.3.4.1 Barker, et. al (2010)

In 2010, Barker et. al. published the results of their global air pollution analysis to improve our understanding of the impact of climate change mitigation on air quality in Mexico under two scenarios – the first with Mexico alone reducing CO₂ emissions by 77% and the other with 80% reductions globally by 2050. As with Dessens, et. al in their 2014 global analysis, these researchers utilized a one-way coupling of the global energy-economy-environment model (E3MG) and the p-TOMCAT global atmospheric chemistry model that has been used in other prominent research collaborations and can be considered a leading tool in global atmospheric chemistry modeling (Barker *et al.*, 2010; Dessens *et al.*, 2014).

Overall, Barker, et. al. show that "substantial investment in low-carbon technologies, such as electric vehicles, heat pumps and geo-thermal power" could leads to many co-benefits including bringing concentrations of tropospheric ozone "close to the WHO guideline levels" (Barker *et al.*, 2010). Furthermore, in their analysis, these researchers saw air pollution concentrations for SO_2 , NO_x , CO, and volatile organic compounds decrease significantly across both scenarios, confirming the already recognized potential for climate change mitigation to have co-benefits for other types of air pollution (Ekins, 1996; Barker *et al.*, 2010).
The work is particularly interesting for the research presented in this thesis in that it examined the potential impacts for mitigation efforts not only on a national level but also specifically in the Mexico City urban area. According to the authors, the existing literature on the specific effects of climate policy on air pollution is "very limited" (Barker *et al.*, 2010). In turn, they used assumptions drawn from studies for Santiago and New York as well as evidence from the PROAIRE programme in Mexico City, which included 22 measures to improve air quality in the metropolitan areas (Barker *et al.*, 2010). Overall, these researchers concluded that "climate control in the form of rapid decarbonisation of the Mexican economy will have substantial effects on air pollution, at no extra cost, especially if the mitigation actions are focused on Mexico City" by evaluating the co-impacts on both greenhouse gas emissions and tropospheric ozone (O₃) (Barker *et al.*, 2010).

The work by Barker, et. al. differs from the work presented in this research in terms of geographic focus, the models utilized, the focus on tropospheric ozone, and the types of insights that can be drawn from the outcomes – in particular, insights on the possible technology transition pathway for achieving the emissions reductions evaluated. However, it still provides significant insights for comparison to outputs from the research project presented in this thesis.

2.3.4.2 Jensen, et. al. (2013)

In 2013, Jensen, et. al. published the results of investigation into the importance of health cobenefits in macroeconomic assessments of greenhouse gas emission reduction strategies in the United Kingdom (Jensen *et al.*, 2013a). This research provides valuable insights related to the relative importance of co-impacts in the UK's climate change mitigation efforts. Of particularl note is that this study included not just the economic co-benefits of these efforts resulting from improved public, but also the negative economic impacts associated with extended lifetimes that are not included in other studies. For the latter, these include the additional cost associated with social security payments for longer-living populations (Jensen *et al.*, 2013a).

Jensen, et. al. framed their research within the observation "that UK policy makers will, most likely, have to adopt elements which involve initial net societal costs in order to achieve future emission targets and longer-term benefits from GHG reductions." Furthermore, "cost-effectiveness of GHG strategies is likely to require technological mitigation interventions and/or demand-constraining interventions with important health co-benefits and other efficiency-enhancing policies that promote internalization of externalities" (Jensen *et al.*, 2013a). These observations articulate a key component in the justification for undertaking this research project. More specifically, the relative importance of co-benefits in supporting climate change mitigation efforts.

In this research by Jensen, et. al., the researchers focused on health co-benefits using a singlecountry computable general equilibrium (CGE) model across four strategies (healthy diet, active travel, household energy efficiency, and cleaner cars). Overall, they found that a strategy including both active travel and cleaner vehicles could be a cost-effective strategy due to its impact on illness related to low activity levels and obesity in the UK, which are important to the discussions presented in this thesis in Chapters 5 and 6. According to these researchers, their results "suggest the need for adopting holistic assessment methodologies which give proper consideration to welfare-improving health co-benefits with potentially negative economic repercussions (such as increased longevity)" (Jensen *et al.*, 2013a). For example, increased longevity can result in a higher cost for social programs such as retirement pensions and social security. Similar to this study are several other publications related to macroeconomic assessments looking at the health co-benefits of reduced air pollution (Garbaccio, Ho and Jorgenson, 2000; Ho, Jorgenson and Di, 2002; Dessus and O'Connor, 2003; Jensen *et al.*, 2013a). For example, Garbaccio, et. al. investigated the co-impacts of a carbon tax policy on particulate matter and sulphur dioxide emissions in China. Their work focused on illustrating a process for evaluating these health co-benefits, using an economy-energy-health modelling framework with rudimentary air quality modelling efforts and generalizations across sectors (Garbaccio, Ho and Jorgenson, 2000).

Ho, et. al. also looked at air pollution in China, examining some pollution control policies and how they might impact economic performance. They placed emphasis on economy-wide policies (e.g. fuel taxes) and examined how these taxes impact fuel use and, in turn, affect air pollution levels and public health damage. They also estimated how these policies could impact economic growth in China over time (Ho, Jorgenson and Di, 2002).

Dessus and O'Connor examined the health co-benefits of climate change policy in Chile using an economy-wide CGE model, focusing on identifying policies that would result in no net loss in welfare. They concluded that direct tax on particulate matter would be a more efficient approach to decreasing the negative health impacts of air pollution. However, they also found that some carbon reduction policies could be economically justified using ancillary health benefits (Dessus and O'Connor, 2003).

2.3.4.3 Pye, et. al (2008a and 2008b)

Pye, et. al. authored a series of two consultancy reports in 2008 that summarised their work to quantify changes in air quality pollutant emissions in policy scenarios for the United Kingdom (Pye and Palmer, 2008; Pye *et al.*, 2008). While these reports were not published in the peer-reviewed scientific literature, they are discussed here because of their close relationship with and relevance to the national-level work undertaken in this research project. They also constitute the most advanced work completed prior to the research presented in this thesis on quantifying the co-impacts (both positive and negative) of energy system technology transitions on air pollution levels in the United Kingdom using the energy system optimisation models that are utilised by the UK government.

In this work, Pye et. al. incorporated three pollutants - sulphur dioxide (SO₂), nitrogen oxides (NO_x) and particulate matter of less than 10-microns in diameter (PM₁₀) – into the UK MARKAL energy systems model. These researchers undertook two model runs at the national level in order to calculate overall changes in air pollution emissions by sector at a national level; the Energy White Paper 2007 base case and 60% carbon reduction runs (Pye *et al.*, 2008). The stated goal of these two runs was to "assess the difference between AQ emissions in the reference case and under a climate policy target case" (Pye *et al.*, 2008).

In this work, Pye et. al. found that "air quality emissions could be significantly reduced in future years as a result of technology improvements, improved efficiency and less use of polluting fuels under a reference case... [and] benefits due to [air quality] emission reductions are estimated at between £0.9 -1 billion in 2050" (Pye *et al.*, 2008).

These benefits were estimated using a damage cost approach but did not complete a full impact pathway analysis due to its "resource intensive" nature. The authors note that the model "could be further developed to assess both climate and air quality targets simultaneously. This could be done by including emission ceilings, for example, for air quality pollutants, which the model would factor in as part of the optimisation process" (Pye *et al.*, 2008).

2.3.4.4 Wadud and Waitz (2011)

In 2011, Wadud and Waitz specifically evaluated the air quality-related mortality impacts of different modes of transportation in the United States. According to the authors, "[k]nowledge about the environmental impacts of various transportation modes is important for understanding trade-offs that may be involved in policy options that affect different transportation modes in different ways" (Wadud and Waitz, 2011). In turn, this paper "reviews the literature on human health impacts attributed to various transportation modes, focusing on premature mortality, to carry out a comparative analysis of the modes" (Wadud and Waitz, 2011). These results articulate the importance of understanding the co-impacts in achieving climate change mitigation goals for the transport sector and are a key component in the motivations behind this research project.

This study considered "the relative contribution of four different modes to degrading air quality and uses the associated health impacts as a metric for comparison" including road transport, ocean shipping, rail and aviation on a national level for the United States (Wadud and Waitz, 2011). Their results presented were normalized to account for the different volumes and types of services provided by each mode of transport. The main output of this study were quantified values for the air pollution-related premature deaths on a per-ton-mile basis (Wadud and Waitz, 2011). While the study was limited to four modes of transport, its results provide valuable insights on exposure and resulting health impacts of air pollution from different modes of transport at the national scale. Their discussion is important to this work in its comparison of the human health impacts attributed to various transportation modes, including road transport technologies that are the focus in Chapter 5 of this thesis.

2.3.4.5 U.S. EPA (2009)

In 2009, the U.S. Environmental Protection Agency (EPA) produced a report of their assessment of the impacts of global climate change on regional U.S. air quality. This work focused on the climate change impacts of ground-level ozone (O_3). The design of this assessment was geared toward exploring and communicating "the potential effects of climate change on air quality in the United States" (U.S. Environmental Protection Agency, 2009).

The motivation for the focus in this report was to "ascertain whether climate change should be considered in the formation of future air quality policy" (U.S. Environmental Protection Agency, 2009). The EPA has also produced reports on the health impacts of air pollutants, many of which can claim energy production and use as a primary source.

These reports have been used as the basis for national clean air regulations including the 2014 Clean Power Plan. This proposed rule would institute additional air pollution restrictions on coal-fired power plants in the United States, focusing on carbon dioxide emissions (U.S. Environmental Protection Agency, 2014). The EPA has yet to publically release work that includes a complete quantification of the air quality and public health impacts of energy technology transitions designed to mitigate climate change. However, their work is still important in illustrating some of the limitations of the approach presented here in that the research in this thesis did not analyse the effects of climate change on air quality itself.

2.3.5 Urban Studies

Perhaps most pertinent to the research questions explored in Chapter 5 of this thesis are those studies that have focused on an urban scale. In the existing literature, there are numerous studies that have looked at large urban areas, including London. While none have had the same focus, scope, and approach as the research undertaken in this thesis, they provided valuable insights that informed the design of this research project as well as a basis for comparison with regards to the final results.

Three key urban-scale studies are discussed in this section. The first, by Jack and Kinney focused broadly on the human health benefits of climate mitigation policies, with an emphasis on urban settings (Jack and Kinney, 2010). The second, by Woodcock, et. al. explored the potentially substantial health co-benefits of measures to reduce greenhouse gas emissions, including two urban-focus areas for the transport sector – London and Delhi – in addition to a number of broader geographical focus areas (Woodcock *et al.*, 2009). Finally, Jarrett et. al. focused specifically on quantifying the economic benefits of shifting to increased levels of active travel (i.e. walking and biking) in London for the National Health Service.

2.3.5.1 Jack and Kinney (2010)

Jack and Kinney's 2010 manuscript is focused on the health co-benefits of climate mitigation in urban areas and includes a detailed review of the methodologies applied in more recent additions to the co-benefits literature including more than a dozen studies (Jack and Kinney, 2010). Jack and Kinney particularly focused on policy, environmental, and health modelling to estimate these co-benefits in urban areas and concluded that "future contributions should look beyond air pollution, analyse developing economies, and draw on research teams that bring sophistication on both the science and the policy aspects of the co-benefits question" (Jack and Kinney, 2010). Their work was aimed at providing policy-relevant estimates of these co-benefits by linking economic behaviour, environmental process, and health models. As a result, their main conclusions were a set of four dimensions where researchers can improve "the salience and credibility of co-benefits research" (Jack and Kinney, 2010).

Specifically, they recommended retrospective evaluations of past policy actions, holistic benefits inclusion beyond air quality changes (e.g. impact of active travel on obesity levels), increasing focus on developing economies, and an interdisciplinary approach that includes experts from across fields of study (Jack and Kinney, 2010). They did not recommend the inclusion of energy sector experts in this discussion, which is consistent with the largely non-technical focus of their discussion (Jack and Kinney, 2010).

2.3.5.2 Woodcock, et. al. (2009)

In 2009, Woodcock, et. al. published their results from a Comparative Risk Assessment using World Health Organization (WHO) methodology in The Lancet medical journal. Their analysis included the quantification of the effects of emissions from motor vehicle combustion for scenarios that included low-carbon-emissions motor vehicles and increased active travel and is a leading paper in the medical literature on the climate change mitigation, air pollution and public health nexus. Overall, Woodcock et. al. found that "although uncertainties remain, climate change mitigation in transport should benefit public health substantially."

Their analysis of two urban areas – London and Delhi – differs from the approach undertaken in this project in a number of ways including the fact that Woodcock et. al. included a much more limited treatment of technology options, did not utilize an energy systems model, included only CO_2 and a limited treatment of particulate matter ($PM_{2.5}$), and did not evaluate nitrogen oxide (NO_x) or non-combustion emissions from motor vehicles (Woodcock *et al.*, 2009). With the last of this list, it is noted that non-tailpipe emissions have been shown to represent the majority of current air pollution emissions from motor vehicle operation in London (Dajnak, 2013).

2.3.5.3 Jarrett et. al (2012)

Jarrett, et. al. discussed the impacts of increasing active travel in urban England and Wales on costs to the United Kingdom's National Health Service (NHS) in their study, published in 2012. This investigation is the leading academic review of the potential economic co-benefits of mode shifting for the NHS and presents a strong case for incentivizing mode shifting in the population to more active travel.

In their work, Jarrett, et. al. focused on the impacts of mode shifting from passive to active travel technologies (i.e. biking and walking) on obesity rates. In turn, the study included the economic benefits of reduced instances of type 2 diabetes, dementia, cerebrovascular disease, breast cancer, colorectal cancer, depression, and ischaemic heart disease resulting from decreased obesity rates using the World Health Organization (WHO) comparative risk assessment method and NHS costing templates. It did not include analysis of the long-term health effects of climate change or the "the effect of walking an cycling on environmental factors such as improved air quality because of reduced vehicle emissions" (Jarrett *et al.*, 2012).

In turn, the work presented in this thesis compliments rather than duplicates Jarrett, et. al.'s findings.

2.4 Modelling Air Quality Co-Impacts

More than ten integrated assessment models have been developed to evaluate the impacts of climate change policy within those modelling efforts evaluated by the Intergovernmental Panel on Climate Change (IPCC) and government entities in the United Kingdom and United States (Nemet, Holloway and Meier, 2010). Of these models, two (2) include an estimate of the air quality co-benefits as shown in Table 2.6 and so are most relevant in informing the research undertaken in this project. Noted here is that, while both of these models have been applied to analysis in the United Kingdom, these applications were not completed in the same manner as undertaken in this research project. Additional details on the models used in the IPCC's Fifth Assessment Report are discussed elsewhere in this chapter.

Table 2.6: The treatment of air quality co-benefits by integrated assessment models of climate change policy (Stern and Taylor, 2006; Intergovernmental Panel on Climate Change (IPCC), 2007; Energy Information Administration, 2008; Department of Energy and Climate Change (DECC) 2009⁻ Nemet Holloway and Meier 2010)

Model Name	Includes GHG impact estimates	Estimates the climate impact value	Estimates the air quality co-benefit	Estimates the value of the air quality co-benefit
IMAGE	Yes	No	No	No
MERGE	Yes	No	No	No
MESSAGE	Yes	No	No	No
MiniCAM	Yes	No	No	No
WIAGEM	Yes	No	No	No
DICE	Yes	Yes	No	No
MARKAL ¹⁵	Yes	Yes	Yes	Yes
PAGE2002	Yes	Yes	Yes	Yes ¹⁶
ADAGE	Yes	No	No	No
IGEM	Yes	No	No	No

The Stern Review quantified the air quality co-benefits of climate change mitigation policy as being "up to 1% of [global] GDP" using a single integrated model, PAGE2002 (Stern and Taylor, 2006). In the United Kingdom, the Department of Energy and Climate Change (DECC) concluded that the air quality co-benefits of the Climate Change Act 2008 could be worth £32 billion (Department of Energy and Climate Change (DECC), 2009). According to Nemet, et. al., the co-benefits of climate change policies could bring a co-benefit of \$2-196 per ton of carbon dioxide, due to the corresponding reductions health-damaging air pollutants (e.g. particulate matter, nitrogen oxides, and sulphur dioxide). Amann, et. al. and Bollen, et. al have estimated that air quality co-benefits could be twice as valuable as climatic benefits (Amann *et al.*, 2009; Bollen *et al.*, 2009). However, reviews of these types of economic studies have highlighted the sensitivity in co-benefits analyses to choices about both methodology and parameter values (Bell *et al.*, 2008; Nemet, Holloway and Meier, 2010).

¹⁵ In the Impact Assessment of the Climate Change Act 2008 – other air pollutants not considered directly in the optimisation pathway in UK MARKAL.

¹⁶ These values are not included in final impact values.

The Dynamic Integrated model of Climate and the Economy (DICE) model used by Nordhaus is a globally aggregated model. Housed at Yale University, this model is intended to represent the economic, policy, and scientific aspects of climate change rather than specifically explore the energy sector (Nordhaus and Sztorc, 2013).

The Applied Dynamic Analysis of the Global Economy (ADAGE) model is a computational general equilibrium model used by the U.S. Environmental Protection Agency to examine the impacts of economic, energy, environmental, climate change mitigation, and trade policies on geographic scales from state-level to international coverage (U.S. Environmental Protection Agency, 2016). Conversely, the U.S. EPA's Intertemporal General Equilibrium Model (IGEM) models the U.S. economy and simulates the effects of policy and other changes on the price, production level, and consumption of energy and pollution emissions (U.S. Environmental Protection Agency, 2016).

As shown in Table 3.2, the Intergovernmental Panel on Climate Change (IPCC) reports have used outputs from at least five (5) integrated assessment models in their evaluation of climate change policy, including:

- The Integrated Model to Assess the Greenhouse Effect (IMAGE) from the National Institute for Public Health and Hygiene in the Netherlands
- A Model for Estimating the Regional and Global Effects of greenhouse gas reductions (MERGE) housed at Stanford University

- The Model for Energy Supply Strategy Alternatives and their General Environmental Impact (MESSAGE) that has been developed by the International institute for Applied Systems Analysis (IIASA) in Austria since the 1980s
- The Mini Climate Assessment Model (MiniCAM) from the Pacific Northwest National Laboratory (PNNL) in the United States
- 5. The World Integrated Assessment General Equilibrium Model (WIAGEM) created at the University of Oldenburg in Germany

Of these five models, MESSAGE is the most closely related to the TIMES-based model utilised in this research project. Similar to TIMES, the MESSAGE systems engineering optimisation model is used by the International Institute for Applied Systems Analysis (IIASA) to produce socioeconomic and technological "response strategies" to major energy challenges including decarbonisation (International Institute for Applied Systems Analysis (IIASA), 2013). Also at IIASA and of relevance to this research is the Greenhouse gas – Air pollution Interactions and Synergies (GAINS) model, which looks specifically at air pollution abatement technologies and how they could be applied to reduce air pollution emissions from the energy sector (Amann *et al.*, 2009; International Institute for Applied Systems Analysis (IIASA), 2011).

Unlike MESSAGE, the MARKAL/TIMES modelling platform has already been successfully adopted and utilized in the United Kingdom for more than a decade and has gained acceptance amongst both UK academics and the government. According to Strachan, et. al. in the 2009 paper published in *Energy Policy* (Strachan, Pye and Kannan, 2009):

"In the UK, the MARKAL family of energy systems models has played an iterative role, providing analytical underpinning into all recent major energy policy reviews."

In 2014, Taylor, et. al. stated that "the ability of MARKAL to perform different roles for different groups has served to embed and institutionalise the model in the energy policy community" (Taylor *et al.*, 2014). In turn, the UK TIMES Model (UKTM-UCL) was used instead of MESSAGE in this research in order to increase its policy relevance in the United Kingdom. It is noted here that a transport-sector specific model was not selected for use in this research project because of its inability to account for non-transport related interactions. For example, the large-scale adoption of electric vehicles and its corresponding impact on electricity demand and emissions from the electricity sector.

Furthermore, with regards to the GAINS model, this tool is currently designed for global analysis and is used for analysis related to the Convention on Long-range Transboundary Air Pollution and the European Union. While the GAINS model can distinguish between 165 regions, including 48 European countries and 46 provinces/states in China and India it is not designed to specifically focus on a particular urban area (International Institute for Applied Systems Analysis (IIASA), 2011, 2016). Furthermore, the model is not open-source, which prevented this researcher from exploring the possibility to disaggregate the GAINS model regions to include a region for "Greater London". Finally, the GAINS model is primarily focused on a set of 2000 defined emission control measures and their costs and not broader energy system technologies (International Institute for Applied Systems Analysis (IIASA), 2016). In turn, for this researcher so that she could include an urban disaggregation for Greater London, as opposed to using the GAINS model.

2.5 Articulation of the research gap

While notable studies exist related to the air-pollution co-impacts of changes in the energy system, in-depth analysis of the nexus between the co-impacts of climate change mitigation on outdoor air pollution (and vice versa) has not yet taken place on a significant scale, with few exceptions (e.g. Barker et. al. 2010 and Woodcock et. al. 2009, IEA 2016) (Woodcock *et al.*, 2009; Barker *et al.*, 2010; International Energy Agency (IEA), 2016). For the United Kingdom in particular, published studies have yet to link an energy systems model (i.e. a TIMES-based or a similarly comprehensive model) to air pollution and public health tools, nor have they holistically evaluated energy technology transition options from a climate and public health co-impacts perspective. Given that these optimisation models are central to energy sector policy assessment in the United Kingdom, the addition of other air pollutants could provide valuable insights on the co-impacts of climate and air quality interventions.

In 2014, Dessens et. al. highlighted this existing research gap, stating that (Dessens *et al.*, 2014):

"in depth analysis of this integration [of air pollution abatement and climate change mitigation policies] has not generally taken place, either in policy literature or in the modelling. Instead the air pollution and other co-benefits have been treated as occasional added benefits for climate change policy (e.g. Stern, 2007 p. 314), or sometimes not mentioned at all (e.g. Nordhaus, 2007)."

In the United Kingdom, initial qualitative discussions of these potential co-benefits have been quantified for a limited number of scenarios (Williams, 2007; Pye and Palmer, 2008; Pye *et al.*, 2008; Milner, Davies and Wilkinson, 2012; Jensen *et al.*, 2013a). Furthermore, a

comparative risk assessment has been used to estimate the health effects of reductions in combustion-related carbon dioxide emissions from urban land transport technologies in London (Woodcock *et al.*, 2009). Outside of the peer-reviewed literature, two consulting reports (Pye and Palmer, 2008; Pye *et al.*, 2008) integrate non-GHG air pollution into a whole energy systems model, quantifying changes in air quality pollutant emissions under different U.K. policy scenarios. In this work, they included three pollutants (SO₂, NO₂, and PM₁₀) into the UK MARKAL energy systems model and found that "air quality emissions could be significantly reduced in future years as a result of technology improvements, improved efficiency and less use of polluting fuels under a reference case... [and] benefits due to [air quality] emission reductions are estimated at between £0.9–1.0 billion in 2050" (Pye *et al.*, 2008). At the time, the authors noted that the model "could be further developed to assess both climate and air quality targets simultaneously. This could be done by including emission ceilings, for example, for air quality pollutants, which the model would factor in as part of the optimisation process" (Pye *et al.*, 2008).

In 2009, Haines, et. al noted that "the varying costs of implementation of [strategies to reduce greenhouse-gas emissions] can be offset at least partly by the benefits to health and development, and these co-benefits should be taken into account in international negotiations" (Haines *et al.*, 2009). Furthermore, "the methods for assessing the health effects of mitigation strategies for climate change...should be further developed and applied to inform policy making."

The statements by Haines, et. al. echoed ideas published previously by Williams in 2006. In his work to qualify and quantify the co-benefits of climate change mitigation strategies, Williams noted "significant synergies and co-benefits are possible through a concerted

88

consideration of air quality and climate change policies." This researcher has subsequently focused his efforts on developing air quality models that quantify the public health impacts of changes in air quality (Beevers *et al.*, 2012, 2013)

In 2011, Thambiran and Diab stated that "air quality and climate change are inextricably linked... this relationship provides a scientific basis for developing integrative policies that derive multiple benefits for simultaneously improving air quality and addressing climate change." Furthermore, they note that "opportunities to use air quality interventions in an innovative manner to contribute toward creating low carbon, resilient communities are mostly overlooked" in their home country of South Africa (Thambiran and Diab, 2011).

The next chapter provides an overview of the methodologies applied in this research project. Subsequent chapters present and discuss the results of the application of these methodologies. This page was intentionally left blank.

Chapter 3 - Methodology

3.1 Overview

This chapter provides details of the methodology applied over the course of this research project to explore the co-impacts of energy technology transitions on climate change mitigation efforts and air pollution. This methodology can be broadly outlined as followed with regards to each research questions explored in this work, namely:

- 1. What are the co-impacts (both positive and negative) on particulate matter and nitrogen oxide air pollution levels for energy sector decarbonisation pathways that are optimised with regards to reducing total greenhouse gas emissions on both a national and urban scale?
- 2. How does considering the impact of these other types of outdoor air pollution (i.e. particulate matter and nitrogen oxides) impact the decarbonisation pathway on both a national and urban scale?

For the United Kingdom, the following steps were undertaken:

Step 1: Understand historic trends to provide a foundation for future scenario development and analysis of air quality and energy-sector air pollution in the United Kingdom in order to quantify the current gap between air pollution levels and World Health Organization recommended levels and estimate the future trajectory given historic trends.

- 1.1 Approach: Sustainability Gap (SGAP) methodology
- 1.2 Determine historic trends of particulate matter (PM₁₀ and PM_{2.5}), nitrogen oxides (NO_x), sulphur oxides (SO_x), ammonia (NH₃), non-methane volatile organic compounds (NMVOCs) and carbon dioxide equivalent (CO₂-eq, Kyoto basket) for the United Kingdom.
- 1.3 Calculate the existing gap between current pollution levels and World Health Organisation (WHO) targets.
- 1.4 Identify pollutants of interest for subsequent evaluation in an energy systems model and an air quality model.

Step 2: Construct a variant of the energy systems model (UKTM-UCL) that includes air pollutants of interest. Use this model (UKTM-UCL-AQ) to calculate the co-impact of decarbonisation scenarios on other air pollutants of interest and determine the impact of including air pollution damage costs in the optimization pathway.

- 2.1 Identify data inputs of interest (e.g. technology emissions factors) and update the UKTM-UCL model to include these values, producing the UKTM-UCL-AQ model.
- 2.2 Produce outputs for scenarios using UKTM-UCL-AQ to 2050, including all sectors.
- 2.3 Incorporate damage costs (using methodology from U.K. Department of Energy and Climate Change and Department for Environment, Food & Rural Affairs, 2011 guidance)

- 2.4 Re-run scenarios with damage costs included for air pollutants of interest, including all sectors in order to account for air pollution changes from other sectors at an aggregated level.
- 2.5 Compare outputs to previous model runs to determine impacts of incorporating health impacts to develop potential answers.

The combination of Step 1 and Step 2 as outlined above allow for the exploration of the answers to the research questions on a national scale. Steps 3 and 4 as outlined below then enable the examination of the two research questions at the urban scale after examining the outputs from the national scale analysis. As previously discussed in this thesis, focus was placed on the Greater London region for this research project.

Step 3: Create a model that can disaggregate UKTM-UCL-AQ outputs into two regions (Greater London, rest-of-UK) and calculate the resulting public health impact for Greater London of changes in air pollution levels.

- 3.1 Gather data, particularly with regards to London-specific population growth and energy demand.
- 3.2 Examine the strength of existing scientific evidence related to air pollution health impacts to identify the sub-set of pollutants examined in the United Kingdom research that can be explored in depth with regards to public health impacts in Greater London.
- 3.3 Incorporate public health impact calculations into the model for identified pollutants of interest.

Step 4: Use the PIONEER model developed in Step 3 to examine the impacts of the technology transitions produced in scenarios along two dimensions - technological and behavioural change – to improve understanding of the potential impacts of each.

- 4.1 Produce output scenarios from PIONEER to 2050 for the Greater London road transport sector using scenarios from United Kingdom analysis.
- 4.2 Run variants to include both technological and behavioural change, completing an iterative loop to ensure model convergence (as needed).
- 4.3 Compare outputs.

The PollutION Emissions from EneRgy (PIONEER) model was developed specifically for this research project by the author of this thesis. This model allows for the quantification of the potential air pollution and public health impacts of national scale technology transition scenarios on the Greater London region. It also allows the user to explore targeted action for the Greater London area to quantify the relative impact of urban versus national action.

In this work, the PIONEER model was soft-linked to a variant of the UK TIMES Model (UKTM-UCL-AQ) that includes endogenized non-greenhouse gas air pollution from the energy sector for particulate matter (PM_{10} and $PM_{2.5}$), nitrogen oxides (NO_x), sulphur oxides (SO_x), ammonia (NH_3) and non-methane volatile organic compounds (NMVOCs). The author

of this thesis is also using PIONEER to support ongoing work to build a multi-region UK TIMES Model (London-TIMES) as is briefly described in this chapter¹⁷.

The chapter begins with an overview of the Sustainability Gap (SGAP) methodology (Section 3.2) followed by a history of energy systems models and their use in energy policy making in the United Kingdom with details on the choice of the energy systems model that was used in this research (Section 3.3). This discussion is followed by background details on the United Kingdom's energy sector, including the country's ongoing climate change mitigation efforts (Section 3.4). This section is followed by sections on the application of the SGAP methodology (Section 3.5) and details on the UKTM-UCL model and the development of the UKTM-UCL-AQ model variation used in this work (Section 3.7). The PIONEER model structure is then discussed including details of its soft-linking with UKTM-UCL-AQ (Section 3.7). The final section briefly discusses the ongoing work to build a multi-region London-TIMES model as well as to extend this work to incorporate explicit analysis of the air quality impacts of energy technology transitions in partnership with Kings College London utilizing their Community Multi-Scale Air Quality (CMAQ) model for the United Kingdom (Section 3.8). The relationships between each of these models with regards to topical coverage and their application in this research project are displayed in Figure 3.1.

As described in more detail elsewhere in this manuscript, the UKTM-UCL-AQ model was developed by a team of researchers at University College London – including the author of this

¹⁷ The author of this thesis built the transportation sector of this UKTM-UCL variant, as well as a portion of the electricity sector with more minor contributions to the other (industry, services, residential, agriculture) portions of the energy sector.

thesis - in partnership with Aether, an air quality and climate change emissions consultancy with offices in the United Kingdom and Spain that specializes in emission inventories, environmental data systems and air quality assessments¹⁸. The PollutION Emissions from EneRgy (PIONEER) model was created specifically for this research project exclusively by the author of this thesis and applied to the Greater London urban area. The Community Multiscale Air Quality Model (CMAQ) is housed at Kings College London and is being used in a related collaborative project involving the author of this thesis and her colleague, Steve Pye at University College London's Energy Institute, and researchers at Kings College London (Williams *et al.*, 2016).





Throughout these sections are explicit details on key assumptions made in the construction of these tools. Specific assumptions related to the implementation of these tools are discussed in

¹⁸ http://www.aether-uk.com/

subsequent chapters, which focus on the analysis conducted at the national (United Kingdom) and urban (Greater London) level.

3.2 Sustainability Gap Methodology

Historic data can be used to explore potential future air pollution trends and air pollutants of research interest using a wide array of assumptions. With regards to this research, the difference between current levels of air pollution and sustainable levels has been defined under the Sustainability Gap (SGAP) methodology. The SGAP methodology also includes a Years-to-Sustainability (YtS) indicator, which assumes a continuation of historic air pollution trends as one moves into the future. For the sake of transparency and ease of understanding among a heterogeneous audience, this approach approximates all trends as being linear over the period evaluated and assumes that they will continue in a linear fashion moving forward (Ekins and Simon, 2001). The SGAP metric and YtS indicator concepts are displayed graphically in Figure 3.2 under this methodology.



Figure 3.2: The Sustainability Gap (SGAP) methodology (Lott, Ekins and Davies, 2014)

Within the SGAP framework, sustainability standards are set according to scientific understanding of the day with respect to the emissions limits that the environment and human

health can tolerate from anthropogenic sources. Where possible, standards are compared with policy targets (also referred to as sustainability targets) and used to quantitatively establish the sustainability gaps for the particular air pollutants that were discussed.

In 2001, Ekins and Simon calculated the SGAP air emissions component with respect to three environmental themes - climate change (C_{eq}), ozone depletion (O_{eq}) and acidification (A_{eq}) (Ekins and Simon, 2001). Furthermore, global carbon dioxide equivalent¹⁹ emissions and six types of local pollutants²⁰ emissions were included in the 2001 SGAP analysis by Ekins and Simon. In 2014, public health impacts were added to this work (Lott, Ekins and Davies, 2014).

The air pollution component of the SGAP methodology includes pollutants that, broadly speaking, significantly:

- 1. contribute to global climate change
- 2. harm/destroy the ozone layer
- 3. negatively impact on human health

The updated SGAP methodology includes air pollution targets that are based both on total emissions levels (tonnes/year) and concentration-based targets ($\mu g/m^3$) in order to capture the health impacts of both prolonged exposure and spikes in pollution levels (Lott, Ekins and Davies, 2014). For the application of the SGAP methodology for the United Kingdom, pollutants are included in the updated methodology based on their current prevalence in the

 $^{^{19}}$ CO_2, CH_4, CFCs, HFCs, PFCs, SF_6, CF_4, N_2O.

²⁰ SO₂, CO, VOCs, Pb, particulate matter, and NO₂

United Kingdom in addition to their impact under one of the three categories previously identified.

While this methodology is both transparent and easily understood by a heterogeneous audience, there are many reasons why YtS values could be overly optimistic, including:

- in the case of pollutants that have already seen dramatic reductions, assuming that historic trends can be extended into the future could be unrealistic.
- for those pollutants with sustainability standards equal to zero, the Years to Sustainability (YtS) values imply that these pollutants could be completely eliminated, which might not be practical depending on the sources for these pollutants.
- for those SGAPs that are policy (and not sustainability) targets, the YtS metric is likely significantly larger. For example, the sustainability target for particulate matter would be "0" as only complete elimination of this type of pollution would eliminate the corresponding health impact. However, a larger target value would provide a more practical goal.

However, this methodology still provides a straightforward structure for evaluating historic emissions datasets and providing one viewpoint on possible future trends. This information also helps in identifying pollutants of interest for future investigations by providing context and a high-level understanding of potential future impacts.

3.3 Energy System Models: History and Their Use in Policy-Making

This section includes a history of energy system models and their use in energy policy making in the United Kingdom. This discussion is followed by a brief overview of the use of integrated assessment models to evaluate the air pollution co-impacts of climate mitigation strategies (and vice versa). This information provides background on the development of energy modelling and justification for the selection of a TIMES-based energy system model (UKTM-UCL-AQ) for use in this research project. This section concludes with an overview of the specific models used in this research project.

3.3.1 A Brief History of Energy Systems Modelling

A wide variety of models have been developed since the early 1970s for analysing energy systems and sub-systems as an extension of previous work with energy balances. These models have included methodologies from disciplines including engineering, economics, and operations research. As a result, they have allowed users to increase their understanding of the present energy system and improve future planning (Bhattacharyya and Timilsina, 2010). Today, energy models can claim a long track record of informing major energy policy initiatives around the globe (Jebaraj and Iniyan, 2006; Strachan, Pye and Kannan, 2009).

Each energy model can differ in its applied techniques as well as its "purpose, philosophy, features, capabilities, possible overlaps and data demand" that make them more or less appropriate for sets of specific applications and given resource availability (Bhattacharyya and Timilsina, 2010). Furthermore, existing energy systems models can be differentiated according to their approach (bottom-up versus top-down), methodology (partial equilibrium, general equilibrium, or hybrid), modelling technology (optimisation, econometric, or accounting) and spatial dimension (sub-national, national, regional, and global).

While most of these terms are self-explanatory, bottom-up versus top-down models bear some additional explanation (Hourcade, Jaccard and Bataille, 2006). The former approach focuses on technical characteristics within the energy sector and can be quite useful in investigating the

100

tradeoffs that come with technology substitutions (Hourcade, Jaccard and Bataille, 2006; Bhattacharyya and Timilsina, 2010). The latter follows an collected view of the energy sector and can be used more effectively than bottom-up models to explore questions of economic competitiveness and wider economy impacts (Hourcade, Jaccard and Bataille, 2006; Bhattacharyya and Timilsina, 2010). Hourcade, et. al. illustrate the tradeoffs between energyeconomy models across three dimensions in their 2006 manuscript, as shown in Figure 3.3. These dimensions include:

- 1. technological explicitness
- 2. microeconomic realism
- 3. macro-economic completeness

Figure 3.3: Three-Dimensional Assessment of Energy-Economy Models (Hourcade, Jaccard and Bataille, 2006)



While the use of models that examine the interactions between energy, resources, and the economy can be traced back to the 1960s, the interactions between energy and the environment,

including climate change came into prominence in the 1990s according to Bhattacharyya and Timilsina in their review of energy system models that was published in 2010 in the *International Journal of Energy Sector Management* (Bhattacharyya and Timilsina, 2010). Environmental effects related to energy production, conversion, and use were incorporated into energy models during this period using environmental and pollution coefficients, which allowed these models to link environmental impacts with economic implications (Bhattacharyya and Timilsina, 2010). In 1990, at the start of this rise to prominence for these models, four approaches were identified for the inclusion of environmental impacts into electricity planning models that also hold for energy system models that wish to include environmental effects (Markandya, 1990):

- 1. models that include environmental costs as part of energy supply costs and then minimise the total costs
- 2. models that include environmental costs in the supply-side but minimise costs subject to environmental constraints
- models that aim for cost minimisation but also include an impact calculation model that is run iteratively to evaluate alternative scenarios
- 4. models not based on optimisation but rather analyse the impacts of alternative system development scenarios

Lists of prominent energy-economy models can be found in the literature with one list being reproduced here in Table 3.1 (Pandey, 2002; Bhattacharyya and Timilsina, 2010).

Table 3.1 Classification of energy-economy	/ models (Bhattacharyya	and Timilsina	, 2010)	from
(Pandey, 2002)					

Paradigm	Space	Sector	Time	Examples
Top-down/simulation	Global, national	Macro- economy, energy	Long term	AIM, SGM2, I/O models
Bottom-up optimisation/ accounting	National, regional	Energy	Long term	MARKAL, TIMES, LEAP
Bottom-up optimisation/ accounting	National, regional, local	Energy	Medium term, short term	Sector models (power, coal)

With regards to dynamic technology-economic models, the Market Allocation Model (MARKAL) is perhaps the most well-known. Developed by the International Energy Agency's Energy Technology Systems Analysis Programme (ETSAP), this modelling platform includes users at more than 75 institutions in more than three dozen countries, including many developing economies. This technology-rich, bottom-up model is tailored for particular applications using input data that is specific to the nation, region, state, or community to which it will be applied (Energy Technology System Analysis Programme (ETSAP), 2014a). In their 2001 publication, Seebregts, et. al. state that the "MARKAL family of models has been contributing to energy/environmental planning since the early 1980s" and that this "family of models is unique, benefiting from application in a wide variety of settings and global technical support from the international research community" (Seebregts, Goldstein and Smekens, 2002).

In 2008, The Integrated MARKAL-EFOM System (TIMES) model generator was chosen to replace MARKAL by its developers in order to conduct in-depth energy and environmental analyses (Loulou *et al.*, 2005). This model "combines two different, but complementary, systematic approaches to modelling energy: a technical engineering approach and an economic

approach" (Energy Technology System Analysis Programme (ETSAP), 2014b). TIMES is a bottom-up, perfect-foresight model that optimizes across all sectors and all periods of time for a prescribed scenario (Loulou *et al.*, 2005). More details on the TIMES methodology are found elsewhere in this Chapter.

3.3.1.1 Choice Of The Energy Systems Model Used For This Research

This section includes discussion of the choice of the specific modelling tools used for this research. Furthermore, details are provided on the limitations of these tools at the onset of this research project in 2013 and the efforts that were required to further develop the core tools.

As discussed elsewhere in this chapter and Chapter 2, a variety of modelling tools have been developed and used to provide a sound analytical framework from which to systematically explore pathways to meet decarbonisation goals via technology transitions in the energy system (Pandey, 2002; Bhattacharyya and Timilsina, 2010). In their 2016 paper on "improving deep decarbonisation modelling capacity for developed and developing country contexts", Pye and Bataille discuss the key motivations for the use of particular models to explore energy transitions in a given context, including their (Pye and Bataille, 2016):

- 1. being fit-for-purpose
- 2. having in-country capacity
- 3. transparency, communicability and policy credibility

In the context of this research project, being "fit-for-purpose" means that any modelling tools used had to be capable of incorporating relatively long time horizons (i.e. through 2050 in line with the UK Climate Change Act) and at varying spatial scales (i.e. both the United Kingdom and an urban subset of that area). Furthermore, this research required the use of a modelling approach that could capture the multiple impacts of energy system decarbonisation that the researcher wished to explore - namely, the co-impact of changes in energy technologies to meet decarbonisation goals on non-greenhouse gas air pollution and public health. In turn, it was important to understand the technology transition pathway in high levels of detail in order to capture these co-impacts.

In determining what type of model to use for this research, the key strengths and weaknesses of the bottom-up versus top-down modelling approaches that are described elsewhere in this chapter were considered. Broadly speaking, top-down models have an advantage over bottom-up models in that they can model the impacts of decarbonisation policies on GDP, employment and the economy but they are more limited in their ability to accurately model detailed technology-focused policies and regulations (Pye and Bataille, 2016). This limitation is particularly problematic for this research, where the co-impacts resulting from technology transitions driven by decarbonisation policies are vital in assessing the air pollution and public health co-impacts.

In comparison, bottom-up models can provide an integrated view of the full energy system with explicit representation of individual technologies and the interactions between energy subsectors (e.g. transport, power, etc). Their detailed representation of energy system technologies allows for the effective modelling of the co-impacts of technology-focused policies and regulations, which are key in both the United Kingdom and Greater London analyses presented in this thesis. Therefore, these bottom-up models are more appropriate for this research than their top-down counterparts. However, it should be noted that bottom-up models have weaknesses that are important to be aware of when drawing insights from the outputs of these types of models. These weaknesses include their significant data requirements and the resulting uncertainties, approach to incorporating human behavior, exogenously defined energy service demands, and lack of ability to model many of the broader economy impacts of changes to the energy system (e.g. the impact of increased efficiency on industrial demand) (Usher and Strachan, 2012; Pye, Sabio and Strachan, 2015; Pye and Bataille, 2016). Furthermore, in the application of these models, the evaluation of transitions over long timeframes inherently introduce significant uncertainties regarding technology development and deployment in the energy system as well as technology costs. Finally, these models do not include the impacts of changes outside of their system boundaries beyond the input assumptions provided (e.g. availability and cost of imported fuel), which limits the insights that can be drawn with regards to air quality impacts that are a function of both changes inside and outside of the spatial boundaries considered.

Of the available bottom-up models, the TIMES modelling framework (i.e. UKTM-UCL) was the most suitable for this research in the context of its having both "in-country capacity" and "transparency, communicability and policy credibility" in the United Kingdom. The history of energy systems modelling in the United Kingdom, including the extensive use of the MARKAL and TIMES models in support of numerous Energy White Papers and other government reports is discussed in more details elsewhere in this thesis (Usher and Strachan, 2012; Ekins *et al.*, 2013; Dodds, Keppo and Strachan, 2014; Pye *et al.*, 2015; Hall and Buckley, 2016). Additional details on the TIMES model methodology can be found elsewhere in Chapter 3. The TIMES model that was chosen for this research project (i.e. UKTM-UCL) required the addition of a set of capabilities that were not previously included at the start of this research in 2013 in order to facilitate its effective implementation in this work. In line with the primary goals sets out for this research project as described in Chapter 1, these capabilities included the need to capture the impact of energy system technology transition pathways on 1) key types of air pollution, including their public health impact and 2) on air pollution emissions in an urban area within the United Kingdom. The process of incorporating these capabilities into the tools used in this work is described elsewhere in Chapter 3 in the decription of the development of both the UKTM-UCL-AQ and the PollutION Emissions from Energy (PIONEER) models.

Specifically, with regards to spatial and temporal resolution, the tools selected and developed for this research examine changes in the energy system in five-year time slices with countryand urban-level resolution. In turn, they are appropriate for the quantification of trends on these scales, which is appropriate given the geographic focus of the United Kingdom's Climate Change Act. However, these tools would not be appropriate for use in examining a number of related research questions that require quite high levels of spatial and/or temporal resolution nor those requiring detailed air pollution chemistry modelling (for example, the impact of air pollution from cars driving on a particular street in Greater London on hourly or daily mean air pollution concentrations).

3.3.2 TIMES Methodology Overview

As described by Dodds, et. al. in their 2014 paper, energy systems models can be both opaque and difficult to understand as new model versions are developed. In turn, according to the authors (Dodds, Keppo and Strachan, 2014):

"energy system models need to be as clear and transparent as possible to ensure quality assurance for users and replicability for practitioners... Model transparency and repeatability are even more relevant for energy system models as these technology-rich, economic optimisation models, such as [MARKAL/TIMES], have become critical tools for informing policy and business decisions in low-carbon energy technologies in many countries"

A full documentation of the TIMES model generator used in this research is provided by ETSAP in their five-part series titled "Documentation for the TIMES Model" (Loulou, Goldstein, *et al.*, 2016; Loulou, Kanudia, *et al.*, 2016; Loulou, Lehtilä, *et al.*, 2016; Loulou, Remme, *et al.*, 2016; Wright *et al.*, 2016). In the interest of transparency and repeatability, the second part of this documentation as it existed during the course of this research project is included in the appendix of this thesis. As stated by the authors (Loulou, Lehtilä, *et al.*, 2016):

"Part II [of this documentation series] constitutes a comprehensive reference manual intended for the technically minded modeler or programmer looking for an in-depth understanding of the complete model details, in particular the relationship between the input data and the model mathematics, or contemplating making changes to the model's equations. Part II includes a full description of the sets, attributes, variables, and equations of the TIMES model."

As this portion of the documentation includes all of the key structural details of the TIMES model, the other portions of the TIMES model documentation provided by ETSAP did not need to be reproduced in the Appendix of this thesis.
In addition to this documentation, an overview of key components of the TIMES model generator that was used to determine scenario outputs in this research is discussed in this section. Of particular importance is the role of user inputs and the structure of the objective function.

3.3.2.1 The TIMES Model Generator

The TIMES model generator is used to create technology explicit, region specific, partial equilibrium models of the energy system that assume energy markets that are competitive and have perfect foresight (Loulou, Goldstein, *et al.*, 2016). According to the ETSAP documentation's authors (Loulou, Goldstein, *et al.*, 2016):

"TIMES (an acronym for The Integrated MARKAL-EFOM System) is an economic model generator for local, national, multi-regional, or global energy systems, which provides a technology-rich basis for representing energy dynamics over a multi-period time horizon."

The TIMES model generator is not intended to create models that will predict the future technology make-up of the energy system. Rather, it is a tool for exploring possible future scenarios in order to better understand their potential impacts and relative trade-offs.

All TIMES-based models use an identical mathematical structure and are economically rational, meaning that they strive to maximize the total surplus (i.e. the sum of producer and consumer surplus) of an energy economy over the entire time horizon being analysed (Loulou, Goldstein, *et al.*, 2016). However, each unique TIMES-based model is ultimately defined by sets of user inputs that are specific to the region being analysed (Loulou, Goldstein, *et al.*, 2016). For this research, the TIMES model used is specific to the United Kingdom.

Included in the TIMES model generator is a reference energy system with a defined set of technologies, commodities, and commodity flows. These components are defined as follows (Loulou, Goldstein, *et al.*, 2016):

- Technologies (also called processes): representations of physical devices that transform commodities into other commodities. Processes may be primary sources of commodities (e.g. mining processes, import processes), or transformation activities such as conversion plants that produce electricity, energy-processing plants such as refineries, end-use demand devices such as cars and heating systems, etc.
- Commodities: energy carriers, energy services, materials, monetary flows, and emissions. A commodity is generally produced by some process(es) and/or consumed by other process(es).
- Commodity flows: the links between processes and commodities. A flow is of the same nature as a commodity but is attached to a particular process, and represents one input or one output of that process.

3.3.2.2 TIMES Model Inputs

In order to create a TIMES model that can be applied in order to produce future energy scenarios, users input four key sets of information as exogenous inputs to the TIMES model framework (Loulou, Goldstein, *et al.*, 2016):

- 1. energy service demands by type (e.g. heating, demand for cars) over time calculated using defined demand drivers (e.g. population, GDP) for a defined time horizon
- 2. resource availability (i.e. the amount of a resource that the model can use to supply demand at a specified cost)

- policies any information on the policy setting under which the scenario is to be run (e.g. carbon reduction targets)
- technical and economic parameters including all descriptive parameters for individual technologies and processes (e.g. cost, efficiency, construction timelines, technical life, commodity use per activity)

With regards to energy service demands by type, these values are typically calculated within TIMES using user-inputted assumptions for both demand drivers and elasticities of demand. Total energy service demand is calculated using the aforementioned demand drivers. Furthermore, the model calculates the resulting demand for individual commodities in order to meet the energy service demands (Loulou, Goldstein, *et al.*, 2016).

Resource availability is set by the modeller in terms of the amount of available resource at a defined cost (Loulou, Goldstein, *et al.*, 2016). In turn, energy supply becomes defined as an energy supply curve that represents the amount of resource that can be utilized in the model's solution at a defined cost. For example, reserves of oil are defined in terms of the amount of oil that is available at each of a series of prices.

The modeller also defines policies that can impact the energy scenario that they wish to be included in their scenarios (Loulou, Goldstein, *et al.*, 2016). For example, a solar mandate is included as a requirement for the model to use a specified amount of the available solar resource. With regards to decarbonisation, constraints are included to meet defined limits on greenhouse gas emissions for specific timeframes as defined by the modeller. Given the model's structure as previously described, these constraints are met just-in-time and with perfect foresight.

As previously discussed, TIMES is a bottom-up model and its technology and economic parameters are defined exogenously by the modeller for all technologies and processes (Loulou, Goldstein, *et al.*, 2016). For example, if the modeller wishes to consider a specific type of car technology that can be used to meet a demand for vehicles, they will define a long list of characteristics for the car including the initial investment cost as well as the costs for maintenance when used. The user will also define the car's fuel requirements, which will - of course - have their own associated costs. All of these parameters can vary over the time horizon being explored in a given scenario or be held constant for the period. A generalised schematic of the TIMES model structure is provided in Figure 3.4.



Figure 3.4: Schematic of a generalised TIMES model (Loulou et al., 2005)

In TIMES-based models, the time horizon and level of temporal resolution is defined exogenously by the modeller. In the runs presented in this research, that time horizon of interest

was defined as 2010 - 2050 with 5-year times slices. In turn, the outputs of the scenario include results for nine points including 2010 (i.e. the baseyear), 2015, 2020, 2025, 2030, 2035, 2040, 2045 and 2050. In practice, the model was set to run to 2060 in order to avoid distortions in the 2050 results. As noted elsewhere in this chapter, these types of distortions are a known issue in the TIMES modelling platform and are avoided by running the model beyond the time horizon that the modeller wishes to analyse (Loulou, Lehtilä, *et al.*, 2016).

3.3.2.3 The TIMES Objective Function

TIMES models are driven to minimize the total discounted cost of the entire energy system over the selected time horizon (Loulou, Lehtilä, *et al.*, 2016). This total discounted cost is represented by the model's objective function, which is defined as the sum of all regional objectives (REG_OBJ) over all years (y) and regions (r), with all of the costs discounted using an exogenously defined discount rate (DISC) to the same user-selected base year (z) (Loulou, Lehtilä, *et al.*, 2016). This function is shown in Equation 1.

$$VAR_Obj(z) = \sum_{r \in REG} REG_OBJ(z,r)$$

Eq. 1

Within each regional objective is a set of nine cost components and two revenue components, as shown in Equation 2 (Loulou, Lehtilä, *et al.*, 2016).

 $REG_OBJ(z,r)$

$$= \sum_{y \in (-\infty, +\infty)} DISC(y, z)$$

$$\times [INVCOST(y) + INVTAXSUB(y) + INVDECOM(y) + FIXCOST(y)$$

$$+ FIXTAXSUB(y) + SURVCOST(y) + VARCOST(y) + VARTAXSUB(y)$$

$$+ ELASTCOST(y) - LATERREVENUES(y) - SALVAGE(z)]$$

Eq. 2

The nine cost and two revenue components are listed below, with corresponding descriptions (Loulou, Lehtilä, *et al.*, 2016):

- Investment Costs (INVCOST): the costs related to the investment, which occur in the year an investment is decided upon and/or during the construction period for that investment. These investment costs can be single payments, or can be period payments made over a series of years.
- 2. Taxes and subsidies on investments (INVTAXSUB): these costs are assumed to be incurred at the same time as the investment cost is incurred.
- 3. Decommissioning/Dismantling capital costs (INVDECOM): these costs are incurred after the end-of-life of an investment and can include a lag period, as defined by the user.
- 4. Fixed annual costs (FIXCOST): the fixed annual cost that is paid during the operation of the investment.
- 5. Annual taxes/subsidies on capacity (FIXTAXSUB): these annual taxes/subsidies are assumed to be paid at the same time as any fixed annual costs.

- Survival cost (SURVCOST): these costs also include annual costs during any lag period between end-of-life and the start of decommissioning.
- Variable operating costs (VARCOST): includes those costs that vary according to the activity being undertaken by the investment
- Variable taxes/subsidies (VARTAXSUB): includes those costs corresponding to annual taxes/taxes
- 9. Cost of demand reductions (ELASTCOST): this cost applies in scenarios where elastic energy service demands are used and represented the cost resulting from the loss of welfare due to the reduction (or increase) of demands
- 10. Late revenues from endogenous commodity recycling (LATEREVENUE): these costs include revenues from any materials and energy that are embedded in a process and are subsequently released after the end of the scenario time horizon.
- 11. Salvage value (SALVAGE): when an investment's technical life extends beyond the scenario time horizon, this value is used to represent the value of the unused portion of the investment

Further details on each of these components is found in the Appendix of this thesis.

There are a few notable issues that result from the structure of the TIMES objective function that can impact a model's evaluation of the cost competitiveness of individual technologies. These challenges have been documented by the model's developers, along with a set of recommended mitigation techniques to reduce distortions to the final costs as calculated by the TIMES model itself (Loulou, Goldstein, *et al.*, 2016). These issues include the following (Loulou, Lehtilä, *et al.*, 2016):

- distortions resulting from the specific assumptions made related to annual payments versus available capacity these distortions are "usually quite small" except in the case of "longer periods having an even number of years" according to the model's developers. (Loulou, Lehtilä, *et al.*, 2016).
- in the case where the end of the time horizon analysed directly corresponds to the end of life for an investment, investment cost accounting can result in distortions in the model's salvage value accounting.
- for investments, capacity availability assumptions can cause "a small distortion in the cost accounting" because any capacity available in a given year has a larger value than the same capacity in the subsequent year (Loulou, Lehtilä, *et al.*, 2016).
- in the case of variable period lengths where investment costs change over time, there can be accounting distortions due to the fact that investment cost data is taken from the start year of each investment step.

In turn, when using the TIMES model generator, researchers should be particularly cautious – in their approach to defining investment timelines and analysis time periods. They should also familiarize themselves with the steps that the TIMES model generator's developers have taken to address the four issues previously mentioned. These steps include optional switches to eliminate the distortions resulting from discounting and annual payment assumptions related to investment and fixed costs (Loulou, Lehtilä, *et al.*, 2016).

3.3.3 Energy System Modelling in the United Kingdom

There exists a long track record of energy modelling supporting policy initiatives in the United Kingdom (Strachan, Pye and Kannan, 2009). In particular, the MARKAL energy system model has been used extensively to inform U.K. energy and climate policy since the turn of the

century and remains a dominate energy systems model today (Dodds, 2014; Taylor *et al.*, 2014; Hall and Buckley, 2016). Outputs and analysis from both UK-MARKAL and, more recently, the United Kingdom TIMES Model (UKTM-UCL) provided inputs for the 2003 Energy White Paper, 2007 Energy White Paper, and 2011 Carbon Plan as well as the Committee on Climate Change reports *Building a Low-Carbon Economy, Fourth Carbon Budget*, and *Fifth Carbon Budget* (Department of Trade and Industry (DTI), 2003; Department of Energy and Climate Change (DECC), 2007; Committee on Climate Change, 2008, 2010, 2015a, 2015b).

According to Taylor, et. al. in their 2014 paper published in *Energy Research and Social Science* on the operation of technical energy models within social systems, "the ability of MARKAL to perform different roles for different groups has served to embed and institutionalise the model in the energy policy community" (Taylor *et al.*, 2014). Furthermore, Taylor, et. al. state that MARKAL has the ability to serve "different but intersecting needs of academic and policy communities over a sustained period of time." In turn, the use of this model has brought together communities across these two worlds, resulting in an influential network of academics and policy makers within the United Kingdom (Taylor *et al.*, 2014).

Noted here is that, prior to this thesis, neither UK MARKAL or the UK TIMES Model (UKTM-UCL) were equipped to directly consider the public health impacts of changes in outdoor air quality resulting from air pollution from the energy sector. Furthermore, the UKTM-UCL model is built to execute a national-level (single region) energy system analysis.

3.4 Background – Energy and Air Pollution in the United Kingdom

This section provides background on energy and air pollution in the United Kingdom. A majority of the discussion centres around 2010, which is the base year used in much of this research. However, some information on changes and pertinent events since 2010 are included. This section is followed by a discussion of the application of UKTM-UCL-AQ in this research.

3.4.1 U.K. Total Primary Energy Supply & Final Energy Consumption

The United Kingdom – including England, Scotland, Wales, and Northern Ireland – had a total primary energy supply of 203 million tonnes of oil equivalent (Mtoe) in 2010 according to energy statistics reported by the U.K. Department of Energy and Climate Change²¹ in the *Energy Consumption in the UK* publication. The vast majority (88%) of this supply came from fossil fuels including natural gas (42%), oil (31%) and coal (15%) (Department of Energy and Climate Change (DECC), 2015a). However, these values have been on the decline as the nation works to achieve an array of climate change and air pollution goals (International Energy Agency (IEA), 2012). One indicator of this trend can be seen in Scotland, which shut-down the last operating coal-fired power plant within its borders in March 2016 (Lott, 2016).

According to these statistics from the Department of Energy and Climate Change, the vast majority (81%) of final energy consumption in the United Kingdom was in England in 2010 as shown in Figure 3.5 (Department of Energy and Climate Change (DECC), 2015a). This fact is unsurprising given that around 83% of the nation's population also live in this country according to data from the United Kingdom Office of National Statistics (Office of National Statistics, 2010). Disaggregating energy consumption for Greater London from the national

²¹ now the Department of Business, Energy and Industrial Strategy (BEIS)

totals reveals that this urban area consumes 9% of the nation's total energy as shown in Figure 3.6. In 2010, the total population for the London area was 8.1 million people or about 13% of the total population for the United Kingdom (Office of National Statistics, 2010).



Figure 3.5: Final Energy Consumption by Country in the United Kingdom (2010) (Department of Energy and Climate Change (DECC), 2015a)

Figure 3.6 Final Energy Consumption by Country in the United Kingdom with Greater London Break-Out (2010) (Department of Energy and Climate Change (DECC), 2015a)



Final energy consumption in the United Kingdom by sector can roughly be divided into thirds, with the largest portion (37% of total) being attributed to the industrial and commercial sectors and transport representing the smallest wedge (30% of total) as shown in Figure 3.7. For

comparison, final energy consumption in Greater London includes a smaller wedge (20%) for transport, with a much larger portion (41%) being dedicated to domestic use as shown in Figure 3.8.

Figure 3.7 Final Energy Consumption by Sector in the United Kingdom (2010) (Department of Energy and Climate Change (DECC), 2015a)



Figure 3.8: Final Energy Consumption in Greater London by Sector (2010) (Department of Energy and Climate Change (DECC), 2015a)



3.4.2 Greenhouse Gas Emissions in the United Kingdom

The United Kingdom's greenhouse gas emissions, as reported by the Department of Energy and Climate Change (DECC) are displayed in Figure 3.9 for 1990 – 2010 (Department of Energy and Climate Change (DECC), 2014a). In this figure, one will see an overall downward trend for greenhouse gas emissions that are weighted for their global warming potential. However, hydrofluorocarbon (HFC) emissions have actually been increasing since the late 1990s though they are still below their 1997 peak.

The values displayed in Figure 3.9 for each of the individual type of air emissions includes the land use, land-use change, and forestry sector (LULUCF) over the UK and Crown Dependencies. However, they exclude UK Overseas Territories. Of note here is that the Kyoto greenhouse gas basket line is not merely a sum of the individual pollutant emissions levels. Rather, this line includes three distinctions compared to the individual pollutant emissions, namely (Department of Energy and Climate Change (DECC), 2014b, 2015b):

- 1. a narrower definition for what is included in the LULUCF sector
- 2. the inclusion of the U.K. Overseas Territories²²
- the inclusion of emissions from flights between the United Kingdom, U.K. Crown Dependencies²³, and U.K. Overseas Territories

²² Anguilla Montserrat Bermuda Pitcairn Island British Antarctic Territory St Helena British Indian Ocean Territory St Helena dependencies (Ascension Island, British Virgin islands Tristan da Cunha) Cayman Islands South Georgia and the South Islands Falkland Islands Turks and Caicos Islands Gibraltar

²³ Isle of Man, Bailiwicks of Jersey and Guernsey

Furthermore, these time series data are not held static each year. Rather, they are updated as emissions accounting methodologies improve over time. As a result, emissions levels could appear different depending on the year in which those data were published.





In 2010, the base year used in this research project's modelling efforts, 97% of total carbon dioxide emissions and 82% of total greenhouse gas emissions in the United Kingdom came from fossil fuel combustion processes (International Energy Agency (IEA), 2012).

3.4.3 Air Pollution Emissions in the United Kingdom

Annual emissions of select local air pollutants in the United Kingdom are reported by the National Atmospheric Emissions Inventory (NAEI) for the Department for Environment, Food and Rural Affairs (DEFRA), the Scottish Government, the Welsh Assembly Government, and the Northern Ireland Department of Environment as well as an array of local authorities and European organizations. The National Atmospheric Emissions Inventory is funded by the Department for Energy and Climate Change (DECC), Department for Environment, Food, and Rural Affairs (DEFRA), Scottish Government, Welsh Government, and Northern Ireland Department of Agriculture, Environment and Rural Affairs. It is developed and maintained by consultants including Ricardo Energy & Environment, Aether, CEH, and Gluckman Consulting.

The most recent total emissions inventory was released in 2014 and includes data through 2012. Annual particulate matter emissions (PM_{10} and $PM_{2.5}$) by sector are displayed in Figures 3.10 and 3.11. Noted here is the fact that these two air pollutants have been declining consistently since 1990 in absolute (total tonnes per year) terms. A map produced by the United Kingdom's Department for Environment, Food and Rural Affairs (Defra) that shows the spatial distribution of sources of particulate matter (PM_{10}) pollution in the United Kingdom is shown in Figure 3.12.







Figure 3.11: Annual emissions of particulate matter $(PM_{2.5})$ by sector, 1970-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)

PM2.5 Emissions in the United Kingdom, 1970 - 2014

Figure 3.12: Map of annual emissions of particulate matter (PM_{2.5}) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016d)



Crown copyright. All rights reserved Defra, Licence number 100022861 [2016].

Annual emissions of non-methane volatile organic compounds (NMVOCs), sulphur oxides $(SO_x \text{ as } SO_2)$, nitrogen oxides $(NO_x \text{ as } NO_2)$, and ammonia (NH_3) as reported in the National Atmospheric Emissions Inventory are displayed in Figures 3.13 thru 3.16. Noted here is that all data in these figures are displayed for 1970-2014 except for ammonia, which is displayed for 1980-2014 as data from the 1970s are not reported in the National Atmospheric Emissions Inventory Database for this pollutant.

Figure 3.13: Annual Emissions of NMVOCs by sector, 1970-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)



Figure 3.14: Annual Emissions of SO_x by sector, 1970-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)







Figure 3.16: Annual Emissions of NH₃ by sector, 1980-2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016f)



Maps produced by the United Kingdom's Department for Environment, Food and Rural Affairs (Defra) that shows the spatial distribution of sources of non-methane volatile organic compounds (NMVOCs), nitrogen oxide (NO_x as NO₂), sulphur oxides (SO_x as SO₂), and ammonia (NH₃) air pollution in the United Kingdom are shown in Figures 3.17 - 3.20.

Figure 3.17: Map of annual emissions of non-methane volatile organic compounds (NMVOCs) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016c)



Figure 3.18: Map of annual emissions of nitrogen oxides (NO_x as NO₂) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016b)



127

Figure 3.19: Map of annual emissions of sulphur oxides (SO_x as SO₂) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016e)



Crown copyright. All rights reserved Defra, Licence number 100022861 [2016].

Figure 3.20: Map of annual emissions of ammonia (NH₃) by source location, 2014 (Department for Environment Food and Rural Affairs (DEFRA), 2016a)



* Some Important sources are limited to 5x5km resolution © Crown copyright. All rights reserved Defra, Licence number 100022861 [2016].

With the exception of ammonia, all of these air pollutants have been declining in terms of total annual emissions at a national level since approximately 1990. But there is still work to be done. For example, in 2015 the United Kingdom's Supreme Court ruled that the government must take action to reduce air pollution levels to meet European Air Quality Directive limits for outdoor air pollution, which it currently violates (United Kingdom Supreme Court, 2015; Carrington, 2016).

3.4.4 Climate Change Legislation

The United Kingdom is both a signatory to the United Nations Framework Convention on Climate Change (UNFCCC) treaty adopted in 1992 and a party to the Kyoto Protocol adopted in 1997 (International Energy Agency (IEA), 2012). The former is an international treaty that focuses on mobilising international efforts in climate change mitigation and adaptation. The latter is an international agreement which commits countries to binding greenhouse gas emission reduction targets, which entered into force in 2005 (United Nations Framework Convention on Climate Change (UNFCCC), 2016).

At the heart of the United Kingdom's domestic climate change policy is the Climate Change Act 2008, which requires that the nation (United Kingdom Parliament, 2008; Department of Energy and Climate Change (DECC), 2009):

- cut its greenhouse gas emissions by at least 34% by 2020 and 80% by 2050 compared to 1990 levels
- set and meet five-year carbon budgets during this period of time, with carbon budgets being set three periods (i.e. 15 years) ahead of the present day to provide clarity on the near-term emissions reduction path.

Most recently, the UK Government set out the 5th carbon budget (2028 – 2032) in late July 2016 based on guidance published by the Committee on Climate Change in 2015 (Committee on Climate Change, 2015b; Department for Business Energy and Industrial Strategy (BEIS), 2016; Department of Energy and Climate Change (DECC), 2016b). The government has also instituted a number of market reforms and infrastructure planning acts to support the energy transition required by the Climate Change Act 2008 as well as investing significantly in communication and consultation to improve public awareness (United Kingdom Parliament, 2008; International Energy Agency (IEA), 2012). The U.K. Climate Change Act 2008 is discussed elsewhere in this thesis in more detail.

3.4.5 Air Quality Legislation

As discussed elsewhere in this thesis, the United Kingdom has adopted a number of air quality objectives at the national, regional, and local levels for pollutants including particular matter (PM), nitrogen dioxide (NO₂), and sulphur dioxide (SO₂) (Department for Environment Food and Rural Affairs (DEFRA), 2013a). The nation is also subject to several directives at the European (EU) level, including the National Emissions Ceilings Directive (2001/81/EC) and the EU Air Quality Directive (2008/50/EC) and its legally binding limits on outdoor air pollution levels. However, these obligations could change now that the United Kingdom has formally indicated its intention to withdraw from the European Union after the referendum vote²⁴ that took place in June 2016 in the United Kingdom.

²⁴ often referred to as the "Brexit" vote

3.5 Sustainability Gap (SGAP) Calculations

As discussed in the methodology chapter of this thesis in Section 3.2, the Sustainability Gap (SGAP) methodology was originally developed by Ekins and Simon at the turn of the century to measure progress toward (or away from) a set of sustainability standards, including targets for air pollution (Ekins and Simon, 2001). The SGAP method was subsequently expanded in 2014 to include sustainability targets for human health (Lott, Ekins and Davies, 2014). This methodology is applied in this research to understand historical trends in key air pollutants in order to identify pollutants of interest to explore further.

Current policy targets for air pollution in the United Kingdom include those set on the regional and national scales (Beevers *et al.*, 2012; Lott, Ekins and Davies, 2014). The existing sustainability targets and standards (where applicable) and resulting SGAP targets are found in Table 3.2 and calculated YtS values are found in Table 3.3 (Lott, Ekins and Davies, 2014).

Table 3.2: Air pollution standards and targets for SGAP calculations (World Health Organization, 2005; Committee on the Medical Effects of Air Pollution (COMEAP), 2011; Department for Environment Food and Rural Affairs (DEFRA), 2013c; Lott, Ekins and Davies, 2014)^{25,26}

Air pollutant	Policy Targets (Sustainability Targets)	Sustainability Standards	SGAP Target
CO ₂ -eq	UK Climate Change Act 2008: 153 million tonnes/year (2050) Kyoto Protocol/UNFCCC: 80% reduction compared to 1990 baseline - 156 million tonnes/year by 2050	OECD/IEA: 0 tonnes/year by 2100 from the energy sector	156 million tonnes
Lead (Pb)	EU Air Quality Directive: 0.5 µg/m ³ (annual mean) DEFRA/Air Quality Strategy: 0.25 µg/m ³ (annual mean)	0 tonnes/year	0 tonnes
NO _x (NO and NO ₂)	Gothenburg Protocol: 707 ktonnes (2050) EU Air Quality Directive -DEFRA/Air Quality Strategy – NO ₂ : 200 μ g/m ³ (1-hour mean) not exceeded more than 18 times/year, 40 μ g/m ³ (annual mean) EU Air Quality Directive -DEFRA/Air Quality Strategy – NO _x : 30 μ g/m ³ (annual mean) National Emissions Ceiling Directive (NECD): 1167 kilotonnes/year by 2010	WHO guidelines (2005) for NO ₂ : 40 μg/m ³ annual mean, 200 μg/m ³ 1-hour mean	707 thousand tonnes 0 instances of violation
O ₃	EU Air Quality Directive -DEFRA/Air Quality Strategy: $100 \mu g/m^3$ (8-hour mean) not to be exceeded more than 10 times per year)	Set to NMVOCs and NO_x emissions	0 instances of violation
PM ₁₀	EU Air Quality Directive -DEFRA/Air Quality Strategy: $40 \mu g/m^3$ (annual mean), $50 \mu g/m^3$ (24 hour mean) not to be exceeded more than 35 times per year.	WHO guidelines (2005): 20 μg/m ³ annual mean, 50 μg/m ³ 24-hour mean	0 tonnes 0 instances of violation
PM _{2.5} (including black carbon)	Gothenburg Protocol: 59 ktonnes (2050) EU Air Quality Directive -DEFRA/Air Quality Strategy (annual mean): 25 µg/m ³ for UK (except Scotland), 12 µg/m ³ for Scotland, 15% reduction in concentrations at urban background	WHO guidelines (2005): 10 μg/m ³ annual mean, 25 μg/m ³ 24-hour mean	0 tonnes 0 instances of violation
SO _x (as SO ₂)	National Emissions Ceilings Directive (NECD): 585 kt Gothenburg Protocol: 282 kt	282 kt	0 tonnes
Ammonia (NH ₃)	National Emission Ceilings Directive (NECD): 297 kt Gothenburg Protocol: 283 kt	283 kt	0 tonnes

million tonnes CO2-e excluding LULUCF) to reach an 80% reduction compared to 1990 levels

²⁵ CO₂-eq includes CO₂, CH₄, N₂O, and Fluorinated gases (HFCs, PFCs, SH₆, NF₃)

²⁶ The UK Climate Change Act 2008 policy target number was calculated using 1990 baseline numbers (767.3

Table 3.3: SGAP and YtS indicator for key air pollutants (total annual emissions level targets) (Lott, Ekins and Davies, 2014)

Air	X	Emissions	У	Emissions	SGAP	SGAP	YtS
pollutant	(year)	level (el) _x	(year)	level (el) _y	Target	(million	(years)
		(million		(million	(million	tonnes)	
		tonnes)		tonnes)	tonnes)		
CO ₂ -eq	1990	591	2012	474	156	318	36
Lead (Pb)	1990	0.00289	2008	0.000067	0	0	1
PM _{2.5}	1990	0.21	2012	0.08	0	0.08	8
(including							
black							
carbon)							
PM ₁₀	1990	0.27	2012	0.11	0	0.11	16
NO _X	1990	2.9	2012	1.1	0.707	0.39	3
(NO and							
NO ₂)							
SO _X	1990	3.7	2012	0.44	0	0.44	3
(as SO ₂)							
NH ₃	1990	0.32	2012	0.28	0	0.28	125

There are many reasons why the YtS values displayed in Table 4.2 could be overly optimistic as previously discussed, including:

- in the case of pollutants that have already seen dramatic reductions, assuming the historic trends moving into the future could be unrealistic.
- for those pollutants with sustainability standards equal to zero, the YtS values imply that these pollutants could be completely eliminated, which might not be practical.
- for those SGAPs that are policy (not sustainability) targets, the YtS metric is likely significantly larger.

As a result of this analysis, air pollution emissions of CO_2 -equivalent, particulate matter (PM_{10} and $PM_{2.5}$), NO_x , SO_x , and NH_3 are initially included in this analysis. Furthermore, non-

methane volatile organic compounds (NMVOCs) are included due to their role in the formation of tropospheric ozone.

3.6 UKTM-UCL-AQ

This section provides an overview of UKTM-UCL-AQ and the set of six (6) scenarios that are used to explore the impacts of incorporating non-greenhouse gas air pollution in UK decarbonisation strategies in Chapter 4 (Lott *et al.*, 2016; Lott, Pye and Dodds, 2017). Particular assumptions related to the implementation of UKTM-UCL-AQ discussed in this chapter with further details included in the Appendix.

3.6.1 Development Process

The UKTM-UCL and UKTM-UCL-AQ models are built using The Integrated MARKAL-EFOM System (TIMES) model generator. The first of these two models, UKTM-UCL, was developed by researchers within the UCL Energy Institute over the period of 2012-2015 and was preceded by the UK MARKAL model (Dodds, 2014; Dodds, Keppo and Strachan, 2014). As mentioned at the beginning of this thesis, UKTM-UCL-AQ was developed in 2015 by researchers at University College London – namely, Birgit Fais, Melissa C. Lott²⁷, Steve Pye, and Paul Dodds – in collaboration with colleagues from Aether, an environmental consultancy company based in the United Kingdom, as a part of a project funded by the former U.K. Department of Energy and Climate Change, which now operates within the Department of Business, Energy and Industrial Strategy. Both of these models use the VErsatile Data Analyst

²⁷ For this project, Melissa C. Lott completed all UKTM-UCL model development for the transport sector as well as a large portion of the electricity sector. She provided input for all other sectors, but was not the lead researcher.

(VEDA) model management software platform for data handling in the manner depicted in Figure 3.21 (Energy Technology Systems Analysis Program (ETSAP), 2016).

Figure 3.21: Overview of the VEDA system for TIMES modelling (Energy Technology Systems Analysis Program (ETSAP), 2016)



This approach to building an energy system model results in a partial equilibrium energy system for the United Kingdom that is technically detailed and suitable for investigating the economic and technological trade-offs of energy scenarios to 2100, though the analyses presented in this thesis were limited to a 2050-time horizon²⁸.

²⁸ Noted here is that longer time horizons have been explored on a limited basis for the United Kingdom. In

^{2015,} Pye et. al. explored potential pathways to complete energy sector decarbonisation with a time horizon of 2100 as a part of the Deep Decarbonisation Pathways Project (DDPP) (Pye *et al.*, 2015).

These models take a bottom-up view of the energy system and energy transitions. For UKTM-UCL-AQ, an air pollutant emissions factor database is incorporated into UKTM-UCL for six (6) air quality pollutants:

- 1. particulate matter that is less than 10 micrometres in diameter (PM_{10})
- 2. particulate matter that is less than 2.5 micrometres in diameter (PM_{2.5})
- 3. nitrogen oxides (NO_x as NO_2)
- 4. sulphur dioxide (SO_x as SO_2)
- 5. ammonia (NH₃)
- 6. non-methane volatile organic compounds (NMVOCs)

This update allows air pollution emissions accounting by year out to 2050 with the model defaulted to five-year time slices, though this level of temporal granularity can be adjusted (Lott, Pye and Dodds, 2017).. This model variant was developed in the fall of 2015 by researchers at the University College London and colleagues from Aether in a project funded by the U.K. Department of Energy and Climate Change. The author of this thesis manuscript completed all TIMES model update work related to the transport sector (including development and implementation) as well as a significant portion of the electricity sector. She provided input to work for all other sectors, but did not directly implement these model updates. Figure 3.22 contains the flow diagram for UKTM-UCL-AQ to compliment the generalised schematic presented in Figure 3.4.

Figure 3.22: Analysis Flow Diagram UKTM-UCL-AQ



3.6.1.1 Emission Factors

In this development process, emission factors (EFs) for the current energy system were compiled for a large number of sectors from the United Kingdom's National Atmospheric Emissions Inventory (NAEI)²⁹ using the 2014 dataset. However, some of the National Atmospheric Emissions Inventory emission factors were confidential due to commercial sensitivity and other emission factors did not directly match the UKTM fuels and technologies.

²⁹ Emission factors (EFs) were mapped from the National Atmospheric Emissions Inventory (NAEI), published online at <u>http://naei.defra.gov.uk</u> (accessed November 2015), which provides the official annual air quality pollutant emission estimates for the United Kingdom. The inventory is structured around reporting under the United Nations Economic Commission for Europe (UNECE) Convention on Long Range Transboundary Air Pollution (CLRTAP) and emission estimates are presented in Nomenclature for Reporting (NFR) format. A full list of the emission factors used in this research are included in the Appendix.

In these cases, the closest match in the NAEI was used or alternative data sources were identified and documented in consultation with experts.

Data from the NAEI was appropriate for use in this research as it provided a transparent and accessible centralised source of emission factors data. Furthermore, its use allowed for comparisons with previous work done in this area by Pye et. al. in 2008 (Pye and Palmer, 2008; Pye *et al.*, 2008). However, there are certainly ample opportunities for future work to examine these emission factors and the uncertainty in the published figures. Furthermore, as the NAEI database is updates on an annual basis, it would be valuable to examine how updated information on these emission factors in the future impact the outputs in this thesis (Department for Business Energy & Industrial Strategy (BEIS), 2016).

As discussed in more detail elsewhere in this Chapter, fuel-based emission factors (EFs) were used for all sectors, with the exception of domestic transport, which used activity-based factors. Fuel-based factors account for emissions based on the amount of fuel that is burned (e.g. grams emitted per PJ) versus activity-based factors that are structured around the activity undertaken (e.g. grams per mile travelled). Activity-based factors are more appropriate for transport as they account for non-tailpipe emissions – including tyre, brake, and road wear – as well as approved European Union Standards (e.g. Euro VI standards for road vehicles) that would be ignored using a fuel-based EF.

Noted here is that these activity-based EFs are based on test cycle emissions as opposed to real world, which could have important implications on the output emissions levels and corresponding policy recommendations. There have been notable scandals recently showing how different real-world versus test-cycle emissions can be, which leads to questions as to the

validity of values in current databases. For example, in 2015, the U.S. Environmental Protection Agency (EPA) found that many diesel vehicles produced by Volkswagen included software that changed the vehicle's performance during emissions testing. All told, this software meant that vehicles could emit up to 40 times the U.S.'s legal limit for nitrogen oxide pollution. This "defeat device" was included in about 11 million cars sold worldwide, including 8 million in Europe (Hotten, 2015). In 2016, Mitsubishi admitted to cheating on its fuel economy tests for more than two decades, which has significant implications for the corresponding emissions factors for all combustion products from the affected vehicles (Soble, 2016). Other research, including a study by Brand published in 2016, has explored the implications of these unaccounted and future air pollutant emissions and energy use for cars in the United Kingdom (Brand, 2016). That being said, test cycle emission factors were still used for the purpose of this research in order to allow for calibration with other air pollution research in the United Kingdom and provide consistency.

For the transport sector, hot exhaust emissions as well as non-tailpipe emissions from tyre wear, brake wear, and road abrasion were included for all road transport technologies. Cold start emissions and evaporative emissions were not included for these technologies because a detailed transport emission model would be needed for proper accounting. These emissions make up about 10% of NO_x emissions from cars and 5% of LGV NO_x emissions. For shipping and aviation, emission factors were calculated by taking the total emissions for each pollutant from the National Atmospheric Emissions Inventory for these sectors and dividing it by the corresponding activity values in UKTM-UCL-AQ for the base year.

For the fuel-based EFs used for all non-transport emissions, it was assumed that pollution levels would be most impacted by changes in the efficiency of fuel use – which could arise from

139

technology changes - and shifts in total fuel demand. When modelling on a time horizon to 2050, there are a range of new technologies, not currently in the system, for which emissions information therefore does not exist. Such technologies include carbon capture and storage (CCS), for which some estimates have been made by organizations including the European Environment Agency (European Environment Agency, 2011). For hydrogen and biofuel production, no emission factors are assumed due to the absence of data estimates (i.e. they are set at zero) (Lott, Pye and Dodds, 2017). For alternative fuel vehicles, additional information published by the National Atmospheric Emissions Inventory is used along with their published values for non-tailpipe emission factors used in UKTM-UCL-AQ can be found in the Appendix of this thesis.

3.6.2 Analysis of Emissions Accounting Coverage

A post-mapping evaluation reveals the extent to which the UKTM accounted for these six (6) air pollutants, since the model only represents the energy system, while significant emissions of specific pollutants come from other parts of the economy (Lott, Pye and Dodds, 2017). All told, a majority of NO_x, SO_x and PM (both PM₁₀ and PM_{2.5}) air pollution were represented in UKTM-UCL-AQ in 2010, with NO_x and SO_x having the most complete coverage as shown in Table 3.4. Conversely, sectoral coverage of NH₃ and NMVOC emissions is limited, representing an opportunity for future model development. For the air pollution emissions that were included in UKTM-UCL-AQ, a validation exercise was undertaken to compare the UKTM-UCL-AQ 2010 base year against the corresponding National Atmospheric Emissions Inventory sector totals, with the objective to be within 10-15% difference for total emission values.

	Type of Air Pollution					
	NOx (as NO ₂)	NMVOC	SOx (as SO ₂)	NH3	PM _{2.5}	PM ₁₀
% of NAEI inventory						
mapped in UKTM	94%	15%	92%	5%	74%	58%

Table 3.4: Air pollution inventory mapping between NAEI and UKTM-UCL-AQ (Lott *et al.*, 2016; Lott, Pye and Dodds, 2017)

In the case of particulate matter, the majority of PM_{10} emissions that are not included are from agricultural sources (livestock and crops) as well as mining and quarrying. A more detailed breakdown of the sources of these excluded emissions is shown in Table 3.5.

Table 3.5: Particulate matter emissions that were excluded from UKTM-UCL-AQ in the 2010 emissions calibration, by sector (Lott *et al.*, 2016; Lott, Pye and Dodds, 2017)

Sector	PM ₁₀	PM _{2.5}
Mining and quarrying	5%	1%
Iron and Steel process	3%	3%
Road Paving	3%	2%
Off road combustion	3%	4%
Waste open burning	1%	2%
Livestock	14%	4%
Crops	4%	1%
Fugitives (exploration and production of fossil fuels)	2%	2%
Other (including glass and other mineral products)	7%	7%
Excluded from UKTM	42 [%]	26%

For NMVOC and NH₃, emissions are dominated by sources not adequately characterised in UKTM-UCL-AQ including solvents, fugitive emissions and emissions from the agricultural sector (e.g. from manure) (Lott *et al.*, 2016; Lott, Pye and Dodds, 2017).

3.6.3 Damage Cost Database

In the United Kingdom, two broad methods have been used to estimate the cost of air pollution – a detailed "impact pathway" and a simpler "damage cost" approach (Miller and Hurley, 2010; Her Majesty's Treasury, 2013). As discussed in this chapter in more detail, the impact pathway approach requires detailed emission, air quality modelling and health impact assessments and is therefore resource intensive. The damage costs approach uses the output of impact pathway studies to quantify the monetary impact of changes per unit of pollutant emitted (Department for Environment Food and Rural Affairs (DEFRA), 2013d; Walton *et al.*, 2015). These damage costs are a more direct way to place an economic value on the impacts of air pollution on both public health and the environment (including both buildings and materials) in UKTM-UCL-AQ, and therefore are more straightforward to include in the optimization process.

Crucially, the damage costs approach does factor in the spatial distribution of air pollution and the likely exposure. It is therefore appropriate to use such nationally-derived damage costs values in a model such as UKTM-UCL-AQ. While recognised as a credible approach for policy appraisal, the limitation in using these values is the implicit assumption that such damage cost values hold for future years, in which this spatial distribution of pollution–exposure–impact may change.

The damage cost values that are used in UKTM-UCL-AQ and within the scenarios presented here were developed by the UK Department for Environment, Food and Rural Affairs (DEFRA) and are shown in Table 3.6. All values represent the cost impact of a change in pollution by one tonne in a given year ("annual pulse damage costs"). These damage cost values include the air pollution impacts of particulate matters (PM₁₀ and PM_{2.5}) on health, including both chronic mortality and morbidity effects as well as building soiling impacts. For nitrogen oxides (NO_x), these values include the health impacts of secondary particulate matter resulting from NO_x emissions but does not include the health impacts of ozone formation as the result of NO_x emissions. The sulphur oxide (SO_x) damage costs include this secondary PM formation and impacts of SO₂ on health and building materials. For ammonia (NH₃), these costs include the health impacts of secondary particulate matter formation (Department for Environment Food and Rural Affairs (DEFRA), 2011).

In the case of particulate matter and nitrogen oxide air pollution, the damage cost values are more disaggregated to reflect the relative impact of pollution source on the population and surrounding built environment (e.g. particulate matter from power plant stacks in rural areas versus cars travelling at ground level on roads). Damage costs are not included for non-methane volatile organic compounds (NMVOCs), as DEFRA does not publish these values. In turn, this type of pollution is inventoried, but is not included in the cost-optimisation process in UKTM-UCL-AQ. For the work presented in this thesis, the "Central" annual pulse damage costs from Table 3.6 were applied.

		Annual Pulse Damage Costs (GBP per tonne - 2010 prices)			
Air Pollutant	Sector				
		Low	High	Central	
	Electricity supplies industries (ESI)	£2,072	£3,007	£2,645	
	Domestic	£24,029	£34,875	£30,690	
PM ₁₀	Agriculture	£8,287	£12,026	£10,583	
	Industrial	£21,543	£31,267	£27,515	
	Waste	£17,815	£25,856	£22,753	
	Transport	£41,429	£60,129	£52,913	
Air Pollutant PM ₁₀ NO _x (as NO ₂) SO _x (as SO ₂) NH ₃ NMVOCs	Electricity supplies industries (ESI)	£383	£1,533	£958	
	Domestic	£4,444	£17,778	£11,111	
	Agriculture	£1,532	£6,130	£3,832	
	Industrial	£3,984	£15,938	£9,962	
	Waste	£3,294	£13,180	£8,238	
	Transport	£7,662	£30,651	£19,157	
SO _x (as SO ₂)		£1,439	£2,025	£1,781	
NH ₃		£1,678	£2,444	£2,151	
NMVOCs		None	None	None	

Table 3.6: Damage costs by sector and subsector (modified from (Department for Environment Food and Rural Affairs (DEFRA), 2011))³⁰

When these damage costs are excluded from individual scenarios, the model simply accounts (i.e. inventories) the total emission levels for each of these air pollutants. In turn, air pollution emissions do not directly affect the model solution.

 $^{^{30}}$ In the model implementation phase, damage cost values were adjusted over time with a 2% per annum lift rate to take into account willingness to pay.
When air pollution damage costs are included, these costs are considered in the optimisation process and so will influence energy technology choices. This functionality allows for analysis related to the impacts of including these costs on scenario outputs. In the implementation stage, these costs are included as additional operating costs that are either fuel- or activity-based, depending on the sector as previously discussed.

3.7 Demand Drivers & Technology Cost Assumptions

Energy flows for the base year (i.e. 2010) are calibrated to match the data found in the Digest of United Kingdom Energy Statistics (DUKES³¹) as published in 2011. Demand for energy services and the cost of energy system technologies are defined exogenously to the UKTM-UCL-AQ model. All input values for the analyses presented in Chapter 4 parallel those used in previous analyses using UKTM-UCL (Committee on Climate Change, 2015a; Pye *et al.*, 2015).

3.8 Scenarios for the United Kingdom

The scenarios presented in Chapter 4 for the United Kingdom cover a range of policy ambitions for both decarbonisation and air pollution. The former was mapped along a range of three scenarios and across the spectrum from 1) where decarbonisation is no longer a priority to 2) where current decarbonisation goals are met. For air pollution, two potential scenarios are considered – the first where non-greenhouse gas air pollution is ignored in the energy system's development and the second where the damage costs associated with these pollutants are included in the development of the energy system. More specifically, these damage cost values are included in the cost-optimisation process for UKTM-UCL-AQ and, in turn, have a direct impact on the model solution. This range of policy ambition for decarbonisation and consideration of air pollution damage costs is displayed in Figure 3.23.

³¹ https://www.gov.uk/government/collections/digest-of-uk-energy-statistics-dukes

Figure 3.23: Scenarios Map for United Kingdom Analysis

ion Costs	With Damage Costs	No decarbonisation goals. Attention paid to non-GHG air pollution costs.	Intermediate level of attention to decarbonisation. Attention paid to non-GHG air pollution costs.	Current decarbonisation goals achieved. Attention paid to non-GHG air pollution costs.
Air Pollut	No Damage Costs	No decarbonisation goals. No attention to non- GHG air pollution costs.	Intermediate level of attention to decarbonisation. No attention to non-GHG air pollution costs.	Current decarbonisation goals achieved. No attention to non-GHG air pollution costs.
		none	intermediate	lowGHG

Decarbonisation Policy Ambition

3.9 The PollutION Emissions from EneRgy (PIONEER) Model

This section includes details on the creation of the PollutION Emissions from EneRgy (PIONEER) model, including its soft-linking with the UKTM-UCL-AQ model for the purpose of the urban-scale analysis presented in Chapter 5. Throughout these sections are explicit details on key assumptions made in the construction of the PIONEER model. Particular assumptions related to the implementation of this tool are discussed in Chapter 5 with additional details in the Appendix.

3.9.1 PIONEER Methodology Overview`

The PollutION Emissions from EneRgy (PIONEER) model is an accounting model designed to disaggregate urban level results from the outputs of national-scale energy systems models to enable the quantification of the air pollution and health impacts of road transport technology transitions in the energy system to 2050. PIONEER is designed specifically for the road transport sector, accounting for demands across five vehicle technology types – motorcycles,

cars, buses, light goods vehicles (LGVs), and heavy goods vehicles (HGVs) as defined by the United Kingdom Department for Transport (Department for Transport (Dft), 2014). It was created for two primary reasons – specifically, to allow for the:

- disaggregation of road transport in an urban region from the outputs of nationallevel energy systems models to enable the evaluation of targeted local interventions in these regions
- 2. accounting of the air pollution and health impacts associated with these targeted interventions

This model was designed to be particularly straightforward to use in conjunction with the UKTM-UCL-AQ model in order to investigate the health impacts associated with technology transitions in the road transport sector. However, its design allows it to be coupled with any bottom-up and technology-rich energy systems model should adequate datasets be available to account for air pollution and public health impacts for the desired analysis region.

Beyond the outputs from UKTM-UCL-AQ, other inputs to PIONEER include demand disaggregation factors (e.g. DfT forecasts, population projections) that are used to separate demand for road transport in Greater London from the total demand values for the United Kingdom. Scenario constraints include technological and behavioural change factors that can impact either or both the technological change pathway or the demand values for Greater London (e.g. technology mandates, degree of modal shifting).

The outputs from the PIONEER are:

- disaggregated road transport demand (billion vehicle kilometres) by technology type (e.g. cars, LGVs) for Greater London
- air pollution emissions levels (kilotonnes) from the road transport sector by vehicle type from 2010-2050
- estimated public health impacts in Greater London resulting from air pollution emissions (PM_{2.5} and NO_x) produced by the local road transport sector

Figure 3.24 contains the flow diagram for the soft-linked PIONEER and UKTM-UCL-AQ models. As shown in this Figure, the outputs of UKTM-UCL-AQ for the transport sector are inputted into PIONEER and then the transport sector is disaggregated into two regions, namely 1) Greater London and 2) Rest-of-UK. When required, the outputs from PIONEER are subsequently used as the inputs to UKTM-UCL-AQ, ensuring harmonization of the two models and allowing the modeller to understand the impacts on the technology transition pathway in the United Kingdom of changes in the transport system that are exclusively applied in Greater London. Practically speaking, the full process, including the "loop" between UKTM-UCL-AQ and PIONEER, is completed with the following steps:

- 1. Run UKTM-UCL-AQ for the desired national-scale scenario
- 2. Use the outputs of UKTM-UCL-AQ for road transport as inputs to PIONEER
- 3. Define and input any additional constraints (e.g. demand reduction measures, technology mandates) for Greater London to be used in the PIONEER-produced scenarios
- 4. Run PIONEER

- 5. Use PIONEER's outputs as inputs to UKTM-UCL-AQ as appropriate (e.g. changes to demand over time due to targeted behavioural change)
- 6. Use the outputs from UKTM-UCL-AQ as inputs to PIONEER to verify convergence of the models.
- 7. Repeat this process as necessary...

Figure 3.24: Analysis Flow Diagram for Soft-linked PIONEER and UKTM-UCL-AQ Models



As discussed in more detail in Chapters 4 and 5, because of the limited impacts to the transport sector that are seen with the national level decarbonisation pathway, the analysis undertaken specifically for the Greater London transport sector is framed along two additional dimensions – technological and behavioural change – in order to more fully explore the "what ifs" and possibilities of targeted local action in this urban region. These dimensions are shown in Figure 3.25 and map a range of potential futures resulting from targeted action to support technological and/or behavioural change. Justification for choosing a range of 0-40% modal

shifting away from cars is discussed elsewhere in this chapter. The technological change pathways that result in zero tailpipe emissions from cars by 2050 are discussed in more detail in Chapter 5.

nge	40% Mode Shift away from cars	NoChange_60:40	<u>UK_60:40</u>	<u>50:50_60:40</u>	Doubling_60:40	<u>CleanLondon_60:40</u>	<u>JustInTime_60:40</u>
vioural Char	20% Mode Shift away from cars	NoChange_80:20	<u>UK_80:20</u>	<u>50:50_80:20</u>	Doubling_80:20	<u>CleanLondon 60:40</u>	JustInTime_80:20
Behav	No Mode Shift	NoChange_NoChange	<u>UK_NoChange</u>	50:50_NoChange	Doubling_NoChange	<u>CleanLondon_NoChange</u>	JustInTime_NoChange
		none	follows U.K. for lowGHG_DAMC	4 pc	No-Tailpipe Emissio Athways to zero tailpi	ons from cars by 2050 pe emission car deploym	ent

Figure 3.25 Scenario Map for Greater London Analysis



3.9.1.1 Disaggregation of Demand in PIONEER

As with UKTM-UCL-AQ, the PIONEER model uses population projections to calculate demand over time by transport type. First, per capita demand (PerCapDem) is calculated for Greater London using the user inputs for both population (Pop) and total demand (Dem) in the base year for each vehicle technology type (tech) as shown in Equation 3. As previously stated, both of these values are user inputs (i.e. exogenous) to PIONEER. The "0" in the equation below indicates that this calculation is completed for the base year.

$$PerCapDem_{GreaterLondon,0,tech} = \frac{Dem_{GreaterLondon,0,tech}}{Population_{GreaterLondon,0,tech}}$$

Eq. 3

Subsequently, this per capita demand value is used in conjunction with population projections for Greater London using a subset of the same ONS statistics that are also used in UKTM-UCL-AQ to calculate road transport demand (Office of National Statistics, 2010). This calculation is completed by multiplying the base year per capita demand value (Equation 3) by the projected population for each time slice (y) and technology type (tech). This equation is shown in Equation 4.

*Demand*_{GreaterLondon,tech,y}

$$= (PerCapDemand_{GreaterLondon,tech,0}) \times (Population_{GreaterLondon,y})$$
Eq. 4

This demand value for Greater London by year is subsequently used to calculate the demand for the Rest-of-UK region by subtracting it from the total UK demand projection using the same process using Equation 6. An internal check is then performed by PIONEER to ensure that the sum of the values totals the demand projection output values provided by UKTM-UCL-AQ.

$$Demand_{Rest-of-UK,tech,y} = Demand_{UK,y} - Demand_{GreaterLondon,y}$$

Eq. 6

For the alternative case where demand in the United Kingdom or Greater London is assumed to follow a demand projection that considers other demand drivers beyond changes population, these values are taken directly from the projections. They are subsequently checked to ensure consistency with the other model. As a default, demands are disaggregated in PIONEER in five-year time slices to match the outputs from UKTM-UCL-AQ, including 2010 to 2050. Linear interpolation is used for any analysis conducted that requires smaller time slides. Should UKTM-UCL-AQ be modified to use a higher number of time slices (e.g. annual, biannual) and assuming appropriate data availability, the PIONEER system can be modified to accommodate this change in temporal resolution.

3.9.1.2 Linking UKTM-UCL-AQ Outputs to PIONEER

Emission factors for each type of air pollution (pollutant) and technology type (e.g. cars, LGVs) resulting from the technology transition pathway outputted by UKTM-UCL-AQ are calculated directly in PIONEER. This calculation is completed using outputs from UKTM-UCL-AQ for the total emissions (Emiss) for each type of pollutant (pollutant) by technology type and corresponding demand values (Dem). These values are used to calculate a set aggregate emission factor for each road transport technology type (tech) as shown in Equation 7. This approach simultaneously ensures consistency in the emission factors used in UKTM-UCL-AQ and PIONEER while creating a convenient opportunity for a simple but important data quality check. Emission factors (EFs) are calculated in units of kilotonnes per billion vehicle kilometres (kt/bvkm).

$$EF_{GreaterLondon,pollutant,tech,y} = \frac{Emiss_{UK,pollutant,tech,y}}{Dem_{UK,tech,y}}$$

Eq. 7

Furthermore, total emissions of a given pollutant in Greater London by a type of road transport technology (e.g. cars, LGVs) is calculated using Equation 8.

 $Emiss_{GreaterLondon,tech,y} = EF_{GreaterLondon,pollutant,tech,y} \times Dem_{GreaterLondon,tech,y}$

Eq. 8

3.9.1.4 Incorporating Scenario Constraints in PIONEER

As a default, PIONEER assumes that Greater London follows the technology transition pathway defined by the outputs from UKTM-UCL-AQ with the demand levels that were previously disaggregated from United Kingdom demand. However, additional scenario constraints can be applied in PIONEER for Greater London through the definition of technological and behavioural change pathways over time. Put another way, these constraints are defined in terms of how they impact either the technology transition pathway (i.e. the aggregate emission factors for each technology type) or demand over time. For example, a scenario could be constructed to represent targeted action in the Greater London urban region to adopt a specific technology subtype (e.g. zero-tailpipe vehicles) or to reduce demand. The former would be defined in terms of the impact that this increased adoption of a technology subtype will impact the aggregate emissions factor for cars. The latter would impact the demand assumptions in both PIONEER and UKTM-UCL-AQ.

3.9.1.4.1 Scenarios with Changes to the Technology Transition Pathway

In the case where the scenario constraints impact the technological transition pathway, the user inputs the amount of demand that must be supplied by an individual technology subtype (subtech) for each year (y) included as a time slice in the defined scenario for Greater London (scenario). For example, this type of constraint is used if the modeller wishes to require that a technology subtype be adopted in a geographically diverse manner (i.e. proportionally more of less electric vehicles in Greater London versus the rest of the United Kingdom).

After these scenario constraints have been inputted, PIONEER does an internal check to ensure that the technologies are available to meet the user inputs by comparing the demand values with the outputs from UKTM-UCL-AQ. An error appears if user constraint cannot be met.

If the user constraint is valid, the user constraints are then utilized by PIONEER to calculate the impact that the inputs have on the aggregate emissions factor for the technology type and, in turn, the total emissions for Greater London over time in the scenario. This process begins with re-calculating the aggregate emissions factor for the technology type as shown in Equation 9 where values are summed over all the subtechs in the UKTM-UCL-AQ model outputs. In this equation:

- $EF_{pollutant,UK,tech,subtech,y}$: emissions factor (EF) for the technology subtype (subtech), which is taken directly from the UKTM-UCL-AQ emission factor assumptions.
- *Dem_{GreaterLondon,tech,subtech,y,scenario*: demand (Dem) for the subtype in Greater London, which is defined by the user in their scenario constraint.}
- Dem_{GreaterLondon,tech,y}: demand for the technology type (e.g. cars, LGVs) in Greater
 London, which is defined in the demand disaggregation process described elsewhere
 in this Chapter.
- $EF_{Rest-of-UK,pollutant,tech,y,scenario}$: emissions factor for the area of the United Kingdom outside of Greater London for the technology type

Dem_{Rest-of-UK,tech,subtech,y,scenario}: as shown in Equation 11, this is defined as the difference between the total demand for the individual technology subtype (e.g. electric vehicles) in the United Kingdom and the demand for the subtype in Greater London as inputted by the user in their scenario constraints.

$EF_{GreaterLondon,pollutant,tech,y,scenario}$

 $=\frac{\sum_{subtech}[(EF_{UK,pollutant,tech,subtech,y}) \times (Dem_{GreaterLondon,tech,subtech,y,scenario})]}{Dem_{GreaterLondon,tech,y}}$

Eq. 9

Correspondingly, the impact of the user constraints on emission factors outside of the Greater London area are calculated as shown in equations 10 and 11.

 $EF_{Rest-of-UK,pollutant,tech,y,scenario}$ $= \frac{\sum_{subtech} [(EF_{UK,pollutant,tech,subtech,y}) \times (Dem_{Rest-of-UK,tech,subtech,y,scenario})]}{Dem_{Rest-of-UK,tech,y}}$

Eq. 10

Dem_{Rest-of-UK},tech,subtech,y,scenario

 $= Dem_{UK,tech,subtech,y} - Dem_{GreaterLondon,tech,subtech,y,scenario}$

Eq. 11

Total emissions (Emis) per year for each year (y) for that technology type for Greater London in the scenario is then calculated in PIONEER by multiplying the scenario-definedemission factor for the technology type by the demand for that technology type in Greater London as shown in Equation 12. The same calculation is completed for the area outside of Greater London using Equation 12 by applying "Rest-of-UK" values for both the emission factors and demands which allows the modeller to examine the impact of the scenario constraints for Greater London on air pollution levels from road transport in the rest of the United Kingdom.

*Emis*_{GreaterLondon,tech}, y, scenario

 $= EF_{GreaterLondon,pollutant,tech,y,scenario} \times Dem_{GreaterLondon,tech,y,scenario}$ Eq. 12

Noted here is that, functionally, this type of scenario constraint only modifies the geographic distribution of technology deployment. It does not change any of the overarching demand or technology assumptions used in UKTM-UCL-AQ. In turn, an iterative loop is not needed between UKTM-UCL-AQ and PIONEER in this type of scenario. This is not the case for changes in overall demand values, as described below.

3.9.1.4.2 Scenarios with Behavioural Change

In the case where the scenario constraints impact the demand pathway for Greater London, the user inputs the amount of demand that will be removed or added for each time slice and technology type in Greater London. This amount of demand change is then incorporated into the demand projections for the United Kingdom using Equation 13 and holding demand in the rest of the UK constant with UKTM-UCL-AQ projections.

 $Demand_{UK,tech,y} = Demand_{GreaterLondon,tech,y} + Demand_{Rest-of-UK,tech,y}$

Eq. 13

In this case, the scenario constraint for Greater London has a direct impact on the assumptions that are used in the UKTM-UCL-AQ model for demand in the United Kingdom. In turn, UKTM-UCL-AQ is re-run using the updated demand values for each technology type. After this iterative "loop", the emission factors for Greater London are re-calculated using the process described previously in this chapter. Furthermore, if the scenario uses a combination of behaviour and technological change constraints, the latter is checked for validity as described in the previous section.

3.9.1.5 Calculating Public Health Impacts in PIONEER

The health impacts in the PollutION Emissions from EneRgy (PIONEER) model are calculated using research published in 2015 by Walton, et. al. from Kings College London, which utilises the impact pathway approach (Favarato *et al.*, 2014; Walton *et al.*, 2015). This research uses emerging techniques to assess the mortality burdens of nitrogen dioxide (NO₂) and also updates previous work on the health impacts of PM_{2.5} in London by the Committee on the Medical Effects of Air Pollution (COMEAP) and Public Health England (Miller and Hurley, 2010; Committee on the Medical Effects of Air Pollution et *al.*, 2015).

Noted here is the existence of uncertainty in the evidence associated with the mortality burdens of NO₂ and so these numbers "need to be used with care" (World Health Organization, 2013a, 2013b; Walton *et al.*, 2015). This study relied on data published in the London Atmospheric Emissions Inventory (LAEI)³² and its associated modelling efforts related to air pollution concentrations throughout the Greater London area. This inventory includes data for 32 London Boroughs, the City of London, and up to the M25 Motorway and includes detailed air

³² http://data.london.gov.uk/dataset/london-atmospheric-emissions-inventory-2010

quality and exposure modelling for the Greater London area (Walton *et al.*, 2015). The latest publication of this inventory is the LAEI 2013, which was published in 2016. The map in Figure 3.26 illustrates the geographical coverage of this inventory (Martin, 2016).



Figure 3.26: Geographies Included in the LAEI 2013 Publication, from the GLA (Martin, 2016)

As discussed elsewhere in this thesis in more detail, the total mortality burden in London in 2010 of $PM_{2.5}$ originating from human activity has been most recently estimated at 3,537 premature deaths, the equivalent of 52,630 life-years lost (Walton *et al.*, 2015). Of these early deaths, air pollution originating from outside of London is the largest contributor though London sources also contributed significantly to the health burden (Walton *et al.*, 2015).

For the same year, the total mortality burden of long-term exposure to NO_2 is estimated to be up to 5,879 premature deaths (88,113 life-years lost) (Walton *et al.*, 2015). This value assumes an up to 30% overlap between the effects of $PM_{2.5}$ and NO_2 that is recommended by the World Health Organization in order to avoid double counting (World Health Organization, 2013a, 2013b). Of these early deaths, air pollution from inside of London (including road transport) is the largest contributor (Walton *et al.*, 2015).

In these estimates, Walton et al. use the following relative risk values for (Walton et al., 2015):

- PM2.5: relative risk of 1.06 (plausibility interval 1.01 to 1.12) for changes in mortality resulting from long-term exposure to PM2.5. These values are derived from the American Cancer Society Study and were subsequently recommended for use in the UK by the Committee on the Medial Effects of Air Pollution with the plausibility interval from COMEAP (Pope *et al.*, 2002; Committee on the Medical Effects of Air Pollution (COMEAP), 2010)
- NO_x (as NO₂): relative risk of 1.039 (95% CI 1.022 1.056) for the change in mortality as a result of long-term exposure to NO₂. These values are derived from studies by Hoek et al. and are recommended for use the World Health Organization Health Risks of Air Pollution in Europe (HRAPIE) project (Hoek *et al.*, 2013; World Health Organization, 2013a).

Of note with regards to the NO_x values is that these relative risks include the 30% maximum adjustment for potential overlap with the health effects of $PM_{2.5}$ as recommended by the World Health Organization in the HRAPIE project. Without these adjustments, this relative risk increases to 1.055 (95% CI 1.031 - 1.080) (World Health Organization, 2013a; Walton *et al.*, 2015).

Combined, these negative health impacts in Greater London have an estimated economic cost of $\pounds 1.4 - 3.7$ billion (Walton *et al.*, 2015). This economic impact is estimated using London-

specific damage cost values that consider the area's particular exposure profiles (Committee on the Medical Effects of Air Pollution (COMEAP), 2011; Atkinson *et al.*, 2014; Mills *et al.*, 2015; Walton *et al.*, 2015).

Overall, of the previously discussed mortality burden estimates for London, a slight majority (52%) come from London sources of $PM_{2.5}$ and NO_2 pollution (Walton *et al.*, 2015). More specifically for road transport, an estimated 30% of total premature deaths in London result from air pollution produced by London road transport – a total of 2,825 premature deaths (42,229 life-years lost). Of these deaths, the vast majority (2,448 premature deaths) result from NO_2 pollution with the balance (377 premature deaths) (Walton *et al.*, 2015). This total value represents the total current opportunity for Greater London that could be realised in terms of targeted local action to reduce $PM_{2.5}$ and NO_2 air pollution from local road transport. Of course, action to reduce emissions in Greater London would also have impacts on areas outside of this urban area.

These values can be used without detailed air quality and exposure calculations to provide an initial estimate the health impacts of changes in levels of air pollution from road transport (road). In PIONEER, these changes are calculated using Equation 14, which assumes a linear relationship between changes in air pollution levels and the corresponding annual attributable premature death (Death) resulting from changes in emissions levels of a specific pollutants (pollutant) from Greater London road transport (road) in a particular year (y). The corresponding years of life lost (YLL) are calculated using Equation 15.

$$Death_{road,pollutant,y} = Death_{road,pollutant,2010} \times \left(\frac{Emis_{road,pollutant,y}}{Emis_{road,pollutant,2010}}\right)$$

Eq. 14

$$YLL_{road,pollutant,y} = YLL_{road,pollutant,2010} \times \left(\frac{Emis_{road,pollutant,y}}{Emis_{road,pollutant,2010}}\right)$$

Eq. 15

Per the defined scope of this research project, this process only accounts for changes in mortality levels in Greater London resulting from changes in road transport in Greater London. Of course, as discussed by Walton, et. al., the realization of the health effects resulting from changes in air pollution will be observed over an extended period of time after the initial change has occurred (Walton *et al.*, 2015). In turn, the resulting values should be viewed as the cumulative health benefit of a change in air pollution levels.

Furthermore, due to the spatial resolution of the models used in this research, this process assumes that air pollution reductions lead to evenly distributed effects on air pollution concentrations and exposure levels across the Greater London area and, in turn, mortality burdens. Furthermore, the knock-on effects of pollution level changes are assumed to be negligible in PIONEER's calculations. For example, dramatic reductions in nitrogen oxide levels could lead to increasing levels of tropospheric ozone due to knock-on effects³³. Each of

³³ This type of knock-on effect is being investigated in a related project that includes the author of this thesis and researchers from Kings College London (Williams *et al.*, 2016).

these assumptions represents an opportunity for additional future work, particularly in air quality modelling to determine the potential impacts of these types of interactions. As discussed elsewhere in this thesis, a portion of this type of work is currently being pursued as a part of a collaborative project between the author of this thesis and researchers at Kings College London.

3.9.2 Mode Shifting Potential in Greater London

The ability to shift personal car use to other forms of travel (e.g. walking, cycling, public transport) has been discussed in this Greater London context via work by both Transport for London in the 2010 report on cycling potential for Greater London and by Pye and Daly in their 2015 paper that focuses on modelling mode shifting in urban areas in the United Kingdom (Transport for London (TfL), 2010; Pye and Daly, 2015).

In the Transport for London report, analysts reviewed data from the London Travel Demand Survey (LTDS) to estimate the extent to which trips made by other modes of travel could be replaced with cycling (Transport for London (TfL), 2010). When identifying trips that could potentially be made by bike, these analysts excluded all trips over 8 kilometres (~ 5 miles) or those trips that would take at least 20% more time if cycled. They also excluded trips that were:

- Already walked or cycled;
- Made by young children, elderly and/or disabled people;
- Made between 8pm and 6am; or
- Carrying heavy or bulky goods.

According to London Travel Demand Survey data, the average distance travelled per day by London residents is 15 kilometres (~9.3 miles) as the "crow-fl[ies]" with an average travel time of 70 minutes per day (Transport for London (TfL), 2010). For those trips made wholly within the London region, average distances travelled per day were 9 kilometres. This value removes any distorting effect potentially caused by long distance trips made by London residents to other parts of the United Kingdom.

This analysis was included in work presented by Pye and Daly in 2015, which focuses on their energy systems approach to modelling modal shift in urban areas in the United Kingdom. In their assessment, they reviewed a number of existing studies to estimate the maximum degree of modal shifting that could be achieved in urban areas in the United Kingdom, including the TfL study (Transport for London (TfL), 2010; Pye and Daly, 2015). Pye and Daly consider the ability for cycling, walking and public transport (bus and train) to offset demand for car travel in Greater London, concluding that a maximum of 44% of per capita car travel in 2050 could potentially be offset by these alternative modes of travel (Pye and Daly, 2015). In turn, while 100% mode shifting might be theoretically possible from a technical standpoint, the values used in the scenarios presented here are grounded by these previous studies to arguably more realistic values of 0-40%.

For context, Pye and Daly note that this value includes a 700% increase in annual per capita cycling, to reach 670 km in 2050 (Pye and Daly, 2015). This level is below the current Dutch average of 850 km per year (Ministerie van Verkeer en Waterstaat, 2009; Pye and Daly, 2015).

3.10 Ongoing Related Projects

This section provides a brief overview of the ongoing work to extend this research to incorporate explicit analysis of the air quality impacts of energy technology transitions in partnership with researchers at Kings College London. In this research, the UKTM-UCL-AQ model has been soft-linked to the Community Multiscale Air Quality Model (CMAQ) for the United Kingdom (Williams *et al.*, 2016). This soft-linking allows for the explicit modelling of the air quality impacts of changes in the energy system in a manner that accounts for non-energy sector air pollution. It also accounts for impacts on tropospheric ozone, a secondary pollutant that is formed through the interaction of nitrogen oxides (NO_x), carbon monoxide (CO), and volatile organic compounds (VOCs) in the presence of sunlight.

The next chapter includes the presentation of results from the application of UKTM-UCL-AQ for the United Kingdom.

Chapter 4 – UK Analysis

4.1 Overview

This chapter presents an overview of the analysis that was done in this research to quantify the co-impacts of energy system decarbonisation on non-greenhouse gases emission levels in the United Kingdom. It begins with an overview of the key outcomes and insights from this research (Section 4.2). This section is followed by details on the application of the UKTM-UCL model variant (UKTM-UCL-AQ) that was developed in Fall 2015 for a set of six scenarios (Section 4.3) with additional details included in the Appendix of this manuscript. The chapter concludes with a brief discussion and series of conclusions (Section 4.4). Additional discussion on the implications of these findings is provided in Chapter 6 of this thesis.

The goal of the analysis presented here was to explore the two research questions identified in this research project with regards to the United Kingdom energy sector, namely:

- 1. What are the co-impacts (both positive and negative) on particulate matter and nitrogen oxide air pollution levels for energy sector decarbonisation pathways that are optimised with regards to reducing total greenhouse gas emissions on a national scale?
- 2. How does considering the impact of these other types of outdoor air pollution (i.e. particulate matter and nitrogen oxides) impact the decarbonisation pathway on a national scale?

These research questions were explored in the context of an energy system technology transition between 2010 (i.e. the base year) and 2050. This time is currently used by the United Kingdom government in setting its long-term decarbonisation goals as is the ultimate target set under the Climate Change Act of 2008 that is discussed in more detail in Chapter 2 of this thesis (United Kingdom Parliament, 2008).

The core tool used in this portion of this research project is the UKTM-UCL-AQ model, which was developed in Fall 2015. As mentioned elsewhere in this manuscript, the UKTM-UCL-AQ model is a bottom-up technoeconomic optimization model that includes air pollution and its associated damage costs. Its topical coverage and relationship to the other models used in the research project and a related ongoing collaborative side project can be seen in Figure 4.1. The UKTM-UCL-AQ model is a single region, national scale model for the United Kingdom while PIONEER is a single region, urban scale model for the Greater London area.

Figure 4.1: Topical Coverage of the UKTM-UCL-AQ Model in Relation to PIONEER and CMAQ



As discussed in Chapter 3, the scenarios presented in this Chapter cover a range of policy ambitions for both decarbonisation and air pollution. For the sake of completeness and to avoid potential confusion, Figure 4.2 is included here as well as in Chapter 3.

Figure 4.2: Scenarios Map for United Kingdom Analysis

ion Costs	With Damage Costs	No decarbonisation goals. Attention paid to non-GHG air pollution costs.	Intermediate level of attention to decarbonisation. Attention paid to non-GHG air pollution costs.	Current decarbonisation goals achieved. Attention paid to non-GHG air pollution costs.
Air Pollut	No Damage Costs	No decarbonisation goals. No attention to non- GHG air pollution costs.	Intermediate level of attention to decarbonisation. No attention to non-GHG air pollution costs.	Current decarbonisation goals achieved. No attention to non-GHG air pollution costs.
		none	intermediate	lowGHG

Decarbonisation Policy Ambition

Portions of the work discussed in this chapter have been presented at three conferences, published in the *Journal of Transport and Health*, and published by *Energy Policy* (Lott, Ekins and Davies, 2014; Lott and Daly, 2015; Lott *et al.*, 2016; Williams *et al.*, 2016; Lott, Pye and Dodds, 2017). The author of this thesis served as primary author/presenter for all of these publications and presentations, with the exception of Williams, et. al. 2016, which relates to ongoing related work in partnership with Kings College London that is described in more detail in the Appendix (Williams *et al.*, 2016).

4.2 Key Outcomes & Insights

 Air pollution emission factors for six (6) air quality pollutants - PM₁₀, PM_{2.5}, NO_x, SO_x, NH₃, and NMVOCs were integrated for all energy sectors into the UK TIMES Model (UKTM-UCL) to create the UKTM-UCL-AQ variant.

- 2. UKTM-UCL-AQ could represent the vast majority of NO_x and SO_x pollution sources in the United Kingdom, a majority of PM_{10} and $PM_{2.5}$ pollution, and a minority of NH_3 and NMVOCs pollution.
- The estimated costs to the environment and human health were integrated into UKTM-UCL-AQ for all energy sectors using damage cost values that are based on an impact pathway approach.
- 4. Incorporating the damage costs of air pollution led to significant changes in air pollution levels over time to 2050, resulting from changes in technology choices and fuel-switching particularly in the non-transport sectors.
- 5. The inclusion of air pollution damage costs had little impact on energy system costs, with an overall increase of 0.15% to 0.5%.
- 6. In particular, the incorporation of damage costs impacted where biomass was used as an input fuel in the energy system, shifting away from residential sector use.
- 7. Results were quite sensitive to assumptions, including the degree to which fuel shifting could occur but not including constraints on the nuclear fleet expansion.
- Transport was the least impacted of the sectors considered within the energy system, suggesting that targeted polices would be needed to address transport-sector air pollution impacts.

4.3 Scenario Outputs

A set of six (6) scenarios are developed to better understand the relative impacts of the inclusion or exclusion of the damage costs for outdoor air pollution. These scenarios include a baseline (base), reference (ref), and low greenhouse gas (lowGHG) both with and without damage costs as shown in Table 4.1. The scenarios which included damage costs for non-greenhouse gas air pollutants as displayed in Table 4.1 are indicated with a " DAMC" at the end of the scenario name. For example, the "base" and "base_DAMC" scenarios are the same except that the latter included damage costs in the optimisation process.

Scenario Name	Carbon target/price?	Damage costs?		
base	No	No		
base_DAMC	_	Yes		
ref	Yes - £30/tonne in 2030	No		
ref_DAMC	_	Yes		
lowGHG	Yes – 80% reduction by 2050 with interim targets (i.e. carbon	No		
lowGHG_DAMC	budgets)	Yes		

 Table 4.1: Scenario Overview

The base and ref scenarios did not include the United Kingdom's 2050 decarbonisation goal or interim targets. The latter included a £30 per tonne carbon price that was linearly phased in from 2015 to 2030 and then held constant to 2050 in order to simulate a central case where the system moves away from the most carbon-intensive technologies (e.g. coal in the electricity sector) but long term decarbonisation goals are not achieved. In the lowGHG scenario, the energy system is required to meet existing U.K. decarbonisation targets for a total reduction in greenhouse gas emissions of 80% by 2050 compared to 1990 levels including interim targets through the 4th Carbon Budget. In late July 2016, the UK Government set a 5th Carbon Budget of 1,725 million tonnes of carbon dioxide equivalent for the 2028–2032 budgetary period in agreement with recommendations from the Committee on Climate Change (Department for Business Energy and Industrial Strategy (BEIS), 2016). The reduction trajectory used in this analysis is broadly consistent with this recently agreed 5th Carbon Budget, though it was not explicitly included as this budget was under development when these scenarios were constructed and executed.

The scenarios examine the period to 2050 for the United Kingdom using demand drivers that rely upon official population and economic growth projections and energy efficiency expectations from the Office of National Statistics (Office of National Statistics, 2010).

4.3.1 Scenarios without damage costs

For the three scenarios without damage costs (i.e. base, ref and lowGHG), the decarbonisation of the energy sector results in significant co-benefits for reducing air pollutant emissions in 2050. For particulate matter, decarbonisation in the lowGHG scenario results in an additional 34% (41 kilotonne) decrease in PM₁₀ emissions and 38% (29 kt) decrease in PM_{2.5} pollution levels in 2050 compared to both the base and ref scenarios in that year as shown in Figure 4.3. These additional decreases were the result of shifts away from fossil fuels (including coal). This result indicates that the carbon tax applied in the ref scenario did not have a significant impact on 2050 levels of particulate matter air pollution (both PM₁₀ and PM_{2.5}), though some intermediate differences are seen as shown in Figure 4.3. A full set of figures showing air pollution by sector for these scenarios can be found in the Appendix of this thesis.

However, decarbonisation in the lowGHG scenario results in increased particulate matter pollution between 2020 and 2045 due to rises in the use of biomass for residential heating. These units would likely be located in areas with higher population densities, giving rise to concerns over pollution exposure levels in urban areas and corresponding policy questions for local governments. This mid-term PM emissions increase is avoided with the inclusion of damage costs, as discussed in the next section.

170

The differences in NO_x emission levels in 2050 across scenarios were also notable, with an additional 25% (125 kilotonne) and 18% (84 kt) reduction in emissions in 2050 in the lowGHG compared to the base and ref scenarios, respectively. These results are displayed in Figure 4.3.

The most dramatic absolute reductions in air pollution emissions between scenarios in 2050 were seen for SO_x pollution levels. All told, decarbonisation in the lowGHG scenario led to a 58% reduction (203 kt) in SO_x emissions compared to the base case in 2050. The difference in SO_x between the lowGHG and ref scenario – with its intermediate carbon price linearly phased in between now and 2030 - was 100 kilotonnes in 2050. These results are displayed in Figure 4.3. Of note here is that – unlike for particulate matter air pollution – emissions of NO_x and SO_x air pollution in 2050 were impacted by the carbon price included in the ref scenario, which leads to different emission levels of these two pollutants in 2050.

Figure 4.3: Total Air Pollution Emissions by Type in the United Kingdom for Scenarios Without Damage Costs, 2010-2050



Noted here is that the ability for fuel-switching to biomass in the residential sector to meet heating demand was limited to the rates shown in the scenario output for the lowGHG scenario. These rates are equal to the maximum allowed by user constraints placed in the model. Without this constraint, the spike in biomass use in the residential sector for heating purposes was much higher and seemingly quite unlikely.

This output is illustrated below in the results from the lowGHG scenario variant "lowGHG_lessconstrainedbio". In this variant, all scenario inputs and constraints were identical to the lowGHG scenario with one exception – that is, the constraint for fuel-switching to biomass was relaxed. This relaxed constraint resulted in increasing levels of mid-term biomass use, resulting in a much higher spike in particulate matter air pollution as displayed in Figures 4.4 and 4.5. This result indicated a high sensitivity of the outputs to this residential biomass constraint.



Figure 4.4 Total Particulate Matter (PM_{10}) Emissions in the United Kingdom for lowGHG lessconstrainedbio" and "lowGHG", 2010-2050



Figure 4.5: Total Particulate Matter ($PM_{2.5}$) Emissions in the United Kingdom for lowGHG lessconstrainedbio" and "lowGHG", 2010-2050

4.3.2 Scenarios that include damage costs

When the damage costs of other air pollutants (i.e. particulate matter, nitrogen oxides, sulphur oxides, ammonia, and non-methane volatile organic compounds) are in the optimisation process, the model selected somewhat different technologies and fuel use patterns across all scenarios. Again, this is because the model explicitly sees the external costs of air pollution, which therefore becomes an economic determinant in energy system choices. For example, coal is replaced by natural gas for electricity generation. There is also a decrease in biomass switching, in particular in the residential sector between 2020 and 2045, showing the inherent air quality risks in decarbonisation pathways that rely heavily on bioenergy use. As discussed in Chapter 3, these scenarios used the "Central" annual pulse damage costs as shown in Table 3.6.

Primary energy consumption in 2050 by fuel type is displayed in Figure 4.6 for all scenarios. Overall, the inclusion of damage costs in the base scenario led to increased use of natural gas and decreased use of biomass and biofuels as well as coal and coke in 2050. Decarbonisation ambitions resulted in increased use of nuclear power for the ref and lowGHG scenarios. For the latter, the inclusion of damage costs had little impact on final primary energy consumption in 2050, though the pathway taken was significantly different.



Figure 4.6: Primary energy consumption (PJ) in 2050 by scenario

Figure 4.7 shows the decrease in particulate matter emissions (for PM₁₀) that results from the inclusion of damage costs in the lowGHG scenario. In particular, the inclusion of damage costs prevents fuel-switching to biomass in the residential sector which, in turn, eliminates the rise in particulate matter pollution between 2020 and 2045. Along with this decrease in fuel-switching to biomass in the residential sector, increased levels of both electricity and other renewables use are seen as well as some smaller increases in natural gas use in the lowGHG_DAMC scenario compared to the scenario where damage costs were not included (i.e. lowGHG).





For the base and ref scenarios, the inclusion of damage costs resulted in lower 2050 air pollution levels across all air pollutants as shown in Figure 4.8 and Table 4.2. The impacts of including damage costs was less dramatic for the lowGHG scenario.



Figure 4.8: Air Pollution Levels in 2050 by Scenario

Pollutant	Scenario	Without damage costs (kt)	With damage costs (kt)	Difference (%)
	base	90	46	-49%
PM ₁₀	ref	90	46	-49%
	lowGHG	59	42	-29%
	base	76	35	-54%
PM _{2.5}	ref	77	34	-55%
	lowGHG	47	30	-35%
	base	504	460	-9%
NO _x	ref	463	408	-12%
	lowGHG	379	362	-4%
	base	346	181	-48%
SO _x	ref	293	178	-39%
	lowGHG	144	131	-8%

Table 4.2: Total Air Pollution by Scenario in 2050

For the transport sector, the inclusion of damage costs results in some limited technology shifts including the electrification of passenger rail across all scenarios. Overall, the relatively small level of change in transport sector pollution levels between scenarios indicates that the level of damage costs assumed were not enough to produce a significant shift in transport sector technologies.

For the base scenario, decreasing emission trends are observed from all forms of road transport, except for cars. For the ref and lowGHG scenarios, less dramatic technology shifts are seen, indicating that energy sector decarbonisation was the driving force behind the technology pathway chosen by the model. Total emissions from transport by emission type and scenario in 2050 as well as the percentage change compared to the base scenario in 2050 are shown in Table 4.3.

	_							
Scenario	PM10	% Change*	PM2 5	% Change*	NO,	% Change*	SO _x	% Change*
base	35.4		25.0		233.6		62.2	
ref	35.4	0.01%	25.0	0.0%	244.7	5%	62.2	0.0%
lowGHG	35.1	-0.81%	25.0	- 1.1%	231.0	-1%	61.9	-0.4%
base_DAMC	35.4	-0.15%	25.0	- 0.2%	225.4	-4%	62.1	- 0.3%
ref_DAMC	35.3	-0.29%	24.9	- 0.4%	229.9	-2%	62.1	- 0.0%
lowGHG_DAMC	35.0	-1.28%	24.5	- 1.8%	222.2	-5%	61.8	-0.4%

Table 4.3: Total Transport Emissions by Type and Scenario in 2050

*compared to the Base scenario

For road transport, total PM₁₀ emissions decline slightly to 2020 across all scenarios and then slowly increase to 2050 to within 5% of 2010 levels as shown in Table 4.4. A similar trend is seen with PM_{2.5} as shown in Table 4.5. These two outputs show the growing importance of non-tailpipe (i.e. road, tyre, and brake wear) particulate matter pollution that is directly a function of distance travelled and not of the type of fuel used. They also illustrate how increasing demand for road transport could slowly outstrip previous improvements in particulate matter pollution mitigation efforts despite improvements in engine technology.

Scenario	2010	2015	2020	2025	2030	2035	2040	2045	2050
base		42.0	38.0	36.7	35.6	35.2	35.3	35.3	35.4
ref		42.0	38.0	36.7	35.6	35.2	35.3	35.3	35.4
lowGHG	45.0	41.9	38.0	36.7	35.5	35.1	35.2	35.2	35.1
base_DAMC	45.0	42.0	37.9	36.6	35.5	35.1	35.2	35.3	35.4
ref DAMC		42.0	37.9	36.6	35.5	35.1	35.2	35.2	35.3
lowGHG_DAMC		41.9	37.9	36.6	35.4	35.0	35.0	35.0	35.0

Table 4.4: Total Transport PM₁₀ Emissions by Scenario, 2010 - 2050

Scenario	2010	2015	2020	2025	2030	2035	2040	2045	2050
base		33.9	29.6	27.9	26.5	25.8	25.6	25.3	25.0
ref		33.9	29.6	28.0	26.5	25.8	25.6	25.3	25.0
lowGHG	27.0	33.8	29.6	27.9	26.5	25.8	25.5	25.1	24.7
base DAMC	37.0	33.9	29.5	27.8	26.4	25.7	25.5	25.2	24.9
ref DAMC		33.9	29.5	27.8	26.4	25.7	25.5	25.2	24.9
lowGHG_DAMC		33.8	29.5	27.8	26.4	25.7	25.3	25.0	24.5

Table 4.5: Total Transport PM_{2.5} Emissions by Scenario, 2010 - 2050

For NO_x pollution, a distinct downward trend in total emissions is seen as more efficient and cleaner road transport technologies – in part related to Euro standards for new road vehicles - are adopted over time as shown in Tables 4.6 and 4.7. Similarly, SO_x emissions from road transport decreased in 2050 compared to the base year, though less dramatically. Of note is that SO_x emissions in the transport sector are predominately produced by non-road transport (in particular, international shipping). As mentioned, there are no options for targeted SO_x abatement for these technologies in the UKTM-UCL-AQ model at this time.

Scenario	2010	2015	2020	2025	2030	2035	2040	2045	2050
base		541.9	429.7	334.8	294.6	267.2	253.5	243.9	233.6
ref		552.5	440.0	344.8	297.8	273.3	262.7	252.4	244.7
lowGHG		556.4	451.4	334.4	275.6	253.3	246.7	234.4	231.0
base_DAMC	642.5	469.2	345.9	305.5	270.2	255.5	246.7	236.3	225.4
ref_DAMC		469.1	345.0	304.9	272.1	258.3	250.6	241.1	229.9
lowGHG_ DAMC		469.2	346.3	298.7	261.4	249.9	239.7	229.8	222.2

Table 4.6: Total Transport NO_x Emissions by Scenario, 2010 – 2050

Scenario	2010	2015	2020	2025	2030	2035	2040	2045	2050
base		106.6	97.4	87.9	78.6	72.9	69.9	66.1	62.2
ref		106.6	97.4	87.9	78.6	72.9	69.7	66.1	62.2
lowGHG		106.6	97.4	87.9	78.5	72.8	69.6	65.9	61.9
base_DAMC	101.1	106.6	97.3	87.8	78.5	72.8	69.6	66.0	62.0
ref DAMC		106.6	97.3	87.8	78.5	72.8	69.6	66.0	62.0
lowGHG_ DAMC		106.6	97.3	87.8	78.4	72.7	69.5	65.8	61.8

Table 4.7: Total Transport SO_x Emissions by Scenario, 2010 – 2050

It is interesting to examine road transport – cars, 2-wheelers (motorcycles), buses, low-gross vehicles, and high-gross vehicles – in isolation in order to better understand the impacts of increasing demand on air pollution emission profiles as it swamps gains made with decreasing emission factors in some cases. In particular, non-GHG air pollution emissions over time for the "cleanest" of the six scenarios (i.e. the lowGHG_DAMC scenario) where the 80% decarbonisation target is met and the damage costs of non-greenhouse gas air pollutants are included in the optimisation process are displayed in Figure 4.9. Here, one sees increasing levels of particulate matter emissions from 2020 due to increasingly dominant non-tailpipe emissions coupled with increasing levels of demand. There is also sees rising levels of nitrogen oxide air pollution levels after 2030 due to increasing demand for car travel.

Figure 4.9: Air Pollution Emissions from Road Transport by Technology for the lowGHG_DAMC scenario, 2010-2050



With regards to cars, the inclusion of damage cost did accelerate the transition away from diesel vehicles to petrol and hybrid electric cars. This trend is shown in Figure 4.10 for the base and base_DAMC scenarios. As discussed elsewhere in this thesis, these scenarios did not include any decarbonisation goal or carbon price. In turn, the differences observed in the outputs of these two scenarios isolate the impact of damage costs on technology trends. For example, diesel vehicles are phased out completely by 2040 in the base scenario versus 2030 when damage costs are included (i.e. in the base_DAMC scenario) as seen in Figure 4.10.


Figure 4.10: Transport demand by engine type (bvkm) for the base and based scenarios

Impacts from the inclusion of damage costs are seen throughout the other sectors as well and lead to a number of interesting areas for future work. For example, total PM_{10} emissions in the agriculture sector was significantly impacted by the inclusion of damage costs in the scenarios run as shown in Figure 4.11. These results indicate that relatively cheap mitigation options exist in this sector. Whether or not these emission reductions could be practically realised in the U.K. agriculture sector – and, indeed, if additional reductions could be realised - represent an opportunity area for future exploration and research.



Figure 4.11: Total PM_{10} emissions from the agriculture sector from 2010-2050 across six scenarios

Similar results, conclusions, and opportunities for future work were found in the industrial sector as shown in Figures 4.12 and 4.13. In the case of SO_x , less dramatic differences were seen in the lowGHG scenario as shown in Figure 4.14.



Figure 4.12 Total PM_{10} emissions from the industrial sector from 2010-2050 across six scenarios

Figure 4.13: Total $PM_{2.5}$ emissions from the industrial sector from 2010-2050 across six scenarios



Figure 4.14: Total SO_x emissions from the industrial sector from 2010-2050 across six scenarios



While the inclusion of damage costs resulted in significant reductions in total pollution levels for non-GHG emissions, they did not dramatically impact total GHG emission levels in the scenarios considered here, as shown in Figure 4.15. Furthermore, there is no noticeable difference in the pace of decarbonisation in the lowGHG scenarios (i.e. lowGHG and lowGHG_DAMC), though differences are observed in individual technology choices across the energy system (e.g. less biomass use if damage costs are included). Perhaps the only

significant exception to this observation is found in the ref scenario, where damage costs noticeably accelerated energy sector decarbonisation between 2020 and 2035 – though 2050 GHG emission levels are essentially unaffected as the U.K. greenhouse emissions reduction target drives the 2050 emission levels. In turn, it is clear that the decarbonisation ambition – and not the damage costs associated with the other types of air pollution - is the dominate driving force behind the technology transition pathway for the scenarios considered.





From a cost perspective, the shifts in the technology choices made in the pathways driven by the inclusion of the damage costs of other types of air pollution have limited impact, as shown in Figure 4.16. In fact, if the air pollution damage cost (Cost_Com) component is removed, the actual additional costs of energy system expenditure were minimal with overall increases of 0.15% to 0.5%. In summary, the inclusion of these damage costs resulted in large air pollution emission benefits but had quite small impact on overall energy system costs.



Figure 4.16: Overall System Costs (annual, undiscounted), including CO₂ or air pollution tax levels

Legend: investment costs, annualised (Cost_Inv); fixed operation and maintenance (Cost_Fom); energy/fuel (Cost_Flo); variable operation and maintenance (Cost_Act); CO2 tax/shadow price (Cost_Comx); air pollution damage costs (Cost_Com)

4.4 Limiting New Nuclear Capacity

As discussed previously in more detail in Chapter 2, the United Kingdom has achieved an average decrease in carbon dioxide equivalent (CO₂e) emissions of 4.5% per year since 2012 (Committee on Climate Change, 2016). According to the Committee on Climate Change in its June 2016 report, these drops were almost entirely due to decarbonisation in the power sector, particularly through the rapid decline in coal for power generation in favour of renewables. Indeed, the report highlights that:

"There has been almost no progress in the rest of the [United Kingdom's] economy, where emissions have fallen less than 1% a year since 2012 on a temperature-adjusted basis. That is because there has been slow uptake of low-carbon technologies and behaviours in the buildings sector (i.e. low rates of insulation improvement, low takeup of low-carbon heat) and improved vehicle efficiency has been offset by increased demand for travel as the economy has grown and fuel prices have fallen...Progress will need to be broader to meet the recommended fifth carbon budget and to prepare sufficiently for 2050. For example, while the complete replacement of coal-fired generation with low-carbon generation in the power sector is an important part of our scenarios, this would provide less than half of the total emissions reduction required by 2030."

Moving forward, the lowGHG case discussed in the previous section includes a significant increase in electricity generation from nuclear power in the United Kingdom with nuclear generation capacity increasing from 10 GW in 2010 to 34 GW in 2050 in this cost-optimised scenario. Electricity production over time for this scenario is displayed in Figure 4.17.



This dramatic increase in new nuclear capacity in conjunction with the recent controversy and delay relating to the proposed third nuclear power station to be built at Hinkley Point in Somerset, England raises significant questions as to the feasibility of this future scenario (Davies, 2016). In turn an alternative scenario ("lowGHG NuclearLimit) is presented where

nuclear capacity is capped at a 50% increase in 2050 compared to 2010 (i.e. an increase from 10 GW to 15 GW over this forty-year period). The resulting electricity production over time outputs for this scenario are displayed in Figure 4.18, which shows the increased role for natural gas both with and without carbon capture and storage (CCS) that resulted from limiting future nuclear capacity growth.



The outputs from this scenario show only minor differences in the emissions profiles between the lowGHG and lowGHG_NuclearLimit scenarios as shown in Figures 4.19 through 4.22. In particular, the carbon emissions trajectory was essentially identical between the two scenarios. Similar results for particulate matter and SO_x emissions indicate a low sensitivity to changes in the nuclear fleet growth constraint that is used in UKTM-UCL-AQ. Overall, the most significant difference in air pollution scenarios between the two scenarios was seen for nitrogen oxides (NO_x), where air pollution emission levels for the two scenarios diverged starting in 2035 with the difference increasing to 2050. In 2050, the lowGHG_NuclearLimit scenario resulted in 41 kilotonnes (11%) more NO_x emissions than the lowGHG scenario as shown in

Figure 4.22.



Figure 4.19: Total annual carbon dioxide equivalent emissions in the U.K. for lowGHG and lowGHG_NuclearLimit scenarios, 2010-2050

Figure 4.20: Total annual PM_{10} emissions in the U.K. for lowGHG and lowGHG_NuclearLimit scenarios, 2010-2050





Figure 4.21: Total annual $PM_{2.5}$ emissions in the U.K. for lowGHG and lowGHG_NuclearLimit scenarios, 2010-2050

Figure 4.22: Total annual nitrogen oxide (NO_x) emissions in the U.K. for lowGHG and lowGHG_NuclearLimit scenarios, 2010-2050



The lowGHG_NuclearLimit scenario, with its limits on nuclear capacity expansions, resulted in a very slight increase in total cumulative undiscounted energy system costs compared to the lowGHG scenario over the period of 2010-2050. These increased costs were predominately realised from 2040-2050 as shown in Table 4.8.

Annual undiscounted ene	rgy system costs	2010	2015	2020	2025	2030	2035	2040	2045	2050
lowGHG		1,252	1,819	1,081	1,132	2,811	4,012	4,237	4,915	5,140
lowGHG_NuclearLimit	Activity costs,	1,252	1,864	1,141	1,190	2,848	3,770	3,649	3,888	4,184
difference	Cost_Act (M£)		45	61	58	37	(242)	(588)	(1,027)	(956)
difference (%)			2.5%	5.6%	5.1%	1.3%	-6.0%	-13.9%	-20.9%	-18.6%
lowGHG		31,991	12,852	31,483	34,704	25,718	27,688	34,437	36,224	37,197
lowGHG_NuclearLimit	Flow costs, Cost_Flo	31,991	12,654	29,545	32,963	26,066	30,124	38,767	42,082	43,996
difference	(M£)		(198)	(1,938)	(1,741)	349	2,436	4,330	5,858	6,799
difference (%)			-1.5%	-6.2%	-5.0%	1.4%	8.8%	12.6%	16.2%	18.3%
lowGHG		106,700	117,096	122,380	130,868	139,732	148,408	156,593	163,991	170,341
lowGHG_NuclearLimit	Fixed O&M costs,	106,700	117,266	122,927	131,291	139,812	148,117	156,338	163,839	169,975
difference	Cost_Fom (M£)		169	547	423	80	(291)	(255)	(151)	(366)
difference (%)			0.1%	0.4%	0.3%	0.1%	-0.2%	-0.2%	-0.1%	-0.2%
lowGHG		98	74,117	121,983	162,200	190,749	210,832	230,845	247,195	264,154
lowGHG_NuclearLimit	Investment costs,	98	74,116	122,551	162,315	190,249	208,763	228,893	245,710	263,296
difference	Cost_Inv (M£)		(0)	567	115	(501)	(2,069)	(1,952)	(1,485)	(858)
difference (%)			0.0%	0.5%	0.1%	-0.3%	-1.0%	-0.8%	-0.6%	-0.3%
lowGHG		140,041	205,884	276,926	328,905	359,011	390,939	426,112	452,324	476,832
lowGHG_NuclearLimit	Sum (M£)	140,041	205,900	276,163	327,759	358,975	390,774	427,647	455,519	481,451
difference	Sum (IVIE)		16	(763)	(1,145)	(36)	(165)	1,535	3,195	4,619
difference (%)			0.0%	-0.3%	-0.3%	0.0%	0.0%	0.4%	0.7%	1.0%

Table 4.8: Annual undiscounted energy systems costs for lowGHG and lowGHG_NuclearLimit scenarios

4.5 Discussion & Conclusions

As mentioned at the start of this chapter, this research was meant to explore the answer to the following two research questions:

- What are the co-impacts (both positive and negative) on particulate matter and nitrogen oxide air pollution levels for energy sector decarbonisation pathways that are optimised with regards to reducing total greenhouse gas emissions on a national scale?
- 2. How does considering the impact of these other types of outdoor air pollution (i.e. particulate matter and nitrogen oxides) impact the decarbonisation pathway on a national scale?

Across all scenarios at the national level, it is clear that climate policy has significant benefits for reducing air pollution emissions in the United Kingdom in 2050. However, some potentially concerning increases are seen with respect to particulate matter emissions in the medium term as the result of rising levels of biomass use in the residential sector for the scenario where the United Kingdom achieved its decarbonisation goals (i.e. the lowGHG scenario).

The consideration of health impacts – in the form of air pollution damage costs included in the optimisation process – result in changes in the fuels and technologies selected by the model. The inclusion of these costs eliminates previously mentioned rises in residential air pollution emission levels before 2045, showing the importance of simultaneously considering the impact of climate policy on efforts to reduce air pollution and vice versa. However, including these damage costs does not significantly impact the CO_2 emissions trajectory, with the exception of the ref scenario in the medium term.

Particulate matter air pollution from transport is not significantly impacted by the inclusion of damage costs in UKTM-UCL-AQ, indicating that targeted policies would be required to substantially reduce these emissions in the future, even if there were a move away from internal combustion engine vehicles. This result is largely due to the fact that, in the scenarios presented here, non-tailpipe particulate matter air pollution increasingly dominates air pollution in road transport over time due to rising demand.

Overall, this work shows that technoeconomic energy systems models can provide significant insight on particulate matter (PM_{10} and $PM_{2.5}$), nitrogen oxide (NO_x), and sulphur oxide (SO_x) air pollution. With respect to other local air pollutants, the vast majority of emission sources for non-methane volatile organic compounds (NMVOC) and ammonia (NH_3) are not within the energy sector, which means that these emissions are not fully captured in UKTM-UCL-AQ and offer interesting areas for future research.

Failure to consider non-GHG air pollution in decarbonisation strategy development creates tension between decarbonisation, air pollution, and public health policies and could result in mid-term air pollution challenges between 2025 - 2040. Considering damage costs in the decarbonisation pathway reduces particulate matter pollution from residential heating systems using biomass fuel 2025 and 2040. Constraints relating to the expansion of the nuclear fleet did not have a significant impact on particulate matter or SO_x air pollution levels in a scenario where long-term decarbonisation goals were reached but damage costs for air pollution were not included. However, the nuclear constraint considered did result in an 11% increase in NO_x emissions in 2050.

This research approach and the resulting insights illustrate the importance of understanding the relationship between greenhouse gas and other air pollution emissions, which share many important sources in the energy sector. The former is a growing concern and the latter is an immediate public health problem in the United Kingdom. Understanding the trade-offs and synergies between these two groups of air pollutants could be critical to effective policy design.

However, it should be noted that this approach did not include all air pollution abatement options, but in effect restricts responses to fuel switching and efficiency gains through technology turnover. Future work is needed in this area to combine work specifically on air quality abatement technologies and their incorporation in energy system optimisation models as this could illuminate the potential role of these technologies in further reducing air pollution levels on a cost-optimised basis. Additional insights would also be gained through the analysis of the model's outputs in a detailed air quality model. This aspect of the research is currently being explored through a partnership with Kings College London. Further improvements could also be made through study of the likely emissions factors for new technologies and refinement of those factors used for existing technologies.

One key limitation of the analysis presented in this chapter lies in the level of spatial disaggregation included in the model given that UKTM-UCL and UKTM-UCL-AQ are national-scale models. In turn, these models are only equipped to provide limited insights on potential air pollution hotspots (e.g. urban areas) and how these hotspots might be impacted by changes in energy system technologies. The next chapter in this thesis explains research completed in this PhD to explore the effects of decarbonisation efforts on urban areas in the United Kingdom, with particular emphasis on the Greater London area.

This page was intentionally left blank.

Chapter 5 – London Analysis

5.1 Overview

This chapter presents details on the application of the PollutION Emissions from EneRgy (PIONEER) model soft-linked to the UKTM-UCL-AQ energy systems model that is described in Chapters 3 and 4. It begins with the presentation of the key outcomes and insights from this work (Section 5.2) followed by an overview of the development of a set of eighteen (18) scenarios using PIONEER soft-linked to UKTM-UCL-AQ (Section 5.3). It then gives a presentation and discussion of key results from this application in scenarios without behaviour change (Section 5.4) and those scenarios with behaviour change including the public health impacts of all scenarios considered (Section 5.5). Included in each of these two sections are sensitivity tests and discussion on the impacts of changing the demand assumptions used in the primary set of scenarios. This chapter concludes with an overview of key insights drawn from this work (Section 5.6).

The goal of this research was to explore the answer to the research questions included in this research project with regards to the Greater London area's road transport sector, namely:

- What are the co-impacts (both positive and negative) on particulate matter and nitrogen oxide air pollution levels for energy sector decarbonisation pathways that are optimised with regards to reducing total greenhouse gas emissions on an urban scale?
- 2. How does considering the impact of these other types of outdoor air pollution (i.e. particulate matter and nitrogen oxides) impact the decarbonisation pathway on an urban scale?

As discussed in Chapter 4 and presented in Table 4.3, the answers to these two questions with specific reference to the road transport sector appears to be that the co-impacts are quite limited. In turn, the research presented in this chapter explores the extent to which local action in the Greater London area could contribute to reductions in locally produced air pollution and its associated public health impacts. This research takes into consideration two dimensions – technological and behavioural change – which is described in more detail in elsewhere in this chapter.

In this research, focus is placed on three key primary non-greenhouse gas air pollutants – namely, particulate matter (PM_{10} and $PM_{2.5}$) and nitrogen oxides (NO_x). This emphasis is placed as discussed in Chapter 1-4 of this thesis. First, these air pollutants are both prevalent in the Greater London area and can be largely attributed to the energy sector. Second, a significant portion of the estimated local public health impact of these pollutants has been linked directly to the local transport sector in recent studies (Miller and Hurley, 2010; Beevers *et al.*, 2013; Atkinson *et al.*, 2014; Gowers, Miller and Stedman, 2014; Walton *et al.*, 2015). Third, these pollutants are of primary concern with regards to air pollution in Greater London and are largely captured in the UKTM-UCL-AQ energy system model.

5.2 Key Outcomes & Insights

The key outcomes of the work presented in this chapter are as follows:

 The PollutION Emissions from EneRgy (PIONEER) model was created and soft-linked to the UKTM-UCL-AQ energy systems model in order to disaggregate the Greater London region transport sector from the broader United Kingdom.

- 2. The air pollution and public health impacts of a range of scenarios were analysed to establish the extent to which the Greater London region could successfully reduce local air pollution levels from road transport using combinations of technology and behavioural change (i.e modal shift away from cars).
- 3. The public health impacts of the scenarios were quantified.

Key insights for this work include:

- 1. Technological change was the principle driver of changes in air pollution emissions, even when considering up to a 40% mode shift away from cars.
- 2. Decarbonisation led to reductions in tailpipe emissions of vehicles and increased the relative importance of non-tailpipe PM_{2.5} emissions.
- 3. In the absence of technological change, particulate matter emissions (PM₁₀ and PM_{2.5}) as well as nitrogen oxide (NO_x) emissions produced by road transport in Greater London increased in all scenarios, including those that incorporated up to a 40% modal shift away from cars.
- 4. For all scenarios that included technological change to meet decarbonisation targets set forth in the UK Climate Change Act, particulate matter (PM₁₀ and PM_{2.5}) and nitrogen oxide (NO_x) emissions decreased over time to below 2010 levels.
- 5. The public health gains from reductions in NO_x emissions overshadowed gains in $PM_{2.5}$ reductions and offered the largest potential health benefit of those considered.

5.3 Scenario Development

For the research presented in this chapter, the PIONEER model is soft-linked to the UKTM-UCL-AQ model as discussed in Chapter 3. Functionally, the outputs from UKTM-UCL-AQ as discussed in Chapter 4 for road transport in the United Kingdom serve as inputs to PIONEER and then the outputs of PIONEER are used as inputs into UKTM-UCL-AQ in a "loop" until

harmonization is achieved between the two models as discussed in Chapter 3. The analysis flow diagram presented in that chapter is reproduced here for the sake of clarity in Figure 5.1.



Figure 5.1: Analysis Flow Diagram for Soft-linked PIONEER and UKTM-UCL-AQ Models

The topical coverage of the results from this soft-linking is shown in Figure 5.2, including the relationship of the PIONEER model with the UKTM-UCL-AQ that is also used in this research project and the Community Multiscale Air Quality (CMAQ) models that is used in a related collaborative project with Kings College London as mentioned elsewhere in this thesis (Williams *et al.*, 2016).





For the research presented in this chapter, scenario outputs from UKTM-UCL-AQ that are described in Chapter 4 for the lowGHG_DAMC scenario are used as inputs to the PollutION Emissions from EneRgy (PIONEER) model. Scenarios are then constructed across the technological and behaviour change dimensions as discussed in Chapter 3. Combined, the analyses along these dimensions result in a set of eighteen (18) scenarios as shown in Figure 5.3.





Technological Change

For the technological change dimension, scenarios are designed to map the range of possible future emissions profiles that would result from the span from 1) zero action to reduce tailpipe emissions from road transport to 2) one where all road transport vehicles with tailpipe emissions are banned from the Greater London region in 2050 along a set of four (4) potential technology deployment pathways in order to more fully explore the decision space. The former represents a sort of "worst case" scenario for locally produced air pollution emissions while the latter represents a "best case" for local action to reduce these emissions. In turn, the outputs of these scenarios effectively map the range of possibilities for future air pollution emissions levels resulting from technological change within Greater London.

For behavioural change, scenarios are designed to map the range of possibilities spanning from no modal shifting in Greater London up to a 40% modal shift away from cars in 2050. This research does not explore the ability to achieve in practice the behavioural changes discussed in this research project. Previous studies on this topic and the justification for exploring this range of mode shift are discussed in Chapter 3 of this thesis.

The combination of the two dimensions – technological and behavioural change – that are considered in this work lead to the construction of eighteen (18) scenarios to 2050, which are presented in Table 5.1. For these scenarios, the outputs of the scenario presented in Chapter 4 that included a national policy ambition toward decarbonisation and damage costs are included (i.e. the lowGHG_DAMC scenario) in the optimisation process in UKTM-UCL-AQ are used as inputs to the PollutION Emissions from EneRgy (PIONEER) model.

	Dimension					
Scenario Name	Technological Change	Behavioural				
		Change				
NoChange_NoChange	No change from 2010	No change from				
Na Champa 90.20		2010				
NoChange_80:20		20% mode shift				
		away noni cars				
NoChange_60:40		40% mode shift				
		away from cars				
UK_NoChange	Follows trends in United	No change from				
	Kingdom for the	2010				
UK_80:20	lowGHG_DAMC scenario.	20% mode shift				
		away from cars				
UK_60:40		40% mode shift				
		away from cars				
50:50_NoChange	Half the availability for zero-	No change from				
	tailpipe emission cars is taken by	2010				
50:50_80:20	Greater London and the other	20% mode shift				
	half is dispersed around the rest	away from cars				
50:50_60:40	of the United Kingdom until	40% mode shift				
	tallpipe emissions from cars are	away from cars				
	emmated in Greater London.					
Doubling NoChange	A ftor initially adopting half of the	No ahanga from				
Doubling_NoChange	availability for zero-tailnine	2010				
Doubling 80:20	emission vehicles in 2025, the	2010 20% mode shift				
2000001 <u>9</u> 00020	total use of zero-tailpipe	away from cars				
Doubling 60:40	emission cars doubles during	40% mode shift				
Doubling_oor to	each five-year period until 2045,	away from cars				
	at which point the zero-tailpipe					
	emission vehicles are adopted					
	rapidly to meet total demand for					
Clean I and an No Change	100% of zero tailning emission	No change from				
	cars that are available in the	2010				
CleanLondon 80:20	United Kingdom are adopted in	20% mode shift				
	Greater London until all of	away from cars				
CleanLondon _60:40	demand for cars in this urban area	40% mode shift				
_	are met by zero-tailpipe emission	away from cars				
	vehicles.					
JustInTime_NoChange	Adoption of zero-tailpipe	No change from				
	emission cars in Greater London	2010				
JustInTime_80:20	is delayed until the last time	20% mode shift				
	period considered (i.e. 2045-	away from cars				
Justin 1 ime _60:40	adopted repidly to most total	40% mode shift				
	demand for cars	away from cars				
	demand for cars.					

Table 5.1: Scenario Names and Definitions with regards to the Analysis Dimensions

Combined, the discussions contained in Chapter 4 and Chapter 5 explore of the range of impacts resulting from national and local action to reduce greenhouse gas emissions and other types of air pollution emissions. A visual mapping of these dimensions is displayed in Figure 5.4, including:

- National dimensions: decarbonisation policy ambition, air pollution costs
- Local dimensions: technological change, behavioural change



Figure 5.4 Scenario Dimension Mapping



 Decarbonisation goals (80% by 2050) met

The assumptions made with regards to demand disaggregation factors, technological change (i.e. emission factors) and behavioural change are discussed in more detail elsewhere in this chapter.

5.3.1 Emission Factors

The technological change dimension for each scenario has direct impacts on each set of emissions profiles for the scenarios constructed for the analysis presented in this chapter. In order to map the range of possible air pollution trajectories moving forward, six sets of technological change pathways (and their corresponding emission factors) as previously outlined in Figure 5.3. A complete list of all emission factor values used for these scenarios is included in the Appendix of this thesis.

With regards to the pathways where tailpipe emissions are eliminated by 2030, these types of transition could be achieved through a technology mandate, ambitious low emission zone initiative or similar measure for Greater London that restricts the use of internal combustion engines for cars in this urban area.

For these scenarios, the emphasis is placed on cars in order to enable a detailed discussion of the tradeoffs and synergies between transitioning to zero tailpipe emission vehicles versus modal shift away from car travel in Greater London. That being said, there is certainly an opportunity for additional work on non-car road transport and the impacts of zero emission vehicles in these portions of the system.

For the lowGHG_DAMC scenario produced using UKTM-UCL-AQ, the transition pathway seen for cars is shown in Figure 5.5. Highlighted in this figure and in Table 5.2 are the deployment pathways seen for two types of no-tailpipe emission cars – electric and hydrogen fuel cell. All told, 164 billion vehicles kilometres are travelled by electric and hydrogen cars in the United Kingdom in 2050 in the lowGHG_DAMC scenario. These technologies begin deploying in 2025 and 2030 (for hydrogen and electric vehicles, respectively) in small numbers, with their deployment accelerating more rapidly from 2035 onwards as shown in Figure 5.5.



Figure 5.5 Car demand by engine type (bvkm) in the United Kingdom for the lowGHG_DAMC scenario

Table 5.2 Car demand by engine type (bvkm) for electric and hydrogen vehicles in the United Kingdom for the lowGHG_DAMC scenario

car travel (bvkm)	2010	2015	2020	2025	2030	2035	2040	2045	2050
EV	0	0	0	0	0	3	8	19	41
Hydrogen	0	0	0	1	5	13	28	60	123
Total	0	0	0	1	5	15	37	79	164

Overall, this technology transition pathway means that there would be be sufficient zerotailpipe emission vehicles in the system to meet the total travel demand for cars in Greater London by just before 2040 as shown in Figure 5.6 if one assumes that all of these vehicles are deployed in Greater London.



Figure 5.6: Total Greater London Car Demand (Group 1) Versus Availability of Zero-Tailpipe Emission Vehicles in the lowGHG-DAMC scenario (2010-2050)

However, it is unlikely that none of these cars will deploy outside of this urban area prior to 2040. In turn, measures are taken to understand the impacts of the relative use of zero-tailpipe emission cars in and out of Greater London as previously discussed in this chapter.

For each of these scenarios, cars that are not zero-tailpipe emission are assumed to follow the overall technology pathway seen for other cars in the UK. For non-car road transport, all vehicles are assumed to follow UK trends. These potential pathways are displayed graphically in Figures 5.7 to provide additional clarity on the resulting technology transition pathway. In this figure, the solid lines display the total vehicle kilometres driven by zero-emission vehicles in Greater London for each of the four scenarios. The dashed lines in Figure 5.7 show total car demand in Greater London and the total availability of zero tailpipe emission cars in the United Kingdom. In all of these scenarios, tailpipe emissions from cars in Greater London are eliminated by 2050.





Figure 5.8 displays the emissions factors used for cars across the scenarios used for Greater London that did not include behavioural change to help in visualizing the range of emission factors assumed. As shown in these figures, zero-tailpipe emission vehicles still produce particulate matter (both PM_{10} and $PM_{2.5}$) via tyre, brake and road wear. In turn, these emission factors never reach zero. For non-tailpipe emissions, the values assume that these emissions are the same as in 2014 on a per kilometre basis, which might not be the case given a number of factors including evolving braking technologies, the increased hybridization of cars where energy is recovered and stored in batteries rather than being dispersed as waste heat through the braking system, tyre design improvements and increasing vehicle weights.

Contrastingly to particulate matter air pollution, the nitrogen oxide emission factors for cars in London go to zero by 2050 for zero-tailpipe emission cars. However, NO_x emissions are still produced by non-car road transport vehicles in the scenarios considered here. Figures 5.9 - 5.12 display the emissions factors used for 2-wheelers (motorcycles), buses, LGVs and HGVs that were used in this analysis.



Figure 5.8: London Emission Factors, Cars (2010 – 2050) for PM₁₀, PM_{2.5} and NO_x







Figure 5.11: London Emission Factors, LGVs (2010 – 2050) for PM₁₀, PM_{2.5} and NO_x PM10 Emission Factors - London, LGVs PM2.5 Emission Factors - London, LGVs





Figure 5.12: London Emission Factors, HGVs (2010 – 2050) for PM₁₀, PM_{2.5} and NO_x PM10 Emission Factors - London, HGVs PM2.5 Emission Factors - London, HGVs

5.3.2 Demand Disaggregation Factors

As discussed previously in Chapters 3 and 4, the demand drivers used within UKTM-UCL-AQ to calculate transport demand over time in the United Kingdom utilise per capital demand values by transport type for 2010 in conjunction with population projections from the Office of National Statistics (Office of National Statistics, 2010). In PIONEER, road transport demand for Greater London is disaggregated from the United Kingdom demand values using the same methodology and data input sources in order to ensure consistency between the two models as discussed in more detail in Chapter 3 (Office of National Statistics, 2010). The demand values used for Greater London in the analysis presented in this chapter are found in Table 5.3.

			-						-
	2010	2015	2020	2025	2030	2035	2040	2045	2050
Group 1: ON	S Long-Term								
Cars	23.70	25.17	26.41	27.47	28.37	29.91	31.53	33.24	35.04
2W	0.70	0.74	0.78	0.81	0.84	0.88	0.93	0.98	1.03
Buses	0.60	0.64	0.67	0.70	0.72	0.76	0.80	0.84	0.89
LGVs	3.80	4.03	4.23	4.41	4.55	4.80	5.06	5.33	5.62
HGVs	1.00	1.06	1.11	1.16	1.20	1.26	1.33	1.40	1.48
TOTAL	29.80	31.64	33.20	34.55	35.67	37.61	39.65	41.79	44.06

Table 5.3 Demand for Transport in Greater London, 2010-2050 (bvkm) for Group 1

5.4 Outputs for Scenarios Without Modal Shift

A set of six (6) scenarios were produced in order to map the impacts of the technological change dimension of the research presented in this Chapter. These scenarios do not include mode shifting to active transport as shown in Table 5.4. The additional set of twelve (12) scenarios that include up to a 40% mode shift away from cars are discussed in a subsequent section of this chapter.

N	Dimension					
Scenario Name	Technological Change	Behavioural Change				
NoChange_NoChange	No change from 2010					
UK_NoChange	Follows trends in United Kingdom for the lowGHG_DAMC scenario.					
50:50_NoChange	Half the availability for zero-tailpipe emission cars is taken by Greater London and the other half is dispersed around the rest of the United Kingdom until tailpipe emissions from cars are eliminated in Greater London.	No change				
Doubling_NoChange	After initially adopting half of the availability for zero-tailpipe emission vehicles in 2025, the total use of zero-tailpipe emission cars doubles during each five-year period until 2045, at which point the zero-tailpipe emission vehicles are adopted rapidly to meet total demand for cars in 2050.					
CleanLondon_NoChange	100% of zero-tailpipe emission cars that are available in the United Kingdom are adopted in Greater London until all of demand for cars in this urban area are met by zero-tailpipe emission vehicles.					
JustInTime_NoChange	Adoption of zero-tailpipe emission cars in Greater London is delayed until the last time period considered (i.e. 2045-2050), at which point they are adopted rapidly to meet total demand for cars.					

Table 5.4: Scenarios Isolating Impacts of Technological Change Dimension

Given that this subset of results is produced for the decarbonisation ambition set in the lowGHG_DAMC scenario presented in Chapter 4 (i.e. an 80% reduction in total CO_2e emissions by 2050 with interim carbon budget targets), the discussion presented in this and the subsequent section of this thesis focus on a portion of the scenario dimension map highlighted in grey within Figure 5.13.

Figure 5.13: Scenario Dimensions Covered in this Discussion



Decarbonisation Policy Ambition: 80% reduction by 2050 (i.e. lowGHG)

5.4.1 Results

For the scenario where no technology transition or behavioural shifts occur (i.e. NoChange_NoChange), PM_{10} emissions from Greater London road transport increase from 1.33 to 1.96 kilotonnes between 2010 and 2050. When Greater London road transport follows the same technology transition pathway as the rest of the United Kingdom in the lowGHG_DAMC scenario presented in Chapter 4, PM_{10} emissions reach their minimum of 0.97 kilotonnes in 2020 and then slowly rise to 1.20 kilotonnes by 2050. This trend is due to the face that increasing demand for road transport in Greater London outstrips gains with improving emission factors as non-tailpipe emissions increasingly dominate. For scenarios

where tailpipe emissions from cars are eliminated by 2050 in Greater London, total PM_{10} emissions decrease to 0.73 kilotonnes in 2050. This pollution is comprised of non-tailpipe emissions (i.e. road, tyre and brake wear) that is not eliminated with the adoption of electric and hydrogen fuel cell cars as well as both tailpipe and non-tailpipe emissions from other forms of road transport. These results are shown in Figure 5.14.

Figure 5.14: Total PM₁₀ Emissions in Greater London area for scenarios without behavioural shifts for Group 1 demand values (2010-2050)



For PM_{2.5}, as shown in Figure 5.15 emissions increase from 0.94 to 1.39 kilotonnes between 2010 and 2050 in the NoChange_NoChange scenario. For the UK_NoChange scenario, PM_{2.5} emissions levels reach their minimum in 2025 and then rise to 0.67 kilotonnes by 2050 due to increasing demand over time. For scenarios where tailpipe emissions from cars in Greater London are eliminated by 2050, total PM₁₀ emissions decrease to 0.38 kilotonnes in 2050.



Figure 5.15: Total $PM_{2.5}$ Emissions in Greater London area for scenarios without modal shifting for Group 1 demand values (2010-2050)

For NO_x pollution, as shown in Figure 5.16, emissions increase from 20.72 to 30.63 kilotonnes between 2010 and 2050 in the NoChange_NoChange scenario. For the UK_NoChange scenario, NO_x emissions levels reach their minimum (2.16 kilotonnes) in 2030 and then rise to 2.55 kilotonnes by 2050. For scenarios where tailpipe emissions from cars in Greater London are eliminated by 2050 in Greater London, total NO_x emissions decrease to 0.39 kilotonnes in 2050. This value represents the total tailpipe emissions from non-car road transport in Greater London in 2050.



Figure 5.16: Total NO_x Emissions in Greater London area for scenarios without modal shifting for Group 1 demand values (2010-2050)

The scenarios presented in this section do not alter any of the assumptions or scenario constraints that produced the national level "lowGHG_DAMC" scenario produced using UKTM-UCL-AQ. Rather, they explore the impacts of decisions with regards to the geographic distribution of road transport technologies by disaggregating technologies and demands that were previously presented as single values for the entire UK by UKTM-UCL-AQ into separate values for Greater London and the area outside of Greater London using PIONEER. In turn, steps 5-7 of the modelling process outlined in Chapter 3 were not required. However, these steps are required for the scenarios that include behavioural change as is described in that section of this chapter.

5.4.2 Sensitivity Analysis for Scenarios without Behavioural Change

The two areas of greatest sensitivity in the scenario outputs discussed in the previous section of this chapter relate to assumptions made for the:

- 1. technology transition pathway for Greater London
- 2. demand for road transport in Greater London

Combined, these six scenarios explore the decision space in a manner that illustrates the sensitivities of the results to the technology transition pathway assumed for Greater London. This section includes additional sensitivity tests to understand the impact of changing the demand disaggregation factors for Greater London.

As discussed in Chapter 3 in more detail, transport demand for Greater London is disaggregated from total demand in the United Kingdom using the same process and input assumptions as in UKTM-UCL-AQ. More specifically, per capita demand values by transport type for 2010 in Greater London are used in conjunction with population projections from the Office of National Statistics in their "long-term migration" scenario, which includes a disaggregation for Greater London (Office of National Statistics, 2010). The demand values for Greater London are the same in the scenario results discussed in the previous section. This section includes an examination of the sensitivity of scenario results to changes in the disaggregated demand values for Greater London.

According to analysis published in 2014 by Transport for London (TfL), demand for transport demand in Greater London is primarily driven by population size (Transport for London (TfL), 2014). In turn, it is appropriate to use population projections to calculate future demand for road transport in Greater London. However, other factors should be acknowledged in order to understand the uncertainty that exists in these future demand calculations. More specifically, according to TfL "modal trends have not uniformly followed population growth" (Transport for London (TfL), 2014). In their analysis, Transport for London identifies several other factors that have influenced transport demand trends in London since the mid-1990s, including:

- income
- economic performance of inner versus outer London and the broader United Kingdom
- demographics
- cost of public transport
- supply³⁴
- road capacity and car ownership saturation
- policies

³⁴ i.e. the availability of public transport

With regard to transport policies, Transport for London identifies parking policies from the mid-1990s and congestion charging as having particularly significant impacts on travel demand in London (Transport for London (TfL), 2014). Specifically, with regards to car travel, TfL observes a number of supply, underlying demand, and structural changes that have ongoing impacts on mode shifting from car travel to public transport. A summary graphic of these factors produced by TfL is displayed in Figure 5.17 (Transport for London (TfL), 2014).

Figure 5.17 Factors Contributing to Modal Shifts from Cars to Public Transport in Greater London (Transport for London (TfL), 2014)



In turn, for the core analysis presented in this Chapter, it is assumed that population was the primary driver of road transport demand in Greater London (as was also the case in UKTM-UCL-AQ). Noted here is that the population projections for the United Kingdom and Greater
London that were used in UKTM-UCL-AQ and in PIONEER were published before the United Kingdom's European Union Referendum vote (often referred to as the "Brexit" vote), where 52% of voters supported the United Kingdom's exit from the European Union. It is unclear what impact this result will have on future population growth. However, it is reasonable to suspect that this decision could slow population growth rates to some degree should the United Kingdom move ahead with this departure from the European Union.

For the sensitivity analyses presented in this section, two new sets of demand input values are used. The first new set of demand values (i.e. Group 2 demands) is disaggregated from national-level demand values using the same process as for the core results. That is, per capita demand as in 2010 for Greater London is used in conjunction with population projections from the Office of National Statistics. However, in this set of demand calculations, the ONS's "short-term migration" projections for Greater London were used to calculate total demand by year to 2050 in five-year time slices while the demand values for the United Kingdom are held constant. In turn, the total demand for the United Kingdom is the same in Groups 1 and 2 with the change being applied to the proportion of this demand that will exist in Greater London. This calculation is completed using the equations presented in Chapter 3.

For the third set of demands (Group 3), projections for road transport demand in Greater London are gathered from the United Kingdom Department for Transport (DfT) and Transport for London (TfL) (Transport for London (TfL), 2011; Department for Transport (Dft), 2014). This set of demand projections accounts for some of the other factors (i.e. beyond population) that influence transport demand in Greater London as previous discussed. As with Groups 1 and 2, this set of demand disaggregation calculations, total demand for road transport in the United Kingdom is held constant.

The outputs of all these calculations resulted in the demand values shown in Table 5.5, where the demand groups are defined as follows:

- Group 1: demand defined using ONS population projections under their "long term migration" scenario for Greater London
- Group 2: demand defined using ONS population projections under their "short term migration" scenario for Greater London
- Group 3: demand inputs defined using Department for Transport projections for Greater London

London (bvkm)												
	2010	2015	2020	2025	2030	2035	2040	2045	2050			
Group 1: ONS Long-Term												
Cars	23.70	25.17	26.41	27.47	28.37	29.91	31.53	33.24	35.04			
2W	0.70	0.74	0.78	0.81	0.84	0.88	0.93	0.98	1.03			
Buses	0.60	0.64	0.67	0.70	0.72	0.76	0.80	0.84	0.89			
LGVs	3.80	4.03	4.23	4.41	4.55	4.80	5.06	5.33	5.62			
HGVs	1.00	1.06	1.11	1.16	1.20	1.26	1.33	1.40	1.48			
TOTAL	29.80	31.64	33.20	34.55	35.67	37.61	39.65	41.79	44.06			
Group 2: ON	Group 2: ONS Short-Term											
Cars	23.70	25.31	26.88	28.26	29.42	31.39	33.50	35.75	38.14			
2W	0.70	0.75	0.79	0.83	0.87	0.93	0.99	1.06	1.13			
Buses	0.60	0.64	0.68	0.72	0.74	0.79	0.85	0.90	0.97			
LGVs	3.80	4.06	4.31	4.53	4.72	5.03	5.37	5.73	6.12			
HGVs	1.00	1.07	1.13	1.19	1.24	1.32	1.41	1.51	1.61			
TOTAL	29.80	31.82	33.80	35.53	36.99	39.47	42.12	44.95	47.96			
Group 3: DfT												
Cars	23.70	26.42	28.24	30.08	31.08	32.23	33.20	34.20	35.22			
2W	0.70	0.78	0.78	0.81	0.84	0.88	0.93	0.98	1.03			
Buses	0.60	0.64	0.67	0.70	0.72	0.76	0.80	0.84	0.89			
LGVs	3.80	4.69	5.32	5.94	6.53	7.05	7.57	8.14	8.75			
HGVs	1.00	1.00	1.03	1.06	1.09	1.13	1.16	1.20	1.23			
TOTAL	29.80	33.52	36.04	38.59	40.26	42.05	43.66	45.36	47.13			

Table 5.5 Demand in Greater London by Road Transport Type (2010-2050) for Group 1, 2, 3

Overall, this process resulted in the demand inputs shown in Figure 5.18. Of note here is the slightly different demand profile seen in Group 3 compared to Group 1 and 2. As previously discussed, Group 3 projections included the potential influence of non-population factors.



Figure 5.18: Demand in Greater London by Road Transport Type (2010-2050) for Group 1, 2, 3

For PM_{10} emissions, the outputs from PIONEER show that changing the demand inputs between the three groups resulted in a range in total air pollution emissions from 1.96 kilotonnes to 2.17 kilotonnes in 2050 for the scenario where emission factors are held constant (i.e. NoChange_NoChange scenario) as shown in Figure 5.19. For the scenario where emission factors follow UK trends (i.e. the UK_NoChange scenario), total PM_{10} emission levels ranged from 1.20 to 1.30 kilotonnes in 2050. Finally, for the scenarios were all tailpipe emissions were eliminated by 2050 (i.e. 50:50_NoChange, Doubling_NoChange, CleanLondon_NoChange and JustInTime_NoChange), total PM_{10} emissions ranged from 0.73 to 0.81 kilotonnes between the three demand groups. The remaining PM_{10} emissions in these scenarios come from the non-tailpipe emissions from cars and the combination of both tailpipe and non-tailpipe pollution from other types of road vehicles.

As shown in Figure 5.19, the scenarios resulted in three distinct clusters of outputs, demonstrating the relative importance of the technology transition pathway compared to the demand input assumptions in determining the final results for the range of demand inputs considered. Cluster A includes the outputs from the "NoChange_NoChange" scenario for the three demand groups, where emission factors were held constant from 2010. Cluster B includes

the results from the "UK_NoChange" scenario, where emission factors changed in line with the results seen in the United Kingdom for the lowGHG_DAMC scenario. Cluster C includes all of the scenarios where tailpipe emissions were eliminated in Greater London by 2050.





For $PM_{2.5}$ emissions, the outputs from PIONEER show that changing the demand inputs between the three groups resulted in a range in total air pollution emissions from 1.39 to 1.56 kilotonnes in 2050 for the NoChange_NoChange scenario as shown in Figure 5.20. For the UK_NoChange scenario, total $PM_{2.5}$ emission levels ranged from 0.67 to 0.73 kilotonnes in 2050. Finally, for the scenarios were all tailpipe emissions were eliminated by 2050, total $PM_{2.5}$ emissions ranged from 0.38 to 0.42 kilotonnes between the three demand scenarios. As with PM_{10} , the remaining $PM_{2.5}$ emissions in these scenarios come from the non-tailpipe emissions from cars and the combination of tailpipe and non-tailpipe pollution from other types of road vehicles. Furthermore, the scenarios again produce three distinct clusters of outputs, demonstrating the importance of the technology transition pathway in determining the final

 $PM_{2.5}$ levels in 2050 as shown in Figure 5.20.



Figure 5.20: Total $PM_{2.5}$ Emissions in Greater London area for scenarios without behavior change for a range of demand groups, 2010-2050

For NOx emissions, the outputs from PIONEER show that changing the demand inputs between the three groups resulted in a range in total air pollution emissions from 30.63 to 33.34 kilotonnes in 2050 for the NoChange_NoChange scenario as shown in Figure 5.21. For the UK_NoChange scenario, total NOx emission levels ranged from 2.55 to 2.77 kilotonnes in 2050. Finally, for the scenarios were all tailpipe emissions were eliminated by 2050, total NOx emissions were 0.38 to 0.42 in 2050 in all three demand scenarios as all NOx emissions from cars were eliminated. The remaining NOx emissions in these scenarios represent the NOx emissions from other types of road transport. As with particulate matter, the scenarios resulted in three distinct clusters of outputs, demonstrating the importance of the technology transition

pathway in determining the final NOx levels in 2050 for the range of demands considered as shown in Figure 5.21.



Figure 5.21: Total NO_x Emissions in Greater London area for scenarios without behavior change for a range of demand groups, 2010-2050

Overall, these results show that the assumptions used for both demand and emission factors in Greater London have significant impact on the outputs from PIONEER. However, the technology transition pathway is the primary driver of changes in pollution levels in 2050 in the range of demand values examined. That being said, demand assumptions had a more significant impact (from the standpoint of final total emission levels) for scenarios where high emission factors were assumed as shown by the larger spread in the final values on a total emissions basis in the NoChange NoChange scenario outputs.

Similarly, with regards to emission factors, one could explore the possibility of higher emission factors or, in the case of particulate matter, lower emission factors to include the possibility of reduced non-tailpipe emissions levels over time (for example, due to changes in braking

technology or driving patterns). However, the range explored above is considered sufficient for the work presented in this thesis as it is able to capture the impacts of an overall trend of decreasing emission factors over time relative to changes in demand. Furthermore, it is in line with the broader focus of this research on the co-impacts of decarbonisation on air pollution (as opposed to the impacts of targeted non-greenhouse gas pollution control measures).

5.5 Scenarios with Behavioural Change

This section includes results from those scenarios that focused on the behavioural change dimension where mode shifting away from car travel would occur in Greater London.

In the previous sections of this chapter, a set of six (6) scenarios were presented in order to explore the impacts of the technological change dimension. In this section, varying degrees of behavioural change (i.e. modal shift) are incorporated, producing a set of twelve (12) additional scenarios as shown in Table 5.6. These scenarios include up to a 40% mode shift away from cars in 2050 for reasons discussed in Chapter 3. Combined with the results in the previous section, these scenarios span the range of mode shift away from cars of 0-40%

Scenario Name	Technological Change	Behavioural
		Change
NoChange_80:20		20% of car travel
	No change from 2010	shifted
NoChange_60:40		40% of car travel
		shifted
UK 80:20	Follows trends in United Kingdom for the	20% of car travel
	lowGHG_DAMC scenario.	shifted
UK 60:40		40% of car travel
		shifted
50:50_80:20	Half the availability for zero-tailpipe emission	20% of car travel
	cars is taken by Greater London and the other half	shifted
50:50_60:40	is dispersed around the rest of the United	40% of car travel
	Kingdom until tailpipe emissions from cars are	shifted
	eliminated in Greater London.	
Doubling_80:20	After initially adopting half of the availability for	20% of car travel
	zero-tailpipe emission vehicles in 2025, the total	shifted
Doubling_60:40	use of zero-tailpipe emission cars doubles during	40% of car travel
	each five-year period until 2045, at which point	shifted
	the zero-tailpipe emission vehicles are adopted	
	rapidly to meet total demand for cars in 2050.	
CleanLondon_80:20	100% of zero-tailpipe emission cars that are	20% of car travel
	available in the United Kingdom are adopted in	shifted
CleanLondon_60:40	Greater London until all of demand for cars in this	40% of car travel
	urban area are met by zero-tailpipe emission	shifted
	vehicles.	
JustInTime_80:20	Adoption of zero-tailpipe emission cars in Greater	20% of car travel
	London is delayed until the last time period	shifted
JustInTime_60:40	considered (i.e. 2045-2050), at which point they	40% of car travel
	are adopted rapidly to meet total demand for cars.	shifted

Table 5.6: Scenarios Focusing on Behavioural Change Dimension

In practical terms, additional mode shifting could be achieved through penalty measures including congestion charging and bans on cars in Greater London. Conversely, incentive measures such as reduced cost public transport, improved access to convenient public transport options, extensive walking and cycling networks, and access to shared and low-cost bicycles could support increased mode shifting away from cars.

Shifting car travel to active travel (i.e. cycling, walking) and public transport (i.e. buses, trains) is represented in PIONEER as a reduction in demand for car travel that is linearly phased in over time from 2020 – 2050 to reach the indicated level of mode shift (i.e. 20% or 40%) by

2050. Demand values over time for each of these scenarios that incorporate behavioural change for Greater London is shown in Figure 5.22 for Group 1 demand values.



While these reductions in total demand due to mode shifting in Greater London are quite small relative to total car demand in the United Kingdom, these shifts do – by definition - impact the total demand assumptions used in UKTM-UCL-AQ. In turn, the lowGHG_DAMC scenario was re-run with the updated to reflect these reduction in expected future demand, resulting in the demand value projections shown in Tables 5.7 and 5.8 for 20% and 40% mode shifting, respectively.

Table 5.7 Updated Road Transport Demand Projections from UKTM-UCL-AQlowGHG_DAMC for the 80:20 Scenarios (Group 1)

bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Total Demand											
Cars	413	439	472	506	526	547	566	587	608		
2W	5.30	4.79	4.79	4.75	4.72	4.69	4.66	4.63	4.60		
Buses	4.80	4.97	5.13	5.29	5.43	5.56	5.67	5.78	5.89		
LGVs	68	74	84	94	104	112	121	131	141		
HGVs	27	27	29	30	31	32	33	35	36		
Change in Demand for Cars	0	0	1	2	3	4	5	6	7		
Compared to Original Run	0.0%	0.0%	0.2%	0.4%	0.6%	0.7%	0.9%	1.0%	1.1%		

Table	5.8:	Updated	Road	Transport	Demand	Projections	from	UKTM-UCL-AQ
lowGH	IG DA	AMC for th	e 60:40	Scenarios (Group 1)			

_			· · · · · · · · · · · · · · · · · · ·	1 /					
bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Total Demand									
Cars	413	439	471	504	523	543	561	581	601
2W	5.30	4.79	4.79	4.75	4.72	4.69	4.66	4.63	4.60
Buses	4.80	4.97	5.13	5.29	5.43	5.56	5.67	5.78	5.89
LGVs	68	74	84	94	104	112	121	131	141
HGVs	27	27	29	30	31	32	33	35	36
Change in Demand for Cars	0	0	2	4	6	8	10	12	14
Compared to Original Run	0.0%	0.0%	0.4%	0.8%	1.1%	1.5%	1.8%	2.0%	2.3%

As discussed in Chapter 3, the scenario constraints in PIONEER were subsequently checked for validity against the results from UKTM-UCL-AQ and the emission factors were recalculated. The updated emission factors and zero-tailpipe emission vehicles deployment curves can be found in the Appendix of this thesis. Overall, these deployment pathways were only minimally impacted by the change in demand assumptions. Of course, should these degrees of mode shifting be applied to the United Kingdom as a whole, one would expect a more significant impact. Noted here is that UKTM-UCL-AQ was re-run for each of the Greater London demand group assumptions that are used later in this chapter in the sensitivity test discussion.

Also, it should be noted that this process inherently assumes that there is capacity in the network to accommodate additional passengers both on the sidewalks and roads as well as within the public transport network (i.e. buses and trains). Any future work using detailed transport models should be cognisant of this assumption and its potential impacts on demand for transport infrastructure including roads and sidewalks, which are not captured in the energy systems model used in this work.

5.5.1 Results

In the case where technologies and their associated emission factors are held constant over time (i.e. the NoChange_80:20 and NoChange_60:40), particulate matter air pollution (PM_{10}) from road transport in Greater London rises over time from 1.33 kilotonnes in 2010 to between 1.52 and 1.74 kilotonnes in 2050 depending on the degree of modal shift assumed in the scenario as shown in Figure 5.23. For comparison, this value rose to 1.96 kilotonnes when no modal shift was assumed. For scenarios where Greater London follows the broader UK technology

transition pathway for the lowGHG_DAMC scenario (i.e. UK_NoChange, UK_80:20 and UK_60:40), emissions in 2050 ranged from 0.87 to 1.03 kilotonnes compared to 1.20 kilotonnes when no modal shift was included. For the other scenarios, where all tailpipe emissions from cars are eliminated by 2050 in Greater London, emissions in 2050 ranged from 0.59 to 0.66 kilotonnes compared to 0.73 kilotonnes when no modal shift was included. Again, the scenario outputs resulted in three distinct clusters, demonstrating the importance of the technology transition pathway in determining the final PM_{10} levels in 2050 for the range of modal shift considered here as shown in Figure 5.23, in particular for the scenarios where tailpipe emissions for cars are not eliminated by 2050.

Figure 5.23: Total PM_{10} Emissions in Greater London area for scenarios both with and without behavioural change, 2010 - 2050 (Group 1)



For PM_{2.5}, air pollution emissions in 2050 ranged from 1.09 to 1.24 kilotonnes in 2050 for scenarios without a technology transition where emission factors were held constant over time but a mode shift away from cars of up to 40% was included (i.e. NoChange_80:20 and NoChange_60:40) compared to 1.39 kilotonnes without modal shift as shown in Figure 5.24. For the scenarios where emissions followed the rest of the UK (i.e. UK_80:20 and UK_60:40), emissions in 2050 ranged from 0.48 to 0.58 kilotonnes compared to 0.67 kilotonnes without modal shift. For the other scenarios, where all tailpipe emissions from cars are eliminated by 2050 in Greater London, emissions in 2050 ranged from 0.31 to 0.35 kilotonnes when a modal shift was included versus 0.38 kilotonnes without modal shift.





For NO_x, air pollution emissions in 2050 ranged from 25.42 to 28.03 kilotonnes in 2050 for scenarios without a technology transition where emission factors were held constant over time (i.e. NoChange_80:20 and NoChange_60:40) compared to 30.63 kilotonnes without modal shift as shown in Figure 5.25. For scenarios that followed UK trends (UK_80:20 and UK_60:40), emissions in 2050 ranged from 1.68 to 2.11 kilotonnes versus 2.55 kilotonnes without modal shift. For the other scenarios, where all tailpipe emissions from cars are eliminated by 2050 in Greater London, emissions in 2050 was 0.39 kilotonnes. There was no range in this final value because, as previously discussed, all tailpipe emissions from cars are eliminated by 2050 and emissions from other vehicles are the same in each of these scenarios. The latter was done in order to isolate the impact of mode shifts away from car travel.

Figure 5.25: Total NO_x Emissions Greater London area for scenarios both with and without behavioural change, 2010 - 2050 (Group 1)



These scenarios help in understanding the tradeoffs and synergies between two measures for reducing air pollution emissions in Greater London – technology change to cleaner vehicles versus mode shifting away from car travel. Overall, for the range of mode shifting evaluated in this research (i.e. 0-40% away from cars) and the technology transition pathways evaluated here, technological change is the primary driver of reductions in emissions by 2050 as shown by the distinct clustering of results in Figures 5.23-5.25. This is particularly interesting because, on a kilometre by kilometre basis, mode shifting eliminates both tailpipe and non-tailpipe emissions whereas zero-tailpipe emission vehicles will still produce non-tailpipe emissions. However, these results show technological change driving the results more than mode shifting, even at a 40% level in 2050.

5.5.2 Sensitivity Analysis for Scenarios with Behavioural Change

As discussed for the scenarios that did not include any degree of modal shift, it is particularly worthwhile to explore the impacts of two factors - namely the inputs used to define the technological change pathway and demand in Greater London – on the results presented. In turn, as was done previously for scenarios without model shift, this section includes outputs from PIONEER that use alternative demand assumptions. These demands are displayed in Figure 5.26 and 5.27 for Group 2 and Group 3 demands, respectively.



Figure 5.26: Demand for Scenarios that Include Behavioural Change, 2010-2050 (Group 2)



As mentioned previously, changing the level of demand for cars in Greater London will, by definition, impact the demand assumptions used in UKTM-UCL-AQ. In turn, UKTM-UCL-AQ is re-run as earlier in this chapter using the methods outlined in Chapter 3. The updated car demand inputs that were included in UKTM-UCL-AQ for all demand groups in scenarios with behavioural shift are included in Figure 5.28 in order to illustrate the relative size of these changes to overall car demand in the United Kingdom. Noted here is that these scenarios only impacted car demand because, as previously noted, these scenarios assume that there is sufficient capacity in the system to accommodate these levels of shift to active travel (i.e. walking and cycling) as well as public transport (i.e. buses and trains).



Figure 5.28 Updated Car Demand Inputs for UKTM-UCL-AQ for All Groups in Scenarios with Behavioral Shift

Air pollution emissions over time are shown in Figures 5.29, 5.30 and 5.31 (PM_{10} , $PM_{2.5}$ and NO_x respectively) for all eighteen scenarios considered in this chapter for Greater London, including the three demand variants used in the sensitivity analysis for a total of fifty-four (54) sets of outputs. When looking at the results of the scenarios across all the technology pathways, behavioural change pathways, and demand groups, one still sees in three distinct clusters of outputs corresponding to the technological change dimension. These results demonstrate the relative importance of the technology transition pathway compared to the behavioural dimension in determining the final results for the range of demand inputs considered. The clusters include:

- Cluster A: outputs from the all the scenarios where the technology profile and corresponding emission factors were held constant from 2010, including three degrees of behavioural change (i.e. none, 20% shift from cars, 40% shift from cars) and three sets of demand assumptions.
- Cluster B: outputs from all scenarios where the technology profile changed in line with the national-scale lowGHG_DAMC scenario, including the same three degrees of behavioural change and three sets of demand assumptions as Cluster A.
- Cluster C: outputs from all scenarios where tailpipe emissions from cars were eliminated in Greater London by 2050, including the same three degrees of behavioural change and three sets of demand assumptions as Clusters A and B.

Emission factor and demand data for each of these scenarios and sensitivities can be found in the Appendix.

Figure 5.29: Total PM₁₀ Emissions in Greater London area for all scenarios and demand groups, 2010-2050



Figure 5.30: Total $PM_{2.5}$ Emissions in Greater London area for all scenarios and demand groups, 2010-2050







5.5.3 Public Health Impacts

As stated elsewhere in this thesis, the public health impacts considered in this work are limited to those resulting directly from changes in air pollution emission levels. There has been significant research on the public health benefits of increasing levels of active travel in urban areas (e.g. decreased obesity rates) that could be used in future work to analyse these potential benefits (Woodcock *et al.*, 2009; Jarrett *et al.*, 2012; Jensen *et al.*, 2013b).

As discussed in Chapter 3, reductions in the mortality burdens and corresponding life-year increases are calculated for each of the scenarios presented in this Chapter using previous work published by Walton, et. al. for $PM_{2.5}$ and NO_x (Walton *et al.*, 2015). The results are displayed in Figures 5.32 and 5.33 for changes in $PM_{2.5}$ and NO_x (as NO_2) pollution across all scenarios and demand groups used in the sensitivity tests shown in the previous sections.

In the "NoChange" scenarios where emission factors were held constant from 2010, net annual premature deaths in Greater London due to $PM_{2.5}$ air pollution produced by Greater London road transport increase from 377 in 2010 to 557 with 2050 emission levels when modal shift away from cars is not included (i.e. the NoChange_NoChange scenario). With a 20% mode shift away from cars (i.e. the NoChange_80:20), this value decreases to of 498. A 40% mode shift away from cars (i.e. the NoChange_60:40 scenario) further decreased this value to 439.

For the scenarios where emission factors decrease in Greater London in line with the rest of the United Kingdom, falling $PM_{2.5}$ emissions lead to a decrease in premature deaths due to $PM_{2.5}$ air pollution to 268 with 2050 emission levels when mode shifting is not included. Including a mode shift away from cars of 20% and 40% led to further decreases in these values to 231 and 194, respectively.

For the scenarios where emission factors decrease in Greater London until all tailpipe emissions from cars reach zero with the rest of the road transport fleet following UK trends (i.e. the 50:50_NoChange, Doubling_NoChange, CleanLondon_NoChange and JustInTime_NoChange scenarios), falling PM_{2.5} emissions lead to a decrease in premature deaths to 153 with 2050 emission levels when mode shifting is not included. Including a mode shift away from cars of 20% and 40% led to further decreases in these values to 139 and 125, respectively.

For nitrogen oxides (as nitrogen dioxide, NO₂), premature deaths increased from 2,448 in 2010 to 3,619 with 2050 emission levels when modal shift away from cars is not included (i.e. the NoChange_NoChange scenario). With a 20% mode shift away from cars (i.e. the

NoChange_80:20), this value decreases to of 3,312. A 40% mode shift away from cars (i.e. the NoChange 60:40 scenario) further decreased this value to 3,004.

For the scenarios where emission factors decrease in Greater London in line with the rest of the United Kingdom (i.e. UK_NoChange, UK_80:20 and UK_60:40), falling NO_x emissions lead to a dramatic decrease in premature deaths due to 301 with 2050 emission levels when mode shifting is not included. Including a mode shift away from cars of 20% and 40% led to further decreases in these values to 250 and 199, respectively.

For the scenarios where emission factors decrease in Greater London until all tailpipe emissions from cars reach zero with the rest of the road transport fleet following UK trends (i.e. the 50:50_NoChange, Doubling_NoChange, CleanLondon_NoChange and JustInTime_NoChange scenarios), decreasing NO_x emissions lead to a decrease in premature deaths to 46. As previously discussed, mode shifting away from cars did not impact these values as NO_x emissions from cars had already been eliminated with the adoption of zero-tailpipe emission vehicles.

Figure 5.32: Annual Premature Deaths by Scenario for PM2.5 and NOx (as NO2) in 2050 for Groups 1, 2 and 3



Annual Premature Deaths (Groups 1, 2 and 3)

Figure 5.33: Life Years Lost by Scenario for $PM_{2.5}$ and NO_x (as $NO_2)$ in 2050 for Groups 1, 2 and 3



Annual Life Years Lost (Groups 1, 2 and 3)

These total mortality burden values do not account for transboundary effects of changes to the road transport fleet in Greater London. In other words, this work does not consider the impacts of changes in air pollution emission levels from Greater London road transport on the rest of the United Kingdom. Should these values be calculated, they are expected to be in the same direction (positive/negative) as for Greater London though with different absolute values depending on the region considered. Furthermore, as discussed in Chapter 3, only a small portion of these health impact changes would be realised in 2050. Longer-term benefits (e.g. through reductions in instances of cancer and other diseases resulting from long-term exposure to these air pollutants) would be increasingly realised over time after 2050 (Walton *et al.*, 2015).

5.6 Discussion & Conclusions

As discussed at the outset of this Chapter, the research presented for the Greater London area explores the extent to which local action in this urban area could contribute to reduction in locally produced air pollution and its associated health impacts. This research takes into consideration two dimensions – technological and behavioural change – and three key primary non-greenhouse gas air pollutants – namely, particulate matter (PM_{10} and $PM_{2.5}$) and nitrogen oxides (NO_x).

After creating the PollutION Emissions from EneRgy (PIONEER) model, it was soft-linked to the UKTM-UCL-AQ energy systems model in order to disaggregate the Greater London region transport sector from the broader United Kingdom. The air pollution and public health impacts of a range of scenarios were analysed to establish the relative impacts of technological versus behavioural change on air pollution emissions from Greater London road transport as well as the resulting health impacts. Overall, it was found that technological change was the primary driver of changes in air pollution emissions and public health across all scenarios and data assumptions considered.

These results are discussed in more detail in Chapter 6.

This page was intentionally left blank.

Chapter 6 – Discussion and Conclusions

6.1 Overview and Key Contributions

This thesis documented the development and use of a technoeconomic energy systems optimisation model (UKTM-UCL-AQ) to quantify the co-impacts of technological transition pathways to achieve decarbonisation targets on air pollution and vice versa for the United Kingdom. This manuscript further documented the development of an air pollution and public health tool (PIONEER) and its subsequent soft-linking to UKTM-UCL-AQ in order to disaggregate the Greater London area from national-level outputs. Finally, this thesis documented the combined use of these tools to both quantify and improve understanding of the air pollution and public health implications of energy system transition pathways to identify "win-win" opportunities.

There are three set of key contributions that were directly made by this work. The first includes the creation of fit-for-purpose tools that allow for the detailed examination of the co-impacts of climate change mitigation efforts in the energy system on other types of air pollution and vice versa at a national scale. Through the development of UKTM-UCL-AQ to include other types of air pollution, this research produced a unique tool to allow the researcher to explore and quantify these co-impacts in the United Kingdom. In turn, it achieved its goal of helping to quantify and understand the synergies and trade-offs between climate change mitigation and air pollution reduction efforts in the United Kingdom.

Second, this research produced a unique tool that allowed for the disaggregation and evaluation of these co-impacts on road transport in an urban area. Through the development of the PIONEER model, this research achieved the goal of understanding the impacts of national scale decarbonisation ambitions on urban air pollution as outlined in Chapter 1. It further enabled the evaluation of the relative impacts of technological versus behavioural change on air pollution from road transport in the Greater London area, revealing the importance of technological change in driving air pollution emission reductions.

Third, through the soft-linking of UKTM-UCL-AQ and PIONEER, this research allowed for the evaluation of the extent to which technological and behavioural change could facilitate air pollution reductions and public health gains in the Greater London urban area. In doing so, it bridged some of the existing gaps between energy system and public health models in the literature as discussed in more detail elsewhere in this chapter as well as in Chapters 1 and 2.

Overall, the research presented in this thesis allowed for the exploration of the research questions posed at the beginning of this thesis, including:

- What are the co-impacts (both positive and negative) on particulate matter and nitrogen oxide air pollution levels for energy sector decarbonisation pathways that are optimised with regards to reducing total greenhouse gas emissions on both a national and urban scale?
- 2. How does considering the impact of these other types of outdoor air pollution (i.e. particulate matter and nitrogen oxides) impact the decarbonisation pathway on both a national and urban scale?

The results from this work are encouraging, as they suggest that there are numerous opportunities for climate change midication and air pollution reduction efforts to be mutually supportive. They also strengthen the evidence base related to the importance of considering air

pollution co-impacts in the evaluation of potential pathways for achieving decarbonisation goals in order to avoid tensions between mitigation and air pollution reduction efforts.

This chapter starts with a discussion on the key insights gained from this research and how they compare with previous work by other researchers (Section 6.2). This section is followed by a discussion of the significance of this work for the modelling community and in policy development (Section 6.3) and an examination of the limitations of the approach taken (Section 6.4), including discussion of the ways that this research can be enhanced and expanded moving forward (Section 6.5).

6.2 Key Insights and Comparisons with Previous Work

This section includes the key insights from this research, presented in the context of the research questions explored. This discussion is followed by a critical comparison of this research and its key insights with previous work in this field, including studies presented in Chapter 2.

6.2.1 Key Insights

As discussed in the preceding chapters, initial focus in this research project was placed on a set of six (6) non-greenhouse gas air pollutants that are inventoried in the National Atmospheric Emissions Inventory (NAEI), including particulate matter (PM_{10} and $PM_{2.5}$), nitrogen oxides (NO_x as NO_2), sulphur oxides (SO_x as SO_2), ammonia (NH_3), and non-methane volatile organic compounds (NMVOCs). These pollutants were integrated into a national-scale TIMES-based energy systems model (UKTM-UCL-AQ), which was able to account for the majority of emissions from PM_{10} , $PM_{2.5}$, NO_x , and SO_x but only a small minority of NH_3 and NMVOCsfor the United Kingdom as discussed in Chapter 4. The UKTM-UCL-AQ model was used to evaluate the co-impacts of energy technology transition pathways on air pollution emissions levels in the United Kingdom. Subsequently, the associated damage costs of the non-greenhouse gas air pollutants were included in the cost-optimisation in UKTM-UCL-AQ to quantify and understand how including these co-impacts might affect the transition pathways. Overall, a set of six (6) scenarios were constructed across two dimensions – decarbonisation ambition and air pollution costs – and the sensitivity of model outputs was examined with regards to constraints placed on biomass and nuclear power.

At an urban level, focus was placed on the Greater London urban area and three key primary non-greenhouse gas air pollutants – namely, particulate matter (PM_{10} and $PM_{2.5}$) and nitrogen oxides (NO_x). These pollutants represent a significant portion of estimated local public health impact of air pollution and are largely captures by the UKTM-UCL-AQ energy systems model as discussed in Chapters 3, 4 and 5. This research then explored a set of eighteen (18) scenarios to explore the relative impacts of technological change and behavioural change as well as a set of sensitivity runs as discussed in Chapter 5.

Overall, this research revealed that the UKTM-UCL-AQ model could represent the vast majority of NO_x and SO_x pollution sources in the United Kingdom, a majority of PM_{10} and $PM_{2.5}$ pollution, and a minority of NH_3 and NMVOCs pollution as discussed in Chapter 4. In turn, use of UKTM-UCL-AQ could provide significant insights on the co-impacts of national scale decarbonisation efforts on the first four pollutants (NO_x , SO_x , PM_{10} and $PM_{2.5}$). Conversely, much less could be said about trends in NH_3 and NMVOCs as the dominate sources of these types of pollution are external to the energy system in the United Kingdom (i.e. outside of the system boundary)

The subsequent implementation of UKTM-UCL-AQ across a range of decarbonisation ambitions showed the potential for tensions to develop between decarbonisation and air pollution reduction efforts. This tension arose from increasing levels of particulate matter air pollution between 2025-2040, resulting from the use of biomass in residential heating systems. However, this tension was alleviated with the inclusion of damage costs in the optimisation pathway, which provided a strong indicator of the importance of considering air pollution co-impacts in the development of the transition pathways to achieve decarbonisation goals.

Furthermore, the inclusion of air pollution damage costs had a quite small impact on total energy system costs in the scenario where the United Kingdom achieved its national scale decarbonisation ambition. In fact, if one removes the air pollution damage cost component from the total system costs, the additional costs of the energy system expenditure amounted to a 0.15% to 0.5% increase (Lott, Pye and Dodds, 2017).

That being said, these results were sensitive to the assumptions used in the UKTM-UCL-AQ model, in particular related to the degree to which fuel-shifting could occur in the residential sector. However, the results did not appear to be sensitive to constraints on nuclear fleet expansion in the United Kingdom. This was a particularly interesting result given that the electricity sector was a primary driver of decarbonisation in the decarbonisation scenario.

The results from UKTM-UCL-AQ also further reinforced evidence on the relative difficulty in achieving significant greenhouse gas reductions in the transport sector compared to other sectors (e.g. electricity) from the viewpoint of this cost-optimised model.

As discussed in Chapter 4, the transport sector was the least impacted of the sectors considered within the energy system. Though, it is noted that the inclusion of damage costs did accelerate the transition to lower emission vehicles. This result suggests that targeted polices and/or significant cost reductions would be needed to address transport-sector air pollution impacts.

As discussed in Chapter 1, the current predominance of road transport as source of local air pollution in urban areas makes this sector of high interest (Woodcock *et al.*, 2009; Sokhi and Kitwiroon, 2011). The results in Chapter 4 further heightened this interest as the transport sector only realised small changes in the technological change pathway. In turn, the focus of Chapter 5 explored the extent to which local action in Greater London could contribute to reductions in locally produced air pollution and its associated public health impacts.

Overall, across the scenarios considered in Chapter 5, the principle driver of air pollution emissions changes was the technological pathway taken in Greater London. This finding held across a range of behavioural change including significant mode shift away from cars as well as sensitivity tests that focused on demand levels for road transport in Greater London.

With regards to public health impacts, the results presented in Chapter 5 illustrated the increasingly dominate role of non-tailpipe emissions in premature deaths due to $PM_{2.5}$ emissions. However, this was overshadowed by the relative importance of nitrogen oxide emissions produced in Greater London on public health. Overall, the results showed that reducing tailpipe emissions from Greater London road transport offers the largest potential health opportunity. Furthermore, as previously discussed, given the recent questions that have been introduced regarding the ability of auto manufacturers to reduce tailpipe emissions of

nitrogen oxides, these results further strengthen arguments for transitioning more quickly to zero-tailpipe emission vehicles (Brand, 2016).

6.2.2 Comparison of United Kingdom Results with Previous Work

As discussed in Chapter 2 (Section 2.3.2), the global study by the International Energy Agency and reports by Pye, et. al. are of particular importance in this research project. The former represents the most advanced work done globally to examine the co-impacts of energy system transitions to reach climate change mitigation targets under the Paris Climate Agreement in a way that considers the air pollution co-impacts. The latter is the most advanced work that has been done in the United Kingdom to analyse the co-impacts of changes to the energy system on air pollution levels using the energy systems model that is at the core of UK government decision making (Pye and Palmer, 2008; Pye *et al.*, 2008; International Energy Agency (IEA), 2016).

6.2.2.1 Comparison with IEA (2016)

At a high level, the IEA's 2016 report quantifies the relative role of the global energy system in the production key types of primary air pollution (International Energy Agency (IEA), 2016). Overall, they found that essentially all of global SO₂ and NO₂ emissions are produced by the energy sector as well as more than 85% of particulate matter (PM_{2.5}). Conversely, the IEA reports that only 3% of global NH₃ emissions are produced by the energy sector.

Similarly, in the development of UKTM-UCL-AQ as described in Chapter 3, this research found that this energy systems model included the sources of the vast majority of NO_x (as NO₂) and SO_x (as SO₂) pollution in the United Kingdom, capturing 94% and 92% of these emissions, respectively. The model also included a majority (74%) of PM_{2.5} pollution. However, UKTM-UCL-AQ only captured a small minority (5%) of NH₃ pollution (Lott, Pye

and Dodds, 2017). These results further reinforce the important role of the energy sector in air pollution production, as well as give an indication of the relative role of energy for each of these key pollutants.

Furthermore, in the IEA's 2016 report the authors find that - in a scenario that includes the Intended Nationally Determined Contributions (INDCs) that have been pledged by countries around the globe - incorporating air pollution mitigation measures results in an early peak in carbon dioxide emissions around the globe (International Energy Agency (IEA), 2016). In their scenarios, targeting air pollution reductions led to a co-benefit in the form of a 13% reduction in total CO₂ emissions in 2040 in the European Union (International Energy Agency (IEA), 2016). Noted here is that the current INDCs generally only include ambitions through 2025 or 2030 and are not expected to achieve the overall goals set forth in the Paris Climate Agreement of limiting global average temperature rise to less than 2 Degrees (Pye *et al.*, 2017).

Similarly, in the research presented in this thesis, modelling showed that the inclusion of damage costs in a scenario with medium climate ambition (i.e. the ref_DAMC scenario) leads to accelerated decarbonisation compared to the same scenario without damage costs (i.e. the ref scenario). This acceleration spanned the period from 2020-2035. This result supports the IEA's finding that attention to air pollution can support accelerated decarbonisation. Though, it is noted here that this accelerated decarbonisation was not observed in the research presented in this thesis in a scenario where a more ambitious decarbonisation target was included (i.e. the lowGHG_DAMC scenario).

Also of note is that actions to reduce non-greenhouse gas emissions in tandem with climate change mitigation were shown to be relatively inexpensive in both the IEA study and the research presented in this thesis in Chapter 4. In the case of the IEA study, a 7% (\$4.8 trillion) increase in total global energy system investment between now and 2040 could significantly reduce global premature deaths due to air pollution exposure (International Energy Agency (IEA), 2016). In this research, scenarios that considered air pollution damage costs resulted in a minimal cost increase of 0.15% to 0.5% while achieving significant decreases in air pollution levels as shown in Chapter 4, Figure 4.24 and Table 4.7. Highlighted in the context of this comparison is the fact that the IEA included a set of targeted air pollution abatement technologies and policies in their analysis that were not explicitly considered in the scope of the research project presented in this thesis.

6.2.2.1 Comparison with Pye, et. al. (2008)

As discussed in Chapter 2, the 2008 report by Pye et. al. accounted for non-greenhouse gas air pollutants in the United Kingdom MARKet ALlocation model (UK MARKAL), which was the precursor to UKTM-UCL (Pye and Palmer, 2008; Pye *et al.*, 2008). In this work, Pye and his co-authors tracked air pollution emissions of particulate matter (PM_{10}), sulphur dioxide (SO_2), and nitrogen oxides (NO_x) arising from all combustion processes in the energy sector. As was done in this research presented in this thesis, Pye et. al. also included marginal damage cost values from the United Kingdom's Department for Environment, Food & Rural Affairs (Defra) in their model, using the 2006 National Atmospheric Emissions Inventory (NAEI) emission factors as far as possible (Pye *et al.*, 2008).

The UK MARKAL model was then applied to two model runs using scenarios released as a part of the 2007 Energy White Paper – namely, the base scenario and a decarbonisation scenario that included a 30% greenhouse gas emission reduction target by 2030 and a 60% reduction target by 2050 (compared to 2000 levels) – in order to "provide examples of outputs" that could be produced using this tool (Department of Energy and Climate Change (DECC),

2007; Pye *et al.*, 2008). This application was similar – though not the same – as the scenarios applied in this research, which included an 80% greenhouse gas reduction target in 2050 and included the interim targets as set out in the UK carbon budgets.

Overall, Pye et. al. concluded that their "analysis shows that air quality emissions could be significantly reduced in future years as a result of technology improvements, improved efficiency and less use of polluting fuels, under the reference case" (Pye *et al.*, 2008). They observed a further reduction of NO_x emissions in 2050 of 131 kt (24.6%) in the decarbonisation scenario versus the base case as well as additional reductions of 11.6 kt for PM₁₀ and 287 kt for SO₂ (Pye *et al.*, 2008). By comparison, the results from UKTM-UCL-AQ presented in this research showed a difference in NO_x emissions between the base and lowGHG scenarios of 125 kt (25%) as well as a 41 kt reduction in PM₁₀ emissions compared to a 2010 base year. As was the case in the results presented in this thesis, Pye et. al. found that transport was only minimally impacted by the inclusion of decarbonisation targets and damage costs for other types of air pollution (Pye *et al.*, 2008).

While the results this study by Pye, et. al. are not directly comparable to the results presented in this thesis in Chapter 4due to differences in the scenario constraints (e.g. the differences in decarbonisation targets, base years, cost assumptions, etc). However, one can still gain insights from comparing the overall trends. In turn, it is noted here that both sets of results support the hypothesis that decarbonisation of the energy system would lead to reductions in other types of air pollution over time. Combined, these studies strengthen the conclusion that targeted policies are needed to effectively reduce the impacts of air pollution from the transport sector. Though, the work presented in this thesis further highlights potential tensions relating to the increased use of biomass.

6.2.3 Comparison of Greater London Results with Previous Work

For Greater London, the results presented in Chapter 5 illustrated the relative importance of technological and behavioural change in reducing air pollution produced within this urban area. Overall, it was shown that technological change was the primary driver of pollution emissions reductions across the range of scenarios considered, though behaviour change did have noticeable impacts. Furthermore, deploying an increased proportion of available zero-tailpipe emission cars (both electric and hydrogen fuel cell) in Greater London had significant air pollution health benefit resulting from decreases in local pollution levels. These benefits outpaced gains realised in scenarios where up to a 40% mode shift away from cars was included.

6.2.3.1 Comparison with Barker et. al (2010)

These conclusions are in broad agreement with those previously made by Barker, et. al in their 2010 study of global mitigation efforts, with a focus on Mexico and within Mexico City. In this study, Barker et. al. concluded that (Barker *et al.*, 2010):

"climate control in the form of rapid decarbonisation of the Mexican economy will have substantial effects on air pollution, at no extra cost, especially if the mitigation actions are focused on Mexico City".

As discussed in Chapter 2, this study by Barker, et. al. differs from the research presented in this thesis not just in terms of modelling approach but in choice of air pollutants to examine (Barker et. al. focused on greenhouse gas emissions and tropospheric ozone). However, they support the findings in this research at a high level in that mitigation efforts resulted in substantial benefits for air pollution at minimal (or no) extra cost. Furthermore, there are substantial benefits to be gained by concentrating mitigation actions in urban areas in both

studies as shown in Chapter 5 of this thesis in the scenarios that included increased adoption of zero-tailpipe emission vehicles in Greater London.

6.2.3.2 Comparison with Woodcock, et. al (2009)

With regards to the health co-benefits from climate change mitigation in transport, the results presented in this thesis in Chapter 5 support the previous findings by Woodcock, et. al. in their 2009 study. As discussed in Chapter 2, these researchers found that "although uncertainties remain, climate change mitigation in transport should benefit public health substantially" (Woodcock *et al.*, 2009).

Furthermore, the research presented in this thesis is complimentary to the work by Woodcock et. al., given that their 2009 evaluation did not consider non-tailpipe or nitrogen oxide emissions. As discussed in Chapter 5, non-tailpipe emissions are increasingly important over time as they are not eliminated with the adoption of zero-tailpipe emission vehicles. Furthermore, nitrogen oxide air pollution provided substantial health co-benefits in Greater London that were not included in the work by Woodcock, et. al. in their 2009 study (Woodcock *et al.*, 2009).

6.2.3.3 Comparison with Jarrett, et. al. (2012)

With regards to the work discussed in Chapter 2 by Jarrett et. al., the results presented in this thesis serves to complement their work related to the co-benefits of increased active travel in urban England and Wales for the National Healthcare Service. As stated in Chapter 2, Jarrett et. al. did not consider the effect of mode shifting "on environmental factors such as improved air quality because of reduced vehicle emissions" (Jarrett *et al.*, 2012).
However, further work would be required in order to fully interlink this research with Jarrett et. al.'s work. In particular, the methodology applied in Chapter 5 would need to be applied to all urban areas in Wales and England in addition to disaggregating the effects of mode shifting to active travel (i.e. walking, cycling) versus public transport (i.e. buses and trains) and expanded to associate direct treatment costs for the NHS to the health co-impacts realised. This activity represents an interesting opportunity for future work in this area.

6.3 Significance for the Modelling Community & Policy Development

As discussed in Chapter 2, researchers have discussed that the potential co-benefits of energy sector decarbonisation on other types of air pollution, hypothesizing that it could be an important benefit of decarbonisation activities. The value in the research presented in this thesis is that these air pollution co-impacts can now be quantified directly when exploring decarbonisation pathways using the same energy systems model (UKTM-UCL-AQ) at a national scale for the United Kingdom. In turn, it provides quantitative outputs instead of more nebulous impressions, allowing researchers to understand the potential implications of policies targeting decarbonisation or air quality in order to eliminate potential tensions between these types of policies and identify "win-win" opportunities. Noted here is that the UKTM-UCL-AQ model is now being used by the former Department of Energy and Climate Change (DECC), which became the Department for Business, Energy & Industrial Strategy (BEIS) in July 2016.

Furthermore, this research enabled the disaggregation of an urban areas from the national level outputs produced by UKTM-UCL-AQ using the PIONEER model. This disaggregation provides additional insights that are particularly pertinent to air quality policy development, where urban areas often represent air quality "hot spots", as is the case with Greater London as discussed in Chapters 1 and 2. For the policy community, the outputs of this research relating

specifically to the road transport sector highlight the need for more targeted solutions (e.g. policies and regulation) in order to support this sector's transition. It also showed the relative importance of technological and behavioural change in reducing local air pollution emisisons.

6.4 Limitations of the Research Approach

At the core of this research are two models, as discussed in more detail in Chapter 3 with details of their application in Chapters 4 and 5. The first of these tools is an energy systems model (UKTM-UCL-AQ) that includes a simplified version of the United Kingdom's energy system, allowing its users to explore the many "what ifs" of energy systems planning and development from a cost-optimised viewpoint without extraordinarily high levels of computational intensity. The second tool is an air pollution and public health model (PIONEER) allows users to further understand the co-impacts of national and local efforts on air pollution and public health. Neither of these models is designed to predict the future, but rather to gain insight on the possible pathways that could be taken to achieve a variety of goals.

As discussed in Chapter 3, the choice of the particular models used in this research was predominately motivated by its:

- 1. being fit-for-purpose
- 2. having in-country capacity
- 3. transparency, communicability and policy credibility

The simplified representations in UKTM-UCL-AQ and PIONEER can provide significant insights into the real-world energy system and the potential impacts of its evolution on air

pollution and public health. There are many reasons for these simplifications, including the following practical considerations:

- 1. resource constraints
- 2. availability of data
- 3. hardware and software access

However, the modelling approach taken in this research – as with any modelling approach – limits the types of insights that can be drawn. These limitations and key considerations to be away of are discussed below.

6.4.1 Cost Optimisation Approach

The optimisation process in UKTM-UCL-AQ is designed to provide insights on the possible pathways to achieve a future energy system that will meet a set of exogenously prescribed demands at a minimum total system cost. Because of its approach, changes in technology costs can have dramatic impacts on the technology transition pathway results – a concept sometimes called the "penny switching" or "bang bang" effect where "small changes in input parameters might lead to considerable modifications in the output" (Held, 2010; Pfluger, 2014). In turn, the assumptions made for these future energy service demands, resource availability, and technology costs (including initial investment, operation and maintenance) are of primary importance in determining the outputs from this model.

6.4.1.1 Energy Service Demand Assumptions

With regards to energy service demands, the core research presented in this thesis used future population growth rate projections from national statistics as the primary driver of demand as discussed in Chapters 3, 4 and 5. These statistics were published prior to the 2016 Brexit vote and the ongoing negotiations related to the United Kingdom's potential departure from the European Union. This is just one example of a source of potential uncertainty for future population trends and the corresponding energy service demands, which supports the completion of the sensitivity tests presented in Chapter 5.

6.4.1.2 Resource Availability and Technology Costs Assumptions

With regards to resource availability and technology costs, inputs to the UKTM-UCL-AQ model included both the availability of resources and their associated costs as well as the technology costs for both initial capital investments and on-going operation and maintenance of each technology. As discussed in Chapter 3, these inputs included factors such as construction timelines, limits on change rates (e.g. fuel switching), learning curves and future innovations including the availability and costs of future technologies. The values used in this research project are in line with other analyses completed using UKTM-UCL to further support transparency and credibility (Committee on Climate Change, 2015a; Pye *et al.*, 2015; Lott, Pye and Dodds, 2017). However, awareness of the role of these assumptions is important both in determining the insights that can be drawn from this research as well as in designing approaches to future work.

In this work, awareness of this sensitivity led to the use of identical cost assumptions across all mode runs presented in Chapter 4 in addition to using assumed values that are in line with previous peer-reviewed studies in order to isolate the co-impacts of the decarbonisation ambition and how these co-impacts area affected when pollution damage costs are included. Furthermore, in the analysis presented in Chapter 5, hydrogen fuel-cell and electric vehicles were aggregated into a "zero-tailpipe" classification. In turn, the conclusions presented in this chapter are dependent on either of these two technologies being deployed and so are not influenced by the relative prices of one of these two car types and the other. That being said, this aggregation does not eliminate the sensitivity of the deployment of these zero-tailpipe cars to their assumed cost relative to internal combustion engine models.

6.4.1.3 Emission Factor and Damage Cost Assumptions

The UKTM-UCL-AQ model also used a series of assumptions related to the emission factors for technologies as described in Chapter 3 as well as the damage costs of these emissions. With regards to the emission factors, this research drew from the National Atmospheric Emissions Inventory (NAEI). As discussed in Chapter 3, the emission factors for future technologies were often based on existing values, which might be inaccurate. Furthermore, assumptions were made related to the lifetime for each current and future technology as well as its emissions performance over that lifetime. For the scenarios that included damage costs in the optimisation pathway (i.e. base_DAMC, ref_DAMC and lowGHG_DAMC), the emission factors are of particular importance.

The NAEI was appropriate for use in this research as it provided a transparent and accessible centralised source of emission factors data. Furthermore, using this database allowed for comparisons of this work to previous efforts by Pye, et. al. (Pye and Palmer, 2008; Pye *et al.*, 2008). As the NAEI database is updated on an annual basis, it would be valuable to examine how updated information on these emission factors in the future impact the outputs in this thesis (Department for Business Energy & Industrial Strategy (BEIS), 2016). There also exist ample opportunities for future work to examine these emission factors and the impact of uncertainty

in the published figures in the light of the recent "dieselgate" scandals as was discussed in Chapter 2 (Brand, 2016).

With regards to damage cost values, as discussed in Chapters 3 and 4, the values assumed in this work for the damage cost values of air pollution were based on previous impact pathway assessments, which explicitly accounted for air pollution and exposure profiles. These damage costs are a more direct way to place an economic value on the impacts of air pollution on both public health and the environment (including both buildings and materials) in UKTM-UCL-AQ, and therefore are more straightforward to include in the optimization process (Lott, Pye and Dodds, 2017).

Crucially, the damage costs approach does factor in the spatial distribution of air pollution and the likely exposure. It is therefore appropriate to use such nationally-derived damage costs values in a model such as UKTM-UCL-AQ. While recognised as a credible approach for policy appraisal, the limitation in using these values is the implicit assumption that such damage cost values hold for future years, in which this spatial distribution of pollution–exposure–impact may change (Lott, Pye and Dodds, 2017).

These values also make assumptions related to healthcare costs for treating conditions resulting from air pollution exposure. By holding these costs constant over time in real terms, this approach does not capture the effects of healthcare and treatment innovations nor changes in standards of care in the healthcare system. These and related topics are beyond the scope of the research presented in this thesis and represent an interesting opportunity for future collaborations and work in partnership with the medical research community.

6.4.2 Spatial Resolution & System Boundaries

As discussed in Chapter 3, with regards to spatial and temporal resolution, the tools selected and developed for this research examine changes in the energy system in five-year time slices with country- and urban-level resolution. In turn, they are appropriate for the quantification of trends on these scales, which is reasonable given the geographic focus and timescale of the United Kingdom's Climate Change Act. However, these tools are not appropriate for use in examining a number of related research questions that require quite higher levels of spatial and/or temporal resolution nor those requiring detailed air pollution chemistry modelling (for example, the co-impact of changing car technologies on air pollution levels on a particular street in Greater London and their corresponding impact on hourly or daily mean air pollution concentrations).

With regards to system boundaries, UKTM-UCL-AQ draws its boundaries around the United Kingdom energy system in each of the scenarios presented in Chapter 4. These boundaries have particularly important implications in this work as it relates to sources of air pollution and the resulting air quality impacts. More specifically, this boundary means that sources of air pollution that exist outside of either the 1) energy system and/or 2) geographical boundaries drawn (i.e. the United Kingdom, Greater London area) are held constant over time. In turn, insights are limited to potential co-impacts of energy systems changes that occur within these system boundaries and do not account for changes outside of their limits.

As discussed elsewhere in this manuscript, the energy system processes included within the system boundaries of UKTM-UCL-AQ represent the majority of man-made particulate matter, nitrogen oxide, and sulphur oxide pollution produced in the United Kingdom (Lott *et al.*, 2016; Lott, Pye and Dodds, 2017). This fact, coupled with the existing body of scientific evidence

related to the direct health impacts of air pollution, mean that the most significant insights of this research related to public health are associated with changes in particulate matter and nitrogen oxide air pollution emission levels (World Health Organization, 2013b; Walton *et al.*, 2015). Future work to enhance this research could include its expansion beyond the energy system or to otherwise account for other air pollution sources.

For the PIONEER model, system boundaries were drawn around the Greater London area as described in Chapter 3. In turn, direct insights resulted from those changes occurring inside of this urban area. However, when coupled with UKTM-UCL-AQ, additional insights could be drawn relating to the impact of national-level action on urban-level air pollution and public health as discussed in Chapter 5. Furthermore, the soft-link (including an iterative loop) between PIONEER and UKTM-UCL-AQ allowed for insights to be drawn on the impact of certain types of actions (e.g. mode shifting) within Greater London on national-level decarbonisation pathways.

6.4.3 Temporal Resolution

As discussed in Chapter 3 with regards to temporal resolution, the tools selected and developed for this research examine changes in the energy system in five-year time slices. In turn, they are appropriate for the quantification of trends over longer periods of type, which is appropriate given the geographic focus of the United Kingdom's Climate Change Act. However, these tools would not be appropriate for use in examining a number of related research questions that require quite high levels of temporal resolution nor those requiring detailed air pollution chemistry modelling. For example, the impact of air pollution from cars on hourly, daily or even seasonal mean air pollution concentrations. Furthermore, this course level of temporal resolution limits the conclusions that can be made with regards to operational viability of the technology transitions output.

6.5 Opportunities for Future Work

Given an appropriate level of resource availability, the logical next steps in this research include the following:

- Expand work relating to the air quality and exposure impacts of changes in air pollution levels.
- 2. Expand work to consider energy transitions outside of the United Kingdom to improve cross-boundary pollution assumptions.
- 3. Increase the temporal granularity of these modelling efforts, to capture additional insights related to the operational implications of these future scenarios.
- 4. Soft-link the outputs of these models with an air quality tool to explore the implications of the scenarios on hourly and daily mean air pollution concentrations.
- 5. Increase spatial granularity of models to study the implications of changing travel patterns on air quality and public health.
- 6. Further develop data on emission factors, in particular for future technologies.
- 7. Explore the potential impacts of energy sector technology innovation, including the car-sharing economy and the impacts of changes in future travel patterns (e.g. through increased remote working).
- 8. Explore the potential impacts of technology innovation in the road transport sector, especially as it pertains to impacts on non-tailpipe emissions.
- 9. Improve the representation of explicit air pollution abatement technologies by including these technology options in UKTM-UCL-AQ.

10. Disaggregate the effects of mode shifting to active travel (i.e. walking, cycling) versus public transport (i.e. buses and trains) to allow for a more granular discussion on behavioural shifts.

From a publication standpoint, the London-specific work presented in Chapter 5 should be used as the basis of an academic journal paper that considers the implications of these results on the country's compliance with the National Emissions Ceiling Directive. This work is currently being pursued by the author of this thesis. Furthermore, the results presented in both Chapters 4 and 5 should be used as the basis for a paper discussing the economic impact of these technology transitions on the United Kingdom's National Healthcare System (NHS).

Bibliography

Abbott, J., Conolly, C., Cooke, S., Passant, N., Stewart, R. and Wagner, A. (2012) Assessment of the effectiveness of measures under the Clean Air Act. Available at: https://ukair.defra.gov.uk/assets/documents/reports/cat05/1303201551_20072012-AEA-Report-CAA.pdf.

Amann, M., Bertok, I., Borken, J., Cofala, J., Heyes, C., Hoglund, L., Klimont, Z., Purohit, P., Rafaj, P., Schöpp, W., Toth, G., Wagner, F. and Winiwarter, W. (2009) *GAINS*. Available at: http://gains.iiasa.ac.at/gains/reports/AnnexI-results.pdf.

Anenberg, S. C., Schwartz, J., Shindell, D., Amann, M., Faluvegi, G., Klimont, Z., Janssens-Maenhout, G., Pozzoli, L., Van Dingenen, R., Vignati, E., Emberson, L., Muller, N. Z., West, J. J., Williams, M., Demkine, V., Hicks, W. K., Kuylenstierna, J., Raes, F. and Ramanathan, V. (2012) 'Global air quality and health co-benefits of mitigating near-term climate change through methane and black carbon emission controls.', *Environmental health perspectives*, 120(6), pp. 831–9. doi: 10.1289/ehp.1104301.

Ashmore, M., de Vries, W., Hettelingh, J.-P., Hicks, K., Posch, M., Reinds, G. J., Tonneijck, F., van Bree, L. and van Dobben, H. (2011) 'Environmental and Health Impacts of Air Pollution', in *World Atlas of Atmospheric Pollution*, pp. 77–93.

Atkinson, R. W., Kang, S., Anderson, H. R., Mills, I. C. and Walton, H. a (2014) 'Epidemiological time series studies of PM2.5 and daily mortality and hospital admissions: a systematic review and meta-analysis.', *Thorax*, 69(7), pp. 660–5. doi: 10.1136/thoraxjnl-2013-204492.

Barker, T., Anger, A., Dessens, O., Pollitt, H., Rogers, H., Scrieciu, S., Jones, R. and Pyle, J. (2010) 'Integrated modelling of climate control and air pollution: Methodology and results from one-way coupling of an energy–environment–economy (E3MG) and atmospheric chemistry model (p-TOMCAT) in decarbonising scenarios for Mexico to 2050', *Environmental Science & Policy*, 13(8), pp. 661–670. doi: 10.1016/j.envsci.2010.09.008.

Beevers, S. D., Kitwiroon, N., Williams, M. L. and Carslaw, D. C. (2012) 'One way coupling of CMAQ and a road source dispersion model for fine scale air pollution predictions.', *Atmospheric environment (Oxford, England : 1994)*. Elsevier Ltd, 59(C), pp. 47–58. doi: 10.1016/j.atmosenv.2012.05.034.

Beevers, S. D., Kitwiroon, N., Williams, M. L., Kelly, F. J., Ross Anderson, H. and Carslaw,
D. C. (2013) 'Air pollution dispersion models for human exposure predictions in London.', *Journal of exposure science & environmental epidemiology*. Nature Publishing Group, 23(6),
pp. 647–53. doi: 10.1038/jes.2013.6.

Bell, M. L., Davis, D. L., Cifuentes, L. a, Krupnick, A. J., Morgenstern, R. D. and Thurston,
G. D. (2008) 'Ancillary human health benefits of improved air quality resulting from climate change mitigation.', *Environmental health : a global access science source*, 7, p. 41. doi: 10.1186/1476-069X-7-41.

Bhattacharyya, S. C. and Timilsina, G. R. (2010) 'A review of energy system models', *International Journal of Energy Sector Management*, 4(4), pp. 494–518. doi: 10.1108/17506221011092742.

Bollen, J., van der Zwaan, B., Brink, C. and Eerens, H. (2009) 'Local air pollution and global climate change: A combined cost-benefit analysis', *Resource and Energy Economics*, 31(3), pp. 161–181. doi: 10.1016/j.reseneeco.2009.03.001.

Brand, C. (2016) 'Beyond "Dieselgate": Implications of unaccounted and future air pollutant emissions and energy use for cars in the United Kingdom', *Energy Policy*. Elsevier, 97(X), pp. 1–12. doi: 10.1016/j.enpol.2016.06.036.

Brimblecombe, P. (1987) *The Big Smoke: A History of Air Pollution in London Since Medieval Times*. New York: Methuen & Co. doi: 10.2307/4050228.

Bruckner, T., Bashmakov, I. A., Mulugetta, Y., Chum, H., De la Vega Navarro, A.,
Edmonds, J., Faaij, A., Fungtammasan, B., Garg, A., Hertwich, E., Honnery, D., Infield, D.,
Kainuma, M., Khennas, S., Kim, S., Nimir, H. B., Riahi, K., Strachan, N., Wiser, R., Zhang,
X., Asayama, Y., Baiocchi, G., Cherubini, F., Czajkowska, A., Darghouth, N., Dooley, J. J.,
Gibon, T., Gujba, H., Hoen, B., Jager, D. de, Jewell, J., Kadner, S., Kim, S. H., Larsen, P.,
Michaelowa, A., Mills, A., Morita, K., Neuhoff, K., Hernandez, A. M., Rogner, H.-H.,

Salvatore, J., Schlömer, S., Seyboth, K., Stechow, C. von and Upadhyay, J. (2014) *Climate Change 2014: Mitigation of climate change, IPCC Fifth Assessment Report.* doi: 10.1017/CBO9781107415416.

Carrington, D. (2015) 'More people die from air pollution than Malaria and HIV/Aids, new study shows', *The Guardian*, 16 September. Available at: http://www.theguardian.com/environment/2015/sep/16/more-people-die-from-air-pollution-than-malaria-and-hivaids-new-study-shows.

Carrington, D. (2016) 'High court rules UK government plans to tackle air pollution are illegal', *The Guardian*. Available at: https://www.theguardian.com/environment/2016/nov/02/high-court-rules-uk-government-plans-to-tackle-air-pollution-are-illegal.

Cohen, A. J., Ross Anderson, H., Ostro, B., Pandey, K. D., Krzyzanowski, M., Künzli, N., Gutschmidt, K., Pope, A., Romieu, I., Samet, J. M. and Smith, K. (2006) 'The global burden of disease due to outdoor air pollution.', *Journal of toxicology and environmental health. Part A*, 68(13–14), pp. 1301–7. doi: 10.1080/15287390590936166.

Cohen, S. M. (2012) *A Techno-economic Plant- and Grid-Level Assessment of Flexible CO2 Capture, PhD Thesis, The University of Texas at Austin.* The University of Texas at Austin. Available at: http://hdl.handle.net/2152/ETD-UT-2012-08-6150.

Committee on Climate Change (2008) *Building a low-carbon economy – the UK 's contribution to tackling climate change*. doi: 10.3763/ehaz.2009.0020.

Committee on Climate Change (2010) *The fourth carbon budget: reducing the emissions through 2020.* Available at: https://www.theccc.org.uk/archive/aws2/4th Budget/CCC_4th-Budget_interactive.pdf.

Committee on Climate Change (2015a) *Sectoral scenarios for the Fifth Carbon Budget*, *Committee on Climate Change*. Available at: https://d2kjx2p8nxa8ft.cloudfront.net/wp-content/uploads/2015/11/Sectoral-scenarios-for-the-fifth-carbon-budget-Committee-on-Climate-Change.pdf.

Committee on Climate Change (2015b) *The Fifth Carbon Budget: The next step towards a low-carbon economy*. Available at: https://www.theccc.org.uk/wp-content/uploads/2015/11/Committee-on-Climate-Change-Fifth-Carbon-Budget-Report.pdf.

Committee on Climate Change (2016) *Meeting Carbon Budgets - 2016 Progress Report to Parliament*. Available at: https://documents.theccc.org.uk/wp-content/uploads/2016/06/2016-CCC-Progress-Report.pdf.

Committee on the Medical Effects of Air Pollution (COMEAP) (2006) *Cardiovascular Disease and Air Pollution*. Available at:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/304668/COM EAP_cardiovascular_disease_and_air_pollution.pdf.

Committee on the Medical Effects of Air Pollution (COMEAP) (2010) *The Mortality Effects* of Long-Term Exposure to Particulate Air Pollution in the United Kingdom.

Committee on the Medical Effects of Air Pollution (COMEAP) (2011) *Review of the UK Air Quality Index*. Available at:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/304633/COM EAP_review_of_the_uk_air_quality_index.pdf.

Costello, A., Abbas, M., Allen, A., Ball, S., Bell, S., Bellamy, R., Friel, S., Groce, N., Johnson, A., Kett, M., Lee, M., Levy, C., Maslin, M., Mccoy, D., Mcguire, B., Montgomery, H., Napier, D., Pagel, C., Patel, J., Antonio Puppim de Oliveira, J., Redclift, N., Rees, H., Rogger, D., Scott, J., Stephenson, J., Twigg, J., Wolff, J., Patterson, C. and Redclif (2009) *Managing the health effects of climate change*.

Dajnak, D. (2013) 'Emissions and Modelling: Remapping London's Air Pollution'. Available at:

http://www.londonair.org.uk/london/asp/LAQNSeminar/pdf/June2013/David_Dajnak_Emissi ons_and_Modelling_Remapping_London's_Air_pollution.pdf.

Davies, R. (2016) 'China-UK investment : key questions following Hinkley Point C delay', *The Guardian*, 8 August, pp. 3–5. Available at:

https://www.theguardian.com/business/2016/aug/09/china-uk-investment-key-questions-following-hinkley-point-c-delay.

Department for Business Energy & Industrial Strategy (BEIS) (2016) *National Atmospheric Emissions Inventory: About the Inventory*. Available at: http://naei.beis.gov.uk/about/methodology (Accessed: 1 November 2016).

Department for Business Energy and Industrial Strategy (BEIS) (2016) *The Carbon Budget Order 2016*. United Kingdom. Available at: http://www.legislation.gov.uk/uksi/2016/785/made.

Department for Environment Food and Rural Affairs (DEFRA) (2010) *Air Pollution: Action in a Changing Climate, Environment.* Available at: http://www.defra.gov.uk/publications/files/pb13378-air-pollution.pdf.

Department for Environment Food and Rural Affairs (DEFRA) (2011) *Air Quality Appraisal* – *Damage Cost Methodology*. Available at:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/182391/airquality-damage-cost-methodology-110211.pdf.

Department for Environment Food and Rural Affairs (DEFRA) (2013a) 'Abatement cost guidance for valuing changes in air quality', (May). Available at: www.gov.uk/defra.

Department for Environment Food and Rural Affairs (DEFRA) (2013b) *Air Quality Pollutant Inventories for England, Scotland, Wales and Northern Ireland: 1990 – 2011*. Available at: https://uk-air.defra.gov.uk/assets/documents/reports/cat07/1309111039_DA_AQPI_1990-2011_MainBody_Issue1.0.pdf.

Department for Environment Food and Rural Affairs (DEFRA) (2013c) *Guide to UK Air Pollution Information Resources*. Available at: http://uk-air.defra.gov.uk/.

Department for Environment Food and Rural Affairs (DEFRA) (2013d) *Impact pathway guidance for valuing changes in air quality*. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/197900/pb139 13-impact-pathway-guidance.pdf.

Department for Environment Food and Rural Affairs (DEFRA) (2016a) *Emission Map Data for Ammonia in 2014*. Available at: http://naei.defra.gov.uk/data/map-uk-das?pollutant_id=21.

Department for Environment Food and Rural Affairs (DEFRA) (2016b) *Emission Map Data* for Nitrogen oxides (NOx expressed as NO2) in 2014. Available at: http://naei.defra.gov.uk/data/map-uk-das?pollutant_id=6.

Department for Environment Food and Rural Affairs (DEFRA) (2016c) *Emission Map Data for Non Methane VOC in 2014*. Available at: http://naei.defra.gov.uk/data/map-uk-das?pollutant_id=9.

Department for Environment Food and Rural Affairs (DEFRA) (2016d) *Emission Map Data* for PM10 (Particulate Matter < 10μm) in 2014. Available at: http://naei.defra.gov.uk/data/map-uk-das?pollutant_id=24.

Department for Environment Food and Rural Affairs (DEFRA) (2016e) *Emission Map Data for Sulphur Dioxide in 2014*. Available at: http://naei.defra.gov.uk/data/map-uk-das?pollutant_id=8.

Department for Environment Food and Rural Affairs (DEFRA) (2016f) 'National Atmospheric Emissions Inventory, 1970-2014'. Available at: http://naei.defra.gov.uk/overview/ap-overview.

Department for Transport (Dft) (2014) 'Table TRA0206: Road traffic (vehicle kilometers) by vehicle type and region and country in Great Britain, annual 2015'. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/523559/tra020 6.xls.

Department of Energy and Climate Change (DECC) (2007) *Meeting the Energy Challengs: A White Paper on Energy May 2007.* Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/243268/7124.p

df.

Department of Energy and Climate Change (DECC) (2009) *Climate Change Act 2008 Impact Assessment*. London. Available at: file:///Users/melissalott/Downloads/CESPIA.pdf.

Department of Energy and Climate Change (DECC) (2014a) *2012 UK Greenhouse Gas Emissions, Final Figures - Revised March 2014*. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/277021/20140 204_2012_UK_Greenhouse_Gas_Emissions_Final_Figures.pdf (Accessed: 21 February 2014).

Department of Energy and Climate Change (DECC) (2014b) 'UK Emissions Statistics: Frequently Asked Questions', (March). Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/371493/uk_em issions statistics faq.pdf.

Department of Energy and Climate Change (DECC) (2015a) *Energy Consumption in the UK (2015)*. Available at:

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/449102/ECUK _Chapter_1_-_Overall_factsheet.pdf.

Department of Energy and Climate Change (DECC) (2015b) *Verification of UK Greenhouse Gas Emissions Estimates*. Available at: https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/573232/Metho dology Report Dec2015.pdf.

Department of Energy and Climate Change (DECC) (2016a) *Final UK greenhouse gas emissions national statistics: 1990-2014.* Available at:

https://www.gov.uk/government/statistics/final-uk-greenhouse-gas-emissions-national-statistics-1990-2014.

Department of Energy and Climate Change (DECC) (2016b) *Impact Assessment for the level* of the fifth carbon budget (DECC0231). Available at: http://www.legislation.gov.uk/ukia/2016/177/pdfs/ukia 20160177 en.pdf.

Department of Trade and Industry (DTI) (2003) *Energy White Paper - Our energy future - creating a low carbon economy*. Available at: http://webarchive.nationalarchives.gov.uk/+/http://www.berr.gov.uk/files/file10719.pdf.

Dessens, O., Anger, A., Barker, T. and Pyle, J. (2014) 'Effects of decarbonising international shipping and aviation on climate mitigation and air pollution', *Environmental Science & Policy*, 44, pp. 1–10. doi: 10.1016/j.envsci.2014.07.007.

Dessus, S. and O'Connor, D. (2003) 'Climate Policy without Tears : CGE-Based Ancillary Benefits Estimates for Chile', *Environmental and Resource Economics*, 25, pp. 287–317. Available at:

http://download.springer.com/static/pdf/688/art%25253A10.1023%25252FA%25253A10244 69430532.pdf?auth66=1412774935_8542f23fb3f090a80c346e692bfe9ea3&ext=.pdf.

Dodds, P. (2014) *UK MARKAL*, *University College London*. Available at: http://www.ucl.ac.uk/energy-models/models/uk-markal.

Dodds, P. E., Keppo, I. and Strachan, N. (2014) 'Characterising the Evolution of Energy System Models Using Model Archaeology', *Environmental Modeling & Assessment*, pp. 83– 102. doi: 10.1007/s10666-014-9417-3.

Ekins, P. (1996) 'The secondary benefits of CO2 abatement: How much emission reduction do they justify?', *Ecological Economics*, 16, pp. 13–24. doi: 10.1016/0921-8009(95)00054-2.

Ekins, P., Keppo, I., Skea, J., Strachan, N., Usher, W. and Anandarajah, G. (2013) *The UK energy system in 2050: Comparing Low-Carbon, Resilient Scenarios*. Available at: http://www.ukerc.ac.uk/asset/F7AF6FFE-5B1F-4BF0-8C076184988238DF/.

Ekins, P. and Simon, S. (2001) 'Estimating sustainability gaps: methods and preliminary applications for the UK and the Netherlands', *Ecological Economics*, 37(1), pp. 5–22. doi: 10.1016/S0921-8009(00)00279-2.

Energy Information Administration (2008) *Energy Market and Economic Impacts of S*. 1766 , *the Low Carbon Economy Act of 2007*. Available at: http://www.eia.gov/oiaf/servicerpt/lcea/pdf/sroiaf(2007)06.pdf.

Energy Technology System Analysis Programme (ETSAP) (2014a) *Energy Technology System Analysis Programme: MARKAL*. Available at: https://iea-etsap.org/index.php/etsaptools/model-generators/markal (Accessed: 1 February 2017). Energy Technology System Analysis Programme (ETSAP) (2014b) *Energy Technology Systems Analysis Program: TIMES*. Available at: http://iea-etsap.org/index.php/etsap-tools/model-generators/times (Accessed: 1 February 2017).

Energy Technology Systems Analysis Program (ETSAP) (2016) *VEDA*. Available at: http://www.iea-etsap.org/web/Veda.asp.

European Environment Agency (2011) *Air pollution impacts from carbon capture and storage (CCS), EEA Technical Report (No 14/2011)*. Available at: https://www.eea.europa.eu/publications/carbon-capture-and-storage.

Favarato, G., Anderson, H. R., Atkinson, R., Fuller, G., Mills, I. and Walton, H. (2014) 'Traffic-related pollution and asthma prevalence in children. Quantification of associations with nitrogen dioxide', *Air Quality, Atmosphere & Health.* doi: 10.1007/s11869-014-0265-8.

Garbaccio, R. F., Ho, M. S. H. and Jorgenson, D. W. (2000) *The Health Benefits of Controlling Carbon Emissions in China*. Available at: http://www.oecd.org/environment/cc/2053233.pdf (Accessed: 6 October 2014).

Gaye, A. (2008) United Nations Development Programme: Human Development Report 2007/2008 - Access to Energy and Human Development. Available at: http://hdr.undp.org/en/content/access-energy-and-human-development.

Gowers, A. M., Miller, B. G. and Stedman, J. R. (2014) *Estimating Local Mortality Burdens associated with Particulate Air Pollution*, *Public Health England*. Available at: http://www.hpa.org.uk/webc/HPAwebFile/HPAweb_C/1317141074607.

Haines, A., McMichael, A. J., Smith, K. R., Roberts, I., Woodcock, J., Markandya, A.,
Armstrong, B. G., Campbell-Lendrum, D., Dangour, A. D., Davies, M., Bruce, N., Tonne, C.,
Barrett, M. and Wilkinson, P. (2009) 'Public health benefits of strategies to reduce
greenhouse-gas emissions: overview and implications for policy makers.', *The Lancet*.
Elsevier Ltd, 374(9707), pp. 2104–14. doi: 10.1016/S0140-6736(09)61759-1.

Haines, A., Smith, K. R., Anderson, D., Epstein, P. R., McMichael, A. J., Woodcock, J. and Woods, J. (2014) 'Policies for accelerating access to clean energy , improving health , advancing development , and mitigating climate change', 370(9594), pp. 1–22. doi: 10.1016/S0140.

Hall, L. M. H. and Buckley, A. R. (2016) 'A review of energy systems models in the UK: Prevalent usage and categorisation', *Applied Energy*. Elsevier Ltd, 169, pp. 607–628. doi: 10.1016/j.apenergy.2016.02.044.

Health Effects Institute (2010) *Traffic-Related Air Pollution: A Critical Review of the Literature on Emissions, Exposure and Health Effects*. Boston, MA. Available at: https://www.healtheffects.org/system/files/SR17Traffic Review.pdf.

Held, A. M. (2010) Modelling the future development of renewable energy technologies in the European electricity sector using agent-based simulation. Available at: https://publikationen.bibliothek.kit.edu/1000021841/1809689.

Her Majesty's Treasury (2013) *Valuing impacts on air quality: Supplementary Green Book guidance*. Available at: https://www.gov.uk/government/publications/green-book-supplementary-guidance-air-quality.

Ho, M. S., Jorgenson, D. W. and Di, W. (2002) *The Health Benefits and Costs of Controlling Air Pollution in China*. Available at: http://live.belfercenter.org/files/local2.pdf.

Hoek, G., Brunekreef, B., Goldbohm, S., Fischer, P. and van den Brandt, P. a (2002) 'Association between mortality and indicators of traffic-related air pollution in the Netherlands: a cohort study.', *Lancet*, 360(9341), pp. 1203–9. doi: 10.1016/S0140-6736(02)11280-3.

Hoek, G., Krishnan, R. M., Beelen, R., Peters, A., Ostro, B., Brunekreef, B. and Kaufman, J.
D. (2013) 'Long-term air pollution exposure and cardio- respiratory mortality: a review.', *Environmental health : a global access science source*, 12(1), p. 43. doi: 10.1186/1476-069X-12-43.

Hoor, P. and Borken-Kleefeld, J. (2009) 'The impact of traffic emissions on atmospheric ozone and OH : results from QUANTIFY', *Atmospheric Chemistry and Physics*, pp. 3113–3136. doi: 10.5194/acp-9-3113-2009.

Hotten, R. (2015) 'Volkswagen: The scandal explained', *BBC News*, pp. 1–14. Available at: http://www.bbc.com/news/business-34324772.

Hourcade, J., Jaccard, M. and Bataille, C. (2006) 'Hybrid Modeling : New Answers to Old Challenges', *The Energy Journal*, (Hybrid Modeling of Energy Environment Policies), pp. 1–12. Available at: https://halshs.archives-ouvertes.fr/halshs-00471234%0A.

Intergovernmental Panel on Climate Change (IPCC) (2007) *IPCC Fourth Assessment Report* (*AR4*): *Climate Change 2007 - Synthesis Report*. Available at: https://www.ipcc.ch/publications_and_data/publications_ipcc_fourth_assessment_report_syn thesis_report.htm.

Intergovernmental Panel on Climate Change (IPCC) (2014) 'Summary for Policy Makers', *Climate Change 2014: Impacts, Adaptation and Vulnerability - Contributions of the Working Group II to the Fifth Assessment Report*, pp. 1–32. doi: 10.1016/j.renene.2009.11.012.

International Energy Agency (IEA) (2004) *World Energy Outlook (WEO) 2004 - Chapter 10 Highlights*. Available at: http://www.worldenergyoutlook.org/media/weowebsite/2008-1994/WEO2004.pdf.

International Energy Agency (IEA) (2012) 'Energy Policies of IEA Countries - The United Kingdom 2012 Review', p. 182. doi: 10.1787/9789264171497-en.

International Energy Agency (IEA) (2015a) 'Energy and Climate Change', *World Energy Outlook Special Report*, pp. 1–200. doi: 10.1038/479267b.

International Energy Agency (IEA) (2015b) *World Energy Model Documentation*. Available at: www.worldenergyoutlook.org.

International Energy Agency (IEA) (2016) 'Energy and Air Pollution', *World Energy Outlook - Special Report*, p. 266. doi: 10.1021/ac00256a010.

International Institute for Applied Systems Analysis (IIASA) (2009) *GAINS Online : Tutorial for advanced users*. Available at: http://www.iiasa.ac.at/web/home/research/researchPrograms/air/GAINS-tutorial.pdf.

International Institute for Applied Systems Analysis (IIASA) (2011) *The GAINS model*. Available at:

http://www.iiasa.ac.at/web/home/research/researchPrograms/air/GAINS.en.html.

International Institute for Applied Systems Analysis (IIASA) (2013) *MESSAGE*. Available at: http://www.iiasa.ac.at/web/home/research/modelsData/MESSAGE/MESSAGE.en.html.

International Institute for Applied Systems Analysis (IIASA) (2016) *The GAINS Model: A scientific tool to combat air pollution and climate change simultaneously*. Available at: http://gains.iiasa.ac.at/gains/EUN/index.login?logout=1 (Accessed: 8 April 2016).

Jack, D. W. and Kinney, P. L. (2010) 'Health co-benefits of climate mitigation in urban areas', *Current Opinion in Environmental Sustainability*. Elsevier B.V., 2(3), pp. 172–177. doi: 10.1016/j.cosust.2010.06.007.

Jarrett, J., Woodcock, J., Griffiths, U. K., Chalabi, Z., Edwards, P., Roberts, I. and Haines, A. (2012) 'Effect of increasing active travel in urban England and Wales on costs to the National Health Service.', *Lancet*, 379(9832), pp. 2198–205. doi: 10.1016/S0140-6736(12)60766-1.

Jebaraj, S. and Iniyan, S. (2006) 'A review of energy models', *Renewable and Sustainable Energy Reviews*, 10(4), pp. 281–311. doi: 10.1016/j.rser.2004.09.004.

Jensen, H. T., Keogh-Brown, M. R., Smith, R. D., Chalabi, Z., Dangour, A. D., Davies, M., Edwards, P., Garnett, T., Givoni, M., Griffiths, U., Hamilton, I., Jarrett, J., Roberts, I., Wilkinson, P., Woodcock, J. and Haines, A. (2013a) 'The importance of health co-benefits in macroeconomic assessments of UK Greenhouse Gas emission reduction strategies', *Climatic Change*, 121(2), pp. 223–237. doi: 10.1007/s10584-013-0881-6.

Jensen, H. T., Keogh-Brown, M. R., Smith, R. D., Chalabi, Z., Dangour, A. D., Davies, M., Edwards, P., Garnett, T., Givoni, M., Griffiths, U., Hamilton, I., Jarrett, J., Roberts, I., Wilkinson, P., Woodcock, J. and Haines, A. (2013b) 'The importance of health co-benefits in macroeconomic assessments of UK Greenhouse Gas emission reduction strategies', *Climatic Change*, 121(2), pp. 223–237. doi: 10.1007/s10584-013-0881-6.

Jerrett, M., Burnett, R. T., Pope, A., Ito, K., Thurston, G., Kewski, D., Shi, Y., Calle, E. and Thun, M. (2009) 'Long-Term Ozone Exposure and Mortality', *The New England Journal of Medicine*, 360, pp. 1085–1095.

Kan, H., Chen, B., Zhao, N., London, S. J., Song, G., Chen, G., Zhang, Y. and Jiang, L.
(2010) Public Health and Air Pollution in Asia (PAPA): Coordinated Studies of Short-term Exposure to Air Pollution and Daily Mortality in Four Cities Part 1: A Time-Series Study of Ambient Air Pollution and Daily Mortality in Shanghai, China.

Kiesewetter, G., Borken-Kleefeld, J., Schöpp, W., Heyes, C., Thunis, P., Bessagnet, B., Terrenoire, E., Gsella, a. and Amann, M. (2014) 'Modelling NO2 concentrations at the street level in the GAINS integrated assessment model: Projections under current legislation', *Atmospheric Chemistry and Physics*, 14(2), pp. 813–829. doi: 10.5194/acp-14-813-2014.

Lelieveld, J., Evans, J. S., Fnais, M., Giannadaki, D. and Pozzer, a. (2015) 'The contribution of outdoor air pollution sources to premature mortality on a global scale', *Nature*, 525(7569), pp. 367–371. doi: 10.1038/nature15371.

Lott, M. C. (2016) 'Scotland Is Now Coal-Free after More Than a Century', *Scientific American - Plugged In*. Available at: http://blogs.scientificamerican.com/plugged-in/scotland-is-now-coal-free-after-more-than-a-century/.

Lott, M. C., Ekins, P. and Davies, M. (2014) 'Using the Air Emissions Component of the "Years to Sustainability" Indicator to Determine Sustainability Trends in the United Kingdom - Health Impacts', in *International Society of Ecological Economics 2014*, pp. 1–10.

Lott, M. C., Pye, S. and Dodds, P. E. (2017) 'Quantifying the co-impacts of energy sector decarbonisation on outdoor air pollution in the United Kingdom', *Energy Policy*. Elsevier, 101(October 2016), pp. 42–51. doi: 10.1016/j.enpol.2016.11.028.

Lott, M. C., Pye, S., Fais, B. and Dodds, P. (2016) 'Up In The Air: A framework for quantifying the co-impacts of energy sector decarbonisation on outdoor air pollution', in *International Energy Workshop 2016 - Book of Abstracts*.

Lott, M. and Daly, H. (2015) 'A62 The Impacts of Transport Sector Decarbonisation Pathways on Air Quality and Public Health in the United Kingdom', *Journal of Transport & Health*, 2(2), p. S37. doi: 10.1016/j.jth.2015.04.550. Loulou, R., Goldstein, G., Kanudia, A., Lettila, A. and Remme, U. (2016) 'Documentation for the TIMES Model Part I', (July), pp. 1–151. Available at: http://iea-etsap.org/index.php/documentation.

Loulou, R., Kanudia, A., Lehtila, A., Remme, U. and Goldstein, G. (2016) 'Documentation for the TIMES Model Part III', *IEA Energy Technology Systems Analysis Programme*, (April), pp. 1–78. Available at: http://iea-etsap.org/index.php/documentation.

Loulou, R., Lehtilä, A., Kanudia, A., Remme, U. and Goldstein, G. (2016) 'Documentation for the TIMES Model Part II', *IEA Energy Technology Systems Analysis Programme*, (April), pp. 1–78. Available at: http://iea-etsap.org/index.php/documentation.

Loulou, R., Remme, U., Kanudia, A., Lehtila, A. and Goldstein, G. (2005) 'Documentation for the TIMES Model Authors ':, (April), pp. 1–78.

Loulou, R., Remme, U., Kanudia, A., Lehtila, A. and Goldstein, G. (2016) 'Documentation for the TIMES Model Part IV', *IEA Energy Technology Systems Analysis Programme*, (April), pp. 1–78. Available at: http://iea-etsap.org/index.php/documentation.

Markandya, A. (1990) 'Environmental costs and power system planning', *Utilities Policy*, October, pp. 13–27. doi: 10.1016/0957-1787(90)90005-6.

Martin, P. (2016) 'London Atmospheric Emissions Inventory (LAEI) 14th April 2016 Workshop', pp. 1–24. Available at: https://data.london.gov.uk/dataset/london-atmosphericemissions-inventory-2013.

Mayor of London. London Assembly (2016a) *Mayor of London announces bold plans to clean up London's toxic air*. Available at: https://www.london.gov.uk/press-releases/mayoral/bold-plans-to-clean-up-londons-toxic-air.

Mayor of London. London Assembly (2016b) *Sadiq Khan unveils action plan to battle London's toxic air*. Available at: https://www.london.gov.uk/press-releases/mayoral/mayor-unveils-action-plan-to-battle-toxic-air (Accessed: 15 July 2016).

Met Office (2015) *The Great Smog of 1952 in London*. Available at: http://history1900s.about.com/od/1950s/qt/greatsmog.htm.

Miller, B. (2010) *Report on estimation of mortality impacts of particulate air pollution in London, Institute of Occupational Medicine (IOM)*. Available at: http://www.aef.org.uk/uploads/IomReport_1.pdf.

Miller, B. G. and Hurley, J. F. (2003) 'Life table methods for quantitative impact assessments in chronic mortality', *Journal of Epidemial Community Health*, 57, pp. 200–206.

Miller, B. G. and Hurley, J. F. (2010) Supporting paper to COMEAP 2010 report: 'The Mortality Effects of Long Term Exposure to Particulate Air Pollution in the United Kingdom'
Technical Aspects of Life Table Analyses. Available at:

http://www.ademloos.be/sites/default/files/gezondheid_docs/supporting paper - miller and hurley.pdf.

Mills, I. C., Atkinson, R. W., Kang, S., Walton, H. and Anderson, H. R. (2015) 'Quantitative systematic review of the associations between short-term exposure to nitrogen dioxide and mortality and hospital admissions.', *BMJ open*, 5(5), p. e006946. doi: 10.1136/bmjopen-2014-006946.

Milner, J., Davies, M. and Wilkinson, P. (2012) 'Urban energy, carbon management (low carbon cities) and co-benefits for human health', *Current Opinion in Environmental Sustainability*. Elsevier B.V., 4(4), pp. 398–404. doi: 10.1016/j.cosust.2012.09.011.

Ministerie van Verkeer en Waterstaat (2009) *Cycling in the Netherlands*. Available at: http://www.fietsberaad.nl/library/repository/bestanden/CyclingintheNetherlands2009.pdf.

Murrels, T. and Pang, Y. (2013) *National Atmospheric Emissions Inventory: Emissions Factors for Alternative Vehicle Technologies (ED57423001)*. Available at: http://naei.beis.gov.uk/resources/NAEI_Emisison_factors_for_alternative_vehicle_technologi es_Final_Feb_13.pdf.

Nemet, G. F., Holloway, T. and Meier, P. (2010) 'Implications of incorporating air-quality co-benefits into climate change policymaking', *Environmental Research Letters*, 5(1), p. 14007. doi: 10.1088/1748-9326/5/1/014007.

Nordhaus, W. and Sztorc, P. (2013) *DICE 2013R : Introduction and User's Manual (Second Edition)*. Available at: dicemodel.net.

Office of National Statistics (2010) *National Population Projections, 2010-Based Projections*. Available at:

http://webarchive.nationalarchives.gov.uk/20160105160709/http://ons.gov.uk/ons/rel/npp/nat ional-population-projections/2010-based-projections/index.html.

Pandey, R. (2002) 'Energy policy modelling: agenda for developing countries', *Energy Policy*, 30, pp. 97–106. doi: 10.1016/S0301-4215(01)00062-3.

Pascal, M., Corso, M., Chanel, O., Declercq, C., Badaloni, C., Cesaroni, G., Henschel, S., Meister, K., Haluza, D., Martin-Olmedo, P. and Medina, S. (2013) 'Assessing the public health impacts of urban air pollution in 25 European cities: results of the Aphekom project.', *The Science of the total environment*. Elsevier B.V., 449(2007105), pp. 390–400. doi: 10.1016/j.scitotenv.2013.01.077.

Pfluger, B. (2014) Assessment of least-cost pathways for decarbonising Europe's power supply : a model-based long-term scenario analysis accounting for the characteristics of renewable energies. KIT Scientific Publishing. Available at: https://www.ksp.kit.edu/download/1000037155.

Pope, C. A., Burnett, R. T., Thun, M. J., Calle, E. E., Krewski, D. and Thurston, G. D. (2002) 'Lung Cancer, Cardiopulmonary Mortality, and Long-term Exposure to Fine Particulate Air Pollution', *Journal of the American Medical Association*, 287(9). doi: 10.1001/jama.287.9.1132.

Pye, S., Anandarajah, G., Fais, B., McGlade, C. and Strachan, N. (2015) *Pathways to Deep Decarbonisation in the United Kingdom*. Available at: http://deepdecarbonization.org/wp-content/uploads/2015/09/DDPP_GBR.pdf.

Pye, S. and Bataille, C. (2016) 'Improving deep decarbonization modelling capacity for developed and developing country contexts', *Climate Policy*. Taylor & Francis, 16(sup1), pp. S27–S46. doi: 10.1080/14693062.2016.1173004.

Pye, S. and Daly, H. (2015) 'Urban transport modal shift : an energy systems approach', 44(April 2014), pp. 1–30.

Pye, S., Li, F. G. N., Price, J. and Fais, B. (2017) 'Achieving net-zero emissions through thereframing of UK national targets in thepost-Paris Agreement era', *Nature Energy*, 2, p. Article 17024. doi: 10.1038/nenergy.2017.24 |.

Pye, S., Ozkan, N. N., Wagner, A. and Hobson, M. (2008) *Air quality emission tracking in the UK MARKAL model*. Available at:

http://webarchive.nationalarchives.gov.uk/20090410160339/http://www.defra.gov.uk/eviden ce/igcb/pdf/markal1.pdf.

Pye, S. and Palmer, T. (2008) *Optimising delivery of carbon reduction targets : integrating air quality benefits using the UK MARKAL model*. Available at: http://webarchive.nationalarchives.gov.uk/20090410160335/http://www.defra.gov.uk/eviden ce/igcb/pdf/markal2.pdf.

Pye, S., Sabio, N. and Strachan, N. (2015) 'An integrated systematic analysis of uncertainties in UK energy transition pathways', *Energy Policy*. Elsevier, pp. 1–12. doi: 10.1016/j.enpol.2014.12.031.

Royal College of Physicians (2016) *Every breath we take: The lifelong impact of air pollution. Report of a working party.* Available at: https://www.rcplondon.ac.uk/projects/outputs/every-breath-we-take-lifelong-impact-air-

pollution.

Seebregts, A. J., Goldstein, G. A. and Smekens, K. (2002) 'Energy/Environmental Modeling with the MARKAL Family of Models', in Chamoni, P., Leisten, R., Martin, A., Minnemann, J., and Stadtler, H. (eds) *Operations Research Proceedings 2001: Selected Papers of the International Conference on Operations Research (OR 2001) Duisburg, September 3--5, 2001.* Berlin, Heidelberg: Springer Berlin Heidelberg, pp. 75–82. doi: 10.1007/978-3-642-50282-8_10.

Soble, J. (2016) 'Mitsubishi Admits Cheating on Fuel-Economy Tests', *The New York TImes*, pp. 1–4. Available at: https://www.nytimes.com/2016/04/21/business/mitsubishi-fuel-economy-tests.html.

Sokhi, R. S. and Kitwiroon, N. (2011) 'Air Pollution in Urban Areas', in *World Atlas of Atmospheric Pollution*. Anthem Press, pp. 19–34. Available at: http://www.anthempress.com/world-atlas-of-atmospheric-pollution-hb.

Stern, N. and Taylor, C. (2006) 'The Stern Review on the Economic Effects of Climate Change', *Population and Development Review*, 32(4), pp. 793–798. doi: 10.1111/j.1728-4457.2006.00153.x.

Strachan, N., Pye, S. and Kannan, R. (2009) 'The iterative contribution and relevance of modelling to UK energy policy', *Energy Policy*, 37(3), pp. 850–860. doi: 10.1016/j.enpol.2008.09.096.

Taylor, P. G., Upham, P., McDowall, W. and Christopherson, D. (2014) 'Energy model, boundary object and societal lens: 35 years of the MARKAL model in the UK', *Energy Research and Social Science*. Elsevier Ltd, 4(C), pp. 32–41. doi: 10.1016/j.erss.2014.08.007.

Tétreault, L.-F., Perron, S. and Smargiassi, A. (2013) 'Cardiovascular health, traffic-related air pollution and noise: are associations mutually confounded? A systematic review.', *International Journal of Public healthealth*, 58(5), pp. 649–66. doi: 10.1007/s00038-013-0489-7.

Thambiran, T. and Diab, R. D. (2011) 'The case for integrated air quality and climate change policies', *Environmental Science & Policy*. Elsevier Ltd, 14(8), pp. 1008–1017. doi: 10.1016/j.envsci.2011.08.002.

Transport for London (TfL) (2010) 'Analysis of Cycling Potential', pp. 1–55. Available at: http://www.tfl.gov.uk/cdn/static/cms/documents/analysis-of-cycling-potential.pdf.

Transport for London (TfL) (2011) *London Travel Demand Survey*. Available at: http://content.tfl.gov.uk/london-travel-demand-survey.pdf.

Transport for London (TfL) (2014) 'Drivers of Demand for Travel in London: A review of trends in travel demand and their causes A review of trends in travel demand and their causes', pp. 1–55. Available at: http://content.tfl.gov.uk/drivers-of-demand-for-travel-in-london.pdf.

Transport for London (TfL) (2015a) *An Ultra Low Emission Vehicle Delivery Plan for London, Transport for London.* Available at: http://content.tfl.gov.uk/ulev-delivery-plan.pdf.

Transport for London (TfL) (2015b) *Ultra Low Emission Zone*. Available at: https://tfl.gov.uk/modes/driving/ultra-low-emission-zone?cid=ultra-low-emission-zone#on-this-page-0.

U.S. Environmental Protection Agency (2009) *Assessment of the Impacts of Global Change on Regional U.S. Air Quality : A synthesis of climate change impacts on ground-level ozone.* Washington, DC. Available at: http://cfpub.epa.gov/ncea/cfm/recordisplay.cfm?deid=203459.

U.S. Environmental Protection Agency (2013) 'Integrated Science Assessment for Oxides of Nitrogen – Health Criteria (First External Review Draft)', (November). Available at: https://cfpub.epa.gov/ncea/isa/recordisplay.cfm?deid=259167.

U.S. Environmental Protection Agency (2014) *Carbon Pollution Emission Guidelines for Existing Stationary Sources: Electric Utility Generating Units*. Available at: https://www.federalregister.gov/documents/2015/10/23/2015-22842/carbon-pollutionemission-guidelines-for-existing-stationary-sources-electric-utility-generating.

U.S. Environmental Protection Agency (2016) *Climate Economic Modeling*. Available at: http://www.epa.gov/climate/climatechange/economics/modeling.html.

U.S. Environmental Protection Agency National Center for Environmental Economics (2014a) *2.3.2.1. Damage Avoided (morbidity)*. Available at: http://yosemite.epa.gov/EE/epa/eed.nsf/dcee735e22c76aef85257662005f4116/d7d14ccfc190 9ea78525779700781a1f!OpenDocument.

U.S. Environmental Protection Agency National Center for Environmental Economics (2014b) *2.3.2.2. Damage Avoided (mortality)*. Available at: http://yosemite.epa.gov/ee/epa/eed.nsf/6058a089548635578525766200639df3/d124ce9528d3 95a485257797007819da!OpenDocument.

United Kingdom Parliament (1956) 'Clean Air Act, 1956'. Available at: http://www.legislation.gov.uk/ukpga/Eliz2/4-5/52/enacted.

United Kingdom Parliament (2008) *Climate Change Act 2008*. Available at: http://www.legislation.gov.uk/ukpga/2008/27/pdfs/ukpga_20080027_en.pdf.

United Kingdom Supreme Court (2015) 'JUDGMENT R (on the application of ClientEarth) (Appellant) v Secretary of State for the Environment, Food and Rural Affairs (Respondent)', (April). Available at: https://www.supremecourt.uk/cases/docs/uksc-2012-0179-judgment.pdf.

United Nations Framework Convention on Climate Change (UNFCCC) (2016) *Kyoto Protocol.* Available at: http://unfccc.int/kyoto_protocol/items/2830.php (Accessed: 1 January 2016).

Usher, W. and Strachan, N. (2012) 'Critical mid-term uncertainties in long-term decarbonisation pathways', *Energy Policy*. Elsevier, 41, pp. 433–444. doi: 10.1016/j.enpol.2011.11.004.

Vaughan, A. (2016) 'Sadiq Khan to more than double size of London' s clean air zone', *The Guardian*, pp. 3–5. Available at:

https://www.theguardian.com/environment/2016/may/13/sadiq-khan-to-double-size-londons-clean-air-zone-pollution.

Vidal, J. (2013) 'More than 90% of people in European cities breathe dangerous air, study finds', pp. 13–16. Available at: https://www.theguardian.com/environment/2013/oct/15/european-cities-dangerous-air-pollution.

Wadud, Z. and Waitz, I. a. (2011) 'Comparison of Air Quality-Related Mortality Impacts of Different Transportation Modes in the United States', *Transportation Research Record: Journal of the Transportation Research Board*, 2233(1), pp. 99–109. doi: 10.3141/2233-12.

Walton, B. H., Dajnak, D., Beevers, S., Williams, M., Watkiss, P. and Hunt, A. (2015)
Understanding the Health Impacts of Air Pollution in London. Available at:
https://www.london.gov.uk/sites/default/files/hiainlondon_kingsreport_14072015_final.pdf.

Watts, N., Adger, W. N., Agnolucci, P., Blackstock, J., Byass, P., Cai, W., Chaytor, S.,Colbourn, T., Collins, M., Cooper, A., Cox, P. M., Depledge, J., Drummond, P., Ekins, P.,Galaz, V., Grace, D., Graham, H., Grubb, M., Haines, A., Hamilton, I., Hunter, A., Jiang, X.,

Li, M., Kelman, I., Liang, L., Lott, M., Lowe, R., Luo, Y., Mace, G., Maslin, M., Nilsson, M., Oreszczyn, T., Pye, S., Quinn, T., Svensdotter, M., Venevsky, S., Warner, K., Xu, B., Yang, J., Yin, Y., Yu, C., Zhang, Q., Gong, P., Montgomery, H. and Costello, A. (2015) 'Health and Climate Change: Policy responses to protect public health', *The Lancet*, 386(10006), pp. 1861–1914. doi: 10.1016/S0140-6736(15)60854-6.

Williams, M. L. (2007) 'UK air quality in 2050—synergies with climate change policies', *Environmental Science & Policy*, 10(2), pp. 169–175. doi: 10.1016/j.envsci.2006.11.001.

Williams, M. L., Atkinson, R. W., Anderson, H. R. and Kelly, F. J. (2014) 'Associations between daily mortality in London and combined oxidant capacity, ozone and nitrogen dioxide', *Air Quality, Atmosphere & Health.* doi: 10.1007/s11869-014-0249-8.

Williams, M. L., Beevers, S. D., Lott, M. C. and Kitwiroon, N. (2016) 'The Impacts of Policies to Meet the UK Climate Change Act Target on Air Quality - An Explicity Modelling Study', in *International Energy Workshop 2016 - Book of Abstracts*.

Woodcock, J., Banister, D., Edwards, P., Prentice, A. M. and Roberts, I. (2007) 'Energy and transport.', *The Lancet*, 370(9592), pp. 1078–88. doi: 10.1016/S0140-6736(07)61254-9.

Woodcock, J., Edwards, P., Tonne, C., Armstrong, B. G., Ashiru, O., Banister, D., Beevers,
S., Chalabi, Z., Chowdhury, Z., Cohen, A., Franco, O. H., Haines, A., Hickman, R., Lindsay,
G., Mittal, I., Mohan, D., Tiwari, G., Woodward, A. and Roberts, I. (2009) 'Public health
benefits of strategies to reduce greenhouse-gas emissions: urban land transport.', *The Lancet*.
Elsevier Ltd, 374(9705), pp. 1930–43. doi: 10.1016/S0140-6736(09)61714-1.

World Health Organization (2000) *Air Quality Guidelines for Europe, Second Edition*. Available at: http://www.euro.who.int/__data/assets/pdf_file/0005/74732/E71922.pdf.

World Health Organization (2005) 'WHO Air quality guidelines for particulate matter, ozone, nitrogen dioxide and sulfur dioxide'. Available at: http://apps.who.int/iris/bitstream/10665/69477/1/WHO_SDE_PHE_OEH_06.02_eng.pdf.

World Health Organization (2007) *Air pollution*. Available at: http://www.who.int/topics/air pollution/en/.

World Health Organization (2013a) 'Health risks of air pollution in Europe – HRAPIE project', p. 60. Available at:

http://www.euro.who.int/__data/assets/pdf_file/0006/238956/Health-risks-of-air-pollution-in-Europe-HRAPIE-project,-Recommendations-for-concentrationresponse-functions-forcostbenefit-analysis-of-particulate-matter,-ozone-and-nitrogen-dioxide.pdf?ua=1.

World Health Organization (2013b) *Review of evidence on health aspects of air pollution* – *REVIHAAP Project*. Available at: http://www.euro.who.int/__data/assets/pdf_file/0004/193108/REVIHAAP-Final-technical-report-final-version.pdf?ua=1.

World Health Organization (2014a) 7 million premature deaths annually linked to air pollution, World Health Organization. Available at: http://www.who.int/mediacentre/news/releases/2014/air-pollution/en/#.

World Health Organization (2014b) *WHO methods for life expectancy and healthy life expectancy*. Available at: http://www.who.int/healthinfo/statistics/LT method 1990 2012.pdf.

World Health Organization (2016a) *Air pollution levels rising in many of the world's poorest cities*. Available at: http://www.who.int/mediacentre/news/releases/2016/air-pollution-rising/en/ (Accessed: 27 July 2016).

World Health Organization (2016b) *Public health, environmental and social determinants of health (PHE) - WHO Global Urban Ambient Air Pollution Database (update 2016).* Available at: http://www.who.int/phe/health_topics/outdoorair/databases/cities/en/.

World Health Organization (2016c) *WHO Ambient (outdoor) Air Pollution Database, by Country and City.* Available at:

http://www.who.int/phe/health_topics/outdoorair/databases/cities/en/.

World Health Organization Media Center (2011) 'Air quality and health', pp. 1–5. Available at: http://www.who.int/mediacentre/factsheets/fs313/en/index.html.

Wright, E., Kanudia, A., Loulou, R. and Goldstein, G. (2016) 'Documentation for the TIMES Model Part V', (July), pp. 1–151. Available at: http://iea-etsap.org/index.php/documentation.

Yim, S. H. L. and Barrett, S. R. H. (2012) 'Public health impacts of combustion emissions in the United Kingdom.', *Environmental Science & Technology*, 46(8), pp. 4291–6. doi: 10.1021/es2040416.

Zanobetti, A., O'Neill, M. S., Gronlund, C. J. and Schwartz, J. D. (2012) 'Summer temperature variability and long-term survival among elderly people with chronic disease.', *Proceedings of the National Academy of Sciences of the United States of America*, 109(17), pp. 6608–13. doi: 10.1073/pnas.1113070109.

Zanobetti, A. and Schwartz, J. (2011) 'Ozone and survival in four cohorts with potentially predisposing diseases.', *American Journal of Respiratory and Critical Care Medicine*, 184(7), pp. 836–41. doi: 10.1164/rccm.201102-0227OC.

This page was intentionally left blank.

Appendix

A.1: Documentation for the TIMES Model - PART II (July 2016)

Energy Technology Systems Analysis Programme

http://www.iea-etsap.org/web/Documentation.asp

Documentation for the TIMES Model

PART II

July 2016

Authors:

Richard Loulou Antti Lehtilä Amit Kanudia Uwe Remme Gary Goldstein

287

1

General Introduction to the TIMES Documentation

This documentation is composed of five Parts.

<u>**Part I**</u> provides a general description of the TIMES paradigm, with emphasis on the model's general structure and its economic significance. Part I also includes a simplified mathematical formulation of TIMES, a chapter comparing it to the MARKAL model, pointing to similarities and differences, and chapters describing new model options.

Part II constitutes a comprehensive reference manual intended for the technically minded modeler or programmer looking for an in-depth understanding of the complete model details, in particular the relationship between the input data and the model mathematics, or contemplating making changes to the model's equations. Part II includes a full description of the sets, attributes, variables, and equations of the TIMES model.

Part III describes the organization of the TIMES modeling environment and the GAMS control statements required to run the TIMES model. GAMS is a modeling language that translates a TIMES database into the Linear Programming matrix, and then submits this LP to an optimizer and generates the result files. Part III describes how the routines comprising the TIMES source code guide the model through compilation, execution, solve, and reporting; the files produced by the run process and their use; and the various switches that control the execution of the TIMES code according to the model instance, formulation options, and run options selected by the user. It also includes a section on identifying and resolving errors that may occur during the run process.

Part IV provides a step-by-step introduction to building a TIMES model in the VEDA-Front End (VEDA-FE) model management software. It first offers an orientation to the basic features of VEDA-FE, including software layout, data files and tables, and model management features. It then describes in detail twelve Demo models (available for download from the ETSAP website) that progressively introduce VEDA-TIMES principles and modeling techniques.

<u>**Part V**</u> describes the VEDA Back-End (VEDA-BE) software, which is widely used for analyzing results from TIMES models. It provides a complete guide to using VEDA-BE, including how to get started, import model results, create and view tables, create and modify user sets, and step through results in the model Reference Energy System. It also describes advanced features and provides suggestions for best practices.

2
PART II: REFERENCE MANUAL

TABLE OF CONTENTS FOR PART II

1	INTRODUCTION	
1.1	Basic notation and conventions	
1.2	2 GAMS modelling language and TIMES implementation	9
2	SETS	10
2.1	Indexes (One-dimensional sets)	
2.2	2 User input sets	
1	2.2.1 Definition of the Reference Energy System (RES)	
	2.2.1.1 Processes	
	2.2.1.2 Commodities	
1	2.2.2 Definition of the time structure	
	2.2.2.1 Time horizon	
	2.2.2.2 Timeslices	
1	2.2.3 Multi-regional models	
1	2.2.4 Overview of all user input sets	
2.3	B Definition of internal sets	
3	PARAMETERS	
3.1	User input parameters	
1	3.1.1 Inter- and extrapolation of user input parameters	
	3.1.1.1 Inter/extrapolation options	
	3.1.1.2 Default inter/extrapolation	
	3.1.1.3 Interpolation of cost parameters	
	3.1.1.4 Examples of using I/E options	
	3.1.1.5 Applicability	
1	3.1.2 Inheritance and aggregation of timesliced input parameters	
	3.1.3 Overview of user input parameters	
3.2	2 Internal parameters	
3.3	B Report parameters	
1	3.3.1 Overview of report parameters	
1	3.3.2 Acronyms used in cost reporting parameters	
	3.3.3 The levelized cost reporting option	
4	USAGE NOTES ON SPECIAL TYPES OF PROCESSES	108
4.1	Combined heat and power	
4	4.1.1 Overview	
4	4.1.2 Defining CHP attributes in TIMES	
	4.1.2.1 Back-pressure turbine systems.	
	4.1.2.2 Pass-out turbine systems	
	4.1.2.3 Alternative choices for defining the activity basis	
4.2	2 Inter-regional exchange processes	112
4	4.2.1 Structure and types of endogenous trade	
4	4.2.2 Input sets and parameters specific to trade processes	
4	4.2.3 Availability factors for trade processes	
4	4.2.4 Notes on other attributes for trade processes	
4.3	B Storage processes	
4	4.3.1 Överview	
4	4.3.2 Timeslice storage	
4	4.3.3 Day/Night storage	
		4

4.3	4 Inter-period storage	121
4.3	5 Auxiliary storage flows	122
4.3	6 Input sets and parameters specifically related to storage processes	123
4.3	Availability factors for storage processes	125
4.5	a indus on outer autobutes for storage processes	145
E 14		27
υ		21
5.1	VAR_ACT(r,v,t,p,s)	129
5.2	VAR_BLND(r,ble,opr)	130
5.3	VAR_CAP(r,t,p)	130
5.4	VAR_COMNET(r,t,c,s)	130
5.5	VAR_COMPRD(r,t,c,s)	131
5.6	VAR_CUMCOM(r,c,type,y1,y2)	131
5.7	VAR_CUMCST(r, y1,y2,costagg,cur)	131
5.8	VAR_CUMFLO(r,p,c,y1,y2)	131
5.9	VAR_DNCAP(r,t,p,u) / VAR_SNCAP(r,t,p)	132
5.10	VAR_DRCAP(r,v,t,p,j)	132
5.11	VAR_ELAST(r,t,c,s,j,l)	132
5.12	VAR_FLO(r,v,t,p,c,s)	133
5.13	VAR_IRE(r,v,t,p,c,s,ie)	133
5.14	VAR_NCAP(r,v,p)	133
5.15	VAR OBJ(v ₀) and related variables	134
5.1	5.1 VAR_OBJR(r, y ₀)	135
5.1	5.2 INVCOST(r,y)	135
5.1	5.3 INVTAXSUB(r,y)	135
5.1	5.4 IN VDECOM(I, y)	135
5.1	$5.5 \text{FIXCOST}(\mathbf{r}, \mathbf{y})$	135
5.1	5.7 VARCOST(r.v)	135
5.1	5.8 VARTAXSUB(r,y)	135
5.1	5.9 ELASTCOST(r,y)	136
5.1	5.10 LATEREVENUES(r,y)	136
5.1	5.11 SALVAGE(r,y ₀)	136
5.16	VAR_RCAP(r,v,t,p)	136
5.17	VAR_SCAP(r,v,t,p)	136
5.18	VAR_SIN/SOUT(r,v,t,p,c,s)	137
5.19	VAR_UPS(r,v,t,p,s,l)	137
5.20	Variables used in User Constraints	137
		138
5.2	0.1 VAR_UC(uc_n)	150
5.2 5.2	0.1 VAR_UC(uc_n) 0.2 VAR_UCR(uc_n,r)	138
5.2 5.2 5.2	0.1 VAR_UC(uc_n). 0.2 VAR_UCR(uc_n,r). 0.3 VAR_UCT(uc_n,t).	138 138
5.2 5.2 5.2 5.2	0.1 VAR_UC(uc_n). 0.2 VAR_UCR(uc_n,r). 0.3 VAR_UCT(uc_n,t). 0.4 VAR_UCRT(uc_n,r,t). 0.5 VAR_UCRT(uc_n,r,t).	138 138 138
5.2 5.2 5.2 5.2 5.2	0.1 VAR_UC(uc_n). 0.2 VAR_UCR(uc_n,r). 0.3 VAR_UCT(uc_n,t). 0.4 VAR_UCRT(uc_n,r,t). 0.5 VAR_UCTS(uc_n,r,t). 0.5 VAR_UCTS(uc_n,t,s). 0.6 VAR_UCTS(uc_n,t,s). 0.7 VAR_UCRS(uc_n,t,s). 0.7 VAR_	138 138 138 138 138
5.2 5.2 5.2 5.2 5.2 5.2 5.2	0.1 VAR_UC(uc_n). 0.2 VAR_UCR(uc_n,r). 0.3 VAR_UCT(uc_n,t). 0.4 VAR_UCRT(uc_n,r,t). 0.5 VAR_UCTS(uc_n,t,s). 0.6 VAR_UCRTS(uc_n,r,t,s).	138 138 138 138 138 138
5.2 5.2 5.2 5.2 5.2 5.2	0.1 VAR_UC(uc_n). 0.2 VAR_UCR(uc_n,r). 0.3 VAR_UCT(uc_n,t). 0.4 VAR_UCRT(uc_n,t,t). 0.5 VAR_UCTS(uc_n,t,s). 0.6 VAR_UCRTS(uc_n,r,t,s). COLIATIONS. 4	138 138 138 138 138 138
5.2 5.2 5.2 5.2 5.2 5.2 5.2	0.1 VAR_UC(uc_n). 0.2 VAR_UCR(uc_n,r). 0.3 VAR_UCT(uc_n,t). 0.4 VAR_UCRT(uc_n,t,t). 0.5 VAR_UCTS(uc_n,t,s). 0.6 VAR_UCRTS(uc_n,r,t,s). CQUATIONS	138 138 138 138 138 138 138
5.2 5.2 5.2 5.2 5.2 5.2 5.2 5.2 6.2 6.1	0.1 VAR UC(uc_n). 0.2 VAR UCR(uc_n,r). 0.3 VAR_UCT(uc_n,t). 0.4 VAR_UCRT(uc_n,r,t). 0.5 VAR_UCRTS(uc_n,r,ts). 0.6 VAR_UCRTS(uc_n,r,t,s). 0.7 VAR_UCRTS(uc_n,r,t,s). 0.8 VAR_UCRTS(uc_n,r,t,s). 1 Notational conventions.	138 138 138 138 138 138 138 39 139
5.2 5.2 5.2 5.2 5.2 5.2 6 E 6.1 6.1 6.1	0.1 VAR_UC(uc_n). 0.2 VAR_UCR(uc_n,r). 0.3 VAR_UCT(uc_n,t). 0.4 VAR_UCRT(uc_n,r,t). 0.5 VAR_UCTS(uc_n,r,t,s). 0.6 VAR_UCRTS(uc_n,r,t,s). 0.6 VAR_UCRTS(uc_n,r,t,s). 0.7 VAR_UCRTS(uc_n,r,t,s). 0.6 VAR_UCRTS(uc_n,r,t,s). 0.7 VAR_UCRTS(uc_n,r,t,s). 0.8 VAR_UCRTS(uc_n,r,t,s). 0.9 VAR_UCRTS(uc_n,r,t,s). 0.6 VAR_UCRTS(uc_n,r,t,s). 0.7 VAR_UCRTS(uc_n,r,t,s). 0.8 VAR_UCRTS(uc_n,r,t,s). 0.9 VAR_UCRTS(uc_n,r,t,s). 0.9 VAR_UCRTS(uc_n,r,t,s). 0.9 VAR_UCRTS(uc_n,r,t,s). 0.9 VAR_UCRTS(uc_n,r,t,s). 0.9 VAR_UCRTS(uc_n,r,t,s). 1 Notation for summations. 2 Notation for logical conditions	138 138 138 138 138 138 138 138 139 139 140

6.1.3	Using Indicator functions in arithmetic expressions	. 140
6.2 Obie	ctive function EO_OBJ	. 141
6.2.1	Introduction and notation	141
6.2.1.1	Notation relative to time	142
6.2.1.2	Other notation	143
6.2.1.3	Reminder of some technology attribute names (each indexed by t)	143
6214	Discounting options	143
6215	Components of the Objective function	144
62.2	Investment costs: INVCOST(v)	144
623	Taxes and subsidies on investments	151
624	Decommissioning (dismantling) capital costs: INVDECOM(v)	152
625	Fixed annual costs: FIXCOST(v) SURVCOST(v)	157
626	Annual taxes/subsidies on capacity: FIXTAXSUB(V)	163
627	Variable annual costs $VAPCOST(v) = v < FOH$	164
628	Cost of demand reductions $FLASTCOST(y)$	164
620	Salvage value: SAT VAGE (EOH+1)	165
6.2.10	I ate revenues from endogenous commodity recycling after FOH I ATEREVENTE(y)	172
6.2.10	Each revenues from endogenous commonly recycling after BOTE EATERE VEROE(y)	172
6212	The discounting methods for annual payments	174
0.2.12	The discounting methods for annual payments	. 174
6.3 Cons	traints	. 177
6.3.1	Bound: BND_ELAST	. 179
6.3.2	Equation EQ(<i>l</i>)_ACTBND	. 180
6.3.3	Equation: EQE_ACTEFF	. 181
6.3.4	Equation: EQ_ACTFLO	. 183
6.3.5	Equation: EQ_ACTPL	. 184
6.3.6	Equation: EQ_ACTRAMP	. 186
6.3.7	Equation: EQL_ACTUPC	. 187
6.3.8	Equation: EQE_ACTUPS	. 189
6.3.9	Equation: EQL_ACTUPS	. 190
6.3.10	Equation: EQ(<i>l</i>)_ASHAR	. 191
6.3.11	Equation: EQ(<i>l</i>)_BLND	. 193
6.3.12	Equation: EQ BNDCST	. 194
6.3.13	Equation: EQ(1)_BNDNET/PRD	. 196
6.3.14	Equation: EQ(<i>l</i>)_CAFLAC	. 197
6.3.15	Equation: EQ(1) CAPACT	. 199
6.3.16	Equation: EQL CAPFLO	. 204
6.3.17	Equation: EQ CAPLOAD	. 205
6.3.18	Equation: $EQ(l)$ _CPT	. 207
6.3.19	Equation: EQ(<i>l</i>)_COMBAL	. 209
6.3.20	Equation: EQE_COMPRD	. 218
6.3.21	Equation: EQ CUMFLO	. 219
6.3.22	Equation: EQ CUMNET/PRD	. 220
6.3.23	Equation: EQ CUMRET	. 222
6.3.24	Equation EQ_DSCNCAP	. 223
6.3.25	Equation: EQ_DSCONE	. 225
6.3.26	Equation: EQ_DSCRET	. 226
6.3.27	Equation: EQ(l)_FLOBND	. 227
6.3.28	Equation: EQ(1)_FLOFR	. 229
6.3.29	Equation: EQ(1) FLMRK	. 231
6.3.30	Equations related to exchanges (EQ IRE, EQ IREBND, EQ XBND)	. 234
6.3.30.	1 Equation EQ IRE	. 235
6.3.30.	2 Equation: EQ(1)_IREBND	238
6.3.30.	3 Equation: EQ(1)_XBND	. 241
6.3.31	Equations: EQ(1) INSHR, EQ(1) OUTSHR	. 243
6.3.32	Equation: EQ PEAK	. 245
6.3.33	Equation: EQ PTRANS	. 248
6.3.34	Equations: EQL SCAP	. 251
6.3.35	Equations: EQ STGAUX	. 252
6.3.36	Equation: EQ_STGTSS/IPS	. 253
6.3.36.	1 EQ STGTSS: Storage between timeslices (including night-storage devices):	254
6.3.36.	2 EQ STGIPS: Storage between periods	. 255

6.3.37	Equations: EQ(l)_STGIN / EQ(l)_STGOUT	256
6.3.38	Equations: EQ STSBAL	257
6.3.39	Equations: EO(l) UCRTP	258
6.4 User	Constraints	260
6.4.1	Overview	260
6.4.2	LHS user constraints	271
6.4.2.1	Equation: EQ(<i>l</i>)_UC / EQE_UC	274
6.4.2.2	Equation: EQ(1)_UCR / EQE_UCR	275
6.4.2.3	Equation: EQ(1)_UCT / EQE_UCT	276
6.4.2.4	Equation: EQ(1) UCRT / EQE UCRT	277
6.4.2.5	Equation: EQ(1) UCRTS / EQE UCRTS	278
6.4.2.6	Equation: EQU UCTS / EQE UCTS	279
643	Timeslice-dynamic user constraints	280
6431	Equation: EQ(1) LICRS / EQE LICRS	280
644	Dynamic user constraints of type (t t+1)	282
6441	Equation: EQ(A) UCSU/EQE UCSU	280
6.4.4.1	Equation: EQ(1)_0CS0 / EQE_0CS0	202
0.4.4.2	Equation EQ(I)_UCKSU / EQE_UCKSU	291
0.4.4.3	Equation. EQ(I)_UCKSUS / EQE_UCKSUS	293
6.4.4.4	Equation: EQ(I)_UCSUS / EQE_UCSUS	295
6.4.5	Dynamic user constraints of type (t-1,t).	297
6.4.5.1	Equation: $EQ(l) _ UCSU / EQE _ UCSU$	301
6.4.5.2	Equation: EQ(1)_UCRSU / EQE_UCRSU	303
6.4.5.3	Equation: EQ(<i>l</i>)_UCRSUS / EQE_UCRSUS	305
6.4.5.4	Equation: EQ(1)_UCSUS / EQE_UCSUS	307
6.4.6	User constraint modifiers	309
6.4.6.1	Overview	309
6.4.6.2	Cost modifiers (COST, TAX, SUB, DELIV)	309
6.4.6.3	Annuity modifiers (INVCOST, INVTAX, INVSUB)	309
6.4.6.4	CAPACT modifier	310
6.4.6.5	CAPFLO modifier	311
6466	5 CUMSUM modifier	311
6467	EFF modifier	311
6468	GROWTH modifier	311
6460) NET modifier	211
6461	0 NEWELO modifier	211
6461	1 ONTINE modifier	212
0.4.0.1	2 DEDIOD - Lifer	212
0.4.6.1	2 PERIOD modifier	312
6.4.6.1	3 SYNC modifier	312
6.4.6.1	4 YES modifier	312
6.4.7	Non-binding user constraints	312
		313
AFFEND		515
APPEND	IX B DAMAGE COST FUNCTIONS	341
		. Т I

1 Introduction

The purpose of the Reference Manual is to lay out the full details of the TIMES model, including data specification, internal data structures, and mathematical formulation of the model's Linear Program (LP) formulation, as well as the Mixed Integer Programming (MIP) formulations required by some of its options. As such, it provides the TIMES modeller/programmer with sufficiently detailed information to fully understand the nature and purpose of the data components, model equations and variables. A solid understanding of the material in this Manual is a necessary prerequisite for anyone considering making programming changes in the TIMES source code.

The Reference Manual is organized as follows:

- Chapter 1 Basic notation and conventions: lays the groundwork for understanding the rest of the material in the Reference Manual;
- Chapter 2 Sets: explains the meaning and role of various sets that identify how the model components are grouped according to their nature (e.g. demand devices, power plants, energy carriers, etc.) in a TIMES model;
 Chapter 3 Parameters: elaborates the details related to the user-provided numerical data,
- as well as the internally constructed data structures, used by the model generator (and report writer) to derive the coefficients of the LP matrix (and prepare the results for analysis);
- Chapter 4 Usage notes on special types of processes: Gives additional information about using input sets and parameters for the modelling of special types of processes: CHP, inter-regional exchange, and storage processes;
- Chapter 5 Variables: defines each variable that may appear in the matrix, both explaining its nature and indicating how it fits into the matrix structure;
- Chapter 6 Equations: states each equation in the model, both explaining its role and providing its explicit mathematical formulation. Includes user constraints that may be employed by modellers to formulate additional linear constraints, which are not part of the generic constraint set of TIMES.
- Appendix A The Climate Module;
- Appendix B The Damage Cost Functions, and
- Appendix C The Endogenous Technological Learning capability.

1.1 Basic notation and conventions

To assist the reader, the following conventions are employed consistently throughout this chapter:

- Sets, and their associated index names, are in lower and bold case, e.g., **com** is the set of all commodities;
- Literals, explicitly defined in the code, are in upper case within single quotes (note that in conformity with the GAMS syntax, single quotes must, in fact, be apostrophes), e.g., 'UP' for upper bound;
- Parameters, and scalars (constants, i.e., un-indexed parameters) are in upper case, e.g., NCAP_AF for the availability factor of a technology;
- Variables are in upper case with a prefix of VAR_, e.g., VAR_ACT corresponds to the activity level of a technology.

• Equations are in upper case with a prefix of EQ_ or EQ(1)_ with the placeholder (1) denoting the equation type (1=E for a strict equality, 1=L for an inequality with the left hand side term being smaller than or equal to the right hand side term and 1=G for an inequality with the left hand side term being greater than or equal to the right hand side term), e.g., EQ_PTRANS is the process transformation equation (strict equality), and EQG_COMBAL is the commodity balance constraint of type G (inequality).

1.2 GAMS modelling language and TIMES implementation

TIMES consists of generic variables and equations constructed from the specification of sets and parameter values depicting an energy system for each distinct region in a model. To construct a TIMES model, a preprocessor first translates all data defined by the modeller into special internal data structures representing the coefficients of the TIMES matrix applied to each variable of Chapter 5 for each equation of Chapter 6 in which the variable may appear. This step is called Matrix Generation. Once the model is solved (optimised) a Report Writer assembles the results of the run for analysis by the modeller. The matrix generation, report writer, and control files are written in GAMS¹ (the General Algebraic Modelling System), a powerful high-level language specifically designed to facilitate the process of building largescale optimisation models. GAMS accomplishes this by relying heavily on the concepts of sets, compound indexed parameters, dynamic looping and conditional controls, variables and equations. Thus there is very a strong synergy between the philosophy of GAMS and the overall concept of the RES specification embodied in TIMES, making GAMS very well suited to the TIMES paradigm.

Furthermore, by nature of its underlying design philosophy, the GAMS code is very similar to the mathematical description of the equations provided in Chapter 6. Thus, the approach taken to implement a TIMES model is to "massage" the input data by means of a (rather complex) preprocessor that handles the necessary exceptions that need to be taken into consideration to properly construct the matrix coefficients in a form ready to be applied to the appropriate variables in the respective equations. GAMS also integrates seamlessly with a wide range of commercially available optimisers that are charged with the task of solving the actual TIMES linear (LP) or mixed integer (MIP) problems that represent the desired model. This step is called the Solve or Optimisation step. CPLEX or XPRESS are the optimisers most often employed to solve the TIMES LP and MIP formulations.

The standard TIMES formulation has optional features, such as lumpy investments and endogenous technology learning. The organization and layout of the TIMES code, along with how it is processed by GAMS during a model run, is discussed in detail in PART III. In addition, a modeller experienced in GAMS programming and the details of the TIMES implementation could define additional equation modules or report routine modules based on this organization, which allows the linkage of these modules to the standard TIMES code in a flexible way. However, any thoughts of modifying the core TIMES code should be discussed and coordinated with ETSAP.

To build, run, and analyse a TIMES model, several software tools have been developed in the past or are currently under development, so that the modeller does not need to provide the input information needed to build a TIMES model directly in GAMS. These tools are the model interfaces VEDA-FE and ANSWER-TIMES, as well as the reporting and analysing tool VEDA-BE.

¹ GAMS A User's Guide, A. Brooke, D. Kendrick, A. Meeraus, R. Raman, GAMS Development Corporation, December 1998.

2 Sets

Sets are used in TIMES to group elements or combinations of elements with the purpose of specifying qualitative characteristics of the energy system. One can distinguish between onedimensional and multi-dimensional sets. The former sets contain single elements, e.g. the set **prc** contains all processes of the model, while the elements of multi-dimensional sets are a combination of one-dimensional sets. An example for a multi-dimensional set is the set **top**, which specifies for a process the commodities entering and leaving that process.

Two types of sets are employed in the TIMES framework: user input sets and internal sets. User input sets are created by the user, and used to describe qualitative information and characteristics of the depicted energy system. One can distinguish the following functions associated with user input sets:

- definition of the elements or building blocks of the energy system model (i.e. regions, processes, commodities),
- definition of the time horizon and the sub-annual time resolution,
- definition of special characteristics of the elements of the energy system.

In addition to these user sets, TIMES also generates its own internal sets. Internal sets serve to both ensure proper exception handling (e.g., from what date is a technology available, or in which time-slices is a technology permitted to operate), as well as sometimes just to improve the performance or smooth the complexity of the actual model code.

In the following sections, the user input sets and the internal sets will be presented. A special type of set is a one-dimensional set, also called index, which is needed to build multidimensional sets or parameters. At the highest level of the one-dimensional sets are the master or "domain" sets that define the comprehensive list of elements (e.g., the main building blocks of the reference energy system such as the processes and commodities in all regions) permitted at all other levels, with which GAMS performs complete domain checking, helping to automatically ensure the correctness of set definition (for instance, if the process name used in a parameter is not spelled correctly, GAMS will issue a warning). Therefore, before elaborating on the various sets, the indexes used in TIMES are discussed.

2.1 Indexes (One-dimensional sets)

Indexes (also called one-dimensional sets) contain in most cases the different elements of the energy model. A list of all indexes used in TIMES is given in Table 2. Examples of indexes are the set **prc** containing all processes, the set **c** containing all commodities, or the set **all_reg** containing all regions of the model. Some of the one-dimensional sets are subsets of another one-dimensional set, e.g., the set **r** comprising the so-called internal model regions is a subset of the set **all_reg**, which in addition also contains the so-called external model regions². To express that the set **r** depends on the set **all_reg**, the master set **all_reg** is put in brackets after the set name **r**: **r**(**all_reg**).

The set cg comprises all commodity groups³. Each commodity c is considered as a commodity group with only one element: the commodity itself. Thus the commodity set c is a subset of the commodity group set cg.

Apart from indexes that are under user control, some indexes have fixed elements to serve as indicators within sets and parameters, most of which should not be modified by the user (Table 1). Exceptions to this rule are the sets defined in the file MAPLISTS.DEF. For

² The meaning and the role of internal and external regions is discussed in Section 2.2.

³ See Section 2.2.1 for a more in-depth treatment of commodity groups.

¹⁰

example, while the process groups (**prc_grp**) listed in Table 1 are used within the code and must not be deleted, other process groups may be added by the user.

Table 1	Sets	with	fixed	elements
---------	------	------	-------	----------

Set/Index	Index Description					
name						
Sets defined in INITSYS.MOD (never to be changed by the user)						
bd(lim)	Index of bound type; subset of the set lim having the internally fixed elements 'LO', 'UP', 'FX'.					
costagg	List of cost ac	gregation types available for user-defined cost constraints:				
	INV investment costs					
	INVTAX	investment taxes				
	INVSUB	investment subsidies				
	INVTAXSUB	investment taxes and subsidies				
	INVALL	all investment costs, taxes and subsidies				
	FOM	fixed O&M costs				
	FOMTAX	fixed operating taxes				
	FOMSUB	fixed operating subsidies				
	FOMTAXSUB	fixed operation taxes and subsidies				
	FOMALL	all fixed operation costs, taxes and subsidies				
	COMTAX	commodity taxes				
	COMSUB	commodity subsidies				
	COMTAXSUB	commodity taxes and subsidies				
	FLOTAX	taxes				
	FLOSUB	subsidies				
	FLOTAXSUB	flow taxes and subsidies				
	FIX	total fixed costs (investment+fixed O&M costs)				
	FIXTAX	total fixed taxes				
	FIXSUB	total fixed subsidies				
	FIXTAXSUB	total fixed taxes and subsidies				
	FIXALL	all fixed costs, taxes and subsidies				
	ALLTAX	all taxes				
	ALLSUB	all subsidies				
	ALLTAXSUB	all taxes and subsidies				
ie	Export/import 'IMP' standing	t exchange index; internally fixed to the two elements: g for import and 'EXP' standing for export.				
io	Input/Output	index; internally fixed elements: 'IN', 'OUT'; used in				
	combination v	with processes and commodities as indicator whether a				
	commodity er	nters or leaves a process.				
lim	Index of limit	types; internally fixed to the elements 'LO', 'UP', 'FX', 'N'.				
side	Index of cons	traint sides; internally fixed to the elements 'LHS', 'RHS'				
tslvl	Index of time	slice levels; internally fixed to the elements 'ANNUAL',				
	'SEASON', 'W	EEKLY', 'DAYNITE'.				
uc_grptype	Index of inter	nally fixed key types of variables:				
	ACT, FLO, IRE, CAP, NCAP, COMPRD, COMNET, COMCON, UCN					
	These are use	ed in association with the user constraints.				
uc_cost	Internally fixe	ed list of cost types that can be used as modifier attributes				
	in user constr	aints: COST, DELIV, TAX, SUB				
uc_name	List of interna	Illy fixed indicators for attributes able to be referenced as				
	coefficients in	user constraints (e.g. the flow variable may be multiplied				
	by the attribu	te FLO_COST in a user constraint if desired):				
	COST, DELIV,	TAX, SUB, EFF, NET, BUILDUP, CAPACT, CAPFLO,				
	GROWTH, NE	WFLO, ONLINE, PERIOD, PERDISC, INVCOST, INVTAX,				
	INVSUB, CUM	SUM, SYNC, YES				
	See Section 6.4.6 for more detailed information.					

Set/Index Description						
name						
Sets defined in	MAPLIST.DEF (additional elements may be added by user)					
com_type Indicator of commodity type; initialized to the following elements:						
	DEM demand					
	NRG energy					
	FNV environment					
	EIN financial					
	The predefined elements should never be deleted					
dem sect	List of demand sectors: internally established in MAPLIST DEF as:					
dem_seet	AGR agriculture					
	RES residential					
	COM commercial and public services					
	IND industry					
	TRN transport					
	NE non-energy					
	OTH other					
	The predefined elements should not be deleted.					
env_type	List of emission types; internally established in MAPLIST.DEF as:					
	GHG greenhouse gas					
	PEM particulate matter emissions					
	OEM other emissions into air or water					
www.docewa	UTHENV other environmental indicator					
Ing_type	EOSCIL fossil fuel					
	SYNTH synthetic fuel					
	ERERENEW free renewable					
	LIMPENEW limited renewable (no commodity balance)					
	FIC electricity					
	HTHEAT high temperature heat					
	LTHEAT low temperature heat					
	CONSRV conservation					
	The predefined elements should not be deleted.					
prc_grp	List of process groups; internally established in MAPLIST.DEF as:					
	CHP combined heat and power plant					
	DISTR distribution process					
	DMD demand device					
	ELE electricity producing technology excluding CHP					
	HPL heat plant					
	IRE Inter-regional exchange process					
	DDE approved to charled up to the table of tabl					
	PRL energy technology not raining in other groups					
	PRW technology with material output measured in weight units					
	REF refinery process					
	RENEW renewable energy technology					
	XTRACT extraction process					
	NST night (off-peak) storage process					
	STG storage process (timeslice storage, unless also STK/NST)					
	STK stockpiling process (inter-period storage)					
	STS generalized timeslice storage					
	The user may augment this list with any additional groups desired. The					
	following predefined groups affect the data processing carried out by the					
	model generator, and should not be deleted by the user: CHP, DISTR,					
	DMD. FLE. HPL. IRE. PRE. PRV. PRW. REE. NST. STG. STK and STS.					

Table 2: Indexes in TIMES

Index⁴	Aliases⁵	Related Indexes ⁶	Description	
age	life, jot		Index for age (number of years since installation) into a parameter shaping curve: default elements 1–200.	
all r	all reg	r	All internal and external regions.	
bd	bnd_type	lim	Index of bound type; subset of lim, having the internally fixed elements 'LO', 'UP', 'FX'.	
c(cg)	com, com1, com2, com3	cg	User defined7 list of all commodities in all regions; subset of cq.	
cg	com_grp, cg1, cg2, cg3, cg4	с	User defined list of all commodities and commodity groups in all regions ⁸ ; each commodity itself is considered a commodity group; initial elements are the members of com type	
com_type			Indicator of commodity type; initialized to the predefined types DEM, NRG, MAT, ENV, FIN (see Table 1).	
costagg			Indicator of cost aggregation type; initialized to list of predefined types (see Table 1).	
cur	cur		User defined list of currency units.	
datayear (year)		year	Years for which model input data are specified.	
dem_sect			Indicator of demand sector; initialized to list of predefined sectors (see Table 1);	
env_type			Indicator of environmental commodity type; initialized to list of predefined elements (see Table 1);	
ie	impexp		Export/import exchange indicator; internally fixed = 'EXP' for exports and 'IMP' for imports.	
io	inout		Input/Output indicator for defining whether a commodity flow enters or leaves a process; internally fixed = 'IN' for enters and 'OUT' for leaves.	
j	jj		Indicator for elastic demand steps and sequence number of the shape/multi curves; default elements 1–999.	
kp			Index for "kink" points in ETL formulation; currently limited to 1-6 {can be extended in <case>.run file by including SET KP / 1*n /; for n-kink points.</case>	
lim	lim_type, l	bd	Index of limit types; internally fixed = 'LO', 'UP', 'FX' and 'N'.	
nrg_type			Indicator of energy commodity type; initialized to predefined types (see Table 1);	
р	prc		User defined list of all processes in all regions9.	
pastyear	pyr	modlyear, year	Years for which past investments are specified; pastyears must be before the beginning of the first period.	
prc_grp			List of process groups; internally established in MAPLIST.DEF (see Table 1).	
r(all_r)	reg	all_r	Explicit regions within the area of study.	

 ⁴ This column contains the names of the indexes as used in this document.
 ⁵ For programming reasons, alternative names (aliases) may exist for some indexes. This information is only relevant for those users who are interested in gaining an understanding of the underlying GAMS code.

⁶ This column refers to possible related indexes, e.g. the index set \mathbf{c} is a subset of the index

set **cg**. ⁷ VEDA/ANSWER compiles the complete list from the union of the commodities defined in each region. ⁸ VEDA/ANSWER compiles the complete list from the union of the commodity groups

defined in each region.

⁹ VEDA/ANSWER compiles the complete list from the union of processes defined in each region.

Index⁴	Aliases⁵	Related Indexes ⁶	Description	
S	all_ts, ts, s2, sl		Timeslice divisions of a year, at any of the tslvl levels.	
side			Side indicator for defining coefficients in user constraints; internally fixed = 'LHS', 'RHS'	
t	milestonyr, tt	year	Representative years for the model periods.	
teg		p	Technologies modelled with endogenous technology learning.	
tslvl			Timeslice level indicator; internally fixed = 'ANNUAL', 'SEASON', 'WEEKLY', 'DAYNITE'.	
u	units	units_com, units_cap, units_act	List of all units; maintained in the file UNITS.DEF.	
uc_grptyp e			Fixed internal list of the key types of variables (see Table 1).	
uc_n	ucn		User specified unique indicator for a user constraint.	
uc_name			Fixed list of indicators associated with various attributes that can be referenced in user constraints to be applied when deriving a coefficient (see Table 1).	
unit			List of capacity blocks that can be added in lumpy investment option; default elements 0–100; the element '0' describes the case when no capacity is added.	
units_act		u	List of activity units; maintained in the file UNITS.DEF.	
units_cap		u	List of capacity units; maintained in the file UNITS.DEF.	
units_co m		u	List of commodity units; maintained in the file UNITS.DEF.	
v(year)	modlyear	pastyear, t	t Union of the set pastyear and t corresponding to all modelling periods.	
ww	allsow	sow, w	States of the world that can be used; default elements 1 64; under user control by the dollar control parameter \$SET MAXSOW <n> in the <case>.RUN file</case></n>	
year	allyear, ll	datayear, pastyear, modlyear, milestonyr	Years that can be used in the model; default range 1850- 2200; under user control by the dollar control parameters \$SET BOTIME <y> and \$SET EOTIME <y> in the <case>.RUN file.</case></y></y>	

2.2 User input sets

The user input sets contain the fundamental information regarding the structure and the characteristics of the underlying energy system model. The user input sets can be grouped according to the type of information related to them:

- One dimensional sets defining the components of the energy system: regions, commodities, processes;
- Sets defining the Reference Energy System (RES) within each region;
- Sets defining the inter-connections (trade) between regions;
- Sets defining the time structure of the model: periods, timeslices, timeslice hierarchy;
- Sets defining various properties of processes or commodities.

The formulation of user constraints also uses sets to specify the type and the features of a constraint. The structure and the input information required to construct a user constraint is covered in detail in Chapter 6, and therefore will not be presented here.

Most of the set specifications are handled for the user by the user shell through process and commodity characterization, and the user does not need to input these sets directly.

In the following subsections first the sets related to the definition of the RES will be described (subsection 2.2.1), then the sets related to the time horizon and the sub-annual representation of the energy system will be presented (subsection 2.2.2). The mechanism for defining trade between regions of a multi-regional model is discussed in subsection 2.2.3. Finally, an overview of all possible user input sets is given in subsection 2.2.4.

2.2.1 Definition of the Reference Energy System (RES)

A TIMES model is structured by regions (all_r) . One can distinguish between external regions and internal regions. The internal regions (\mathbf{r}) correspond to regions within the area of study, and for which a RES has been defined by the user. Each internal region may contain processes and commodities to depict an energy system, whereas external regions serve only as origins of commodities (e.g. for import of primary energy resources or for the import of energy carriers) or as destination for the export of commodities. A region is defined as an internal region by putting it in the internal region set (\mathbf{r}) , which is a subset of the set of all regions **all_r**. An external region needs no explicit definition, all regions that are member of the set **all_r** but not member of \mathbf{r} are external regions is arbitrary. The main building blocks of the RES are processes (\mathbf{p}) and commodities (\mathbf{c}), which are connected by commodity flows to form a network. An example of a RES with one internal region (UTOPIA) and two external regions (IMPEXP, MINRNW) is given in Figure 1.

All components of the energy system, as well as nearly the entire input information, are identified by a region index. It is therefore possible to use the same process name in different regions with different numerical data (and description if desired), or even completely different commodities associated with the process.



Figure 1: Example of internal and external regions in TIMES

2.2.1.1 Processes

A process may represent an individual plant, e.g. a specific existing nuclear power plant, or a generic technology, e.g. the coal-fired IGCC technology. TIMES distinguishes three main types of processes:

- Standard processes;
- Inter-regional exchange processes, and
- Storage processes.

2.2.1.1.1 Standard processes

The so-called standard processes can be used to model the majority of the energy technologies, e.g., condensing power plants, heat plants, CHP plants, demand devices such as boilers, coal extraction processes, etc. Standard processes can be classified into the following groups:

- PRE for generic energy processes;
- PRW for material processing technologies (by weight);
- PRV for material processing technologies (by volume);
- REF for refinery processes;
- ELE for electricity generation technologies;
- HPL for heat generation technologies;
- CHP for combined heat and power plants;
- DMD for demand devices;
- DISTR for distribution systems;
- MISC for miscellaneous processes.

The process classification is done via the set **prc_map(r,prc_grp,p)**. This grouping is mainly intended for reporting purposes, but in some cases it also affects the properties of the

processes¹⁰ and the constraint matrix. The set is maintained in the MAPLIST.DEF file, and may be adjusted by user with additional technology groups of interest, with some restrictions as noted in Table 1.

The topology of a standard process is specified by the set **top(r,p,c,io)** of all quadruples such that the process **p** in region **r** is consuming (**io** = 'IN') or producing (**io** = 'OUT') commodity **c**. Usually, for each entry of the topology set **top** a flow variable (see **VAR_FLO** in Chapter 5) will be created. When the so-called *reduction algorithm* is activated, some flow variables may be eliminated and replaced by other variables (see PART III, Section 3.7 for details).

The activity variable (VAR_ACT) of a standard process is in most cases equal to the sum of one or several commodity flows on either the input or the output side of a process. The activity of a process is limited by the available capacity, so that the activity variable establishes a link between the installed capacity of a process and the maximum possible commodity flows entering or leaving the process during a year or a subdivision of a year. The commodity flows that define the process activity are specified by the set $prc_actunt(r,p,cg,u)$ where the commodity index cg may be a single commodity or a user-defined commodity group, and u is the activity unit. The commodity group defining the activity of a process is also called Primary Commodity Group (PCG).



Figure 2: Example of the definition of a commodity group and the activity of a normal process.

User-defined commodity groups are specified by means of the set **com_gmap(r,cg,c)**, which indicates the commodities (c) belonging to the group (cg). Once a user-defined commodity group has been defined, one can use it for any processes for defining attributes that require a commodity group (not only for the definition of the process activity, but also for other purposes, e.g., in the transformation equation EQ_PTRANS), as long as the members of the group are valid for the particular process and the process characteristic to be defined.

An example for the definition of the activity of a process is shown in Figure 2. In order to define the activity of the process SRE as the sum of the two output flows of gasoline (GSL) and diesel (DSL), one has to define a commodity group called CG_SRE containing these two commodities. The name of the commodity group can be arbitrarily chosen by the modeller.

¹⁰ Important cases are the process type CHP, which activates the CHP attributes, storage process indicators (STG, STS, STK, NST), and material conversion process types PRW and PRV, which may affect the creation of the internal set **prc_spg** (see Table 5).

¹⁷

In addition to the activity of a process, one has to define the capacity unit of the process. This is done by means of the set **prc_capunt(r,p,cg,u)**, where the index **cg** denotes the primary commodity group. In the example in Figure 3 the capacity of the refinery process is defined in mtoe/a (megatonne oil equivalent). Since the capacity and activity units are different (mtoe for the capacity and PJ for the activity), the user has to supply the conversion factor from the energy unit embedded in the capacity unit to the activity unit. This is done by specifying the parameter **prc_capact(r,p)**. In the example **prc_capact** has the value 41.868.



Conversion factor from capacity to activity unit PRC_CAPACT UTOPIA.SRE = 41.868



It might occur that the unit in which the commodity(ies) of the primary commodity group are measured, is different from the activity unit. An example is shown in Figure 4. The activity of the transport technology CAR is defined by commodity TX1, which is measured in passenger kilometres PKM. The activity of the process is, however, defined in vehicle kilometres VKM, while the capacity of the process CAR is defined as number of cars NOC.





The conversion factor from capacity to activity unit **prc_capact** describes the average mileage of a car per year. The process parameter **prc_actflo**(**r**,**y**,**p**,**cg**) contains the conversion factor from the activity unit to the commodity unit of the primary commodity group. In the example this factor corresponds to the average number of persons per car (1.5).

2.2.1.1.2 Inter-regional exchange processes

Inter-regional exchange (IRE) processes are used for trading commodities between regions. They are needed for linking internal regions with external regions as well as for modelling trade between internal regions. A process is specified as an inter-regional exchange process by specifying it as a member of the set **prc_map(r,'IRE',p)**. If the exchange process is connecting internal regions, this set entry is required for each of the internal regions trading with region r. The topology of an inter-regional exchange process **p** is defined by the set **top_ire(all_reg,com,all_r,c,p)** stating that the commodity **com** in region **all_reg** is exported to the region **all_r** (the traded commodity may have a different name **c** in region **all_r** than in region **all_reg**). For example the topology of the export of the commodity electricity (ELC_F) from France (FRA) to Germany (GER), where the commodity is called ELC_G via the exchange process (HV GRID) is modelled by the **top_ire** entry:

top_ire('FRA', 'ELC_F', 'GER', 'ELC_G', 'HV_GRID')

The first pair of region and commodity ('FRA', 'ELC_F') denotes the origin and the name of the traded commodity, while the second pair ('GER', 'ELC_G') denotes the destination. The name of the traded commodity can be different in both regions, here 'ELC_F' in France and 'ELC_G' in Germany, depending on the chosen commodity names in both regions. As with standard processes, the activity definition set **prc_actunt(r,p,cg,u)** has to be specified for an exchange process belonging to each internal region. The special features related to interregional exchange processes are described in subsection 2.2.3.

2.2.1.1.3 Storage processes

Storage processes are used to store a commodity either between periods or between timeslices. A process (p) can be specified to be an inter-period storage (IPS) process for commodity (c) by defining the process to be of the type 'STK' and c as its PCG (or, alternatively, including it as a member of the set prc_stgips(r,p,c)). In a similar way, a process is characterised as a timeslice storage by defining the process to be of the type 'STG' and c as its PCG (alternatively, by inclusion in the set prc stgtss(r,p,c)). A special case of timeslice storage is a so-called night-storage device (NST) where the commodity for charging and the one for discharging the storage are different. An example for a night storage device is an electric heating technology which is charged during the night using electricity and produces heat during the day. Including a process in the set prc nstts(r,p,s) indicates that it is a night storage device which is charged in timeslice(s) s. More than one timeslice can be specified as charging timeslices, the non-specified timeslices are assumed to be discharging timeslices. The charging and discharging commodity of a night storage device are specified by the topology set (top). It should be noted that for inter-period storage and normal timeslice storage processes (non-NST) the commodity entering and leaving the storage (the charged and discharged commodity) should be a member of the PCG (and both should be, if they are different). Other auxiliary commodity flows are also permitted in combination with these two storage types, by including them in the topology (see Section 4.3.5).

As for standard processes, the flows that define the activity of a storage process are identified by providing the set **prc_actunt(r,p,c)** entry. In contrast to standard processes, the activity of a storage process is however interpreted as the amount of the commodity being

stored in the storage process. Accordingly the capacity of a storage process describes the maximum commodity amount that can be kept in storage.

Internally, a **prc_map(r,'STG',p)** entry is always generated for all storage processes to put the process in the group of storage processes. A further **prc_map** entry is created to specify the type of storage ('STK' for inter-period storage, 'STS' for general time-slice storage and 'NST' for a night-storage device), unless already defined so by the user.

2.2.1.2 Commodities

As mentioned before, the set of commodities (c) is a subset of the commodity group set (cg). A commodity in TIMES is characterised by its type, which may be an energy carrier ('NRG'), a material ('MAT'), an emission or environmental impact ('ENV'), a demand commodity ('DEM') or a financial resource ('FIN'). The commodity type is indicated by membership in the commodity type mapping set (com_tmap(r,com_type,c)). The commodity type affects the default sense of the commodity balance equation. For NRG, ENV and DEM the commodity production is normally greater than or equal to consumption, while for MAT and FIN the default commodity balance constraint is generated as an equality. The type of the commodity limit set (com_lim(r,c,lim)). The unit in which a commodity is measured is indicated by the commodity unit set (com_unit(r,c,units_com)). The user should note that within the GAMS code of TIMES no unit conversion, e.g., of import prices, takes place when the commodity unit is changed from one unit to another one. Therefore, the proper handling of the units is entirely the responsibility of the user (or the user interface).

2.2.2 Definition of the time structure

2.2.2.1 Time horizon

The time horizon for which the energy system is analysed may range from one year to many decades. The time horizon is usually split into several *periods* which are represented by so-called *milestone years* (t(allyear) or milestonyr(allyear), see Figure 5). Each milestone year represents a point in time where decisions may be taken by the model, e.g. installation of new capacity or changes in the energy flows. The activity and flow variables used in TIMES may therefore be considered as average values over a period. The shortest possible duration of a period is one year. However, in order to keep the number of variables and equations at a manageable size, periods are usually comprised of several years. The durations of the periods do not have to be equal, so that it is possible that the first period, which usually represents the past and is used to calibrate the model to historic data, has a length of one year, while the following periods may have longer durations. Thus in TIMES both the number of a period a the duration of each period are fully under user control. The beginning year of a period t, **B**(t), and its ending year, **E**(t), have to be specified as input parameters by the user (see Table 13 in subsection 3.1.3).

To describe capacity installations that took place before the beginning of the model horizon, and still exist during the modeling horizon, TIMES uses additional years, the socalled *past years* (**pastyear(allyear)**), which identify the construction completion year of the already existing technologies. The amount of capacity that has been installed in a pastyear is specified by the parameter **NCAP_PASTI(r,allyear,p)**, also called *past investment*. For a process, an arbitrary number of past investments may be specified to reflect the age structure in the existing capacity stock. The union of the sets **milestonyr** and **pastyear** is called **modelyear** (or **v**). The years for which input data is provided by the user are called datayears (**datayear(allyear)**). The datayears do not have to coincide with modelyears, since the preprocessor will interpolate or extrapolate the data internally to the modelyears. All pastyears

are by default included in datayears, but, as a general rule, any other years for which input data is provided should be explicitly included in the set **datayear** or that information will not be seen by the model. Apart from a few exceptions (see Table 3), all parameter values defined for years other than datayears (or pastyears) are ignored by the model generator. Due to the distinction between of modelyears and datayears, the definition of the model horizon, e.g., the duration and number of the periods, may be changed without having to adjust the input data to the new periods. The rules and options of the inter- and extrapolation routine are described in more detail in subection 3.1.1.



Figure 5: Definition of the time horizon and the different year types

One should note that it is possible to define past investments (NCAP_PASTI) not only for pastyears but also for any years within the model horizon, including the milestone years. Since the first period(s) of a model may cover historical data, it is useful to store the already known capacity installations made during this time-span as past investments and not as a bound on new investments in the model database. If one later changes the beginning of the model horizon to a more recent year, the capacity data of the first period(s) do not have to be changed, since they are already stored as past investments. This feature therefore supports the decoupling of the datayears, for which input information is provided, and the definition of the model horizon. Defining past investments for years within the actual model horizon may also be useful for identifying already planned (although not yet constructed) capacity expansions in the near future¹¹.

 $^{^{11}}$ In this case the model may still decide to add additional new capacity, if this is economical and not inhibited by any investment bounds.

²¹

Table 3: Parameters that can have values defined for any year, irrespective of datayear $^{12}\,$

Attribute name	Description		
G_DRATE	General discount rate for currency in a particular year		
MULTI	Parameter multiplier table with values by year		
ACT_CUM	Cumulative limit on process activity		
FLO_CUM	Cumulative limit on process flow		
COM_CUMPRD	Cumulative limit on gross production of a commodity for a block of years		
COM_CUMNET	Cumulative limit on net production of a commodity for a block of years		
REG_CUMCST	Cumulative limit on regional costs, taxes or subsidies		
UC_CUMACT	Coefficient for a cumulative amount of process activity in a user constraint		
UC_CUMFLO	Coefficient for a cumulative amount of process flow in a user constraint		
CM_EXOFORC	Radiative forcing from exogenous sources; included in the climate module extension (see Appendix A for a description of the climate module).		
CM_HISTORY	Climate module calibration values; included in the climate module extension (see Appendix A for a description of the climate module).		
CM_MAXC	Maximum level of climate variable; included in the climate module extension (see Appendix A for a description of the climate module).		

2.2.2.2 <u>Timeslices</u>

The milestone years can be further divided in sub-annual timeslices in order to describe for the changing electricity load within a year, which may affect the required electricity generation capacity, or other commodity flows that need to be tracked at a finer than annual resolution. Timeslices may be organised into four hierarchy levels only: 'ANNUAL', 'SEASON', 'WEEKLY' and 'DAYNITE' defined by the internal set tslvl. The level ANNUAL consists of only one member, the predefined timeslice 'ANNUAL', while the other levels may include an arbitrary number of divisions. The desired timeslice levels are activated by the user providing entries in set ts group(r,tslvl,s), where also the individual user-provided timeslices (s) are assigned to each level. An additional user input set ts_map(r,s1,s2) is needed to determine the structure of a timeslice tree, where timeslice s1 is defined as the parent node of s2. Figure 6 illustrates a timeslice tree, in which a year is divided into four seasons consisting of working days and weekends, and each day is further divided into day and night timeslices. The name of each timeslice has to be unique in order to be used later as an index in other sets and parameters. Not all timeslice levels have to be utilized when building a timeslice tree, for example one can skip the 'WEEKLY' level and directly connect the seasonal timeslices with the daynite timeslices. The duration of each timeslice is expressed as a fraction of the year by

¹² The purpose of this table is to list those parameters whose year values are independent of the input **datayear**s associated with most of the regular parameters, and therefore need not be included in the set **datayear**. For example, a value for MULTI(j,'2012') would not require including 2012 in **datayear**s if 2012 were not relevant to the other input parameters.

²²

the parameter $G_YRFR(r,s)$. The user is responsible for ensuring that each lower level group sums up properly to its parent timeslice, as this is not verified by the pre-processor. The definition of a timeslice tree is region-specific.¹³ When different timeslice names and durations are used in two regions connected by exchange processes, the mapping parameters IRE_CCVT(r,c,reg,com) for commodities and IRE_TSCVT(r,s,reg,ts) for timeslices have to be provided by the user to map the different timeslice definitions. When the same timeslice definitions are used, these mapping tables do not need to be specified by the user.



Figure 6: Example of a timeslice tree

Commodities may be tracked and process operation controlled at a particular timeslice level by using the sets **com_tsl(r,c,tslvl)** and **prc_tsl(r,p,tslvl)** respectively. Providing a commodity timeslice level determines for which timeslices the commodity balance will be generated, where the default is 'ANNUAL'. For processes, the set **prc_tsl** determines the timeslice level of the activity variable. Thus, for instance, condensing power plants may be forced to operate on a seasonal level, so that the activity during a season is uniform, while hydropower production may vary between days and nights, if the 'DAYNITE' level is specified for hydro power plants. Instead of specifying a timeslice level, the user can also identify individual timeslices for which a commodity or a process is available by the sets **com_ts(r,c,s)** and **prc_ts(r,p,s)** respectively. Note that when specifying individual timeslices for a specific commodity or process by means of **com_ts** or **prc_ts** they all have to be on the same timeslice level.

The timeslice level of the commodity flows entering and leaving a process are determined internally by the preprocessor. The timeslice level of a flow variable equals the timeslice level of the process when the flow variable is part of the primary commodity group (PCG) defining the activity of the process. Otherwise the timeslice level of a flow variable is set to whichever level is finer, that of the commodity or the process.

¹³ By setting G_YRFR(r,s)=0 one can exclude any individual timeslices from specific regions, even if only a global timeslice tree is defined for all regions (as it is the case when using VEDA-FE). In this way each region can employ a different subset of the global tree.

²³

2.2.3 Multi-regional models

If a TIMES model consists of several internal regions, it is called a multi-regional model. Each of the internal regions contains a unique RES to represent the particularities of the region. As already mentioned, the regions can be connected by inter-regional exchange processes to enable trade of commodities between the regions. Two types of trade activities can be depicted in TIMES: bi-lateral trade between two regions and multilateral trade between several supply and demand regions.

Bi-lateral trade takes place between specific pairs of regions. A pair of regions together with an exchange process and the direction of the commodity flow are first identified, where the model ensures that trade through the exchange process is balanced between these two regions (whatever amount is exported from region A to region B must be imported by region B from region A, possibly adjusted for transportation losses). The basic structure is shown in Figure 7. Bi-lateral trading may be fully described in TIMES by defining an inter-regional exchange process and by specifying the two pair-wise connections by indicating the regions and commodities be traded via the set top ire(r,c,reg,com,p). If trade should occur only in one direction then only that direction is provided in the set top_ire (export from region r into region reg). The process capacity and the process related costs (e.g. activity costs, investment costs) of the exchange process can be described individually for both regions by specifying the corresponding parameters in each regions. If for example the investment costs for an electricity line between two regions A and B are 1000 monetary units (MU) per MW and 60 % of these investment costs should be allocated to region A and the remaining 40 % to region B, the investment costs for the exchange process have to be set to 600 MU/MW in region A and to 400 MU/MW in region B.



Figure 7: Bilateral trade in TIMES

Bi-lateral trade is the most detailed way to specify trade between regions. However, there are cases when it is not important to fully specify the pair of trading regions. In such cases, the so-called *multi-lateral trade* option decreases the size of the model while preserving enough flexibility. Multi-lateral trade is based on the idea that a common marketplace exists for a traded commodity with several supplying and several consuming regions for the commodity, e.g. for crude oil or GHG emission permits. To facilitate the modelling of this kind of trade scheme the concept of marketplace has been introduced in TIMES. To model a marketplace first the user has to identify one internal region that participates both in the production and consumption of the traded commodity. Then only one exchange process is

used to link the supply and demand regions with the marketplace region using the set top_ire .¹⁴

The following example illustrates the modelling of a marketplace in TIMES. Assume that we want to set up a market-based trading where the commodity CRUD can be exported by regions A, B, C, and D, and that it can be imported by regions C, D, E and F (Figure 8).



Figure 8: Example of multi-lateral trade in TIMES

First, the exchange process and marketplace should be defined. For example, we could choose the region C as the marketplace region. The exchange process has the name XP. The trade possibilities can then be defined simply by the following six **top_ire** entries:

```
SET PRC / XP /;
SET TOP_IRE /
A .CRUD .C .CRUD .XP
B .CRUD .C .CRUD .XP
D .CRUD .C .CRUD .XP
C .CRUD .D .CRUD .XP
C .CRUD .E .CRUD .XP
C .CRUD .F .CRUD .XP
/;
```

To complete the RES definition of the exchange process, only the set **prc_actunt(r,p,c,u)** is needed to define the units for the exchange process XP in all regions:

SE	T PR	C_ACTU	NT /
А	.XP	. CRUD	.PJ
В	.XP	. CRUD	.PJ
С	.XP	. CRUD	.PJ
D	.XP	. CRUD	.PJ
Е	.XP	. CRUD	.PJ
F	.XP	. CRUD	.PJ
/;			

¹⁴ Note however that some flexibility is lost when using multilateral trade. For instance, it is not possible to express transportation costs in a fully accurate manner, if such cost depends upon the precise pair of trading regions in a specific way.

These definitions are sufficient for setting up of the market-based trade. Additionally, the user can of course specify various other data for the exchange processes, for example investment and distribution costs, and efficiencies.

2.2.4 Overview of all user input sets

All the input sets which are under user control in TIMES are listed in Table 4. For a few sets default settings exist that are applied if no user input information is given. Set names starting with the prefix 'com_' are associated with commodities, the prefix 'prc_' denotes process information and the prefix 'uc_' is reserved for sets related to user constraints. Column 3 of Table 4 is a description of each set. In some cases (especially for complex sets), two (equivalent) descriptions may be given, the first in general terms, followed by a more precise description within square brackets, given in terms of n-tuples of indices.

Remark

Sets are used in basically two ways:

- as the domain over which summations must be effected in some mathematical expression, or
- as the domain over which a particular expression or constraint must be enumerated (replicated).

In the case of n-dimensional sets, some indexes may be used for **enumeration and others for summation**. In each such situation, the distinction between the two uses of the indexes is made clear by the way each index is used in the expression.

An example will illustrate this important point: consider the 4-dimensional set **top**, having indexes r,p,c,io (see table 4 for its precise description). If some quantity $A(\mathbf{r},\mathbf{p},\mathbf{c},\mathbf{io})$ must be enumerated for all values of the third index (c=commodity) and of the last index (io=orientation), but summed over all processes (p) and regions (r), this will be mathematically denoted:

$$EXPRESSION1_{c,io} = \sum_{r,p,c,io \in top} A(r, p, c, io)$$

It is thus understood from the indexes listed in the name of the expression (c,io), that these two indexes are being enumerated, and thus, by deduction, only r and p are being summed upon. Thus the expression calculates the total of A for each commodity c, in each direction io ('IN' and 'OUT'), summed over all processes and regions.

Another example illustrates the case of nested summations, where index r is enumerated in the inner summation, but is summed upon in the outer summation. Again here, the expression is made unambiguous by observing the positions of the different indexes (for instance, the outer summation is done on the r index)

EXPRESSION
$$2_{c,io} = \sum_{r,p,c,io \in lop} B(r) \sum_{p} A(r,p)$$

Table 4:	User i	nput sets	in	TIMES
----------	--------	-----------	----	-------

Set ID/Indexes ¹⁵	Alias ¹⁶	Description
all_r	all_reg	Set of all regions, internal as well as external; a region is defined as internal by putting it in the internal region set (r), regions that are not member of the internal region set are per definition external.
c (cg)	com, com1, com2, com3	User defined list of all commodities in all regions; subset of cg .
cg	com_grp, cg1, cg2, cg3, cg4	User defined list of all commodities and commodity groups (see Figure 2) in all regions.
clu (p)		Set of cluster technologies in endogenous technology learning.
com_desc (r,c)		Commodities by region, only to facilitate different descriptions by region. The elements are pairs {r,c} , for which the description is specified according to the GAMS syntax.
com_gmap (r,cg,c)		Mapping of commodity c to user-defined commodity group cg , including itself [set of triplets { r,cg,c } such that commodity c in in group cg in region r]. ¹⁷
com_lim (r,c,lim)		Definition of commodity balance equation type [set of triplets { r,c,lim }such that commodity c has a balance of type lim (lim='UP','LO','FX', 'N') in region r]; Default: for commodities of type NRG , DM and ENV production is greater or equal consumption, while for MAT and FIN commodities the balance is a strict equality.
com_off (r,c,y1,y2)		Specifying that the commodity c in region r is not available between the years y1 and y2 [set of quadruplets { $\mathbf{r,c,y1,y2}$ } such that commodity c is unavailable from years y1 to y1 in region r]; note that y1 may be 'BOH' for the first year of the first period and y2 may be 'EOH' for the last year of the last period.
com_peak (r,cg)		Set of pairs { r,cg } such that a peaking constraint is to be generated for commodity cg in region r; note that the peaking equation can be generated for a single commodity (cg also contains single commodities c) or for a group of commodities, e.g. electricity commodities differentiated by voltage level.
com_pkts (r,cg,s)		Set of triplets { r , cg , s } such that a peaking constraint for a single commodity or a group of commodities cg (e.g. if the model differentiates between three electricity commodities: electricity on high, middle and low voltage) is to be generated for the timeslice s ; Default: all timeslices of com_ts ; note that the peaking constraint will be binding only for the timeslice with the highest load.
com_tmap (r,com_type,c)		Mapping of commodities to the main commodity types (see com_type in Table 1); [set of triplets { r,com_type,c } such that commodity c has type com_type];

¹⁵ The first row contains the set name. If the set is a one-dimensional subset of another set, the second row contains the parent set in brackets. If the set is a multi-dimensional set, the second row contains the index domain in brackets.
¹⁶ For programming reasons, alternative names (aliases) may exist for some indexes. This

¹⁶ For programming reasons, alternative names (aliases) may exist for some indexes. This information is only relevant for those users who are interested in gaining an understanding of the underlying GAMS code.

¹⁷ For multidimensional sets such as this one, two definitions are sometimes given, one as an indicator function or mapping, the other (in square brackets) as a set of n-tuples.

²⁷

Set ID/Indexes ¹⁵	Alias ¹⁶	Description
com_ts (r,c,s)		Set of triplets { r , c , s } such that commodity c is available in timeslice s in region r ; commodity balances will be generated for the given timeslices; Default: all timeslices of timeslice level specified by com_tsl .
com_tsl (r,c,tslvl)		Set of triplets { r , c , tslvl } such that commodity c is modelled on the timeslice level tslvl in region r; Default: 'ANNUAL timeslice level.
com_unit (r,c,units_com)		Set of triplets { r,c,units_com } such that commodity c is expressed in unit units_com in region r .
cur		User defined list of currency units.
datayear (year)		Years for which model input data are to be taken; No default.
dem_smap (r,dem_sect,c)		Mapping of demands to main demand sectors (see dem_sect in Table 1); [set of triplets { r,dem_sect,c } such that commodity c belongs to sector dem_sect];
env_map (r,env_grp,c)		Mapping of environmental commodities to main types (see env_grp in Table 1); [set of triplets { r,env_grp,c } such that commodity c is of type env_grp].
nrg_tmap (r,nrg_type,c)		Mapping of energy commodities to main types (see nrg_type in Table 1); [set of triplets { r,nrg_type,c } such that commodity c is of type nrg_type].
р	prc	User defined list of all processes in all regions
pastyear (year)	pyr	Years for which past investments are specified; pastyears have to lie before the beginning of the first period; No default.
prc_actunt		Definition of activity [Set of quadruples such that the
(r,p,cg,units_act)		commodity group cg is used to define the activity of the process p , with units units_act , in region r].
prc_aoff (r,p,y1,y2)		Set of quadruples { r , p , y1 , y2 } such that process p cannot operate (activity is zero) between the years y1 and y2 in region r; note that y1 may be 'BOH' for first year of first period and y2 may be 'EOH' for last year of last period.
prc_capunt (r,p,cg,units_cap)		Definition of capacity unit of process p [set of quadruples { r,p,cg,units_cap } such that process p uses commodity group cg and units units cap to define its capacity in region r].
prc_desc (r,p)		Processes by region, only to facilitate different descriptions by region. The elements are pairs {r,p} , for which the process description is specified according to the GAMS syntax.
prc_dscncap (r,p)		Set of processes \mathbf{p} to be modelled using the lumpy investment formulation in region \mathbf{r} ; Default: empty set. If \mathbf{p} is not in this set, then any lumpy investment parameters provided for \mathbf{p} are ignored.
prc_foff (r,p,c,s,y1,y2)		Set of sextuples specifying that the flow of commodity c at process p and timeslice s is not available between the years y1 and y2 in region r; note that y1 may be 'BOH' for first year of first period and y2 may be 'EOH' for last year of last period.
prc_grp		List of process groups, used mainly for reporting purposes; Predefined list of groups (defined in MAPLIST.DEF) is shown in section 2.2.1.
prc_map		Grouping of processes into process groups (prc_grp) [set of
(r,prc_grp,p)		triplets { r , prc_grp , p } such that process p belongs to group prc_grp in region r]. Note: used strictly for reporting purposes.
prc_noff (r,p,y1,y2)		Set of quadruples { r , p , y 1, y 2} such that new capacity of process p cannot be installed between the years y1 and y2 in region r ; note that y1 may be 'BOH' for first year of first period and y2 may be 'EOH' for last year of last period.
prc_nstts (r,p,s)		Set of triplets { r , p , s } such that process p is a night storage device with charging timeslices s in region r; note that for night storage devices the commodity entering and the commodity leaving the storage may be different, as defined via the set ton

Set ID/Indexes ¹⁵	Alias ¹⁶	Description
prc_pkaf		Set of pairs $\{all, r, n\}$ such that the availability factor
(all r n)		(ncap af) is to be used as value for the fraction of capacity of
(un_r/p)		process p that can contribute to the peaking constraints
		(ncap nkcnt) in region r
nrc nkno		Set of pairs $\{all r, n\}$ such that process n cannot be used in the
(all r n)		peaking constraints in region r
prc_rcap		Set of pairs $\{\mathbf{r}, \mathbf{n}\}$ such that early retirements are activated for
(r.p)		process p in region r .
nrc ts	prc_ts2	Set of triplets {all r.p.s} such that process p can operate at
(all rns)	p. 0_002	timeslice s in region r: Default: all timeslices on the timeslice
(un_()p)0)		level specified by prc tsl
prc tsl		Set of triplets { r.p.tslvi } such that process p can operate at
(r.p.tslvl)		timeslice level tsivi in region r : Default: 'ANNUAL' timeslice
		level.
nrc vint		Set of processes p that are vintaged technologies in region r
(r.p)		i.e. technical characteristics are tied to when the capacity was
		installed not the current period. Default: process is not
		vintaged: note that vintaging increases the model size.
r	rea	Set of internal regions: Subset of all r .
(all reg)		
s	all ts ts s2	Set of all timeslices (define the sub-annual divisions of a
L.	sl	period). Timeslices effectively defined for specific processes and
		technologies are subsets of this set
+	milestonyr	Set of representative years (middle years) for the model
(vear)	tt	periods within the modelling horizon
tea		Set of technologies selected for endogenous technology
(n)		learning: Subset of set p : if p not in teg then any FTI
		investment parameters provided are ignored.
ton		RES topology definition indicating that commodity c enters
(r n c io)		$(i\mathbf{n}='IN')$ or leaves $(i\mathbf{n}='OUT')$ the process n [set of quadruples
(1,12,10)		$\{\mathbf{r}, \mathbf{n}, \mathbf{c}, \mathbf{io}\}$ such that process n has a flow of commodity c with
		orientation io in region r].
top ire		RES topology definition for trade between regions [Set of
(all reg.com.		quintuples indicating that commodity com from region all reg
all r.c.p)		is traded (exported) via exchange process p (where it is
		imported) into region all r as commodity c1; note: the name of
		the traded commodity may be different in the two regions.
		By using all reg=all r , one can also define bi-directional
		processes within a region, e.g. for modeling transmission lines.
ts group		Set of triplets {all r,tsivi,s} such that timeslice s belongs to
(all_r,tslvl,s)		the timeslice level tslvl in region r ; needed for the definition of
		the timeslice tree; only default is that the 'ANNUAL' timeslice
		belongs to the 'ANNUAL' timeslice level.
ts_map		Set of triplets { all_r,s,ts } such that s is an intermediate node
(all_r,s,ts)		s of the timeslice tree (neither 'ANNUAL' nor the lowest level),
		and ts is a node directly under s in region r ; the set is further
		extended by allowing $ts = s$ (see figure 1).
uc_attr		Set of quintuples such that the UC modifier specified by the
(r,uc_n,side,		uc_name (e.g., cost, conversion factor, etc.) will be applied to
uc_grptype,		the coefficient for the variable identified by uc_grptype in the
uc_name)		user constraint uc_n , for the side side ('LHS' or 'RHS') in
		region r; if uc_name='GROWTH' the user constraint represents
		a growth constraint.
uc_n	ucn	List of user specified unique indicators of the user constraints.
uc_dynbnd		List of user constraint names uc_n that will be handled as
(uc_n,bd)		simplified process-wise dynamic bound constraints of type bd.
		Can be used together with UC_ACT, UC_CAP, and UC_NCAP for
		defining the growth/decay coefficients and RHS constants for
		the dynamic bounds. See EQ_UCRTP for information on usage.
uc_r_each		Set of pairs {all_r,uc_n} such that the user constraint uc_n is
(all r.uc n)		to be generated for each specified region all r .

Set ID/Indexes ¹⁵	Alias ¹⁶	Description
uc_r_sum (all_r,uc_n)		Set of pairs {all_r,uc_n} indicating that the user constraint uc_n is summing over all specified regions all_r (that is these constraints do not have a region index). Note that depending on the specified regions in ur_r_sum, the summation may be done only over a subset of all model regions. For example if the model contains the regions FRA, GER, ESP and one wants to create a user constraint called GHG summing over the regions FRA and GER but not ESP, the set uc_r_sum contains has the two entries {'FRA', 'GHG'} and {'GER', 'GHG'}.
uc_t_each		Indicator that the user constraint uc_n is to be generated for
uc_t_succ (r,uc_n,t)		Indicator that the user constraint uc_n is to be generated between the two successive periods t and t +1.
uc_t_sum (r,uc_n,t)		Indicator that the user constraint uc_n is to be generated summing over the periods t .
uc_ts_each (r,uc_n,s)		Indicator that the user constraint uc_n will be generated for each specified timeslice s .
uc_ts_sum (r,uc_n,s)		Indicator that the user constraint uc_n is to be generated summing over the specified timeslice s .
uc_tsl (r,uc_n,side,tslvl)		Indicator of the target timeslice level tslvi of a timeslice- dynamic (or pseudo-dynamic) user constraint uc_n .
V	modlyear	Union of the sets pastyear and t corresponding to all the years (periods) of a model run (thus actually an internal set).

2.3 Definition of internal sets

The sets internally derived by the TIMES model generator are given in Table 5. The list of internal sets presented here concentrates on the ones frequently used in the model generator and the ones used in the description of the model equations in Chapter 6. Some internal sets are omitted from Table 5 as they are strictly auxiliary sets of the preprocessor whose main purpose is the reduction of the computation time for preprocessor operations.

Set ID ¹⁸ Indexes ¹⁹	Description
afs (r,t,p,s,bd)	Indicator that the internal parameter COEF_AF, which is used as coefficient of the capacity (new investment variable VAR_NCAP plus past investments NCAP_PASTI) in the capacity utilization constraint EQ(I)_CAPACT, exists.
bohyear (*) ²⁰	Set allyear plus element ' BOH ' (Beginning Of Horizon).
dm_year (year)	Union of sets datayear and modlyear
eachyear (year)	Set of all years between scalars MINYR (first year needed for cost calculation in objective function) and MIYR_VL + DUR_MAX (estimation of last year possible cost terms may occur).
eohyear (*)	Set allyear plus element 'EOH' (Ending OF Horizon)
eohyears (year)	Set of all years between scalars MINYR (first year needed for cost calculation in objective function) and MIYR_VL (last year of model horizon).
finest (r,s)	Set of finest timeslices s used in region r .
fs_emis (r,p,cg,c,com)	Indicator that the flow variable (VAR_FLO) associated with emission com can be replaced by the flow variable of c multiplied by the emission factor FLO_SUM , which is used in the transformation equation (EQ_PTRANS) between the commodity group cg and the commodity com ; used in the reduction algorithm (see Part III).
g_rcur (r,cur)	Indicator of main currency cur by region r . For regions having several discounted currencies, the one having highest present value factors is selected; used for undiscounting the solution marginals.
invspred (year,jot,k,y)	Set of investment years y and commissioning years k belonging to the investment spread starting with year and having jot number of steps (used for investment and fixed cost accounting).
invstep (year,jot,y,jot)	Set of investment years y belonging to the investment spread starting with year and having jot number of steps (used for investment and fixed cost accounting).
miyr_1 (t)	First milestonyr .
no_act (r,p)	List of processes p in region r not requiring the activity variable; used in reduction algorithm
no_cap (r,p)	List of processes p in region r not having any capacity related input parameters; used in reduction algorithm.
no_rvp (r v p)	New investment in process p in region r is not possible in period v and previously installed capacity does not exist anymore

Table 5: Internal sets in	TIMES
---------------------------	-------

¹⁸ Name of the internal set as used in this documentation and the GAMS code.

¹⁹ Index domain of the internal set is given in brackets (Note: the symbols y, y1, y2, k, and **ll** all refer to **year**). ²⁰ The asterisk denotes in the modeling system GAMS a wildcard, so that domain

checking is disabled and any index may be used.

³¹

Set ID ¹⁸ Indexes ¹⁹	Description
obj 1a	Investment case small investment (NCAP ILED/D(v) \leq G ILEDNO) and
(r,v,p)	no repetition of investment (NCAP_TLIFE + NCAP_ILED >= $D(v)$) for process p in region r and vintage period v .
obj_1b	Investment case small investment (NCAP_ILED/D(v) $\leq G_{ILEDNO}$) and
(r,v,p)	repetition of investment (NCAP_TLIFE + NCAP_ILED < $D(v)$) for process p in region r and vintage period v .
obj_2a	Investment case large investment (NCAP_ILED/D(v) > G_ILEDNO) and no
(r,v,p)	repetition of investment (NCAP_TLIFE + NCAP_ILED >= $D(v)$) for process p in region r and vintage period v .
obj_2b	Investment case large investment (NCAP_ILED/D(v) > G_ILEDNO) and
(r,v,p)	repetition of investment (NCAP_TLIFE + NCAP_ILED < $D(v)$) for process p in region r and vintage period v .
obj idc	Summation control for calculating the interest during constriction (IDC)
(r,v,p,life,k,age)	for investment Cases 2.a and 2.b.
obj_sumii	Summation control for investment and capacity related taxes and
(r,v,p,life,y,jot)	subsidies of the in the annual objective function, with lifetime life, spread
	starting in commissioning year y , having jot number of steps in the
	spread, and vintage period ${f v}.$
obj_sumiii	Summation control for decommissioning costs with for the running year
(r,v,p,ll,k,y)	index y of annual objective function, vintage period v , startup-year II , and
	commissioning year k (e.g. for spreading decommissioning costs over
nalize recorden	decommissioning time).
ODJ_SUMIV	Summation control for fixed costs in the annual objective function with
(r,v,p,me,y,joc)	of stops in the spread starting in commissioning year y, having jot number
obi cumive	Summation control for decommissioning suppoillance costs with running
(r y p k y)	vear index w of annual objective function, wintage period w and
(', ', P, N, Y)	commissioning year k
obi sums	Indicator that process p in region r with vintage period v has a salvage
(r.v.p)	value for investments with a (technical) lifetime that extends past the
(model horizon.
obj_sums3	Indicator that process p in region r with vintage period v has a salvage
(I,V,P)	Indicator that for commissioning years k process n in region r with
(r v n k)	vintage period v has a salvage value due to investment decommissioning
	or surveillance costs arising from the technical lifetime extending past the
	model horizon.
periodyr	Mapping of individual years \mathbf{y} to the modlyear (milestonyr or pastyear; \mathbf{v})
(v,y)	period they belong to; if v is a pastyear, only the pastyear itself belongs
10. TOA 81	to the period; for the last period of the model horizon also the years until
	the very end of the model accounting horizon (MIYR_VL + DUR_MAX) are
	elements of periodyr .
prc_act	Indicator that a process p in region r needs an activity variable (used in
(r,p)	reduction algorithm).
prc_cap	Indicator that a process p in region r needs a capacity variable (used in reduction place)
	Shadow primary group (SPC) of a process n : all commodities on the
(r p cq)	opposite process side of the primary commodity group (PCG) which have
(1,12,09)	the same commodity type as the PCG_usually internally determined
	(though it may be specified by the user under special circumstances (e.g.,
	when not all the commodities on the opposite side of the process, which
	should be in the SPG, are of the same commodity type com_type).
	If no commodity of the same type is found:
	if PCG is of type 'DEM' and process is a material processing process (PRV
	or PRW), then the SPG contains all material commodities on the SPG side;
	otherwise the SPG is selected as the first type among the commodity
	types on the SPG side, in the flowing order:
	When PCG type is DEM: (NRG, MAT, ENV)
	When PCG type is NRG: (MAT, DEM, ENV)
	When PCG type IS MAT: (NKG, DEM, ENV)
	WHEN PUG LYPE IS ENV: (INKG, MAT, DEM)

	·
Set ID ¹⁸	Description
Indexes ¹⁹	Description
prc_stgips	Set of triplets { r,p,c } such that process p is an inter-period storage for
(r,p,c)	the commodity c in region r ; the commodity c enters and/or leaves the
	storage according to the set top ; the storage can only operate at the
	ANNUAL level.
prc_stgtss	Set of triplets { r,p,c } such that process p is a storage process between
(r,p,c)	timeslices (e.g., seasonal hydro reservoir, day/night pumped storage) for
N 80 EDI 0	commodity c in region r ; commodity c enters and/or leaves the storage
	according to set top ; the storage operates at the timeslice level prc_tsl .
rc	List of all commodities c found in region r .
(r,c)	
rc_agp	Indicator of which commodities c are aggregated into other commodities
(r,c,lim)	by aggregation type lim .
rc_cumcom	Indicator of a cumulative constraint of type com_var defined for
(r,com_var,y1,y2,c)	commodity c between years y1 and y2
rcj	Steps j used in direction bd for the elastic demand formulation of
(r,c,j,bd)	commodity c .
rcs_combal	Indicator of which timeslices (s) associate with commodity c in region r
(r,t,c,s,bd)	for time period t the commodity balance equation (EQ(I)_COMBAL) is to
	be generated, with a constraint type corresponding to bd .
rcs_comprd	Indicator of which timeslices (s) associate with commodity c in region r
(r,t,c,s,bd)	for time period t the commodity production equation (EQ(I)_COMBAL) is
	to be generated, with a constraint type according to bd , when a
	corresponding rhs_comprd indicator exists.
rcs_comts	All timeslices s being at or above timeslice level (com_tsl) of commodity
(r,c,s)	c in region r.
rdcur	List of currencies cur that are discounted (G_DRATE provided) in each
(r,cur)	
rns_combai	Indicator that the commodity net variable (VAR_COMINET) is required in
(r,t,c,s)	commodity balance (EQE_COMBAL), owing to bounds/costs imposed on
ale al la average d	the net amount.
	required in commodity production variable (VAR_COMPRD) is
(1,1,0,5)	impressed on the production
rn	List of processos (p) in each region (p)
(rp)	
In aire	List of exchange processes (\mathbf{p}) in each region (\mathbf{r}) with indicators (\mathbf{ie})
(r p ie)	corresponding to the activity being defined by imports/exports or both
rn flo	List of all processes in region (r) except inter-regional exchange
(r,p)	processes (ire).
rp inout	Indicator as to whether a process (p) in a region (r) is input or output (io
(r,p,io)	= 'IN'/'OUT') normalized with respect to its activity.
rp_ire	List of inter-regional exchange processes (p) found in each region (all_r).
(all_r,p)	
rp_pg	The primary commodity group (cg) of each process (p) in a region (r).
(r,p,cg)	n 22 , 23 vino n 052 lantado na prostugina vincula 053 lantado
rp_pgtype	The commodity type (com_type) of primary commodity group of a
(r,p,com_type)	process (p) in a region (r).
rp_sgs	List of those standard processes (p) in each region (r), which have been
(r,p)	defined to have a night storage (NST) capability.
rp_std	List of standard processes (p) in each region (r).
(r,p)	
rp_stg	List of storage processes (p) in each region (r).
(r,p)	
rp_sts	List of generalized timeslice storage processes (p) in each region (r).
(r,p)	
rp_upl	List of those processes (\mathbf{p}) in each region (\mathbf{r}) that have dispatching
(r,p,lim)	Tattributes ACI_MINLD/ACI_UPS defined, with qualifier lim.
rp_ups	Timestices (s) of a process (p) in a region (r) during which start-ups are
(r,p,tsivi,lim)	permitted (used for processes in the set rp_upi(r,p, FX))
rpc	List of commodities (c) associated with a process p in region r (by top or
(r,p,c)	top ire).

	122
Set ID ¹⁸	Description
Indexes ¹⁹	Description
rpc act	Indicator that the primary commodity group of a process (p), except
(r,p,c)	exchange processes (see rpc_aire) consists of only one commodity (c),
	enabling the corresponding flow variable to be replaced by the activity
	variable (used in reduction algorithm).
rpc_aire	Indicator that the primary commodity group of an exchange process (p)
(r,p,c)	consists of only one commodity (c), enabling the corresponding flow
	variable to be replaced by the activity variable (used in reduction
	algorithm).
rpc_capflo	Indicator that a commodity flow ${f c}$ in region ${f r}$ is associated with the
(r,v,p,c)	capacity of a process (p , due to NCAP_ICOM, NCAP_OCOM, or NCAP_COM
	being provided).
rpc_cumflo	Indicator of a cumulative constraint defined for commodity flow ${f c}$ of
(r,p,c,y1,y2)	process p between years y1 and y2
rpc_noflo	A subset of rpc_capflo indicating those processes (p) in a region (r)
(r,p,c)	where a commodity (\mathbf{c}) is only consumed or produced through capacity
	based flows, and thus has no flow variable for the commodity.
rpc_emis	Indicator that the flow variable of an emission commodity (cg) associated
(r,p,cg)	with process (p) in a region (r) can be replaced by the fuel flow causing
	the emission multiplied by the emission factor (used in reduction
rno ogiro	digonunini).
rpc_eqire	Indicator of the commodities (\mathbf{c}) associated with inter-regional exchange
(r,p,c)	(EQ_IRE) is to be generated; the set does not contain the marketplace
	region (rnc, market)
rpcc ffunc	Eleve variable of a commodity (a) accorded with a process (b) that can
(rpc)	be replaced by another flow variable of the process due to a direct
(1,0,0)	ELO EUNC or ELO SUM relationship
rnc ire	Commodities (c) imported or exported (ie='TMP'/'FXP') via process p in
(all rncie)	a region (all r)
rpc_market	List of market regions (subset of all \mathbf{r}) that trade a commodity (\mathbf{c})
(all r.p.c.ie)	through a process (n) either by only multidirectional export links
	(ie='EXP') or by both import and export links (ie='IMP'). The market
	structure is user-defined through the set top_ire .
rpc_pg	Mapping of the commodities (c) in a region (r) that belong to the primary
(r,p,cg,c)	commodity group (cg) associated with process p.
rpc_spg	The list of commodities (c) in a region (r) belonging to the shadow
(r,p,c)	primary group of process (p).
rpc_stg	List of stored (charged/discharged) commodities (c) of storage processes
(r,p,c)	(p) in region (r).
rpc_stgn	List of those stored (charged/discharged) commodities (c) of storage
(r,p,c,io)	processes (p) in region (r), which are connected to the commodity
	balance on one side (io) only.
rpcg_ptran	Indicator of the transformation equations (EQ_PTRANS) that can be
(r,p,c1,c2,cg1,cg2)	eliminated by the reduction algorithm.
rpcs_var	The list of valid timeslices for the flow variable (VAR_FLO) of commodity c
(1,p,c,s)	associated with process p in region r , now variables of commodules which
	the process (pro_tsl) while all other flow variables are created according
	to the rns s1 timeslices
rns nrcts	$\Delta II (permitted) timeslices (s) at or above the process (n) timeslice level$
(rps)	(prc_tsl) in a region (r)
rps s1	All (permitted) timeslices (s) belonging to the finest timeslice level of the
(r.p.s)	process (p , prc tsl) and the commodity timeslice level (com tsl) of the
() / / / / / / / / / / / / / / / / / /	shadow primary commodity group.
rps s2	For an ANNUAL level NST process, contains all permitted timeslices (s) at
(r,p,s)	the level above the finest commodity timeslice levels (com tsl) of the
S 10-1-7	shadow primary group (spg). For all other processes, rps s2 = rps s1.
rps stg	Process level timeslices (s) of timeslice storage process (p) in a region
(r,p,s)	(r).
rreg	Indicator that trade exists from region all_reg to region all_r .
(all reg all r)	ter ner analare odenter konstente terterterter en ander itte odente tert i tre itte itte i itte i itte itte

a	1
Set ID ¹⁸	Description
rs below	All timeslices (a) strictly below the higher timeslice (ta) in the timeslice
(all r tr c)	
	All timeslines (a) immediately (one level) below the higher timesline (to)
(all r to c)	In the timeslice tree
	For a timeslice (te) all timeslices (a) that are an the same nothe within
(all r to c)	the timeslice tree or a lifter SP. WD in Fig. 6 wolid timeslices a pro-
(all_1,ts,s)	
ute europet	I ANNUAL, SP, SP_WD, SP_WD_D, SP_WD_N Indicator that the commodity not variable (VAD_COMNET) for commodity
rtc_cumnet	Indicator that the commodity net variable (VAR_COMINET) for commodity
(r,t,c)	c in region r for period t has a cumulative bound applied.
rtc_cumpra	Indicator that the commodity production variable (VAR_COMPRD) for
(r,t,c)	commodity \mathbf{c} in region \mathbf{r} for period \mathbf{t} has a cumulative bound applied.
rtcs_sing	Indicator that a commodity c is not available in a specific period t and
(r,t,c,s,io)	timeslice s, since the all the processes producing (Io= OUT) or consuming
	It (\mathbf{IO} = \mathbf{IN}) are turned-off. In the case of \mathbf{IO} = \mathbf{OUT} , the commodity is not
	available, meaning that processes having only this commodity as input
	cannot operate. Similar reasoning applies to the case IO = IN.
rtcs_varc	For commodity (c) in region (r) indicator for the timeslices (s) and the
(r,t,c,s)	periods (t) the commodity is available.
rtp = rvp	Indication of the periods and pastyears for which process (p) in region (r)
(r,v,p)	is available; all other RIP_* control sets are based on this set.
rtp_cptyr	For each vintage period (\mathbf{v}) an indication of the periods (\mathbf{t}) for which
(r,v,t,p)	newly installed capacity of process (p) in a region (r) is available, taking
	Into account construction lead-time (NCAP_ILED) and technical lifetime
	(NCAP_TLIFE).
rtp_off	Indication of the periods (t) in which no new investment is permitted for a
(r,t,p)	process (p) in a region (r).
rtp_vara	Indication of the periods (t) for which a process (p) in a region (r) is
(r,t,p)	
rtp_varp	Indicator that the capacity variable (VAR_CAP) will be generated for
(r,t,p)	process (p) in a region (r) in period (t).
rtp_vintyr	An indication of for which periods (t) a process (p) in a region (r) is
(r,v,t,p)	available since it was first installed (v); for vintaged processes (prc_vint)
	Identical to rtp_cptyr , for non-vintaged processes the v index in the
	[rtp_cptyr entries is ignored by setting it to \mathbf{t} ($\mathbf{v} = \mathbf{t}$).
(rtpc)	available (rtn) and commodifies associated with it (rnn)
	An indication for process (n) of the timeslices (n) for which the process is
(rtpc)	turned off (used in reduction algorithm)
	The list of valid timeslines (a) and periods (b) for the flow variable
(rtpcs_vall	(1) (AB, ELO) of process (b) and commodity (c); taking into account the
(1,0,0,0,5)	availability of the activity, capacity and flow (rth, yara, rnce, yar and
	prc foff) The timeslice level of a flow variable equals the process
	timeslice level (prc_tsl) when the flow is part of the primary commodity
	aroun of the process. Otherwise the timeslice level of a flow variable is set
	to the finest level of the commodities in the shadow group (SPG) or the
	process level, whichever is finer
uc dyndir	If side = 'RHS', indicator for growth constraints to be generated between
(r.uc. n.side)	the periods t-1 and t : if side = 'I HS', the set is ignored.
	Indicator that a commodity variable (VAR_COMCON or VAR_COMPRD) for
(r.uc n.uc arptype.c)	commodity (c) in a region (r) appears in a user constraint (uc n).
uc amap p	Indicator that a variable (VAR_ACT, VAR_NCAP or VAR_CAP) associated
(r.uc n.uc arptype.p)	with a process (p) in a region (r) appears in a user constraint (uc n).
uc gmap u	Indicator that a variable (VAR_UCRT) associated with a user constraint
$(r_{\rm uc} n_{\rm ucn})$	(ucn) in a region (r) appears in another user constraint (ucn) .
uc map flo	Indicator that the flow variable (VAR_FLO) for region r , process p and
(uc n.r.p.c)	commodity c is involved in user constraint uc n .
uc map ire	Indicator that an import/export (according to top ire) trade variable
(uc n,r,p,c)	(VAR IRE) for region r , process p , and commodity c is involved in a user
	constraint (uc_n).
V	Union of the input sets pastyear and t , corresponding to all the periods
	of a model run (=modlyear).

3 Parameters

While sets describe structural information of the energy system or qualitative characteristics of its entities (e.g. processes or commodities), parameters contain numerical information. Examples of parameters are the import price of an energy carrier or the investment cost of a technology. Most parameters are time-series where a value is provided (or interpolated) for each year (datayear). The TIMES model generator distinguishes between user input parameters and internal parameters. The former are provided by the modeller (usually by way of a data handling system or "shell" such a VEDA-FE or ANSWER-TIMES), while the latter are internally derived from the user input parameters, in combination with information given by sets, in order to calculate for example the cost coefficients in the objective function. This Chapter first covers the user input parameters in Section 3.1 and then describes the most important internal parameters as far as they are relevant for the basic understanding of the equations (Section 3.2). Section 3.3 presents the parameters used for reporting the results of a model run.

3.1 User input parameters

This section provides an overview of the user input parameters that are available in TIMES to describe the energy system. Before presenting the various parameters in detail in Section 3.1.3 two preprocessing algorithms applied to the user input data are presented, namely the inter/extrapolation and the inheritance/aggregation routines. User input parameters that are time-dependent can be provided by the user for those years for which statistical information or future projections are available, and the inter-/extrapolation routine described in Section 3.1.1 used to adjust the input data to the years required for the model run. Timeslice dependent parameters do not have to be provided on the timelice level of a process, commodity or commodity flow. Instead the so-called inheritance/aggregation routine described in Section 3.1.2 assigns the input data from the user provided timeslice level to the appropriate timeslice level as necessary.

3.1.1 Inter- and extrapolation of user input parameters

Time-dependent user input parameters are specified for specific years, the so-called *datayears* (datayear). These datayears do not have to coincide with the modelyears (v or modelyear) needed for the current run. Reasons for differences between these two sets are for example that the period definition for the model has been altered after having provided the initial set of input data leading to different milestoneyears (t or milestoneyr) or that data are only available for certain years that do not match the modelyears. In order to avoid burdening the user with the cumbersome adjustment of the input data to the modelyears, an inter/extrapolation (I/E) routine is embedded in the TIMES model generator. The inter/extrapolation routine distinguishes between a default inter-/extrapolation that is automatically applied to the input data and an enhanced user-controlled inter-/extrapolation that allows the user to specify an inter-/extrapolation rule for each time-series explicitly. Independent of the default or user-controlled I/E options, TIMES inter-/extrapolates (using the standard algorithm) all cost parameters in the objective function to the individual years of the model as part of calculating the annual cost details (see section 3.1.1.3 below).

The possibility of controlling interpolation on a time-series basis improves the independence between the years found in the primary database and the data actually used in the individual runs of a TIMES model. In this way the model is made more flexible with respect to running scenarios with arbitrary model years and period lengths, while using basically the very same input database.

3.1.1.1 Inter/extrapolation options

The TIMES interpolation/extrapolation facility provides both a default I/E method for all time-series parameters, and options for the user to control the interpolation and extrapolation of each individual time series (Table 6). The option 0 does not change the default behavior. The specific options that correspond to the default methods are 3 (the standard default) and 10 (alternative default method for bounds and RHS parameters).

Non-default interpolation/extrapolation can be requested for any parameter by providing an additional instance of the parameter with an indicator in the YEAR index and a value corresponding to one of the integer-valued Option Codes (see Table 6 and example below). This control specification activates the interpolation/extrapolation rule for the time series, and is distinguished from actual time-series data by providing a special control label ('0') in the YEAR index. The particular interpolation rule to apply is a function of the Option Code assigned to the control record for the parameter. Note that for log-linear interpolation the Option Code indicates the year from which the interpolation is switched from standard to loglinear mode. TIMES user shell(s) will provide mechanisms for imbedding the control label and setting the Option Code through easily understandable selections from a user-friendly drop-down list, making the specification simple and transparent to the user.

The enhanced interpolation/extrapolation facility provides the user with the following options to control the interpolation and extrapolation of each individual time series:

- Interpolation and extrapolation of data in the default way as predefined in TIMES. This option does not require any explicit action from the user.
 - No interpolation or extrapolation of data (only valid for non-cost parameters).
- Interpolation between data points but no extrapolation (useful for many bounds). See option codes 1 and 11 in Table 2 below.
- Interpolation between data points entered, and filling-in all points outside the interpolation window with the EPS (zero) value. This can useful for e.g. the RHS of equality-type user constraints, or bounds on future investment in a particular instance of a technology. See option codes 2 and 12 in Table 2 below.

Option code	Action	Applies to
0 (or none)	Interpolation and extrapolation of data in the default way as predefined in TIMES (see below)	All
< 0	No interpolation or extrapolation of data (only valid for non-cost parameters).	All
1	Interpolation between data points but no extrapolation.	All
2	Interpolation between data points entered, and filling-in all points outside the interpolation window with the EPS value.	All
3	Forced interpolation and both forward and backward extrapolation throughout the time horizon.	All
4	Interpolation and backward extrapolation	All
5	Interpolation and forward extrapolation	All
10	Migrated interpolation/extrapolation within periods	Bounds, RHS
11	Interpolation migrated at end-points, no extrapolation	Bounds, RHS
12	Interpolation migrated at ends, extrapolation with EPS	Bounds, RHS
14	Interpolation migrated at end, backward extrapolation	Bounds, RHS
15	Interpolation migrated at start, forward extrapolation	Bounds, RHS
YEAR (≥ 1000)	Log-linear interpolation beyond the specified YEAR, and both forward and backward extrapolation outside the interpolation window.	All

Table 6: Option codes for the control of time series data interpolation

- Forced interpolation and extrapolation throughout the time horizon. Can be useful for parameters that are by default not interpolated. See option codes 3, 4, and 5 as well as 14 and 15 in Table 2 below.
- Log-linear interpolation beyond a specified data year, and both forward and backward extrapolation outside the interpolation window. Log-linear interpolation is guided by relative coefficients of annual change instead of absolute data values.

Migration means that data points are interpolated and extrapolated within each period but not across periods. This method thus migrates any data point specified for other than milestoneyr year to the corresponding milestoneyr year within the period, so that it will be effective in that period.

Log-linear interpolation means that the values in the data series are interpreted as coefficients of annual change beyond a given YEAR. The YEAR can be any year, including modelyears. The user only has to take care that the data values in the data series correspond to the interpretation given to them when using the log-linear option. For simplicity, however, the first data point is always interpreted as an absolute value, because log-linear interpolation requires at least one absolute data point to start with.

3.1.1.2 Default inter/extrapolation

The standard default method of inter-/extrapolation corresponds to the option 3, which interpolates linearly between data points, while it extrapolates the first/last data point constantly backward/forward. This method, full interpolation and extrapolation, is by default applied to most TIMES time series parameters. However, the parameters listed in Table 7 are by default **NOT** inter/extrapolated in this way, but have a different default method.

3.1.1.3 Interpolation of cost parameters

As a general rule, all cost parameters in TIMES are densely interpolated and extrapolated. This means that the parameters will have a value for every single year within the range of years they apply, and the changes in costs over years will thus be accurately taken into account in the objective function. The user can use the interpolation options 1–5 for even cost parameters. Whenever an option is specified for a cost parameter, it will be first sparsely interpolated/extrapolated according to the user option over the union of modelyear and datayear, and any remaining empty data points are filled with the EPS value. The EPS values will ensure that despite the subsequent dense interpolation the effect of user option will be preserved to the extent possible. However, one should note that due to dense interpolation, the effects of the user options will inevitably be smoothed.

3.1.1.4 Examples of using I/E options

Example 1:

Assume that we have three normal data points in a FLO_SHAR data series:

FLO_SHAR('REG','1995','PRC1','COAL','IN_PRC1','ANNUAL','UP') = 0.25; FLO_SHAR('REG','2010','PRC1','COAL','IN_PRC1','ANNUAL','UP') = 0.12; FLO_SHAR('REG','2020','PRC1','COAL','IN_PRC1','ANNUAL','UP') = 0.05;

FLO_SHAR is by default NOT interpolated or extrapolated in TIMES. To force interpolation/extrapolation of the FLO_SHAR parameter the following control option for this data series should be added:

FLO_SHAR('REG', '0', 'PRC1', 'COAL', 'IN_PRC1', 'ANNUAL', 'UP') = 3;
Parameter	Justification	Default I/E
ACT_BND		
CAP_BND		
NCAP_BND		
NCAP_DISC		
FLO_FR		
FLO_SHAR		
STGIN_BND		
STGOUT_BND	Bound may be intended at specific periods only	10
COM_BNDNET	bound may be intended at specific periods only.	(migration)
COM_BNDPRD		
COM_CUMNET		
COM_CUMPRD		
REG_BNDCST		
RCAP_BND		
IRE_BND		
IRE_XBND		
PRC_MARK	Constraint may be intended at specific periods only	11
PRC_RESID	Residual capacity usually intended to be only interpolated	1*
UC_RHST		10
UC_RHSRT	User constraint may be intended for specific periods only	(migration)
UC_RHSRTS		(
NCAP_AFM	Interpolation meaningless for these parameters (parameter	
NCAP_FOMM	value is a discrete number indicating which MULTI curve	10
NCAP_FSUBM	should be used).	(migration)
NCAP_FIAXM		
	Interpolation meaningless for these parameters (parameter	10
	value is a discrete number indicating which SHAPE curve	
NCAP_FUMX	should be used).	(migration)
NCAP_FSUBX	20	
NCAP_FIAXA	Parameter describes pact investment for a single vintage vear	nono
NCAP DASTY	Parameter describes pumber of years over which to distribute	none
	past investments.	none
CM_MAXC	Bound may be intended at specific years only	none
PEAKDA_BL	Blending parameters at the moment not interpolated	none

- Table 7: rarameters not being funy inter/extrapolated by def
--

* If only a single PRC_RESID value is specified, assumed to decay linearly over NCAP_TLIFE years

Example 2:

Assume that we define the following log-linear I/E option for a FLO_SHAR data series: FLO_SHAR('REG', '0', 'PRC1', 'COAL', 'IN_PRC1', 'ANNUAL', 'UP') = 2005;

This parameter specifies a log-linear control option with the value for the threshold YEAR of log-linear interpolation taken from 2005. The option specifies that all data points up to the year 2005 should be interpreted normally (as absolute data values), but all values beyond that year should be interpreted as coefficients of annual change. By using this interpretation, TIMES will then apply full interpolation and extrapolation to the whole data series. It is the responsibility of the user to ensure that the first data point and all data points up to (and including) the year 2005 represent absolute values of the parameter, and that all subsequent data points represent coefficients of annual change. Using the data of the example above, the first data point beyond 2005 is found for the year 2010, and it has the value of 0.12. The interpretation thus requires that the maximum flow share of COAL in the commodity group IN_PRC1 is actually meant to increase by as much as 12% per annum between the years 1995 and 2010, and by 5% per annum between 2010 and 2020.

3.1.1.5 Applicability

All the enhanced I/E options described above are available for all TIMES time-series parameters, excluding PRC_RESID and COM_BPRICE. PRC_RESID is always interpolated, as if option 1 were used, but is also extrapolated forwards over TLIFE when either I/E option 5 or 15 is specified. COM_BPRICE is not interpolated at all, as it is obtained from the Baseline solution. Moreover, the I/E options are not applicable to the integer-valued parameters related to the SHAPE and MULTI tables, which are listed in Table 8.

Table 8: Parameters which cannot be interpolated.

Parameter	Comment				
NCAP_AFM					
NCAP_FOMM	Parameter value is a discrete numbers indicating which				
NCAP_FSUBM	MULTI curve should be used, and not a time series datum.				
NCAP_FTAXM					
COM_ELASTX					
FLO_FUNCX	Parameter value is a discrete number indicating which SHAPE curve should be used, and not a time series datum.				
NCAP_AFX					
NCAP_FOMX					
NCAP_FSUBX					
NCAP_FTAXX					

Nonetheless, a few options are supported also for the extrapolation of the MULTI and SHAPE index parameters, as shown in Table 9. The extrapolation can be done either only inside the data points provided by the user, or both inside and outside those data points. When using the inside data points option, the index specified for any **datayear** is extrapolated to all modelyears (v) between that **datayear** and the following **datayear** for which the SHAPE index is specified. The extrapolation options are available for all of the SHAPE and MULTI parameters listed in Table 8.

Table 9: Option codes for the extrapolation of SHAPE/MULTI indexes.

Option code	Action
<= 0 (or none)	No extrapolation (default)
1	Extrapolation between data points only
2	Extrapolation between and outside data points
11	Extrapolation between data points only, migration at ends

Example:

The user has specified the following two SHAPE indexes and a control option for extrapolation:

NCAP_AFX('REG', '0', 'PRC1') = 1; NCAP_AFX('REG', '1995', 'PRC1') = 12; NCAP_AFX('REG', '2010', 'PRC1') = 13;

In this case, all modelyears (v) between 1995 and 2010 will get the shape index 12. No extrapolation is done for modelyears (v) beyond 2010 or before 1995.

3.1.2 Inheritance and aggregation of timesliced input parameters

As mentioned before, processes and commodities can be modelled in TIMES on different timeslice levels. Some of the input parameters that describe a process or a commodity are timeslice specific, i.e. they have to be provided by the user for specific timeslices, e.g. the availability factor NCAP_AF of a power plant operating on a 'DAYNITE' timeslice level. During the process of developing a model, the timeslice resolution of some processes or even the entire model may be refined. One could imagine for example the situation that a user starts developing a model on an 'ANNUAL' timeslice level and refines the model later by refining the timeslice related parameters to be re-entered again for the finer timeslices, TIMES supports inheritance and aggregation of parameter values along the timeslice tree.

Inheritance in this context means that input data being specified on a coarser timeslice level (higher up the tree) are inherited to a finer timeslice level (lower down the tree), whereas aggregation means that timeslice specific data are aggregated from a finer timeslice level (lower down the tree) to a coarser one (further up the tree). The inheritance feature may also be useful in some cases where the value of a parameter should be the same over all timeslices, since in this case it is sufficient to provide the parameter value for the 'ANNUAL' timeslice which is then inherited to the required finer target timeslices.²¹

The TIMES pre-processor supports different inheritance and aggregation rules, which depend on the type of attribute. The main characteristics of the different inheritance and

Inheritance rules	Description
Direct inheritance	A value on a coarser timeslice is inherited by target timeslices below (in the timeslice tree), without changing the numeric values.
Weighted inheritance	A value on a coarser timeslice is inherited by target timeslices below (in the timeslice tree) by weighting the input value with the ratio of the duration of the target timeslices to the duration of the coarser timeslice. Example: Parameter COM_FR.
No inheritance	Absolute bound parameters specified on a coarser timeslice level than the target timeslice level are not inherited. Instead a constraint summing over related variables on the finer timeslices is generated, e.g. an annual ACT_BND parameter specified for a process with a 'DAYNITE' process timeslice level (prc_tsl) leads to a constraint (EQ_ACTBND) with the summation over the activity variables on the 'DAYNITE' level as LHS term and with the bound as RHS term.
Aggregation rules	Description
Standard aggregation	The values specified on finer timeslices are aggregated to the target timeslice being a parent node in the timeslice tree by summing over the values on the finer timeslices.
Weighted aggregation	The values specified for finer timeslices are aggregated to the target timeslice being a parent node in the timeslice tree by summing over the weighted values on the finer timeslices. The ratios of the duration of the finer timeslices to the duration of the target timeslice serve as weighting factors.

Table 10: Inheritance and aggregation rules

²¹ The term *target timeslice level* or *target timeslice* is used in the following as synonym for the timeslice level or timeslices which are required by the model generators depending on the process or commodity timeslice resolution (**prc_tsl** and **com_tsl** respectively).

⁴¹

aggregation rules are summarised in Table 10. The specific rules applied to each individual parameter are listed in the detailed reference Table 13 further below.

The different aggregation rules are illustrated by examples in Figure 9. It should be noted that if input data are specified on two timeslice levels different from the target level, then especially the weighted inheritance/aggregation method may lead to incorrect results. Therefore, at least for the parameters where weighted methods are applied, it is recommended to provide input data only for timeslices on one timeslice level. However, for parameters that are directly inherited, specifying values at multiple levels may sometimes be a convenient way to reduce the amount of values to be specified.²²

Bound parameters are in most cases not levelized by inheritance, only by aggregation. Exceptions to this rule are the relative type bound parameters NCAP_AF and FLO_SHAR, which are inherited by the target timeslices. One should also notice that, due to levelization, fixed bounds that are either inherited or aggregated to the target timeslice level will always override any upper and lower bounds simultaneously specified.



Figure 9: Inheritance and aggregation rules for timeslice specific parameters in TIMES

²² Note that as an exception, for NCAP_AF direct inheritance and aggregation will be disabled if any values are specified at the process timeslice level. However, this may be circumvented by using NCAP_AFS for defining the values at process timeslices.

3.1.3 Overview of user input parameters

A list of all user input parameters is given in Table 13. In order to facilitate the recognition by the user of to which part of the model a parameter relates the following naming conventions apply to the prefixes of the parameters (Table 11).

Prefix	Related model component			
G_	Global characteristic			
ACT_	Activity of a process			
CAP_	Capacity of a process			
COM_	Commodity			
FLO_	Process flow			
IRE_	Inter-regional exchange			
NCAP_	New capacity of a process			
PRC_	Process			
RCAP_	Retiring capacity of a process			
REG_ / R_	Region-specific characteristic			
STG_	Storage process			
UC_	User constraint			

Table 11: Naming conventions for user input parameters

For brevity, the default interpolation/extrapolation method for each parameter is given by using the abbreviations listed in Table 12.

Table 12: Abbreviations for default I/E method in Table 13.

Abbreviation	Description
STD	Standard full inter-/extrapolation (option 3)
MIG	Migration (option 10)
<number></number>	Option code for any other default method
none	No default inter-/extrapolation
N/A	Inter-/extrapolation not applicable

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
ACT_BND (r,datayear,p,s,bd)		Units of activity [0,∞); default value: none Default i/e ²⁸ : MIG	Since inter-/extrapolation default is MIG, the bound must be explicitly specified for each period, unless an inter-/extrapolation option is set. If the bound is specified for a timeslice resolution (prc_tsl), the bound is applied to the sum of the activity variables according to the timeslice tree. Standard aggregation.	Bound on the overall activity a process.	Activity limit constraint (EQ(I)_ACTBND) when s is above prc_tsl. Direct bound on activit variable (VAR_ACT) when at the prc_tsl level.
ACT_COST (r,datayear,p,cur)	OBJ_ACOST, CST_ACTC, CST_PVP	Monetary unit per unit of activity [open]; default value: none Default i/e: STD		Variable costs associated with the activity of a process.	Applied to the activity variable (VAR_ACT) as a component of the objective function (EQ_OBJVAR). May appear in user constraints (EQ_UC*) if

Table 13: User input parameters in TIMES

²³ The first row contains the parameter name, the second row contains in brackets the index domain over which the parameter is defined.
²⁴ This column gives references to related input parameters (in upper case) or sets (in lower case) being used in the context of this parameter as well as internal parameters/sets or result parameters being derived from the input parameter.
²⁵ This column lists the unit of the parameter, the possible range of its numeric value [in square brackets] and the inter-/extrapolation rules that apply.

²⁵ This column lists the unit of the parameter, the possible range of the manners, the second secon

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
					specified in UC_NAME.
ACT_CSTPL (r,datayear,p,cur)	ACT_MINLD ACT_LOSPL	Monetary unit per unit of activity [0,∞); default value: none Default i/e: STD	Used as an alternative or supplement to using ACT_LOSPL(r, γ, p, FX'). When used as an alternative, the fuel increase at the minimum operating level that should be included in the cost penalty must be embedded in the ACT_CSTPL coefficient.	Partial load cost penalty, defined as an additional cost per activity at the minimum operating level, corresponding to the efficiency loss at that load level. Added as an extra term to variable costs in the objective and reporting.	Generates an additional term in EQ_OBJVAR for the increase in operating cost.
ACT_CSTUP (r,datayear,p,tslvl,cur)	ACT_MINLD ACT_UPS	Monetary unit per unit of capacity $[0,\infty)$; default value: none Default i/e: STD	The tslvl level refers to the timeslice cycle for which the start-up cost is defined. Only applicable when the min. stable operating level has been defined with ACT MINLD.	Cost of process start- up per unit of started- up capacity. Added as an extra term to variable costs in the objective and reporting.	Activates generation of EQL_ACTUPS eqs. Generates an additional term in the variable operating costs included in EQ_OBJVAR.
ACT_CUM (r,p,y1,y2,bd)	FLO_CUM	Activity unit [0,∞); default value: none Default i/e: N/A	The years y1 and y2 may be any years of the set allyear; where y1 may also be 'BOH' for first year of first period and y2 may be 'EOH' for last year of last period.	Bound on the cumulative amount of annual process activity between the years y1 and y2, within a region.	Generates an instance of the cumulative constraint (EQ_CUMFLO)
ACT_EFF (r,datayear,p,cg,s)		Activity unit per flow unit [0,∞); default value: none Default i/e: STD	The group cg may be a single commodity, group, or commodity type on the shadow side, or a single commodity in the PCG; cg='ACT refers to the default shadow group. If no group officiancy is defand	Activity efficiency for process, i.e. amount of activity per unit of commodity flows in the group cg. For more information on usage, see Section	Generates instances of the activity efficiency constraint (EQE_ACTEFF)

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			shadow group is assumed to be the commodity type. Individual commodity efficiencies are multiplied with the shadow group efficiency (default=1). Direct inheritance. Weighted aggregation.	EQE_ACTEFF.	
ACT_FLO (r,datayear,p,cg,s)		Flow unit per activity unit $[0,\infty);$ default value: none Default i/e: STD	Inherited/aggregated to the timeslice levels of the process activity. Direct inheritance. Weighted aggregation.	Flow of commodities in cg in proportion to activity, in timeslice s.	Establishes a transfor- mation relationship (EQ_PTRANS) between the flows in the PCG and one or more input (or output) commodities.
ACT_LOSPL (r,datayear,p,bd)	ACT_MINLD ACT_CSTPL	Decimal fraction [0,∞); default values: FX: none LO: default value is ACT_MINLD or 0.1 if that is not defined UP: 0.6 Default i/e: STD	Endogenous partial load modeling can only be used for processes that have their efficiency modelled by the ACT_CFF parameter. For other processes, the ACT_CSTPL parameter can be used for modeling a cost penalty at partial loads.	Partial load efficiency parameters. 1) (bd='FX'): Proportional increase in specific fuel con- sumption at minimum operating level 2) (bd='LO'): Minimum operating level of partial load operation 3) (bd='UP'): Fraction of feasible load range above the minimum operating level, below which the efficiency losses are assumed to occur	Generates instances of the partial load efficiency constraint EQ_ACTPL.
ACT_MINLD	ACT_UPS	Decimal fraction	Can only be used for standard processes (not	Assumed to occur Minimum stable	Generates instances of

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
	ACT_CSTPL ACT_LOSPL	default value: none Default i/e: STD	IRE or STG). Must be defined if ACT_CSTUP or ACT_TIME is specified.	dispatchable process.	EQ_CAPLOAD and EQE_ACTUPS.
ACT_TIME (r,datayear,p,lim)	ACT_MINLD ACT_CSTUP ACT_UPS	Hours $[0,\infty)$; default value: none Default i/e: STD	Requires that start-up costs have been modeled for the process, using both ACT_MINLD and ACT_CSTUP at the DAYNITE/WEEK/Y level. The lim type 'FX' is not supported and is ignored.	 Minimum online (UP) / offline (LO) hours of a process with start-up costs modeled (lim=LO/UP) Maximum number of start-up cycles within the process timeslice cycles (lim=N). 	Generates instances of EQL_ACTUPC
ACT_UPS (r,datayear,p,s,bd)	ACT_MINLD ACT_CSTUP ACT_CSTPL ACT_LOSPL	Decimal fraction [0,∞); default value: none Default i/e: STD	Inherited/aggregated to the timeslice levels of the process activity. Direct inheritance. Weighted aggregation. The ramp rates can only be specified with bd=LO/UP.	Maximum ramp-rate (down/up) of process activity as a fraction of nominal on-line capacity per hour.	Generates instances of equation EQ_ACTRAMP.
B (t)	M, D, E, COEF_CPT, rtp_vintyr			Beginning year of period t.	
CAP_BND (r,datayear,p,bd)	PAR_CAPLO, PAR_CAPUP	Capacity unit [0,∞); default value: none Default i/e: MIG	Since inter-/extrapolation is default is MIG, a bound must be specified for each period desired, if no explicit inter-/extrapolation option is given.	Bound on investment in new capacity.	Imposes an indirect limit on the capacity transfer equation (EQ_CPT) by means of a direct bound on the capacity variable (VAR_CAP).
CM_CONST (item)		Constant specific unit [open]; default value: See Appendix	See Appendix on Climate Module for details.	Various climate module constants, e.g. phi and sigma values between	EQ_CLITOT EQ_CLICONC EQ_CLITEMP EQ_CLIBEOH

	1				
Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
		Default i/e: N/A		reservoirs.	
CM_EXOFORC (year)		Forcing unit [open]; default value: none Default i/e: STD	Default values are provided. See Appendix on Climate Module for details.	Radiative forcing from exogenous sources	EQ_CLITOT
CM_GHGMAP (r,c,cm_var)		Units of climate module emissions per units of regional emissions $[0, \infty);$ default value: none	The global emissions in the climate module (cm_var) are 'CO2-GtC' (GtC), 'CH4- Mt' (Mt) and 'N2O-Mt' (Mt). See Appendix on Climate Module for details.	Mapping and conversion of regional GHG emissions to global emissions in the climate module	EQ_CLITOT
CM_HISTORY (year,item)		Climate variable unit [0, ∞); default value: none Default i/e: STD	Default values are provided until 2010. See Appendix on Climate Module for details.	Calibration values for CO2 and forcing	EQ_CLITOT EQ_CLICONC EQ_CLITEMP EQ_CLIBEOH
CM_LINFOR (datayear,item,lim)		Forcing unit per concentration unit [open]; default value: none Default i/e: STD	With lim types LO/UP, CO2 forcing function can be automatically linearized between the concentration levels given. For CH4 and N2O, lim types FX/N must be used (N=concentration multiplier, FX=constant term). See Appendix on Climate Module for details.	Parameters of linearized forcing functions	EQ_CLITOT
CM_MAXC (datayear,item)		Climate variable unit [0, ∞); default value: none Default i/e: none	Since no default inter- /extrapolation, bounds must be explicitly specified for each desired year, unless an explicit inter- /extrapolation option is set. See Appendix on Climate Module for details.	Maximum level of climate variable	EQ_CLIMAX

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
COM_AGG (r,dayayear,c1,c2)		Commodity units [open]; default value: none Default i/e: STD	When commodity type is LO, VAR_COMNET of c1 is aggregated to c2; When commodity type is FX/N, VAR_COMPRD of c1 is aggregated to c2.	Aggregation of commodity NET/PRD production to the production side of the balance of another commodity.	Adds a term in EQ(I)_COMBAL and EQ(I)_COMPRD.
COM_BNDNET (r,datayear,c,s,bd)	rhs_combal, rcs_combal	Commodity unit [open]; default value: none Default i/e: MIG	Since inter-/extrapolation default is MIG, a bound must be specified for each period desired, if no explicit inter-/extrapolation option is given. If the bound is specified for a timeslice s above the commodity timeslice resolution (com_tsl), the bound is applied to the sum of the net commodity variables (VAR_COMNET) below it, according to the timeslice tree. Standard aggregation.	Limit on the net amount of a commodity within a region for a particular timeslice.	The balance constraint is set to an equality (EQE_COMBAL). Either the finer timeslice variables are summed (EQ(I)_BNDNET) or the bound applied direct to the commodity net variable(VAR_COMNET) when at the commodity level (com_tsl).
COM_BNDPRD (r,datayear,c,s,bd)	rhs_comprd, rcs_comprd	Commodity unit [0,∞); default value: none Default i/e: MIG	Since inter-/extrapolation default is MIG, a bound must be specified for each period desired, if no explicit inter-/extrapolation option is given. If the bound is specified for a timeslice s being above the commodity timeslice resolution (com_tsl), the bound is applied to the sum of the commodity production variables	Limit on the amount of a commodity produced within a region for a particular timeslice.	The balance constraint is set to an equality (EQE_COMBAL). Finer timeslice variables summed (EQ(1)_BNDPRD). or the bound is applied direct to the commodity production variable (VAR_COMPRD) when at the commodity level (com_tsi).

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			(VAR_COMPRD) below it, according to the timeslice tree. Standard aggregation.		
COM_BPRICE (r,t,c,s,cur)	COM_ELAST, COM_STEP, COM_VOC	Monetary unit per commodity unit [open]; default value: none Default i/e: none	The control parameter \$SET TIMESED YES' to activate elastic demands must be set.	Base price of a demand commodity for the elastic demand formulation.	Controls the inclusion of the elastic demand variable (VAR_ELAST) in the commodity balance equation(EQ(I)_COMBA L) Applied to the elastic demand variable (VAR_ELAST) in the objective function (EQ_OBJELS).
COM_CSTNET (r,datayear,c,s,cur)	OBJ_COMNT, CST_COMC, CST_PVC, rhs_combal, rcs_combal	Monetary unit per commodity unit [open]; default value: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Cost on the net amount of a commodity within a region for a particular timeslice.	Forces the net commodity variable (VAR_COMNET) to be included in the equality balance constraint (EQE_COMBAL). Applied to said variable in the cost component of the objective function (EQ_OBJVAR).
COM_CSTPRD (r,datayear,c,s,cur)	OBJ_COMPD, CST_COMC, CST_PVC, rhs_comprd, rcs_comprd	Monetary unit per commodity unit [open]; default value: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Cost on the production of a commodity, within a region for a particular timeslice.	Forces the commodity production variable (VAR_COMPRD) to be included in the equality balance constraint (EQE_COMBAL). Applied to said variable in the cost component of the objective

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
COM_CUMNET (r,y1,y2,bd)	bohyear, eohyear, rhs_combal, rcs_combal, rtc_cumnet	Commodity unit [0,∞); default value: none Default i/e: not possible	The years y1 and y2 may be any years of the set allyear; where y1 may also be 'BOH' for first year of first period and y2 may be 'EOH' for last year of last period.	Bound on the cumulative net amount of a commodity between the years y1 and y2, within a region for a particular timeslice.	function (EQ_OBJVAR). Forces the net commodity variable (VAR_COMNET) to be included in the equality balance constraint (EQE_COMBAL). Generates the cumulative commodity constraint (EO(I)_CUMNET).
COM_CUMPRD (r,y1,y2,bd)	bohyear, eohyear, rhs_comprd, rcs_comprd, rtc_cumprd	Commodity unit [0,∞); default value: none Default i/e: not possible	The years y1 and y2 may be any years of the set allyear; where y1 may also be 'BOH' for first year of first period and y2 may be 'EOH' for last year of last period.	Bound on the cumulative production of a commodity between the years y1 and y2 within a region for a particular timeslice.	Forces the net commodity variable (VAR_COMPRD) to be included in the balance equation (EQE_COMBAL). The cumulative constraint is generated (EQ(I)_CUMPRD).
COM_ELAST (r,datayear,c,s,bd)	COM_BPRICE, COM_STEP, COM_VOC	Dimensionless [open]; default value: none Default i/e: STD	The control parameter \$SET TIMESED YES' to activate elastic demands must be set. An elasticity is required for each direction the demand is permitted to move. The index bd = 'LO' corresponds to the direction of decreasing the demand, while bd = 'LP' denotes the direction for demand increase. A different value may be provided for each direction,	Elasticity of demand indicating how much the demand rises/falls in response to a unit change in the marginal cost of meeting a demand that is elastic.	Controls the inclusion of the elastic demand variable (VAR_ELAST) in the commodity balance equation(EQ(I)_COMBA L) Applied to the elastic demand variable (VAR_ELAST) in the objective function costs (EQ_OBJELS).

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			thus curves may be asymmetric.		
COM_ELASTX (r,datayear,c,bd)	COM_ELAST	Integer scalar [1,999]; default value: none Default extrapolation: MIG	Provided when shaping of elasticity based upon demand level is desired. Note: Shape index 1 is reserved for constant 1.	Shape index for the elasticity of demand	Affects the demand elasticities applied in EQ_OBJELS
COM_FR (r,datayear,c,s)	COM_PROJ, com_ts, com_tsl, RTCS_TSFR	Decimal fraction [0,1]; default value: timeslice duration (G_YRFR) Default i/e: STD	Only applicable to demand commodities (com_type = 'DEM'). Affects timeslice resolution at which a commodity is tracked (RTCS_TSFR), and thereby may affect when a process cannot operate (rtps_off). Weighted inheritance. Weighted aggregation.	Fraction of the annual demand (COM_PROJ) occurring in timeslice s; describes the shape of the load curve.	Applied to the annual demand (COM_PROJ) as the RHS of the balance equation (EQ(I)_COMBAL). Enters the peaking equation (EQ_PEAK), if a peaking commodity. Applied when setting the upper bound of an elastic demand step (VAR_ELAST).
COM_IE (r,datayear,c,s)		Decimal fraction (0,∞); default value: 1 Default i/e: STD	Direct inheritance. Weighted aggregation.	Infrastructure or transmission efficiency of a commodity.	Overall efficiency applied to the total production of a commodity in the commodity balance equation (EQ(I)_COMBAL).
COM_PKFLX (r,datayear,c,s)	com_peak, com_pkts, COM_PKRSV, FLO_PKCOI	Scalar [open]; default value: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Difference between the average demand and the peak demand in timeslice s, expressed as fraction of the average demand.	Applied to the total consumption of a commodity to raise the capacity needed to satisfy the peaking constraint (EQ_PEAK).
COM_PKRSV (r,datayear,c)	com_peak, com_pkts,	Scalar [0,∞);		Peak reserve margin as fraction of peak	Applied to the total consumption of a

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
	COM_PKFLX, FLO_PKCOI	default value: none Default i/e: STD		demand, e.g. if COM_PKRSV = 0.2, the total installed capacity must exceed the peak load by 20%.	commodity to raise the capacity needed to satisfy the peaking constraint (EQ_PEAK).
COM_PROJ (r,datayear,c)	COM_FR	Commodity unit [0,∞); default value: none Default i/e: STD	Only applicable to demand commodities (com_type = 'DEM').	Projected annual demand for a commodity.	Serves as the RHS (after COM_FR applied) of the commodity balance constraint (EQ(I)_COMBAL). Enters the peaking equation (EQ_PEAK), if a peaking commodity. Applied when setting the upper bound of an elastic demand step (VAR_ELAST).
COM_STEP (r,c,bd)	COM_BPRICE, COM_ELAST, COM_VOC, rcj	Integer number [1,∞); default value: none	The control parameter \$SET TIMESED YES' to activate elastic demands must be set. The number of steps is required for each direction the demand is permitted to move. The index bd = LO corresponds to the direction of decreasing the demand, while bd = UP denotes the direction for demand increase. A different value may be provided for each direction, thus curves may be asymmetric.	Number of steps to use for the approximation of change of producer/consumer surplus when using the elastic demand formulation.	Controls the instance of the elastic demand variable (VAR_ELAST) in: the commodity balance equation (EQ(I)_COMBAL); setting of the step limit for the elastic demand variable (VAR_ELAST); enters the objective function costs (EQ_OBJELS).

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
COM_SUBNET (r,datayear,c,s,cur)	OBJ_COMT, CST_COMX, CST_PVC, rhs_combal, rcs_combal	Monetary unit per commodity unit [0,∞); default value: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Subsidy on the net amount of a commodity within a region for a particular timeslice.	Forces the net commodity variable (VAR_COMNET) to be included in the equality balance constraint (EQE_COMBAL). Applied (-) to said variable in the cost component of the objective function (EQ_OBJVAR).
COM_SUBPRD (r,datayear,c,s,cur)	OBJ_COMPD, CST_COMX, CST_PVC, rhs_comprd, rcs_comprd	Monetary unit per commodity unit $[0,\infty)$; default value: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Subsidy on the production of a commodity within a region for a particular timeslice.	Forces the commodity production variable (VAR_COMPRD) to be included in the equality balance constraint (EQE_COMBAL). Applied (-) to said variable in the cost component of the objective function (EQ_OBJVAR).
COM_TAXNET (r,datayear,c,s,cur)	OBJ_COMNT, CST_COMX, CST_PVC, rhs_combal, rcs_combal	Monetary unit per commodity unit [0,∞); default value: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Tax on the net amount of a commodity within a region for a particular timeslice.	Forces the net commodity variable (VAR_COMNET) to be included in the equality balance constraint (EQE_COMBAL). Applied to said variable in the cost component of the objective function (EQ_OBJVAR).
COM_TAXPRD (r,datayear,c,s,cur)	OBJ_COMPD, CST_COMX, CST_PVC,	Monetary unit per commodity unit [0,∞);	Direct inheritance. Weighted aggregation.	Tax on the production of a commodity within a region for a	Forces the commodity production variable (VAR_COMPRD) to be

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
	rhs_comprd, rcs_comprd	default value: none Default i/e: STD		particular timeslice.	included in the equality balance constraint (EQE_COMBAL). Applied to said variable in the cost component of the objective function (EQ_OBJVAR).
COM_VOC (r,datayear,c,bd)	COM_BPRICE, COM_STEP, COM_ELAST	Dimensionless [0,∞); default: none Default i/e: STD	The control parameter \$SET TIMESED 'YES' to activate elastic demands must be set. A number is required for each direction the demand is permitted to move. The index bd = LO corresponds to the direction of decreasing the demotes the direction for demand increase. A different value may be provided for each direction, thus curves may be asymmetric.	Possible variation of demand in both directions when using the elastic demand formulation.	Applied when setting the bound of an elastic demand step (VAR_ELAST). Applied to the elasticity variable in the objective function cost (EQ_OBJELS).
DAM_BQTY (r,c)	DAM_COST	Commodity unit [0,∞); default value: none Default i/e: N/A	Only effective when DAM_COST has been defined for commodity c.	Base quantity of emissions for damage cost accounting	EQ_DAMAGE EQ_OBJDAM
DAM_COST (r,datayear,c,cur)	DAM_BQTY	Monetary unit per commodity unit [0,∞); default value: none Default i/e: STD	Damage costs are by default endogenous (included in the objective). To set them exogenous, use \$SET DAMAGE NO	Marginal damage cost of emissions at Base quantity.	EQ_DAMAGE EQ_OBJDAM
DAM_ELAST (r,c,lim)	DAM_COST DAM_BQTY	Dimensionless [0,∞);	Only effective when DAM_COST has been	Elasticity of damage cost in the lower or	EQ_OBJDAM

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
		default value: none Default i/e: N/A	defined for commodity c.	upper direction from Base quantity.	
DAM_STEP (r,c,lim)	DAM_COST DAM_BQTY	Integer number [1,∞); default value: none Default i/e: N/A	Only effective when DAM_COST has been defined for commodity c.	Number of steps for linearizing damage costs in the lower or upper direction from Base quantity.	EQ_DAMAGE EQ_OBJDAM
DAM_VOC (r,c,lim)	DAM_COST DAM_BQTY	Decimal fraction LO: [0,1]; UP: [0,∞); default value: none Default i/e: N/A	Only effective when DAM_COST has been defined for commodity c.	Variance of emissions in the lower or upper direction from Base quantity as a fraction of Base quantity.	EQ_OBJDAM
E (t)	B, D, M, COEF_CPT, rtp_vintyr		For each modelyear period	End year of period t, used in determining the length of each period	The amount of new investment (VAR_NCAP) carried over in the capacity transfer constraint (EQ(I)_CPT). Amount of investments (VAR_NCAP) remaining past the modelling horizon that needs to be credited back to the objective function (EQ_OBJINV).
FLO_BND (r,datayear,p,cg,s,bd)		Commodity unit [0,∞); default: none Default i/e: MIG	If the bound is specified for a timeslice s being above the flow timeslice resolution (rtpcs_varf), the bound is applied to the sum of the flow variables (VAR_FLO) according to the timeslice tree, otherwise directly to the flow variable.	Bound on the flow of a commodity or the sum of flows within a commodity group.	Flow activity limit constraint (EQ(I)_FLOBND) when s is above rtpcs_varf Direct bound on activity variable (VAR_FLO) when at the rtpcs_varf level.

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			No aggregation. ²⁹		
FLO_COST (r,datayear,p,c,s,cur)	OBJ_FCOST, CST_FLOC, CST_PVP	Monetary unit per commodity unit [open]; default: none Default i/e: STD	Direct inheritance Weighted aggregation	Variable cost of a process associated with the production/ consumption of a commodity.	Applied to the flow variable (VAR_FLO) when entering the objective function (EQ_OBJVAR). May appear in user constraints (EQ_UC*) if specified in UC_NAME.
FLO_CUM (r,p,c,y1,y2,bd)	ACT_CUM	Flow unit [0,∞); default value: none Default i/e: N/A	The years y1 and y2 may be any years of the set allyear; where y1 may also be 'BOH' for first year of first period and y2 may be 'EOH' for last year of last period.	Bound on the cumulative amount of annual process activity between the years y1 and y2, within a region.	Generates an instance of the cumulative constraint (EQ_CUMFLO)
FLO_DELIV (r,datayear,p,c,s,cur)	OBJ_FDELV, CST_FLOC, CST_PVP	Monetary unit per commodity unit [open]; default: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Cost of a delivering (consuming) a commodity to a process.	Applied to the flow variable (VAR_FLO) when entering the objective function (EQ_OBJVAR). May appear in user constraints (EQ_UC*) if specified in UC_NAME.
FLO_EFF (r,datayear,p,cg,c,s)		Commodity unit of c / commodity unit of cg [open]; default value: none Default i/e: STD	Inherited/aggregated to the timeslice levels of the flow variables of the commodities in group cg. All parameters with the same process (p) and target commodity (c) are combined in the same	Defines the amount of commodity flow of commodity (c) per unit of other process flow(s) or activity (cg).	Generates process transformation equation (EQ_PTRANS) between one or more input (or output) commodities and one output (or input) commodities.

²⁹ Standard aggregation not implemented for FLO_BND.

Input parameter	Related sets /	Units / Ranges & Default values &	Instances ²⁶ (Required / Omit / Special	Description	Affected equations or variables ²⁷
(Indexes) ²³	parameters24	Default inter- /extrapolation ²⁵	conditions)		
			transformation equation.		
FLO_EMIS (r,datayear,p,cg,com,s)	FLO_EFF (alias)	Commodity unit of c / commodity unit of cg [open]; default value: none Default i/e: STD	See FLO_EFF. If com is of type ENV and is not in the process topology, it is added to it as an output flow.	Defines the amount of emissions (c) per unit of process flow(s) or activity (cg).	See FLO_EFF.
FLO_FR (r,datayear,p,c,s,bd)		Decimal fraction [0,1] / [0,∞); default value: none Default i/e: MIG	FLO_FR may be specified as lower, upper or fixed bounds, in contrast to COM_FR. Can be specified for any flow variable having a subannual timeslice resolution. Weighted aggregation. Direct inheritance, if defined at the ANNUAL level.	 Bounds the flow of commodity (c) entering or leaving process (p) in a timeslice, in propor- tion to annual flow. If specified also at the ANNUAL level, bounds the flow level in proportion to the average level under the parent timeslice 	A share equation (EQ(I)_FLOFR) limiting the amount of commodity (c) is generated according to the bound type (bd = I indicator).
FLO_FUNC (r,datayear,p,cg1,cg2,s)	FLO_SUM, FLO_FUNCX, COEF_PTRAN, rpc_ffunc, rpcg_ptran	Commodity unit of cg2/commodity unit of cg1 [open]; default value: see next column Default i/e: STD	If for the same indexes the parameter FLO_SUM is specified but no FLO_FUNC, the FLO_FUNC is set to 1. Important factor in determining the level at which a process operates in that the derived transformation parameter (COEF_PTRAN) is inherited/aggregated to the timeslice levels of the flow variables associated with the commodities in the group cg1.	A key parameter describing the basic operation of or within a process. Sets the ratio between the sum of flows in commodity group cg2 to the sum of flows in commodity group cg1, thereby defining the efficiency of producing cg2 from cg1 (subject to any FLO_SUM). cg1 and cg2 may be also single commodities.	Establishes the basic transformation relationship (EQ_PTRANS) between one or more input (or output) commodities and one or more output (or input) commodities. Establishes the relationship between storage charging / discharging and a related commodity flow (VAR_FLO) in the auxiliary storage flow equation (EQ_STGAUX).
FLO_FUNCX	FLO_FUNC,	Integer scalar	Provided when shaping	Age-based shaping	Applied to the flow

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
(r,datayear,p,cg1,cg2)	FLO_SUM, COEF_PTRAN	[1,999]; default value: none Default extrapolation: MIG	based upon age is desired. Vintaged processes only. Note: Shape index 1 is reserved for constant 1. ACT_EFF(g): cg1=cg, cg2='ACT' ACT_FLO(g): cg1=ACT', cg2=cg FLO_EMS(cg,c): cg1=cg2=c FLO_EFF(cg,c): cg1=cg2=c FLO_EFF(cg,c): cg1=cg2=c	curve (SHAPE) to be applied to the flow parameters (ACT_EFF/ACT_FLO/ FLO_FUNC/FLO_SUM/ FLO_EMIS/FLO_EFF)	variable (VAR_FLO) in a transformation equation (EQ_PTRANS / EQE_ACTEFF) to account for changes in the transformation efficiency according to the age of each process vintage.
FLO_MARK (r,datayear,p,c,bd)	PRC_MARK	Decimal fraction [0,1]; default value: none Default i/e: STD	The same given fraction is applied to all timesilces of the commodity (this could be generalized to allow time-slice-specific fractions, if deemed useful). If an ANNUAL level market- share is desired for a timesliced commodity, PRC_MARK can be used instead.	Process-wise market share in total commodity production.	The individual process flow variables (VAR_FLO, VAR_IN, VAR_STGIN/OUT) are constrained (EQ(I)_FLMKK) to a fraction of the total production of a commodity (VAR_COMPRD). Forces the commodity production variable (VAR_COMPRD) to be included in the equality balance constraint (EOE COMPRL).
FLO_PKCOI (r,datayear,p,c,s)	COM_PKRSV, COM_PKFLX, com_peak, com_pkts	Scalar [open]; default value: 1 Default i/e: STD	FLO_PKCOI is specified for individual processes p consuming the peak commodity c. Direct inheritance. Weighted aggregation. Used when the timeslices are not necessarily fine	Factor that permits attributing more (or less) demand to the peaking equation (EQ_PEAK) than the average demand calculated by the model, to handle the situation where peak	Applied to the flow variable (VAR_FLO) to adjust the amount of a commodity consumed when considering the average demand contributing to the peaking constraint (EO_PEAK).

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			enough to pick up the actual peak within the peak timeslices.	usage is typically higher (or lower) due to coincidental (or non-coincidental) loads at the time of the peak demand.	
FLO_SHAR (r,datayear,p,c,cg,s,bd)		Decimal fraction [0,1]; default value: none Default i/e: MIG over milestoneyears, STD over pastyears	Direct inheritance. Weighted aggregation. A common example of using FLO_SHAR is to specify the power-to-heat ratio of CHP plants in the backpressure point. For example, for a heat output of a CHP technology, the FLO_SHAR parameter would have the value CHPR/(1-LHPR), with CHPR/being the heat-to- power ratio.	Share of flow commodity c based upon the sum of individual flows defined by the commodity group cg belonging to process p.	When the commodity is an input an EQ(I)_INSHR equation is generated. When the commodity is an output an $EQ(I)_OUTSHR equation$ is generated.
FLO_SUB (r,datayear,p,c,s,cur)	OBJ_FSUB, CST_FLOX, CST_PVP	Monetary unit per commodity unit $[0,\infty)$; default value: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Subsidy on a process flow.	Applied with a minus sign to the flow variable (VAR_FLO) when entering the objective function (EQ_OBJVAR). May appear in user constraints (EQ_UC*) if specified in UC_NAME.
FLO_SUM (r,datayear,p,cg1,c,cg2, s)	FLO_FUNC FLO_FUNCX COEF_PTRANS, fs_emis, rpc_emis, rpc_ffunc, rpcg_ptran	Commodity unit of cg2/commodity unit of c [open]; default value: see next column Default i/e: STD	If a FLO_SUM is specified and no corresponding FLO_FUNC, the FLO_FUNC is set to 1. If FLO_FUNC is specified for a true commodity group cg1, and no FLO_SUM is	Multiplier applied for commodity c of group cg1 corresponding to the flow rate based upon the sum of individual flows defined by the	The FLO_SUM multiplier is applied along with FLO_FUNC parameter in the transformation coefficient (COEF_PTRANS), which is applied to the flow

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			specified for the commodities in cg1, these FLO_SUM are set to 1. The derived parameter COEF_PTRANS is inherited/aggregated to the timeslice level of the flow variable of the commodity c.	commodity group cg2 of process p. Most often used to define the emission rate, or to adjust the overall efficiency of a technology based upon fuel consumed.	variable (VAR_ELO) in the transformation equation (EQ_PTRANS).
FLO_TAX (r,datayear,p,c,s,cur)	OBJ_FTAX, CST_FLOX, CST_PVP	Monetary unit per commodity unit $[0,\infty)$; default: none Default i/e: STD	Direct inheritance. Weighted aggregation.	Tax on a process flow.	Applied to the flow variable (VAR_FLO) when entering the objective function (EQ_OBJVAR). May appear in user constraints (EQ_UC*) if specified in UC_NAME.
G_CUREX (cur1,cur2)	R_CUREX	Scalar (0,∞) Default value: none	The target currency cur2 must have a discount rate defined with G_DRATE.	Conversion factor from currency cur1 to currency cur2, with cur2 to be used in the objective function.	Affects cost coefficients in EQ_OBJ
G_DRATE (r,allyear,cur)	OBJ_DISC, OBJ_DCEOH, NCAP_DRATE, COR_SALVI, COR_SALVD, COEF_PVT VDA_DISC	Decimal fraction (0,1]; default value = none Default i/e: STD	A value must be provided for each region. Interpolation is dense (all individual years included).	System-wide discount rate in region r for each time-period.	The discount rate is taken into consideration when constructing the objective function discounting multiplier (OBJ_DISC), which is applied in each components of the objective function (EQ_OBJVAR, EQ_OBJFIX, EQ_OBJFIX, EQ_OBJSALV,

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
					EQ_OBJELS).
G_DYEAR	OBJ DISC COEF_PVT	Year [BOTIME,EOTIME]; default value = M(MIYR_1), i.e. the first milestone year		Base year for discounting.	The year to which all costs are to be discounted is taken into constructing the objective function discounting multiplier (OBJ_DISC), which is applied in each of the components of the objective function (EQ_OBJINV, EQ_OBJINV, EQ_OBJELS).
G_ILEDNO	NCAP_ILED	Decimal fraction [0,1]; default value: 0.1	Only provided when the costs associated with the lead-time for new capacity (NCAP_ILED) are not to be included in the objective function. Not taken into account if the OBLONG switch or any alternative objective formulation is used.	If the ratio of lead- time (NCAP_ILED) to the period duration (D) is below this threshold then the lead-time consideration will be ignored in the objective function costs.	Prevents the investment costs associated with investment lead-times from energy the investment component of the objective function (EQ_OBJINV).
G_NOINTERP	All parameters that are normally subjected to interpolation / extrapolation	Binary indicator [0 or 1]; default value = 0	Only provide when interpolation / extrapolation is to be turned off for all parameters. Interpolation of cost parameters is always done.	Switch for generally turning-on (= 0) and turning-off (= 1) sparse inter- / extrapolation.	
G_OFFTHD	PRC_NOFF	Scalar	Setting G_OFFTHD=1 will	Threshold for	Affects availability of

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
(datayear)	PRC_AOFF PRC_FOFF COM_OFF	[0,1] Default value: 0 Default i/e: 5	make the *_OFF attributes effective only for periods fully included in the OFF range specified.	considering an *_OFF attribute disabling a process/commodity variable in period.	VAR_NCAP, VAR_ACT, VAR_FLO, VAR_COMNET/PRD
G_OVERLAP		Scalar [0,100] Default value: TIMESTEP/2	Used only when time- stepped solution is activated with the TIMESTEP control variable.	Overlap of stepped solutions (in years).	-
G_TLIFE	NCAP_TLIFE	Scalar $[1,\infty)$; default value = 10		Default value for the technical lifetime of a process if not provided by the user.	
G_YRFR (all_r,s)	RTCS_TSFR, RS_STGPRD	Fraction [0,1]; default value: none; only for the ANNUAL timeslice a value of 1 is predefined	Must be provided for each region and timeslice.	Duration of timeslice s as fraction of a year. Used for shaping the load curve and lining up timeslice duration for inter-regional exchanges.	Applied to various variables (VAR_NCAP+PASTI, VAR_COMX, VAR_IRE, VAR_FLO, VAR_SIN/OUT) in the commodity balance equation (EQ(I)_COMBAL).
IRE_BND (r,datayear,c,s,all_r,ie, bd)	top_ire	Commodity unit [0,∞); default value: none Default i/e: MIG	Only applicable for inter- regional exchange processes (IRE). If the bound is specified for a timeslice cost (s) being above the commodity (c) timeslice resolution, the bound is applied to the sum of the imports/exports according to the timeslice tree. Standard aggregation.	Bound on the total import (export) of commodity (c) from (to) region all_r in (out of) region r.	Controls the instances for which the trade bound constraint (EQ(I)_IREBND) is generated, and the RHS.
IRE_CCVT	IRE_TSCVT,	Scalar	Required for mapping	Conversion factor	The conversion factor is
(r1,C1,r2,C2))	top ire	[(U,∞)]	commodities involved in	petween commodity	applied to the flow

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
		Default value: 1 if commodity names are the same in both regions I/e: N/A	inter-regional exchanges between two regions whenever commodities traded are in different units in the regions.	units in region r1 and region r2. Expresses the amount of commodity c2 in region r2 equivalent to 1 unit of commodity c1 in region r1.	variable (VAR_IRE) in the inter-regional balance constraint (EQ_IRE). Similarly, applied to the flow variable (VAR_IRE) when an inter-regional exchange is bounded in the limit constraint (EQ(I)_IREBND). Similarly, applied to the flow variable (VAR_IRE) when an exchange with an external region is bounded (EQ(I) XBND).
IRE_FLO (r1,datayear,p,c1,r2,c2, s2)	top_ire	Commodity unit c2/commodity unit c1 [0,∞); default value: 1 Default i/e: STD	Only applicable for inter- regional exchange processes (IRE) between two internal regions. Note that for each direction of trade a separate IRE_FLO. needs to be specified. Similar to FLO_FUNC for standard processes. Direct inheritance.	Efficiency of exchange process from commodity c1 in region r1 to commodity c2 in the region2 in timeslice s2; the timeslice s2; the timeslice refers to the r2 region.	Applied to the exchange flow variable (VAR_IRE) in the inter-regional trade equation (E_{O} _IRE). Applied to the exchange flow variable (VAR_IRE) when a bound on inter- regional trade is to be applied (EQ(I)_IREBND).
IRE_FLOSUM (r,datayear,p,c1,s,ie,c2, io)	top_ire	Commodity unit c2/commodity unit c1 [open]; default value: none Default i/e: STD	Only applicable for inter- regional exchange processes (IRE). Since the efficiency IRE_FLO can only be used for exchange between internal regions, IRE_FLOSUM may be used	Auxiliary consumption (io = IN, owing to the commodity entering the process) or production/ emission (io = OUT, owing to the commodity leaving the process)	The multiplier is applied to the flow variable (VAR_IRE) associated with an inter-reginal exchange in the commodity balance constraint (EQ(I)_COMBAL).

Tanut neversitor	Deleted esta (Units / Ranges &	Instances ²⁶	Description	Affected equations
(Induce)23	Related sets /	Default values &	(Required / Offic / Special		or variables*
(Indexes)~~	parameters**	Default Inter-			
		/extrapolation ²⁵			
			to define an efficiency for an import/export with an external region by specifying the same commodity for c1 and c2 and the value 1-efficiency as auxiliary consumption. Direct inheritance. Weighted aggregation.	or commodity c2 due to the IMPort / EXPort (index ie) of the commodity c1 in region r ³⁰	If a flow snare (FLO_SHAR) is provided for an inter-regional exchange process then the multiplier is applied to the flow variable (VAR_IRE) in the share constraint (EQ(I)_IN/OUTSHR). If a cost is provided for the flow (FLO_COST or FLO_DELIV) then the flow variable (VAR_IRE) in the variable (VAR_IRE) in the variable component of the objective function (EQ_OBJVAR).
IRE_PRICE	OBJ_IPRIC,	Monetary unit /	Only applicable for inter-	IMPort/EXPort price	The price of the
(r,datayear,p,c,s,all_r,i	CST_COMC,	commodity unit	regional exchange	(index ie) for to/from	exchange commodity is
e,cur)	CST_PVP,	[0,∞);	processes (IRE).	an internal region of a	applied to the trade
	top_ire	default value: none	Ignored if all_r is an	commodity (c)	flow variable (VAR_IRE)
		Default i/e: STD	internal region.	originating	in the variable costs

³⁰ The indexing of auxiliary consumption flows or emissions of inter-regional exchange processes is illustrated in the figure below. Indexing of auxiliary consumption/emission



Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			Direct inheritance. Weighted aggregation.	from/heading to an external region all_r.	component of the objective function (EQ_OBJVAR).
IRE_TSCVT (r1,s1,r2,s2)	IRE_CCVT, top_ire	$\begin{array}{l} & \text{Scalar} \\ (0,\infty); \\ & \text{default value: 1 if} \\ & \text{timeslice tree and} \\ & \text{names are the same} \\ & \text{in both regions} \\ & \text{I/e: N/A} \end{array}$	Used for mapping timeslices in different regions. Required if timeslice definitions are different in the regions.	Matrix for mapping timeslices; the value for (r1,s1,r2,s2) gives the fraction of timeslice 2 in region r2 that falls in timeslice s1 in region r1.	The conversion factor is applied to the flow variable (VAR_IRE) in the inter-regional balance constraint (EQ_IRE). Similarly, applied to the flow variable (VAR_IRE) when an inter-regional exchange is bounded in the limit constraint (EQ(I)_IREND). Similarly, applied to the flow variable (VAR_IRE) when an exchange with an external region is bounded (EQ(I) XBND).
IRE_XBND (all_r,datayear,c,s ie,bd)	top_ire	Commodity unit [0,∞); default value: none Default i/e: MIG	Only applicable for inter- regional exchange processes (IRE). Provide whenever a trade flow is to be constrained. Note that the limit is either imposed by summing lower or splitting higher flow variables (VAR_IRE) when specified at other than the actual flow level (as determined by the commodity and process levels (COM_TSL/ PRC_TSL	Bound on the total IMPort (EXPort) (index ie) of commodity c in region all_r with all sources (destinations).	The trade limit equation EQ(I)_XBND generated either sums lower flow vanables (VAR_IRE) or splits (according to the timeslice tree) coarser vanables.

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
MULTI (j,aliyear)	NCAP_AFM, NCAP_FOMM, NCAP_FSUBM, NCAP_FTAXM	Scalar [open]; default value: none I/e: Full dense interpolation and extrapolation). Only provided when the related shaping parameters are to be used.	Multiplier table used for any shaping parameters (*_*M) to adjust the corresponding technical data as function of the year; the table contains different multiplier curves identified by	{See Related Parameters}
NCAP_AF (r,datayear,p,s,bd)	NCAP_AFA, NCAP_AFS, NCAP_AFM, NCAP_AFK, COEF_AF	Decimal fraction [0,1]; default value: 1 Default i/e: STD Remark: In special cases values >1 can also be used (when PRC_CAPACT does not represent the max. technical level of activity per unit of capacity).	NCAP_AF, NCAP_AFA and NCAP_AFS can be applied simultaneously. Direct inheritance. Weighted aggregation. (Important remark: No inheritance/aggregation if any value is specified at process timeslices.)	the index j. Availability factor relating a unit of production (process activity) in timeslice s to the current installed capacity.	The corresponding capacity-activity constraint (EQ(I)_CAPACT) will be generated for any timeslice s. If the process timeslice level (PRC_TSL) is below said level, the activity variables will be summed.
NCAP_AFA (r,datayear,p,bd)	NCAP_AFA, NCAP_AFS, NCAP_AFM, NCAP_AFM, COEF_AF	Decimal fraction [0,1]; default value: none Default i/e: STD Remark: In special cases values >1 can also be used (when PRC_CAPACT has been chosen not to represent the max.	Provided when 'ANNUAL' level process operation is to be controlled. NCAP_AF, NCAP_AFA and NCAP_AFS can be applied simultaneously. NCAP_AFA is always assumed to be non-virtage dependent, even if the process is defined as a vintaged one; for virtage-	Annual availability factor relating the annual activity of a process to the installed capacity.	The corresponding capacity-activity constraint (EQ(I)_CAPACT) will be generated for the 'ANNUAL' timeslice. If the process timeslice level (PRC_TSL) is below said level, the activity variables will be summed.

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- / extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
		technical level of activity per unit of capacity).	dependent annual availability NCAP_AFS with s='ANNUAL' can be used.		
NCAP_AFC (r,datayear,p,cg,tsl)	NCAP_AFCS	Decimal fraction $[0,\infty)$; default value: none Default i/e: STD	If the commodities are in the PCG constraint is applied to the flows in the PCG as a whole (linear combination of flows). Independent equations are generated for commodities not in the PCG, or when NCAP_AFC(r,'0', p,'ACT',ts) =-1 is also specified.	Commodity-specific availability of capacity for commodity group cg, at given timeslice level.	Generates instances of EQ(I)_CAFLAC (thereby disabling EQ(I)_CAPACT generation), or EQL_CAPFLO.
NCAP_AFCS (r,datayear,p,cg,ts)	NCAP_AFC	Decimal fraction [0,∞); default value: none Default i/e: STD	See NCAP_AFC. NCAP_AFCS is similar to NCAP_AFC but is defined on individual timeslices. Overrides NCAP_AFC.	Commodity-specific availability of capacity for commodity group cg, timeslice-specific.	See NCAP_AFC.
NCAP_AFM (r,datayear,p)	NCAP_AF, NCAP_AFA, NCAP_AFS, MULTI, COEF_AF	Integer number Default value: 0 (no multiplier applied) Default extrapolation: MIG	Provided when multiplication of NCAP_AF / NCAP_AFS based upon year is desired. Note: Multiplier index 1 is reserved for constant 1.	Period sensitive multiplier curve (MULTI) to be applied to the availability factor parameters (NCAP_AF/AFA/AFS) of a process.	{See Related Parameters}
NCAP_AFS (r,datayear,p,s,bd)		Decimal fraction [0,1]; default value: none Default i/e: STD Remark: In special cases values >1 can also be used (in cases where PRC_CAPACT has been chosen not	NCAP_AF, NCAP_AFA and NCAP_AFS can be applied simultaneously. NCAP_AFS being specified for timeslices s being below the process timeslice level are ignored. No inheritance. No aggregation.	Availability factor relating the activity of a process in a timeslice s being at or above the process timeslice level (prc_tsl) to the installed capacity. If for example the process timeslice	The corresponding capacity-activity constraint (EQ(I)_CAPACT) will be generated for a timeslice s being at or above the process timeslice level (prc_tsl). If the process timeslice level is below said level,

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
		to represent the maximum technical level of activity per unit of capacity).	Can be used also on the process timeslices, and will then override the levelized NCAP_AF availability factors.	level is 'DAYNITE' and NCAP_AFS is specified for timeslices on the 'SEASONAL' level, the sum of the 'DAYNITE' activities within a season are restricted, but not the 'DAYNITE' activities directly.	the activity variables will be summed.
NCAP_AFX (r,datayear,p)	NCAP_AF, NCAP_AFA, NCAP_AFS, SHAPE, COEF_AF	Integer number Default value: 0 (no Shape curve applied) Default extrapolation: MIG	Provided when shaping based upon age is desired. NCAP_AFX is applied to NCAP_AF and NCAP_AFS, but not the annual availability NCAP_AFA. For non-vintaged process, the SHAPE parameter is only applied to NCAP_AF, i.e. availabilities at process timeslices will be vintaged. Note: Shape index 1 is reserved for constant 1.	Age-based shaping curve (SHAPE) to be applied to the availability factor parameters (NCAP_AF/AFA/AFS) of a process.	{See Related Parameters}
NCAP_BND (r,datayear,p,bd)		Capacity unit [0,∞); default value: none Default i/e: MIG	Provided for each process to have its overall installed capacity (VAR_NCAP) limited in a period. Since inter-/extrapolation default is MIG, a bound must be specified for each period desired, if no explicit inter-/extrapolation option is given, e.g. NCAP_BND(R,'0',P) = 2.	Bound on the permitted level on investment in new capacity	Imposes an indirect limit on the capacity transfer equation (EQ_CPT) by means of a direct bound on the new investments capacity variable (VAR_NCAP).
NCAP_BPME (r,datayear,p)	NCAP_CDME	Decimal fraction [0,∞);	The parameter is only taken into account when	Back pressure mode efficiency (or total	Process transformation equation, either

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
		default value: none Default i/e: STD	the process is of type CHP, and NCAP_CDME has been also defined.	mode).	EQE_ACTEFF or EQ_PTRANS
NCAP_CDME (r,datayear,p)	NCAP_BPME	Decimal fraction [0,∞); default value: none Default i/e: STD	The parameter can only be used for standard process- es having electricity output in the PCG. The efficiency is applied between the default shadow group and the electricity. If the process is also defined as a CHP, heat efficiency is also included.	Condensing mode efficiency	Process transformation equation, either EQE_ACTEFF or EQ_PTRANS
NCAP_CEH (r,datayear,p)	NCAP_CHPR ACT_EFF	Decimal fraction [−1,∞]; default value: none Default i/e: STD	The parameter is only taken into account when the process is defined to be of type CHP. According to the CEH value, the process activity will be defined as: CEH ≤ 0 : Max. electricity output according to CHPR $0 < CEH \leq 1$: Condensing mode electricity output $CEH \leq 1$: Total energy output in [II] CHP mode.	Coefficient of electricity to heat along the iso-fuel line in a pass-out CHP technology.	Process transformation equation, either EQE_ACTEFF or EQ_PTRANS
NCAP_CHPR (r,datayear,p,bd)	FLO_SHAR	Decimal fraction [0,∞); default value: none Default i/e: STD	The parameter is only taken into account when the process is defined to be of type CHP.	Heat-to-power ratio of a CHP technology (fixed / minimum / maximum ratio).	The ratios are implemented with EQ(I)_OUTSHR
NCAP_CLAG (r,datayear,p,c,io)	NCAP_CLED NCAP_COM	Years [open]; default value: none Default i/e: STD	Provided when there is a delay in commodity output after commissioning new capacity. So, if the process is available in the year K, the commodity is produced	Lagtime of a commodity after new capacity is installed.	Applied to the investment variable (VAR_NCAP) in the commodity balance (EQ(I)_COMBAL) of the investment period or

	1				
Tenut parameter	Delated sate /	Units / Ranges &	(Beguired / Omit / Special	Description	Affected equations
(Indexes)23	Related Sets /	Default inter	(Required / Onlic / Special		
(Indexes) 20	parameters-	Default Inter-	conditionsy		
		/extrapolation	during the surgery IVA CLAC		
			K+NCAP_TLIFE-1].		previous periods.
NCAP_CLED (r,datayear,p,c)	NCAP_ICOM COEF_ICOM	Years [open]: default value: = NCAP_ILED Default i/e: STD	Provided when a commodity must be available prior to available prior to available prior to available inthe year B(v) +NCAP_ILED-1, the commodity is produced during the time span [B(v)+ILED-CLED, B(v) +NCAP_ILED-1]. Usually used when modelling the need for fabrication of reactor fuel the period before a reactor ropes online.	Lead time requirement for a commodity during construction (NCAP_ICOM), prior to the initial availability of the capacity.	Applied to the investment variable (VAR_NCAP) in the commodity balance (EQ(I)_COMBAL) of the investment period or previous periods.
NCAP_COM (r,datayear,p,c,io)	rpc_capflo, rpc_conly	Commodity unit per capacity unit [open]; default value: none Default i/e: STD	Provided when the consumption or production of a commodity is tied to the level of the installed capacity.	Emission (or land- use) of commodity c associated with the capacity of a process for each year said capacity exists.	Applied to the capacity variable (VAR_CAP) in the commodity balance (EQ_COMBAL).
NCAP_COST (r,datayear,p)	OBJ_ICOST, OBJSCC, CST_INVC, CST_PVP	Monetary unit per capacity unit $[0,\infty)$; default value: none Default i/e: STD	Provided whenever there is a cost associated with putting new capacity in place.	Investment costs of new installed capacity according to the installation year.	Applied to the investment variable (VAR_NCAP) when entering the objective function (EQ_OBJNV). May appear in user constraints (EQ_UC*) if specified in UC_NAME.
NCAP_DCOST	NCAP_DLAG,	Monetary unit per	Provided when there are	Cost of dismantling a	Applied to the current
(r,datayear,p,cur)	COR_SALVD,	capacity unit	decommissioning costs	facility after the end	capacity subject to
	OBJ_DCOST,	[0,∞); default value:	associated with a process.	of its lifetime.	decommissioning

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
	CST_DECC, CST_PVP	none Default i/e: STD	Decommissioning of a process and the payment of decommissioning costs may be delayed by a lag time (NCAP_DLAG).		(VAR_NCAP+NCAP_PAS TI) when entering the objective function (EQ_OBJNV).
NCAP_DELIF (r,datayear,p)	NCAP_DLIFE, COR_SALVD, DUR_MAX, OBJ_CRFD, SALV_DEC	Years (0,∞); default value: NCAP_DLIFE Default i/e: STD	Provided when the timeframe for paying for decommission is different from that of the actual decommissioning.	Economic lifetime of the decommissioning activity.	Applied to the investment variable (VAR_NCAP) when entering the salvage portion of the objective function (EQ_OBJSALV).
NCAP_DISC (r,datayear,p,unit)	rp_dscncap	Capacity unit [0,∞); default value: none Default i/e: MIG	Used for lumpy investments. Requires MIP. Since inter-/extrapolation default is MIG, a value must be specified for each period desired, if no explicit inter-/extrapolation option is given.	Size of capacity units that can be added.	Applied to the lumpy investment integer variable (VAR_DNCAP) in the discrete investment equation (EQ_DSCNCAP) to set the corresponding standard investment variable level (VAR_NCAP).
NCAP_DLAG (r,datayear,p)	COEF_OCOM, DUR_MAX, OBJ_DLAGC	Years [0,∞); default value: none Default i/e: STD	Provided when there is a lag in the decommissioning of a process (e.g., to allow the nuclear core to reduce its radiation).	Number of years delay before decommissioning can begin after the lifetime of a technology has ended.	Delay applied to a decommissioning flow (VAR_FLO) in the balance equation (EQ(I)_COMBAL) as production. Delay applied to the current capacity subject to decommissioning (VAR_NCAP+NCAP_PAS TI) when entering the objective function components

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
					(EQ_OBJINV, EQ_OBJFIX, EQ_OBJSALV).
NCAP_DLAGC (r,datayear,p,cur)	NCAP_DLAG, OBJ_DLAGC, CST_DECC, CST_PVP	Monetary unit per capacity unit [0,∞); default value: none Default i/e: STD	Provided when there is a cost during any lag in the decommissioning (e.g., security).	Cost occurring during the lag time after the technical lifetime of a process has ended and before its decommissioning starts.	Cost during delay applied to the current capacity subject to decommissioning (VAR_NCAP+NCAP_PAS TI) when entering the objective function components (EQ_OBJFIX, EQ_OBJSALV).
NCAP_DLIFE (r,datayear,p)	DUR_MAX	Years (0,∞); default value: none Default i/e: STD	Provided when a process has a decommissioning phase.	Technical time for dismantling a facility after the end its technical lifetime, plus any lag time (NCAP_DLAG).	Decommissioning time impacting (VAR_NCAP+NCAP_PAS T1) when entering the objective function components (EQ_OBJINV, EQ_OBJSALV).
NCAP_DRATE (r,datayear,p)	G_DRATE, COR_SALVI, COR_SALVD	Percent (0,∞); default value: G_DRATE Default i/e: STD	Provided if the cost of borrowing for a process is different from the standard discount rate.	Technology specific discount rate.	Discount rate applied to investments (VAR_NCAP+NCAP_PAS TI) when entering the objective function components (EQ_OBJINV, EQ_OBJSALV).
NCAP_ELIFE (r,datayear,p)	NCAP_TLIFE, COR_SALVI, OBJ_CRF	years (0,∞); default value: NCAP_TLIFE Default i/e: STD	Provided only when the economic lifetime differs from the technical lifetime (NCAP_TLIFE).	Economic lifetime of a process.	Economic lifetime of a process when costing investment (VAR_NCAP+NCAP_PAS TI) or capacity in the objective function

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
					components (EQ_OBJINV, EQ_OBJSALV, EQ_OBJFIX).
NCAP_FOM (r,datayear,p,cur)	OBJ_FOM, CST_FIXC, CST_PVP	Monetary unit per capacity unit $[0,\infty)$; default value: none Default i/e: STD	Provided when there is a fixed cost associated with the installed capacity.	Fixed operating and maintenance cost per unit of capacity according to the installation year.	Fixed operating and maintenance costs associated with total installed capacity (VAR_NCAP+NCAP_PAS TI) when entering the objective function components (EQ_OBJTA).
NCAP_FOMM (r,datayear,p)	NCAP_FOM, MULTI	Integer number Default value: 0 (no multiplier curve applied) Default i/e: MIG	Provided when shaping based upon the period is desired. Note: Multiplier index 1 is reserved for constant 1.	Period sensitive multiplier curve (MULTI) applied to the fixed operating and maintenance costs (NCAP_FOM).	{See Related Parameters}
NCAP_FOMX (r,datayear,p)	NCAP_FOM, SHAPE	Integer number Default value: 0 (no shape curve applied) Default i/e: MIG	Provided when shaping based upon age is desired. Note: Shape index 1 is reserved for constant 1.	Age-based shaping curve (SHAPE) to be applied to the fixed operating and maintenance cost.	{See Related Parameters}
NCAP_FSUB (r,datayear,p,cur)	OBJ_FSB, CST_FIXX, CST_PVP	Monetary unit per capacity unit $[0,\infty)$; default value: none Default i/e: STD	Provided when there is a subsidy for associated with the level of installed capacity.	Subsidy per unit of installed capacity.	Fixed subsidy associated with total installed capacity (VAR_NCAP+NCAP_PAS TI) when entering the objective function component (EQ_OBJFIX) with a minus sign.
NCAP_FSUBM (r,datayear,p)	NCAP_FSUB, MULTI	Integer number Default value: 0 (no	Provided when shaping based upon the period is	Period sensitive multiplier curve	{See Related Parameters}

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
		multiplier curve applied) Default i/e: MIG	desired. Note: Multiplier index 1 is reserved for constant 1.	(MULTI) applied to the subsidy (NCAP_FSUB).	
NCAP_FSUBX (r,datayear,p)	NCAP_FSUB, SHAPE	Integer number Default value: 0 (no shape curve applied) Default i/e: MIG	Provided when shaping based upon age is desired. Note: Shape index 1 is reserved for constant 1.	Age-based shaping curve (SHAPE) to be applied to the fixed subsidy (NCAP_FSUB).	{See Related Parameters}
NCAP_FTAX (r,datayear,p,cur)	OBJ_FTX, CST_FIXX, CST_PVP	monetary unit per capacity unit [open]; default value: none Default i/e: STD	Provided when there is a fixed tax based upon the level of the installed capacity.	Tax per unit of installed capacity.	Fixed subsidy associated with total installed capacity (VAR_NCAP+NCAP_PAS TI) when entering the objective function components (EQ_OBJFIX).
NCAP_FTAXM (r,datayear,p)	NCAP_FTAX, MULTI	Integer number Default value: 0 (no multiplier curve applied) Default i/e: MIG	Provided when shaping based upon the period is desired. Note: Multiplier index 1 is reserved for constant 1.	Period sensitive multiplier curve (MULTI) applied to the tax (NCAP_FTAX).	{See Related Parameters}
NCAP_FTAXX (r,datayear,p)	NCAP_FTAX, SHAPE	Integer number Default value: 0 (no shape curve applied) Default i/e: MIG	Provided when shaping based upon age is desired. Note: Shape index 1 is reserved for constant 1.	Age-based shaping curve (SHAPE) to be applied to the fixed tax (NCAP_FTAX).	{See Related Parameters}
NCAP_ICOM (r,datayear,p,c)	NCAP_CLED, rpc_capflo, rpc_conly	Commodity unit per capacity unit [open]; default value: none Default i/e: STD	Provided when a commodity is needed in the period in which the new capacity is to be available, or before NCAP_CLED. If NCAP_CLED is provided, the commodity is required during the years [B(v)+NCAP_CLED,B(v)+N CAP_LLED_NCAP_CLED]. If	Amount of commodity (c) required for the construction of new capacity.	Applied to the investment variable (VAR_NCAP) in the appropriate commodity constraints (EQ(I)_COMBAL) as part of consumption.

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
			this time spans more than one period, the commodity flow is split up proportion- ally between the periods. For the commodity balance the commodity requirement in a period is converted to an average annual commodity flow for the entire period, although the construction may take place only for a few years of the period. Negative value describes production (e.g. emissions) at the time of a new investment.		
NCAP_ILED (r,t,p)	NCAP_ICOM, NCAP_COST, COEF_CPT, COEF_ICOM, DUR_MAX	Yeans [open]; default value: none Default i/e: STD	Provided when there is a delay between when the investment decision occurs and when the capacity (new capacity or past investment) is initially available. If NCAP_ILED>0, the investment decision is assumed to occur at $B(v)$ and the capacity becomes available at $B(v)$ +NCAP- ILED. If NCAP_ILED<0, the investment decision is assumed to occur at $B(v)$ - NCAP_ILED and the capacity becomes available at $B(v)$. Causes an IDC overhead in the investment	Lead time between investment decision and actual availability of new capacity (= construction time).	Applied to the investment variable (VAR_NCAP) balance constraints (EQ(I)_COMBAL) as part of consumption, if there is an associated flow (NCAP_ICOM). Used as to distinguish between small and large investments (VAR_NCAP) and thus influences the way the investment and fixed costs are treated in the objective function (EQ_OBJINX,

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
NCAP_ISUB (r,datayear,p,cur)	OBJ_ISUB, OBJSCC, CST_INVX, CST_SALV, CST_PVP	monetary unit per capacity unit [0,∞); default value: none Default i/e: STD	costs accounting. Provided when there is a subsidy for new investments in a period.	Subsidy per unit of new installed capacity.	EQ OBJSALV). Applied to the investment variable (VAR_NCAP) when entering the objective function (EQ_OBJNV) with a minus sign. May appear in user constraints (EQ_UC*) if specified in UC_NAME.
NCAP_ITAX (r,datayear,p,cur)	OBJ_ITAX, OBJSCC, CST_INVX, CST_SALV, CST_PVP	monetary unit per capacity unit [0,∞); default value: none Default i/e: STD	Provided when there is a tax associated with new investments in a period.	Tax per unit of new installed capacity	Applied to the investment variable (VAR_NCAP) when entering the objective function (EQ_OBJNV). May appear in user constraints (EQ_UC*) if specified in UC_NAME.
NCAP_OCOM (r,datayear,p,c)	NCAP_VALU, rpc_capflo, rpc_conly	Commodity unit per capacity unit [open]; default value: none Default i/e: STD	Provided when there is a commodity release associated with the decommissioning. The year index of the parameter corresponds to the vintage year. If the decommissioning time (NCAP_DLIFE) falls in more than one period, is split up proportionally among the periods. For the commodity balance the commodity balance the converted to an average annual commodity flow for the entire period.	Amount of commodity c per unit of capacity released during the dismantling of a process.	Applied to the investment variable (VAR_NCAP) in the appropriate commodity constraints (EQ(I)_COMBAL) as part of production in the appropriate period.

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
			although the dismantling may take place only for a few years of the period.		
NCAP_OLIFE (r,datayear,p)	NCAP_TLIFE	Years (0,∞); default value: none Default i/e: STD	Requires that early retirements are enabled and the process is vintaged.	Maximum operating lifetime of a process, in terms of full-load years.	EQL_SCAP
NCAP_PASTI (r,pastyear,p)	NCAP_PASTI, OBJ_PASTI, PAR_PASTI, PRC_RESID	capacity unit [0,∞); default value: none No i/e	Past investment can also be specified for milestone years, e.g. if the milestone year is a historic year, so that capacity additions are known or if planned future investments are already known.	Investment in new capacity made before the beginning of the model horizon (in the year specified by pastyear).	EQ(I)_COMBAL EQ_CPT EQ_OBJINV, EQ_OBJSALV, EQ_OBJFIX
NCAP_PASTY (r,pastyear,p)	NCAP_PASTI	Years [1,999]; default value: none No i/e	Provided to spread a single past investment (NCAP_PASTI) back over several years (e.g., cars in the period before the 1 st milestoneyr were bought over the previous 15 years). If overlaps with other past investments, the capacity values are added.	Number of years to go back to calculate a linear build-up of past investments	{See NCAP_PASTI}
NCAP_PKCNT (r,datayear,p,s)	com_peak, com_pkts, prc_pkaf, prc_pkno	Decimal fraction [0,1]; default value: 1 Default i/e: STD	If the indicator PRC_PKAF is specified, the NCAP_PKCNT is set equal to the availabilities NCAP_AF. Direct inheritance. Weighted aggregation.	Fraction of capacity that can contribute to peaking equations.	Applied to investments in capacity (VAR_NCAP, NCAP_PASTI) in the peaking constraint (EQ_PEAK).
NCAP_SEMI (r,datayear,p)	NCAP_DISC	Capacity unit (0,∞);	Upper bound for the capacity must be defined	Semi-continuous new capacity, lower	Applied to the semi- continuous investment

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
		default value: none Default i/e: MIG	by NCAP_BND; if not defined, assumed to be equal to the lower bound. Requires MIP.	bound. (See Section 5.9)	variable VAR_SNCAP in the discrete investment equation EQ_DSCNCAP
NCAP_START (r,p)	PRC_NOFF	Year [1000,∞); default value: none	NCAP_START(r,p)=y is equivalent to PRC_NOFF(r,p,BOH,y-1).	Start year for new investments	Affects the availability of investment variable (VAR_NCAP)
NCAP_TLIFE (r,datayear,p)	NCAP_ELIFE, COEF_CPT, COEF_RPTI, DUR_MAX	Years (0,∞); default value: G_TLIFE Default i/e: STD	Expected for all technologies that have investment costs. Values below 0.25 cannot be well accounted in the objective function, and should thus be avoided (they are automatically resetted to 1).	Technical lifetime of a process.	Impacts all calculations that are dependent upon the availability of investments (VAR_NCAP) including capacity transfer (EQ_CPT), commodity flow (EQ(I)_COMBAL), costs (EQ_OBJINV, EQ_OBJINV, EQ_OBJINZ, EQ_OBJIALV).
NCAP_VALU (r,datayear,p,c,cur)	NCAP_OCOM	Monetary unit / commodity unit $[0,\infty)$; default value: none Default i/e: STD	Provided when a released commodity has a value.	Value of a commodity released at decommissioning (NCAP_OCOM).	Applied to the investment related (VAR_NCAP, NCAP_PASTI) release flow at decommissioning in the objective function (EQ_OBJSALV).
PRC_ACTFLO (r,datayear,p,cg)	PRC_CAPACT, prc_actunt, prc_spg, rpc_aire	Commodity unit / activity unit $(0,\infty)$; default value: 1 Default i/e: STD	Only (rarely) provided when either the activity and flow variables of a process are in different units, or if there is a conversion efficiency between the activity and the flow(s) in the PCG.	 Conversion factor from units of activity to units of those flow variables that define the activity (primary commodity group), or, 2) Conversion 	Applied to the primary commodity (prc_pcg) flow variables (VAR_FLO, VAR_IRE) to relate overall activity (VAR_ACT in EQ_ACTFLO). When the Reduction

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Default values & Default values & Default inter- /extrapolation ²⁵	(Required / Omit / Special conditions)	Description	or variables ²⁷
			The group (cg) can be the whole PCG or any individual commodity in the PCG, or 'ACT' (=PCG).	multiplier representing the amount of flow(s) in the cg per 1 unit of activity.	algorithm activated it is applied to the activity variable (VAR_ACT) in those cases where the flow variable (VAR_FLO) can be replaced by the activity variable (e.g. the activity is defined by one commodity flow).
PRC_CAPACT (r,p)	PRC_ACTFLO, PRC_ACTUNT	Activity unit / capacity unit (0,∞); default value: 1 Default i/e: none		Conversion factor from capacity unit to activity unit assuming that the capacity is used for one year.	Applied along with the availability factor (INCAP_AF) to the investment (VAR_NCAP + NCAP_PASTI) in the utilization equations (EQ(I)_CAFLAC). Applied to the investment (VAR_NCAP + NCAP_PASTI) in the peak constraint (EQ_PEAK). Applied to the investment (VAR_NCAP + NCAP_PASTI) in the capacity utilization constraint for CHP plants (ECT_AFCHP) and peak constraint in the IER extension (see Part III).
PRC_MARK (r,datayear,p,item,c,bd)	FLO_MARK	Decimal fraction [open]; default value: none	Combined limit on commodity production is derived as the sum of the	Process group-wise market share, which defines a constraint	EQ(I)_FLOMRK VAR_COMPRD

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter-	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
		/extrapolation ²⁵			
		Default i/e: 11	process-specific productions multiplied by the inverse values of PRC_MARK. The constraint is applied to the annual production of commodity.	for the combined market share of multiple processes in the total commodity production.	
PRC_RESID (r,datayear,p)	NCAP_PASTI	Capacity unit $[0,\infty)$; default value: none Default value: none Default i/e: 1 (options 5/15 may be used for extrapolation over TLIFE)	If only a single data point is specified, linear decay of the specified residual capacity over technical lifetime is assumed. Used as an alternative to NCAP_PASTI, not to use both for the same process.	Residual existing capacity stock of process (p) still available in the year specified (datayear). PRC_RESID is most useful for describing the stock of capacity with mixed vintages, while NCAP_PASTI is suited for capacities of a certain vintages, such as an individual power plants.	EQ(I)_CAPACT EQ(I)_CAFLAC EQL_CAPFLO EQ(I)_CPT VAR_CAP
R_CUREX (r,cur1,cur2)	G_CUREX	Scalar (0,∞) Default value: none Default i/e: N/A	The target currency cur2 must have a discount rate defined with G_DRATE.	Conversion factor from currency cur1 to currency cur2 in region r, in order to use cur2 in the objective function.	Affects cost coefficients in EQ_OBJ
RCAP_BLK (r,datayear,p)	PRC_RCAP RCAP_BND	Capacity unit [0,∞); default value: none Default i/e: STD	Only effective when lumpy early capacity retirements are active (RETIRE=MIP). Requires MIP.	Retirement block size.	EQ_DSCRET VAR_DRCAP VAR_SCAP
RCAP_BND (r,datayear,p,bd)	PRC_RCAP RCAP_BLK	Capacity unit [0,∞); default value: none Default i/e: STD	Unless the control variable DSCAUTO=YES, requires that PRC_RCAP is defined for process p.	Bound on the retired amount of capacity in a period (same bound for all vintages).	VAR_RCAP VAR_SCAP
REG_BNDCST	REG_CUMCST	Monetary unit	The cost aggregations	Bound on regional	EQ_BNDCST

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
(r,datayear,agg,cur,bd)		[0,∞); default value: none Default i/e: MIG	(agg) supported are listed in the set COSTAGG (see Table 1).	costs by type of cost aggregation.	VAR_CUMCST
REG_CUMCST (r,y1,y2,agg,cur,bd)	REG_BNDCST	Monetary unit [0,∞); default value: none Default i/e: N/A	The cost aggregations (agg) supported are listed in the set COSTAGG (see Table 1).	Cumulative bound on regional costs by type of cost aggregation.	EQ_BNDCST VAR_CUMCST
REG_FIXT (all_r)		Year [1000,∞); default value: none	Only taken into account when the first periods are fixed by using the FIXBOH control variable.	Year up to which periods are fixed by period	-
RPT_OPT (item,j)		Integer value [open]; default value: none	See Part III, Table 15 for a list and descriptions of available options.	Miscellaneous reporting options	-
SHAPE (j,age)	FLO_FUNC, FLO_SUM, NCAP_AFX, NCAP_FOMX, NCAP_FSUBX, NCAP_FTAXX	Scalar [open]; default value: none I/e: Full dense interpolation and extrapolation	Provided for each age dependent shaping curve that is to be applied.	Multiplier table used for any shaping parameters (*_*X) to adjust the corresponding technical data as function of the age; the table can contain different multiplier curves that are identified by the index j.	{See Related Parameters}
STG_CHRG (r,datayear,p,s)	prc_nstts, prc_stgips, prc_stgtss	Scalar [0,∞); default value: none Default i/e: STD	Only applicable to storage processes (STG): timeslice storage, inter-period storage or night storage devices.	Annual exogenous charging of a storage technology in a particular timeslice s.	Exogenous charging of storage enters storage equations (EQ_STGTSS, EQ_STGIPS) as right- hand side constant.
STG_EFF (r,datayear,p)	prc_nstts, prc_stgips, prc_stgtss	Decimal fraction [0,∞); default value: 1 Default i/e: STD	Only applicable to storage processes (STG): timeslice storage, inter-period storage or night storage	Efficiency of storage process.	Applied to the storage output flow (VAR_SOUT) in the commodity balance

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			devices.		(EQ(I)_COMBAL) for the stored commodity.
STG_LOSS (r,datayear,p,s)	prc_nstts, prc_stgips, prc_stgtss	Scalar [open]; default value: none Default i/e: STD	Only applicable to storage processes (STG): timeslice storage, inter-period storage or night storage devices. STG_LOSS-0 defines the loss in proportion to the initial storage level during one year's storage time. STG_LOSS-0 defines an equilibrium loss, i.e. how much the annual losses would be if the storage level is kept constant.	Annual loss of a storage process per unit of average energy stored.	Timeslice storage process (EQ_STGTSS): applied to the average storage level (VAR_ACT) between two consecutive timeslices. Inter-period storage process (EQ_STGIPS): applied to the average storage level from the pre-period (VAR_ACT) and the net inflow (VAR_SIN-VAR_SOUT) of the current period.
STGIN_BND (r,datayear,p,c,s,bd)	prc_nstts, prc_stgips, prc_stgtss	Commodity unit [0,∞); default value: none Default i/e: MIG	Only applicable to storage processes (STG): timeslice storage, inter-period storage or night storage devices.	Bound on the input flow of a storage process in a timeslice s.	Storage input bound constraint (EQ(I)_STGIN) when s is above prc_tsl of the storage process. Direct bound on storage input flow (VAR_SIN) when at the prc_tsl level.
STGOUT_BND (r,datayear,p,c,s,bd)	prc_nstts, prc_stgips, prc_stgtss	Commodity unit [0,∞); default value: none Default i/e: MIG	Only applicable to storage processes (STG): timeslice storage, inter-period storage or night storage devices.	Bound on the output flow of a storage process in a timeslice s.	Storage output bound constraint (EQ(I)_STGIN) when s is above prc_tsl of the storage process. Direct bound on storage output flow variable (VAR_SOUT) when at the prc_tsl level.

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
TL_CCAP0 (r,teg)	(Alias: CCAP0) PAT, CCOST0	Capacity unit [open]; default value: none	Requires using ETL. For learning technologies teg when ETL is used.	Initial cumulative capacity of a learning technology.	Cumulative investment constraint (EQ_CUINV) and cumulative capacity variable (VAR_CCAP) in endogenous technological learning formulation.
TL_CCAPM (r,teg)	(Alias: CCAPM) CCOSTM	Capacity unit [open]; default value: none	Requires using ETL. For learning technologies teg when ETL is used.	Maximum cumulative capacity.	Core ETL equations.
TL_CLUSTER (r,teg,prc)	(Alias: CLUSTER) TL_MRCLUST	Decimal fraction. [0-1]; default value: none	Requires using ETL (MIP). • Provided to model clustered endogenous technology learning. • Each of the learning parameters must also be specified for the key learning technology.	Indicator that a technology (teg) is a learning component that is part of another technology (prc) in region r; teg is also called key component.	EQ_CLU
TL_MRCLUST (r,teg,reg,p)	TL_CLUSTER	Decimal fraction. [0-1]; default value: none	Requires using ETL (MIP). • Provided to model clustered endogenous technology learning. • Each of the learning parameters must also be specified for the key learning technology.	Mapping for multi- region clustering between learning key components (teg) and processes (p) that utilize the key component.	EQ_MRCLU
TL_PRAT (r,teg)	(Alias: PRAT) ALPH BETA CCAPK CCOST0 PAT PBT	Scalar [0,1]; default value none	Requires using ETL. Provided for learning technologies (teg) when ETL is used.	Progress ratio indicating the drop in the investment cost each time there is a doubling of the installed capacity.	Fundamental factor to describe the learning curve and thus effects nearly all equations and variables related to endogenous technology learning (ETL).
TL_SC0 (r.teg)	(Alias: SC0)	Monetary unit / capacity unit	Requires using ETL. For learning technologies	Initial specific investment costs.	Defines together with CCAP0 initial point of

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
		[open]; default value: none	teg when ETL is used.		learning curve and affects thus the core equations and variables of endogenous technological learning (ETL).
TL_SEG (r,teg)	(Alias: SEG)	Integer [open];	Requires using ETL. For learning technologies teg when ETL is used. Currently limited to six segments by set kp.	Number of segments.	Influences the piecewise linear approximation of the cumulative cost curve (EQ_COS, EQ_LA1, EQ_LA2).
UC_ACT (uc_n,side,r,datayear,p, s)	uc_n, uc_gmap_p	None [open]; default value: none Default: i/e: STD	Used in user constraints. Direct inheritance. Weighted aggregation.	Coefficient of the activity variable VAR_ACT in a user constraint.	EQ(I)_UCXXX
UC_CAP (uc_n,side,r,datayear,p)	uc_n, uc_gmap_p	None [open]; default value: none Default: i/e: STD	Used in user constraints.	Coefficient of the activity variable VAR_CAP in a user constraint.	EQ(I)_UCXXX
UC_CLI (uc_n,side,r,datayear, item)		Dimensionless [open]; default value: none Default i/e: STD	Used in user constraints. Climate variable can be at least any of CO2-GTC, CO2-ATM, CO2-UP, CO2- LO, FORCING, DELTA-ATM, DELTA-LO (for carbon). See Appendix on Climate Module for details.	Multiplier of climate variable in user constraint	EQ(I)_UCXXX
UC_COMCON (uc_n,side,r,datayear,c, s)	uc_n, uc_gmap_c	None [open]; default value: none Default: i/e: STD	Used in user constraints. No inheritance/aggregation (might be changed in the future).	Coefficient of the commodity consumption variable VAR_COMCON in a user constraint.	EQ(I)_UCXXX
UC_COMPRD (uc_n,side,r,datayear,c,	uc_n, uc_gmap_c	None [open];	Used in user constraints. No inheritance/aggregation	Coefficient of the net commodity	EQ(I)_UCXXX

		Units / Ranges &	Instances ²⁶	Description	Affected equations
Input parameter	Related sets /	Default values &	(Required / Omit / Special	-	or variables ²⁷
(Indexes) ²³	parameters ²⁴	Default inter-	conditions)		
		/extrapolation ²⁵			
s)		default value: none	(might be changed in the	production variable	
		Default: I/e: STD	future).	VAR_COMPRD in a	
	ACT CUM	Dimensionless	Used in cumulative user	Multiplier of	
(uc n.r.p.v1.v2)	Act_con	[open]:	constraints only.	cumulative process	FO(I) UCR
		default value: none		activity variable in	VAR CUMFLO
		I/e: N/A		user constraint.	
UC_CUMCOM	COM_CUMNET	Dimensionless	Used in cumulative user	Multiplier of	EQ(I)_UC
(uc_n,r,type,c,y1,y2)	COM_CUMPRD	[open];	constraints only.	cumulative	EQ(I)_UCR
		default value: none	Type=NET/PRD determines	commodity variable in	VAR_CUMCOM
		I/e: N/A	(CUMNET/ CUMPRD).	user constraint.	
UC CUMFLO	FLO CUM	Dimensionless	Used in cumulative user	Multiplier of	EQ(I)_UC
(uc_n,r,p,c,y1,y2)	_	[open];	constraints only.	cumulative process	EQ(I)_UCR
		default value: none		flow variable in user	VAR_CUMFLO
	100000 Dea	I/e: N/A	I to an I the second second second second	constraint.	E6/11 (18)00/
UC_FLO	uc_n	None	Used in user constraints.	Coefficient of the flow	EQ(I)_UCXXX
(uc_n,side,i,datayeai,p,		default value: none	Weighted aggregation	VAR_FLO Valiable III a	
(,3)		Default: i/e: STD	Weighted aggregation.	user consciante.	
UC_IRE	uc_n	None	Used in user constraints.	Coefficient of the	EQ(I)_UCXXX
(uc_n,side,r,datayear,p,		[open];	Direct inheritance.	trade variable	
c,s)		default value: none	Weighted aggregation.	VAR_IRE in a user	
LIC NCAD		Default: i/e: STD	lleed in the second second second	constraint.	
UC_INCAP	uc_n, uc_gmap_p	[open]:	osed in user constraints.	activity variable	
(uc_n,side,i,datayear,p		default value: none		VAR NCAP in a user	
,		Default: i/e: STD		constraint.	
UC_RHS	uc_n, uc_r_sum,	None	Used in user constraints.	RHS constant with	RHS (right-hand side)
(uc_n,lim)	uc_t_sum,	[open];	Binding user constraints	bound type of bd of a	constant of a user
	uc_ts_sum	default value: none	are defined using bound	user constraint.	constraint, which is
		Derauit i/e: none	types IIm=UP/LO/FX.		summing over regions
			constraints can be defined		(uc_i_sum), periods
L	I	1	Leonar annea can de delineu	L	Lac_c_auni ana

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
			using the lim type lim=N.		timeslices (uc_ts_sum) (EQ(I)_UC).
UC_RHSR (r,uc_n,lim)	uc_n, uc_r_each, uc_t_sum, uc_ts_sum	None [open]; default value: none Default i/e: none	Used in user constraints. Binding user constraints are defined using bound types lim=UP/LO/FX. Non-binding (free) user constraints can be defined using the lim type lim=N.	RHS constant with bound type of bd of a user constraint.	RHS constant of user constraints, which are generated for each specified region (uc_r_each) and are summing over periods (uc_t_sum) and timeslices (uc_ts_sum) (EQ(I)_UCR).
UC_RHSRT (r,uc_n,datayear,lim)	uc_n, uc_r_each, uc_t_each, uc_t_succ, uc_ts_sum	None [open]; default value: none Default i/e: MIG	Used in user constraints. Binding user constraints are defined using bound types lim=UP/LO/FX. Non-binding (free) user constraints can be defined using the lim type lim=N.	RHS constant with bound type of bd of a user constraint.	RHS constant of user constraints, which are generated for each specified region (uc_t_each) and period (uc_t_each) and are summing over timeslices (uc_ts_sum) (EQ(I)_UCRT). If uc_t_succ instead of uc_t_each is specified the constraints will be generated as dynamic constraint between the two successive periods (EQ(I)_UCRSU).
UC_RHSRTS (r,uc_n,datayear,s,lim)	uc_n, uc_r_each, uc_t_each, uc_t_succ, uc_ts_each	None [open]; default value: none Default i/e: MIG	Used in user constraints. No inheritance / aggregation, unless the target timeslice level is specified by UC_TSL. Direct inheritance, if the target timeslice level is specified by UC_TSL.	RHS constant with bound type of bd of a user constraint.	RHS constant of user constraints, which are generated for each specified region (uc_r_each), period (uc_t_each) and timeslice (uc_ts_each) (EQ(I)_UCRTS).

Input parameter	Related sets /	Units / Ranges & Default values & Default inter-	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
(Indexes)	parameters	/extrapolation ²⁵	,		
			Binding user constraints are defined using bound types lim=UP/LO/FX. Non-binding (free) user constraints can be defined using the lim type lim=N.		If uc_t_succ instead of uc_t_each is specified the constraints will be generated as dynamic constraint between the two successive periods (EQ(I)_UCRSUS).
UC_RHST (uc_n,datayear,lim)	uc_n,uc_r_sum, uc_t_each, uc_t_succ, uc_ts_suc	None [open]; default value: none Default i/e: MIG	Used in user constraints. Binding user constraints are defined using bound types lim=UP/LO/FX. Non-binding (free) user constraints can be defined using the lim type lim=N.	RHS constant with bound type of bd of a user constraint.	RH5 constant of user constraints, which are generated for each specified period (uc_t_each) and are summing over regions (uc_r_sum) and timeslices (uc_ts_sum) (EQ(1)_UCT). If uc_t_succ instead of uc_t_each is specified the constraints will be generated as dynamic constraint between the two successive periods (EQ(1)_UCSU).
UC_RHSTS (uc_n,datayear,s,lim)	uc_n, uc_r_sum, uc_t_each, uc_t_succ, uc_ts_each	None) [open]; default value: none Default i/e: MIG	Used in user constraints. No inheritance/aggregation. Binding user constraints are defined using bound types lim=UP/LO/FX. Non-binding (free) user constraints can be defined using the lim type lim=N.	RHS constant with bound type of bd of a user constraint.	RHS constant of user constraints, which are generated for each specified period (uc_t_each) and timeslice (uc_ts_each) and are summing over regions (uc_r_sum) (EQ(I)_UCTS). If uc_t_succ instead of uc_t_each is specified the constraints will be

Input parameter (Indexes) ²³	Related sets / parameters ²⁴	Units / Ranges & Default values & Default inter- /extrapolation ²⁵	Instances ²⁶ (Required / Omit / Special conditions)	Description	Affected equations or variables ²⁷
					generated as dynamic constraint between the two successive periods (EQ(I)_UCSUS).
UC_TIME (uc_n,r,datayear)		Dimensionless [open]; default value: none Default i/e: STD	Used in user constraints. Adds a time constant to the RHS side.	Multiplier for the number of years in model periods (static UCs), or between milestone years (dynamic UCs)	EQ(I)_UCXXX
UC_UCN (uc_n,side,r,datayear, ucn)	UC_RHSRT	Dimensionless [open]; default value: none Default i/e: STD	Only taken into account if the user constraint is by region & period, and summing over timeslices and the RHS side is activated (EQ(I)_UCRSU).	Multiplier of user constraint variable in another user constraint.	EQ(I)_UCRSU VAR_UCRT
VDA_EMCB (r,datayear,c,com)	FLO_EMIS FLO_EFF	Emission units per flow units default value: none Default i/e: STD	Available in the VEDA shell. Any process-specific FLO_EMIS / FLO_EFF with the commodities c and com will override VDA EMCB.	Emissions (com) from the combustion of commodity (c) in region (r).	EQ_PTRANS

3.2 Internal parameters

Table 14 gives an overview of internal parameters generated by the TIMES preprocessor. Similar to the description of the internal sets, not all internal parameters used within TIMES are discussed. The list given in Table 14 focuses mainly on the parameters used in the preparation and creation of the equations in Chapter 6. In addition to the internal parameters listed here, the TIMES preprocessor computes additional internal parameters which are either used only as auxiliary parameters being valid only in a short section of the code or which are introduced to improve the performance of the code regarding computational time.

Internal	Instances	Description
parameter ³¹	(Required / Omit / Special	
(Indexes)	conditions)	
ALPH	For learning technologies teg when	Axis intercept on cumulative cost axis for description of linear
(r,kp,teg)	ETL is used.	equation valid for segment kp.
BETA	For learning technologies teg when	Slope of cumulative cost curve in segment kp (= specific
(r,kp,teg)	ETL is used.	investment cost).
CCAPK	For learning technologies teg when	Cumulative capacity at kinkpoint kp.
(r,kp,teg)	ETL is used.	
CCOST0(r,teg)	For learning technologies teg when	Initial cumulative cost of learning technology teg.
	ETL is used.	
CCOSTK	For learning technologies teg when	Cumulative investment cost at kinkpoint kp.
(r,kp,teg)	ETL is used.	
CCOSTM	For learning technologies teg when	Maximum cumulative cost based on CCAPM.
(r,teg)	ETL is used.	
COEF_AF	For each technology, at the level of	Availability coefficient of the capacity (new investment variable
(r,v,t,p,s,bd)	process operation (PRC_TSL).	VAR_NCAP plus still existing past investments NCAP_PASTI) in
		EQ(I)_CAPACT; COEF_AF is derived from the availability input
		parameters NCAP_AF, NCAP_AFA and NCAP_AFS taking into
		account any specified MULTI or SHAPE multipliers.

Table 14: Internal parameters in TIMES

 $\frac{31}{10}$ The first row contains the parameter name, the second row contains in brackets the index domain, for which the parameter is defined.

Internal	Instances	Description
parameter ³¹	(Required / Omit / Special	beschption
(Indexes)	conditions)	
COEF_CPT	For each technology the amount of	Fraction of capacity built in period v that is available in period t;
(r,v,t,p)	an investment (VAR_NCAP) available	might be smaller than 1 due to NCAP_ILED in vintage period or
	in the period.	the fact that the lifetime ends within a period.
COEF_ICOM	Whenever there is a commodity	Coefficient for commodity requirement during construction in
(r,v,t,p,c)	required during construction, the	period t due to investment decision in period v (see also
	consuming being taken from the	NCAP_ICOM).
	balance constraint (EQ(I)_COMBAL).	
	Applied to the investment variable	
	(VAR_NCAP) of period v in the	
	commodity balance (EQ(I)_COMBAL)	
	of period t.	
	The duration during which the	
	commodity is produced starts in the	
	year B(v)+NCAP_ILED(v)-	
	NCAP_CLED(v) and ends in the year	
	$B(v)+NCAP_ILED(v)-1.$	
COEF_OCOM	Whenever there is a commodity	Coefficient for commodity release during decommissioning time
(r,v,t,p,c)	released during decommissioning,	in period t due to investment made in period v.
	the production being added to the	
	balance constraint (EQ(I)_COMBAL).	
	Applied to the investment variable	
	(VAR_NCAP) of period v in the	
	commodity balance (EQ(I)_COMBAL)	
	of period t.	
	The release occurs during the	
	decommissioning lifetime	
	NCAP_DLIFE.	
COEF_PTRAN	For each flow through a process.	Coefficient of flow variable of commodity c belonging to
(r,v,t,p,cg,c,com_grp)		commodity group cg in EQ_PTRANS equation between the
		commodity groups cg and com_grp.
COEF_PVT	For each region, the present value of	Coefficient for the present value of periods, used primarily for
(r,t)	the time in each period.	undiscounting the solution marginals.

Internal	Instances	Description
parameter ³¹	(Required / Omit / Special	
(Indexes)	conditions)	
COEF_RPTI	For each technology whose technical	Number of repeated investment of process p in period v when
(r,v,p)	life (NCAP_TLIFE) is shorter than the	the technical lifetime minus the construction time is shorter than
	period.	the period duration; Rounded to the next largest integer
COD SALVD	For each technology existing past	number.
(r v n cur)	the end of the modelling horizon	technical discount rates and economic decommissioning times
(1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1, 1	with decommissioning costs.	teeninear discount rates and ceonomic decommissioning arres.
	adjustment in the objective function.	
COR_SALVI	For each process extending past the	Correction factor for investment costs taking into account
(r,v,p,cur)	end of the modelling horizon	technical discount rates, economic lifetimes and a user-defined
	adjustment in the objective function.	discount shift (triggered by the control switch MIDYEAR (see
-		Section 6.2 EQ_OBJ).
D	For each period, $D(t) = E(t)-B(t)+1$.	Duration of period t.
(t)	Plane Alexan contend of	Maniana (NOAD TIED - NOAD TITEE - NOAD DIAG
DUR_MAX	For the model.	Maximum of NCAP_ILED + NCAP_ILIFE + NCAP_DLAG +
		NCAP_DLIFE + NCAP_DELIF OVER all regions, periods and
M	For each period, if the duration of	Middle year of period t
(v)	the period is even, the middle year	
	of the period is $B(t) + D(t)/2 - 1$, if	
	the period is uneven, the middle	
	year is B(t) + D(t)/2 - 0.5.	
MINYR	For the model	Minimum year over $t = M(t) - D(t) + 1$; used in objective
		function.
MIYR_V1	For the model	First year of model horizon.
MIYR_VL	For the model	Last year of model horizon.
NICHIEG	For learning technologies teg when	Number of processes using the same key technology teg.
(r,teg)	EIL with technology clusters is used.	Teter (Extra clated unichle sector (ACT_COCT) (s
OBJ_ACOST	For each process with activity costs.	Inter-/Extrapolated variable costs (ACI_COST) for activity
(i,y,p,cur)	(FO_OBIVAR)	Variable (VAK_ACT) for each year.
L	11-2-000 0000	1

	1	
Internal	Instances	Description
parameters	(Required / Omit / Special	
(Indexes)	conditions)	
OBJ_COMNI	For each commodity with costs,	Inter-/Extrapolated cost, tax and subsidy (distinguished by the
(r,y,c,s,type,cur)	taxes or subsidies on the net	type index) on net production of commodity (c) for each year
	production.	associated with the variable VAR_COMNET. Cost types (type) are
	Enters the objective function	COST, TAX and SUB.
	(EQ_OBJVAR).	
OBJ_COMPD	For each commodity with costs,	Inter-/Extrapolated cost, tax and subsidy (distinguished by the
(r,y,c,s,type,cur)	taxes or subsidies on the commodity	type index) on production of commodity (c) for each year
	production.	associated with the variable VAR_COMPRD. Cost types (type) are
	Enters the objective function	COST, TAX and SUB.
	(EQ_OBJVAR).	
OBJ_CRF	For each technology with investment	Capital recovery factor of investment in technology p in objective
(r,y,p,cur)	costs.	function taking into account the economic lifetime (NCAP_ELIFE)
	Enters objective function	and the technology specific discount rate (NCAP_DRATE) or, if
	(EQ_OBJINV).	the latter is not specified, the general discount rate (G_DRATE).
OBJ_CRFD	For each technology with	Capital recovery factor of decommissioning costs in technology p
(r,y,p,cur)	decommissioning costs.	taking into account the economic lifetime (NCAP_DELIF) and the
	Enters objective function	technology specific discount rate (NCAP_DRATE) or, if the latter
	(EQ_OBJINV).	is not specified, the general discount rate (G_DRATE).
OBJ_DCEOH	Enters objective function	Discount factor for the year EOH + 1 based on the general
(r,cur)	(EQ_OBJSALV).	discount rate (G_DRATE).
OBJ_DCOST	For each technology with	Inter-/Extrapolated decommissioning costs (NCAP_DCOST) for
(r,y,p,cur)	decommissioning costs.	each year related to the investment (VAR_NCAP) of process p.
	Enters objective function	
	(EQ_OBJINV).	
OBJ_DISC	Enters objective function	Annual discount factor based on the general discount rate
(r,y,cur)	EQ_OBJINV, EQ_OBJVAR,	(G_DRATE) to discount costs in the year y to the base year
	EQ_OBJFIX, EQ_OBJSALV,	(G_DYEAR).
	EQ_OBJELS).	
OBJ_DIVI	Enters objective function	Divisor for investment costs (period duration, technical lifetime
(r,v,p)	(EQ_OBJINV).	or investment lead time depending on the investment cases 1a,
		[1b, 2a, 2b).

Internal	Instances	Description
parameter ³¹	(Required / Omit / Special	
(Indexes)	conditions)	
OBJ_DIVIII	Enters objective function	Divisor for decommissioning costs and salvaging of
(r,v,p)	(EQ_OBJINV).	decommissioning costs (period duration, technical lifetime or
		decommissioning time depending on the investment cases 1a,
		1b, 2a, 2b).
OBJ_DIVIV	Enters objective function	Divisor for fixed operating and maintenance costs and salvaging
(r,v,p)	(EQ_OBJFIX).	of investment costs.
OBJ_DLAGC	Enters objective function	Inter-/Extrapolated fixed capacity (VAR_NCAP+NCAP_PASTI)
(r,y,p,cur)	(EQ_OBJFIX).	costs between the end of the technical lifetime and the beginning
The statement is seen		of the decommissioning for each year.
OBJ_FCOST	For each flow variable with flow	Inter-/Extrapolated flow costs (FLO_COST) for each year for the
(r,y,p,c,s,cur)	related costs.	flow or trade variable (VAR_FLO, VAR_IRE) as well as capacity
	Enters objective function	related flows (specified by NCAP_COM, NCP_ICOM,
	(EQ_OBJVAR).	NCAP_OCOM).
OBJ_FDELV	For each flow with delivery costs.	Inter-/Extrapolated delivery costs (FLO_DELIV) for each year for
(r,y,p,c,s,cur)	Enters objective function	the flow or trade variable (VAR_FLO, VAR_IRE) as well as
	(EQ_OBJVAR).	capacity related flows (specified by NCAP_COM, NCP_ICOM,
		NCAP_OCOM).
OBJ_FOM	For each process with fixed	Inter-/Extrapolated fixed operating and maintenance costs
(r,y,p,cur)	operating and maintenance costs.	(NCAP_FOM) for the installed capacity
	Enters the objective function	(VAR_NCAP+NCAP_PASTI) for each year.
	(EQ_OBJFIX).	
OBJ_FSB	For each process with subsidy on	Inter-/Extrapolated subsidy (NCAP_FSUB) on installed capacity
(r,y,p,cur)	existing capacity.	(VAR_NCAP+NCAP_PASTI) for each year.
	Enters objective function	
	(EQ_OBJFIX).	
OBJ_FSOR	For each flow variable with subsidies.	Inter-/Extrapolated subsidy (FLO_SUB) for the flow or trade
(r,y,p,c,s,cur)	Enters objective function	variable (VAR_FLO, VAR_IRE) for each year as well as capacity
	(EQ_OBJVAR).	related flows (specified by NCAP_COM, NCP_ICOM,
		INCAP_OCOM).
ORT-LIX	For each flow variable with taxes.	Inter-/Extrapolated tax (FLU_IAX) for flow or trade variable
(r,y,p,c,s,cur)	Enters objective function	(VAK_FLU, VAK_IKE) for each year as well as capacity related
	(EQ_OBJVAR).	TIOWS (SPECIFIED BY NCAP_COM, NCP_ICOM, NCAP_OCOM).

Internal parameter ³¹	Instances (Required / Omit / Special	Description
(Indexes)	conditions)	
OBJ_FTX	For each process with taxes on	Inter-/Extrapolated tax (NCAP_FTAX) on installed capacity
(r,y,p,cur)	existing capacity.	(VAR_NCAP+NCAP_PASTI) for each year.
	Enters objective function	
	(EQ_OBJFIX).	
OBJ_ICOST	For each process with investment	Inter-/Extrapolated investment costs (NCAP_COST) for
(r,y,p,cur)	costs.	investment variable (VAR_NCAP) for each year.
	Enters objective function	
	(EQ_OBJINV).	
OBJ_IPRIC	For each import/export flow with	Inter-/Extrapolated import/export prices (IRE_PRICE) for
(r,y,p,c,s,all_r,ie,cur)	prices assigned to it.	import/export variable (VAR_IRE) for each year.
	Enters objective function	
	(EQ_OBJVAR).	
OBJ_ISUB	For each process with subsidy on	Inter-/Extrapolated subsidy (NCAP_ISUB) on new capacity
(r,y,p,cur)	new investment.	(VAR_NCAP) for each year.
1 00 1 00 00 00 00 00 00 00 00 00 00 00	Enters objective function	
	(EQ_OBJINV).	
OBJ_ITAX	For each process with taxes on new	Inter-/Extrapolated tax (NCAP_ITAX) on new capacity
(r,y,p,cur)	investment.	(VAR_NCAP) for each year.
LATA PERSONA AN AN ANALY	Enters objective function	
	(EQ_OBJINV).	
OBJ_PASTI	Enters objective function	Correction factor for past investments.
(r,v,p,cur)	(EQ_OBJINV).	
OBJ_PVT	Used as a multiplier in objective	Present value of time (in years) in period t, according to
(r,t,cur)	function in a few sparse cases.	currency cur in region r , discounted to the base year.
OBJSIC	For learning technologies.	Investment cost related salvage value of learning technology teg
(r,v,teg)	Enters objective function	with vintage period v at year EOH+1.
	(EQ_OBJINV).	
OBJSSC	For processes with investment costs.	Investment cost related salvage value of process p with vintage
(r,v,p,cur)	Enters objective function	period v at year EOH+1.
	(EQ_OBJSALV).	
PAT	For learning technologies teg when	Learning curve coefficient in the relationship:
(r.tea)	ETL is used.	$ISC = PAT * VAR CCAP^{(-PBT)}$

Internal	Instances	Description
parameter ³¹	(Required / Omit / Special	
(Indexes)	conditions)	
PBT	For learning technologies teg when	Learning curve exponent PBT(r,teg) =
(r,teg)	ETL is used.	LOG(PRAT(r,teg))/LOG(2).
PYR_V1	For the model	Minimum of pastyears and MINYR.
RS_FR	Defined for all commodities. Applied	Fraction of timeslice s in timeslice ts, if s is below ts, otherwise
(r,s,ts)	to flow variables in all equations in	1. In other words, RS_FR(r,s,ts) = G_YRFR(r,s) / G_YRFR(r,ts),
	order to take into account cases	if s is below ts, and otherwise 1.
	where the variables may be defined	5
	at a different timeslice level than the	
	level of the equation.	
RS_STG	Mainly applied for the modelling of	Lead from previous timeslice in the same cycle under the parent
(r,s)	storace cycles, but also in	timeslice.
	dispatching equations.	
RS_STGAV	Only applicable to storage processes	Average residence time of storage activity.
(r,s)	(STG): timeslice storage devices, to	
	calculate activity costs in proportion	
	to the time the commodity is stored.	
RS_STGPRD	Only applicable to storage processes	Number of storage periods in a year for each timeslice.
(r,s)	(STG): timeslice storage, inter-	
	period storage or night storage	
	devices.	
RS_UCS	Applied in timeslice-dynamic user	Lead from previous timeslice in the same cycle under the parent
(r,s,side)	constraints, to refer to the previous	timeslice.
	timeslice in the same cycle.	
RTP_FFCX	The efficiency parameter	Average SHAPE multiplier of the parameter FLO_FUNC and
(r,v,t,p,cg,c,cg)	COEF_PIRAN is multiplied by the	FLO_SUM efficiencies in the EQ_PTRANS equation in the period
	factor (1+RTP_FFCX).	(t) for capacity with vintage period (v). The SHAPE curve that
	Enters EQ_PTRANS equation.	should be used is specified by the user parameter FLO_FUNCX.
		The SHAPE feature allows to alter technical parameter given for
L		the vintage period as a function of the age of the installation.

Internal	Instances	Description
parameter ³¹	(Required / Omit / Special	
(Indexes)	conditions)	
RTCS_TSFR	Defined for each commodity with	The effective handling of timeslice aggregation/disaggregation. If
(r,t,c,s,ts)	COM_FR. Applied to flow variables	ts is below s in the timeslice tree, the value is 1, if s is below ts
	in all equations in order to take into	the value is COM_FR(r,s) / COM_FR(r,ts) for demand
	account cases where some of the	commodities with COM_FR given and G_YRFR(r,s) /
	variables may be defined at a	G_YRFR(r,ts) for all other commodities.
	different timeslice level than the	The parameter is used to match the timeslice resolution of flow
	level of the equation.	variables (VAR_FLO/VAR_IRE) and commodities. RTCS_TSFR is the
	e	coefficient of the flow variable, which is producing or consuming
		to the commodity c, in the commodity balance of c. If timeslice is corresponds
		timeslice resolution of the flow variable two cases may occur:
		The flow variables are on a finer timeslice level than the commodity
		balance: in this case the flow variables with timeslices s being below ts
		in the timeslice tree are summed to give the aggregated flow within
		timeslice ts. RTCS_TSFR has the value 1.
		The flow variables are on coarser timeslice level than the commodity
		balance: In this case the flow variable is split-up on the finer timeslice
		duration of s to ts: PTCS_TSER has the value = COM_EP(r.s.) /
		COM_ER(r s1) for demand commodities and G_YBER(r s) /
		G YRFR(r.s1) otherwise. When COM FR is used, the demand load curve
		is moved to the demand process. Thus, it is possible to model demand
		processes on an ANNUAL level and ensure at the same time that the
		process follows the given load curve COM_FR.
SALV_DEC	For those technologies with salvage	Salvage proportion of decommissioning costs made at period v
(r,v,p,k,ll)	costs incurred after the model	with commissioning year k.
	horizon the contribution to the	
CALLY THE	objective function.	
SALV_INV	For those technologies with salvage	Salvage proportion of investment made at period v with
(r,v,p,k)	costs incurred after the model	commissioning year k.
	norizon the contribution to the	
	objective function.	
YEARVAL	A value for each year.	Numerical value of year index (e.g. YEARVAL('1984') equals
1 (V)		119841.

3.3 Report parameters

3.3.1 Overview of report parameters

The parameters generated internally by TIMES to document the results of a model run are listed in Table 15. These parameters can be imported into the **VEDA-BE** tool for further result analysis. They are converted out of the **GDX**³² file via the **gdx2veda** GAMS utility into a **VEDA-BE** compatible format according to the file **times2vedavdd**³³. Note that some of the results are not transferred into parameters, but are directly accessed through the **times2vela.vdd** file (levels of commodity balances and peaking equation, total discounted value of objective function). The following naming conventions apply to the prefixes of the report parameters:

• CST_: detailed annual undiscounted cost parameters; note that also the costs of past investments, which are constants in the objective function, are being reported;
PAR: various primal and dual solution parameters;
EQ(1): directly accessed GAMS equation levels/marginals

- REG_: regional total cost indicators.

Table 15: Report parameters in TIMES

Report parameter ³⁴ (Indexes)	VEDA-BE attribute name	Description
AGG_OUT	VAR_FOut	Commodity production by an aggregation process:
(r,t,c,s)		Production of commodity (c) in period (t) and timeslice (s) from other commodities aggregated into c.
CAP_NEW	Cap_New	Newly installed capacity and lumpsum investment by vintage and commissioning period:
(r,v,p,t,uc_n)		New capacity and lumpsum investment of process (p) of vintage (v) commissioned in period (t).
CM_RESULT	VAR_Climate	Climate module results for the levels of climate variable (c) in period (t).
(c,t)		

³² GDX stands for GAMS Data Exchange. A GDX file is a binary file that stores the values of one or more GAMS symbols such as sets, parameters variables and equations. GDX files can be used to prepare data for a GAMS model, present results of a GAMS model, store results of the same model using different parameters etc. They do not store a model formulation or executable statements. ³³ The use of the **gdx2veda** tool together with the **times2veda.vdd** control file and the **VEDA-BE** software are described in Part V. ³⁴ Time target results reasoned raw (in breakets) the index domain for which the narameter is defined.

³⁴ First row: parameter name; second row (in brackets): the index domain, for which the parameter is defined.

Report parameter ³⁴ (Indexes)	VEDA-BE attribute name	Description
CM_MAXC_M (c,t)	Dual_Clic	Climate module results for the duals of constraint related to climate variable (c) in period (t).
CST_ACTC (r,v,t,p,uc_n)	Cost_Act	Annual activity costs: Annual undiscounted variable costs (caused by ACT_COST) in period (t) associated with the operation (activity) of a process (p) with vintage period (v). Additional indicator (uc_n) for start-up costs.
CST_COMC (r,t,c)	Cost_Com	Annual commodity costs: Annual undiscounted costs for commodity (c) (caused by COM_CSTNET and COM_CSTPRD) in period (t).
CST_COME (r,t,c)	Cost_Els	Annual elastic demand cost term: Annual costs (losses) due to elastic demand changes of commodity (c). When elastic demands are used the objective function describes the total surplus of producers and consumers, which reaches its maximum in the equilibrium of demand and supply.
CST_COMX (r,t,c)	Cost_Comx	Annual commodity taxes/subsidies: Annual undiscounted taxes and subsidies for commodity (c) (caused by COM_TAXNET, COM_SUBNET, COM_TAXPRD, COM_SUBPRD) in period (t).
CST_DAM (r,t,c)	Cost_Dam	Annual damage cost term: Annual undiscounted commodity (c) related costs, caused by DAM_COST, in period (t).
CST_DECC (r,v,t,p)	Cost_Dec	Annual decommissioning costs: Annual undiscounted decommissioning costs (caused by NCAP_DCOST and NCAP_DLAGC) in period (t), associated with the dismantling of process (p) with vintage period (v).
CST_FIXC (r,v,t,p)	Cost_Fom	Annual fixed operating and maintenance costs: Annual undiscounted fixed operating and maintenance costs (caused by NCAP_FOM) in period (t) associated with the installed capacity of process (p) with vintage period (v).
CST_FIXX (r,v,t,p)	Cost_Fixx	Annual fixed taxes/subsidies: Annual undiscounted fixed operating and maintenance costs (caused by NCAP_FTAX, NCAP_FSUB) in period (1) associated with the installed capacity of process (p) with vintage period (v).
CST_FLOC (r,v,t,p,c)	Cost_Flo	Annual flow costs (including import/export prices): Annual undiscounted flow related costs (caused by FLO_COST, FLO_DELV, IRE_PRICE) in period (t) associated with a commodity (c) flow in/out of a process (p) with vintage period (v) as well as capacity related commodity flows (specified by NCAP_COM, NCAP_ICOM, NCAP_COCM).
CST_FLOX (r,v,t,p,c)	Cost_Flox	Annual flow taxes/subsidies: Annual undiscounted flow related costs (caused by FLO_TAX, FLO_SUB) in period (t) associated with a commodity (c) flow in/out of a process (p) with vintage period (v) as well as capacity related commodity flows (specified by NCAP_COM, NCAP_ICOM, NCAP_OCOM).
CST_INVC (r,v,t,p,uc_n)	Cost_Inv	Annual investment costs: Annual undiscounted investment costs (caused by NCAP_COST) in period (t) spread over the economic

Report parameter ³⁴ (Indexes)	VEDA-BE attribute	Description
(Indexes)	name	lifetime (NCAP_ELIEE) of a process (p) with viotage period (v)
CST INVX	Cost Invx	Annual investment taxes/subsidies:
(r,v,t,p,uc_n)		Annual undiscounted investment costs (caused by NCAP_ITAX, NCAP_ISUB) in period (t) spread over the economic lifetime (NCAP_FLIFF) of a process (n) with vintage period (v).
CST IREC	Cost ire	Annual implied costs of endogenous trade:
(r,v,t,p,c)		Annual undiscounted costs from endogenous imports/exports of commodity (c) in period (t) associated with process (p) and vintage period (v), valued according to the marginal(s) of the trade equation of process p.
CST PVC	Cost NPV	Total discounted costs by commodity (optional):
(uc_n,r,c)		Total present value of commodity-related costs in the base year, by type (with types COM, ELS, DAM). See Part III, Section 3.10 on the reporting options, and Table 16 below for acronym explanations.
CST PVP	Cost NPV	Total discounted costs by process (optional):
(uc_n,r,p)		Total present value of process-related costs in the base year, by type (with types INV, INV+, FIX, ACT, FLO, IRE, where INV+ is only used for the split according to hurdle rate). See Part III, Section 3.10 on the reporting explanation of the 15 below for second and explanation and according to have a second s
CST SALV	Cost Salv	Salvag values of canacities at EOH+1:
(r,v,p)	COSC_Daily	Salvage value of investment cost, taxes and subsidies of process (p) with vintage period (v), for which the technical lifetime exceeds the end of the model britzon, value at year EOH+1.
CST TIME	Time NPV	Discounted value of time by period:
(r,t,s,uc_n)		Present value of the time in each model period (t) by region (r), with s='ANNUAL' and uc n='COST'/'LEVCOST' depending on whether the \$SET ANNCOST LEV reporting option has been used.
EO PEAK.L	EO Peak	Peaking Constraint Slack:
(r,t,c,s)	c	Level of the peaking equation (EQ_PEAK) of commodity (c) in period (t) and timeslice (s).
EQE_COMBAL.L	EQ_Combal	Commodity Slack/Levels:
(r,t,c,s)		Level of the commodity balance equation (EQE_COMBAL) of commodity (c) in period (t) and timeslice (s), where the equation is a strict equality.
EQG_COMBAL.L	EQ_Combal	Commodity Slack/Levels:
(r,t,c,s)		Level of the commodity balance equation (EQG_COMBAL) of commodity (c) in period (t) and timeslice (s), where the equation is an inequality.
F_IN	VAR_FIn	Commodity Consumption by Process:
(r,v,t,p,c,s)		Input flow (consumption) of commodity (c) in period (t) and timeslice (s) into process (p) with vintage period (v), including exchange processes.
F_OUT	VAR_FOut	Commodity Production by Process:
(r,v,t,p,c,s)		Output flow (production) of commodity (c) in period (t) and timeslice (s) into process (p) with vintage period (y), including exchange processes.
OBJZ.L	ObjZ	Total discounted system cost:
Report parameter ³⁴ (Indexes)	VEDA-BE attribute name	Description
--	------------------------------	---
0		Level of the ObjZ variable, equal to the value of the objective function.
PAR_ACTL	VAR_Act	Process Activity:
(r,v,t,p,s)		Level value of activity variable (VAR_ACT) in period (t), timeslice (s) of process (p) in vintage period (v).
PAR_ACTM	VAR_ActM	Process Activity – Marginals:
(r,v,t,p,s)		Undiscounted annual reduced costs of activity variable (VAR_ACT) in period (t) and timeslice (s) of process (p) with vintage period (v); when the variable is at its lower (upper) bound, the reduced cost describes the increase (decrease) in the objective function caused by an increase of the lower (upper) bound by one unit; the reduced cost can also be interpreted as the necessary decrease or increase of the to cost coefficient of the activity variable in the objective function, for the activity variable to leave tis lower (upper) bound.
PAR_CAPL	VAR_Cap	Technology Capacity:
(r,t,p)		Capacity of process (p) in period (t), derived from VAR_NCAP in previous periods summed over all vintage periods. For still existing past investments, see PAR_PASTI.
PAR_CAPLO	PAR_CapLO	Capacity Lower Limit:
(r,t,p)		Lower bound on capacity variable (CAP_BND('LO')), only reported, if the lower bound is greater than zero.
PAR_CAPM (r,t,p)	VAR_CapM	Technology Capacity – Marginals: Undiscounted reduced costs of capacity variable (VAR_CAP); only reported in those cases, in which the capacity variable is generated (bound CAP_BND specified or endogenous technology learning is used); the reduced costs describe in the case, that the capacity variable is at its lower (upper) bound, the cost increase (decrease) of the objective function caused by an increase of the lower (upper) bound by one unit. The reduced cost is undiscounted with COEF_PVT.
PAR_CAPUP (r.t.p)	PAR_CapUP	Capacity Upper Limit: Upper bound on capacity variable (CAP_BND('UP')), only reported, if upper bound is smaller than infinity.
PAR COMBALEM	EO CombalM	Commodity Slack/Levels - Marginals:
(r,t,c,s)	<u> </u>	Undiscounted annual shadow price of commodity balance (EQE_COMBAL) being a strict equality. The marginal value describes the cost increase in the objective function, if the difference between production and consumption is increased by one unit. The marginal value can be determined by the production side (increasing production), but can also be set by the demand side (e.g., decrease of consumption by energy saving or substitution measures).
PAR_COMBALGM (r,t,c,s)	EQ_CombalM	Commodity Slack/Levels – Marginals: Undiscounted annual shadow price of commodity balance (EQG_COMBAL) being an inequality (production being greater than or equal to consumption); positive number, if production equals consumption; the marginal value describes the cost increase in the objective function, if the difference between production and consumption is increased by one unit. The marginal value can be determined by the production side (increasing production), but can also be set by the demand side (e.g., decrease of consumption by energy saving or substitution measures)

Report	VEDA-BE	Description
parameter ³⁴	attribute	
(Indexes)	name	
PAR_COMNETL	VAR_Comnet	Commodity Net:
(r,t,c,s)		Level value of the variable corresponding the net level of a commodity (c) (VAR_COMNET). The net level of
		a commodity is equivalent to the total production minus total consumption of said commodity. It is only
		reported, if a bound or cost is specified for it or it is used in a user constraint.
PAR_COMNETM	VAR_ComnetM	Commodity Net – Marginal:
(r,t,c,s)		Undiscounted annual reduced costs of the VAR_COMNET variable of commodity (c). It is only reported, if a
280.80.80.00.000		bound or cost is specified for it or it is used in a user constraint.
PAR_COMPRDL	VAR_Comprd	Commodity Total Production:
(r,t,c,s)		Level value of the commodity production variable (VAR_COMPRD). The variable represents the total
		production of a commodity. It is only reported, if a bound or cost is specified for it or it is used in a user
		constraint.
PAR_COMPRDM	VAR_ComprdM	Commodity Total Production – Marginal:
(r,t,c,s)		Undiscounted annual reduced costs of the commodity production variable (VAR_COMPRD). It is only
		reported, if a bound or cost is specified for it or it is used in a user constraint.
PAR_CUMCST	VAR_CumCst	Cumulative costs by type (if constrained);
(r,v,t,uc_n,c)		Level of cumulative constraint for costs of type (uc_n) and currency (c) in region (r).
PAR_CUMFLOL	EQ_Cumflo	Cumulative flow constraint – Levels:
(r,p,c,v,t)		Level of cumulative constraint for flow of commodity (c) of process (p) between the year range (v-t).
PAR_CUMFLOM	EQ_CumfloM	Cumulative flow constraint – Marginals:
(r,p,c,v,t)		Shadow price of cumulative constraint for flow of commodity (c) of process (p) between the year range (v-
		t). Not undiscounted.
PAR_EOUT	VAR_Eout	Electricity supply by technology and energy source (optional):
(r,v,t,p,c)		Electricity output of electricity supply processes by energy source; based on using NRG_TMAP to identify
		electricity commodities, but excludes standard and storage processes having electricity as input.
D40 510		(Opted out by default - set RPI_OPI(FLO', 5')=1 to activate; see Part III, Section 3.10).
PAR_FLO	see:	How of commodity (c) entering or leaving process (p) with vintage period (v) in period (t).
(r,v,t,p,c,s)	F_IN/F_001	
PAR_FLU	none	Discounted reduced costs of flow variable of commodity (c) in period (t) or process (b) with vintage period
(r,v,t,p,c,s)		(v); the reduced costs describe that the now variable is at its lower (upper) bound, and give the cost
		Increase (decrease) of the objective function caused by an increase of the lower (upper) bound by one
		unit, the undiscounted reduced costs can be interpreted as the necessary decrease / increase of the cost
DAD THE	5001	Uter regional evolver and be of commendity (a) in period (b) with evolver (upper) boothd.
(rytocsie)	E IN/E OUT	Interregional exchange now of commonly (c) in period (1) via exchange process (p) entering region (r) as $\lim_{n \to \infty} \frac{1}{n} \lim_{n \to \infty} 1$
DAD TREM		Discounted reduced costs of intervencional exchange flow variable of commodity (c) in period (t) of
		Discounced reduced costs of mice regional exchange now variable of commodity (c) in period (c) of

Report parameter ³⁴ (Indexes)	VEDA-BE attribute name	Description
(r,v,t,p,c,s,ie)		exchange process (p) with vintage period (v); the reduced costs describe that the flow variable is at its lower (upper) bound, and give the cost increase (or decrease) of the objective function caused by an increase of the lower (upper bound) by one unit; the undiscounted reduced costs can be interpreted as the necessary decrease / increase of the cost coefficient of the flow variable in the objective function, such that the flow will leave its lower (upper) bound.
PAR_IPRIC (r,t,p,c,s,uc_n)	EQ_IreM	Inter-regional trade equations – Marginals: Undiscounted shadow price of the inter-regional trade equation of commodity (c) via exchange process (p) in period (t) and timeslice (s). The undiscounted shadow price can be interpreted as the import/export price of the traded commodity. Note: ucn={IMP/EXP}.
PAR_NCAPL (r,t,p)	VAR_Ncap	Technology Investment – New capacity: Level value of investment variable (VAR NCAP) of process (p) in period (v).
PAR_NCAPM (r,t,p)	VAR_NcapM	Technology Investment – Marginals: Undiscounted reduced costs of investment variable (VAR_NCAP) of process (p); only reported, when the capacity variable is at its lower or upper bound; the reduced costs describe in the case, that the investment variable is at its lower (upper) bound, the cost increase (decrease) of the objective function caused by an increase of the lower (upper) bound by one unit; the undiscounted reduced costs can be interpreted as the necessary decrease / increase in the investment cost coefficient, such that the investment variable will leave its lower (upper) bound.
PAR_NCAPR (r,t,p,uc_n)	VAR_NcapR	Technology Investment – BenCost + ObjRange (see Part III, Section 3.10 for more details): Cost-benefit and ranging indicators for process (p) in period (t), where uc_n is the name of the indicator: COST - the total unit costs of VAR_NCAP (in terms of an equivalent investment cost) CGAP - competitiveness gap (in terms of investment costs), obtained directly from the VAR_NCAP marginals (and optional ranging information) GGAP - competitiveness gap (in terms of investment costs), obtained by checking also the VAR_ACT, VAR_FLO and VAR_CAP marginals, in case VAR_NCAP is basic at zero RATIO - benefit / cost ratio, based on CGAP GRATIO - benefit / cost ratio, based on GGAP RNGLO - ranging information (LO) for VAR_NCAP (if ranging is activated; in terms of investment costs) RNGLO - ranging information (LP) for VAR_NCAP (if ranging is activated; in terms of investment costs)
PAR_PASTI (r,t,p,v)	VAR_Cap	Technology Capacity: Residual capacity of past investments (NCAP_PASTI) of process (p) still existing in period (t), where vintage (v) is set to '0' to distinguish residual capacity from new capacity.
PAR_PEAKM (r,t,c,s)	EQ_PeakM	Peaking Constraint Slack – Marginals: Undiscounted annual shadow price of peaking equation (EQ_PEAK) associated with commodity (c); since the peaking equation is at most only binding for one timeslice (s), a shadow price only exists for one timeslice. The shadow price can be interpreted as an additional premium to the shadow price of the

Report parameter ³⁴ (Indexes)	VEDA-BE attribute name	Description
		commodity balance that consumers of commodity (c) have to pay for consumption during peak times. The premium is used (besides other sources) to cover the capacity related costs (e.g., investment costs) of capacity contributing reserve capacity during peak times.
PAR_TOP (r,t,p,c,uc_n)	PAR_Top	Process topology: Process topology indicators for reporting use. Values are all zero, period (t) is the first milestone year, and $u_c n = IN/OUT$. (Opted out by default – SET RPT_TOP YES to activate.)
PAR_UCMRK (r,t,uc_n,c,s)	User_conFXM	Marginal cost of user constraint: Undiscounted shadow price of group-wise market share constraint (defined with PRC_MARK) for commodity c, identified with name uc_n, in period t and timeslice s.
PAR_UCRTP (uc_n,r,t,p,c)	User_DynbM	Marginal cost of dynamic process bound constraint: Undiscounted shadow price of dynamic process-wise bound constraint, identified with name uc_n, for variable c (CAP / NCAP / ACT), in period t and timeslice s.
PAR_UCSL (uc_n,r,t,s)	User_con	Level of user constraint (or its slack) (only reported when the VAR_UC variables are used): The level of user constraint (uc_n) by region (r), period (t) and timeslice (s). The levels should be zero whenever the RHS constant is zero and the equation is binding. If the constraint is not binding, the level together with the RHS constant gives the gap for the equation to become binding.
PAR_UCSM (uc_n,r,t,s)	User_conFXM	Marginal cost of fixed bound user constraint Marginal of user constraint (uc_n) by region (r), period (t) and timeslice (s). The marginals are undiscounted, if the constraint is defined by region and period. The marginals of cumulative and multi- region user constraints are thus not undiscounted, due to ambiguity.
REG_ACOST (r,t,uc_n)	Reg_ACost	Regional total annualized costs by period: Total annualized costs in region (r) by period (t) and cost category. The cost categories are INV, INVX, FIX, FIXX, VAR, VARX, IEL SL and DAM (see Table 16 below for more information).
REG_IREC (r)	Reg_irec	Regional total discounted implied trade cost: Total discounted implied trade costs in region (r), derived by multiplying the shadow prices of the trade equations by the trade volumes. The sum of REG_IREC over regions is zero.
REG_OBJ (r)	Reg_obj	Regional total discounted system cost: Discounted objective value (EQ_OBJ) for each region (r).
REG_WOBJ (r,uc_n,c)	Reg_wobj	Regional total discounted system cost by component: Discounted objective value (EQ_OB) for each region (r), by cost type (uc_n) and currency (c). The cost types are: INV, INVX, FIX, FIX, VAR, VARX, ELS, DAM (see Table 16 below for more information).
VAL_FLO (r,v,t,p,c)	Val_Flo	Annual commodity flow values: Flows of process (p) multiplied by the commodity balance marginals of those commodities (c) in period (t); the values can be interpreted as the market values of the process inputs and outputs.

3.3.2 Acronyms used in cost reporting parameters

The acronyms used in the reporting parameters for referring to certain types of costs are summarized in Table 16. The acronyms are used as qualifiers in the **uc_n** index of each reporting attribute, and are accessible in VEDA-BE through that same dimension.

Table 16: Acronyms used in the cost reporting parameters.

Cost parameter	Component acronyms		
CST_PVC	Total discounted costs by commodity (optional):		
(uc_n,r,c)	COM Commodity-related costs, taxes and subsidies		
	ELS Losses in elastic demands		
	DAM Damage costs		
CST_PVP	Total discounted costs by process (optional):		
(uc_n,r,p)	INV Investment costs, taxes and subsidies, excluding portions attributable to hurdle rates in excess of the general discount rate		
	INV+ Investment costs, taxes and subsidies, portions attributable to hurdle rates in excess of the general discount rate		
	FIX Fixed costs taxes and subsidies		
	ACT Activity costs		
	FLO Flows costs taxes and subsidies (including exogenous IRE prices)		
	IRE Implied trade costs minus revenues		
REG_ACOST	Regional total annualized costs by period:		
(r,t,uc_n)	INV Annualized investment costs		
	INVX Annualized investment taxes and subsidies		
	FIX Annual fixed costs		
	FIXX Annual fixed taxes and subsidies		
	VAR Annual variable costs		
	VARX Annual variable taxes and subsidies		
	IRE Annual implied trade costs minus revenues		
	ELS Annual losses in elastic demands		
	DAM Annual damage costs		
REG_WOBJ	Regional total discounted system cost by component:		
(r,uc_n,c)	INV Investment costs		
	INVX Investment taxes and subsidies		
	FIX Fixed costs		
	FIXX Fixed taxes and subsidies		
	VAR Variable costs		
	VARX Variable taxes and subsidies		
	ELS Losses in elastic demands		
	DAM Damage costs		

3.3.3 The levelized cost reporting option

As indicated in Table 15 above, the reporting of levelized costs for each process can be requested by setting the option RPT_OPT('NCAP', '1'). The results are stored in the VEDA-BE Var_NcapR result attribute, with the qualifier 'LEVCOST (with a possible system label prefix).

The levelized cost calculation option looks to weight all the costs influencing the choice of a technology by TIMES. It takes into consideration investment, operating, fuel, and other costs as a means of comparing the full cost associated with each technology.

Levelized cost can be calculated according to the following general formula:

$$LEC = \frac{\sum_{t=1}^{n} \frac{IC_{t}}{(1+r)^{t-1}} + \frac{OC_{t} + VC_{t} + \sum_{t} FC_{i,t} + FD_{i,t} + \sum_{j} ED_{j,t}}{(1+r)^{t-0.5}} - \frac{\sum_{k} BD_{k,t}}{(1+r)^{t-0.5}}$$
(1)

where

- r = discount rate (e.g. 5%)
- IC_t = investment expenditure in (the beginning of) year t
- $OC_t =$ fixed operating expenditure in year t
- VC_t = variable operating expenditure in year t
- FC_{it} = fuel-specific operating expenditure for fuel *i* in year *t*
- FD_{it} = fuel-specific acquisition expenditure for fuel *i* in year *t*
- ED_{jt} = emission-specific allowance expenditure for emission j in year t (optional)
- BD_{kt} = revenues from by-product k in year t (optional; see below)
- MO_{mt} = output of main product m in year t

The exponent t-0.5 in the formula indicates the good practice of using mid-year discounting for continuous streams of annual expenditures.

In TIMES, the specific investment, fixed and variable O&M costs and fuel-specific flow costs are calculated directly from the input data. However, for the fuel acquisition prices, emission prices and by-product prices, *commodity marginals* from the model solution are used. All the unit costs are multiplied by the corresponding *variable levels* as given by the model solution: investment cost and fixed operating costs are multiplied by the amounts of capacity installed / existing, variable operation costs by the activity levels, and fuel-specific costs by the process flow levels. Mid-year discounting can also be activated.

The outputs of the main products are taken from the flow levels of the commodities in the primary group (PG) of the process. An exception is CHP processes, for which the electricity output is considered the sole main output, and heat is considered as a by-product.

Options for variants of levelized cost reporting:

1. Do not include emission prices or by-product revenues in the calculation (RPT OPT('NCAP','1') = -1):

In this option emission prices are omitted from the calculation, in accordance with the most commonly used convention for LEC calculation. Consequently, any by-product revenues need to be omitted as well, because if emissions have prices, the by-product prices in the solution would of course be polluted by those prices, and thus it would be inconsistent to use them in the calculation. Instead, in this case any amount of by-product energy produced by ELE, CHP and HPL processes is indirectly credited by reducing the fuel-specific costs in the calculation to the fraction of the main output in the total amount of energy produced.

2. Include both emission prices and by-product revenues in the calculation $(RPT_OPT('NCAP', '1') = 1)$:

In this option both emission prices and by-product revenues are included in the calculation. The levelized cost thus represents the unit cost after subtracting the levelized value of all by-products from the gross value of the levelized cost. This approach of crediting for by-products in the LEC calculation has been utilized, for example, in the IEA *Projected Costs of Generating Electricity* studies.

3. <u>Include not only emission prices and by-product revenues, but also the revenues</u> <u>from the main product in the calculation (RPT_OPT('NCAP', '1') = 2):</u>

This option is similar to option (2) above, but in this case all product revenues are included in the calculation, including also the peak capacity credit from the TIMES peaking equation (when defined). The calculated LEC value thus represents the levelized **net** unit cost after subtracting the value of all products from the gross levelized cost. For competitive new capacity vintages, the resulting levelized cost should in this case generally be *negative*, because investments into technologies that enter the solution are normally profitable. For the marginal technologies that have been in some way forced into the solution, e.g. by specifying lower bounds on the capacity or by some other types of constraints, should normally have a positive levelized cost when using this option.

In the TIMES calculation, the expenditures for technology investments and process commodity flows include also taxes minus subsidies, if such have been specified. The levelized costs are calculated by process vintage, but only for new capacity vintages, as for them both the full cost data influencing technology choice and the operating history starting from the commissioning date are available, which is rarely the case for existing vintages.

Alternatively, if it would seem more convenient to define both the condensing mode efficiency and the full CHP efficiency, that can be done by using the parameters NCAP_CDME (condensing mode efficiency) and NCAP_BPME (back-pressure mode efficiency). When these two parameters are used, the NCAP_CEH and ACT_EFF parameters should then not be used at all. The activity will in this alternative approach always represent the electricity output in condensing mode.

For pass-out turbine technologies that have a reduction operation capability, one can enable the reduction option by adding to the process a third output of a dummy commodity, which is of type NRG and has a limit type 'N', and is also a member of the PCG. The model generator will automatically assign such a dummy output to the reduction operation, and will adjust the process transformation equation accordingly.

4.1.2.3 Alternative choices for defining the activity basis

As indicated above, the recommended basis of the activity of a CHP technology is the maximum electricity output, because the available technology data is usually best suited for using the electricity output as the basis for the activity. However, also the total energy output in full CHP mode can be used as the basis for the activity, should that be a more convenient way of defining the process data.

The below summarizes the different options modelling CHP processes according to the choice of the main efficiency parameters. Note that the cases with $-1 < CEH \le 0$ and $0 \le CEH < 1$ are identical when there is no lower bound for NCAP_CHPR specified.

Characteristic	Choices of	parameters for	modelling CHP ef	fficiencies
Efficiency parameters	j	ACT_EFF + NCAP_CE	H	NCAP_CDME+ NCAP_BPME
Value of CEH	-1 <ceh≤0< td=""><td>0≤CEH<1</td><td>CEH ≥ 1</td><td>None</td></ceh≤0<>	0≤CEH<1	CEH ≥ 1	None
Interpretation of CEH	Decrease in electricity output per unit of heat gained (when moving towards full CHP mode)	Loss in electricity output per unit of heat gained (when moving towards full CHP mode)	Loss in heat output per unit of electricity gained (when moving towards con- densing mode)	None
Activity	Max. electricity output	Electricity output in full condensing mode	Total energy output in full CHP mode	Electricity output in condensing mode
Capacity	Electrical capacity	Electrical capacity	Electrical+heat capacity	Electrical capacity
Efficiency specification	Max. electrical efficiency (=ACT_EFF) + the CEH specification	Electrical efficiency in full condensing mode (=ACT_EFF) + the CEH specification	Total efficiency in full CHP mode (=ACT_EFF) + the CEH specification	Electrical efficiency in condensing mode + total efficiency in full CHP mode
Investment & fixed O&M costs	Per electrical capacity	Per electrical capacity	Per electrical+heat capacity	Per electrical capacity
Variable costs	Per activity (see above)	Per activity (see above)	Per activity (see above)	Per activity (see above)

Table 18: Alternative ways of modelling efficiencies of CHP processes.

4.2 Inter-regional exchange processes

4.2.1 Structure and types of endogenous trade

In TIMES, the inter-regional trading structure of a given commodity basically consists of one or several exchange processes (called IRE processes), each of which defines a portion of the trading network for the commodity. The individual sub-networks can be linked together through common intermediating regions. As an example, electricity trade can be conveniently described by bi-lateral exchange processes (see Figure 12). But bi-lateral trading between all pairs of regions may become onerous in terms of data and model size. It is therefore useful to consider the other trade structure of TIMES, called multi-lateral trade, where regions trade with a common market (Figure 11). For either structure, the topology of the trading possibilities are all defined via the set top_ire of quintuples $\{r1,c1,r2,c2,p\}$, where r1, r2 are the exporting and importing regions respectively, c1, c2 are the names of the trade commodity in regions r1 and r2 respectively, and p is the process identifier. Process p is a process in both regions. It has to be defined only once, but one can add parameters to it in both regions (e.g. costs, bounds, etc.). Nearly every piece of data in TIMES has to be assigned to a region.

TIMES provides considerable flexibility in the definition of trading structures. Each sub-network defined for a single exchange process can have the general structure shown in Figure 11. A trading structure that involves both several supply (export) regions and several demand (import) regions cannot be defined without introducing an intermediating 'market' region (R_M). Whenever such an intermediate region is defined between (at least) two different regions, the model generator will assume that the structure is actually meant to ignore the intermediate node-region shown in Figure 11, by generating a single trade



Figure 11. General structure of the pair-wise specification of the trading subnetwork allowed in TIMES for a single exchange process.

balance equation directly between all the export and all the import flows. If the intermediate step should nonetheless be included, for example, to reflect a physical market hub in the region R_M , this can be accomplished by dividing the sub-network into two parts, by using two exchange processes. Consequently, depending on the user's choice, the trading relationships shown in Figure 11 can be modeled both with and without the intermediate transportation step through the market region.

The general structure allowed for the trading sub-networks can be further divided into four cases, which will be discussed below in more detail:

- Case 1: Bi-lateral trading.
- Case 2: Unidirectional trade from some export regions into a single importing region
- Case 3: Multi-directional trade from a single export region to several importing regions
- Case 4: General multi-lateral trading structure

Trading without need for explicit marketplace definition

Cases 1, 2 and 3 fall in this category. Bi-lateral trade takes place between pairs of regions. An ordered pair of regions together with an exchange process is first identified, and the trade through the exchange process is balanced between these two regions. Whatever amount is exported from region *i* to region *j* is imported by region *j* from region *i* (possibly with an adjustment for transportation losses). The basic structure is shown in Figure 12. Bi-lateral trading can be fully described in TIMES by specifying the two pairwise connections in **top_ire**. The capacity and investment costs of the exchange process can be described individually for both regions. For Cases 2 and 3, the general structure of the trade relationships is shown in Figure 13. Also in these cases the definition of the trading structure is easy, because the relationships can be unambiguously described by pairwise **top_ire** specifications between two regions.

Trading based on marketplace

Case 4 is covered by the generic structure shown in Figure 11. Trading occurs in this case between at least three regions, and involves both several exporting regions and several importing regions. In this type of trade, the commodity is 'put on the market' by each region participating in the supply side of the market and may be bought by any region participating in the demand side of the market. This case is convenient for global commodities such as emission permits or crude oil where the transportation cost from R_i to



Figure 12. Case 1: Bi-lateral trade (both R1 and R2 qualify as RM).



Figure 13. General structure of unidirectional trade into a single import region (Case 2, left) and multidirectional trade from a single export region (Case 3, right).

 R_j may be approximated by $Cost_i+Cost_j$ (rather than a more accurate cost such as C_{ij}). When the exact cost (or losses) are strictly dependent on the pair i,j of trading regions, it may be more accurate to use bilateral trade.

In general, there are many different possibilities for defining the multi-lateral structure by using the pair-wise **top_ire** specifications. In order to comply with the structure allowed in TIMES, the user has to decide which of the regions represents the 'marketplace', i.e. is chosen to be the R_M shown in Figure 11. Note that the market region will participate both in the supply and demand side of the market. The TIMES model generator automatically identifies this general type of trading on the basis of the **top_ire** topology defined by the user. Therefore, the user only needs to define the possible trading relationships between regions into the set **top_ire**. If there are *n* supply regions and *m* demand regions, the total number of entries needed in **top_ire** for defining all the trade possibilities is n+m-2 (counting the market region has to be defined to be an intermediate node in the structure, the model generator will actually <u>not</u> introduce any intermediate step between the export and import regions.

The timeslice levels of the traded commodity may be different in each region (as well as the commodity name). However, some appropriate common timeslice level must be chosen for writing the market balance equation. *That common level is the level attached to the exchange process in the market region*. In all other respects, *the market region is not treated in any way differently* from the other regions participating in the market. Nevertheless, the user can of course provide different data for the different regions, for example investment costs or efficiencies for the exchange process can be differentiated by region.

If the sets of supply and demand regions participating in the market should actually be disjoint, even in that case the user has to choose one of the regions to be used as the intermediate market region. The imports to or exports from the market region can then be switched off by using an IRE XBND parameter, if that is considered necessary.

Remarks on flexibility

- 1. Any number of exchange processes can be defined for describing the total trade relationships of a single commodity (but see warning 1 below).
- 2. The names of traded commodities can be different in each region participating in the trade. In addition, also the import and export names of the traded commodities can be different (but see warning 2 below). This could be useful e.g. in the case of electricity, for which it is common to assume that the export commodity is taken from the system after grid transport, while the import commodity is introduced into the system before the grid.
- 3. Any number of commodities can be, in general imported to a region or exported from a region through the same process (but see warning 2 below).

<u>Warnings</u>

- For each exchange process of any traded commodity, the total structure of the trading sub-network, as defined in top_ire, must comply with one of the basic structures supported by TIMES (Cases 1–4). If, for example, several bi-lateral trading relationships are defined for the same commodity, they should, of course, not be defined under the same process, but each under a different process.
- 2. If the export and import names for a market-based commodity (c) are different in the market region, no other commodities should be imported to the market region through the same exchange process as commodity c.
- 3. The model generator combines the trading relationships of a single process into a single market whenever there is an intermediate region between two different regions. If, however, the intermediate exchange step should be explicitly included in the model, the trading sub-network should be divided between two different exchange processes.

<u>Example</u>

Assume that we want to set up a market-based trading where the commodity CRUD can be exported by regions A, B, C, and D, and that it can be imported by regions C, D, E and F. First, the exchange process and marketplace should be defined. For example, we may choose (C,XP,CRUD) as the marketplace, where XP has been chosen to be the name of the exchange process (recall that process XP is declared only once but exists in all trading regions, possibly with different parameters). The trade possibilities can then be defined simply by the following six **top_ire** entries:

SET	PRC /	XP	/;	
SET	TOP_I	RE	/	
A	. CRUD	. C	. CRUD	. XP
В	. CRUD	.С	. CRUD	.XP
D	. CRUD	. C	. CRUD	. XP
С	. CRUD	. D	. CRUD	. XP
С	. CRUD	.E	. CRUD	.XP
С	. CRUD	. F	. CRUD	. XP
1:				

To complete the RES definition needed for the exchange process, in addition only the set **prc_actunt(r,p,c,u)** needs be defined for the exchange process XP:

4 Usage notes on special types of processes

4.1 Combined heat and power

4.1.1 Overview

Cogeneration power plants or combined heat and power plants (CHP) are plants that consume one or more commodities and produce two commodities, electricity and heat. One can distinguish two different types of cogeneration power plants according to the flexibility of the outputs, a back-pressure turbine process and a pass-out turbine process.

Back-pressure turbines are systems where the ratio of heat production to electricity production is fixed, and the electricity generation is therefore directly proportional to the heat generation. Pass-out turbines are systems where the ratio of heat production to electricity production is flexible, usually having a minimum value of zero and a maximum value usually in the range of 0.8–3 (but can be even smaller or larger).

However, both types of CHP systems often additionally support so-called reduction operation, where the turbine can be by-passed, whereby all the steam is directed to a heat exchanger for producing heat. As a result, in a back-pressure turbine system, the ratio of heat production to electricity production may in such systems vary from the fixed value to infinity, and in a pass-out turbine system it may vary from zero to infinity.

All these different cases are illustrated in Figure 10 below, which shows the relations between heat and electricity production in different modes of a flexible CHP system, of which the back-pressure turbine system is a special case. Taking into account that thermal power plants usually have a minimum stable operation level, the operating area of the fixed back-pressure turbine system is represented by the line E-F in the Figure. The corresponding operating area of a pass-out turbine system (without reduction operation) is represented by the polygon A-B-F-E. In some cases the turbine characteristics require a minimum level of heat production in proportion to electricity, and with such a constraint the feasible operating area is reduced to C-D-F-E. Finally, with a reduction operation the Figure. Similarly, the operating area of a back-pressure turbine system with a reduction operation capability would be expanded to E-F-H-G.

Denoting the electrical efficiency in the full condensing mode (point B) by η_B , the total efficiency in the full CHP mode (point F) by η_F , the heat-to-power ratio (inverse slope of line E–F) by *R*, and the slope of the iso-fuel line (B–F) by *S*, we can easily write the relations between these as follows:

$$\eta_{\scriptscriptstyle B} = \frac{\eta_{\scriptscriptstyle F} \times (1 + R \times S)}{(1 + R)}$$
$$\eta_{\scriptscriptstyle F} = \frac{\eta_{\scriptscriptstyle B} \times (1 + R)}{1 + R \times S}$$
$$S = \frac{\eta_{\scriptscriptstyle B} \times (1 + R) - \eta_{\scriptscriptstyle F}}{\eta_{\scriptscriptstyle F} \times R}$$

The core TIMES parameters for modeling the CHP attributes are listed in Table 17.

Table 17: Core TIMES parameters related to the modelling of CHP processes.

Attribute name	Description
ACT_EFF	Efficiency: amount of activity produced by 1 unit of input flow
ACT_MINLD	Minimum stable level of operation
NCAP_CHPR	Heat-to-power ratio *
NCAP_CEH	Coefficient of electricity to heat *
NCAP_CDME	Efficiency in full condensing mode
NCAP_BPME	Efficiency in back-pressure mode (full CHP mode) *
NCAP_AFA / NCAP_AFC	Bound on the annual utilization factor

* Only taken into account for processes defined to be of type CHP with the set prc_map.

4.1.2 Defining CHP attributes in TIMES

4.1.2.1 Back-pressure turbine systems

For modelling a fixed back-pressure turbine system in TIMES, the following approach is recommended:

- Define the PCG of the process to consist of both the electricity and heat output commodities (using the set **prc_actunt**);
- Define the process type to be CHP (using the set prc_map);
- Use the electrical output as the basis of the process activity, and choose the capacity unit accordingly (using the parameter PRC_CAPACT).
- Define the process electrical efficiency (by using the parameter ACT_EFF);



Figure 10: Illustration of basic CHP characteristics supported in TIMES.

- Define the process cost parameters accordingly; for example, specify the investment and fixed O&M costs per electrical capacity;
- Define the fixed heat-to-power ratio (using the parameter NCAP CHPR);
- Optionally, define also a maximum annual utilization factor considering the typical optimal sizing of CHP plants in proportion to the heat demand in the heat network represented (using the parameter NCAP_AFA);
- Optionally, define a minimum stable operation level (using ACT_MINLD).

All the input data specifications mentioned above should be quite straightforward. Note that the NCAP CEH parameter is not needed at all in the fixed turbine case.

For back-pressure turbine technologies that have a reduction operation capability, one can enable the reduction option by adding to the process a third output of a dummy commodity, which is of type NRG and has a limit type 'N', and is also a member of the PCG. The model generator will automatically assign such a dummy output to the reduction operation, and will adjust the process transformation equation accordingly.

4.1.2.2 Pass-out turbine systems

For modelling a flexible pass-out turbine system in TIMES, the following approach is recommended (but see additional remarks below):

- Define the PCG of the process to consist of both the electricity and heat output commodities (using the set prc_actunt);
- Define the process type to be CHP (using the set prc_map);
- Use the maximum electrical output as the basis of the process activity, and choose the capacity unit accordingly (using the parameter PRC_CAPACT). ³⁵
- Define the process electrical efficiency according to the maximum electrical efficiency (at point D in Figure 10), by using the parameter ACT_EFF;
- Define the process cost parameters accordingly, for example, specify the investment and fixed O&M costs per unit of electrical capacity;
- Define the maximum heat-to-power ratio (excluding any reduction operation), and optionally also the minimum heat-to-power ratio (using the parameter NCAP_CHPR);
- Define the slope S of the iso-fuel line (the line B–F in Figure 10) by specifying NCAP_CEH=S (where -1 ≤ S ≤ 0, as in Figure 10);
- Optionally, define also a maximum annual utilization factor considering the typical optimal sizing of CHP plants in the heat network represented (using the parameter NCAP_AFA and/or NCAP_AFC);
- Optionally, define a minimum stable operation level (using ACT MINLD).

Again, the specifications should be quite straightforward. The slope S of the iso-fuel line represents the amount of electricity lost per heat gained. In the example of Figure 10, the inverse of the slope has the value 7 and so one would define NCAP_CEH = -1/7.

³⁵ The activity remains constant over the iso-fuel line, but the electricity output varies when moving along it. Maximum electrical output is thus usually the most convenient quantity along this line for defining the basis of the process activity and capacity. This choice should then be consistently reflected in the input data (see Table 18).



Alternatively, if it would seem more convenient to define both the condensing mode efficiency and the full CHP efficiency, that can be done by using the parameters NCAP_CDME (condensing mode efficiency) and NCAP_BPME (back-pressure mode efficiency). When these two parameters are used, the NCAP_CEH and ACT_EFF parameters should then not be used at all. The activity will in this alternative approach always represent the electricity output in condensing mode.

For pass-out turbine technologies that have a reduction operation capability, one can enable the reduction option by adding to the process a third output of a dummy commodity, which is of type NRG and has a limit type 'N', and is also a member of the PCG. The model generator will automatically assign such a dummy output to the reduction operation, and will adjust the process transformation equation accordingly.

4.1.2.3 Alternative choices for defining the activity basis

As indicated above, the recommended basis of the activity of a CHP technology is the maximum electricity output, because the available technology data is usually best suited for using the electricity output as the basis for the activity. However, also the total energy output in full CHP mode can be used as the basis for the activity, should that be a more convenient way of defining the process data.

The below summarizes the different options modelling CHP processes according to the choice of the main efficiency parameters. Note that the cases with $-1 < CEH \le 0$ and $0 \le CEH < 1$ are identical when there is no lower bound for NCAP_CHPR specified.

Characteristic	Choices of	parameters for	modelling CHP ef	fficiencies
Efficiency parameters	j	ACT_EFF + NCAP_CE	H	NCAP_CDME+ NCAP_BPME
Value of CEH	-1 <ceh≤0< td=""><td>0≤CEH<1</td><td>CEH ≥ 1</td><td>None</td></ceh≤0<>	0≤CEH<1	CEH ≥ 1	None
Interpretation of CEH	Decrease in electricity output per unit of heat gained (when moving towards full CHP mode)	Loss in electricity output per unit of heat gained (when moving towards full CHP mode)	Loss in heat output per unit of electricity gained (when moving towards con- densing mode)	None
Activity	Max. electricity output	Electricity output in full condensing mode	Total energy output in full CHP mode	Electricity output in condensing mode
Capacity	Electrical capacity	Electrical capacity	Electrical+heat capacity	Electrical capacity
Efficiency specification	Max. electrical efficiency (=ACT_EFF) + the CEH specification	Electrical efficiency in full condensing mode (=ACT_EFF) + the CEH specification	Total efficiency in full CHP mode (=ACT_EFF) + the CEH specification	Electrical efficiency in condensing mode + total efficiency in full CHP mode
Investment & fixed O&M costs	Per electrical capacity	Per electrical capacity	Per electrical+heat capacity	Per electrical capacity
Variable costs	Per activity (see above)	Per activity (see above)	Per activity (see above)	Per activity (see above)

Table 18: Alternative ways of modelling efficiencies of CHP processes.

4.2 Inter-regional exchange processes

4.2.1 Structure and types of endogenous trade

In TIMES, the inter-regional trading structure of a given commodity basically consists of one or several exchange processes (called IRE processes), each of which defines a portion of the trading network for the commodity. The individual sub-networks can be linked together through common intermediating regions. As an example, electricity trade can be conveniently described by bi-lateral exchange processes (see Figure 12). But bi-lateral trading between all pairs of regions may become onerous in terms of data and model size. It is therefore useful to consider the other trade structure of TIMES, called multi-lateral trade, where regions trade with a common market (Figure 11). For either structure, the topology of the trading possibilities are all defined via the set top_ire of quintuples $\{r1,c1,r2,c2,p\}$, where r1, r2 are the exporting and importing regions respectively, c1, c2 are the names of the traded commodity in regions r1 and r2 respectively, and p is the process identifier. Process p is a process in both regions. It has to be defined only once, but one can add parameters to it in both regions (e.g. costs, bounds, etc.). Nearly every piece of data in TIMES has to be assigned to a region.

TIMES provides considerable flexibility in the definition of trading structures. Each sub-network defined for a single exchange process can have the general structure shown in Figure 11. A trading structure that involves both several supply (export) regions and several demand (import) regions cannot be defined without introducing an intermediating 'market' region (R_M). Whenever such an intermediate region is defined between (at least) two different regions, the model generator will assume that the structure is actually meant to ignore the intermediate node-region shown in Figure 11, by generating a single trade



Figure 11. General structure of the pair-wise specification of the trading subnetwork allowed in TIMES for a single exchange process.

balance equation directly between all the export and all the import flows. If the intermediate step should nonetheless be included, for example, to reflect a physical market hub in the region R_M , this can be accomplished by dividing the sub-network into two parts, by using two exchange processes. Consequently, depending on the user's choice, the trading relationships shown in Figure 11 can be modeled both with and without the intermediate transportation step through the market region.

The general structure allowed for the trading sub-networks can be further divided into four cases, which will be discussed below in more detail:

- Case 1: Bi-lateral trading.
- Case 2: Unidirectional trade from some export regions into a single importing region
- Case 3: Multi-directional trade from a single export region to several importing regions
- Case 4: General multi-lateral trading structure

Trading without need for explicit marketplace definition

Cases 1, 2 and 3 fall in this category. Bi-lateral trade takes place between pairs of regions. An ordered pair of regions together with an exchange process is first identified, and the trade through the exchange process is balanced between these two regions. Whatever amount is exported from region *i* to region *j* is imported by region *j* from region *i* (possibly with an adjustment for transportation losses). The basic structure is shown in Figure 12. Bi-lateral trading can be fully described in TIMES by specifying the two pairwise connections in **top_ire**. The capacity and investment costs of the exchange process can be described individually for both regions. For Cases 2 and 3, the general structure of the trade relationships is shown in Figure 13. Also in these cases the definition of the trading structure is easy, because the relationships can be unambiguously described by pairwise **top_ire** specifications between two regions.

Trading based on marketplace

Case 4 is covered by the generic structure shown in Figure 11. Trading occurs in this case between at least three regions, and involves both several exporting regions and several importing regions. In this type of trade, the commodity is 'put on the market' by each region participating in the supply side of the market and may be bought by any region participating in the demand side of the market. This case is convenient for global commodities such as emission permits or crude oil where the transportation cost from R_i to



Figure 12. Case 1: Bi-lateral trade (both R1 and R2 qualify as RM).



Figure 13. General structure of unidirectional trade into a single import region (Case 2, left) and multidirectional trade from a single export region (Case 3, right).

 R_j may be approximated by $Cost_i+Cost_j$ (rather than a more accurate cost such as C_{ij}). When the exact cost (or losses) are strictly dependent on the pair i,j of trading regions, it may be more accurate to use bilateral trade.

In general, there are many different possibilities for defining the multi-lateral structure by using the pair-wise **top_ire** specifications. In order to comply with the structure allowed in TIMES, the user has to decide which of the regions represents the 'marketplace', i.e. is chosen to be the R_M shown in Figure 11. Note that the market region will participate both in the supply and demand side of the market. The TIMES model generator automatically identifies this general type of trading on the basis of the **top_ire** topology defined by the user. Therefore, the user only needs to define the possible trading relationships between regions into the set **top_ire**. If there are *n* supply regions and *m* demand regions, the total number of entries needed in **top_ire** for defining all the trade possibilities is n+m-2 (counting the market region has to be defined to be an intermediate node in the structure, the model generator will actually <u>not</u> introduce any intermediate step between the export and import regions.

The timeslice levels of the traded commodity may be different in each region (as well as the commodity name). However, some appropriate common timeslice level must be chosen for writing the market balance equation. *That common level is the level attached to the exchange process in the market region*. In all other respects, *the market region is not treated in any way differently* from the other regions participating in the market. Nevertheless, the user can of course provide different data for the different regions, for example investment costs or efficiencies for the exchange process can be differentiated by region.

If the sets of supply and demand regions participating in the market should actually be disjoint, even in that case the user has to choose one of the regions to be used as the intermediate market region. The imports to or exports from the market region can then be switched off by using an IRE XBND parameter, if that is considered necessary.

Remarks on flexibility

- 1. Any number of exchange processes can be defined for describing the total trade relationships of a single commodity (but see warning 1 below).
- 2. The names of traded commodities can be different in each region participating in the trade. In addition, also the import and export names of the traded commodities can be different (but see warning 2 below). This could be useful e.g. in the case of electricity, for which it is common to assume that the export commodity is taken from the system after grid transport, while the import commodity is introduced into the system before the grid.
- 3. Any number of commodities can be, in general imported to a region or exported from a region through the same process (but see warning 2 below).

<u>Warnings</u>

- For each exchange process of any traded commodity, the total structure of the trading sub-network, as defined in top_ire, must comply with one of the basic structures supported by TIMES (Cases 1–4). If, for example, several bi-lateral trading relationships are defined for the same commodity, they should, of course, not be defined under the same process, but each under a different process.
- 2. If the export and import names for a market-based commodity (c) are different in the market region, no other commodities should be imported to the market region through the same exchange process as commodity c.
- 3. The model generator combines the trading relationships of a single process into a single market whenever there is an intermediate region between two different regions. If, however, the intermediate exchange step should be explicitly included in the model, the trading sub-network should be divided between two different exchange processes.

<u>Example</u>

Assume that we want to set up a market-based trading where the commodity CRUD can be exported by regions A, B, C, and D, and that it can be imported by regions C, D, E and F. First, the exchange process and marketplace should be defined. For example, we may choose (C,XP,CRUD) as the marketplace, where XP has been chosen to be the name of the exchange process (recall that process XP is declared only once but exists in all trading regions, possibly with different parameters). The trade possibilities can then be defined simply by the following six **top_ire** entries:

SET	PRC /	XP	/;	
SET	TOP_I	RE	/	
Α	. CRUD	. C	. CRUD	. XP
В	. CRUD	. C	. CRUD	. XP
D	. CRUD	.С	. CRUD	.XP
С	. CRUD	. D	. CRUD	. XP
С	. CRUD	. E	. CRUD	. XP
С	. CRUD	. F	. CRUD	.XP
1:				

To complete the RES definition needed for the exchange process, in addition only the set **prc_actunt(r,p,c,u)** needs be defined for the exchange process XP:

```
SET PRC_ACTUNT
   A .XP
          . CRUD
                 .́РЈ
   в
     .XP .CRUD
                 .PJ
   C
D
                 .PJ
     .XP .CRUD
     . XP
                 .PJ
          . CRUD
   E
F
     .XP
          . CRUD
                 .PJ
      . XP
          . CRUD
                 .PJ
1:
```

These definitions are sufficient for setting up of the market-based trade. Additionally, the user can, of course, specify various other data for the exchange processes, for example investment and distribution costs, efficiencies and bounds.

4.2.2 Input sets and parameters specific to trade processes

TIMES input SETs that have a special role in trade processes are the following:

• **top_ire(r1,c1,r2,c2,p):** For bi-lateral trade, unidirectional trade into a single destination region, and multidirectional trade from a single source region, **top_ire** should contain the corresponding entries from the exporting region(s) **r1** to the importing region(s) **r2**.

For market-based trade, **top_ire** must contain entries for each exporting region to the intermediate market region, and from the market region to each importing region. Each region may be both exporting and importing. One may thus force even a bi-lateral exchagne to be modeled as market-based trade, by introducing an additional **top_ire** entry within the desired market region between the exported and imported commodity. Instead of two trade balance equations, only one market balance equation is then generated.

 prc_aoff(r,p,y1,y2): Override used to control in what years (not periods) a process is unavailable. This set is not specifically related to exchange processes. However, in the case of market-based trading it can be used to switch off the entire commodity market for periods that fall within the range of years given by prc_aoff. The market will be closed for all commodities exchanged through the process (p). If trading should be possible only between certain years, even multiple entries of prc_aoff can be specified.

All the **top_ire** specifications are handled for the user by the user shell (VEDA/ANSWER) according to the characterization of the trade processes.

Additional remarks:

- 1. Commodity type can be used as the primary group of IRE processes. All commodities of that type, traded through the process, will then be included in the PCG.
- 2. Topology entries are automatically created on the basis of IRE_FLOSUM and FLO_EMIS defined for IRE processes (the latter only for ENV commodities).
- 3. In any non-bilateral trade, the marketplaces are automatically set by the model generator for any trade that involves an intermediate region between two different regions for the same exchange process (**p**) and same commodity (**c**), or if there are multiple destination (importing) regions for the same exporting region.

4. In market-based trade with r as the market region, the import/export regions participating in the market consist of all those regions that import/export commodity c from/into region r through process p (as defined in top_ire). The market region r by itself always participates in the market both as an importing and exporting region. However, the imports/exports of commodity (c) to/from the market region (r) can be switched off by using an IRE_XBND parameter, if necessary.

Input parameters

Input parameters specific to inter-regional exchange processes are listed in Table 19.

Table 19: Specific TIMES parameters related to the modelling of trade processes.
--

Attribute name	Description
(indexes)	22
IRE_FLO	Coefficient that represents the efficiency of exchange from r1 to r2,
(r1,y,p,c1,r2,c2,s2)	inside an inter-regional process where both regions are internal. Note
20 10000000000 00 07 00 00 00	that separate IRE_FLOs are required for import and export. Default =1
	for each top_ire direction specified. Time slice s2 refers to the region
	where the commodity arrives. Units: none
IRE_FLOSUM	Special attribute to represent auxiliary consumption (io = 'IN'), or
(r,y,p,c1,s,ie,c2,io)	production/emission (io = 'OUT') of commodity c2 due to the IMPort /
	EXPort (index ie) of the commodity c1 in region r by an inter-regional
	process \mathbf{p}^{30} . It is a fixed FLO_SUM with (one of) the pcg in that region.
	These relate commodities on the same side of the process. Auxiliary
	flows can also be specified on the process activity, by setting c1= AC I
	In the IRE_FLOSUM parameter (or in a FLO_EMIS parameter).
IRE_BND	Bound on the total import/export (index ie) into/from internal region
(r1,y,c,s,r2,ie,ba)	r1 , from/to region r2 , where region r2 may be internal or external ³⁷ ; c
	Beund an total imports (synaptic of some adity a in region with (from all
(rycania bd)	destinctions (sources, where a may be an internal or outernal ragion
(1,y,c,s,ie,bu)	(Default value: pope)
	Conversion factor between commedity units from unit of at in region
(r1 c1 r2 c2)	region region region ratio finter-regional exchanges
(11,01,12,02)	Default = 1, when exchange permitted Units; none
IRE TSOVT	Δ matrix that transforms timeslices of region r1 to region r2 as part of
(r1 s1 r2 s2)	inter-regional exchanges, including both internal and external. Default
	value = 1 when exchange permitted. Units: none.

³⁶ The indexing of auxiliary consumption flows or emissions of inter-regional exchange processes is illustrated in the figure below.



³⁷ The equation EQ(1)_XBND may have an external regional as region index (bounding the import from one external regions to all other regions).

<u>Remarks:</u>

- In market-based trading the IRE_FLO parameter is taken into account on the export side only (representing the efficiency from the export region to the common marketplace). By using this convention, any bi-lateral exchange can be represented by a fully equivalent market-based exchange simply by choosing one of the two regions to be the marketplace, and adding the corresponding entry to the set rpc_market(r,p,c). The efficiency of the exports from the market region itself to the marketplace should also be specified with an IRE_FLO parameter, when necessary (r1=r2=market region).
- If the user wants to specify efficiency on the import side of a market-based exchange, this can be done by using an IRE FLOSUM parameter on the import side.
- 3. Similarly to any other pair of regions, the total amount of commodity imported to a region from the commodity market can be constrained by the IRE_BND parameter, by specifying the market region as the export region. Correspondingly, the total amount of commodity exported from a supply region to the marketplace can be constrained by the IRE_BND parameter by specifying the market region as the import region.

4.2.3 Availability factors for trade processes

In TIMES, capacity by default bounds only the activity. However, with the NCAP_AFC / NCAP_AFCS attributes, one can bound the import / export flows instead. Capacity then also refers to the nominal maximum import (or export) capacity, e.g. the capacity of a transmission line in either direction. One can thus simultaneously bound the import and export flows by the same capacity but with different availabilities, which can be useful with bi-directional exchange links with different availabilities in the import/export direction. All these availability factors can be defined either on a desired timeslice level (NCAP_AFC), or on individual timeslices (NCAP_AFCS).

The rules for defining the availabilities for trade flows can be summarized as follows:

- If the import/export commodities are different (c1/c2): Use NCAP_AFC(c1) for bounding the import flow and NCAP_AFC(c2) for bounding the export flow, or use NCAP_AFC('NRG') for applying the same availability to both flows.
- If input=output=c, specifying *either* NCAP_AFC(c) *or* NCAP_AFC('NRG') alone applies to both imports and exports (unless the process type is DISTR, see Section 4.2.4 below). However, if they are both specified, then NCAP_AFC(c) applies to the import flow while NCAP_AFC('NRG') applies to the export flow.

Remarks:

- 1. As any process has only a single capacity variable, the availabilities specified for the import/export flows are always proportional to the same overall capacity.
- 2. Note that any the availability factors defined by NCAP_AFC are multiplied by any NCAP_AF/NCAP_AFS/NCAT_AFA value if defined for the same timeslice.

4.2.4 Notes on other attributes for trade processes

There are important limitations of using the parameters for standard processes for IRE processes. The most important limitations are summarized Table 20 with regard to the parameters with the prefixes 'ACT_', 'FLO_' and 'PRC_'. In addition, none of the CHP parameters, storage parameters (STG_*), or dispatching parameters (ACT_MINLD, ACT_UPS, ACT_CSTUP, ACT_LOSPL, ACT_CSTPL, ACT_TIME), can be used for IRE processes, and are ignored if used.

Table 20:	: Limitations	of using standard	process parameters for	or IRE processes.
-----------	---------------	-------------------	------------------------	-------------------

Attribute	Description	Limitations
name	20	
ACT_EFF	Activity efficiency	Can not be used
FLO_BND	Bound on a process flow variable	The bound will apply to the sum of both imports and exports of the given commodity, or, alter- natively, to the net imports when a true commodity group is speci- fied in the parameter (e.g. NRG).
FLO_EFF	Amount of process flow per unit of other process flow(s) or activity.	Same as for FLO_EMIS.
FLO_EMIS	Amount of emissions per unit of process flow(s) or activity.	Can only be used on the activity, by specifying 'ACT' as the source group.
FLO_FR	Process flow fraction	Can not be used
FLO_FUNC	Relationship between 2 groups of flows	Can not be used
FLO_MARK	Process market share bound	The bound will apply to import flow if FLO_MARK≥0, and to export flow if FLO_MARK≤0.
FLO_SHAR	Process flow share	Can not be used
FLO_SUM	Multiplier for a commodity flow in a relationship between 2 groups of flows	Can not be used
PRC_MARK	Process group-wise market share bound	Same as for FLO_MARK.

Additional remarks with respect to inter-regional trade (IRE) processes:

- By using the process type indicator 'DISTR', the activity and capacity of an IRE process will be based on the import flow only, if the same commodity is both imported and exported. In this case also NCAP_AFC(c) will only apply to the import flow of c.
- In peaking equations, IRE processes are by default taken into account by having gross imports on the supply side and gross exports on the consumption side. By defining the IRE process as a member of the set PRC_PKNO, and also defining NCAP_PKCNT>0, only the net imports are taken into account on the supply side, which can be useful for regions having trade flows passing through the region.

4.3 Storage processes

4.3.1 Overview

The TIMES model generator provides tools for specifying the following types of storage processes:

- Standard timeslice storage (STG without additional storage type qualifier)
- Generalized timeslice storage (STG+STS)
- Day/night storage (STG+NST, or just NST if at ANNUAL level)
- Inter-period storage (STG+STK)

The process type indicator STG is automatically assigned also to all processes that have been defined to be of type STS, NST or STK, with the exception of ANNUAL level NST processes, which are implemented as normal processes (see Section 4.3.3 below). Therefore, the user only needs to specify one of {STG, STS, NST, STK} as the process type of a storage process.

In addition to the charged and discharged commodity, storage processes can also produce and consume auxiliary commodities (emissions, electricity, fuels, waste etc.). The flows of such auxiliary commodities can be defined to be proportional either to the activity, the main input flows, or the main output flows of the storage (see Section 4.3.5 below).

4.3.2 Timeslice storage

The standard timeslice storage operates within the timeslice cycles under the timeslices of the level immediately above the process timeslice level. Consequently, the commodity charged can be only stored over the cycle of timeslices under a single parent timeslice, and not between timeslices under different parent timeslices. For example, a standard DAYNITE level storage can only store the charged commodity over the timeslices under one season, and not between seasons.

The activity of a timeslice storage represents the storage level, i.e. the amount of energy/material stored in the storage, measured at the beginning of each timeslice. However, one should note that for a DAYNITE level storage, the level of the activity variable for each timeslice is the actual storage level multiplied by the number of days under the parent timeslice, in the same way as the level of the activity variables for standard processes is the daily activity in that timeslice multiplied by the number of days under the parent timeslice.

If a storage technology is capable of storing energy for longer periods than over daily cycles, one may consider combining a SEASON/WEEKLY level storage process with a DAYNITE storage. However, a DAYNITE level storage may also be generalized to provide a storage capability between seasons, and even between periods, by using the generalized timeslice storage type qualifier 'STS' (and both 'STS' and 'STK', if the interperiod storage capability should be included). Because the same storage capacity can be utilized on all timeslice levels, the general storage process type may thus provide a somewhat improved modeling of a multi-cycle storage.

4.3.3 Day/Night storage

Day/Night storage (NST) is a timeslice storage, which can store energy over the day-night cycles, but not over weekly or seasonal cycles. In its basic functionality, an NST storage does not differ much from a standard timeslice storage, the main difference being that one can define the charging timeslices by specifying them in the set **prc nstts**.

Day/Night storage processes that produce ANNUAL level demand commodities can be modeled either as genuine storage processes or as standard processes with a night storage capability. In both cases 'NST' should be specified as the process type. If the process itself is defined to operate at the DAYNITE level, the process will be a genuine storage process, but if it is defined to operate at the ANNUAL level, it will be a standard process. For any such night storage devices, the charging and discharging commodity may be different, as defined via the set **top**.

When the NST process **p** is a genuine storage process, the input set **prc_nstts**(r,p,s) may be used for defining the charging timeslices **s**. Discharging can then only occur in timeslices other than the charging timeslices. Defining **prc_nstts** is required for all other genuine NST processes, except those serving an ANNUAL level demand, which can always discharge at the level of the demand, regardless of any **prc_nstts** defined.

In both types of NST storage, if the process is serving any ANNUAL level demand, the demand commodity is produced according to the load curve, while the charging can be optimized so that it occurs at night timeslices only. However, when the NST process is a normal process, it can be described in all other respects just as any other end-use technologies. For example, electric heating systems with accumulators can be described basically in the same way as direct electric heating systems, but with the additional night storage capability.

4.3.4 Inter-period storage

An inter-period storage process is able to store energy or material over periods. For example, a coal stockpile or a waste disposal site can be modeled as an inter-period storage. All inter-period storage processes should be defined to operate at the ANNUAL level, unless the generic timeslice process characterization (STS) is also specified.

The initial stock of an inter-period storage process can be specified by using the STG_CHRG parameter, which is interpolated such that it always includes the year at the beginning of the model horizon (B(t1)–1). The value of STG_CHRG in the year B(t1)–1 is used as the initial stock for inter-period storage. The allocation of the initial stock between the process vintages that are available at the beginning of the model horizon is left to be optimized by the model.

The activity of an inter-period storage is measured at the end of each period. Therefore, either by setting a lower bound on the activity or on the process availability, the storage can be prevented from getting fully discharged during any period. However, as there is no explicit accounting of the salvage value of the remaining contents of an inter-period storage, it may also be considered reasonable to allow discharging the storage fully in the last period, for taking into account the value of the storage.

4.3.5 Auxiliary storage flows

Storage processes can have any amount of auxiliary input or output commodities, as long as they are distinct from the main storage commodity. The flows of the auxiliary commodities can only be defined to be fixedly proportional either to the activity, the main input flows, or the main output flows. The main flows of timeslice and inter-period storage processes are the flows of the charged and discharged commodities included in the set primary commodity group PCG of the process. In the day/night storage processes, the main flows consist of all commodities in the primary and shadow groups of the process (see documentation).

The relation between the auxiliary flows and the activity or main flows should be defined by using the PRC_ACTFLO and the FLO_FUNC parameters. For example, if the main storage flows of the process consist of the commodity 'STORED', and the auxiliary commodity is 'AUX', the auxiliary flow can be defined in the three following ways, corresponding to the cases where the auxiliary flow is proportional to the activity, the input flow, or the output flow, respectively:

PRC_ACTFLO(r,t,p,'AUX')	! AUX proportional to activity
FLO_FUNC(r,t,p,'STORED','AUX',s)	! AUX proportional to input flow
FLO_FUNC(r,t,p,'AUX','STORED',s)	! AUX inversely proportional to output flow

These auxiliary storage flow relations have been implemented by adding a new TIMES equation EQ_STGAUX(r, v, t, p, c, s). As the auxiliary storage flows are represented by standard flow variables, any flow-related cost attributes and UC constraints can be additionally defined on these auxiliary flows. However, no transformation equations can be defined between any auxiliary storage flows. Therefore, if, for example, some auxiliary flows should also produce emissions, also these emissions should be defined on the basis of the activity or main flows, and not by defining a relation between the auxiliary flows related to storage processes, whether energy, material, or emissions, are described by using the three types of relations shown above.

A concrete example where these enhancements to the storage processes can be very useful is the modeling of waste management, and, in particular, the modeling of landfilling of different types of waste. Using inter-period storage processes for this purpose makes it possible to conveniently incorporate e.g. the following features in the waste management model:

- Modeling of methane emissions from landfilling in a dynamic way by using firstorder decay functions for the gradual waste decomposition (optionally with different rates of decay for different waste qualities);
- Modeling of other waste management and emission reduction options both before and after landfilling;
- Incorporating gate fees to landfill sites (by defining costs on an input-based auxiliary storage flow).

4.3.6 Input sets and parameters specifically related to storage processes

Input sets

There is only one TIMES input SETs specifically related to storage: prc_nstts. However, there are important storage-specific aspects related to each of the following input sets:

- prc_map(r,prc_grp,p): Defines the process as a storage process, where prc_grp=STG/STS/NST/STK according to the desired storage type.
- **prc_actunt(r,p,cg,units_act)**: Definition of the commodity/commodities in the PCG, i.e. those that are stored. Set of quadruples such that the members of the commodity group **cg** is used to define the charged and discharged commodity of storage process **p**, with activity units **units_act**, in region **r**. If the charged and discharged commodities are different, the group **cg** should preferably contain both of them, but if the user shell does not allow that, the model generator will automatically assign to the PCG any commodities on the shadow side that are of the same type than those already in the PCG, and are not verified to be auxiliary commodities. A commodity type can also be used as the primary group of storage processes. All commodities of that type will then be included in the PCG.
- top(r,p,c,io): Definition of the charged (io=IN) discharged (io=OUT) and optional auxiliary input/output commodities for storage process p in region r. The set top_ire should thus first and foremost contain the input/output indicators for the stored commodities defined by prc_actunt (see above), but should include also any auxiliary input/output commodities assumed for the process. When the charged and discharged commodity is the same, that commodity can optionally be defined only as an input or only as an output, and in that case it will be connected to the commodity balance equations either only on the production or only on the consumption side, instead of being connected on both sides.
- prc_nstts(r,p,s): For genuine night storage process p in region r, defines the timeslices s to be the charging timeslices, at which discharging cannot occur.

<u>Remarks</u>

In TIMES, the input (charge) and output (discharge) commodity of a storage process is usually the same commodity (input=output). When so, and this commodity is defined both as an input and an output of the process, the input and output flows will be taken into account in the commodity balance equations on different sides: the input on the consumption side, and the output on the production side.

However, in some cases this design has proven to be undesirable, because due to the nature of the storage processes, the input and output flows can usually be made arbitrarily large without affecting the storage operation or costs. That is so because the input flow may also by-pass the storage in the same timeslice or period, without being stored, and will then be directly converted into the output flow, without any costs or efficiency losses (unless STG_EFF is being used). Such arbitrary input/output flows can also make the total commodity production arbitrarily large, thereby rendering VAR_COMPRD a very unreliable measure of the size of the commodity market. This can be undesirable with

respect to various market-share constraints that are usually defined on the basis of the VAR COMPRD values.

In order to avoid any arbitrary storage flows on the production or consumption side, the input/output flows can be defined to be both connected either on the production or consumption side, instead of being on different sides. This will prevent the undesirable impacts of such arbitrary flows. The desired side can be chosen by the user by defining the commodity only as an output (production side) or as an input (consumption side).

Input parameters

The TIMES input parameters that are specific to storage processes or have a specific functionality for storage processes are summarized in Table 21.

Table 21: Specific TIMES parameters related to the modelling of storage processes.

Attribute name (indexes)	Description
STG_CHRG (r,y,p,s)	Exogenous amount assumed to be charged into storage \mathbf{p} , in timeslice \mathbf{s} and year \mathbf{y} . For timeslice storage this parameter can be specified for each period, while for inter-period storage this parameter is only taken into account for the first period, to describe the initial content of the storage at the beginning of the model horizon. Units: Unit of the storage input flow.
STG_EFF (r,y,p)	Coefficient that represents the storage efficiency of a storage process ${\bf p}$ in region ${\bf r}.$ Applied at the commodity balance to the output flow.
STG_LOSS (r,y,p,s)	Coefficient that represents the annual storage losses of a storage process \mathbf{p} in region \mathbf{r} , as a fraction of the (average) amount stored, corresponding to a storage time of one year. If the value specified is negative, the corresponding annual losses are interpreted as an annual equilibrium loss (under exponential decay).
STGIN_BND (r v p c s bd)	Bound on the input flow of commodity c of storage process p in a timeslice s . Units: Unit of the storage input flow. (Default value: none)
STGOUT_BND (r,y,p,c,s,bd)	Bound on the output flow of commodity \mathbf{c} of storage process \mathbf{p} in a timeslice \mathbf{s} . Units: Unit of the storage input flow. (Default value: none)
FLO_FUNC (r,y,p,c1,c2,s)	Defines the ratio between the flow of commodity c2 and the flow of commodity c1 , in timeslice s , in other words, an efficiency coefficient giving the flow of commodity c2 per one unit of flow of commodity c1 . For storage processes, can be used for defining amount of discharge in c2 per unit of auxiliary flow of c1 , or amount of auxiliary flow of c2 per unit of charging in c1 .
PRC_ACTFLO (r,y,p,c)	Defines a conversion coefficient between the activity and the flow in commodity \mathbf{c} . For storage processes, PRC_ACTFLO can be used for the commodities in the PCG in the standard way, but also for defining the amount of auxiliary flow of \mathbf{c} per unit of activity.
NCAP_AFC (r,y,p,cg,tslvl)	Can be used for defining availability factors for the process activity (amount stored), process output flow, or process input flow, or any combination of these. See Section 6.3 for additional information.
NCAP_AFCS (r,y,p,cg,s)	As NCAP_AFC above, but can be specified for individual timeslices. NCAP_AFCs values override NCAP_AFC values defined at the same level.

4.3.7 Availability factors for storage processes

In TIMES, capacity by default bounds only the activity. For storage, this means the amount of stored energy. However, with the NCAP_AFC/NCAP_AFCS attributes, one can bound the output (or input) flows instead. Capacity then also refers to the nominal output (or input) capacity, e.g. electrical capacity of a pumped hydro power plant. In addition, one can bound simultaneously both the output and input flows by the capacity, which can be useful if the charging rate is limited by the capacity as well. Moreover, one can simultaneously define a bound also for the activity (the amount stored) in proportion to the same capacity variable. All these availability factors can be defined either on a desired timeslice level (NCAP_AFC), or on individual timeslices (NCAP_AFCS).

The rules for defining the availabilities for storage flows/activity can be summarized as follows:

- If the input/output commodities are different (c1/c2): Use NCAP_AFC(c1) for bounding the input flow and NCAP_AFC(c2) for bounding the output flow.
- If input=output=c, NCAP_AFC('NRG') will define the availability factor for both the input and output flow, while NCAP_AFC(c) will define the availability factor for the output flow only, overriding any NCAP_AFC('NRG') value if that is also specified (assuming NRG is the type of the stored commodity).
- NCAP_AFC('ACT') can additionally be used for bounding the activity (the amount stored); in this case one must bear in mind that any capacity expressed in power units (e.g. MW/GW) is assumed to represent a storage capacity equivalent to the amount produced by full power during one full year/week/day for SEASON/WEEKLY/DAYNITE level storage processes, respectively. Knowing this, the availability factor can be adjusted to correspond to the assumed real storage capacity. For example, a capacity of 1 GW is assumed to represent a storage capacity of 24 GWh for a DAYNITE storage, and if the real daily storage capacity is, say 8 GWh / GW, the maximum availability factor should be 0.333.

Remarks:

- 1. As any storage process has only a single capacity variable, the assumption is that the availabilities specified for the output/input flows and the activity are all proportional to the same capacity.
- 2. Note that any the availability factors defined by NCAP_AFC are multiplied by any NCAP_AF/NCAP_AFS/NCAT_AFA value if defined for the same timeslice.

4.3.8 Notes on other attributes for storage processes

There are important limitations of using standard processes parameters for storage processes. The most important limitations are summarized in Table 22, with regard to the parameters with the prefixes 'ACT_', 'FLO_' and 'PRC_'. In addition, none of the CHP parameters, IRE parameters (IRE_*), or dispatching parameters (ACT_MINLD, ACT_UPS, ACT_CSTUP, ACT_LOSPL, ACT_CSTPL, ACT_TIME), can be used for storage processes, and are ignored if used.

Attribute	Description	Limitations	
name			
ACT_EFF	Activity efficiency	Can not be used	
FLO_BND	Bound on a process flow variable	Can only be used for bounding	
		auxiliary storage flows.	
FLO_COST	Added variable cost for commodity flow	Can only be used for the	
		charging (input) flow(s), and for	
		all auxiliary flows.	
FLO_DELIV	Delivery cost for commodity flow	Can only be used for the	
		discharge (output) flow(s), and	
		for all auxiliary flows.	
FLO_EFF,	Amount of process flow per unit of	Can only be used for defining an	
FLO_EMIS	other process flow(s) or activity.	auxiliary flow per unit of activity,	
(r,y,p,cg,c,s)		by specifying 'ACT' as the source	
		group (cg).	
FLO_FR	Process flow fraction	Can only be used for auxiliary	
		storage flows.	
FLO_FUNC	Relationship between 2 groups of flows	Can only be used for defining	
		auxiliary storage flows.	
FLO_MARK	Process market share bound	For a stored commodity, the	
		bound will apply to discharge	
		flow when FLO_MARK≥0, and to	
		charging flow if FLO_MARK≤0.	
FLO_SHAR	Process flow share	Can only be used among	
(r,y,p,c,cg,s,bd)		auxiliary flows, and for bounding	
		the output flow (c) in proportion	
		to the activity (cg='ACT')	
FLO_SUM	Multiplier for a commodity flow in a	Can only be used among	
	relationship between 2 groups of flows	auxiliary flows.	
FLO_TAX,	Tax/subsidy for the production/use of	Can only be used for auxiliary	
FLO_SUB	commodity by process	storage flows	
PRC_MARK	Process group-wise market share	Same limitations as for	
	bound	FLO_MARK.	

Table 22: Limitations of using standard process parameters for storage processes.

Additional remark on peaking equations:

• In peaking equations, storage processes producing the peaking commodity are by default taken into account by their capacity on the supply side, and not at all by their flows (charging/discharging). By defining the storage process as a member of the set PRC_PKNO, and also defining NCAP_PKCNT>0, the discharge from the storage is taken into account on the supply side instead of the capacity, and the charging into the storage is included on the consumption side (should such happen in the peak timeslice). That can be recommended, whenever the capacity represents the amount stored, and not the output capacity, and may be reasonable even for storage processes where the capacity represents the nominal maximum output flow.

5 Variables

This chapter describes each variable name, definition, and role in the TIMES Linear Program. To facilitate identification of the variables when examining the model's source code, all variable names start with the prefix VAR_. The value assigned to each variable indexed by some time period, represents the average value in that time period, but the case of VAR_NCAP(v) is an exception, since that variable represents a point-wise investment decided at time period v. VAR NCAP is discussed in detail below.

Table 23 is a list of TIMES variables by category, with brief description of each variable.

Remarks on Table 23:

- Many variables that are related to a process have two period indexes: t represents the current period, and v represents the vintage of a process, i.e. the period when the investment in that process was decided. For the VAR_NCAP variable, t is by definition equal to v. For other variables, $t \ge v$, if the process is vintaged (prc_vint), i.e., the characteristics of the process depend on the vintage year. If the process is non-vintaged, the characteristics of the capacity of a process are not differentiated by its vintage structure, so that the vintage index is actually not needed for the variables of a non-vintaged process. In these cases, the vintage index v is by convention set equal to the period index t.
- In Table 23, the variables are listed according to five categories, depending on what TIMES entity they represent. In the rest of the chapter, the variables are listed and fully described in alphabetical order.
- Table 23 does not list the variables used in the Climate Module, Damage Cost and ETL extensions of TIMES, which are fully documented in Appendices A, B, and C, respectively.
- In the Objective function category, Table 23 also lists several parameters that stand for certain portions of the objective functions. These are not bona fide GAMS variables, but mostly serve as convenient placeholders for this documentation, and also as useful parameters that may be reported in the solution.

Table 23. List of TIMES variables by category

Category	Variable name	Brief description
Region relate	ed	
	VAR_CUMCST	Cumulative amount of regional cost/tax/subsidy
Process relation	ted	• · · · · · · · · · · · · · · · · · · ·
	VAR_ACT	Annual activity of a process
	VAR_CAP	Current capacity of a process, all vintages together
	VAR_NCAP	Investment (new capacity) in a process
	VAR_DNCAP	Binary variable (VAR_DNCAP) and semi-continuous
	VAR_SNCAP	variable (VAR_SNCAP) used with the discrete investment option (see EO_DSCNCAP)
	VAR RCAP	Retired capacity of a process in a period by vintage
	VAR SCAP	Cumulative retired capacity of a process in a period
	VAR DRCAP	Binary variable for discrete capacity retirements
	VAR UPS	Started-up, shut-down, and off-line capacities
Commodity r	elated	,
	VAR_BLND	Blending variable (for oil refining)
	VAR_COMNET	Net amount of a commodity
	VAR_COMPRD	Gross production of a commodity
	VAR_CUMCOM	Cumulative gross/net production of commodity
	VAR_ELAST	Variables used to linearize elastic demand curves
Flow (Proces	s and Commodity) related
	VAR_FLO	Flow of a commodity in or out of a process
	VAR_CUMFLO	Cumulative amount of process flow/activity
	VAR_IRE	Flow of a commodity in or out of an exchange
		process (trade variable)
	VAR_SIN/OUT	Flow of a commodity in or out of a storage process
Objective fu	nction related	
	VAR_OBJ	Variable representing the overall objective function (all regions together)
The following	10 parameters are i	not true variables of the LP matrix
	OBJR	Parameter representing a regional component of the objective function.
	INVCOST	Parameter representing the investments portion of
		a regional component of the objective function
	INVTAXSUB	Parameter representing the taxes and subsidies
		attached to the investments portion of a regional
		Parameter representing the capital cost attached to
	INVDECON	the dismantling (decommissioning) portion of a
		regional component of the objective function
	FIXCOST	Parameter representing the fixed annual costs
		portion of a regional component of the objective
		function
	FIXTAXSUB	Parameter representing the taxes and subsidies
		component of the objective function
	VARCOST	Parameter representing the variable annual cost
		portion of a regional component of the objective
		function
	VARTAXSUB	Parameter representing the variable taxes and
		subsidies of a regional component of the objective
		TUNCTION

Category	Variable name	Brief description
category	FLASTCOST	Variable representing the demand loss portion of a
	LEASTCOST	regional component of the objective function
		Parameter representing the late revenue portion of
	LATEREVENUES	a regional component of the objective function.
	SALVAGE	Parameter representing the salvage value portion
		or a regional component of the objective function
User Constra	aint related ³⁸	
	VAR_UC	variable representing the LHS expression of a user
		constraint summing over regions (uc_r_sum),
		periods (uc_t_sum) and timeslices (uc_ts_sum).
	VAR_UCR	Variable representing the LHS expression of a user
		constraint summing over periods (uc_t_sum) and
		timeslices (uc_ts_sum) and being generated for
		the regions specified in uc_r_each .
	VAR_UCT	Variable representing the LHS expression of a user
		constraint summing over regions (uc_r_sum) and
		timeslices (uc_ts_sum) and being generated for
		the periods specified in uc_t_each.
	VAR_UCRT	Variable representing the LHS expression of a user
		constraint summing over timeslices (uc_ts_sum)
		and being generated for the regions specified in
		uc_r_each and periods in uc_t_each.
	VAR UCTS	Variable representing the LHS expression of a user
	_	constraint summing over regions (uc r sum) and
		being generated for the periods specified in
		uc t each and timeslices in uc ts each.
	VAR UCRTS	Variable representing the LHS expression of a user
		constraint summing over periods being generated
		for the regions specified in uc r each, the periods
		in uc t each and timeslices in uc ts each.

Notation for indexes: The following indexes are used in the remainder of this chapter:

r, **r**' = region; **v** = vintage; **t**, **t**' = time period; **y** = year; **p** = process; **c**, **c**' = commodity; **s**, **s**' = timeslice; **ie** = import or export; l = sense of a constraint (\geq , =, or \leq). In addition, some indexes (**u**; **ble**; **opr**; **j**; **uc_n**) are used for specific variables only and are defined in their context.

5.1 VAR_ACT(r,v,t,p,s)

Definition: the overall activity of a process. VAR_ACT is defined by the EQ_ACTFLO equation either as the sum of outflows or as the sum of inflows of a particular (user selected) group of commodities, adequately normalized. If the process is not vintaged, the vintage index \mathbf{v} is by convention set equal to the period index \mathbf{t} .

Role: reports the activity of a process and implicitly defines how the capacity is measured, since the activity is bounded by the available capacity in the constraint

³⁸ In case the dollar control parameter VAR_UC is set to YES, the user constraints are always strict equalities (l=E) with the RHS constants replaced by the user constraint variables given in the table. The RHS bound parameter (UC_RHS(R)(T)(S)) are then applied to these user constraint related variables. See Section 5.20.

¹²⁹

EQ(l)_CAPACT, e.g. if the activity of a coal power plant is defined over its electricity output, the capacity is measured in terms of the output commodity, e.g. MW_{electric}. Similarly, if the activity variable represents the input flow of coal, the capacity of the coal plant is measured in terms of the input commodity, e.g. MW_{coal}.

Bounds: Can be directly bounded by ACT_BND User constraints: Can be directly referred to by UC ACT

5.2 VAR_BLND(r,ble,opr)

Definition: amount of the blending stock **opr** in energy, volume or weight units needed for the production of the blending product **ble** in oil refinery modeling.

Role: used for specifying constraints on quality of the various refined petroleum products.

Bounds: Cannot be bounded. User constraints: Cannot be referred to in user constraints.

5.3 VAR_CAP(r,t,p)

Definition: the installed capacity in place in any given year \mathbf{t} , of all vintages of a process determined by the equation EQ(l)_CPT. The variable is equal to the sum of all previously made investments in new capacity, plus any remaining residual capacity installed before the modeling horizon,that has not yet reached the end of its technical lifetime, and minus any capacity that has been retired early.

Role: Its main purpose is to allow the total capacity of a process to be bounded. The variable is only created when

- capacity bounds (CAP_BND) for the total capacity installed are specified. In case only one lower or one upper capacity bound is specified, the variable is not generated, but the bound is directly used in the EQ(1)_CPT constraint.
- \circ $\;$ the capacity variable is needed in a user constraint, or
- the process is a learning technology (teg) in case that endogenous technological learning is used.

Bounds: Can be directly bounded by CAP_BND **User constraints:** Can be directly referred to by UC_CAP

5.4 VAR_COMNET(r,t,c,s)

Definition: the net amount of a commodity at period t, timeslice s. It is equal to the difference between amount procured (produced plus imported) minus amount disposed (consumed plus exported).

Role: The variable is only created if a bound is imposed, or a cost is explicitly associated with the net level of a commodity.

Bounds: Can be directly bounded by COM_BNDNET User constraints: Can be directly referred to by UC_COMNET

5.5 VAR_COMPRD(r,t,c,s)

Definition: the amount of commodity c procured at time period t, timeslice s.

Role: this variable is only created if a bound is imposed on total production of a commodity, or a cost is explicitly associated with production level of a commodity. The variable is defined through the equation EQE_COMPRD.

Bounds: Can be directly bounded by COM_BNDPRD User constraints: Can be directly referred to by UC_COMPRD

5.6 VAR_CUMCOM(r,c,type,y1,y2)

Definition: the cumulative amount of commodity c produced in region r between years y1 and y2, over all timeslices. The type indicator (PRD/NET) distinguishes between gross and net production.

Role: this variable is only created if a bound is imposed on cumulative gross/net production of a commodity. The variable is defined through the equations EQ_CUMPRD and EQ_CUMNET.

Bounds: Can be directly bounded by COM_CUMNET/ COM_CUMPRD User constraints: Can be directly referred to by UC_CUMCOM

5.7 VAR_CUMCST(r, y1,y2,costagg,cur)

Definition: the cumulative amount of costs/taxes/subsidies according to the aggregation costagg in region \mathbf{r} between years $\mathbf{y1}$ and $\mathbf{y2}$, over all timeslices. The available cost aggregations are identified by the pre-defined members of the fixed index set costagg.

Role: this variable is only created if a bound is imposed on the cumulative amount of regional costs, taxes, and/or subsidies. The variable is defined through the equation EQ BNDCST.

Bounds: Can be directly bounded by REG_CUMCST User constraints: Cannot be referred to in user constraints

5.8 VAR_CUMFLO(r,p,c,y1,y2)

Definition: the cumulative amount of flow in commodity c by process p in region r between years y1 and y2, over all timeslices. With the commodity name c='ACT (reserved system label), the variable represents the cumulative amount of process activity.

Role: this variable is only created if a bound is imposed on the cumulative amount of process flow or activity. The variable is defined through the equation EQ_CUMFLO.

Bounds: Can be directly bounded by FLO_CUM / ACT_CUM User constraints: Can be directly referred to by UC_CUMFLO/UC_CUMACT

5.9 VAR_DNCAP(r,t,p,u) / VAR_SNCAP(r,t,p)

Definition: VAR_DNCAP is only used for processes selected by the user as being discrete, i.e. for which the new capacity in period t may only be equal to one of a set of discrete sizes, specified by the user. For such processes, VAR_DNCAP is a binary decision variable equal to 1 if the investment is equal to size 'u' and 0 otherwise. Thanks to an additional constraint, only one of the various potential sizes allowed for the investment at period t is indeed allowed.

VAR_SNCAP is only used for processes selected by the user as having semicontinuous amounts of new capacity, i.e. for which new capacity in period t may only be zero or between positive lower and upper bounds specified by the user.

Role: useful to mathematically express the fact that investment in process \mathbf{p} at period \mathbf{t} may only be done in discrete or semi-continuous sizes. See equation EQ DSCNCAP in Chapter 6.

Bounds: Direct bounding not available, indirectly by NCAP_BND User constraints: Not available

5.10 VAR_DRCAP(r,v,t,p,j)

Definition: this variable is used only for processes selected by the user as having discrete early capacity retirements, i.e. for which the retirement at period t may only be a multiple of a block size, specified by the user. For such processes, VAR_DRCAP is an integer decision variable equal to the number of blocks retired.

Role: needed for mathematically expressing the fact that early retirement in capacity of process \mathbf{p} at period \mathbf{t} may only be done in discrete amounts. See equation EQ DSCRET in Chapter 6.

Bounds: Direct bounding not available, indirectly by RCAP_BND User constraints: Not available

5.11 VAR_ELAST(r,t,c,s,j,l)

Definition: these variables are defined whenever a demand is declared to be price elastic. These variables are indexed by **j**, where **j** runs over the number of steps used for discretizing the demand curve of commodity \mathbf{c} (\mathbf{c} = energy service only). The **j**th variable stands for the portion of the demand that lies within discretization interval **j**, on side *l* (*l* indicates either increase or decrease of demand w.r.t. the reference case demand). Each ELAST variable is bounded upward via virtual equation EQ BNDELAS.

Role: Each elastic demand is expressed as the sum of these variables. In the objective function, these variables are used to bear the cost of demand losses as explained in Part I, Chapter 4.

Bounds: Direct bounding not available, indirectly by COM_VOC/COM_STEP User constraints: Not available
5.12 VAR_FLO(r,v,t,p,c,s)

Definition: these variables stand for the individual commodity flows in and out of a process. If the process is not vintaged, the vintage index \mathbf{v} is by convention set equal to the period index \mathbf{t} .

Role: The flow variables are the fundamental quantities defining the detailed operation of a process. They are used to define the activity of a process (VAR_ACT) in a user chosen manner. They are also essential for expressing various constraints that balance the flows of a commodity, or that control the flexibility of processes.

Bounds: Can be directly bounded by FLO_BND User constraints: Can be directly referred to by UC_FLO

5.13 VAR_IRE(r,v,t,p,c,s,ie)

Definition: the inter-regional exchange variable (i=IMPort, e=EXPort) that tracks import (ie=i) or export (ie=e) of a commodity between region \mathbf{r} and other regions. The region(s) \mathbf{r} ' trading with \mathbf{r} is (are) not specified via this variable, but rather via the process(es) \mathbf{p} through which the import/export is accomplished. The topology set top_ire(r,c,r',c',p) of an exchange process indicates the (single) region \mathbf{r} ' with which region \mathbf{r} is trading commodity \mathbf{c} (which may have a different name \mathbf{c} ' in region \mathbf{r} '). Each trade process may trade more than one commodity. Otherwise, VAR_IRE operates in a manner similar to VAR_FLO for conventional processes. An option exists for trading with an external region that is not modeled explicitly (exogenous trading). If the process is not vintaged, the vintage index \mathbf{v} is by convention set equal to the period index \mathbf{t} .

Role: the role of an IRE variable is to embody the amount of a commodity in or out of a trading process.

Bounds: Can be bounded by IRE_BND (directly for bilateral trade) User constraints: Can be directly referred to by UC IRE

5.14 VAR_NCAP(r,v,p)

Definition: the amount of new capacity (or what has traditionally been called "investment" in new capacity, or capacity build-up) at period **v**. As will be explained in Section 6.2.2, VAR_NCAP represents the total investment in technology **p** at period **v** only when ILED+TLIFE $\geq D(v)$, where D(v) is the period length. And, as discussed further in that Section, when ILED+TLIFE $\leq D(v)$, the model assumes that the investment is repeated as many times as necessary within the period so that the life of the last repetition is beyond the end of period **v**. In this case VAR_NCAP represents the capacity level of the single investments. Figure 1 illustrates a case where the investment is made twice in period **v** (and some capacity still remains after period **v**). The average capacity in period **v** resulting from the investment VAR_NCAP(v) is less than VAR_NCAP(v), due to the delay ILED (it is equal to VAR_NCAP(v)* D(v)/TLIFE). The average capacity in period **v**+1 due to VAR_NCAP(v) is also less than VAR_NCAP(v) because the end of life of the second round of investment occurs before the end of period **v**. In these adjustments are made in every equation involving VAR_NCAP by the internal parameter COEF_CPT.



Figure 1: Example of a repeated investment in same period

Role: The new capacity (i.e. investment) variables are fundamental in defining the investment decisions, and many other quantities derived from it (for instance process capacities). They play a key role in the model structure and intervene in the majority of constraints. They are notably used in equations that define the conservation of capacity and those that tie the activity of a process to its capacity. The omnipresence of VAR_NACP is in part due to the fact that the VAR_CAP variable is not always defined in TIMES, by design. Note that residual capacity, or capacity in place prior to the initial model year, is handled as a constant in place of VAR_NCAP given by the input parameter NCAP_PASTI(y), which describes the investment made prior to the first period in the pastyear **y**.

Bounds: Can be directly bounded by NCAP_BND User constraints: Can be directly referred to by UC_NCAP

5.15 VAR_OBJ(y₀) and related variables

Definition: equal to the objective function of the TIMES LP, i.e. the total cost of all regions, discounted to year y₀.

Role: this is the quantity that is minimized by the TIMES optimizer.

Remark: The next 10 'variables' do not directly correspond to GAMS variables. They are used in the documentation (especially Section 6.2) as convenient intermediate placeholders that capture certain portions of the cost objective function. The reader is invited to look at Section 6.2 for detailed explanations on how these various costs enter the composition of the objective function. Most of these 'variables' are defined as reporting parameters that are made available to the VEDA-BE results analyser, as shown in Section 3.3.

5.15.1 VAR_OBJR(r, y₀)

Definition: equal to the sum of the various pieces of the total cost of region \mathbf{r} discounted to year \mathbf{y}_0 .

Role: this is not a true variable in the GAMS code. It is used only as a convenient placeholder for writing the corresponding portion of the objective function in this documentation. It may also be reported in VEDA-BE.

5.15.2 INVCOST(r,y)

Definition: equal to the portion of the cost objective for year \mathbf{y} , region \mathbf{r} , that corresponds to investments.

Role: it is used mainly as a convenient placeholder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.15.3 INVTAXSUB(r,y)

Definition: equal to the portion of the cost objective for year y, region r, that corresponds to investment taxes and subsidies.

Role: it is used mainly as a convenient placeholder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.15.4 INVDECOM(r,y)

Definition: equal to the portion of the cost objective for year **y**, region **r**, that corresponds to capital costs linked to decommissioning of a process.

Role: it is used mainly as a convenient placeholder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.15.5 FIXCOST(r,y)

Definition: equal to the portion of the cost objective for year \mathbf{y} , region \mathbf{r} , that corresponds to fixed annual costs.

Role: it is used mainly as a convenient placeholder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.15.6 FIXTAXSUB(r,y)

Definition: equal to the portion of the cost objective for year y, region r, that corresponds to taxes and subsidies attached to fixed annual costs.

Role: it is used mainly as a convenient placeholder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.15.7 VARCOST(r,y)

Definition: equal to the portion of the cost objective for year y, region r, that corresponds to variable annual costs.

Role: it is used mainly as a convenient placeholder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.15.8 VARTAXSUB(r,y)

Definition: equal to the portion of the cost objective for year y, region r, that corresponds to variable annual taxes and subsidies.

Role: it is used mainly as a convenient place holder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.15.9 ELASTCOST(r,y)

Definition: equal to the portion of the cost objective for year y, region r, that corresponds to the cost incurred when demands are reduced due to their price elasticity. **Role:** it is used mainly as a convenient placeholder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.15.10 LATEREVENUES(r,y)

Definition: equal to the portion of the cost objective for year \mathbf{y} , region \mathbf{r} , that corresponds to certain late revenues from the recycling of materials from dismantled processes that occur after the end-of-horizon.

Role: this is not a true variable in the GAMS code. It is used only as a convenient placeholder for writing the corresponding portion of the objective function in this documentation. It may also be reported in VEDA-BE as a convenient replacement for the sum of the components of the total cost.

5.15.11 SALVAGE(r,y0)

Definition: equal to the portion of the cost objective for region r, that corresponds to the salvage value of investments and other one-time costs. It is discounted to some base year y_0

Role: it is used mainly as a convenient placeholder for writing the corresponding portion of the objective function. It may also be reported in VEDA-BE.

5.16 VAR_RCAP(r,v,t,p)

Definition: this variable is used only for processes selected by the user as having early capacity retirements. For such processes, VAR_RCAP represents the amount of capacity of vintage v retired in period t.

Role: introduced for supporting bounds on the amount of retired capacity of process \mathbf{p} and vintage \mathbf{v} in period \mathbf{t} .

Bounds: Can be directly bounded by RCAP_BND User constraints: Not available

5.17 VAR_SCAP(r,v,t,p)

Definition: this variable is used only for processes selected by the user as having early capacity retirements. For such processes, VAR_SCAP represents the cumulative amount of capacity of vintage v retired in periods $tt \le t$.

Role: needed in several TIMES equations for adjusting the overall available capacity of process \mathbf{p} at period \mathbf{t} according to the amount of capacity already retired.

Bounds: Not directly available; indirectly by RCAP_BND / CAP_BND User constraints: Not available

5.18 VAR_SIN/SOUT(r,v,t,p,c,s)

Definition: flow entering/leaving at period **t** a storage process **p**, storing commodity **c**. The process may be vintaged. If the process is not vintaged, the vintage index **v** is by convention set equal to the period index **t**. For storages between timeslices (**prc_stgtss**) and night-storage devices (**prc_nsttss**) the timeslice index s of the storage flows is determined by the timeslice resolution of the storage (e.g. DAYNITE for a day storage). For a storage operating between periods (**prc_stgips**), the storage flows are always on an annual level and hence the timeslice **s** is then always set to ANNUAL.

Role: to store some commodity so that it may be used in a time slice or period different from the one in which it was procured; enters the expressions for the storage constraints.

Bounds: Can be directly bounded by STGIN_BND/ STGOUT_BND User constraints: Not directly available; indirectly by using auxiliary storage flows

5.19 VAR_UPS(r,v,t,p,s,l)

Definition: amount of off-line capacity (l='N'), started-up capacity (l='UP'), shutdown capacity (l='LO'), or efficiency losses due to partial loads (l='FX') in period t process p vintage index v.

Role: used for modeling capacity dispatching, start-up costs as well as partial load efficiencies, but only when requested so by the user.

Bounds: Not available

User constraints: Not directly available; in timeslice-dynamic constraints on-line capacity can be referred by UC_CAP, using the ONLINE modifier for CAP

5.20 Variables used in User Constraints

The remaining TIMES variables are all attached to user constraints. User constraints are quite flexible, and may involve any of the usual TIMES variables. Two variants of formulating user constraints exist. In the first case a LHS expression, containing expressions involving the different TIMES variables, are bounded by a RHS constant (given by the input parameter UC_RHS(R)(T)(S)). In the second case, the constant on the RHS is replaced by a variable. The bound UC_RHS(R)(T)(S) is then applied to this variable. In the latter case, the user constraints are always generated as strict equalities, while in the first case the equation sign of the user constraint is determined by the bound type.

- Case 1 (RHS constants): <LHS expression> UC_RHS(R)(T)(S)
- Case 2 (UC variables): $\langle LHS expression \rangle = VAR UC(R)(T)(S)$

These user constraint variables are in fact redundant, but quite useful in providing streamlined expressions constraints (see Chapter 6), and allow for reporting the slack level of each UC. Morever, in the case of range constraints, they will reduce model size and the amount of input data. By setting the dollar control parameter VAR_UC to YES in the run-file, the variable based formulation is activated (second case). By default, the formulation without user constraint variables will be used, and only the marginals of the equations are reported.

Non-binding user constraints (intoduced for reporting purposes) can only be defined when the user constraint variables are used (i.e. VAR_UC == YES).

Each of the listed variables is related to a specific class of user constraint depending on whether the user constraint is created for each period, region, or time slice or only a subset of these indices. In addition, some user constraints are defined for pair of successive time periods (dynamic user constraint or growth constraint). Each variable has at least one index (representing the user constraint **uc_n** for which this variable is defined), and may have up to three additional indexes among **r**, **t**, and **s**.

5.20.1 VAR_UC(uc_n)

Variable representing the LHS expression of the user constraint EQE_UC(uc_n) summing over regions (uc_r_sum), periods (uc_t_sum) and timeslices (uc_ts_sum).

5.20.2 VAR_UCR(uc_n,r)

Variable representing the LHS expression of the user constraint EQE_UCR(r,uc_n) summing over periods (uc_tsum) and timeslices (uc_tsum) and being generated for the regions specified in uc_reach .

5.20.3 VAR_UCT(uc_n,t)

Variable representing the LHS expression of the user constraint EQE_UCT(uc_n,t) and the combined LHS-RHS expression of the user constraint EQE_UCSU(uc_n,t), summing over regions (uc_r_sum) and timeslices (uc_ts_sum) and being generated for the periods specified in uc_t_each/uc_t_succ.

5.20.4 VAR_UCRT(uc_n,r,t)

Variable representing the LHS expression of the user constraint EQE_UCRT(r,uc_n,t) and the combined LHS-RHS expression of the user constraint EQE_UCRSU(r,uc_n,t), summing over timeslices (uc_ts_sum) and being generated for the regions specified in uc_r_each and periods in uc_t_each/uc_t_succ .

5.20.5 VAR_UCTS(uc_n,t,s)

Variable representing the LHS expression of the user constraint EQE_UCTS(uc_n,t,s) and the combined LHS-RHS expression of the user constraint EQE_UCSUS(uc_n,t,s), summing over regions (uc_r_sum) and being generated for the periods specified in uc_t_each/uc_t_succ and timeslices in uc_ts_each.

5.20.6 VAR_UCRTS(uc_n,r,t,s)

Variable representing the LHS expression of the user constraint EQE_UCRTS(r,uc_n,t,s) and the combined LHS-RHS expression of the user constraint EQE_UCRSUS(r,uc_n,t,s), being generated for the regions specified in uc_r_each, the periods in uc_t_each/uc_t_succ and the timeslices in uc_ts_each.

6 Equations

This chapter is divided into four sections: the first section describes the main notational conventions adopted in writing the mathematical expressions of the entire chapter. The next two sections treat respectively the TIMES objective function and the standard linear constraints of the model. The fourth section is devoted to the facility for defining various kinds of user constraints. Additional constraints and objective function additions that are required for the Climate Module, Damage Cost and Endogenous Technology Learning options are described in Appendices A, B and C, respectively.

Each equation has a unique name and is described in a separate subsection. The equations are listed in alphabetical order in each section. Each subsection contains successively the name, list of indices, and type of the equation, the related variables and other equations, the purpose of the equation, any particular remarks applying to it, and finally the mathematical expression of the constraint or objective function.

The mathematical formulation of an equation starts with the name of the equation in the format: $EQ_XXX_{i,j,k,l}$, where XXX is a unique equation identifier, and i,j,k,..., are the equation indexes, among those described in chapter 2. Some equation names also include an index *l* controlling the sense of the equation. Next to the equation name is a *logical* condition that the equation indexes must satisfy. That condition constitutes the domain of definition of the equation. It is useful to remember that the equation is created in multiple instances, one for each combination of the equation indexes that satisfies the logical condition, and that each index in the equation's index list remains fixed in the expressions constituting each instance of the equation.

6.1 Notational conventions

We use the following mathematical symbols for the mathematical expressions and relations constituting the equations:

The conditions that apply to each equation are mathematically expressed using the \exists symbol (meaning "such that" or "only when"), followed by a logical expression involving the usual logic operators: \land (AND), \lor (OR), and NOT.

Within the mathematical expressions of the constraints, we use the usual symbols for the arithmetic operators $(+, -, \times, /, \Sigma, \text{ etc})$.

However, in order to improve the writing and legibility of all expressions, we use some simplifications of the usual mathematical notation concerning the use of multiple indexes, which we describe in the next two subsections.

6.1.1 Notation for summations

When an expression A(i,j,k,...) is summed, the summation must specify the range over which the indexes are allowed to run. Our notational conventions are as follows:

When a single index j runs over a one-dimensional set A, the usual notation is used, as in: $\sum_{j \in A} Expression_j$ where A is a single dimensional set.

When a summation must be done over a subset of a multi-dimensional set, we use a simplified notation where some of the running indexes are omitted, if they are not active for this summation.

Example: consider the 3-dimensional set *top* consisting of all quadruples {*r*,*p*,*c*,*io*} such that process *p* in region *r*, has a flow of commodity *c* with orientation *io* (see table 3 of chapter 2). If is it desired to sum an expression $A_{r,p,c,lo}$ over all commodities *c*, keeping the region (r), process (p) and orientation (io) fixed respectively at r_l , p_l and 'IN', we will write, by a slight abuse of notation: $\sum_{cetop(r), p_l, (2N')} A(r_l, p_l, c, 'IN')$, or even more simply:

 $\sum_{c \in lop} A(r_1, p_1, c, 'IN')$, if the context is unambiguous. Either of these notations clearly

indicates that r, p and io are fixed and that the only active running index is c.

(The traditional mathematical notation would have been: $\sum_{(r_1, p_1, c_1, D') \neq top} A(r_1, p, c_1, 'DN'),$

but this may have hidden the fact that c is the only running index active in the sum).

6.1.2 Notation for logical conditions

We use similar simplifying notation in writing the logical conditions of each equation. A logical condition usually expresses that some parameter exists (i.e. has been given a value by the user), and/or that some indexes are restricted to certain subsets.

A typical example of the former would be written as: $\exists ACTBND_{r,t,p,s,bd}$, which reads: "the user has defined an activity bound for process *p* in region *r*, time-period *t*, timeslice *s* and sense *bd*". The indexes may sometimes be omitted, when they are the same as those attached to the equation name.

A typical example of the latter is the first condition for equation $EQ_ACTFLO_{r,v,t,p,s}$ (see section 6.3.4), which we write simply as: **rtp_vintyr**, which is short for: $\{r, v, t, p\} \in \mathbf{rtp}$ _vintyr, with the meaning that "some capacity of process p in region r, created at period v, exists at period t". Again here, the indices have been omitted from the notation since they are already listed as indices of the equation name.

6.1.3 Using Indicator functions in arithmetic expressions

There are situations where an expression A is either equal to B or to C, depending on whether a certain condition holds or not, i.e.:

A = B if Cond A = C if NOT Cond

This may also be written as:

 $A = B \times (Cond) + C \times (NOTCond)$

where it is understood that the notation (Cond) is the *indicator function* of the logical condition, i.e. (Cond)=1 if Cond holds, and 0 if not.

This notation often makes equations more legible and compact. A good example appears in EQ_CAPACT.

6.2 Objective function EQ_OBJ

```
Equation EQ_OBJ
```

Indices: region (r); state of the world (w); process (p); time-slice (s); and perhaps others ...

Type: = Non Binding (MIN)

Related Variables: All

Purpose: the objective function is the criterion that is minimized by the TIMES model. It represents the total discounted cost of the entire, possibly multi-regional system over the selected planning horizon. It is also equal to the negative of the discounted total surplus (plus a constant), as discussed in PART I, chapters 3 and 4.

6.2.1 Introduction and notation

The TIMES objective function includes a number of innovations compared to those of more traditional energy models such as MARKAL, EFOM, MESSAGE, etc. The main design choices are as follows:

- The objective function may be thought of as the discounted sum of *net annual costs* (i.e. costs minus revenues), as opposed to *net period costs*³⁹. Note that some costs and revenues are incurred after the end of horizon (EOH). This is the case for instance for some investment payments and more frequently for payments and revenues attached to decommissioning activities. The past investments (made before the first year of the horizon) may also have payments within horizon years (and even after EOH!) These are also reflected in the objective function. However, it should be clear that such payments are shown in OBJ only for reporting purposes, since such payments are entirely *sunk*, i.e. they are not affected by the model's decisions.
- The model uses a general discount rate d(y) (year dependent), as well as technology specific discount rates d_s(t) (period dependent). The former is used to: a) discount fixed and variable operating costs, and b) discount investment cost payments from the point of time when the investment actually occurs to the base year chosen for the computation of the present value of the total system cost. The latter are used only to calculate the annual payments resulting from a lump-sum

³⁹ The actual implementation of OBJ in the GAMS program is different from the one described in the documentation, since the annualizing of the various cost components is not performed in the GAMS code of the OBJ equation, but rather in the reporting section of the program, for improved code performance. However, despite the simplification, the GAMS code results in an objective function that is fully equivalent to the one in this documentation.

investment in some year. Thus, the only place where $d_s(t)$ intervenes is to compute the Capital Recovery Factors (*CRF*) discussed further down.

For convenience, we summarize below the notation which is more especially used in the objective function formulation (see Section 6.1 for general notes on the notation).

6.2.1.1 Notation relative to time

MILESTONEYEARS: th	e set of	all	milestone	years	(by	convention:	middle	years,	see
below M(t))								

PASTYEARS: Set of years p	rior to start of horizon, :	for which there is	a past investment
(after interpol	ation of user data).		

MODE	ELYEAI	RS: any year within the model's horizon
FUTU	REYEA	IRS: set of years posterior to EOH
YEAR	S	set of years before during and after planning horizon
t		any member of MILESTONEYEARS or PASTYEARS. By convention, a
		period t is represented by its middle year (see below M(t)). This convention
		can be changed without altering the expressions in this document.
B(t)	:	the first year of the period represented by t
E(t)	:	the last year of the period represented by t
D(t)	:	the number of years in period t . By default, $D(t)=1$ for all past
		years. Thus, $D(t)=E(t)-B(t)+1$
M(t):		the "middle" year or milestone year of period t . Since period n may have
		an even or an odd number of years, M(t) is not always exactly centered at
		the middle of the period. It is defined as follows: $M(t) = [B(t)+(D(t)-1)/2],$
		where $[x]$ indicates the largest integer less than or equal to x. For example,
		period from 2011 to 2020 includes 10 years, and its "middle year" is
		[2011+4.5] or 2015 (slightly left of the middle), whereas the period from
		2001 to 2015 has 15 years, and its "middle year" is : [2001+7] or 2008
		(i.e. the true middle in this example)
v	:	running year, ranging over <i>MODELYEARS</i> , from B_{θ} to <i>EOH</i> .
k	:	dummy running index of any year, even outside horizon
V:		running index for a year, used when it represents a vintage year for some
		investment.
v(p)		vintage of process p (defined only if p is vintaged)
B_{θ}	:	initial year (the single year of first period of the model run)
EOH	:	Last year in horizon for a given model run.

Similarly, by a slight abuse of notation, the above entities are extended as follows, when the argument is a particular year, rather than a model year:

- M(y) : "middle year" of the period containing year y (instead of M(T(y)))
- D(y) : number of years of the period containing year y (instead of D(T(y)))

6.2.1.2 Other notation

<i>d(y)</i> :	general (social) discount rate (time dependent, although not shown in notation)
r(y) :	general discount factor: $r(y)=1/(1+d(y))$ (time dependent, although not shown in notation)
$d_s(t)$:	technology specific discount rate (model year dependent)
$r_s(t)$:	technology specific discount factor: $r_s(t) = 1/(1+d_s(t))$
DISC(y,z):	Value, discounted to the beginning of year z, of a \$1 payment made at
	beginning of year y, using general discount factor. $DISC(y,z) = \prod_{u=z \text{ to } y=1} r(u)$
CRF _s (t):	Capital recovery factor, using a (technology specific) discount rate and an
	economic life appropriate to the payment being considered. This quantity
	is used to replace an investment cost by a series of annual payments
	spread over some span of time $CRF_s = \{1 - r_s(t)\}/\{1 - r_s(t)\}^{ELIFE}\}^{40}$. Note that a
	CRF using the general discount rate is also defined and used in the
	SALVAGE portion of the objective function.
OBJ(z):	Total system cost, discounted to the beginning of year z
INDIC(x):	1 if logical expression x is true, θ if not
$\langle E \rangle$	is the smallest integer larger than of equal to ${\cal E}$
6.2.1.3 <u>Remi</u>	nder of some technology attribute names (each indexed by t)
TLIFE	Technical life of a technology
ELIFE	Economic life of a technology, i.e. period over which investment
	payments are spread (default = $TLIFE$)
DLAG	Lag after end of technical life, after which decommissioning may start
DLIFE	Duration of decommissioning for processes with <i>ILED>0</i> , (otherwise =1)
DELIF	Economic life for decommissioning purposes (default DLIFE).
ILED	Lead-time for the construction of a process. TLIFE starts after the end of
	ILED
ILED _{Min}	=Min {1/10 * D(t), 1/10 * TLIFE} This threshold serves to distinguish small
	from large projects; it triggers a different treatment of investment timing.
6.2.1.4 <u>Disco</u>	punting options

There are alternate discounting methods in TIMES. The default method is to assume that all payments occur at the beginning of some year. Alternate methods (activated by a switch, see PART III) assume that investments are incurred at the beginning of some year, but that all annual (or annualized) payments occur at the middle or at the end of the corresponding year. Section 0 explains the different methods.

 $^{^{40}}$ This is the default definition adopted for *CRF*, corresponding to beginning-of-year discounting. For other discounting options, see Section 0.

6.2.1.5 Components of the Objective function

The objective function is the sum of all regional objectives, all of them discounted to the same user-selected base year, as shown in equation (A) below

$$EQ_OBJ(z) \qquad \Rightarrow z \in ALLYEARS$$

$$VAR_OBJ(z) = \sum_{r \in REG} REG_OBJ(z, r) \qquad (A)$$

Each regional objective OBJ(z,r) is decomposed into the sum of nine components, to facilitate exposition, as per expression (B) below.

$$EQ_OBJ(z,r) \qquad \Rightarrow z \in ALLYEARS, r \in REG$$

$$REG_OBJ(z,r) = \sum_{y \in (-\infty, +\infty)} DISC(y,z) \times \begin{cases} INVCOST(y) + INVTAXSUB(y) + INVDECOM(y) + FIXTAXSUB(y) + SURVCOST(y) + FIXTAXSUB(y) + SURVCOST(y) + VARCOST(y) + VARCOST(y) + VARCOST(y) + VARTAXSUB(y) + ELASTCOST(y) - LATEREVENUES(y) \\ -SALVAGE(z) \end{cases}$$
(B)

The regional index r is omitted from the nine components for simplicity of notation.

The first and second terms are linked to investment costs. The third term is linked to decommissioning capital costs, the fourth and fifth terms to fixed annual costs, the seventh and eighth terms to all variable costs (costs proportional to some activity), and the ninth to demand loss costs. The tenth cost (actually a revenue) accounts for commodity recycling occurring after EOH, and the eleventh term is the salvage value of all capital costs of technologies whose life extends beyond EOH. The 11 components are presented in the nine subsections 6.2.2 to 6.2.10.

6.2.2 Investment costs: INVCOST(y)

This subsection presents the components of the objective function related to investment costs, which occur in the year an investment is decided and/or during the construction lead-time of a facility.

Remarks

a) The investment cost specified by using the input attribute NCAP_COST should be the overnight investment cost (excluding any interests paid during construction) whenever the construction lead time is explicitly modeled (i.e. cases 2 are used, see below). In such a case, the interests during construction are endogenously calculated by the model itself, as will be apparent in the sequel. If no lead-time is specified (and

thus cases 1 are used), the full cost of investments should be used (including interests during construction, if any)⁴¹.

- b) Each individual investment physically occurring in year k, results in a stream of annual payments spread over several years in the future. The stream starts in year k and covers years k, k+1, ..., k+ELIFE-1, where ELIFE is the economic life of the technology. Each yearly payment is equal to a fraction CRF of the investment cost (CRF = Capital Recovery Factor). Note that if the technology discount rate is equal to the general discount rate, then the stream of ELIFE yearly payments is equivalent to a single payment of the whole investment cost located at year k, inasmuch as both have the same discounted present value. If however the technology's discount rate is chosen different from the general one, then the stream of payments has a different present value than the lump sum at year k. It is the user's responsibility to choose technology dependent discount rates, and therefore to decide to alter the effective value of investment costs.
- c) In addition to spreading the payments resulting from investment costs, a major TIMES refinement is that the physical investment itself does not occur in a single year, but rather as a series of annual increments. For instance, if the model invests 3 GW of electric capacity in a period extending from 2011 to 2020, the physical capacity increase may be delayed and/or may be spread over several years. The exact way the delaying and spreading are effected depends on several conditions, which are specified further down as four separate cases, and which are functions both of the nature of the technology and of the length of the period in which the investment takes place relative to the technology's technical life. The spreading of investments and the spreading of payments described in the previous paragraph help guarantee a smooth trajectory for most investment payments, a more realistic representation than what happens in other models. The Case 1.a example given below shows a case where the physical investment is spread over 3 years.
- d) The above two remarks entail that payments of investment costs may well extend beyond the horizon. We shall also see that some investment payments occur in years prior to the beginning of the planning horizon (cases 1 only).
- Taxes and subsidies on investments are treated exactly as investment costs in the objective function.
- f) Since the model has the capability to represent *sunk* materials and energy carriers (i.e. those embedded in a technology at construction time, such as the uranium core of a nuclear reactor, or the steel imbedded in a car), these sunk commodities have an impact on cost. Two possibilities exist: if the material is one whose production is explicitly modeled in the RES, then there is no need to indicate the cost corresponding to the sunk material, which will be implicitly accounted for by the model just like any other flow. If on the other hand the material is not specifically modeled in the RES, then the cost of the sunk material should be included in the technology's investment cost, and will then be handled exactly as investment costs.

⁴¹ Ideally, it would be desirable that cases 1 be used only for those investments that have no lead time (and thus no interest during construction). However, if cases 1 are employed even for projects with significant IDC's, these should have their IDC included in the investment cost.



The four investment cases

As mentioned above, the timing of the various types of payments and revenues is made as realistic and as smooth as possible. All investment decisions result in increments and/or decrements in the capacity of a process, at various times. These increments or decrements may occur, in some cases, in one large lump, for instance in the case of a large project (hydroelectric plant, aluminum plant, etc.), and, in other cases, in small additions or subtractions to capacity (e.g. buying or retiring cars, or heating devices). Depending on which case is considered, the assumption regarding the corresponding streams of payments (or revenues) differs markedly. Therefore, the distinction between small and large projects (called cases 1 and 2 below) will be crucial for writing the capital cost components of the objective function. A second distinction comes from the relative length of a project's technical life vs. that of the period when the investment occurs. Namely, if the life of an investment is less than the length of the period, then it is clear that the investment must be repeated all along the period. This is not so when the technical life extends beyond the period's end. Altogether, these two distinctions result in four mutually exclusive cases, each of which is treated separately. In what follows, we present the mathematical expression for the INVCOST component and one graphical example for each case.

Case 1.a If $ILED_t \leq ILED_{MDt,t}$ and $TLIFE_t + ILED_t \geq D(t)$ (Small divisible projects, non-repetitive, progressive investment in period)

Here, we make what appears to be the most natural assumption, i.e. that the investment occurs in small yearly increments spread linearly over D(t) years. Precisely, the capacity additions start at year M(t)-D(t)+1, and end at year M(t), which means that payments start earlier than the beginning of the period, and end at the middle of the period, see example. This seems a more realistic compromise than starting the payments at the beginning of the period and stopping them at the end, since that would mean that during the whole period, the paid for capacity would actually not be sufficient to cover the capacity selected by the model for that period.



 $\{M(t) - D(t) + 1, M(t) + ELIFE_t - 1\}$

(I.1.a)

Comments: The summand represents the payment effected in year y, due to the investment increment that occurred in year v (recall that investment payments are spread over *ELIFE*). The summand consists of three factors: the first is the amount of investment in year v, the second is the capital recovery factor, and the third is the unit investment cost.

The outer summation is over all periods (note that periods later than T(y) are relevant, because when y falls near the end of a period, the next period's investment may have already started). The inner summation is over a span of D(t) centered at B(t), but truncated at year y. Also, the lower summation bound ensures that an investment increment which occurred in year v has a payment in year y only if y and v are less than *ELIFE* years apart.

Case 1.b if $ILED_t \leq ILED_{Min,t}$ and $TLIFE_t + ILED < D(t)$ Small projects, repeated investments in period

Note that in this case the investment is repeated as many times as necessary to cover the period length (see figure). In this case, the assumption that the investment is spread over D(t) years is not realistic. It is much more natural to spread the investment over the technical life of the process being invested in, because this ensures a smooth, constant stream of small investments during the whole period (any other choice of the time span over which investment is spread, would lead to an uneven stream of incremental investments). The number of re-investments in the period is called C, and is easily computed so as to cover the entire period. As a result of this discussion, the first investment cycle starts at year $\langle B(t) - TLIFE_t/2 \rangle$ (meaning the smallest integer not less than the operand), and ends TLIFE years later, when the second cycle starts, etc, as many times as necessary to cover the entire period. The last cycle extends over the next period(s), and that is taken into account in the capacity transfer equations of the model. As before, each capacity increment results in a stream of ELIFE payments at years v, v+1, etc.



INVCOST(y) =

$$\sum_{t \in \textit{MILESTONE}} \textit{INDIC}(1.b) \times \sum_{v=\textit{Max}(\langle \textit{B}(t) - \textit{TLIFE}_t/2 \rangle, v - \textit{ELIFE}_t+1)}^{\textit{Min}(v,\langle \textit{B}(t) - \textit{TLIFE}_t/2 \rangle, v - \textit{ELIFE}_t-1)} \frac{\textit{VAR}_\textit{NCAP}_t}{\textit{TLIFE}_t} \times \textit{CRF}_s \times \textit{NCAP}_\textit{COST}_v$$

Relevant range for y:

$$\left\langle \left\langle B(t) - TLIFE_t / 2 \right\rangle, \left\langle B(t) - TLIFE_t / 2 \right\rangle + C \times TLIFE_t + ELIFE_t - 2 \right\rangle$$
(I.1.b)

Comments: the expression is similar to that in case **1.a.**, except that i) the investment is spread over the technical life rather than the period length, and ii) the investment cycle is repeated more than once.

Case 2.a: $ILED_t > ILED_{Min,t}$ and $ILED_t + TLIFE_t \ge D(t)$ (Large, indivisible projects, unrepeated investment in period)

Here, it is assumed that construction is spread over the lead-time (a very realistic assumption for large projects), and capacity becomes available at the end of the lead time, **in a lump quantity** (see figure).



Useful Range for y: {B(t), $B(t) + ILED_t + ELIFE_t - 2$ }

(I.2.a)

Comment: the main difference with case I.1.a) is that the investment's construction starts at year B(t) and ends at year $B(t)+ILED_{t-1}$ (see example). As before, payments for each year's construction spread over *ELIFE* years.

Case 2.b: *ILED* > *ILED*_{*Min,t*} and *TLIFE*_{*t*} + *ILED*_{*t*} < D(t)(Large, indivisible Projects, repeated investments in period)



This case is similar to case I.2.a, but the investment is repeated more than once over the period, each cycle being *TLIFE* years long. As in case I.2.a, each construction is spread over one lead time, *ILED*. In this case, the exact pattern of yearly investments is complex, so that we have to use an algorithm instead of a closed form summation.

<u>ALGORITHM</u> (Output: the vector of payments $P_t(y)$ at each year y, due to VAR_NCAP_t)

Step 0: Initialization (*NI(u)* represents the amount of new investment made in year *u*)

 $NI_t(u) := 0$ $\forall B(t) \le u \le B(t) + ILED_t + (C-1) \times TLIFE_t - 1$

Step 1: Compute number of repetitions of investment

$$C = \left\langle \frac{D(t) - ILED_t}{TLIFE_t} \right\rangle$$

Step 2: for each year *u* in range:

 $B(t) \le u \le B(t) + ILED_t + (C-1) \cdot TLIFE_t - 1$

Compute:

$$\begin{aligned} For I &= 1 \text{ to } C \\ For u &= B(t) + (I-1) \cdot TLIFE_t \text{ to } B(t) + (I-1) \cdot TLIFE_t + ILED_t - 1 \\ NI_t(u) &:= NI_t(u) + \frac{NCAP - COST_{B(t) + (I-1) \times TLIFE_t + ILED_t}}{ILED_t} \end{aligned}$$

Next u

Next I

<u>Step 3:</u> Compute payments incurred in year *y*, and resulting from variable VAR_NCAP_t For each *y* in range:

$$B(t) \le y \le B(t) + (C-1) \cdot TLIFE_t + ILED_t + ELIFE_t - 2$$

Compute:

$$P_t(y) = \sum_{u=Max\{B(t), y=ELIFE, +1\}}^{y} NI_t(u) \times VAR _NCAP_t \times CRF_s$$

END ALGORITHM

$$INVCOST(y) = \sum_{t \in MILESTONES, t \leq T(y)} INDIC(2.b) \times P_t(y)$$

6.2.3 Taxes and subsidies on investments

We assume that taxes/subsidies on investments occur at precisely the same time as the investment. Therefore, the expressions INVTAXSUB(y) for taxes/subsidies are identical to those for investment costs, with $NCAP_COST$ replaced by: $(NCAP_ITAX - NCAP_ISUB)$.

151

(I.2.b)

6.2.4 Decommissioning (dismantling) capital costs: INVDECOM(y)

Remarks

- a) Decommissioning physically occurs after the end-of-life of the investment, and may be delayed by an optional lag period *DLAG* (e.g. a "cooling off" of the process before dismantling may take place). The decommissioning costs follow the same patterns and rules as those for investment costs. In particular, the same four cases that were defined for investment costs are still applicable.
- b) The same principles preside over the timing of payments of decommissioning costs as were defined for investment costs, namely, the decomposition of payments into a stream of payments extending over the economic life of decommissioning, *DELIF*.
- c) At decommissioning time, the recuperation of embedded materials is allowed by the model. This is treated as explained for investment costs, i.e. either as an explicit commodity flow, or as a credit (revenue) subtracted by the user from the decommissioning cost.
- g) Decommissioning activities may also receive taxes or subsidies which are proportional to the corresponding decommissioning cost.

 $EQ_COSTDECOM(y) \rightarrow y \in ALLYEARS$

Case 1.a) If $ILED_t \leq ILED_{MDt,t}$ and $TLIFE_t + ILED_t \geq D(t)$ (Small divisible projects, non-repetitive, progressive investment in period)

In this case, decommissioning occurs exactly TLIFE+DLAG years after investment. For small projects (cases **1.a** and **1.b**), it is assumed that decommissioning takes exactly one year, and also that its cost is paid that same year (this is the same as saying that DLIFE=DELIF=1). Any user-defined DLIFE/DELIF is in this case thus ignored. This is a normal assumption for small projects. As shown in the example below, also payments made at year y come from investments made at period T(y) or earlier. Hence the summation stops at T(y).

$$\begin{split} INVDECOM(y) &= \\ \sum_{\substack{t \in MILESTONES \ \cup \ PASTYFEARS}} INDIC(1.a) \times \left(\frac{VAR_NCAP_t}{D(t)} + NCAP_PASTI_t \right) \times NCAP_DCOST_{y-TLIFE_t} \\ \times \begin{cases} 1 & \text{if} \quad M(t) - D(t) + 1 + TLIFE_t + DLAG_t \le y \le M(t) + TLIFE_t + DLAG_t \\ 0 & otherwise \end{cases}$$
(III.1.a)

Comment: Note that the cost attribute is indexed at the year when the investment started to operate. We have adopted this convention throughout the objective function.



Case 1.b) if $ILED_t \leq ILED_{Min,t}$ and $TLIFE_t + ILED < D(t)$ (Small projects, repeated investments in period)

This cost expression is similar to I.1.b, but with payments shifted to the right by TLIFE (see example). The inner summation disappears because of the assumption that DELIF=1. Note also that past investments have no effect in this case, because this case does not arise when D(t)=1, which is always the case for past periods.

$$INVDECOM(y) = \sum_{\substack{t \in MILESTONES\\ t \leq T(y)}} INDIC(1.b) \times \left(\frac{VAR _NCAP_t}{TLIFE_t}\right) \times NCAP _DCOST_{y-TLIFE_t}$$
$$\times \begin{cases} 1 \text{ if } B(t) + \left[\frac{TLIFE_t}{2}\right] \leq y \leq B(t) + \left[\frac{TLIFE_t}{2}\right] + C \cdot TLIFE_t - 1 \\ 0 \text{ otherwise} \end{cases}$$

where $C = \left\langle \frac{D(t)}{TLIFE_t} \right\rangle$

(III.1.b)





In this situation, it is assumed that decommissioning of the plant occurs over a period of time called DLIFE, starting after the end of the technical process life *plus a time* DLAG (see example). DLAG is needed e.g. for a reactor to "cool down" or for any other reason. Furthermore, the payments are now spread over DELIF, which may be larger than one year.

INVDECOM(y) =

 $\sum_{\substack{t \in MILESTONES\\t \leq T(y)}} INDIC(2.a) \times \sum_{\substack{k = Max \{B(t) + ILED_t + TLIFE_t + DLAG_t, y - DELIE_t + 1\}}}^{Mm(y,B(t) + ILED_t + TLIFE_t + DLAG_t, DLAG_t, z - DLAG_t, z$

$$+ \sum_{t \in PASTTEARS} INDIC(2.a) \times \sum_{k=Max(t+TLIFE_t+DLAG_t, y-DELIF_t+1)}^{Mot(y,t+TLIFE_t+DLAG_t+DLIFE_t-1)} \left(\frac{NCAP_PASTI_t}{DLIFE_t} \right) \times CRF_s \times NCAP_DCOST_t$$

(III.2.a)

Useful Range for y: $\{B(t) + ILED_t + TLIFE_t + DLAG_t - 1, same + DELIF_t - 1\}$



Case 2.b: $ILED_t > ILED_{Min,t}$ and $TLIFE_t + ILED_t < D(t)$ (Big projects, repeated investments in period)

Here too, the decommissioning takes place over *DLIFE*, but now, contrary to case 2.a, the process is repeated more than once in the period. The last investment has life extending over following periods, as in all similar cases. The resulting stream of yearly payments is complex, and therefore, we are forced to use an algorithm rather than a closed form summation. See also example below.





 $INVDECOM(y) = \sum_{t \in MILESTONES, t \leq T(y)} INDIC(III.2.b) \times P_t(y) \times VAR_NCAP_t \times CRF$



6.2.5 Fixed annual costs: FIXCOST(y), SURVCOST(y)

The fixed annual costs are assumed to be paid in the same year as the actual operation of the facility. However, the spreading of the investment described in subsection 5.1.1 results in a tapering in and a tapering out of these costs. Taxes and subsidies on fixed annual costs are also accepted by the model.

There are two types of fixed annual costs, FIXCOST(y), which is incurred each year for each unit of capacity still operating, and SURVCOST(y), which is incurred each year for each unit of capacity in its DLAG state (this is a cost incurred for surveillance of the facility during the lag time before its demolition). Again here, the same classification of cases is adopted as in previous subsections on capital costs. Note that by assumption, SURVCOST(y) occurs only in cases 2. DLAG is allowed to be positive even in case 1a, but that in this case the surveillance costs are assumed to be negligible. Finally, note that FIXCOST(y) need be computed only for years y within the planning horizon, whereas SURVCOST(y) may exist for years beyond the horizon

Remark on early retirements:

In TIMES, any capacity may also be retired before the end of its technical lifetime, if so-called early retirements are enabled for a process. In such cases, the plant is assumed to be irrevocably shut down, and therefore fixed O&M costs would no longer occur. This situation is not taken into account in the standard formulations given below, but it has been taken into account in the model generator. To see that the expressions for the fixed annual costs, taxes and subsidies could be easily adjusted for early retirements, consider the standard expressions for FIXCOST(y), which can all be written as follows.

$$FIXCOST(r, y) = \sum_{(r, v, p) \in r_{\mathbf{p}}} \left(\frac{VAR _ NCAP_{r, v, p} (\Rightarrow \mathbf{t}_{v})}{+ NCAP _ PASTI_{r, v, p}} \right) \times CF_{r, v, p, y}$$

Here, $CF_{r,v,p,y}$ is the compound fixed cost coefficient for each capacity vintage in year y, as obtained from the original expressions for FIXCOST(y). Recalling that fixed costs are accounted only within the model horizon, these expressions can be adjusted as follows:

$$FIXCOST^{\circ}(r, y) = \sum_{\substack{(r, y, p) \in \mathbf{rip} \\ prc_reap_{rp} \\ periodyr_y}} \begin{pmatrix} VAR_NCAP_{r, y, p} & (\mathbf{i} \mathbf{t}_y) \\ + NCAP_PASTI_{r, y, p} \\ - \sum_{\substack{prc_reap_{rp} \\ periodyr_y}} VAR_SCAP_{r, y, t, p} \end{pmatrix} \times CF_{r, y, p, y}$$

As one can see, the expressions for FIXCOST(r,y) can be augmented in a straightforward manner, obtaining the expressions $FIXCOST^{\circ}(r,y)$ that take into account early capacity retirements of each vintage, represented by the $VAR_SCAP_{r,v,t,p}$ variables.

Case 1.a) If $ILED_t \leq ILED_{Min,t}$ and $TLIFE_t + ILED_t \geq D(t)$ (Small projects, single investment in period)

 $EQ_FIXCOST(y)$, $y \leq EOH$

The figure of the example shows that payments made in year y may come from investments made at periods before T(y), at T(y) itself, or at periods after T(y). Note that the cost attribute is multiplied by two factors: the *SHAPE*, which takes into account the vintage and age of the technology, and the *MULTI* parameter, which takes into account the pure time at which the cost is paid (the notation below for *SHAPE* and *MULTI* is simplified: it should also specify that these two parameters are those pertaining to the *FOM* attribute).

FIXCOST(y) =

 $\sum_{t \in MILESTONTR \cup PASTTEARS} INDIC(1.a) \times \sum_{v=Max\{M(t)-D(t)+1, y-TLIFE_t+1\}}^{Mn(M(t), y)} \left(\frac{VAR_NCAP_t}{D(t)} + NCAP_PASTI_t \right) \times NCAP_FOM_v \times SHAPE(v, y - v) \times MULTI(y)$

The useful range for y is: $\{M(t) - D(t) + 1, M(t) + TLIFE_t - 1\}$ and $y \le EOH$







Case 1.b, if $ILED_t \leq ILED_{Min,t}$ and $TLIFE_t + ILED < D(t)$ (Small projects, repeated investments in period)

The figure shows that payments made at year y may come from investments made at, before, or after period T(y). Note that our expression takes into account the vintage and age of the *FOM* being paid, via the *SHAPE* parameter, and also the pure time via *MULTI*, both pertaining to the *FOM* attribute.

$$FIXCOST(y) = \sum_{t \in MILESTONTR} INDIC(1.b) \times \sum_{v=Max\{(B(t)-TLIFE_t/2),v=TLIFE_t-1\}}^{Mm(y,\{B(t)-TLIFE_t/2),v=TLIFE_t-1\}} \left(\frac{VAR_NCAP_t}{TLIFE_t}\right) \times NCAP_FOM_v$$
$$\times SHAPE(t, y - v) \times MULTI(y)$$
(IV.1.b)

where

$$C = \left\langle \frac{D(t)}{TLIFE_t} \right\rangle$$

Useful Range for y:

$$\left\{ \left\langle B(t) - \frac{TLIFE_t}{2} \right\rangle, \left\langle B(t) - \frac{TLIFE_t}{2} \right\rangle + (C+1) \times TLIFE_t \right\}$$

and
 $y \le EOH$

Example:



Case 2.a: $ILED_t > ILED_{Min,t}$ and $ILED_t + TLIFE_t \ge D(t)$ (Large, indivisible projects, unrepeated investment in period)

i) FIXCOST(y)

The figure of the example shows that payments made in year y may come from investments made at period T(y) or earlier, but not later. Again here the *SHAPE* has the correct vintage year and age, as its two parameters, whereas *MULTI* has the current year as its parameter. Both pertain to *FOM*.

$$FIXCOST(y) = \sum_{t \in MILBETONYR, t \leq T(y)} INDIC(2.a) \times (VAR _NCAP_{t}) \times NCAP _FOM_{B(t)+ILED_{t}}$$

$$\times \left\{ \begin{array}{l} 1 \text{ if } B(t) + ILED_{t} \leq y \leq B(t) + ILED_{t} + TLIFE_{t} - 1 \\ 0 & otherwise \end{array} \right\} \times SHAPE(t, y - B(t) + ILED_{t}) \times MULTI(y)$$

$$+ \sum_{t \in PASTYEARS} INDIC(2.a) \times (NCAP _PASTI_{t}) \times NCAP _FOM_{t}$$

$$\times \left\{ \begin{array}{l} 1 \text{ if } t \leq y \leq t + TLIFE_{t} - 1 \\ 0 & otherwise \end{array} \right\} \times SHAPE(t, y - t) \times MULTI(y)$$
(IV.2.a)

Useful Range for y:

 $\{B(t) + ILED_t, B(t) + ILED_t + TLIFE_t - 1\}$ and $y \le EOH$

ii) SURVCOST (Surveillance cost for same case 2.a. See same example)

$$\begin{split} SURVCOST(y) &= \sum_{\substack{t \in MILESTONYR, \\ t \leq T(y)}} INDIC(2.a) \times (VAR_NCAP_t) \times NCAP_DLAGC_{B(t)+ILED_t} \\ &\times \begin{cases} 1 \ if \ B(t) + ILED_t + TLIFE_t \leq y \leq B(t) + ILED_t + TLIFE_t + DLAG_t - 1 \\ 0 \ otherwise \end{cases} \end{split}$$

$$+ \sum_{t \in PASTTEARS} INDIC(2.a) \times (NCAP _ PASTI_t) \times NCAP _ DLAGC_t \\ \times \begin{cases} 1 \text{ if } t + TLIFE_t \le y \le t + TLIFE_t + DLAG_t - 1 \\ 0 & otherwise \end{cases}$$

(IV.2.a')

Useful Range for y: $\{B(t) + ILED_t + TLIFE_t, same + DLAG_t - 1\}$ note that y may be larg er than EOH



Remark: again here, the cost attribute is indexed by the year when investment started its life. Also, note that, by choice, we have not defined the *SHAPE* or *MULTI* parameters for surveillance costs.

Case 2.b: $ILED_t > ILED_{Min,t}$ and $TLIFE_t + ILED_t < D(t)$ (Big projects, repeated investments in period)

i. Fixed O&M cost

The cost expression takes into account the vintage and the age of the *FIXOM* being paid at any given year y. See note in formula and figure for an explanation.

Range for y:

$$\{B(t) + ILED_t, B(t) + ILED_t + C \times TLIFE_t - 1\}$$
and
$$y \le EOH$$

Remark: same as above, concerning the indexing of the cost attribute

ii. SURVCOST(y) (surveillance cost for same case; the same example applies)

$$\begin{split} &SURVCOST(y) = \sum_{\substack{i \in MLESTONES \\ t \leq T(y)}} INDIC(2.b) \times (VAR_NCAP_t) \times NCAP_DLAGC_{\mathcal{B}(t)+ILED_t+I-TLIFE_t} \\ &\times \begin{cases} 1 \ if \ B(t) + ILED_t + (I+1) \times TLIFE_t \leq y \leq same + DLAG_t - 1 \ and \ 0 \leq I \leq C - 1 \\ 0 \ otherwise \end{cases} \end{split}$$

162

(IV.2.b)

where : $I = \left[\frac{y - B(t) - ILED_t - TLIFE_t}{TLIFE_t}\right]$ and $C = \left\langle \frac{D(t) - ILED_t}{TLIFE_t} \right\rangle$



 $\it Remark:$ same as precedently regarding the indexing of the cost attribute $\it NCAP_DLAGC$

6.2.6 Annual taxes/subsidies on capacity: FIXTAXSUB(Y)

It is assumed that these taxes (subsidies) are paid (accrued) at exactly the same time as the fixed annual costs. Therefore, the expressions IV of subsection 5.1.4 are valid, replacing the cost attributes by $NCAP_FTAX-NCAP_FSUB$.

6.2.7 Variable annual costs VARCOST(y), $y \leq EOH$

Variable operations costs are treated in a straightforward manner (the same as in MARKAL), assuming that each activity has a constant activity over a given period.

In this subsection, the symbol VAR_XXX_t is any variable of the model that represents an activity at period *t*. Therefore, *XXX* may be *ACT*, *FLO*, *COMX*, *COMT*, etc. Note that, if and when the technology is vintaged, the variable has an index *v* indicating the vintage year, whereas T(y) indicates the period when the activity takes place. Similarly, the symbol *XXX_COST_k* represents the value in year *k* of any cost attribute that applies to variable VAR_XXX .

Finally, the expressions are written only for the years within horizon, since past years do not have a direct impact on variable costs, and since no variable cost payments occur after EOH. Note also that the SHAPE and MULTI parameters are not applicable to variable costs.

As stated in the introduction, the payment of variable costs is constant over each period. Therefore, the expressions below are particularly simple.

$$VARCOST(y) = VAR _XXX_{y,T(y)} \times XXX _COST_{y}$$

$$VARTAXSUB(y) = VAR _XXX_{y,T(y)} \times (XXX _TAX_{y} - XXX _SUB_{y})$$

$$y \le EOH$$
(VI)

6.2.8 Cost of demand reductions ELASTCOST(y)

When elastic demands are used, the objective function also includes a cost resulting from the loss of welfare due to the reduction (or increase) of demands in a given run compared to the base run. See PART I for a theoretical justification.

ELASTCOST(y) =

$$\sum_{j=1}^{COM_STER_{b}} COM_BPRICE_{T(y)} \times \left\{ \left(1 - \frac{(j-1/2) \times COM_VOC_{lo,T(y)}}{COM_STEP_{lo}} \right)^{\frac{1}{COM_BLAST_{b,T(y)}}} \right\} \times VAR_ELAST_{lo,j,T(y)}$$

$$- \sum_{j=1}^{COM_STER_{bp}} COM_BPRICE_{T(y)} \times \left\{ \left(1 + \frac{(j-1/2) \times COM_VOC_{up,T(y)}}{COM_STEP_{up}} \right)^{\frac{1}{COM_BLAST_{up,T(y)}}} \right\} \times VAR_ELAST_{up,J,T(y)}$$

$$y \le EOH$$

(VII)



6.2.9 Salvage value: SALVAGE (EOH+1)

Investments whose technical lives exceed the model's horizon receive a SALVAGE value for the unused portion of their technical lives. Salvage applies to several types of costs: investment costs, sunk material costs, as well as decommissioning costs and surveillance costs. SALVAGE is reported as a single lump sum revenue accruing precisely at the end of the horizon (and then discounted to the base year like all other costs).

The salvaging of a technology's costs is an extremely important feature of any dynamic planning model with finite horizon. Without it, investment decisions made toward the end of the horizon would be seriously distorted, since their full value would be paid, but only a fraction of their technical life would lie within the horizon and produce useful outputs.

What are the costs that should trigger a salvage value? The answer is: any costs that are directly or indirectly attached to an investment. These include investment costs and decommissioning costs. Fixed annual costs and variable costs do not require salvage values, since they are paid each year in which they occur, and their computation involves only years within the horizon. However, surveillance costs should be salvaged, because when we computed them in section 6.2.5, we allowed y to lie beyond EOH (for convenience). Finally, note that any capacity prematurely retired within the model horizon is not assumed to have a salvage value (although this detail is not explicitly shown in the formulation below).

Thus, SALVAGE is the sum of three salvage values

SALVAGE(EOH + 1) = SALVINV(EOH + 1) + SALVDECOM(EOH + 1) + SALVSURV(EOH + 1)

We treat each component separately, starting with SALVINV.

A). Salvaging investment costs (from subsections 6.2.2 and 6.2.3)

The principle of salvaging is simple, and is used in other technology models such as MARKAL, etc: a technology with technical life TLIFE, but which has only spent x years within the planning horizon, should trigger a repayment to compensate for the unused portion TLIFE-x of its active life. The computation of the salvage value obeys a simple rule, described by the following result:

Result 1				
The salvage value (calculated at year k) of a unit investment made in year k , and whose technical life is TL , is:				
$S(k,TL) = 0$ if $k + TL \le EOH$				
S(k,TL) = 1 if $k > EOH$				
$S(k,TL) = \frac{(1+d)^{TL-EOH-1+k}-1}{(1+d)^{TL}-1}$ otherwise				
where d is the general discount rate				

Note that the second case may indeed arise, because some investments will occur even after *EOH*.

Since we want to calculate all salvages at the single year (EOH+1), the above expressions for S(k,TL) must be discounted (multiplied) by:

 $(1+d)^{EOH+1-k}$

Finally, another correction must be made to these expressions, whenever the user chooses to utilize a technology specific discount rate. The correction factor which must multiply every investment (and of course every salvage value) is:

$$\frac{CRF_{z}}{CRF} = \frac{\left(1 - \frac{1}{1 + i_{z}}\right) \times \left(1 - \frac{1}{(1 + i)^{ELLFE}}\right)}{\left(1 - \frac{1}{1 + i}\right) \times \left(1 - \frac{1}{(1 + i_{z})^{ELLFE}}\right)}$$

where *i* is the general discount rate *i_s* is the technolog y specific discount rate and ELIFE is the economic life of the investment

Note: the time indexes have been omitted for clarity of the expression.

The final result of these expressions is *Result 2* expressing the salvage value discounted to year EOH+1, of a unit investment with technical life *TL* made in year *k* as follows. Result 2 will be used in salvage expressions for investments and taxes/subsidies on investments.

Result 2 $SAL(k,TL) = 0 if k + TL \le EOH$ $SAL(k,TL) = \frac{CRF_s}{CRF} if k \ge EOH + 1$ $SAL(k,TL) = \frac{1 - (1+d)^{EOH + 1 - k - TL}}{1 - (1+d)^{-TL}} \times \frac{CRF_s}{CRF} otherwise$ where d is the general discount rate and d_s is the technology specific discount rate

These expressions may now be adapted to each case of investment (and taxes/subsidies on investments). We enumerate these cases below. Note that to simplify the equations, we have omitted the second argument in *SAL* (it is always *TLIFE_t* in the expressions).

Case 1.a $ILED_t \leq ILED_{Mn,t}$ and $TLIFE_t + ILED_t \geq D(t)$ (Small divisible projects, non-repetitive, progressive investment in period)

$$SALVINV(EOH + 1) = \sum_{t} INDIC(I.1.a) \times \sum_{\nu=M(t)-D(t)+1}^{M(t)} \left(\frac{VAR_NCAP_{t}}{D(t)} + NCAP_PASTI_{t} \right) \times NCAP_COST_{\nu} \times SAL(\nu)$$

where SAL(v) is equal to $SAL(v, TLIFE_t)$ defined in Result 2.

Note that SAL(v) = 0 whenever $v + TLIFE_t \le EOH + 1$ (see Result 2)

(VIII.1.a)

Case 1.b $ILED_t \leq ILED_{Min,t}$ and $TLIFE_t + ILED < D(t)$ Small Projects, repeated investments in period

 $SALVINV(EOH + 1) = \sum_{t} INDIC(I.1.b) \times \sum_{v=B(t)-)TL/2 \langle +(C-1) \times TLIFE_t}^{B(t)-\rangle TL/2 \langle +(C-1) \times TLIFE_t-1} \frac{VAR_NCAP_t}{TLIFE_t} \times NCAP_COST_v \times SAL(v)$

Note again here that SAL(v) equals 0 if $v + TLIFE \le EOH + 1$

(VIII.1.b)

Case 2.a: $ILED_t > ILED_{Min,t}$ and $ILED_t + TLIFE_t \ge D(t)$ (Large, indivisible projects, unrepeated investment in period)

$$SALVINV(EOH + 1) = \sum_{t \in MILESTONESYEARS} VAR_NCAP_t \times NCAP_COST_{B(t)+ILED_t} \times SAL(B(t) + ILED_t)$$

Note that $SAL(B(t) + ILED_t) = 0$ whenever $B(t) + ILED_t + TLIFE_t \le EOH + 1$ (VIII.2.a)

Case 2.b: *ILED* > *ILED*_{Min,t} and *TLIFE*_t + *ILED*_t < D(t)(Large, indivisible Projects, repeated investments in period)

 $SALVINV(EOH + 1) = \sum VAR _NCAP_t \times NCAP_COST_{B(t)+(C-1)\times TLIFE_t + ILED_t} \times SAL(B(t) + (C-1)\times TLIFE_t + ILED_t)$

Note again that $SAL(B(t) + (C-1) \times TLIFE_t + ILED_t) = 0$ whenever $B(t) + (C-1) \times TLIFE_t + ILED_t + TLIFE_t \leq EOH + 1$ (VIII.2.b)

NOTE: salvage cost of taxes/subsidies on investment costs are identical to the above, replacing NCAP_COST by {NCAP_ITAX – NCAP_ISUB}.

B). Savage value of decommissioning costs (from subsection 6.2.4)

For decommissioning costs, it should be clear that the triggering of salvage is still the fact that some residual life of the *investment itself* exists at EOH+1. What matters is *not* that the decommissioning occurs after EOH, but that some of the investment life extends beyond EOH. Therefore, Result 1 derived above for investment costs, still applies to decommissioning. Furthermore, the correction factor due to the use of technology specific discount rates is also still applicable (with *ELIFE* replaced by *DELIF*).

However, the further discounting of the salvage to bring it to EOH+1 is now different from the one used for investments. The discounting depends on the year l when the decommissioning occurred and is thus equal to:

 $(1+d)^{EOH+1-l}$ where l is the year when decommissioning occurs.

l depends on each case and will be computed below

In cases 1.a and 1.b,

l=TLIFE + k
In case 2.a	k is fixed at $B(t)$ +ILED, but l varies from $(B(t)$ +ILED+TLIFE+DLAG) to (same +DLIFE-1)
In case 2.b	<i>k</i> is fixed at <i>B(t)+ILED+(C-1)×TLIFE</i> , but <i>l</i> varies from (<i>B(t)+ILED+C×TLIFE+DLAG</i>) to (same + DLIFE-1)

It is helpful to look at the examples for each case in order to understand these expressions.

Finally, the equivalent of Result 2 is given as Result 3, for decommissioning.

Result 3		
The Salvage	Value of a decommissioning cost occuring at year l , for an investment taking place at year k , is:	
	$SAL(k, l) = 0$ if $k + TL \le EOH$	
SAL	$U(k,l) = \frac{CRF_s}{CRF} \times (1+i)^{EOH+1-l} \qquad if \ k \ge EOH+1$	
SAL(k,	$l) = \frac{(1+d)^{TLIFE+k-l} - (1+d)^{EOH+1-l}}{(1+d)^{TLIFE} - 1} \times \frac{CRF_s}{CRF} \text{ otherwise}$	
where d is the general discount rate and d_s is the technology specific discount rate		

We are now ready to write the salvage values of decommissioning cost in each case.

Case 1.a $ILED_t \leq ILED_{Mn,t}$ and $TLIFE_t + ILED_t \geq D(t)$ (Small divisible projects, non-repetitive, progressive investment in period)

$$\begin{split} SALVDECOM(EOH + 1) = \\ \sum_{t} INDIC(1.a) \times \sum_{v=\mathcal{M}(t)-\mathcal{D}(t)+1}^{\mathcal{M}(t)} \left(\frac{VAR_NCAP_{t}}{D(t)} + NCAP_PASTI_{t} \right) \times \\ NCAP_DCOST_{v} \times SAL(v, v + TLIFE_{t}) \end{split}$$

where SAL(k, l) is defined in Result 3.

Note that SAL(v,x) is always 0 whenever $v + TLIFE \le EOH + 1$ (IX.1.a)

Case 1.b $ILED_t \leq ILED_{Min,t}$ and $TLIFE_t + ILED < D(t)$ (Small Projects, repeated investments in period)

$$\begin{aligned} SALVDECOM(EOH + 1) &= \\ \sum_{t} INDIC(1.b) \times \sum_{v=B(t)-/TL/2 \langle +(C-1) \lor TLIFE_{t}}^{B(t)-/TL/2 \langle +(C-1) \lor TLIFE_{t}-1} \frac{VAR_NCAP_{t}}{TLIFE_{t}} \times NCAP_DCOST_{v} \times SAL(v, v + TLIFE_{t}) \end{aligned}$$

Note again here that SAL(k, l) equals 0 if $k + TLIFE \le EOH + 1$

(IX.1.b)

Case 2.a: $ILED_t > ILED_{Min,t}$ and $ILED_t + TLIFE_t \ge D(t)$ (Large, indivisible projects, unrepeated investment in period)

SALVDECOM(EOH + 1) =

$$\sum_{t \in MILESTONESTEARS} INDIC(2.a) \times VAR_NCAP_t \times NCAP_COST_{B(t)+ILED_t} \times \sum_{l=B(t)+TLIFE+DLAG}^{same+ULIFE-1} SAL(B(t)+ILED_t, l)$$

Note that SAL is 0 whenever $B(t) + ILED_t + TLIFE_t \le EOH + 1$

(IX.2.a)

Case 2.b: $ILED_t > ILED_{Mn,t}$ and $TLIFE_t + ILED_t < D(t)$ (Large, indivisible Projects, repeated investments in period)

$$\begin{aligned} SALVDECOM(EOH + 1) &= \sum_{t \in MILESTONYEARS} INDIC(2.b) \times VAR _NCAP_t \times NCAP _DCOST_{B(t)+(C-1)\times TLIFE_t+ILED_t} \\ &\times \sum_{l=B(t)+,ILED_t + C \times TLIFE_t+DLAG_t} SAL[B(t) + ILED_t + (C-1) \times TLIFE_t, l] \end{aligned}$$

where

$$C = \left\langle \frac{D(t) - ILED_t}{TLIFE_t} \right\rangle$$

Note again that SAL is 0 whenever $B(t) + C \times TLIFE_t + ILED_t \leq EOH + 1$

(IX.2.b)

C) Salvage Value of Surveillance Costs

Similarly to the salvaging of decommissioning costs, the basic salvage value fractions S(k,m) defined in *Result 1* at the beginning of Section 6.2.9 are used as the basis for the salvage value of surveillance costs. However, unlike with decommissioning costs, there is no need to make corrections for technology-specific discount rates, as the costs do not represent capital costs. In addition, the discounting to EOH+1 must be made separately for each surveillance year. Note that only Cases 2 have surveillance costs.

 $\begin{aligned} & \textbf{Case 2.a: } ILED_t > ILED_{Min,t} \quad \text{and} \quad ILED_t + TLIFE_t \ge D(t) \\ & \textbf{(Large, indivisible projects, unrepeated investment in period)} \\ & SALVSURV(EOH + 1) = \\ & \sum_{t \in MILESTONESYEARS} INDIC(2.a) \times S(B(t) + ILED_t, TLIFE_t) \times \\ & VAR_NCAP_t \times NCAP_DLAGC_{B(t)+ILED_t} \times \sum_{l=B(t)+ILED_t+TLIFE_t}^{same+DLAG_t-1} DISC(l, EOH + 1) \end{aligned}$

Note that S(k,m) = 0 whenever $k + m \le EOH + 1$.

(X.2.a)

Case 2.b: $ILED_t > ILED_{Min,t}$ and $TLIFE_t + ILED_t < D(t)$ (Large, indivisible projects, repeated investments in period)

SALVSURV(EOH + 1) =

$$\sum INDIC(2.b) \times S[B(t) + ILED_{t} + (C - 1) \times TLIFE_{t}, TLIFE_{t}] \times VAR _NCAP_{t} \times NCAP_{t} \otimes ILED_{t}(C-1) \times TLIFE \times \sum_{l=B(t)+ILED_{t}+C \times TLIFE_{t}}^{same+DLAG_{t}-1} DISC(l, EOH + 1)$$

where : $C = \left\langle \frac{D(t) - ILED_t}{TLIFE_t} \right\rangle$

Note again that S(k,m) = 0 whenever $k + m \le EOH + 1$.

(X.2.b)

6.2.10 Late revenues from endogenous commodity recycling after EOH LATEREVENUE(y)

Late revenues consist of revenues from any materials and energy which had been embedded in some processes, and which are released after *EOH*. Such revenues exist only if an exogenous salvage value was declared by the user for the sunk material.

Note: For materials released within the horizon, the revenue is either explicit (and then it is the user's responsibility to indicate a negative $\cot - \arctan d$ dismantling time), or the revenue is implicit, and then the user must specify a physical release of the material at dismantling time, and the model will correctly 'price' this material within the RES.

 $LATEREVENUES(y) \quad y \ge EOH+1$

The late revenues come *only* from the resale at dismantling time, of materials and/or energy that were sunk at construction time. Therefore, the *LATEREVENUES* expressions are identical to the decommissioning cost expressions, with the NCAP_DCOST attribute replaced by

$$\sum_{c} -NCAP_VAL(c) \times NCAP_OCOM(c)$$

where the summation extends over all commodities c for which an $NCAP_OCOM$ attribute is defined (defaults to zero if undefined)

LATEREVENUES(y) is reported as a lump sum discounted to the user selected base year.

6.2.11 Known issues in the standard objective function formulation

There are a few known issues in the standard objective function formulation that may cause small distortions in the cost accounting and, subsequently, in the relative competitiveness of technologies. The distortions only occur when using period lengths D(t) > 1. The issues can be briefly summarized as follows:

- In the investment cases 1.a and 1.b, the timing of the annual payments for the investment costs and fixed operation and maintenance costs are not fully in sync with the assumed amounts of available capacity. Although the effective difference is usually quite small, with longer periods having an even number of years, the distortion may become considerable.
- In the investment cases 1.a and 1.b, the spreading of the investment cost over D(t) or TLIFE(p) years causes some distortions in the salvage value accounting, which are at the highest in cases where B(v)+TLIFE = EOH+1, (capacity is retired exactly at the end of the horizon), because in such cases the capacity is assumed fully available within the model horizon, but it still has a salvage value according to the standard formulation.
- In all investment cases, the capacity is assumed to be available in each period according to the proportion of the period being covered by the years [B(v)+ILED(v),B(v)+ILED(v)+TLIFE(v)-1]. If all periods contain only a single year, this is quite accurate, but, due to discounting, it is no longer accurate with longer periods. That is because any capacity available in year y has a larger value than the same capacity available in year y+1. But again, this causes only a small distortion in the cost accounting.
- With variable period lengths, investments for period t can start even before the previous milestone year t-1. If the investment costs are changing over time, in such cases the costs are not accounted in a fully consistent way, because the investment cost data is taken from the start year of each investment step.

The first three of these issues have been addressed by introducing an optional switch (\$SET OBLONG YES), which, when activated, will eliminate all those three issues. For the first two issues, the discounting of the annual payments for the investment costs and fixed operation and maintenance costs is slightly modified, such that the weighted average of the commissioning years over the investment steps is exactly equal to B(v) (the weights being the present value factors for the commissioning years). In other words, the modification introduces a small additional discounting multiplier, which moves the whole investment spread slightly in time, such that the resulting costs will effectively always be in sync with the assumed available capacity (and activity).

For the third issue, the capacity transfer coefficients are slightly modified to reflect the true value of the capacity in each period, based on the *discounted* proportion of the period being covered by the process lifetime.

The modified objective function has been verified to produce results that are fully consistent with single-year period results, assuming that process parameters do not change over time, which is the best what one can expect. The fourth issue can only be

addressed by using any of the alternative objective formulations (see separate *Objective Function Variants* documentation, available at the ETSAP documentation website).

6.2.12 The discounting methods for annual payments

In the standard objective function of TIMES, all costs and payments are assumed to occur at the beginning of each year. In the case of investment costs, this means that the annualized payments made in the beginning of each year within the economic lifetime are equivalent to a lump-sum investment cost paid at the beginning of the first operation year, if the annual payments are discounted back to that point by the technology-specific discount rate (for instance, in case 1a, each lump sum is equal to $NCAP_COST/D(t)$). Similarly, in the case of operation costs (e.g. $NCAP_FOM$), the total annual costs are assumed to occur at the beginning of each operating year.

Because the operating costs can nevertheless be assumed to be spread continuously throughout the year, this kind of 'beginning-of-year' discounting method introduces a small bias in the discounting of different cost components. For example, the operating costs in the first year of operation should be assumed to occur about half a year later in time compared to the investment, and not at the same time, as assumed in TIMES. One may well argue that this time-difference should be reflected in the discounting applied.

In TIMES, there is an option to correct this small bias by using mid-year discounting, or even end-of-year discounting. The options can be activated by the switch $MID_YEAR / DISCSHIFT$ (see Part III, Control switches). The modifications needed in the discounting are basically quite similar for employing both mid-year and end-of-year discounting. Therefore, only the corrections for mid-year discounting are described in detail below.

The corrections needed for employing mid-year discounting in TIMES can be made in the following two steps:

- First, simply assume that instead of the beginning of each year, all payments are made in the mid-point of each year in TIMES. As such, this assumption doesn't change the objective function in any way; it is only a change in thinking. However, it also means that instead of the beginning of the base year, all costs are assumed to be discounted to the mid-point of the base year.
- Second, make the necessary corrections to the discounting of all those cost components that cannot be assumed to be actually paid at the mid-point of the year.

By going through the various cost components, the following conclusions hold for step 2:

- All variable, fixed operation and surveillance costs can be assumed to be paid in the mid-point of each year, and no change is needed for them in the discounting.
- The lump-sum investment costs in Cases 1 (*NCAP_COST/D(T)*) should be assumed to occur at the beginning of the investment year instead of the mid-point.
- All the lump-sum investment costs in Cases 2 (NCAP_COST/ILED) can be assumed to occur in the mid-point of each construction year. Therefore, no change is needed in the discounting of the annualized investment payments.
- Decommissioning costs in Cases 1 can be assumed to be paid in the mid-point of the year, because in these cases decommissioning is assumed to take exactly one year, and one may assume that, on the average, the costs occur at the mid-point.

• The lump-sum decommissioning costs in Cases 2 (*NCAP_DCOST/DLIFE*) can be assumed to occur at the mid-point of each year within the decommissioning lifetime. Therefore, no change is needed in the discounting of the annualized payments.

Consequently, the initial overall conclusion is that the only correction needed in the discounting of various cost components is related to the investment costs in Cases 1. If we assume that the Capital Recovery Factor used in the beginning-of-year discounting (CRF_{beg}) is still valid for mid-year discounting, we should simply shift the position of both the lump-sum investment and the annualized payments half a year backwards. In terms of discounting, this means that in Cases 1 the annualized investment payments should be multiplied by the factor $(1+d(y))^{0.5}$, where d(y) is the general discount rate. Perhaps the simplest way to apply this correction in the objective function is to make the adjustment to the Capital Recovery Factor. Thus, for Cases 1 we could define a 'CRF corrected for mid-year discounting' (*CRF*_{1,mid}) as follows:

$$CRF_{1,mid} = CRF_{beg} \times (1 + d(T(y)))^{0.5}$$

However, one could additionally argue that the Capital Recovery Factor CRF_{beg} is no longer valid for mid-year discounting. The annualized investment payments can also be assumed to represent a continuous stream of costs, which should thus be assumed to be paid at the mid-point of each year. The shortcoming of the original CRF_{beg} can be seen by calculating its value for an investment with an economic lifetime of just one year. The value of CRF_{beg} is in this case exactly 1, although it seems obvious that some interest should be involved as well. Assuming that the single payment represents a continuous stream of costs, the payment can be assumed to occur at the mid-point of the year, and would thus include interest for half-year's time.

Accordingly, we should correct the definition of the *CRF proper* by assuming that the annualized payments occur *half a year forward* in time with respect to the lump-sum investment, which means that we must increase the nominal size of the payments by the corresponding interest for the half-year's time. Combining these corrections together, the general discount rate d(y) should be simply replaced by the **technology-specific discount** rate $d_s(T(y))$ in the expression above, because in addition to the nominal change in the *CRF*, the time of the annualized payments has been restored back to original. However, to maintain consistency between Cases 1 and 2, the same basic correction to the *CRF* proper should be applied to all cases. Therefore, the total adjustments needed when taking into account the correction to the *CRF* proper are the following:

$$CRF_{mid}^{proper} = CRF_{beg} \times (1 + d_s(T(y)))^{0.5}$$
(XI.1)

$$CRF_{I,mid} = CRF_{mid}^{proper} \times (1+d(T(y)))^{0.5} \times (1+d(T(y)))^{-0.5} =$$

$$CRF_{I,mid} \times (1+d(T(y)))^{0.5} \times (1+d(T(y)))$$

$$CRF_{2,mid} = CRF_{mid}^{proper} \times (1+d(T(y)))^{-0.5} = CRF_{beg} \times (1+d(T(y)))^{-0.5} \times (1+d_s(T(y)))^{0.5}$$
(XI.3)

Consequently, in both cases the annualized investment payments are then assumed to occur at the mid-point of each fiscal year starting at the time of the lump-sum investment, and

the annual payments are equivalent to the lump-sum investment when discounted back to that point by the technology-specific discount rate. The implementation of the optional corrections for mid-year discounting corresponds to equations (XI.1 to XI.3). To be consistent, the expression (XI.3) for $CRF_{2,mid}$ should also be used for decommissioning costs.

6.3 Constraints

The constraints available in standard TIMES are shown in Table 23 below, and later fully described in the following subsections. The constraints related to the Climate Module (CLI), Damage Cost Functions (DAM) and Endogenous Technology Learning (ETL) are shown and described in three separate chapters (Appendices A, B and C respectively).

Table 24. List of TIMES equations

Equation Name	Short description
BND FLAST	Upper bound on each of the step variables used to linearize the demand
5110_001	function when elastic demand feature is used
EQ(I) ACTBND	Bound on the activity of a process
EQE ACTEFF	Equality relationship that defines the activity efficiency of a process
EQ_ACTFL0	Equality relationship that defines the activity of a process in terms of its flow variables
EQ ACTPL	Defines the efficiency deterioration of a process at partial loads
EQ_ACTRAMP	Defines bounds on the ramping of process activity, in proportion to its online capacity, in either direction (LO/UP)
EQL_ACTUPC	Sets a lower limit on the successive on-line / off-line hours of capacity
EQE_ACTUPS	Expresses that the change in process on-line capacity between successive timeslices must be equal to the capacity started-up – shut-down
EQL_ACTUPS	Expresses that the sum of process started-up capacity over a cycle must be at least equal to the max. amount of capacity put off-line in the cycle
EQ(I)_ASHAR	Establishes advanced share constraints between process flows
EQ(I)_BLND	Special blending constraints used to specify the composition of refined oil products
EQ_BNDCST	Establishes a variable representing the cumulative amount of process costs, taxes and/or subsidies over a time interval, for defining a bound
EQ(I)_BNDNET	Bound on the net amount (production minus consumption) of a commodity
EQ(I)_BNDPRD	Bound on the total production of a commodity
EQ(I)_CAFLAC	Relates the flows in the primary group of a process to its available capacity; may be rigid $(=)$ or flexible (\leq)
EQ(I)_CAPACT	Relates the activity of a process to its available capacity; may be rigid (=) or flexible (\leq, \geq)
EQL_CAPFLO	Relates a flow not in the primary group of a process to its available capacity; only an upper bound for the flow \leq is supported
EQ_CAPLOAD	Relates the activity of a process to its available on-line capacity in each timeslice; only for processes with flexible availability (\leq,\geq)
EQ(I)_CPT	Calculates the current capacity of a process in terms of all past and current investments in that process
EQ(I)_COMBAL	Balance equation of a commodity
EQE_COMPRD	Definition of the total production of a commodity
EQ_CUMFLO	Bound on the cumulative flow or activity of a process over a time interval
EQ_CUMNET	Bound on the cumulative production of a commodity over a time interval
EQ_CUMPRD	Bound on the cumulative net quantity of a commodity over a time interval
EQ_CUMRET	Establishes a variable representing the cumulative amount of retired capacity of a process

- Standard State - States	
Equation Name	Short description
EQ DSCNCAP	These two constraints ensure that some investments may only be made in
and	certain discrete sizes
EO DSCONE	
FO DSCRFT	Ensures that early canacity retirements may only be made in multiples of
EQ_DOCKET	a certain discrete block size
EO(I) FLOBND	Bound on the sum over a commodity group, of the commodity flows of a
	process
EO(I) ELOER	Relationship between a flow in one timeslice and the annual flow for a
	given process
	Expresses for a given commodity that the amount produced/consumed by
	a process is tied to the total amount produced/consumed of that
	a process is the to the total amount produced/consumed of that
	Everyone that imports of a commodity by region r must be equal to all
EQ_IKE	expresses that imports of a commonly by region r must be equal to all
	Exports by other regions to region 1
EQ_IREBND	Bound on exchange of a commonity between two regions
EQ_XBND	Bound on total exchanges of a commodity by one region
EQ(I)_INSHR	For a given process, expresses that the inflow of a commodity is tied to
	the total inflows of all commodities in a certain group
EQ(I)_OUTSHR	For a given process, expresses that the outflow of a commodity is tied to
	the total outflows of all commodities in a certain group
EQ_PEAK	Expresses that capacity available must exceed demand of a selected
and a second of the second sec	commodity in any time slice by a certain margin
EO PTRANS	Establishes an equality relationship between (groups of) inputs and
·—	certain (groups of) outputs of a process
FOL SCAP	Bounds the amount of capacity salvaged if early retirements are active.
EQ STGAUX	Establishes an equality relationship between storage main flows or activity
24_010/10/1	and an auxiliary storage flow
EQ_STGIPS	Ensures the storage of a commodity between two time periods
EQ_STGTSS	Ensures the storage of a commodity between two timeslices
EQ(I) STGIN	Bounds the input into a storage process
EO(I) STGOUT	Bounds the output of a storage process
EQ STSBAL	Defines balances between timeslice levels in a general timeslice storage
EQ(I) LICRTP	Defines a dynamic bound on the growth / decay in the installed canacity
2.2(1)_001(1)	new capacity or activity of a process over successive periods
Llcor	User defined constraints that have a user defined constant PHS
Constraints of	
the LHS type	
Timoglico	Licer defined constraints that involve only a single region r and period t
Timesice-	but both timestice a and the averaging timestice a up ato (a c)
dynamic User	but both timeslice s and the preceding timeslice s-rs_stg (r,s)
Constraints	
User	User defined constraints that involve both period t and the succeeding
Constraints of	period t+1
dynamic type	
(t,t+1)	
User	User defined constraints that involve both period ${f t}$ and the preceding
Constraints of	period t-1
dynamic type	
(t-1,t)	

6.3.1 Bound: BND_ELAST

Indices: region (r), year (t), commodity (c), time slice (s), linearization step (j), direction of elastic demand change (l)

Type: ≤

Related variables: VAR_ELAST

Related equations: EQ(I)_COMBAL, EQ_OBJELS, EQ_OBJ

Purpose: Upper Bounds on the step variables used to represent the demand when the elasticity is non-zero.

Remarks:

- These bounds are applied whenever a demand is price elastic, i.e. when the COM_ELAST (elasticity) and COM_VOC (total range) parameters are specified and not zero.
- If COM_ELAST and COM_VOC are specified, and COM_STEP (number of steps) is not, the latter defaults to 1 (single step discretization)
- Attributes COM_VOC and COM_STEP do not have a timeslice index. The user can still control elasticities in each time slice through COM_ELASTs.

Bound:

 $BND_ELAST_{r,t,c,s,j,l} \ni COM_STEP_{r,c,l} \land (s \in com_ts_{r,c,s})$

 $V\!A\!R_E\!L\!A\!ST_{r,t,c,s,j,j} \leq \frac{COM_PROJ_{r,t,c} \times COM_FR_{r,t,c,s} \times COM_VOC_{r,t,c,j}}{COM_STEP_{r,c,j}}$

6.3.2 Equation EQ(*l*)_ACTBND

Indices: region (r), model year (t), process (p), time slice (s)

Type: Any type, as determined by the index bd of ACT BND:

- l = G' for **bd** = LO' (lower bound) yields \geq .
- l = 'E' for **bd** = 'FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .

Related variables: VAR_ACT

Related equations: EQ_COMBAL, EQ_ACTFLO, EQ_PTRANS

Purpose: This equation bounds the total activity of a process in a period independently of the vintage years of the installed capacities. The equation will either be generated when the activity bound is specified for a timeslice being at a timeslice level above the timeslice level of the process (**prc_tsl**), e.g. ACT_BND is specified for an ANNUAL timeslice but the process operates on a DAYNITE timeslice level, or irrespectively of the timeslices when the process is characterized as a vintaged one (**prc_vint**). If activity bounds are specified for timeslice level by standard aggregation (see Section 3.1.2) and then directly applied to the activity variable for non-vintaged processes.

Remarks:

- The equation is required because for the two cases described above (bound specified for a timelslice above the process timeslice level or process is characterized as a vintaged one), no single variable exists which can be bounded directly.
- The bound is only directly applied to VAR_ACT for non-vintaged processes, when ACT_BND is applied at the level **prc_ts(r,p,s)**.

Interpretation of the results:

- Primal: The level value of the equation describes the activity of the process in the considered period t and timeslice s.
- Dual: The dual variable describes in the case of a lower (upper) bound the cost increase (decrease) caused by an increase of the activity bound by one unit.



6.3.3 Equation: EQE_ACTEFF

Indices: region (r), vintage year (v), period (t), process (p), commodity group (cg), side (io), timeslice (s)

Type: =

Related variables: VAR_ACT, VAR_FLO

Related equations: EQ_PTRANS, EQ_ACTPL

Purpose: This equation is generated when the process activity efficiency has been defined with the input attribute $ACT_EFF_{r,v,p,cg,s}$ for a group of flows on the shadow side. **Remarks**:

- The group cg in the equation may be either directly specified in ACT_EFF, or, if ACT_EFF is only specified for single commodity, determined as the commodity type, or, if ACT_EFF is specified for the reserved group name 'ACT, determined as the default shadow group of the process.
- The parameter $ACT_EFF_{r,v,p,cg,s}$ can be specified using any of the following as the cg:
 - commodity groups; these define a common efficiency for all member commodities in the group that are on the shadow side of the process;
 - commodity types (NRG/MAT/ENV/DEM/FIN); as above, these define a common efficiency for all member commodities in the group that are on the shadow side of the process;
 - the predefined commodity group 'ACT'; this defines a common efficiency for all members of the default shadow group of the process;
 - single commodities on the shadow side without an associated group efficiency; these define commodity-specific efficiencies, and the shadow group will consist of all commodities of the same type; if no commodity efficiency is defined for some member in the group, the default efficiency 1 is assumed;
 - single commodities on the shadow side with an associated group efficiency; these
 define commodity-specific efficiencies as above, but are multiplied by the efficiency specified for the group; if no efficiency is defined for some member in the
 group, the group efficiency is applied directly to that member;
 - single commodities C that are members of the PCG of the process; these define commodity-specific multipliers for the process efficiency when producing the commodity C; if no efficiencies are additionally defined on the shadow side of the process, the whole standard shadow group of the process is assumed to be involved in the transformation (as when using 'ACT'), with the default efficiency of 1 on the shadow side.
- The ACT_EFF parameter can also be shaped by using a FLO_FUNCX parameter of the following form: FLO_FUNCX(reg,datayear,p,CG,'ACT') = shape index. Here, the CG should correspond to the group of commodities on the shadow side involved in the EQE_ACTEFF equation (the group, commodity type, or 'ACT' that was either explicitly or implicitly used in the ACT_EFF parameters that should be shaped).

Equation:

 $EQE_ACTEFF_{r,v,t,p,cg,i\sigma,s} \quad \ni (\mathsf{rtp_vintyr}_{r,v,t,p} \land \neg \mathsf{rp_inout}_{r,p,io} \land ACT_EFF_{r,v,p,cg,s})$

$$\sum_{\substack{\text{com} gman_{reg} \\ \text{rtep_vart_{stepss}}}} \begin{pmatrix} VAR_FLO_{r,v,t,c,ts} \times \\ (ACT_EFF_{r,v,p,c,ts} & \text{if } ACT_EFF_{r,v,p,c,ts} & \text{given} \\ 1 & \text{otherwise} \end{pmatrix} \end{pmatrix}$$

$$=$$

$$\sum_{\substack{\text{rtep_vart_{stepss}}}} \begin{pmatrix} VAR_ACT_{r,v,t,p,ts} & \text{if } RP_PGACT_{r,p} \\ \frac{VAR_FLO_{r,v,t,p,c,ts}}{PRC_ACTFLO_{r,v,p,c}} & \text{otherwise} \end{pmatrix} \times \\ \begin{pmatrix} 1/ACT_EFF_{r,v,p,cg,ts} & \text{if } ACT_EFF_{r,v,p,cg,ts} & \text{given} \\ 1 & \text{otherwise} \end{pmatrix} \times \\ \begin{pmatrix} 1/ACT_EFF_{r,v,p,cg,ts} & \text{if } ACT_EFF_{r,v,p,cd,ts} & \text{given} \\ 1 & \text{otherwise} \end{pmatrix} \times \\ \begin{pmatrix} 1/ACT_EFF_{r,v,p,cd,ts} & \text{if } ACT_EFF_{r,v,p,cd,ts} & \text{given} \\ 1 & \text{otherwise} \end{pmatrix} \end{pmatrix}$$

$$\sum_{\substack{\text{Prc_fs}_{r,p,s}}} \begin{pmatrix} VAR_UPS_{r,v,t,p,ts} \times \\ ACT_LOSPL_{r,v,p,ts} \end{pmatrix} & \text{if } ACT_LOSPL_{r,v,p,ts} & \text{given} \\ \times RS_FR_{r,s,ts} \end{pmatrix}$$

6.3.4 Equation: EQ_ACTFLO

Indices: region (r), vintage year (v), milestone year (t), process (p), time slice (s)

Type: =

Related variables: VAR_ACT, VAR_FLO, VAR_IRE

Related equations: EQ_COMBAL, EQ_CAPACT, EQ_PTRANS

Purpose: This equation defines the VAR_ACT activity variable in terms of the "primary flows" of a process. The primary flows are defined by the user through the **prc_actunt** set attribute.

Remarks:

- The internal set **rtp_vintyr** ensures that (v,t) expressions are generated for the vintaged processes and (t,t) for the non-vintaged ones.
- The constraint defines the activity of a process. The activity of a process is limited in the equation EQ(1)_CAPACT by the available capacity.
- rtp_vara(r,t,p) controls the valid periods in which the process can operate.
- **rp_aire(r,p)** controls which sides of an import/export process should define activity
- If the activity of a process is defined by a single flow, the flow variable is replaced by the activity variable in case that the reduction algorithm is activated. Then, in all equations where the flow occurs, the activity variable is used instead. In this case the equation EQ_ACTFLO is not generated.

Equation:

 $E\!\mathcal{Q}_\textit{ACTFLO}_{r,v,t,p,s} \quad \ \ \text{ \Rightarrow rtp_vintyr}_{r,v,t,p} \land prc_ts_{r,p,s} \land rtp_vara_{r,t,p}$

IF NOT rpc_ire	4	The process is not an inter-
$VAR_ACT_{v,t} = \sum_{c \in \mathbf{prc_actume}}$	$\frac{VAR_FLO_{r,v,t,p,c,s}}{PRC_ACTFLO_{r,v,p,c}}$	regional process
IF rpc_ire		The process is an inter- regional trade process.
$VAR_ACT_{t,v} = \sum_{c \in \mathbf{prc_actum}}$	$\frac{\sum_{i e \in \mathbf{r_p} \text{ aire}} VAR _ IRE_{r, y, t, p, c, s, ie}}{PRC _ ACTFLO_{r, y, p, c}}$	

6.3.5 Equation: EQ_ACTPL

Indices: region (r), vintage year (v), period (t), process (p), time slice (s)

Type: =

Related variables: VAR_ACT, VAR_UPS

Related equations: EQE_ACTEFF

Purpose: This equation defines the variable proportional to the efficiency loss under partial loads, if endogenous partial load efficiencies are modeled for a process, or a corresponding cost penalty under partial loads.

Remarks:

- Endogenous partial load efficiencies can only be modeled for processes that have their efficiency modelled by the ACT_EFF parameter.
- The input parameter ACT_LOSPL(r,y,p,'FX') defines the proportional increase in specific fuel consumption at the minimum operating level, when modelling partial load efficiencies endogenously (for process p, vintage y, region r).
- The input parameter ACT_LOSPL(r,y,p,'LO') defines the minimum operating level used for the partial load efficiency function; default value is taken from ACT_UPS(r,y,p, 'ANNUAL','FX'), but if neither is specified, is set to 0.1.
- The input parameter ACT_LOSPL(r,y,p,'UP') defines the fraction of the feasible load range above the minimum operating level, below which the efficiency losses are assumed to occur; default value = 0.6.
- It is recommended that the minimum operating level is defined by the ACT_MINLD(r,v,p) parameter, which is then used as the default value for ACT_LOSPL(r,y,p,'LO'). However, if desired, the minimum level to be assumed can also be defined by explicitly specifying ACT_LOSPL('LO').
- When the ACT_CSTPL input parameter is defined instead of (or as a supplement to) ACT_LOSPL, the cost coefficient is applied in the objective function directly to the VAR_UPS_{r,v,p,s,TX} variable as defined by the EQ_ACTPL equation.

Notation:

- *AF_MIN_{r,v,p,s}* minimum operating level of online capacity of process *p*, vintage *v* in timeslice *s*, as defined by ACT_MINLD (default) or ACT_LOSPL('LO');
- *PL_LDL_{p,v}* the load level below which partial load efficiency losses start to occur for process *p*, vintage *v*;
- *SUP(s)* is the set of timeslices above timeslice *s* in the timeslice tree, but including also *s* itself;
- UPS(p) is the set of timeslices with start-ups/shut-downs allowed for process p.

Equations:

 $EQ_ACTPL_{r,y,t,p,s,} \quad \ni (\textbf{rtp_vintyr}_{r,y,t,p} \land \textbf{prc_ts}_{r,p,s} \land (ACT_LOSPL_{r,y,p,'EX'} > 0))$

$$\begin{split} &VAR_UPS_{r,\mathbf{y},t,p,s,'EX'} \geq \\ & \left(COEF_CPT_{r,\mathbf{y},t,p} \left(VAR_NCAP_{r,\mathbf{y},p} - \sum_{ts \in SUP(s) \cap UPS(p)} VAR_UPS_{r,\mathbf{y},t,p,ts,'N'} \right) \times \right) \times \\ & PL_LDL_{r,\mathbf{y},p} \cdot PRC_CAPACT_{r,p} \cdot G_YRFR_s - VAR_ACT_{r,\mathbf{y},t,p,s} \right) \\ & \frac{AF_MIN_{r,\mathbf{y},p,ANNUAL}}{PL_LDL_{r,\mathbf{y},p} - AF_MIN_{r,\mathbf{y},p,ANNUAL}} \end{split}$$

6.3.6 Equation: EQ_ACTRAMP

Indices: region (r), vintage year (v), period (t), process (p), time slice (s), bound (bd)

Type: =

Related variables: VAR_ACT, VAR_NCAP, VAR_UPS

Related equations: EQE_CAPLOAD

Purpose: This equation defines maximum ramp-up and ramp-down rates for a standard process. The maximum ramp-rates are specified with the input parameter ACT_UPS(r,v,p,s,'UP') and ACT_UPS(r,v,p,s,'LO'), as fractions of the nominal on-line capacity per hour.

Remarks:

• The amount of on-line capacity is the full available capacity, unless start-ups / shutdowns have been enabled by using the parameter ACT_MINLD.

Notation:

- SUP(s) is the set of timeslices above timeslice s in the timeslice tree, but including also s itself
- UPS(p) is the set of timeslices with start-ups/shut-downs allowed for process p.

Equations:

$$\begin{split} EQ_ACTRAMP_{r,v,t,p,s,UP'} & \ni (\mathbf{rtp_vintyr}_{r,v,t,p} \land \mathbf{prc_ts}_{r,p,s} \land (ACT_UPS_{r,v,p,UP'} > 0)) \\ & \left(\frac{VAR_ACT_{r,v,t,p,s}}{G_YRFR_{r,s}} - \frac{VAR_ACT_{r,v,t,p,s-1}}{G_YRFR_{r,s-1}} - \left(VAR_UPS_{r,v,t,p,s,UP'} - VAR_UPS_{r,v,t,p,s,UO'} \right) \land ACT_UPS_{r,v,p,s,FX'} \right) \times \\ & \frac{2 \cdot RS_STGPRD_{r,s}}{8760 \times \left(G_YRFR_{r,s} + G_YRFR_{r,s-1} \right)} \leq \left(VAR_NCAP_{r,v,p} - \sum_{t \in SUP(s) \land UPS(p)} VAR_UPS_{r,v,t,p,ts,VP'} \right) \times \\ & COEF_CPT_{r,v,t,p} \times PRC_CAPACT_{r,p} \times ACT_UPS_{r,v,p,s,UP'} \\ & EQ_ACTRAMP_{r,v,t,p,s,UO'} \rightarrow (\mathbf{rtp_vintyr}_{r,v,t,p} \land \mathbf{prc_ts}_{r,p,s} \land (ACT_UPS_{r,v,p,UO'} > 0)) \end{split}$$

6.3.7 Equation: EQL_ACTUPC

Indices: region (r), vintage year (v), period (t), process (p), timeslice level (tsl), lim_type (l), time slice (s)

Type: \leq

Related variables: VAR_UPS, VAR_NCAP, VAR_RCAP

Related equations: EQE_ACTUPS, EQ_CAPLOAD

Purpose: This equation has two purposes, according to the lim_type (l):

- 1. It defines a lower limit for consecutive on-line / off-line hours of process capacity, such that capacity started-up cannot be immediately shut down again, or capacity shut-down cannot be immediately started up again. This purpose is served when lim type=LO/UP.
- 2. It defines a maximum number of start-up cycles for the process capacity within the timeslice cycle under the parent timeslice.

Remarks:

• The minimum on-line / off-line hours are defined by using the input attribute *ACT_TIME*_{*r*,*v*,*p*,*bd*}, where bd = LO/UP. The maximum number of start-up cycles is defined by using the input attribute *ACT_TIME*_{*r*,*v*,*p*,*W*}.

Notation:

- *SUP(s)* is the set of timeslices above timeslice *s* in the timeslice tree, but including also *s* itself,
- UPS(p) is the set of timeslices with start-ups/shut-downs allowed for process p,
- *P(s)* and *C(s)* refer to the parent timeslice of *s* and the set of child timeslices of *s*, respectively.

Equations:

Case A: Lower limit for consecutive on-line / off-line hours

$$\begin{split} EQ_ACTUPC_{r,v,t,p,dsl,UP,s} & \ni (\mathbf{rtp_vintyr}_{r,v,t,p} \land \mathbf{prc_ts}_{r,p,s} \land \mathbf{ts_group}_{r,dsl,s} \land \\ & (ACT_TIME_{r,t,p,UP'} > 0)) \\ & \sum_{ts \in C(P(s))} VAR_UPS_{r,v,t,p,ds'UP'} \times (\mathsf{mod}(Hour(s) - Hour(ts), 24) < ACT_TIME_{r,v,p,'UP'}) \leq \\ & \left(VAR_NCAP_{r,v,p} - \sum_{ts \in SUP(s) \cap UPS(p)} VAR_UPS_{r,v,t,p,ds'N'} \right) \end{split}$$

$$\begin{split} EQ_ACTUPC_{r,v,t,p,tsl,',LO',s} & \Rightarrow (\mathbf{rtp_vintyr}_{r,v,t,p} \land \mathbf{prc_ts}_{r,p,s} \land \mathbf{ts_group}_{r,tsl,s} \land \\ & (ACT_TIME_{r,t,p,',LO'} > 0)) \\ & \sum_{ts \in C(P(s))} VAR_UPS_{r,v,t,p,ts,',LO'} \land (\mathsf{mod}(Hour(s) - Hour(ts), 24) < ACT_TIME_{r,v,p,',LO'}) \leq \\ & (VAR_UPS_{r,v,t,p,s,',V'}) \end{split}$$

Case B: Maximum number of start-up cycles within parent timeslice cycle

$$\begin{split} EQ_ACTUPC_{r,y,t,p,ts|,N',s} & \ni (\mathbf{rtp_vintyr}_{r,y,t,p} \land \left(s \in \bigcup_{ts} \{P(ts) \mid \mathbf{prc_ts}_{r,p,ts}\}\right) \\ & \land \mathbf{ts_group}_{r,tsl,s} \land (ACT_UPS_{r,y,p,ANNUAL',N'} > 0)) \\ & \sum_{ts \in C(s)} VAR_UPS_{r,y,t,p,ts,UP'} \leq \\ \left(VAR_NCAP_{r,y,p} - \sum_{ts \in SUP(s) \cap UPS(p)} VAR_UPS_{r,y,t,p,ts,'N'}\right) \times ACT_UPS_{r,y,p,ANNUAL',N'} \end{split}$$

6.3.8 Equation: EQE_ACTUPS

Indices: region (r), vintage (v), period (t), process (p), timeslice level (tsl), timeslice (s)

Type: =

Related variables: VAR_UPS

Related equations: EQL_ACTUPS, EQ_CAPLOAD

Purpose: This equation establishes the relation between start-ups/shut-downs and the change in the amount of on-line capacity between successive timeslices. It is generated only when start-up costs have been defined for a standard process with *ACT_CSTUP*.

Notation:

• *UPS⁺(r,p,tsl)* is the set of timeslice levels with start-ups/shut-down costs defined for process p.

Equation:

 $EQE_ACTUPS_{r,v,t,p,tsl,s} \quad \Rightarrow (\mathbf{rtp_vintyr}_{r,v,t,p} \land UPS^+_{r,p,tsl} \land \mathbf{ts_group}_{r,tsl,s})$ $VAR_UPS_{r,v,t,p,s,'UP'} - VAR_UPS_{r,v,t,p,s,'LO'}$ = $VAR_UPS_{r,v,t,p,s-1,'N'} - VAR_UPS_{r,v,t,p,s,'N'}$

6.3.9 Equation: EQL_ACTUPS

Indices: region (r), vintage year (v), period (t), process (p), timeslice level (tsl), lim_type (l), time slice (s)

Type: \leq

Related variables: VAR_UPS

Related equations: EQE_ACTUPS, EQ_CAPLOAD

Purpose: This equation ensures that startup costs are being consistently applied when startup costs have been defined on multiple timeslice levels.

Notation:

- UPS⁺(r,p,tsl) is the set of timeslice levels with start-ups/shut-down costs defined for process **p**.
- P(r,s) refers to the parent timeslice of s in region r.

Equations:

Case A: lim_type='N'

$$\begin{split} EQL_ACTUPS_{r,v,t,p,tsl,'N',s} & \Rightarrow \left(\mathsf{rtp_vintyr}_{r,v,t,p} \land UPS^+_{r,p,tsl} \land \mathsf{ts_group}_{r,tsl,s} \right) \\ VAR_UPS_{r,v,t,p,s,'N'} & \leq VAR_UPS_{p,v,t,P(s),'EX'} \end{split}$$

Case B: lim_type='FX'

$$EQL_ACTUPS_{r,v,t,p,dsl,'FX',s} \rightarrow \left\{ \begin{aligned} \mathsf{rtp_vintyr}_{r,v,t,p} \wedge UPS_{r,p,tsl}^+ \wedge \\ \mathsf{ts_group}_{r,tsl,s} \wedge s \in \left\{ \bigcup_{sl} P(sl) \mid sl \in UPS^+(p) \right\} \end{aligned} \right\}$$

$$VAR_UPS_{r,v,t,p,s,'FX'} \leq \sum_{ts \in C(s)} VAR_UPS_{r,v,t,p,ts,'UP'}$$

6.3.10 Equation: EQ(l)_ASHAR

Indices: region (r), vintage year (v), period (t), process (p), time slice (s)

Type: As determined by the bd index of the input parameter FLO_SHAR:

- l = 'G' for bd = 'LO' yields \geq ,
- l = 'L' for **bd** = 'UP' yields \leq ,
- l = 'E' for **bd** = 'FX' yields =.

Related variables: VAR_FLO, VAR_ACT, VAR_SOUT

Related equations: EQ(*l*)_INSHR, EQ(*l*)_OUTSHR

Purpose: A share equation between process flows/activity is generated a process (**p**) in region (**r**) for time period (**t**) and each time-slice (**s**). The equation is similar to the equations EQ(1)_INSHR and EQ(1)_OUTSHR, but is only generated when the input parameter $FLO_SHAR_{r,v,p,c,c,g,s,bd}$ is specified in a non-standard way, for a commodity **c** and group **cg** such that **c** is not a member of group **cg**, or such that **c**=**cg**.

Remarks:

- Internally, the non-standard FLO_SHARs are converted into the corresponding FLO ASHAR parameters.
- In general, the constraint is generated on the level of the process flow variable for c, unless c='ACT, which will result in an annual level constraint.
- When **c**='ACT', the equation defines a bound on the amount of activity in proportion to the flows in the group **cg**, on the ANNUAL level.
- When cg='ACT, the equation defines a bound on the amount of flow of c in proportion to the activity, on the level of the process flow variable for c.
- When c=cg, and c is a member of the default shadow group, the share equation is generated for the flow of c in the total flow of commodities in the SPG, and either on the group level or on the WEEKLY level, whichever is higher. This feature makes it easy to define e.g. daily share constraints for a DAYNITE level process, such as fuel shares for plug-in hybrid cars.
- When the process is a storage process, the only valid share specification is *FLO_SHAR*_{r,v,p,c,ACT,s,bd}, where c is the discharge commodity of a timeslice storage. This generates a constraint between the output flow and the storage activity, which can be useful e.g. for preventing the use of the storage for a by-pass operation. The cg is set automatically to the SPG when the *FLO_SHAR* is converted into *FLO_ASHAR*.

$$EQ(l)_ASHAR_{r,y,t,p,c,cg,s} \rightarrow \begin{pmatrix} \mathsf{rtp_vintyr}_{r,v,t,p} \land (\mathsf{rpcs_var}_{r,p,c,s} \lor ((c = ACT') \land \mathsf{annual}_{s})) \\ \land \sum_{ts_map} FLO_ASHAR_{r,y,p,c,cg,ts,bd} \end{pmatrix}$$

Case A: Standard processes:

$$\sum_{\mathbf{rps}_{s} \leq \mathbf{2}_{r,p,sl}} FLO_ASHAR_{r,v,p,c,cg,sl,bd} \times RS_FR_{r,s,sl} \times \\ \left(\sum_{com \in \mathcal{C}g} \sum_{\mathbf{rtps}_{s} \leq \mathbf{varf}_{r,t,p,com,ts}} VAR_FLO_{r,v,t,p,com,ts} \times RTCS_TSFR_{r,t,p,com,sl,ts} + \\ \sum_{com \in \left[\mathsf{rpc}_{p} \geq \mathsf{pg}_{r,p,som} \right]} \sum_{\mathbf{prc}_{s} \in \mathsf{f}_{r,p,ts}} \frac{VAR_FLO_{r,v,t,p,com,ts}}{PRC_ACTFLO_{r,v,p,com}} \times RTCS_TSFR_{r,t,p,com,sl,ts} \\ \left\{ =; \leq; \geq \right\} \end{cases} \right)$$

$$\sum_{\substack{\text{rtpcs}_varf_{,tp,c,ts}}} VAR_FLO_{r,v,t,p,c,ts} \times RTCS_TSFR_{r,t,p,c,s,ts} + \\ \sum_{\substack{\text{come} \left\{ \frac{\text{rpc}_pg_{e,p,com}}{|c=^{+} \wedge CT'} \right\}} \sum_{\substack{\text{prc}_ts_{r,p,ts}}} \frac{VAR_FLO_{r,v,t,p,com,ts}}{PRC_ACTFLO_{r,v,p,com}} \times RTCS_TSFR_{r,t,p,com,s,ts}$$

Case B: Storage processes:

$$\sum_{\mathbf{rps}_s2_{r,p,d}} FLO_ASHAR_{r,v,p,c,'ACT',sl,bd} \times RS_FR_{r,s,sl} \times \left(\sum_{com\in \left\{ \begin{array}{c} rpe_pg_{r,p,cm} \\ |cg='ACT' \end{array} \right\}} \sum_{pre_br_{r,p,ts}} VAR_FLO_{r,v,l,p,com,ls} \times RTCS_TSFR_{r,l,p,com,sl,ls} \right) \\ \left\{ =; \leq ; \geq \right\} \\ \sum_{rms_eve} \frac{VAR_SOUT_{r,v,l,p,c,ls}}{PRC_ACTFLO_{r,v,l,p,c,ls}} \times RTCS_TSFR_{r,l,p,c,s,ls}$$

$$\sum_{\substack{\text{rpc}_{stg_{r,p,c}}\\\text{rps}_{vat_{p,c,ts}}}} \overline{PRC_ACTFLO_{r,v,p,c}}$$

6.3.11 Equation: EQ(l)_BLND

Indices: region (r), year (t), refinery product (ble), specification (spe)

Type: Any type, as determined by the value of the input parameter BL_TYPE(r,ble,spe):

- l = L for a value of 1 yields \leq .
- l = G' for a value of 2 yields ³.
- l = 'E' for a value of 3 yields =.

Related variables: VAR_BLND

Related equations: EQ_COMBAL

Purpose: The blending equations ensure that the characteristics of petroleum products (e.g. sulfur content, density, octane number, etc.) lie within specified limits, if desired.

Remarks:

- Parameter BL_COM contains the values of the blending specifications spe for the blending streams opr.
- Parameter BL_SPEC contains the value of the specification **spe** of the blending product **ble**.
- The blending variables VAR_BLND are expressed in volume units. If the characteristics of the blending streams **opr** and the product **ble** are not given in volume units (indicated by input parameter REFUNIT), the user has to provide a conversion parameter CONVERT which contains the density and energy content (by weight or by volume) of each blending stream. The conversion parameters are used to derive the coefficients RU_CVT of the blending streams in the blending equation.

Equation:

 $EQ(l)_BLND_{r,t,ble,spe}$ $\Rightarrow bl_type_{r,ble,spe}$

 $\sum_{opreble_ops} BL_COM_{r,ble,opr,spe} \cdot RU_CVT_{r,ble,spe,opr} \cdot VAR_BLND_{r,t,ble,opr}$

 $\left\{\!\leq\!\!;=\!\!;\geq\right\}$

$$\sum_{opreble_opr_{bk,opr}} BL_SPEC_{r,ble,opr,spe} \cdot RU_CVT_{r,ble,spe,opr} \cdot VAR_BLND_{r,t,ble,opr}$$

6.3.12 Equation: EQ_BNDCST

Indices: region (r), year1 (y1), period (t), year2 (y2), cost aggregation (costagg), currency (cur)

Type: =

Related variables: VAR_CUMCST

Related equations: EQ_OBJ

Purpose: This equation is generated when a bound is specified on regional costs, taxes and/or subsidies, either cumulative over a year range (using $REG_CUMCST_{r,yl,y2,agg,.cur,bd}$) or in given milestone years (using $REG_BNDCST_{r,y,agg,.cur,bd}$). It sets the level of the variable $VAR_CUMCST_{r,yl,y2,costagg,cur}$ equal to the cost expression, to be bounded accordingly.

Remarks:

- The available cost aggergations that can be bounded are listed in the table below.
- All the cost components related to investments are expressed in terms of annualized capital costs, i.e. as annuities paid in the year(s) in question. These components thus include interest during both construction and payback time.
- In all combined cost aggregations, subsidies are treated as negative costs when summed up with other cost/taxes, but when bounded alone they are treated as positive.

Cost aggregation ID	Description
INV	investment costs (annuities)
INVTAX	investment taxes (annuities)
INVSUB	investment subsidies (annuities)
INVTAXSUB	investment taxes-subsidies (annuities)
INVALL	= INV+INVTAXSUB (annuities)
FOM	fixed OM costs
FOMTAX	fixed operating taxes
FOMSUB	fixed operating subsidies
FOMTAXSUB	fixed operating taxes-subsidies
FOMALL	= FOM+FOMTAXSUB
FIX	= INV+FOM
FIXTAX	= INVTAX+FOMTAX
FIXSUB	= INVSUB+FOMSUB
FIXTAXSUB	= FIXTAX-FIXSUB
FIXALL	= FIX+FIXTAXSUB
COMTAX	commodity taxes
COMSUB	commodity subsidies
COMTAXSUB	commodity taxes-subsidies
FLOTAX	process commodity flow taxes
FLOSUB	process commodity flow subsidies
FLOTAXSUB	process commodity flow taxes-subsidies
ALLTAX	= FIXTAX+COMTAX+FLOTAX
ALLSUB	= FIXSUB+COMSUB+FLOSUB
ALLTAXSUB	= ALLTAX-ALLSUB

Notation:

- INVTAX(r,y) = the tax portion of the (virtual) variable INVTAXSUB
- INVSUB(r,y) = the subsidy portion of the (virtual) variable INVTAXSUB
- FIXTAX(r,y) = the tax portion of the (virtual) variable FIXTAXSUB
- FIXSUB(r,y) = the subsidy portion of the (virtual) variable FIXTAXSUB
- COMTAX(r,y) = the commodity tax portion of the (virtual) variable VARTAXSUB
- COMSUB(r,y) = the commodity subsidy portion of the (virtual) variable VARTAXSUB
- FLOTAX(r,y) = the flow tax portion of the (virtual) variable VARTAXSUB
- FLOSUB(r,y) = the flow subsidy portion of the (virtual) variable VARTAXSUB
- **cost_map**_{agg,costagg} = mapping coefficient between all cost aggregations and the component aggregations to be summed up (value = 0 / 1 / -1).

Remark: See the Section on the objective function for details on the expressions for the (virtual) cost variables mentioned above.

Equation:

$$\begin{split} EQ_BNDCST_{r,yl,t,y2,agg,cur} & \Rightarrow \left(\left(\sum_{bd} REG_CUMCST_{r,yl,y2,agg,cur,bd} <>0 \right) \land \left(t = \underset{\{t|M}{\operatorname{tr}(t-1) < y2} (M(tt)) \right) \right) \\ VAR_CUMCST_{r,yl,y2,agg,cur} = \\ & \sum_{yl \leq y \leq y2} INVCOST(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(INVTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} INVTAX(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(INVTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} INVSUB(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(INVTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLXCOST(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(INVTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLXUB(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(INVTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLXUB(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FOMTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLXUB(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FOMTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} COMTAX(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FOMTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLOTAX(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FOMTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLOTAX(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FOMTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLOTAX(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FOMTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLOSUB(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FOMTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLOSUB(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FOMSUB')}} \right) \\ & \sum_{yl \leq y \leq y2} FLOSUB(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FLOTAX')}} \right) \\ & \sum_{yl \leq y \leq y2} FLOSUB(r, y) \times \left(\operatorname{cost_map}_{\operatorname{agg(FLOSUB')}} \right) \end{aligned}$$

6.3.13 Equation: EQ(*l*)_BNDNET/PRD

Indices: region (r), period (t), commodity (c), timeslice (s)

Type: Any type, as determined by the bound index bd of COM_BNDNET/PRD:

- l = G' for **bd** = LO' (lower bound) yields \geq .
- l = E' for **bd** = FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .

Purpose: If the bound on the net or gross production of a commodity is specified for a timeslice being above the timeslice level of the commodity, the equation described here is generated. The bound on the net or gross production of a commodity is directly applied to the variable (VAR_COMNET, VAR_COMPRD), if the bound parameter is specified for a commodity timeslice (**com_ts**).

Remarks:

- The internal set **rcs_comts** used in the equation contains all timeslices at or above the timeslice level being defined for the commodity.
- The internal set **rtcs_varc** used in the summation part of the equation contains all timeslices (out of **com_ts**) and periods for which the commodity is available.
- The internal set **ts_map(r,s,ts)** used in the summation part of the equation contains for a given timeslice (s) all timeslices (ts) being at or below s in the timeslice tree.

Interpretation of the results:

Primal: Value of the net production of a commodity (production minus consumption) Dual: marginal cost of increasing the bound by one unit

Equation

 $EQ(l)_BND(NET / PRD)_{r,t,c,s}$ $\Rightarrow \left\{ \mathbf{rcs_com ts}_{r,c,s} \land (NOT \ \mathbf{com_ts}_{r,c,s}) \land COM _BND(NET / PRD)_{r,t,c,s,bd} \right\}$

$$\sum VAR _COM(NET / PRD)_{r,t,c,ts}$$

 $\mathit{ts}{\in}rtcs_varc_{r,t,\ell,ts}{\frown}ts_map_{r,s,ts}$

$$(\leq l \geq l =)$$

 $COM_BND(NET / PRD)_{r,l,c,s,bd}$
Sign according to the *l* equation index (must coincide with the **bd** index in parameter COM_BNDNET/PRD).

6.3.14 Equation: EQ(l)_CAFLAC

Indices: region (r), vintage year (v), period (t), process (p), time slice (s)

Type: As determined by the bd index of the standard availability parameter:

- l = L' for **bd** = UP' yields \leq ,
- l = 'E' for $\mathbf{bd} = 'FX'$ yields =.

Related variables: VAR_NCAP, VAR_FLO, VAR_IRE, VAR_SIN, VAR_SOUT

Related equations: EQ(l)_CAPACT, EQL_CAPFLO

Purpose: This equation relates the flows in the primary group of a process to its available existing capacity in period **t**. The existing capacity consists of investments made in the current and previous periods (VAR_NCAP) and investment decisions that have been made exogenously (*NCAP_PASTUPRC_RESID*). The availability of the existing capacity in a specific period **t** and timeslice **s** can be specified by the input attribute *NCAP_AFC*_{*r*,*v*,*p*,*c*,*g*,*st*}/*NCAP_AFCS*_{*r*,*v*,*p*,*c*,*g*,*st*}, where cg can be a single commodity in the PG, thereby making the process availability factor dependent on the output mix.

Remarks:

- The cg index in the input attributes NCAP_AFCr,v,p,cg,tsl / NCAP_AFCSr,v,p,cg,s can be
 either a single commodity in the PG, the reserved group name 'ACT' denoting the PG
 itself (for other than storage processes, see below), or a commodity type of the PG. Any
 NCAP_AFCSr,v,p,cg,s specified overrides an NCAP_AFCCr,v,p,cg,tsl specified on the same
 level, and any value specified for a single commodity overrides a value specified for a
 group containing the commodity.
- For storage and trade processes, which both can have the same commodity IN and OUT
 of the process, defining an NCAP_AFC/NCAP_AFCS both for the commodity type and
 for the single commodity itself results in the commodity availability being applied to the
 output flow while the group availability is applied to the input flow.
- For storage processes, defining an *NCAP_AFC/NCAP_AFCS* on the reserved group name 'ACT' defines a separate availability factor constraint (EQL_CAPFLO) for the storage level, and not for the flows in the PG.
- For trade processes, **prc_map**r,'DISTR',p can be also used for removing exports from contributing to the availability equation, if the process is bi-directional.

Special notation used for the equation formulation:

• $SX_{r,v,p,c,s}$ denotes an adjustment coefficient for storage inputs:

$$SX_{r,y,p,\varrho,\varphi} = \begin{cases} 0 & \text{if } \operatorname{top}_{\operatorname{OUT}}(c) \wedge \operatorname{top}_{\operatorname{IN}}(c) \wedge \neg NCAP _ AFCS_{r,y,p,\varrho,\varphi,\varphi} \\ \\ \frac{NCAP _ AFCS_{r,y,p,\varrho,\varphi,\varphi}}{\sum NCAP _ AFCS_{r,y,p,\varrho,\varphi,\varphi}} \\ 1 & \text{otherwise} \end{cases} \quad if \quad \operatorname{top}_{\operatorname{OUT}}(c) \wedge \operatorname{top}_{\operatorname{IN}}(c) \wedge \\ NCAP _ AFCS_{r,y,p,\varrho,\varphi,\varphi} \\ NCAP _ AFCS_{r,y,p,\varrho,\varphi,\varphi} \end{cases}$$

• *IX_{r,v,p,c,s}* denotes an adjustment coefficient for trade process exports:

$I\!X_{r,v,p,c,s}$	= <	$\left(\frac{NCAP_AFCS_{r, \mathbf{y}, p, c, s}}{\sum_{\mathbf{p}_{p}_Pg_{prg}}NCAP_AFCS_{r, \mathbf{y}, p, cg, s}}\right)$	<i>if</i> imp (c) \land exp (c) \land <i>NCAP</i> $_$ <i>AFCS</i> _{r,y,p,c,s} \land <i>NCAP</i> $_$ <i>AFCS</i> _{r,y,p,cg,s}
	ļ	1	otherwise

Equation:

$$\begin{split} EQ(l)_CAFLAC_{r,v,l,p,s} \rightarrow & \left(\begin{aligned} \mathbf{rtp_vintyr}_{r,v,l,p} \wedge NCAP_AFCS_{r,v,p,s} \wedge \\ (NCAP_AF_{r,v,p,p,d} \vee NCAP_AFCS_{r,v,p,s,d} \vee NCAP_AFA_{r,l,p,bd} \\ \\ \hline RC_ACTFLO_{r,v,p,c} \\ \left(\begin{array}{c} 1 \\ \overline{PRC_ACTFLO}_{r,v,p,c} & if NCAP_AFCS_{r,v,p,s,s} & given \\ \\ \sum \prod_{\mathbf{p}_\mathbf{p}_{\mathbf{b}_{\mathbf{b}_{\mathbf{s}_{\mathbf{n}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{n}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{s}_{\mathbf{n}}_{\mathbf{n}_{n$$

6.3.15 Equation: EQ(l)_CAPACT

Indices: region (r), vintage year (v), period (t), process (p), time slice (s)

Type: Determined by the bound index bd of NCAP_AF, NCAP_AFS or NCAP_AFA:

- $l = {}^{\prime}E'$ for **bd** = {}^{\prime}FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .
- l = G' for $\mathbf{bd} = LO'$ (lower bound) yields \geq .

Related variables: VAR_ACT, VAR_NCAP, VAR_FLO

Related equations: EQ_ACTFLO, EQ_COMBAL, EQ_INSHR, EQ_OUTSHR, EQ_PTRANS

Purpose: The capacity-activity equation relates the activity of a process to its available existing capacity in period **t**. The existing capacity consists of investments made in the current and previous periods (VAR_NCAP) and investment decisions that have been made exogenously (NCAP_PASTI/PRC_RESID). The availability of the existing capacity in a specific period **t** and timeslice **s** is specified by the availability factor. Three availability factors exist:

• NCAP_AF(r,v,p,s,bd):

Availability factor specified for a specific period and timeslice. If this availability factor is not specified for the process timeslices (**prc_ts**), the availabilities are aggregated/inherited according to the timeslice tree. Thus, for a process operating on the DAYNITE level it is sufficient to specify only one availability for the ANNUAL timeslice, which is then inherited to the DAYNITE timeslices.

• NCAP AFS(r,v,p,s,bd):

Availability factor specified for a specific period and timeslice. In contrast to NCAP_AF, this availability is not inherited/aggregated along the timeslice tree. If this availability is specified for a seasonal timeslice for a process operating on the DAYNITE level, the capacity-activity constraint is generated for the seasonal timeslice and sums over the DAYNITE activities. This gives the process flexibility how to operate within the seasonal timeslice as long as the overall seasonal availability restriction is fulfilled.

• NCAP AFA(r,v,p,bd):

Annual availability factor similar to NCAP_AFS being specified for the ANNUAL timeslice with the difference that NCAP_AFA is always applied in such a way as if the process is non-vintage dependent, even if it is specified as a vintaged one (**prc_vint**). Thus the annual availability factor is especially useful to calibrate the activity of a process in the first period(s) to the statistics irrespectively of its vintage structure and the vintage dependent activities (NCAP_AFS), which can be specified in addition to NCAP_AFA.

If the process is defined as a vintaged one (prc_vint), for each vintage year (v) of the existing capacity stock in period (t) a separate capacity-activity constraint will be generated

(exception NCAP_AFA), while for a non-vintaged process one capacity-activity constraint is generated that sums over all vintage years. In the latter case the vintage index of the equation $(EQ(I)_CAPACT(r,v,t,p,s))$ always equals the period index (v = t).

Remarks:

- For all process timeslices (**prc_ts**), NCAP_AF(r,t,p,s,'UP') is by default set to 1, unless NCAP_AF('UP') or NCAP_AF('FX') has been specified by the user. Thus, it is ensured that the activity of a process can never exceed its capacity. If for example only NCAP_AFA is specified by the modeler as annual availability for a process with a DAYNITE timeslice resolution, in addition to the annual activitycapacity constraint activity-capacity constraints with an availability of 100% are generated for the DAYNITE process timeslices.
- An average value of the availability factors (NCAP_AF/S) over the years in the period is used when an age-dependent 'Shape' is specified for them.
- **rtp_cptyr** identifies the capacities installed in period **v** still available in period **t**. This set takes into account that investments may be turned-off for certain periods (by PRC_NOFF). The condition is as under:

```
v,t such that

B(v) \ge B(t) - (COEF_RPTI*TLIFE) - ILED + 1

and
```

- B(v) ≤ E(t) ILED
 prc_vint is a set of processes for which attributes are changing over time and vintaging is required.
- Entries in **rtp_vintyr** are controlled by the same logic as applied to COEF_CPT combined with the vintaging consideration. Note **v** = **t** when no vintaging is required, or vintaging is turned off for a particular processes, where the sum over the previous investments is used instead of individual variables.
- COEF_AFr,v,t,p,s,bd will be read off a pre-processed table, after application of SHAPE and MULTI to the user provided availabilities (NCAP_AF/A/S).
- COEF_AF is calculated in the following manner:
 - aggregate if possible (pp_lvlbd.mod), otherwise inherit (in ppmain.mod)
 apply SHAPE and MULTI to NCAP AF/S
- For storage processes, the capacity describes the volume of the storage and the activity the storage content. For storage processes between timeslices (**prc_tgtss**, **prc_nstts**) parameter RS_STGPRD is used instead of G_YRFR. RS_STGPRD(r,s) equals the number of storage periods on the timeslice level of timeslice s in the whole year multiplied with the duration of its parent timeslice ts. Thus, the storage level VAR_ACT (and indirectly the storage in- and output flows VAR_SIN and VAR_SOUT) are scaled-up for the entire year.
 - Consequently, the value of RS_STGPRD(r,s) is:
 - o 1 for a seasonal storage,
 - o 365/7*G_YRFR(r,ts) for a weekly storage, where ts is the parent node of s,
 - 365*G YRFR(r,ts) for a daynite storage, where ts is the parent node of s.

Interpretation of the results:

- Primal: In case of an inequality constraint and no past investments (i.e. RHS is zero), the primal value describes the difference between the activity level and the maximum possible activity due to the installed capacity in the considered period and timeslice. If the primal value is negative, it means that the capacity is not fully utilized. In case of past investments, the RHS is not zero⁴², but has a positive value and corresponds to the possible activity due to the past investments. If the primal value equals the RHS value, the capacity is fully utilized. If not the difference (RHS minus primal value), where the primal value may also be negative, describes the possible unused activity production.
- Dual: The dual value is in case of an inequality constraint a negative number, when the constraint is binding. It describes the cost reduction caused by an additional capacity unit and can thus be interpreted as the value of the capacity. For a power plant for example it can be viewed as the part of the electricity price that can be used for covering the fixed operating and investment costs of the capacity (multiplied by the corresponding coefficient in the dual equation of the electricity flow variable). If NCAP AFS or NCAP AFA are applied for timeslices above the process timeslice level, in addition capacity-activity constraints (with a default value for NCAP_AF of 1 as upper bound) are generated for the process timeslices. The dual value of the constraints related to NCAP AFS or NCAP AFA serve as benchmark value of the capacity between the process timeslices. If for example NCAP_AFA is given for a power plant with a DAYNITE timeslice resolution (e.g. WD, WN, SD, SN), the NCAP AF related capacity constraints with an availability of 1 are usually binding only in one process timeslice level, e.g. WD. Now the dual variable of NCAP AFA can be seen as rent that must be covered in other process timeslices (WN, SD, SN) by the then prevailing electricity price, so that the model would decide to shift the scarce annual capacity from WD to another timeslice.

 $^{^{42}}$ GAMS moves all constants (e.g. past investments) on the RHS and the variables on the LHS of the equation. In the listing file the primal value of the equation can be found in the solution report under the LEVEL column. The RHS value is given under the column UPPER column in case of a <= inequality and in the LOWER column for a >= inequality. For an equality LOWER, LEVEL and UPPER value are the same.

Equation:

$$\begin{split} EQ(l)_CAPACT_{r,\mathbf{v},t,p,s} &\Rightarrow \mathsf{rtp_vintyr}_{r,\mathbf{v},t,p} \land \mathsf{prc_ts}_{r,p,s} \land \mathsf{rtp_vara}_{r,t,p} \land \\ & \left(NCAP_AF_{r,t,p,s} \lor NCAP_AFS_{r,t,p,s} \lor NCAP_AFA_{r,t,p}\right) \end{split}$$

$$\begin{cases} \sum_{ts \in [\text{prc_ts}_{r,p,ts} \cap \text{ts_map}_{r,s,ts}]} VAR_ACT_{r,y,t,p,ts} & \text{if } \neg \text{prc_map}_{r,'STGp} \\ \sum_{ts \in [\text{prc_ts}_{r,p,ts}]} \frac{VAR_ACT_{r,y,t,p,ts}}{RS_STGPRD_{r,ts}} \times RS_FR_{r,ts,s} & \text{if } \text{prc_map}_{r,'STG'p} \end{cases} \end{cases}$$

 $\left\{\!\leq\!\!;=\!\!;\geq\right\}$

$$\begin{aligned} & \text{Case1: Non-vintaged } \text{process}(\text{v}=\text{t}):\\ & \sum_{\{v \geq | r \texttt{tp}_cpt \texttt{y} \texttt{r}_{v,v2t_{P}}\}} \begin{pmatrix} \text{COEF}_AF_{r,v2,l,p,s,bd} \times \text{COEF}_CPT_{r,v2,l,p} \times \\ & (\text{VAR}_NCAP_{r,v2,p} + NCAP_PASTI_{r,v2,p} - \\ & (\text{VAR}_SCAP_{r,v2,l,p} & \text{if } PRC_RCAP_{r,p}) \end{pmatrix} \end{pmatrix} \quad \text{if } \neg \texttt{prc_vint}_{r,p} \end{aligned}$$

$$\begin{array}{c} \text{Case2: Vintaged process}(\textit{v} = \textit{vintage}): \\ & \left(\begin{matrix} \text{COEF}_AF_{r,\textit{v},t,p,\textit{s},bd} \times \text{COEF}_CPT_{r,\textit{v},t,p} \times \\ \left(\textit{VAR}_NCAP_{r,\textit{v},p} + \textit{NCAP}_PASTI_{r,\textit{v},p} - \\ \left(\textit{VAR}_SCAP_{r,\textit{v},t,p} \quad \textit{if} \; PRC_RCAP_{r,p} \right) \end{matrix} \right) \end{matrix} \quad \textit{if} \; \textit{prc_vint}_{r,p}$$

 $\times \Big[G_\textit{YRFR}_{r,s} \times (\textit{p} \not\in \textit{prc_map}_{r,\texttt{STG},p}) + 1 \times (\textit{p} \in \textit{prc_map}_{r,\texttt{STG},p}) \Big]$



D(t) = Duration of the period containing t

6.3.16 Equation: EQL_CAPFLO

Indices: region (r), vintage (v), period (t), process (p), commodity (c), time slice (s)

Type: \leq

Related variables: VAR_NCAP, VAR_SCAP, VAR_FLO, VAR_ACT, VAR_UPS

Related equations: EQ(l)_CAFLAC

Purpose: The equation defines a maximum level for a process flow (standard processes) or activity (only for storage processes) in relation to its capacity, according to an *NCAP_AFC/NCAP_AFCS* parameter specified by the user.

Remarks:

- The equation is generated only for process flows not in the PG, as the PG flows are handled by EQ_CAFLAC. However, independent EQL_CAPFLO constraints may be requested also for the PG flows by setting NCAP_AFCr, vp.p./ACT/tsl =-1.
- When defined for storage activity, note that the capacity is assumed to represent an
 annual production capacity equivalent to the amount produced by full power during
 one full year/week/day for SEASON/WEEKLY/DAYNITE level storage processes,
 respectively. The availability factor should be adjusted to correspond to the actual
 storage capacity. For example, a capacity of 1 GW is equal to 24 GWh for a
 DAYNITE storage, and if the real daily storage capacity is, say 8 GWh / GW, the
 maximum availability factor should be 0.333.
- The equation formulation is shown below is for the vintaged case only; the nonvintaged case differs in the RHS in the same way as in EQ(l)_CAPACT.

Notation:

• *P(s)* denotes the parent timeslice of *s*.

Equation (vintaged case only):

$$EQ_CAPFLO_{r,v,t,p,c,s} \rightarrow (\mathbf{rtp_vintyr}_{r,v,t,p} \land NCAP_AFC_{r,v,p,c,s} \land \neg \mathbf{rpc_pg}_{r,p,c})$$

$$\begin{pmatrix} \sum_{rtpc_varf_{rtpr,is}} VAR_FLO_{r,y,t,p,c,s} & \text{if } c \neq 'ACT' \\ \sum_{prc_fs_{rps,is}} \frac{VAR_ACT_{r,y,t,p,is} \times RS_FR_{r,is,s} \times G_YRFR_{r,s}}{G_YRFR_{r,P(ts)}} & \text{if } c = 'ACT' \end{pmatrix}$$

$$\leq$$

$$NCAP_AFC_{r,y,t,p,\varphi,\sigma} \times \begin{pmatrix} VAR_NCAP_{r,y,p} + NCAP_PASTI_{r,y,p} - \\ \sum_{pr \in rcap_{rp}} VAR_SCAP_{r,y,t,p} - \sum_{ts \in SUP(s) \cap UFS(p)} VAR_UPS_{r,y,t,p,ts,N'} \end{pmatrix}$$

$$COEF_CPT_{r,y,p,t} \times PRC_CAPACT_{r,p} \times G_YRFR_{r,s}$$
6.3.17 Equation: EQ_CAPLOAD

Indices: region (r), vintage year (v), period (t), process (p), time slice (s), lim_type (l)

Type: \leq

Related variables: VAR_ACT, VAR_NCAP, VAR_UPS

Related equations: EQE_ACTUPS

Purpose: This equation is used as a replacement for the standard EQ(I)_CAPACT equations for the process timeslices, whenever flexible minimum operating limits are defined for a standard process. It defines the maximum and minimum levels of activity in relation to the available capacity, taking also into account capacity that may have been shut-down during some timeslices. The difference to the standard EQ(I)_CAPACT equations is thus that EQ_CAPLOAD refers to the on-line capacity in each timeslice, while EQ(I)_CAPACT refers to the full available capacity.

Remarks:

- The flexible minimum operating limits are defined with the parameter ACT_MINLD(r,y,p). Any fixed lower bound availability factor at the process timeslice level is ignored when ACT_MINLD is defined.
- Star-ups/shut-downs of capacity are by default only allowed on the SEASON level, and without costs. More general dispatchability features can be activated by defining start-up costs, with the parameter ACT_CSTUP. Start-up costs can be optionally defined even on the SEASON level, if desired (see the Table below).
- Start-ups and shut-downs will always occur in pairs, and therefore any shut-down costs can be directly included in the ACT_CSTUP parameter. If start-ups on some level can be assumed without additional costs, it is advisable to leave ACT_CSTUP unspecified at that level. If the start-up costs are assumed zero on some timeslice level, they must be zero also on any higher levels.

	Input parameters specified			Resulting start-up capability				
			ACT	_CSTUP(TS	SLVL)	on	timeslice lev	vels
Case	ACT_MINLD	ACT_TIME(N)	SEASON	WEEKLY	DAYNITE	SEASON	WEEKLY	DAYNITE
0	No	NA	NA	NA	NA	(S)	(S)	(S)
1	Yes	No	Ţ.	-	-	S	-	. –
2	Yes	Yes	-	-	-	S	S	-
3	Yes	*	Yes	-	-	SC	-	1-1
4	Yes	*	-	Yes	-	S	SC	-
5	Yes	*	-	-	Yes	S	S	SC
6	Yes	*	Yes	Yes	-	SC	SC	-
7	Yes	*	Yes	_	Yes	S	S	SC
8	Yes	*	-	Yes	Yes	S	SC	SC
9	Yes	*	Yes	Yes	Yes	SC	SC	SC

S = start-ups enabled without cost

SC = start-ups enabled with costs

Notation:

- *AF_MAX_{r,v,p,t,s}* maximum operating level of online capacity of process *p*, vintage *v*, in period *t* and timeslice *s*, as defined by NCAP_AF('UP')
- *AF_MIN_{r,v,p,s}* minimum operating level of online capacity of process *p*, vintage *v* in timeslice *s*, as defined by ACT_MINLD
- *SUP(s)* is the set of timeslices above timeslice *s* in the timeslice tree, but including also *s* itself
- UPS(p) is the set of timeslices with start-ups/shut-downs allowed for process p;

Note: Only vintaged case shown below, see the RHS of EQ_CAPACT for the differences in the non-vintaged case.

Equations:

 $EQ_CAPLOAD_{r,v,t,p,s,UP} \rightarrow (\mathbf{rtp_vintyr}_{r,v,t,p} \land \mathbf{prc_ts}_{r,p,s} \land (ACT_UPS_{r,v,p,s,'EX'} > 0))$

 $VAR_ACT_{r,v,t,p,s} \leq$

$$\begin{split} AF_MAX_{r,y,t,p,s} \times & \begin{pmatrix} VAR_NCAP_{r,\sharp(y),p} + NCAP_PASTI_{r,y,p} - \\ \sum_{\textbf{prc_rcap}_{r,y}} VAR_SCAP_{r,y,t,p} - \sum_{ts\inSUP(s) \cap UPS(p)} VAR_UPS_{r,y,t,p,\sharps,\mathcal{W}} \end{pmatrix} \\ COEF_CPT_{r,y,p,t} \cdot PRC_CAPACT_{r,p} \cdot G_YRFR_{r,s} \end{split}$$

$$\begin{split} EQ_CAPLOAD_{r,v,t,p,s,LO} & \Rightarrow (\texttt{rtp_vintyr}_{r,v,t,p} \land \texttt{prc_ts}_{r,p,s} \land (ACT_UPS_{r,v,p,s,'EX'} > 0)) \\ VAR_ACT_{r,v,t,p,s} \geq \end{split}$$

6.3.18 Equation: EQ(*l*)_CPT

Indices: region (r), period (t), process (p)

Type: Any type, as determined either by the bound index **bd** of CAP_BND or the need to have a capacity variable (learning technology or capacity variable used in user constraint):

- l = G' for **bd** = LO' (lower bound) yields \geq , if no upper bound at the same time
- l = 'E' for **bd** = 'FX' (fixed bound), or for lower and upper capacity bound at the same time, or for learning technology or for capacity variable used in user constraint yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq , if no lower bound at the same time.

Related variables: VAR_NCAP, VAR_CAP

Related equations: EQ(l)_CAPACT

Purpose: This equation adds up the investments (VAR_NCAP), which have been made in the current and previous periods and still exist in the current period, and past investments being made before the beginning of the model horizon and either assigns it to the capacity variable VAR_CAP or applies directly lower or upper capacity bounds to it.

Remarks:

- It is generated only for those milestone year & process combinations that have a corresponding CAP_BND specification, for processes where there is a user constraint involving a capacity variable, and for processes being a learning technology (teg).
- In case that only a lower or an upper capacity bound is specified, the capacity bounds are directly used as RHS constants. In the other cases, the capacity variable is used instead.
- The set **rtp_varp(r,t,p)** describes the cases where a capacity variable is needed:
 - Capacity variable is used in a user constraint,
- Lower and upper capacity bound are specified for the same period. In this case it is more efficient to generate one capacity variable by one EQE_CPT equation and bound the variable instead of generating the two equations EQL_CPT and EQG_CPT.

Equation:

$$\begin{split} & EQ(l)_CPT_{r,t,p} \ni CAP_BND_{r,t,p,bd} \lor \mathbf{teg_p} \lor \mathbf{rtp_varp}_{r,\mathbf{t,p}} \\ & VAR_CAP_{r,t,p} \asymp \left(\mathbf{rtp_varp}_{r,\mathbf{t,p}} \lor CAP_BND_{r,t,p,'FX'} \lor \mathbf{teg_p}\right) \\ & + CAP_BND_{r,t,p,'LO'} \asymp \left[\left(NOT\ \mathbf{rtp_varp}_{r,\mathbf{t,p}}\right) \land CAP_BND_{r,t,p,'LO'} \right] \\ & + CAP_BND_{r,t,p,'UP'} \asymp \left[\left(NOT\ \mathbf{rtp_varp}_{r,\mathbf{t,p}}\right) \land CAP_BND_{r,t,p,'UP'} \right] \end{split}$$

 $\left\{\!\leq\!\!;=\!\!;\geq\!\right\}$

$$\sum_{v \in r \mathbf{p_cplyr_{r,v,t,p}}} COEF_CPT_{r,v,t,p} \times \begin{pmatrix} VAR_NCAP_{r,v,p} \times (v \in MILESTONYR) \\ + NCAP_PASTI_{r,v,p} \times (v \in PASTYEAR) \\ -VAR_SCAP_{r,v,t,p} \times ((r, p) \in \mathbf{prc_rcap}) \end{pmatrix}$$

where

 $COEF_CPT_{r,v,t,p}$ as defined in equation $EQ(l)_CAPACT$

6.3.19 Equation: EQ(l)_COMBAL

Indices: region (r), period (t), commodity (c), timeslice (s)

Type: Determined by the user-supplied set com_lim. Defaults are:

- *l* = 'G' (lim = 'LO' in com_lim) for energy carriers (com_tmap(r,c, 'NRG')), demands (com_tmap(r,c, 'DEM')), and emissions (com_tmap(r,c, 'ENV')); yields ≥ type of equation; production has to be greater or equal consumption if no upper bound at the same time
- l = 'E' (**lim** = 'FX' in **com_lim**) for materials (**com_tmap**(r,c, 'MAT')) and financial commodities (**com_tmap**(r,c, 'FIN')); yields = type of equation; production has to be equal consumption if no upper bound at the same time

Related variables: VAR_ACT, VAR_FLO, VAR_COMNET, VAR_COMPRD, VAR_IRE, VAR_NCAP, VAR_SIN/OUT, VAR_BLND, VAR_ELAST

Purpose: This equation ensures that at each period and time-slice, the total procurement of a commodity balances its total disposition. A commodity may be procured in several different ways: imported, produced by technologies (activity and capacity based), released at retirement of some investments. A commodity may be disposed of in several other ways: exported, consumed by technologies (activity or capacity based) or by a demand, or "sunk" at investment time of a process. The default type for the balance constraint of an energy carrier and for an emission is \geq , which allows procurement to exceed disposition. This may be important in order to avoid some infeasibilities dues to rigid processes with many outputs or inputs. The default sign is = for materials. Both defaults may be modified by the user by the set **com_lim**.

Remarks:

- The commodity balance is generated for the timeslices (s) according to the user defined sets com tsl or com ts.
- When there are one or more of the attributes BND/CST/SUB/TAX/CUM relating to
 production of the commodity, EQE_COMPRD is generated in addition to this equation.
 EQE_COMPRD simply creates a new variable (VAR_COMPRD) equal to the production
 part of the LHS of the balance constraint (see expression COMSUP below)
- Similarly, if there are relevant coefficients for the net production of the commodity, the expression VAR_COMNET is created, containing the net production, and used in the RHS (see below).
- Note that CAL_FLOFLO(r,t,p,c,s,io) table stores the complete expressions (*coefficients and variables*) giving the flow of each commodity.
- The investment related input flows are assumed to be spread uniformly throughout the commodity lead-time, NCAP_CLED, ending exactly at the end of NCAP_ILED (default value for NCAP_CLED is NCAP_ILED).
- Commodity output flows related to dismantling are assumed to occur uniformly over NCAP_DLIFE, and to start right after NCAP_DLAG (default value: NCAP_DLIFE =1).

- Net/gross production of other commodities can be aggregated to the production side of the commodity balance by using the COM_AGG attribute.
- Capacity-related input/output flows can be defined with NCAP_COM, which has the 'io' index. Examples exist for (physical) consumption as well as release, land use by hydro dams and methane emissions from them, respectively.

EQ_COMBAL reads schematically as follows: $Procurement - Disposition \ \geq \text{ or } = \ COEF \ FBRHS$

where COEF_FBRHS is 0 for all balance equations, except for demand balances where it is equal to a positive parameter. In addition, COEF_FBRHS is equal to a variable when the equation is used to define the variables VAR_COMPRD or VAR_COMNET.

This is expressed mathematically as the following equation, whose coefficients will be further developed in what follows.

Interpretation of the results:

- Primal: In case of an inequality constraint of the commodity balance, the primal value corresponds to the value which is obtained when all terms with variables are moved to the LHS of the equation and all constants, e.g. terms with the demand parameter COM_PROJ, are moved to the RHS side. The primal value equals the value of the LHS side. Thus, the commodity balance is binding when its primal value equals its RHS constant, it is non-binding, i.e., production exceeds consumption if the primal value is greater than the RHS constant⁴³.
- Dual: The dual variable (shadow price) of the commodity balance describes the internal value of the commodity. If the commodity balance is binding, i.e., consumption equals production, the shadow price describes the cost change in the objective function induced by an increase of the commodity demand by one unit. Since the LHS of the commodity balance describes the difference between production and consumption, this additional demand may be covered by an increase in production or by a decrease in consumption. In the first case the shadow price is determined by activities on the supply side of the commodity, while in the latter case saving measures on the demand side of the commodity are setting the shadow price. Note that when a peaking constraint (EQ PEAK) for the considered commodity exists, the price consumers must pay during peak hours depends not only on the shadow price of the commodity balance but also on the shadow of the peaking constraint (if the flow variable of the consuming technology has the same timeslice resolution as the commodity and the peaking parameters COM PKFLX=0 and FLO PKCOI=1, the price to consumers is simply the sum of the two shadow prices; in other cases the dual constraint of the flow variable should be inspected to identify the correct coefficients for the two shadow prices).

 $^{^{43}}$ The primal value and the RHS constant of an equation can be found in the GAMS listing file in solution report part. The LEVEL value column corresponds to the primal value, the LOWER level value equals the RHS of a constraint of type >= and the UPPER level value equals the RHS of a constraint of a type <=.





We now show the detailed calculation of the Right-hand-side

 $COEF _ FBRHS$:

DoCase

Case	$\ni COM$	$BNDNET \lor COM$	_CUMNET	
	$\lor COM$	$_CSTNET \lor COM$	$_SUBNET \lor COM$	_TAXNET

COEF _FBRHS = VAR _COMNET

Case >COM_BNDPRD \COM_CUMPRD \COM_CSTPRD \COM_SUBPRD \COM_TAXPRD

 $COEF_FBRHS = VAR_COMPRD$

Case COM_PROJ

 $COEF _FBRHS = COM _PROJ \times COM _FR$

Otherwise

 $COEF _FBRHS = 0$

Endcase



The parameter RTCS_TSFR is used to match the timeslice resolution of flow variables (VAR_FLO/VAR_IRE) and commodities. RTCS_TSFR is the coefficient of the flow variable, which is producing or consuming commodity (\mathbf{c}), in the commodity balance of \mathbf{c} . If timeslice \mathbf{s} corresponds to the commodity timeslice resolution of \mathbf{c} and timeslice $\mathbf{s1}$ to the timeslice resolution of the flow variable two cases may occur:

- The flow variables are on a finer timeslice level than the commodity balance (first case in the formula above, ts_map(r,s,s1) is true): in this case the flow variables with timeslices s being below ts in the timeslice tree are summed to give the aggregated flow within timeslice s1. RTCS TSFR has the value 1.
- 2) The flow variables are on coarser timeslice level than the commodity balance: in this case the flow variable is split-up on the finer timeslice level of the commodity balance according to the ratio of the timeslice duration of s to s1: RTCS_TSFR has the value = COM_FR(r,s) / COM_FR(r,s1) for demand commodities and G_YRFR(r,s) / G_YRFR(r,s1) otherwise. When COM_FR is used, the demand load curve is moved to the demand process. Thus, it is possible to model demand processes on an ANNUAL level and ensure at the same time that the process follows the given load curve COM_FR.



Investment Related Flow Coefficients

Intermediate Notation:

 $BCF = B(v) + NCAP_ILED - NCAP_CLED$ Beginning year of commodity flow $ECF = B(v) + NCAP_ILED - 1$ Ending year of commodity flow

Note that these flows never need to be carried across 'long' periods, because the construction never exceeds the end of period v if v is 'long'

 $COEF_ICOM$:

$if (v = t) \land (IL + TL < D(t))$		
$= COEF _ RPTINV \times \frac{NCAP _ ICOM_{y}}{D(t)}$	Case IV	
where $COEF_RPTINV = \left\langle \frac{D(t) - ILED_t}{TLIFE_t} \right\rangle$		1
else		
$= Max \left(\frac{1 + Min(ECF, E(t)) - Max(BCF, B(t))}{D(t)} \times \frac{NCAP - ICOM_{v}}{NCAP - CLED_{v}}\right),$	$0 \right) \qquad \qquad Cases I, II, III$	
endif		

Case I	Case II
NCAP_CLED	-
+ t +	-t t
Case III	Case IV
<→	$\longleftrightarrow \longleftrightarrow \longleftrightarrow$
t t	+ t

Dismantling Related Flow Coefficients

Intermediate Notation:

 $BCF = B(v) + NCAP_ILED + NCAP_TLIFE + NCAP_DLAG$ Start year of commodity flow.

 $ECF = B(v) + NCAP _ ILED + NCAP _ TLIFE + NCAP _ DLAG + NCAP _ DLIFE - 1$ End year of commodity flow.





6.3.20 Equation: EQE_COMPRD

Indices: region (r), period (t), commodity (c), timeslice (s)

Type: =

Related variables: VAR_ACT, VAR_FLO, VAR_COMNET, VAR_COMPRD, VAR_IRE, VAR_NCAP, VAR_SOUT, VAR_BLND, VAR_ELAST

Related equations: EQ(1)_COMBAL, EQ(1)_BNDPRD, EQ(1)_CUMPRD, EQ_OBJVAR

Purpose: This equation generates a variable VAR_COMPRD equal to the total supply of the commodity, i.e. import + production (activity and capacity based) + investment-time outflow + dismantling related outflows, in each period and time slice. Note that this excludes demand reduction (in the case of a demand commodity).

Remarks:

• Enables the application of bounds to the annual or cumulative production of commodities. This is also needed to incorporate cost/sub/tax attributes on commodity production.

Equation:

$$\begin{split} EQE _COMPRD_{r,t,c,s} & \Rightarrow COM _BNDPRD \lor COM _CUMPRD \\ & \lor COM _CSTPRD \lor COM _SUBPRD \lor COM _TAXPRD \end{split}$$

 $COMSUP = VAR _ COMPRD_{r,t,c,s}$

This refers to the term marked "COM_SUP", on equation EQ_COMBAL

6.3.21 Equation: EQ_CUMFLO

Indices: region (r), process (p), commodity (c), year1 (y1), year2 (y2)

Type: =

Related variables: VAR_ACT, VAR_FLO, VAR_CUMFLO

Related equations: EQ_CUMNET, EQ_CUMPRD

Purpose: This equation is generated whenever the input parameter $FLO_CUM_{r,p,c,yl,y2}$ or $ACT_CUM_{r,p,yl,y2}$ has been specified, for bounding the cumulative amount of process flow or activity. It is also generated when the input parameter UC_CUMFLO or UC_CUMACT has been specified. It sets the variable $VAR_CUMFLO_{r,p,c,yl,y2}$ equal to the cumulative flow/activity expression, to be bounded accordingly or to be referred to in a user constraint.

Remarks:

• The internal set **rpc_cumflo**_{r,p,c,y1y2} is set according to any user-defined *FLO_CUM*_{r,p,c,y1y2}, *ACT_CUM*_{r,p,y1,y2}, *UC_CUMFLO or UC_CUMACT*, with the reserved commodity name 'ACT' used for *ACT_CUM* and *UC_CUMACT*.

Equation:

$EQ_CUMFLO_{r,p,c,yl,t,y2} \rightarrow (rpc_{r,p,c} \land rpc_cumflo_{r,p,c,y1,y2})$

if $c \neq$ 'ACT':

 $\sum_{t=T(y1)}^{t=T(y2)} \sum_{\substack{s \in \mathsf{rpes}_v \text{ warf}_{tipes} \\ v \in \mathsf{rp}_v \text{ large}_t \text{ set}_{p} \neq s}} [Min\{E(t), y2\} - Max\{B(t), y1\} + 1] \times VAR_FLO_{r, v, t, p, c, s} = VAR_CUMFLO_{r, p, c, y1, y2}$

if c = 'ACT':

 $\sum_{t=\mathcal{T}(y^2)}^{t=\mathcal{T}(y^2)} \sum_{\substack{s \in p^* c_t_{s_{tps}} \\ v \in r_p_vinty_{r_{syp}}}} [Min\{E(t), y^2\} - Max\{B(t), y^1\} + 1] \times VAR_ACT_{r, y, t, p, s} = VAR_CUMFLO_{r, p, o, y^1, y^2}$

Bounds:

$$\begin{split} &VAR_CUMFLO.LO_{r,p'ACT',yl,y2} = ACT_CUM_{r,yl,y2,'LO'}\\ &VAR_CUMFLO.UP_{r,p'ACT',yl,y2} = ACT_CUM_{r,yl,y2,'UP'}\\ &VAR_CUMFLO.FX_{r,p'ACT',yl,y2} = ACT_CUM_{r,yl,y2,'TX'}\\ &VAR_CUMFLO.LO_{r,p,c,yl,y2} = FLO_CUM_{r,p,c,yl,y2,'UP'}\\ &VAR_CUMFLO.UP_{r,p,c,yl,y2} = FLO_CUM_{r,p,c,yl,y2,'UP'}\\ &VAR_CUMFLO.FX_{r,p,c,yl,y2} = FLO_CUM_{r,p,c,yl,y2,'TX'}\\ \end{split}$$

6.3.22 Equation: EQ_CUMNET/PRD

Indices: region (r), year1 (y1), year2 (y2), commodity (c)

Type: =

Related variables: VAR_COMNET/VAR_COMPRD, VAR_CUMCOM

Related equations: EQ(l)_COMBAL, EQE_COMPRD

Purpose: This equation defines a variable representing the cumulative amount of net release or total gross production of a commodity, primarily for bounding the variable according to the bound parameter COM_CUMNET/PRD. The constraint concerns net release/production over an arbitrary number of consecutive years between the year (y1) and year (y2) as given in the parameter COM_CUMNET/PRD.

Remarks:

- It is possible to have multiple cumulative bounds of any type.
- The total time span for calculating the cumulative production need not consist of an exact number of periods.
- The cumulative bounds are expressed annually only.
- The sign of the bound is indicated by the l equation index.

Interpretation of the results:

Primal: The primal value describes the cumulative net release/the cumulative production of commodity c between the years y1 and y2.

Dual: The dual value of the constraint describes the change in the objective function if the bound parameter is increased by one unit. The increase of an upper bound yields a reduction of the total costs (dual value is negative), since the system wants to use more of this commodity. The increase of a lower bound yields an increase of the total costs (dual value is positive), since the system has to be forced to use more of an uncompetitive commodity (the commodity itself or the technologies utilizing it maybe too expensive). The dual value of a cumulative production constraint can also be interpreted as a tax/subsidy that is applied between the years y1 and y2 to reach the same cumulative productions as specified in the bound (the tax/subsidy has to be adjusted by the discount rate).

Equation:

 $EQ(l) _CUMNET_{r,y1,y2,c} \ni COM _CUMNET_{r,y1,y2,c,l}$

 $\sum_{t=\mathcal{T}(y1)}^{t=\mathcal{T}(y2)} \sum_{s \in \mathsf{rtes}} [Min\{E(t), y2\} - Max\{B(t), y1\} + 1] \times VAR _COMNET_{r,t,c,s} = VAR _CUMCOM_{r,c,NET',y1,y2} + 1] \times VAR _COMNET_{r,t,c,s} + 1] \times VAR _COMNET_{r,t,c,$

The internal set **rtcs_varc** gives the periods at which the commodity is available (usually all periods, but the user can turn off periods by the set **com_off**), and the timeslices as defined by the user in **com_tsl** or **com_ts**.

 $EQ(l) _CUMPRD_{r,y1,y2,c,s} \ni COM _CUMPRD_{r,y1,y2,c,s}$

 $\sum_{t=T(y1)}^{t=T(y2)} \sum_{s \in \mathsf{rts}_var_{\mathsf{r},t,s}} [Min\{E(t), y2\} - Max\{B(t), y1\} + 1] \times VAR_COMPRD_{r,t,c,s} = VAR_CUMCOM_{r,c,!PRD',y1,y2} + 1] \times VAR_CUMCOM_{r,c,!PRD',y1,y2$

Bounds:

$$\begin{split} &VAR_CUMCOM.LO_{r,c,NET',yl,y2} = COM_CUMNET~r,~yl,y2,c,'LO'\\ &VAR_CUMCOM.UP_{r,c,NET',yl,y2} = COM_CUMNET~r,~yl,y2,c,'UP'\\ &VAR_CUMCOM.FX_{r,c,'NET',yl,y2} = COM_CUMNET~r,~yl,y2,c,'LO'\\ &VAR_CUMCOM.LO_{r,c,'PRD',yl,y2} = COM_CUMPRD~r,~yl,y2,c,'LO'\\ &VAR_CUMCOM.UP_{r,c,'PRD',yl,y2} = COM_CUMPRD~r,~yl,y2,c,'UP'\\ &VAR_CUMCOM.FX_{r,c,'PRD',yl,y2} = COM_CUMPRD~r,~yl,y2,c,'UP'\\ &VAR_CUMCOM.FX_{r,c,'PRD',yl,y2} = COM_CUMPRD~r,~yl,y2,c,'TX'\\ \end{split}$$

6.3.23 Equation: EQ_CUMRET

Indices: region (r), vintage year (v), period (t), process (p)

Type: =

Related variables: VAR_RCAP, VAR_SCAP

Related equations: EQ_DSCRET, EQ_SCAP

Purpose: This equation defines the relation between the early retirements of capacity occurring in each period t and the cumulative retirements over all periods $tt \le t$, by vintage. Its main purpose is to define the early retirements of capacity by each period, in order to be able to bound them directly with the attribute RCAP_BND.

Equation:

$$EQ_CUMRET_{r,v,t,p} \quad \Rightarrow (\mathbf{rtp_cptyr}_{r,v,t,p} \land \mathbf{prc_rcap}_{r,p})$$
$$VAR_SCAP_{r,v,t,p} = VAR_RCAP_{r,v,t,p} + \sum_{t-1 \in \{u| \mathsf{rtp_cptyr}_{r,v,x,p}\}} VAR_SCAP_{r,v,t-1,p}$$

6.3.24 Equation EQ_DSCNCAP

Indices: region (r), milestone year (t), process (p)

Type: =

Related variables: VAR_DNCAP, VAR_SNCAP, VAR_NCAP

Related equations: EQ_DSCONE

Purpose: The investment variable of the technology p in period t and region r can take only specific unit sizes given by the parameter NCAP_DISC. This equation defines the investment variable to be equal to the sum over the different unit sizes each multiplied by the corresponding decision variable VAR_DNCAP. However, the sister equation EQ_DSCONE restricts this sum to a single term only (i.e. a single unit – of a specific size – is allowed to be invested in at period t). Alternatively, if NCAP_SEMI is defined, the equation defines the investment variable to be equal to the semi-continuous variable VAR_SNCAP.

Remarks:

- The set **unit** contains the names of capacity blocks/units that can be added, the set contain integer numbers going from '0' to '100'. The unit name '0' is used to describe the decision that no capacity should be added.
- The set **prc_dscnap(r,p)** contains the processes **p** (in region **r**) for which the discrete capacity formulation should be used
- The parameter NCAP_DISC(r,t,p,u) is the allowed capacity size of unit u; e.g. the size of unit '1' could be 50 MW, unit '2' 100 MW and unit '3' 500 MW. The size of unit '0' is automatically set to zero (EPS). If all unit sizes are taken equal, the formulation allows the repeated investment of a basic unit (as many as 100 times, in integer numbers).
- The parameter NCAP_SEMI(r,t,p) can alternatively be used for defining the investment variable VAR_NCAP semi-continuous, with the lower bound defined by NCAP_SEMI, and upper bound by NCAP_BND. If NCAP_BND is not defined, upper bound is assumed equal to the lower bound.
- VAR_DNCAP(r,t,p,u) is a binary decision variable describing whether the capacity unit of technology **p** should be added in period **t** or not. Some solvers for mixedinteger problems, as CPLEX or XPRESS, allow the definition of variables as socalled SOS1 sets (special ordered sets) in order to improve the solution process. An SOS1 set is defined as a set of variables of which only one variable can take a nonzero value. VAR_DNCAP is currently defined as an SOS1 variable. Not all solvers support this option, in these cases the variable type should be changed to a binary variable in the file **mod_vars.dsc**.

Equation:

$$EQ_DSCNCAP_{r,t,p} \qquad \rightarrow (rp_dscncap_{r,p} \land rtp_{r,t,p})$$

$$VAR_NCAP_{r,t,p} = \sum_{\mu \in unit} (VAR_DNCAP_{r,t,p,\mu} \times NCAP_DSC_{r,t,p,\mu}) + (VAR_SNCAP_{r,t,p} \quad if \ NCAP_SEMI_{r,t,p} \ given)$$

6.3.25 Equation: EQ_DSCONE

Indices: region (r), milestoneyear (t), process (p)

Type: =

Related variables: VAR_DNCAP, VAR_NCAP

Related equations: EQ_DSCNCAP

Purpose: The equation ensures that only one of the multiple unit sizes allowed for technology \mathbf{p} (described by NCAP_DSC(r,t,p,u)) can be added in period t.

Equation

 $EQ_DSCONE_{r,t,p} \rightarrow \left(\mathbf{rp_dscncap}_{r,p} \land \mathbf{rtp}_{r,t,p} \right)$

 $\sum_{u \in \textbf{unit}} V\!A\!R_DNC\!A\!P_{r,t,p,u} = 1$

Note that VAR _DNCAP must be declared as a binary var iable (taking values 0 or 1 only)

6.3.26 Equation: EQ_DSCRET

Indices: region (r), vintage year (v), period (t), process (p)

Type: =

Related variables: VAR_NCAP, VAR_SCAP, VAR_DRCAP

Related equations: EQ_CUMRET

Purpose: This equation defines the cumulative early retirement variable *VAR_SCAP* to be a multiple of a user-defined block size, specified by *RCAP_BLK*. The amount of capacity retired early can thus only take discrete values $n \times RCAP_BLK$, n=0,1,2,3,....

Remarks:

- Because the residual capacity can be defined rather freely by PRC_RESID, a forced component (*RTFORC*) is added into the cumulative retirements for processes having existing capacities defined with PRC_RESID, corresponding to the trajectory given by PRC_RESID.
- Because it should always be possible to retire the remaining residual capacity in full (regardless of the block size specified), that amount is added as a second alternative block size, which can only be retired in a multiple of 1.

Equation:

$$\begin{split} EQ_DSCRET_{r,y,t,p} & \rightarrow (\mathbf{rtp_cptyr}_{r,y,t,p} \land RCAP_BLK_{r,y,p}) \\ VAR_SCAP_{r,y,t,p} - RTFORC_{r,y,t,p} = \\ & RCAP_BLK_{r,y,p} \times VAR_DRCAP_{r,y,t,p,2} + \\ & (NCAP_PASTI_{r,y,p} - RTFORC_{r,y,t,p}) \times VAR_DRCAP_{r,y,t,p,1} \end{split}$$

6.3.27 Equation: EQ(*l*)_FLOBND

Indices: region (r), period (t), process (p), commodity group (cg), timeslice (s)

Type: Any type, as determined by the bound index bd of FLO_BND:

- l = G' for **bd** = LO' (lower bound) yields \geq .
- $l = {}^{\circ}E'$ for **bd** = {}^{\circ}FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .

Purpose: Bound on the sum of process flows in a given commodity group (**cg**) for a particular process (**p**) in period (**t**) and timeslice (**s**).

Remarks:

- The constraint bounds the flows in a specific period (t) irrespectively of the vintage years of the process capacity.
- The bound can be defined for a single commodity or a group of commodities linked to the process (**p**). In the latter case, a commodity group (**cg**) must be defined by the user (through **com_gmap**).
 - The constraint is generated if one of the following conditions is true:
 - Process (p) is vintaged, or
 - The sum of several process flows given by the commodity group (cg), and not only a single process flow, should be bounded, or
 - The timeslice resolution of the flow variables are below the timeslice (s) of the bound parameter.

In other cases, the bound can be directly applied to the corresponding flow variable, so that no extra equation is needed.

- The timeslice level (s) of the bound must be at or higher than the timeslice level of the process flows (rtpcs_varf).
- If *FLO_BND* is defined for a trade process, the constraint bounds the sum of imports and exports when **cg** is a single commodity, but net imports if **cg** is a true commodity group (i.e. is itself not a commodity).

Interpretation of the results:

Primal: If the primal value equals the bound parameter, the constraint is binding.

Dual: The dual value describes for a lower/upper bound the cost increase/decrease in the objective function, if the bound is increased by one unit. It may also be interpreted as subsidy/tax needed to reach the given bound value.

Notation used in formulation:

 XS_{cg,ie} denotes a sign coefficient for trade flows, such that XS_{cg,TMP} = 1 for all cg, XS_{c,TXP} = 1 for all c, and XS_{cg,TXP} = −1 for all other (true) cg.



where the equation sign is indicated by equation index I based on the bound ty pebd.

6.3.28 Equation: EQ(l)_FLOFR

Indices: region (r), period (t), process (p), commodity (c), timeslice (s)

Type: Any type, as determined by the bound index bd of FLO_FR:

- l = G' for **bd** = LO' (lower bound) yields \geq .
- l = 'E' for **bd** = 'FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .
- Purpose: 1) Relationship in period (t) between the total annual flow and the flow in a particular timeslice (s) for a specific process (p). This is the standard usage of the *FLO_FR* parameter, which may be used even for defining a full load curve for a process flow.
 2) Relationship in period (t) between the the flow level in a particular flow timeslice (s) and the average level under all timeslices under its parent timeslice for a specific process (p). This variant will only be used when *FLO_FR* is

levelized to the flow timeslices (**rpc_var**), which is triggered by defining any *FLO_FR* value for that process flow at the ANNUAL level.

Remarks:

The sign of the equation determines whether the flow in a given timeslice is rigidly (=) or flexibly $(\geq; \leq)$ linked to the annual flow (or the parent flow level). The constraint bounds the flows irrespectively of the vintage years of the process capacity.

Equation:

Case A: Standard EQ(l)_FLOFR: fraction of flow in total ANNUAL flow



where the equation sign is indicated by equation index 1.

Case B: Levelized EQ(l)_FLOFR: flow level in proportion to average level under parent

 $EQ(l)_FLOFR_{r,t,p,c,s} \quad \rightarrow \left\{ \mathbf{rtpcs_var}_{\mathbf{r,t,p,c,s}} \land FLO_FR_{r,t,p,c,s,bd} \right\}$

where the equation sign is indicated by equation index l.

6.3.29 Equation: EQ(*l*)_FLMRK

Indices: region (r), period (t), process (p), commodity (c), time-slice (s)

Type: Any type, as determined by the bound index bd of FLO MARK/PRC MARK:

- l = G' for **bd** = LO' (lower bound) yields \geq .
- $l = {}^{2}E'$ for **bd** = {}^{6}FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .

Related variables: VAR_FLO, VAR_IRE, VAR_SIN/SOUT, VAR_COMPRD

Related equations: EQ(l)_COMBAL, EQE_COMPRD

Purpose: Relationship to facilitate constraints on the market share of process (p) in the total production of commodity (c). Indicates that the flow of commodity (c) from/to process (p) is bounded by the given fraction of the total production of commodity (c). The time-slice level of the constraint is that of the commodity (c) when using FLO_MARK, and ANNUAL when using PRC_MARK. The same given fraction is applied to all timeslices.

Variables involved:

- VAR_FLO(r,v,t,p,com,s) the average flow to/from a process built in period v, during time-slice s, during each year of period t. The variable for an input flow appears on the consumption side of the balance equation without any coefficients, and the variable for an output flow on the production side multiplied with the commodity efficiency (COM IE).
- VAR_IRE(r,v,t,p,com,s,ie) the average flow to/from an exchange process built in period v, during time-slice s, during each year of period t. The export variable appears on the consumption side of the balance equation without any coefficients, and the import variable on the production side multiplied by the commodity efficiency (COM_IE).
- VAR_SIN/SOUT(r,v,t,p,c,s) flows entering/leaving a storage process p storing a commodity c. The variable for charging appears on the consumption side of the balance equation without any coefficients; the import variable on the production side multiplied by both the storage efficiency and commodity efficiency (SGT EFF, COM IE).
- VAR_COMPRD(r,t,com,s) variable equal to the total import + production (activity and capacity based) in each period and time slice. This balance is defined by the equation EQE_COMPRD, which is automatically generated for all commodities used in the FLO MARK or PRC MARK parameters.

Parameters:

- FLO_MARK(r,t,p,c,l) Market share of single process in total production of commodity c.
- **PRC_MARK(r,t,p,grp,c,l)** Market share of a group **grp** of processes in total production of commodity **c**.

Remarks:

- All the FLO_MARK parameters are internally converted to PRC_MARK parameters by the model generator, using the process name of the FLO_MARK parameter as the process group index (grp) in PRC_MARK. Therefore, below references to the parameters are mostly given in terms of PRC_MARK only.
- 2. Market-share constraints can be specified for standard processes, as well as for exchange and storage processes. For standard processes, the PRC_MARK parameter value can be unambiguously applied to the process flow, and the value should normally be non-negative. However, because exchange and storage processes may have both input and output flows of the same commodity, for these processes the sign of the parameter value determines whether it is applied to the input or output flow, by using the following simple conventional rules:
 - Value ≥ 0: Constraint is applied to the output flow (imports or storage discharge)
 - Value ≤ 0: Constraint is applied to the negative of input flow (exports or storage charge)

• Value=EPS: Constraint is applied to the net output flow (output-input flow) These simple rules provide reasonable flexibility for specifying market share bounds also for exchange and storage processes, in addition to ordinary processes. Although these rules preclude individually bounding the input or output flow to zero, this could always be accomplished by using the IRE BND, STG OUTBND, and STG INBND parameters when necessary.

- 3. The default timeslice level of the constraint is the commodity timeslice level for the constraints defined by using FLO_MARK by the user, and ANNUAL level for those defined by using the PRC_MARK parameter. For overriding the default, see remark 4 below.
- 4. The commodity used in the parameter does not actually need to be in the topology, but it should contain some commodity that does exist in the process topology. This feature can be utilized for defining market-share equations at any desired timeslice level. For example, if ELC is a DAYNITE level commodity, the user could define a dummy commodity ELC_ANN that includes ELC as a group member (through COM_GMAP membership), and use the ELC_ANN commodity in the PRC_MARK parameter instead of ELC. The constraint would then be defined at the timeslice level of the ELC_ANN commodity, which is ANNUAL if not explicitly defined.
- 5. In the equation formulation below, the set mrk ts_{r,grp,c,s} denotes the timeslices assigned to the constraints associated with group grp and commodity c in region r, as explained in remarks 3 and 4 above.
- 6. Zero market shares are either removed (for bound type 'LO') or converted into flow bounds (bound types 'UP' and 'FX'), because the formulation employs inverse values.

Examples:

- Define an upper market share bound of 5% for technology WIND1 in total ELC production in the 2010 period.
- Define an upper market share of 25% for diesel export (through exchange process DSLXHG) of total DSL production in the 2010 period. Note that because the bound is for exports, in this case the parameter value should be negative and the bound type LO instead of UP.

PARAMETER FLO_MARK / REG.2010.WIND1.ELC.UP 0.05 REG.2010.DSLXHG.DSL.LO -0.25 /:

Interpretation of the results:

- Primal: If the primal value is zero, the constraint is binding. If the primal value is positive for a lower PRC_MARK bound or negative for an upper bound, the constraint is non-binding.
- Dual: The dual value describes for example for a lower bound, the subsidy needed to guarantee the market share of the technology being forced into the market. The subsidy is needed, since the production of the technology is too expensive compared to other competing technologies. The value of the subsidy, which the technology receives, is equal to (1–PRC_MARK)*(dual variable). This subsidy has to be paid by the other technologies producing the same commodity. Thus, the costs of these technologies are increased by the amount PRC_MARK*(dual variable). The constraint can therefore be interpreted as a quota system for the production of a specific technology, e.g. a certificate system for electricity by a wind technology: each non-wind producer has to buy certificates according to the quota. The price of the certificates equals the dual value of the constraint.

Equation:

 $EQ(l)_{FLMRK_{r,l,g:rp,c,s}} \forall (r, t, grp, c, s) \in \left(\{ \mathsf{rtpc}_{\mathsf{r},\mathsf{t},\mathsf{p},\mathsf{e}} \mid PRC_{MARK_{r,l,p,g:rp,c,s}} \neq 0 \} \cap \mathsf{mrk_ts}_{\mathsf{r},\mathsf{grp},\mathsf{c},\mathsf{s}} \right)$

$$\sum_{\substack{(\text{com}, y, ts) \in \\ \cap RPC_p \cap COM_{-}GMAP_c \\ \cap RPC_p \cap COM_{-}GMAP_c \\ \cap RPC_p \cap COM_{-}GMAP_c \\ (VAR_{-}IRE_{r,y,t,p,com,ts}, imp \\ (VAR_{-}SOUT_{r,y,t,p,com,ts}, STG_{-}EFF_{r,y,p}) \times \begin{bmatrix} COM_{-}IE_{r,com,ts} & if PRC_{-}MARK_{r,t,p,grp,c,s,l} \ge 0 \\ 0 & if PRC_{-}MARK_{r,t,p,grp,c,s,l} < 0 \end{bmatrix} \\ \begin{pmatrix} VAR_{-}IRE_{r,y,t,p,com,ts} \times STG_{-}EFF_{r,y,p} \\ VAR_{-}SIN_{r,y,t,p,com,ts} \end{pmatrix} \times \begin{bmatrix} 1 & if PRC_{-}MARK_{r,t,p,grp,c,s,l} \le 0 \\ 0 & if PRC_{-}MARK_{r,t,p,grp,c,s,l} < 0 \end{bmatrix} \\ \begin{pmatrix} VAR_{-}IRE_{r,y,t,p,com,ts} \end{pmatrix} \times \begin{bmatrix} 1 & if PRC_{-}MARK_{r,t,p,grp,c,s,l} \le 0 \\ 0 & if PRC_{-}MARK_{r,t,p,grp,c,s,l} < 0 \end{bmatrix} \end{bmatrix} \times \begin{pmatrix} RS_{-}FR_{r,s,ts} \\ PRC_{-}MARK_{r,t,p,grp,c,s,l} \end{pmatrix} \\ \{=; \leq \} \\ \sum_{\substack{\text{com} \in \\ PRC \cap COM_{-}GMAP_{c}} & \underbrace{\text{RHS}_{-}COMPRD_{r,t,com,ts}} \end{pmatrix} \begin{pmatrix} VAR_{-}COMPRD_{r,t,com,ts} \times \begin{pmatrix} RS_{-}FR_{r,s,ts} \\ RS_{-}FR_{r,s,ts} \end{pmatrix} \end{pmatrix}$$

6.3.30 Equations related to exchanges (EQ_IRE, EQ_IREBND, EQ_XBND)

The three equations in this section concern trade between regions. Since these equations involve (directly or indirectly) more than one region, we start their presentation by a complete description of the modeling approach used, which, as we shall see, involves various schemes for representing different types of trade. The description already given in Chapter 4 is also relevant to these equations.

<u>Variables</u>

• VAR_IRE(r, v, t, p, c, s, ie)

Description: The total amount of traded commodity (c) imported/exported (ie) to/from region (r), through process (p) vintage (v) in each time period (t)

Purpose: The trade variables facilitate trade of commodities between exporting and importing regions

Bounds: The amount of commodity imported to a region from each exporting region can be directly constrained by the IRE_BND parameter.

Remarks:

- Note that there is a one-to-one correspondence between the VAR_IRE variables and the top_ire entries (one variable for the supply region/commodity and one variable for the demand region/commodity for each instance of top_ire).
- In market-based trade, the VAR_IRE variables for the market region describe the net imports to, and exports from, the market region, not the total market volume.
- There is no variable for the total volume of the commodity market in market-based trade. The total volume can only be addressed by means of UC_IRE parameters (summing over all imports to or exports from the market).
- In market-based trade, only the amount of commodity imported to a region from the market, or exported from the region to the market, can be constrained by the IRE_BND parameter. The imports and exports thus cannot be attributed to a specific supply or demand region on the other side of the trade.
- The amount of commodity exported from / imported to a region may also be limited by various user constraints. However, unless the trade is modeled with bilateral processes, such bounds can only apply to the total exports from or imports to a region, and cannot apply to e.g. imports from a specific region.

There are only three trade equations, namely a generic trade balance equation EQ_IRE, and two bounds, EQ(1)_IREBND and EQ(1)_XBND. The generic balance equation, EQ_IRE, can be further divided into two flavors:

- A. Balance equations for bilateral and other unidirectional trade into a single destination region (Cases 1 and 2).
- B. Balance equations for multidirectional trade from single export region and multilateral market-based trade (Cases 3 and 4).

6.3.30.1 Equation EQ_IRE

Indices: region (r), year (t), process (p), commodity (c), timeslice (s)

Type: =

Related variables: VAR_IRE

Related equations: EQ(I)_IREBND, EQ(I)_XBND, EQ(I)_COMBAL, EQ_ACTFLO

Purpose: This equation defines the balance between the imports of each traded commodity (c) into region (r) and the corresponding exports through each exchange process (p) in each time period (t) and timeslice (s) of the process.

Units: Units of commodity traded. Normally PJ for energy, Mton or kton for materials or emissions.

Remarks:

- Flows into individual regions may be limited by the IRE_BND and IRE_XBND parameters.
- The equation has two flavors: The first one is for bilateral and unidirectional trade with a single destination region, and the second is for market-based trade and multidirectional trade from a single source region.

6.3.30.1.1 Case A. Bi-lateral or multilateral unidirectional trade to a single import region



Remarks:

- The IRE_TSCVT conversion coefficients are in practice provided only for some pairs of mapped timeslices between **r2** and **r**. Therefore, the timeslice conversion is actually done in two stages: First, the timeslices of the VAR_IRE variables are converted to the mapped timeslices, and then the mapped timeslices in **r2** to those in **r** as follows:
 - The mapping coefficients IRE_TSCVT do not have to be provided by the user if the timeslice definitions in both regions are identical.
 - If the timeslice definitions are different, the user provides the mapping coefficients IRE_TSCVT to convert the timeslice s2 in region r2 to the timeslice s in region r. Since the timeslice level of s2 may be different from the timeslice level ts of the exchange variable in region r2, the parameter RTCS_TSFR is used to match ts and s2.
- Note that the equation is generated for each period in rtp only, not for each vintage in rtp_vintyr as in the original code. This is because prc_vint is region-specific. If prc_vint is set to YES in one region and to NO in another, that would create serious sync problems, if the equation were generated for each vintage in rtp_vintyr. In addition, differences in e.g. NCAP_PASTI, NCAP_TLIFE, and NCAP_AF could create sync problems, even if prc_vint would be set to YES in all regions.

6.3.30.1.2 Case B. Multidirectional and market-based trade between regions.

Equation:

 $EQ_IRE_{r_{t,p,c,s}} \rightarrow \{r, t, p, c, s \in (\mathbf{rtp}_{r,t,p} \land \mathbf{rpcs_var}_{r,p,c,s} \land \mathbf{rpc_eqire}_{r,p,c})\}$



Remarks:

- The IRE_TSCVT conversion coefficients are in practice provided only for some pairs of mapped timeslices between *r2* and *r*. Therefore, the timeslice conversion is actually done in two stages: First, the timeslices of the VAR_IRE variables are converted to the mapped timeslices, and then the mapped timeslices in *r2* to those in *r*.
- In the case of market-based trading, **prc_aoff** can be used to switch off the entire commodity market for periods that fall within a range of years. It is also possible to specify multiple entries of **prc_aoff**, if, for example trading should be possible only between selected years.
- The top_ire entry between the export and import commodity in the market region itself is automatically defined by the TIMES model generator when necessary, i.e. there is no need to provide it by the user.

6.3.30.2 Equation: EQ(1) IREBND

Indices: region (r), year (t), commodity (c), timeslice (s), region2 (all_r), import/export (ie)

Type: Any type, as determined by the bound index bd of IRE BND:

- 1 = G' for **bd** = LO' (lower bound) yields \geq .
- 1 = E' for **bd** = FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .

Related variables: VAR_IRE

Related equations: EQ_IRE, EQ(I)_XBND, EQ(I)_COMBAL

Description: Sets a bound for the amount of commodity (c) imported/exported (ie) to/from region (r), from/to another region (all_r) in time period (t) and timeslice (s).

Purpose: The equation is optional and can be used to set a bound for a pair-wise interregional exchange. The generation of the equation is triggered by the user-specified parameter IRE_BND.

Units: Units of commodity traded. Normally PJ for energy, Mton or kton for materials or emissions.

Type: Set according to the 'l' index in IRE_BND.

Remarks:

• Total trade flows into/from individual regions may be limited by using the IRE XBND parameter.

Interpretation of the results:

- Primal: If the primal value equals the bound parameter, the constraint is binding.
- Dual: The dual value describes for a lower/upper bound the cost increase/decrease in the objective function, if the bound is increased by one unit. It may also be interpreted as subsidy/tax needed to reach the given bound value.

Equation:

Case A. Imports from an exten $EQ(I)_IREBND_{r,t,c,s,all_r,je} \forall \begin{cases} \\ \\ \\ \end{cases}$	nal region or market region $r, t, c, s, all _ r, ie : (RCS _COMTS_{r,c,s} \land$ $(\exists p : RPC _IE_{r,p,c,ie}) \land IRE _BND_{r,j,c,s,all_r,je})$:
$\sum_{p:(\exists c2:TOP_IRE_{all_r,c1,r,p})} \sum_{v \in RTP_V}$	$\sum_{\text{NTYR}_{r,y,t,p}} \sum_{s2} VAR _ IRE_{r,y,t,p,c,s2,exp} \times$
	$\begin{pmatrix} 1 & \text{if } s2 \in TS_MAP(r,s,s2) \\ \hline FR(s) & \text{if } s2 \in RS_BELOW(r,s2,s) \end{pmatrix}$

 $\{\leq;=;\geq\}$ IRE _BND_{r,t,c,s,all_r,ie}

Case B. Imports from an internal non-market region

$$\begin{split} EQ(l)_IREBND_{r,t,c,s,all_r,se} & \forall \begin{cases} r,t,c,s,all_r,ie:(RCS_COMTS_{r,c,s} \land \\ (\exists p \in RPC_IE_{r,p,c,se}) \land IRE_BND_{r,t,c,s,all_r,se} \end{cases} ; \\ \sum_{\substack{(c2,p) \in TOP_IRE_{all_r,s2,r,s}}} \sum_{\nu \in RPP_VDVTR_{all_r,s2,r,s}} \sum_{e_2} VAR_IRE_{all_r,\nu,t,p,c2,s2,eep} \times IRE_FLO_{all_r,\nu,p,c2,r,c,e1} \times \\ IRE_CCVT_{all_r,c2,r,c} \times IRE_TSCVT_{all_r,s2,r,r,s1} \times \begin{pmatrix} 1 & \text{if } s1 \in TS_MAP(r,s,s1) \\ FR(s) & \text{if } s1 \in RS_BELOW(r,s1,s) \end{pmatrix}; \end{cases}$$

 $\{\leq;=;\geq\}$ IRE _BND_{r,t,c,s,all_r,ie}

Case C. Exports from a non-market region to an internal or external region

 $EQ(I)_IREBND_{r,t,c,s,dl_r,ie} \quad \forall \begin{cases} r,t,c,s,all_r,ie:(RCS_COMTS_{r,c,s} \land \\ (\exists p:RPC_IE_{r,p,c,ie}) \land IRE_BND_{r,t,c,s,dl_r,ie}) \end{cases} :$ $\sum_{p:(\exists c2:TOP_IRE_{r,c,dl_r,c,p}) \lor \in RTP_VBATYR_{r,s,t,p}} \sum_{e2} VAR_IRE_{r,v,j,p,c,s2,ep} \times \\ \left(\frac{1}{FR(s)} if s2 \in TS_MAP(r,s,s2) \\ \frac{FR(s2)}{FR(s2)} if s2 \in RS_BELOW(r,s2,s) \right) \end{cases}$

 $\{\leq;=;\geq\}$ IRE _BND_{r,t,c,s,all_r,ie}

Case D. Exports from a market region to an internal region

$$EQ(I) _IREBND_{r,t,c,s,all_r,ie} \quad \forall \begin{cases} r,t,c,s,all_r,ie : (RCS_COMTS_{r,c,s} \land \\ (\exists p:RPC_IE_{r,p,c,ie}) \land IRE_BND_{r,t,c,s,all_r,ie}) \end{cases} ; \\ (\exists p:RPC_IE_{r,p,c,ie}) \land IRE_BND_{r,t,c,s,all_r,ie}) \end{cases} ; \\ RE_CCVT_IE_VENTTR_{all_r,s,p}} \sum_{e2} VAR_IRE_all_r,v,t,p,c2,s2,eep} \times \\ IRE_CCVT_all_r,c2,r,c} \land IRE_TSCVT_all_r,s2,r,s} \land \left(\frac{1}{FR(s)} if \ s2 \in RS_BELOW(r,s2,s)\right) ; \end{cases} ;$$

 $\left\{\!\leq\!;=;\geq\right\}\mathit{IRE}_\mathit{BND}_{r,t,c,s,\mathit{all}_r,\mathit{ie}}$

Remarks:

• The IRE_TSCVT conversion coefficients are in practice provided only for some pairs of mapped timeslices between **all_r** and **r**. Therefore, the timeslice conversion is actually done in two stages: First, the timeslices of the VAR_IRE variables are converted to the mapped timeslices, and then the mapped timeslices in **all_r** to those in **r**.
6.3.30.3 Equation: EQ(l) XBND

Indices: region (r), year (t), commodity (c), timeslice (s), imp/exp (ie)

Type: Any type, as determined by the bound index **bd** of IRE_XBND:

- l = G' for **bd** = LO' (lower bound) yields \geq .
- l = 'E' for **bd** = 'FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .

Related variables: VAR_IRE

Related equations: EQ(*l*)_IRE, EQ(*l*)_IREBND, EQ(*l*)_COMBAL

- **Description:** Bound on the total amount of traded commodity (c) imported/exported (ie) to/from region (all r) in a period (t) and timeslice (s).
- **Purpose:** This equation bounds inter-regional or exogenous exchanges in a particular region, across all other regions.
- Units: Units of commodity traded. Normally PJ for energy, Mton or kton for materials or emissions.
- **Remarks:** Flows into/from individual regions may be limited by the IRE_BND parameter.

Interpretation of the results:

Primal: If the primal value equals the bound parameter, the constraint is binding.

Dual: The dual value describes for a lower/upper bound the cost increase/decrease in the objective function, if the bound is increased by one unit. It may also be interpreted as subsidy/tax needed to reach the given bound value.

Equation:



 $EQ(l)_XBND_{all_r,t,c,s,ie}$ \Rightarrow $IRE_XBND_{all_r,t,c,s,iebd}$

6.3.31 Equations: EQ(*l*)_INSHR, EQ(*l*)_OUTSHR

Indices: region (r), year (t), process (p), commodity (c), commodity group (cg), timeslice (s)

Type: Any type, as determined by the bound index bd of FLO SHAR:

- l = 'G' for **bd** = 'LO' (lower bound) yields \geq .
- l = 'E' for **bd** = 'FX' (fixed bound) yields =.
- l = 'L' for **bd** = 'UP' (upper bound) yields \leq .

Related variables: VAR_FLO, VAR_ACT

Related equations: EQ(l)_COMBAL, EQ_PTRANS, EQ_ACTFLO

Purpose: A market/product allocation constraint equation is generated for each process (\mathbf{p}) for each time period (\mathbf{t}) and each time-slice (\mathbf{s}) in each region (if desired). It ensures that the share of an inflow/outflow of a commodity (\mathbf{c}) is lower/higher/equal a certain percentage of the total consumption/production of this process for a specified commodity group (\mathbf{cg}) .

Quality Control Checks:

$$\sum_{o \in og} FLO_SHAR_{r,v,p,c,og,s,'LO'} \forall (FLO_SHAR_{r,v,p,c,og,s,'LO'} \ni l =" \geq") \leq 1 - \sum_{o \in og} FLO_SHAR_{r,v,p,c,og,s,'EX'}$$
$$\sum_{o \in og} FLO_SHAR_{r,v,p,c,og,s,'UP'} \forall (FLO_SHAR_{r,v,p,c,og,s,'UP'} \ni l =" \leq") \geq 1 - \sum_{o \in og} FLO_SHAR_{r,t,p,c,og,s,'EX'}$$

 $\forall FLO_SHAR > 0$

Remarks:

- Exchanging top(r,p,c,'IN')=Input vs. top(r,p,c'OUT') = Output in this equation yields EQ(1)_OUTSHR since c is only member of one cg.
- The period index of the parameter FLO_SHAR is related to the vintage period (v) of the process, i.e., if the process is vintaged (**prc_vint**), a constraint will be generated for each period (t) the installation made in the vintage period (v) still exists (these period pairs are internally provided by the set **rtp_vintyr**).

Interpretation of the results:

- Primal: If the primal value is zero, the constraint is binding. If the primal value is positive for a lower FLO_SHAR bound or negative for an upper bound, the constraint is non-binding.
- Dual: The dual value describes, for a lower bound, the subsidy needed to guarantee that the flow is at the given lower bound. The subsidy is needed, since for an output flow the shadow price of the produced commodity is too low to cover the

production costs of the flow variable (for an input flow the opposite is true, the commodity is too expensive to be used in the process). The value of the subsidy that the flow receives is equal to (1-FLO_SHAR)*(dual variable). This subsidy has to be paid by the other flows forming the denominator in FLO_SHAR constraint, thus, the costs for these flows are increased by the amount FLO_SHAR*(dual variable). In a similar way, an upper bound FLO_SHAR can be interpreted as a tax being added to the costs of a flow.

Equation:



6.3.32 Equation: EQ_PEAK

Indices: region (r), period (t), commodity group (cg), time-slice (s)

Type: ≥

Related variables: VAR_ACT, VAR_NCAP, VAR_FLO

Related equations: EQ(*l*)_COMBAL, EQ(*l*)_CAPACT

Purpose: The commodity peaking constraint ensures that the capacity installed is enough to meet the highest demand in any timeslice, taking into consideration both adjustments to the average demands tracked by the model and a reserve margin requiring excess capacity to be installed.

Remarks:

• In the description below, the production and consumption components resemble those of the EQ(1)_COMBAL commodity balance equation, but with a peak contribution/co-incident factor applied to the terms. These factors are process dependent and as such are actually applied within the referenced expression during the summing operation.

Sets and parameters involved:

- com_peak(r,cg) is a flag that a peaking constraint is desired. It is optional if com_pkts(r,cg,s) is provided
- **com_pkts(r,cg,s)** are the explicit time slices for which peaking constraints are to be constructed. A post-optimization QC check will be done to ensure that the timeslice with highest demand is in said list. Default is all **com_ts(r,c,s)**.
- COM_PKRSV(r,t,c) is the peak reserve margin. Default 0.
- COM_PKFLX(r,t,c,s) is the difference (fluctuation) between the average calculated demand and the actual shape of the peak. Default 0
- FLO_PKCOI(r,t,p,c,s) is a factor that permits increasing the average demand calculated by the model to handle the situation where peak usage is typically higher due to coincidental usage at peak moment (e.g., air condition). Default 1 for each process consuming c. User can prevent a process from contributing to the calculation of the peak by specifying = 0
- NCAP_PKCNT(r,t,p,s) is the amount of capacity (activity) to contribute to the peak. Default 1 for each process producing commodity c. User can prevent a process from contributing to the peak by specifying = EPS
- prc_pkaf(r,p) switch to set NCAP_PKCNT=NCAP_AF/1 as default. Default: no
- prc_pkno(r,p) switch to disable process p from contributing to the peak by its capacity, and to disable also assigning the default value of NCAP_PKCNT for the process.
- rpc_pkc(r,p,c) is an internal set defined to contain those processes (p) and peaking commodities (c) that will be assumed to contribute to the peak by their capacity. Derived by the preprocessor from all those process producing commodity c, which either have c as their primary group PG or have prc_pkaf defined, but are in neither case included in the set prc_pkno.

Interpretation of the results:

Primal: When the equation is binding, the primal value of the equation is equal to the RHS constant of the equation, i.e. corresponds to the maximum output from the existing capacity in the peak timeslice, adjusted with the peak reserve requirement. When the equation is non-binding, the primal level also includes the amount of output capacity exceeding the capacity requirements during the timeslice.

Dual: The dual value of the peaking equation describes the premium consumers have to pay in addition to the commodity price (dual variable of EQ(1)_COMBAL) during the peak timeslice. The premium equals

(1+COM_PKFLX)*FLO_PKCOI*RTCS_TSFR*(dual variable).

Equation:

 $EQ_PEAK_{r,t,cg,s} \ni \mathbf{com_peak}_{r,cg} \land s \in \mathbf{com_pkts}_{r,cg,s}$

$$\begin{split} &\sum_{c \in cg} 1/(1 + COM_PKRSV_{r,t,c}) \times COM_IE_{r,t,c} \times \sum_{p \in (iop_{r,p,c}) \cup Tpc_ire_{r,p,c}(MP^{c})} \\ & \left[if \left(rpc_pkc_{r,p,c} \wedge prc_cap_{r,p} \right) \\ & \left(G_YRFR_{r,s} \times \sum_{v \in rp_opyr_{r,v,t,p}} \left[NCAP_PKCNT_{r,v,p,s} \times COEF_CPT_{r,v,t,p} \\ & \times \left[VAR_NCAP_{r,tt(v),p} + NCAP_PASTI_{r,v,p} - \\ VAR_SCAP_{r,v,t,p} \times (\ni prc_rcap_{r,p}) \\ PRC_CAPACT_{r,p} \times PRC_ACTFLO_{r,v,p,c} \\ & \left[else \\ & \sum_{v \in rp_vimyr_{r,v,t,p}} FLOFLO_{r,v,t,p,c,s,'OUT'} \times NCAP_PKCNT_{r,v,p,s} \\ & + \\ & \sum_{p \in sq_{u,p,c}} \sum_{v \in rpe_vimyr_{r,v,t,p}} VAR_SOUT_{r,v,t,p,c,ts} \times RS_FR_{r,s,ts} \times NCAP_PKCNT_{r,v,p,s} \\ & + \\ & \sum_{v \in rp_vimyr_{r,v,t,p}} CAL_IRE_{r,v,t,p,c,s,'IMP'} \times NCAP_PKCNT_{r,v,p,s} \\ & - \\ & \sum_{v \in rp_vimyr_{r,v,t,p}} CAL_IRE_{r,v,t,p,c,s,'IMP'} \times NCAP_PKCNT_{r,v,p,s} \\ & if \ prc_pkno_{r,p} \\ & \end{bmatrix} \\ \end{split}$$

$$\sum_{\text{orig}} (1 + COM_PKFLX_{t,l,\sigma,s}) \times$$

$$\left[\left(+ COM_PKFLX_{t,l,\sigma,s}) \times \left[\sum_{\substack{\text{verp}_\text{vurp}_{t,v,q}}} CAL_FLOFLO_{t,v,l,p,c,s}, (OUT)} \times FLO_PKCOI_{t,l,p,c,s} + \sum_{\substack{\text{perp}_\text{vurp}_{t,v,q}}} CAL_IRE_{t,v,l,p,c,s}, (EP)} \times FLO_PKCOI_{t,l,p,c,s} + \sum_{\substack{\text{perp}_\text{vurp}_{t,v,q}}} CAL_IRE_{t,v,l,p,c,s}, (EP) \times FLO_PKCOI_{t,l,p,c,s} + \sum_{\substack{\text{perp}_\text{vurp}_{t,v,q}}} CAL_IRE_{t,v,l,p,c,s}, (EP) \times FLO_PKCOI_{t,l,p,c,s} + \sum_{\substack{\text{perp}_\text{vurp}_{t,v,q}}} CAL_IRE_{t,v,l,p,c,s} \times FLO_PKCOI_{t,v,l,p,c,s} \times FLO_PKCOI_{t,v,l,p,c,s} + \sum_{\substack{\text{perp}_\text{vurp}_{t,v,q}} CAL_IRE_{t,v,l,p,c,s} \times FLO_PKCOI_{t,v,l,p,c,s} \times FLO_PKCOI_$$

6.3.33 Equation: EQ_PTRANS

Indices: region (r), year (y), process (p), commodity group1 (cg1), commodity group2 (cg2), time-slice (s)

Type: =

Related variables: VAR_FLO, VAR_ACT

Related equations: EQ(*l*)_COMBAL, EQ(*l*)_INSHR, EQ(*l*)_OUTSHR, EQ_ACTFLO

Purpose: Allows specifying an equality relationship between certain inputs and certain outputs of a process e.g. efficiencies at the flow level, or the modeling of emissions that are tied to the inputs. It is generated for each process for each time period and each time-slice in each region.

Remarks:

- Internal set **rps_s1(r,p,s**): The finer of (set of time slices of the most finely divided member of the commodities within the shadow primary group (commodities being not part of primary commodity group and on the process side opposite to the primary commodity group) and the process timeslice level (**prc_tsl**)).
- The flow variables of the commodities within the primary commodity group are modelled on the process level (**prc_tsl**). All other flow variables on the timeslice level of **rps_s1**.
- The internal parameter COEF_PTRAN(r,v,t,p,cg1,c,cg2) is the coefficient of the flow variables of commodity **c** belonging to the commodity group **cg2**. While FLO_FUNC(r,v,p,cg1,cg2,s) establishes a relationship between the two commodity groups **cg1** and **cg2**, FLO_SUM(r,v,p,cg1,c,cg2,s) can be in addition specified as multiplier for the flow variables of **c** in **cg2**.

COEF_PTRAN is derived from the user specified FLO_FUNC and FLO_SUM parameters based on the following rules:

- If FLO_FUNC is given between cg1 and cg2 but no FLO_SUM for the commodities c in cg2, it is assumed that the FLO_SUMs are 1.
- If FLO_SUM is specified but no FLO_FUNC, the missing FLO_FUNC is set to 1.
- If FLO_SUM(r,v,p,cg1,c,cg2) and FLO_FUNC(r,v,p,cg2,cg1,s) are specified, the reciprocal of FLO_FUNC is taken to calculate COEF_PTRAN.
- FLO_SUMs can only be specified for the flows within one commodity group cg1 or cg2 of EQ_PTRANS between these two commodity groups, but not for both commodity groups at the same time.
- By specifying a SHAPE curve through the parameter FLO_FUNCX(r,v,p,cg1,cg2) the efficiencies FLO_FUNC and FLO_SUM can be described as function of the age of the installation. The internal parameter RTP_FFCX contains the average SHAPE multiplier for the relevant years in a period (those years in which the installed capacity exists).

Interpretation of the results:

- Primal: The primal value of the transformation is usually zero.
- Dual: Due to the flexibility of the transformation equation the interpretation of its dual value depends on the specific case. For a simple case, a process with one input flow **c1** and one output flow **c2** being linked by an efficiency FLO_FUNC(c1,c2), the dual variable, which is being defined as the cost change when the RHS is increased by one unit, can be interpreted as cost change when the efficiency of the process is increased by 1/VAR_FLO(r,v,t,p,c1,s):

$$VAR_FLO_{r,y,t,p,c2,s} - FLO_FUNC_{r,y,t,p,c1,c2,s} \times VAR_FLO_{r,y,t,p,c1,s} = 1$$

$$VAR_FLO_{r,y,t,p,c2,s} - FLO_FUNC_{r,y,t,p,c1,c2,s} \times VAR_FLO_{r,y,t,p,c1,s} - 1 = 0$$

$$VAR_FLO_{r,y,t,p,c2,s} - \left(FLO_FUNC_{r,y,t,p,c1,c2,s} + \frac{1}{VAR_FLO_{r,y,t,p,c1,s}}\right) \times VAR_FLO_{r,y,t,p,c1,s} = 0$$

Equation:

 $EQ_PTRANS_{r,\nu,t,p,\text{sg1,sg2,sl}} \ni (r,\nu,t,p) \in \left(\textbf{rp_flo}_{r,p} \cap \textbf{rtp_vintyr}_{r,\nu,t,p} \right) \land sl \in \textbf{rps_s1}_{r,p,s1}$

 $\wedge \left(s2 \in \mathbf{ts_map}_{r,s2,s1} \land \left(\begin{matrix} FLO_SUM_{r,t,p,og1,o,g2,s2} & \lor \\ FLO_FUNC_{r,t,p,og1,o,g2,s2} & \land NOT \left(FLO_SUM_{r,t,p,og1,o,g2,s2} & \lor FLO_SUM_{r,t,p,og2,o,g2,s2} \\ \end{matrix} \right) \right) \right)$

 $\sum_{\sigma \in og2} \sum_{\sigma \in (\mathbf{s}_{_} m \, \mathbf{ap}_{\mathbf{r}, \mathbf{s}, \mathbf{s}})} \nabla T A R_{_} FLO_{r, \nu, t, p, \sigma, \sigma} \times RTCS_TSFR_{r, t, \sigma, \sigma 1, \sigma} =$

 $COEF_PTRAN_{r,v,t,p,cg1,c,cg2,ts} \quad ts \in \mathbf{rpcs_varc}_{r,p,c,ts}$

=

-

$$\sum_{s \in prc_ts_{r,p,s}} \left(1 \times (if \ ts_map_{r,ts,s}) + \frac{G_YRFR_{r,s}}{G_YRFR_{r,s}} \times (if \ rs_below_{r,s,ts}) \right)$$

$$\frac{FLO_FUNC_{r,v,t,p,cg1,cg2,s}}{FLO_FUNC_{r,v,t,p,cg2,cg1,s} \times (if \ FLO_SUM_{r,v,t,p,cg1,c,cg2,s})} \times \left(if \ FLO_FUNC_{r,v,t,p,cg2,cg1,s} \right) \times \left(1 \times (if \ NOT \ FLO_SUM_{r,v,t,p,cg1,c,cg2,s}) + FLO_SUM_{r,v,t,p,cg1,c,cg2,s} \right)$$

Calculation of SHAPE parameter RTP_FFCX

<u>Case A: Lifetime minus construction time is longer than the construction period</u> $PRC_YMIN_{r,v,p} = B_v + NCAP_ILED_{r,v,p}$ $PRC_YMAX_{r,v,p} = PRC_YMIN_{r,v,p} + NCAP_TLIFE_{r,v,p} - 1$

 $RTP_FFCX_{r,v,t,p,cg1,c,cg2} \quad \ni FLO_FUNCX_{r,v,p,cg1,cg2}$

$$\frac{\sum_{y \in (\text{periody}_{r,y,p})} SHAPE(FLO_FUNCX_{r,y,p,\text{C2}1,\text{C2}2}, 1 + MIN(y, PRC_YMAX_{r,y,p}) - PRC_YMIN_{r,y,p})}{MAX[1, MIN(E(t), PRC_YMAX_{r,y,p}) - MAX(B(t), PRC_YMIN_{r,y,p}) + 1]} - 1$$

<u>Case B: Lifetime minus construction time is shorter than the construction period =></u> Investment is repeated in construction period

$$PRC_YMAX_{r,v,p} = NCAP_TLIFE_{r,v,p} - 1$$

$$RTP_FFCX_{r,v,t,p,cg1,c,cg2} \rightarrow FLO_FUNCX_{r,v,p,cg1,cg2}$$

$$= \sum_{v \in rep_vintyr_{t,v,tp}} \frac{SHAPE(FLO_FUNCX_{r,v,p,cg1,cg2}, PRC_YMAX_{r,v,p})}{PRC_YMAX_{r,v,p}} - 1$$

6.3.34 Equations: EQL_SCAP

Indices: region (r), vintage (v), process (p), indicator (ips)

Type: \leq

Related variables: VAR_ACT, VAR_NCAP, VAR_SCAP

Related equations: EQ_CUMRET

Purpose: Establishes an upper bound for the cumulative retirements and salvaged capacity by process vintage, as well as for the cumulative process activity. The equation is only generated when early retirements are allowed for the process, or if the process is vintaged and a maximum operating life is specified with NCAP_OLIFE.

Notation:

• $RVPRL_{r,v,p}$ is defined as the time (in years) between the vintage year (v) and the last period **t** of availability for that vintage: $RVPRL_{r,v,p} = M(t)-M(v)$.

Equation:

$$Equation:$$

$$EQL_SCAP_{r,v,p,ips} \rightarrow \left(rtp_{r,v,p} \land \left((prc_rcap_{r,p} \land (\neg prc_vint_{r,p} \lor obj_sums_{r,v,p})) \land (prc_vint_{r,p} \land NCAP_OLIFE_{r,v,p}) \right) \right)$$

$$\left(\sum_{t=v+RVPRL_{r,v,p}} VAR_SCAP_{r,v,t,p} \qquad if \ ips ='N' \\ \sum_{rtp_eptyr_{r,v,p}} \sum_{prc_ts_{r,p,s}} \frac{VAR_ACT_{r,v,t,p,s} \times D_t}{PRC_CAPACT_{r,p} \times NCAP_OLIFE_{r,v,p}} \quad otherwise \right)$$

$$\leq$$

$$VAR _ NCAP_{r,u(v),p} + NCAP _ PASTI_{r,v,p} - \sum_{\substack{\text{obj_sums}_{r,v,p} \\ pre_rcap_{r,p}}} VAR _ SCAP_{r,v,'0',p}$$

6.3.35 Equations: EQ_STGAUX

Indices: region (r), vintage (v), period (t), process (p), commodity (c), timeslice (s)

Type: =

Related variables: VAR_ACT, VAR_FLO, VAR_SIN, VAR_SOUT

Related equations: EQ_STGTSS, EQ_STGIPS

Purpose: Establishes the relations between the main storage flows / activity and auxiliary storage flows.

Equation:

 $EQ_STGAUX_{r,y,t,p,c,s} \rightarrow (\mathbf{rtp_vintyr}_{r,y,t,p} \land \mathbf{rpcs_var}_{r,p,c,s} \land \mathbf{prc_map}_{r,\forall STG',p} \land \neg \mathbf{rpc_stg}_{r,p,c})$ $VAR_FLO_{r,y,t,p,c,s} =$

$$PRC_ACTFLO_{r,y,p,c} \times \begin{pmatrix} \left(\sum_{\substack{c \in \{\text{imp}_{N'} \\ prc_stgp \}}} \frac{VAR_SIN_{r,y,t,p,c,s}}{PRC_ACTFLO_{r,y,p,c}} - \sum_{\substack{c \in \{\text{top}_{OUT} \\ prc_stgp \}}} \frac{VAR_SOUT_{r,y,t,p,c,s}}{PRC_ACTFLO_{r,y,p,c}} \right) \times \\ \begin{pmatrix} \sum_{\substack{c \in \{\text{top}_{OUT} \\ prc_stgp \}}} \frac{VAR_SOUT_{r,y,t,p,c,s}}{PRC_ACTFLO_{r,y,p,c}} \right) \\ \times \\ \begin{pmatrix} \sum_{\substack{v \in \{protody_{T_y} \\ y \ge M(t) \}}} \left(1 - STG_LOSS_{r,y,p,s} \right)^{\mathcal{B}(t)-y+0.5} \right) \\ \times \\ \times \\ \begin{pmatrix} \left(1 - STG_LOSS_{r,y,p,s} \right)^{\mathcal{M}(t)-\mathcal{Z}(t)-Mod(D(t)/2,1)} \right)^{1/f} prc_map_{r,STC_{P}} \end{pmatrix} \\ \times \\ \end{pmatrix} \\ + \\ \sum_{cometop_{OUT}} VAR_SOUT_{r,y,t,p,com,s} \times \\ \begin{pmatrix} 1 \\ COEF_PTRAN_{r,y,p,c,x,com,s} \end{cases} f COEF_PTRAN_{r,y,p,c,c,com,s} > 0 \end{pmatrix} \end{pmatrix}$$

6.3.36 Equation: EQ_STGTSS/IPS

Indices: region (r), vintage year (v), period (t), process (p), time-slice (s)

Туре: "="

Related variables: VAR_FLO, VAR_ACT

Related equations: EQ(l)_COMBAL, EQ(l)_CAPACT, EQ(l)_STGIN/OUT

Purpose

- The model allows two kinds of storage: inter-period storage (IPS), and storage across time-slices (or time-slice storage TSS). A special type of the TSS storage is a night-storage device, which may have an input commodity different from its output commodity. The input and output commodity of a night-storage device are given by the topology set **top**.
- Storage processes are special, as they have the same commodity as input and output. Also, all other processes transform energy within their time-slices and time periods. Since topology (with the exception of night-storage devices) does not determine in/out, different variables have to be used for this purpose. Similarly, since the transformation is special, EQ_PTRANS is replaced by new equations for the two types of storage.

Sets:

- prc_stgips(r,p,c): The set of inter-period storage processes. They are forced to operate annually.
- prc_stgtss(r,p,c): The set of time-slice storage processes. A storage process can operate
 only at one particular time slice level.
- prc_nstts(r,p,s): The set contains the allowed charging timeslices for a night-storage device.

Variables:

- VAR_SIN(r,v,t,p,c,s) the average in flow to a process built in period v, during timeslice s, during each year of period t. This variable would appear on the consumption side of the balance equation, without any coefficients.
- VAR_SOUT(r,v,t,p,c,s) the average out flow from a process built in period v, during time-slice s, during each year of period t. This variable would appear on the supply side of the balance equation, multiplied by *STG_EFF* and *COM_IE*.
- VAR_ACT(r,v,t,p,s) the energy stored in a storage process at the beginning of timeslice s (for a timeslice storage) or end of period t (for an inter-period storage). Note that this is a special interpretation of 'activity', to represent 'storage level.' Therefore, EQ_ACTFLO will not be generated for storage processes.
- In EQ_STGIPS only annual flows are allowed; the timeslice s index is set to ANNUAL in this case.

Equations:

- EQ_STGTSS(r,t,p,s) transforms input to output for the timeslice storage processes.
- EQ_STGIPS(r,t,p) transforms input to output for the interperiod storage processes.

Parameters:

- STG_LOSS(r,v,p,s) annual energy loss from a storage technology, per unit of (average) energy stored.
- STG_CHRG(r,t,p,s) exogenous charging of a storage technology. For timeslice storage this parameter can be specified for each period, while for interperiod storage this parameter can only be specified for the first period, to describe the initial content of the storage.

6.3.36.1 EQ STGTSS: Storage between timeslices (including night-storage devices):

Equation:

 $EQ _STGTSS_{r,v,t,p,s} \forall (r,v,t,p,s) \in (rtp_vintyr_{r,v,t,p} \land rps_stg_{r,p,s} \land prc_map_{r,STG,p})$



 $+STG_CHRG_{r,t,p,s-1}$

6.3.36.2 EQ_STGIPS: Storage between periods

Equation:

$$\begin{split} & EQ_STGIPS_{r,v,t,p} \ \forall (r,v,t,p) \in \left(\mathsf{rtp_vintyr}_{r,v,t,p} \cap \mathsf{prc_map}_{r,\mathsf{STK',p}} \right) \\ & VAR_ACT_{r,v,t,p,'ANNUAL'} = \end{split}$$

$$\sum_{\substack{\text{vertp}_v \text{inty}_{r,v,t-1p} \\ p \in \text{prc}_v \text{int}}} \left[\frac{VAR_ACT_{r,v,t-1,p;ANNUAL'}}{\times (1 - STG_LOSS_{r,v,p;ANNUAL'})^{D(t)}} \right] + \sum_{p \in \text{prc}_v \text{int}} \left[\frac{VAR_ACT_{r,t-1,d-1,p;ANNUAL'}}{\times (1 - STG_LOSS_{r,v,p;ANNUAL'})^{D(t)}} \right] + \left[\sum_{\substack{v \in \text{top}_{BU} \\ v \in \text{top}_{BU}}} \frac{VAR_SIN_{r,v,t,p,e;ANNUAL'}}{PRC_ACTFLO_{r,v,p,e}} - \sum_{\substack{v \in \text{top}_{BU} \\ v \in \text{top}_{BU}}} \frac{VAR_SUUT_{r,v,t,p,e;ANNUAL'}}{PRC_ACTFLO_{r,v,p,e}} \right] \times (1 - STG_LOSS_{r,v,p;ANNUAL'})^{(E(t)-y+0.5)}$$

+ $STG_CHRG_{r,t,p,ANNUAL'}(when ORD(t) = 1)$

6.3.37 Equations: EQ(*l*)_STGIN / EQ(*l*)_STGOUT

Indices: region (r), period (t), process (p), commodity (c), timeslice (s)

Type: Any type, as determined by the bound index bd of STGIN/OUT_BND:

- l = G' for $\mathbf{bd} = LO'$ (lower bound) yields \geq .
- $l = {}^{\circ}E'$ for **bd** = {}^{\circ}FX' (fixed bound) yields =.
- l = L' for **bd** = UP' (upper bound) yields \leq .

Related variables: VAR_SIN, VAR_SOUT

Related equations: EQ_STGTSS, EQ_STGIPS

Purpose: Bound on the input/output flow of a storage process of commodity (c) for a particular process (**p**) in period (**t**) and timeslice (**s**).

Remarks:

.

- The constraint bounds the flows in a specific period (t) irrespectively of the vintage years of the process capacity.
 - The constraint is generated if one of the following conditions is true:
 - Process (\mathbf{p}) is vintaged, or
 - the timeslice resolution of the flow variables (VAR_SIN/OUT) are below the timeslice (s) of the bound parameter.
 - In other cases, the bound can be directly applied to the flow variable
 - (VAR SIN/SOUT), so that no extra equation is needed.
- The timeslice level (s) of the bound must be at or higher than the timeslice level at which the storage operates.

Interpretation of the results:

Primal: If the primal value equals the bound parameter, the constraint is binding.

Dual: The dual value describes for a lower/upper bound the cost increase/decrease in the objective function, if the bound is increased by one unit. It may also be interpreted as subsidy/tax needed to reach the given bound value.

Equation:

$$EQ(l)_STGIN/OUT_{r_{L},p_{\mathcal{L},\mathcal{S}}} \Rightarrow \begin{cases} (r,t,p,c) \in \mathbf{rtpc}_{r,t,p,c} \land STGIN/OUT_BND_{r_{L},p,c,s,b,d} \land s \in \mathbf{rps_prcts}_{r,p,s} \land \\ (\mathbf{prc_vint}_{r,p} \lor (NOT \ \mathbf{prc_ts}_{r,p,s})) \end{cases}$$

$$All \ timeslices \ \mathbf{s} \ at \ or \ above \ the \ timeslice \ level \ of \ the \ process \ (\mathbf{prc_tsl}). \end{cases}$$

where the equation sign is indicated by equation index I based on the bound typebd.

6.3.38 Equations: EQ_STSBAL

Indices: region (r), vintage (v), period (t), process (p), timeslice (s) Type: =

Related variables: VAR_ACT, VAR_FLO, VAR_SIN, VAR_SOUT

Related equations: EQ_STGTSS

Purpose: Establishes the balance between different levels of a general timeslice storage.

Equation:

$$\begin{split} EQ_STSBAL_{r,y,l,p,s} &\forall (r, v, t, p, s) \in \left(\mathsf{rtp_vintyr}_{r,v,t,p} \cap \mathsf{prc_ts}_{r,p,s} \cap \neg \mathsf{rps_stg}_{r,p,s} \right) \\ &\sum_{\substack{\sum VAR_ACT_{r,y,l,p,s} = \\ \mathsf{rs_bdow}_{r,\mathsf{ANNUALls}}}} \\ & \mathsf{rs_bdow}_{r,\mathsf{ANNUALls}} \\ & \mathsf{fs_bdow}_{r,\mathsf{ANNUALls}} \\ \\ & \mathsf{fs_bdow}_{r,\mathsf{ANNUALls}} \\ & \mathsf{fs_bdow}_{r,\mathsf{ANNUALls}} \\ & \mathsf{fs_bdow}_{r,\mathsf{ANNUALls}} \\ \\ & \mathsf{fs_bdow}_{r,\mathsf{ANNUALls}} \\ \\ & \mathsf{fs_bdow}_{r,\mathsf{ANNUALls}} \\ \\ & \mathsf{fs_bdow}_{r,\mathsf{ANNUALls}} \\ \\ & \mathsf{fs_bdow}_{r,\mathsf{ANU},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs}}} \\ \\ & \mathsf{fs_bdow}_{r,\mathsf{ANU},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r,\mathsf{a},\mathsf{fs_bdow}_{r$$

6.3.39 Equations: EQ(*l*)_UCRTP

Indices: name (uc_n), region (r), period (t), process (p), type (uc_grptype), bound (bd)

Type: Any type, as determined by the bound index **bd** of **uc_dynbnd**:

- l = 'N' for **bd** = 'LO' (lower bound) yields \geq .
- l = 'N' for **bd** = 'UP' (upper bound) yields \leq .
- l = 'E' for **bd** = 'FX' (fixed bound) yields =.

Related variables: VAR_ACT, VAR_CAP, VAR_NCAP

Purpose: Dynamic bound on the growth/decay in the capacity (CAP), new capacity (NCAP) or activity level (ACT) of a particular process (**p**) between period (**t**) and previous period (**t**-1).

Remarks:

- The input set **uc_dynbnd** must be used for flagging the pairs (uc_n,bd) to be reserved for dynamic bound constraints.
- The input parameters UC_CAP, UC_NCAP, and UC_ACT should be used for defining the growth/decay coefficients (side='LHS') and RHS constants (side='RHS').
- The growth/decay coefficients (side='LHS') are given as annual multipliers (e.g. 1.1 for a 10% annual growth). The RHS constants (side='RHS') represent annual absolute values of additional growth/decay.
- The LHS is by default interpolated using option 5. If no LHS is specified, the RHS is by default interpolated with the option 10, like other bounds. However, if the LHS is also specified, the RHS is by default interpolated by the same option as the LHS.
- Whenever any *RHS values* are specified, the constraints will be generated for those
 periods for which the RHS is defined after the interpolation/extrapolation. If no
 RHS is specified, the constraints are generated for the periods that have the LHS
 defined, but excluding the first period of technology availability.
- In the case of dynamic bounds on the activity (ACT), the UC_ACT values must be specified at the ANNUAL level, and constraint bounds the change in the total activity in a specific period (t), summing over the process vintages and timeslices.

Equations:

 $\begin{aligned} \text{Case A. For CAP:} \\ EQ(l)_UCRTP_{uc_n,r,t,p,CAP',bd} \rightarrow \left(\mathbf{rtp}_{r,t,p} \wedge \mathbf{uc_dynbnd}_{uc_n,bd} \wedge \left(\sum_{ade} (UC_CAP_{uc_n,side,r,t,p}) > 0 \right) \right) \\ VAR_CAP_{r,t,p} \\ & \{\leq :=, \geq \} \\ VAR_CAP_{r,t-1,p} \times (UC_CAP_{uc_n,'LHS',r,t,p})^{(M(t)-M(t-1))} + UC_CAP_{uc_n,'RHS',r,t,p} \times (M(t)-M(t-1))) \\ \text{Case B. For NCAP:} \\ EQ(l)_UCRTP_{uc_n,r,t,p,'NCAP,bd} \rightarrow \left(\mathbf{rtp}_{r,t,p} \wedge \mathbf{uc_dynbnd}_{uc_n,bd} \wedge \left(\sum_{side} (UC_NCAP_{uc_n,side,r,t,p}) > 0 \right) \right) \\ VAR_NCAP_{r,t,p} \\ & \{\leq :=, \geq \} \\ VAR_NCAP_{r,t-1,p} \times (UC_NCAP_{uc_n,'LHS',r,t,p})^{(M(t)-M(t-1))} + UC_NCAP_{uc_n,'RHS',r,t,p} \times (M(t)-M(t-1))) \end{aligned}$

Case C. For ACT:

$$EQ(l)_UCRTP_{uc_n,r,t,p,'ACT',bd} \rightarrow \left(\mathbf{rtp}_{r,t,p} \land \mathbf{uc_dynbnd}_{uc_n,bd} \land \left(\sum_{xde} \left(UC_ACT_{uc_n,side,r,t,p,'ANNUAL'} \right) > 0 \right) \right) \right)$$

$$\sum_{\mathbf{vertp_vintyr}_{r,r,t,p}} \sum_{s \in \mathbf{prc_s}} VAR_ACT_{r,t,p,s} \quad \{ \leq := ; \geq \}$$

$$\sum_{\mathbf{vertp_vintyr}_{r,r,t,p}} \sum_{s \in \mathbf{prc_s}} VAR_ACT_{r,t-1,p,s} \times \left(UC_ACT_{uc_n,'LHS',r,t,p,'ANNUAL'} \right)^{(M(t)-M(t-1))} + UC_ACT_{uc_n,'RHS',r,t,p,'ANNUAL'} \times (M(t)-M(t-1)) \right)$$

6.4 User Constraints

This section on TIMES User Constraints explains the framework that may be employed by modellers to formulate additional linear constraints, which are not part of the generic constraint set of TIMES, without having to bother with any GAMS programming.

6.4.1 Overview

Indexes: region (r), time period (t), time slice (s), user constraint (uc_n)

Type: Any type, as determined by the bound index **bd** of $UC_RHS(R)(T)(S)_{(r),uc_n,(t),(s),bd}$:

- l = 'G' for **bd** = 'LO' (lower bound) yields \geq .
- l = 'E' for **bd** = 'FX' (fixed bound) yields =.
- l = 'L' for **bd** = 'UP' (upper bound) yields \leq .

Related variables: VAR_ACT, VAR_CAP, VAR_FLO, VAR_IRE, VAR_NCAP, VAR_COMPRD, VAR_COMNET, VAR_CUMCOM, VAR_CUMFLO, VAR_UPS

Related equations: EQ(1)_COMBAL, EQ(1)_CPT

Purpose: The user constraints in TIMES provide a modeler with a flexible framework to add case-study specific constraints to the standard equation set embedded in TIMES. With the help of the user constraints virtually any possible linear relationship between variables in TIMES can be formulated. Examples of user constraints are quotas for renewables in electricity generation or primary energy consumption, GHG reduction targets, absolute bounds on the minimum amount of electricity generated by various biomass technologies, etc.

Four types of user constraints can be distinguished in TIMES:

- Pure LHS (left hand side) user constraints,
- Timeslice-dynamic user constraints,
- Dynamic user constraints of type (t, t+1), and
- Dynamic user constraints of type (t-1, t).

In addition, the dynamic bound constraints (see EQ_UCRTP) also employ user constraint names and UC_* attributes, but these constraints are based on prescribed expressions and are thus not considered genuine user constraints.

In the following four subsections, the different types of user constraints are briefly presented. Their mathematical formulations are then presented in a new section.

LHS user constraints

The so-called LHS user constraints have the following main structure:

$$EQ(l)_UC(R)(T)(S)_{(r),\mu\sigma_n,(l),(s)} \forall \begin{cases} UC_RHS(R)(T)(S)_{(r),\mu\sigma_n,(l),(s),bd} \land (r \in uc_r_each_{r,u_n}) \\ \land (t \in uc_t_each_{r,u_n,l}) \land (s \in uc_ts_each_{r,u_n,s}) \end{cases}$$

To identify the user constraint, the modeller has to give it a unique name **uc_n**. The LHS expression $LHS_{r,t,s}$ consists of the sum of various TIMES variables (VAR_ACT, VAR_FLO, VAR_COMPRD, VAR_COMNET, VAR_NCAP, VAR_CAP), multiplied by corresponding coefficients (UC_ACT, UC_FLO, UC_COMPRD, UC_COMCON, UC_NCAP, UC_CAP). The coefficients are input data given by the modeller and serve thus also as an indicator of which variables are being components of the user constraint.

With respect to region r, time period t and timeslice s, the user constraint is either specified for specific regions, periods or timeslices or the expression within the user constraint is summed over subsets of regions, periods and timeslices. In the first case, the regions, periods or timeslices for which the user constraint should be generated are given by the sets uc_r_each, uc_t_each or uc_ts_each, while in the latter case, summation sets are specified by the sets uc_r_sum, uc_t_sum and uc_ts_sum. The corresponding sets uc x each/sum are exclusive, so that for example, if uc t each has been specified, the set uc_t_sum cannot be specified and vice versa. By choosing uc_x_each/sum also the name and the index domain of the user constraint are specified, e.g. if uc r each, uc t each and uc_ts_sum are given, the user constraint has the name and index domain EQ(l)_UCRT_{r,uc n,t}. It is generated for each region and period specified by uc_r_each and uc_t_each, respectively, and is summing within the user constraint over the timeslices given in uc ts each. The name of the RHS constraint depends in the same way on the choice of uc x each/sum. In the previous example, the RHS constant has the name and index domain UC_RHSRT_{r,uc_n,t,bd}. The knowledge of these naming rules is important, since the modeller has to give the correct RHS parameter names depending on the choice of uc_x_each/sum when defining a user constraint.

Since for each of the three dimensions (region, period, timeslice), two options (EACH or SUM) exist, this would result in 8 possible combinations of user constraint equations (Figure 5.6). However, the combinations EQ(1)_UCS and EQ(1)_UCRS, which would lead to a constraint being generated for specific timeslices while summing over time periods at the same time, have been considered unrealistic, so that 6 variants remain. It should be noted that the sets uc_r_each/sum, uc_t_each/sum and uc_ts_each/sum can contain an arbitrary combination of elements, e.g. the periods specified in uc_t_each/sum do not have to be consecutive.



Figure 14: The allowed combinations of region, period and timeslice for user constraints.

The RHS (right hand side) of this category of user constraint consists of a constant $UC_RHS(R)(T)(S)_{(r)uc_n,(t),(s),bd}$ which is provided by the modeller. The RHS constant also defines the equation type of the user constraint. If the RHS constant has the index FX, the user constraint is generated as strict equality (=). If the RHS index is LO (respectively UP), the constraint has \geq (respectively \leq) inequality sign. It should be noted that a RHS user constraint is only generated when a RHS constant is specified (this feature may be used to easily turn-on/off user constraints between different scenarios).

In addition to the coefficients UC_ACT, UC_FLO, etc. also some model input attributes may be used as coefficient for the variables in a user constraint. The model attribute being used as coefficient in a user constraint is specified by the set $UC_ATTR_{r,uc_n,THS,VAR,ATTR}$ with the indicator VAR for the variable (ACT, FLO, IRE, NCAP, CAP, COMNET, COMPRD) and the index ATTR representing the attribute being used (COST, SUB, TAX, DELIV, INVCOST, INVSUB, INVTAX, CAPACT, CAPFLO, NEWFLO, ONLINE, EFF, NET, CUMSUM, PERIOD, GROWTH, see Section 6.4.6 for more information).

Instead of defining different equality types of user constraints depending on the bound type of $UC_RHS(R)(T)(S)_{(r),uc_n,(t),(s),bd}$ an alternative formulation can be used in TIMES.

In this formulation a variable $VAR_UC(R)(T)(S)_{(r),uc_n,(t),(s)}$ is created that is set equal to the LHS expression. The RHS bounds are then applied to these variables.

$$EQE_UC(R)(T)(S)_{(r),\muc_n,(t),(s)} \forall \begin{cases} UC_RHS(R)(T)(S)_{(r),\muc_n,(t),(s),bd} \land (r \in uc_r_each_{r,uc_n}) \\ \land (t \in uc_t_each_{r,uc_n,t}) \land (s \in uc_ts_each_{r,uc_n,s}) \end{cases}$$

$$\left(\sum_{r\in uc_r_sum_{r,uc_n,t}} \right) (\sum_{s\in uc_ts_sum_{r,uc_n,s}} \sum_{s\in uc_ts_sum_{r,uc_n,s}} \sum_{s= uc_ts_sum_{r,uc_ts_sum_{r,uc_n,s}} \sum_{s= uc_ts_sm_{r,uc_n,s}} \sum_{s= uc_ts_sm_{r,uc_ts_sm_{r,uc_n,s}$$

The alternative formulation is created when the dollar control parameter VAR_UC (see Part III for the use of dollar control parameters) is set to YES by the modeller, while in the default case the first formulation is used.

Timeslice-dynamic user constraints

Timeslice-dynamic user constraints establish a relationship between two successive timeslices within a timeslice cycle. The LHS expression $LHS_{r,t,s}$ is generated for timeslice s, whereas the RHS expression $RHS_{r,t,s-1}$ is generated for the preceding timeslice s–RS_STG(r,s) under the same parent timeslice. Timeslice-dynamic user constraints of type can thus be written as follows:

$$EQ(I)_UCRS_{r,uc_n.t.tsl,s} \ni \begin{cases} UC_RHSRTS_{r,uc_n.t.s,bd} \land (r \in uc_r_each_{r,uc_n}) \land (t \in uc_t_each_{r,uc_nt}) \\ \land (s \in \{ts \mid ts_grp_{r,tsl,s} \land \bigcup_{slde} uc_tsl_{r,uc_n,slde,tsl}\}) \end{cases}$$

$$LHS_{r,t,s} \qquad \left\{= / \ge / \le\right\} \sum_{\mathbf{uc_tsl}(r,\mu;cn';RHS';tsl)} RHS_{r,t,s-RS_STG(r,s)} + UC_RHS(R)T(S)_{(r),\mu;c_n,t,(s),bal}$$

Timeslice-dynamic user constraints are always specific to a single region and period. To build a timeslice-dynamic user constraint, the modeller must identify the desired timeslice level of the constraint, by using the set $uc_tsl_{r,uc_n,side,tsl_n}$ and the RHS constants must be defined by using the UC_RHSRTS parameter. The constraint will be genuinely dynamic only if uc_tsl is specified on the RHS. This is the only type of user constraint for which the RHS constant parameter is levelized, according the timeslice level identified by uc_tsl . That can make the RHS specification much easier.

Dynamic user constraints

Dynamic user constraints establish a relationship between two *consecutive* periods. The LHS expression $LHS_{r,t,s}$ is generated for period **t**, whereas the for the RHS expression either the term $RHS_{r,t-1,s}$ corresponding to the period **t+1**, or the term $RHS_{r,t-1,s}$ corresponding to the period **t+1** is generated, according to the dynamic type.

Dynamic user constraints of type (t,t+1) can thus be written as follows:

$$EQ(I)_UC(R)SU(S)_{(r),uc_n,t,(s)} \Rightarrow \begin{cases} UC_RHS(R)T(S)_{(r),uc_n,t,(s),bd} \land (r \in uc_r_each_{r,uc_n}) \\ \land (t \in uc_t_succ_{r,uc_n,t}) \land (s \in uc_ts_each_{r,uc_n,s}) \end{cases}$$

 $UC_RHS(R)T(S)_{(r),uc_n,t,(s),bd}$

Similarly, dynamic user constraints of type (t-1,t) can be written as follows:

$$EQ(I)_UC(R)SU(S)_{(r),uc_nt,(s)} \ni \begin{cases} UC_RHS(R)T(S)_{(r),uc_nt,(s),bd} \land (r \in uc_r_each_{r,uc_n}) \\ \land (t \in uc_t_succ_{r,uc_nt}) \land (s \in uc_ts_each_{r,uc_ns}) \end{cases}$$

$$\left(\sum_{r \in uc_r_sum_{ruc_n}} \left(\sum_{s \in uc_ts_sum_{ruc_n,s}}\right) LHS_{r,t,s} \left\{= / \ge / \le\right\} \left(\sum_{r \in uc_r_sum_{ruc_n}} \left(\sum_{s \in uc_ts_sum_{ruc_n,s}}\right) RHS_{r,t-1,s} + UC_RHS(R)T(S)_{(r),uc_n,t,(s),bd}\right)$$

To build a dynamic user constraint of type (t,t+1), between the periods t and t+1, the modeller identifies the desired set of time periods that will be used as first periods in the pairs (t, t+1). This set is named **uc_t_succ** (note that the sets **uc_t_sum** and **uc_t_each** are not used in the context of dynamic user constraints, and are reserved for the pure LHS user constraints described in the previous section). In addition, the RHS constant parameter must be defined for all of these time periods.

To build a dynamic user constraint of type (t-1,t), between the periods t-1 and t, the modeller should indicate the desired type by defining for the constraint any UC_ATTR attribute using the 'RHS' side. In addition, the desired set of time periods that will be used as the second period in the pairs(t-1,t) should be identified by defining the RHS constant parameter for those periods t.

The choice between the dynamic types (t,t+1) or (t-1,t) is usually only a matter of convenience. However, while using the (t,t+1) type requires explicit specification of **uc_t_succ**, for using the (t-1,t) type, any UC_ATTR on the RHS is sufficient to trigger that dynamic type and will cause auto-generation of **uc_t_succ** for all milestone years.

For both types of dynamic constraints, only four combinations with respect to the region and timeslice domain are possible:

- EQ(1)_UCSU: dynamic user constraint summing r over uc_r_sum and s over uc_ts_sum,
- EQ(1)_UCRSU: dynamic user constraint being generated for each region uc_r_each and summing s over uc_ts_sum,
- EQ(1)_UCRSUS: dynamic user constraint being generated for each region **uc_r_each** and timeslice **uc_ts_each** and
- EQ(1)_UCSUS: dynamic user constraint summing **r** over **uc_r_sum** and being generated for each timeslice **s** in set **uc_ts_each**.

The input parameters for defining the coefficients, UC_ACT, UC_FLO, UC_IRE, UC_COMCON, UC_COMNET, UC_COMPRD, UC_NCAP and UC_CAP all have an index side, which can be either LHS or RHS, to identify on which side of the user constraint the corresponding variables should appear. The LHS index corresponds always to the period t, while the RHS index is related either to the t+1 or the t-1 term.

As for LHS user constraints, setting the dollar control parameter VAR_UC to YES yields a strict equality type of dynamic user constraint (EQE_UCSU, EQE_UCRSU, EQE_UCRSUS, EQE_UCSUS) with the RHS constant replaced by a user constraint variable (VAR_UCT, VAR_UCRT, VAR_UCRTS, VAR_UCTS). The bound given by the RHS constant is then applied to the user constraint variable.

Growth constraints

Growth (or decay) constraints are a special type of dynamic constraints. A growth constraint may for example express that the capacity increase between two periods is limited by an annual growth rate. So, growth constraints relate variables in one period to the ones in the previous or following period as in dynamic constraints described in the previous section. In growth constraints, however, in addition some of the variable coefficients UC_ACT, UC_FLO, UC_IRE, UC_COMNET, UC_COMPRD, UC_NCAP, UC_CAP can represent annual growth (or decay) rates⁴⁴ by specifying the set $UC_ATTR_{r,wc_nTHS',VARATTR}$ with the index ATTR being set to GROWTH. This will cause the coefficient of the corresponding variable being interpreted as an annual growth rate. If for example the input information $UC_ATTR_{RBGI',G_J',THS',CAP',GROWTH'}$ is given for the user constraint G_1, the coefficient $UC_CAP'_{G_J',THS', REGI',LP}$ of the capacity variable of technology **p** will be interpreted as annual growth rate and the final coefficient of the variable VAR_CAP in the user constraint will be calculated in the following way:

 $\left(UC_CAP_{G}]_{1'LHS'',REG1't,p}\right)^{M(t+1)-M(t)}$.

With the help of the input set UC_ATTR, growth coefficients can be defined for the variables in LHS expression (as in the example) or for the variables in RHS expression. If a

⁴⁴ If the coefficient UC_ACT, UC_FLO, etc. is greater than one, it represents an annual growth rate, while a coefficient smaller than one describes an annual decay rate.

growth rate is defined for variables on the LHS, the exponent is M(t+1)-M(t), whereas for RHS variables the exponent is equal to M(t)-M(t+1).

If at least one growth coefficient is defined for a LHS variable, the dynamic constraint will be assumed to be of type (t,t+1) described above. In this case, the growth constraints are generated for the period pairs t and t+1 for all periods t of the model horizon with the exception of the last period.

If, however, all growth coefficients are specified for the RHS variables, the dynamic constraint will be assumed to be of type (t-1,t), and the growth constraints are now generated for the period pairs t-1 and t for all periods of the model horizon. In this alternative RHS formulation, it is possible to introduce boundary conditions that are usually needed for the first period.

Example of defining a simple growth constraint:

The annual capacity increase of technology E01 between two periods should not exceed 2% for model covering the three ten-year periods 1990, 2000 and 2010. So one wants to create user constraints expressing:

```
\begin{split} 1.02^{10} \times VAR\_CAP_{{}_{\textit{EEG1'},1990,'E01'}} + 1 \ge VAR\_CAP_{{}_{\textit{EEG1'},2000,'E01'}} \\ 1.02^{10} \times VAR\_CAP_{{}_{\textit{EEG1'},2000,'E01'}} + 1 \ge VAR\_CAP_{{}_{\textit{EEG1'},2010,'E01'}} \end{split}
```

The summand 1 on the LHS expresses an initial capacity value, so that capacity growth can start from this starting point, e.g. if $VAR_CAP_{REGI',1990',TE0I'}$ is zero, the model can invest at most 1 capacity unit in the year 2000: $1 \ge VAR_CAP_{REGI',2000',TE0I'}$.

Since growth constraints should be generated for the first two periods, but not the last one, the growth constraint should be of type (t,t+1). The specification of the growth constraint called 'G 1' in GAMS looks like:

```
SET UC_N /
G_1
/
* Specify growth of capacity on the LHS
SET UC_ATTR /
REG1.G_1.LHS.CAP.GROWTH
/
* Specify growth coefficient for E01 on LHS (period 1) and coefficient
* for capacity on RHS (period t+1)
PARAMETER UC_CAP /
* on the LHS
G1.LHS.REG1.2000.E01 1.02
* on the RHS
G1.RHS.REG1.2000.E01 1
/
* Specify RHS constant for the years t to have the constraint
PARAMETER UC_RHSRTS /
REG1.G_1.1990.ANNUAL.LO -1
REG1.G_1.2000.ANNUAL.LO -1
/;
```

One should note that the period index used for the UC_CAP on the LHS is related to the period t, while the period index on the RHS is related to the period t+1. The RHS UC_RHSRTS constant is provided for the time period t of the LHS.

Since a growth coefficient is specified for the LHS, the user constraint is automatically identified as a dynamic growth constraint, so that the set **uc_t_succ** does not need to be provided by the user. The constraint will be generated for all periods for which the RHS parameter UC_RHSRTS is given.

In the following section, we give the full descriptions of the available user constraints in each category, along with a reminder of the corresponding variables.

Mathematical descriptions of user constraints

List of user constraints and variables

We first show the complete list of user constraints in the three categories.

The following types of LHS user constraints exist:

• $EQ(l)_UC_{uc_n}$:	user constraint summing over regions uc_r_sum, periods
	uc_t_sum and timeslices uc_ts_sum,
• $EQ(l)_UCR_{r,uc_n}$:	user constraint generated for regions uc_r_each and summing
	over periods uc_t_sum and timeslices uc_ts_sum,
• $EQ(l)_UCT_{uc_n,t}$:	user constraint generated for periods uc_t_each and summing
	over regions uc r sum and timeslices uc ts sum,

- EQ(l)_UCRT_{r,nc_n,t}: user constraint generated for regions uc_r_each and periods uc_t_each and summing over timeslices uc_ts_sum,
- EQ(1)_UCTS_{uc_n,t,s}: user constraint generated for periods uc_t_each, timeslices uc_ts_each and summing over regions uc_r_sum,
- EQ(1)_UCRTS_{r,uc_n,t,s}: user constraint generated for regions uc_r_each, periods uc_t_each and timeslices uc_ts_each.

The placeholder l reflects the equation type of the user constraint (l=E, G or L) corresponding to the bound type of the RHS constant. In case the dollar control parameter VAR_UC is set to YES, the user constraints are always strict equalities (l=E) with the RHS constants replaced by the following user constraint variables:

•	$VAR_UC_{uc_n}$:	user constraint variable for EQE_UC,
٠	VAR_UCR_{r,uc_n} :	user constraint variable for EQE_UCR,
•	VAR_UCT _{uc_n,t} :	user constraint variable for EQE_UCT,
•	VAR_UCRT _{r,uc_n,t} :	user constraint variable for EQE_UCRT,
•	VAR_UCTSuc_n,t,s:	user constraint variable for EQE_UCTS,
٠	VAR_UCRTSuc_n,r,t,s:	user constraint variable for EQE_UCRTS.

The following types of dynamic user constraints and growth constraints exist:
 EQ(l)_UCSU_{uc_nt}: user constraint generated for periods uc_t_succ, summing over regions uc_r_sum and timeslices uc_ts_sum,

٠	$EQ(l)_UCRSU_{r,uc_n,t}$:	user constraint generated for regions uc_r_each and periods
		<pre>uc_t_succ and summing over timeslices uc_ts_sum,</pre>
•	$EQ(l)_UCSUS_{uc_n,t,s}$:	user constraint generated for periods uc_t_succ, timeslices
		uc_ts_each and summing over regions uc_r_sum,
٠	$EQ(l)_UCRSUS_{r,uc}$ n,t,s:	user constraint generated for regions uc_r_each, periods
	107407. 100 mm	uc t succ and timeslices uc ts each.

The placeholder l reflects the equation type of the user constraint (l=E, G or L) corresponding to the bound type of the RHS constant. In case the dollar control parameter VAR_UC is set to YES, the user constraints are always strict equalities (l=E) with the RHS constants replaced by the following user constraint variables:

٠	$VAR_UCT_{uc_n,t}$:	user constraint variable for EQE_UCSU,
•	VAR_UCRT _{r,uc_n,t} :	user constraint variable for EQE_UCRSU,
•	VAR_UCTSuc_n,t,s:	user constraint variable for EQE_UCSUS,
•	$VAR_UCRTS_{uc_n,r,t,s}$:	user constraint variable for EQE_UCRSUS.

Sets and parameters related to user constraints

The following sets and parameters are related to the user constraint framework in TIMES.

Sets

Predefined internal sets:

- side : set having the two elements *LHS* and *RHS* (elements are fixed and not under user control),
- uc_grptype : set having the elements ACT, FLO, IRE, COMCON, COMPRD, NCAP, CAP used in the multi-dimensional set UC_ATTR (elements are fixed and not under user control),
- uc_name: set having the following attribute names as elements: COST, SUB, TAX, DELIV, INVCOST, INVSUB, INVTAX, CAPACT, CAPFLO, NEWFLO, ONLINE, EFF, NET, CUMSUM, PERIOD, GROWTH and SYNC used in the multi-dimensional set UC_ATTR (elements are fixed and not under user control).

User-specified sets:

- uc_n: unique name of the user constraint,
- uc r eachr, uc n : regions r for which the user constraint uc n is generated,
- uc_r_sum_{r,uc_n}: regions r being summed over in the user constraint uc_n,
- uc_t_eachr,uc_n,t: periods t for which the user constraint uc_n is generated,
- uc_t_sumr,uc_n,t: periods t being summed over in the user constraint uc_n,
- uc_ts_eachr,uc_n,ts: timeslices ts for which the user constraint uc_n is generated,
- uc_ts_sumr,uc_n,ts: timeslices ts being summed over in the user constraint uc_n,
- uc_tsl_{r,uc_n,side,tslvl}: timeslice level tslvl of user constraint uc_n,

• uc_attr_{r,uc_n,side,uc_grptype,uc_name}: indicator that the attribute uc_name on the RHS or LHS side of the user constraint uc_n as coefficient of the variable given by uc_grptype.

If neither uc_r_each nor uc_r_sum are given, the default is set to all uc_r_each containing all internal regions. In a similar fashion uc_t_each being set to all milestoneyears is the default, if neither uc_t_each or uc_t_sum are specified. The default for the timeslice dimension is uc_ts_each being set to all timeslices for which the RHS constants UC_RHSRS or UC_RHSRTS are being specified.

Parameters

User-specified coefficients of variables:

•	$UC_ACT_{uc_n,side,r,t,p,s}$:	coefficient of the activity variable $VAR_ACT_{r,v,t,p,s}$ in the
		user constraint uc_n on the LHS or RHS side ,
•	UC_FLO_{uc} n, side, r, t, p, c, s:	coefficient of the flow variable $VAR_FLO_{r,v,t,p,c,s}$ in the
		user constraint uc n on the LHS or RHS side,
•	UC IREnc nodertne sie:	coefficient of the inter-regional exchange variable
	nc_n, side, r, i, p, c, s, ie -	VAR IREman in the user constraint up n on the LHS
		or RHS side
	UC COMCON	as officiant of the virtual commodity consumption
	UC_COMCON uc_n, side, r, t, c, s.	interview of the virtual commodity consumption
		variable $(VAR_COMPRD_{r,t,c,s}-VAR_COMNET_{r,t,c,s})$ in the
	AND TRANSPORT TO AND TRANSPORT	user constraint uc_n on the LHS or RHS side,
•	UC_COMPRDuc_n, side, r, t, c, s:	coefficient of the gross commodity production variable
		$VAR_COMPRD_{r,t,c,s}$ in the user constraint uc_n on the
		LHS or RHS side,
•	UC COMNETuc n, side, r, t, c, s:	coefficient of the net commodity production variable
		VAR COMNET _{r,t,c,s} in the user constraint uc n on the
		LHS or RHS side.
•	UC CUMACT 16 NERVINE	coefficient of the cumulative process activity variable
	= = = = = = = = = = = = = = = = =	VAR CUMFLO _{n n'ACTURE} in the user constraint up n
		(only in cumulative constraints)
	UC CUMCOM	configure of the sumulative commodity not or group
	UC_COMCOMuc_n,r,type,c,y1,y2	reduction variable VAD CUDICOM
		production variable VAR_COMCOM _{r,c,type,y1,y2} in the user
		constraint uc_n, where type =PRD/NE1 (only in
		cumulative constraints),
•	$UC_CUMFLO_{uc_n,r,p,c,yI,y2}$:	coefficient of the cumulative process flow variable
		$VAR_CUMFLO_{r,p,c,yl,y2}$ in the user constraint uc_n (only
		in cumulative constraints),
•	$UC_NCAP_{uc_n,side,r,t,p}$:	coefficient of the investment variable VAR_NCAP _{r,t,p} in
		the user constraint uc_n on the LHS or RHS side,
•	$UC_CAP_{uc_n, side, r, t, p}$:	coefficient of the capacity variable $VAR_CAP_{r,t,p}$ in the
		user constraint uc n on the LHS or RHS side.

User-specified RHS constants:

- $UC_RHS_{uc_n,bd}$: RHS constant with bound type **bd** of the user constraint
- *UC_RHSR*,*uc_n,bd*: EQ1_UC_{uc_n} of type 1,
 RHS constant with bound type bd of the user constraint
- $UC_RHST_{uc_n,t,bd}$: EQl_UCR_{r,uc_n} of type **l**, RHS constant with bound type **bd** of the user constraint
- UC_RHSRT_{r,uc_n,t,bd}: RHS constant with bound type **bd** of the user constraint
- $EQL_UCRT_{r,uc_n,t}$ of type **I**,
- $UC_RHSTS_{uc_n,t,s,bd}$: RHS constant with bound type **bd** of the user constraint $EQl_UCTS_{uc_n,t,s}$ of type **l**,
- $UC_RHSRTS_{r,uc_n,t,s,bd}$: RHS constant with bound type **bd** of the user constraint $EQl_UCRTS_{r,uc_n,t,s}$ of type **l**.
- $UC_TIME_{uc_n,r,t}$: Defines an additional term in the RHS constant, which is either the time (in years) covered by the user constraint multiplied by UC_TIME (for static and cumulative constraints), or the time between the milestone years of the successive periods in the constraint (for dynamic user constraints).

6.4.2 LHS user constraints

Mathematical formulation of LHS user constraints

In the mathematical description of the different variants of LHS user constraints the following placeholders are used for clarity reasons: $ACT_{r,t,p,s',THS'}$, $FLO_{r,t,p,s',THS'}$, $IRE_{r,t,p,s',THS'}$, $COMPRD_{r,t,s,THS'}$, $COMNET_{r,t,s,THS'}$, $NCAP_{r,t,p,s',THS'}$, $CAP_{r,t,p,s',THS'}$, $CUMCOM_r$ and $CUMFLO_r$. For example the placeholder $ACT_{r,t,p,s',THS'}$ includes the part of the user constraint related to the activity variable.

$$ACT_{r,t,s,'LHS'}$$

$$= \left(\begin{array}{c} VAR_ACT_{r,v,t,p,te} \times UC_ACT_{uo_n,'LHS',r,t,p,te} \times \left(RS_FR_{r,s,te} \right) \\ \times \\ \left(\begin{array}{c} \sum_{(v,p) \in rtp_vintyr_{r,v,tp}} \sum_{t \neq e pre_6_{r,p,te}} \left(\begin{array}{c} VAR_ACT_{r,v,t,p,te} \times UC_ACT_{uo_n,'LHS',r,t,p,te} \times \left(RS_FR_{r,s,te} \right) \right) \\ \end{array} \right) \\ \end{array} \right)$$

$$FLO_{r,t,s,'LHS'}$$

Ξ

$$\left[\begin{array}{c} \sum_{(p, c, t, b) \in rpec_varf_{clp,clb} \lor erp_vlntyr_{v,vlp}} \\ \left[VAR_FLO_{r, v, t, p, c, to} \times UC_FLO_{uc_n, LHS', r, t, p, c, to} \times \left(RTCS_TSFR_{r, t, c, \sigma, to} \right) \right] \\ \times \\ \left[\sum_{\substack{(U) \in rdeur_{r,ver}}} \left(\begin{array}{c} OBJ_FCOST_{r, t, p, c, to, cur} & if UC_ATTR_{r, uc_n, 'LHS', FLO', COST'} & is given \\ + \\ OBJ_FDELV_{r, t, p, c, to, cur} & if UC_ATTR_{r, uc_n, 'LHS', FLO', 'DELIV'} & is given \\ - \\ OBJ_FSUB_{r, t, p, c, to, cur} & if UC_ATTR_{r, uc_n, 'LHS', FLO', 'DELIV'} & is given \\ + \\ OBJ_FTAX_{r, t, p, c, to, cur} & if UC_ATTR_{r, uc_n, 'LHS', FLO', 'SUB'} & is given \\ \end{array} \right] \right]$$



$$COMPRD_{r,t,s,1LHS'}$$

$$= \sum_{(c,tz) \in \mathsf{rts}_varc_{ut,ets}} \left(VAR_COMPRD_{r,t,c,ts} \times UC_COMPRD_{uc_n,'LHS',r,t,c,s} \times (RTCS_TSFR_{r,t,c,sts}) \times \sum_{\substack{(c,tz) \in \mathsf{rts}_varc_{ut,ets}}} (OBJ_COMPD_{r,t,c,ts,\mu c_cost,cur} \quad if \ uc_attr_{r,\mu c_n,'LHS','COMPRD',\mu c_cost} \times (CMPRD', cost, cur) \right)$$

$$COMNET_{r,t,s,'LHS}$$

 $NCAP_{r,t,'LHS'}$

Ξ

$$\sum_{p} \begin{bmatrix} VAR_NCAP_{rt,p} \times UC_NCAP_{uc_n,'LHS',r,t,p} \times \\ \sum_{curerdenr_{rew}} OBJ_ICOST_{r,t,p,cur} & if UC_ATTR_{r,uc_n,'LHS','NCAP,'COST} & is given \\ - \\ \sum_{curerdenr_{rew}} OBJ_ISUB_{r,t,p,cur} & if UC_ATTR_{r,uc_n,'LHS','NCAP,'SUB'} & is given \\ + \\ \sum_{curerdenr_{rew}} OBJ_ITAX_{r,t,p,cur} & if UC_ATTR_{r,uc_n,'LHS','NCAP,'TAX'} & is given \\ \end{bmatrix}$$

$$CAP_{r,t,p,'LHS'}$$

 $= \sum_{p} \begin{bmatrix} VAR _ CAP_{r,t,p} \times UC _ CAP_{uc_n,'LHS',r,t,p} \\ \times \\ PRC _ CAPACT_{r,p} \quad if \ UC _ ATTR_{r,uc_n,'LHS','CAP,'CAPACT'} \ is \ given \end{bmatrix}$

 $CUMCOM_r$

$$= \sum_{re_cumcom_{r,c,com_var,yl,y2} \times UC_CUMCOM_{uc_n,r,com_var,c,yl,y2}} VUC_CUMCOM_{uc_n,r,com_var,c,yl,y2}$$

$$CUMFLO_{r}$$

$$= \sum_{re_cumflo_{r,p,c}} (VAR_CUMFLO_{r,p,c,yl,y2} \times UC_CUMFLO_{uc_n,r,p,c,yl,y2}) + \sum_{re_cumflo_{r,p,c}} (VAR_CUMFLO_{r,p'ACT',yl,y2} \times UC_CUMACT_{uc_n,r,p,yl,y2})$$

6.4.2.1 Equation: EQ(1) UC / EQE_UC

Indices: user constraint (uc_n)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET, VAR_CUMCOM, VAR_CUMFLO

Purpose: The user constraint $EQ(l)_UC$ is a user constraint, which is summing over specified regions (uc_r_sum), periods (uc_t_sum) and timeslices (uc_ts_sum).

Equation:

=

$$EQ(l)_UC_{uc_n} \ni UC_RHS_{uc_n,bd} \land uc_ts_sum_{r,uc_n,s} \land uc_r_sum_{r,uc_n} \land uc_t_sum_{r,uc_n,t}$$

$$\sum_{r \in \mathbf{u} \in \underline{r}_{s} \text{sum}} \sum_{t \in \mathbf{u} \in \underline{t}_{s} \text{sum}} \sum_{s \in \mathbf{u} \in \underline{b}_{s} \text{sum}} \begin{pmatrix} ACT_{r,t,s,'LHS'} + FLO_{r,t,s,'LHS'} + IRE_{r,t,s,'LHS'} \\ + COMNET_{r,t,s,'LHS'} + COMPRD_{r,t,s,'LHS'} \end{pmatrix} + \sum_{r \in \underline{u} \in \underline{t}_{s} \text{sum}} \left(\sum_{t \in \underline{u} \in \underline{t}_{s} \text{sum}} \left(NCAP_{r,t,'LHS'} + CAP_{r,t,'LHS'} \right) + \left(CUMCOM_{r} + CUMFLO_{r} \right) \right)$$

when control parameter VAR UC is set to NO by the user or is missing:

$$\begin{split} &\{\leq;=;\geq\} \\ &UC_RHS_{uc_n!} + \sum_{t \in \texttt{uc_n}} \sum_{\textit{m}} UC_TIME_{uc_n:r!} \times D_t \end{split}$$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the LHS is set equal to the variable VAR_UC. The bounds UC_RHS are then applied to the variable VAR_UC.

$$VAR_UC_{uc_n} + \sum_{t \in uc_t_sum} \sum_{r \in uc_r_sum} UC_TIME_{uc_n,r,t} \times D_t$$

with $VAR_UC.LO_{uc_n} = UC_RHS_{uc_n,'LC}$ $VAR_UC.UP_{uc_n} = UC_RHS_{uc_n,UP}$ $VAR_UC.FX_{uc_n} = UC_RHS_{uc_n,'FX'}$

6.4.2.2 Equation: EQ(1)_UCR / EQE_UCR

Indices: user constraint (uc_n), region (r)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET, VAR_CUMCOM, VAR_CUMFLO

Purpose: The user constraint EQ(*l*)_UCR is a user constraint, which is created for each region of **uc_r_each** and is summing over periods (**uc_t_sum**) and timeslices (**uc_ts_sum**).

Equation:

=

$$\begin{split} EQ(l)_UCR_{r,uc_n} \ni UC_RHSR_{r,uc_n,bd} \wedge uc_ts_sum_{r,uc_n,s} \wedge uc_r_each_{r,uc_n} \\ & \wedge uc_t_sum_{r,uc_n,t} \\ \\ \sum_{l \in uc_t_sum} \sum_{s \in uc_t_sum} \begin{pmatrix} ACT_{r,l,s,'LHS'} + FLO_{r,l,s,'LHS'} + IRE_{r,l,s,'LHS'} \\ + COMNET_{r,l,s,'LHS'} + COMPRD_{r,l,s,'LHS'} \end{pmatrix} \\ + \\ \\ \sum_{l \in uc_t_sum} (NCAP_{r,l,'LHS'} + CAP_{r,l,'LHS'}) + (CUMCOM_r + CUMFLO_r) \end{split}$$

en control parameter ...__ $\{\leq;=;\geq\}$ $UC_RHSR_{r,uc_n,j} + \sum_{t \in uc_Lsum} UC_TIME_{uc_n,r,t} \times D_t$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the LHS is set equal to the variable VAR_UCR. The bounds UC_RHSR are then applied to the variable VAR_UCR.

$$VAR_UCR_{r,uc_n} + \sum_{t \in uc_t_sum} UC_TIME_{uc_n,r,t} \times D_t$$

with $VAR_UCRLO_{r,\muc_n} = UC_RHSR_{r,\muc_n,'LC'}$ $VAR_UCRUP_{r,\muc_n} = UC_RHSR_{r,\muc_n,'UP'}$ $VAR_UCRFX_{r,\muc_n} = UC_RHSR_{r,\muc_n,'EC'}$

6.4.2.3 Equation: EQ(1) UCT / EQE_UCT

Indices: user constraint (uc_n), period (t)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET, VAR_CUMCOM, VAR_CUMFLO

Purpose: The user constraint **EQ(***l***)_UCT** is a user constraint, which is created for each period of **uc_t_each** and is summing over regions (**uc_r_sum**) and timeslices (**uc_ts_sum**).

Equation:

$$EQ(l)_UCT_{uc_n,t} \ni UC_RHST_{uc_n,t,bd} \land uc_ts_sum_{r,uc_n,s} \land uc_r_sum_{r,uc_n}$$

$$\land uc_t_each_{r,uc_n,t}$$

$$\sum_{r \in uc_r_sum} \sum_{s \in uc_(ts_sum} \begin{pmatrix} ACT_{r,t,s,'LHS'} + FLO_{r,t,s,'LHS'} + IRE_{r,t,s,'LHS'} \\ + COMNET_{r,t,s,'LHS'} + COMPRD_{r,t,s,'LHS'} \end{pmatrix} + \sum_{r \in uc_r_sum} \begin{pmatrix} NCAP_{r,t,'LHS'} + CAP_{r,t,'LHS'} \end{pmatrix}$$

when control parameter VAR_UC=NO:

$$\{\leq;=;\geq\}$$

$$UC_RHST_{uc_nf,i} + \sum_{r \in \mathbf{nc_sum}} UC_TIME_{uc_n,r,i} \times D_t$$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the LHS is set equal to the variable VAR_UCT. The bounds UC_RHST are then applied to the variable VAR_UCT.

$$VAR_UCT_{uc_nt} + \sum_{r \in uc_rsum} UC_TIME_{uc_n,r,t} \times D_t$$

with

$$\begin{split} &VAR_UCT.LO_{uc_nt}=UC_RHST_{uc_nt,!LO'}\\ &VAR_UCT.UP_{uc_nt}=UC_RHST_{uc_nt,!DP'}\\ &VAR_UCT.FX_{uc_nt}=UC_RHST_{uc_nt,!EK'} \end{split}$$
6.4.2.4 Equation: EQ(*l*)_UCRT / EQE_UCRT

Indices: user constraint (uc_n), region (r), period (t)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET, VAR_CUMCOM, VAR_CUMFLO

Purpose: The user constraint EQ(t)_UCRT is a user constraint, which is created for each region of uc_r_each and each period of uc_t_each and is summing over timeslices (uc_ts_sum).

Equation:

$$\begin{split} EQ(l)_UCRT_{r,uc_n,t} \ni UC_RHSRT_{r,uc_n,t,bd} \wedge uc_ts_sum_{r,uc_n,s} \wedge uc_r_each_{r,uc_n} \\ & \wedge uc_t_each_{r,uc_n,t} \\ \\ \sum_{s \in uc_6_sum} \begin{pmatrix} ACT_{r,t,s,lHS^{s}} + FLO_{r,t,s,'LHS^{s}} + IRE_{r,t,s,'LHS^{s}} \\ + COMNET_{r,t,s,'LHS^{s}} + COMPRD_{r,t,s,'LHS^{s}} \end{pmatrix} \\ \\ + \\ (NCAP_{r,t,'LHS^{s}} + CAP_{r,t,'LHS^{s}}) \\ \end{split}$$
when control parameter VAR_UC=NO: $\{ \leq ; = ; \geq \}$

 $UC_RHSRT_{r,uc} + UC_TIME_{uc} + D_t$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the LHS is set equal to the variable VAR_UCRT. The bounds UC_RHSRT are then applied to the variable VAR_UCRT.

 $VAR_UCRT_{r\mu c_n,t} + UC_TIME_{uc_n,r,t} \times D_t$

with $VAR_UCRT.LO_{r,\mu c_n,t} = UC_RHSRT_{r,\mu c_n,t,LO}$ $VAR_UCRT.UP_{r,\mu c_n,t} = UC_RHSRT_{r,\mu c_n,t,UP}$ $VAR_UCRT.FX_{r,\mu c_n,t} = UC_RHSRT_{r,\mu c_n,t,FE'}$

6.4.2.5 Equation: EQ(1) UCRTS / EQE_UCRTS

Indices: user constraint (uc_n), region (r), period (t), timeslice (s)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET, VAR_CUMCOM, VAR_CUMFLO

Purpose: The user constraint EQ(l)_UCRTS is a user constraint, which is created for each region of **uc_r_each**, each period of **uc_t_each** and each timeslice of **uc_ts_each**.

Equation:

$$EQ(l)_UCRTS_{r,uc_n,t,s} \ni UC_RHSRTS_{r,uc_n,t,s,bd} \land uc_ts_each_{r,uc_n,s} \land uc_r_each_{r,uc_n} \land uc_t_each_{r,uc_n,t}$$

$$\begin{pmatrix} ACT_{r,t,s;LHS'} + FLO_{r,t,s;LHS'} + IRE_{r,t,s;LHS'} \\ + COMNET_{r,t,s;LHS'} + COMPRD_{r,t,s;LHS'} \end{pmatrix}$$

+
$$(NCAP_{r,t;LHS'} + CAP_{r,t;LHS'})$$

when control parameter VAR_UC=NO:

{≤;=;≥}

 $UC_R\!H\!S\!RT\!S_{r, \mu c_n, t, \sharp, \sharp} + UC_TI\!M\!E_{\mu c_n, r, \sharp} \times D_t$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the LHS is set equal to the variable VAR_UCRTS. The bounds UC_RHSRTS are then applied to the variable VAR_UCRTS.

$$\begin{split} &VAR_UCRTS_{r,\mu c_n,t,\sigma} + UC_TIME_{\mu c_n,r,t} \times D_t \\ &\text{with} \\ &VAR_UCRTS.LO_{r,\mu c_n,t,\sigma} = UC_RHSRTS_{r,\mu c_n,t,\sigma,t,LO} \end{split}$$

$$\begin{split} &V\!AR_UCRTS.UP_{r,\mu c_n,t,\sigma} = UC_RHSRTS_{r,\mu c_n,t,\sigma,UP'} \\ &V\!AR_UCRTS.FX_{r,\mu c_n,t,\sigma} = UC_RHSRTS_{r,\mu c_n,t,\sigma,TEX'} \end{split}$$

6.4.2.6 Equation: EQ(l)_UCTS / EQE_UCTS

Indices: user constraint (uc_n), period (t), timeslice (s)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET, VAR_CUMCOM, VAR_CUMFLO

Purpose: The user constraint **EQ(l)_UCTS** is a user constraint, which is created for each period of **uc_t_each** and each timeslice of **uc_ts_each** and is summing over regions (**uc_r_sum**).

Equation:

$$EQE_UCTS_{uc_n,t,s} \ni UC_RHSRTS_{uc_n,t,s,bd} \land uc_ts_each_{r,uc_n,s} \land uc_r_sum_{r,uc_n}$$

 \wedge uc_t_each_{r,uc_n,t}

$$\sum_{r \in uc, r_s sum} \left(ACT_{r,t,s,',LHS'} + FLO_{r,t,s,',LHS'} + IRE_{r,t,s,',LHS'} + COMNET_{r,t,s,',LHS'} + COMPRD_{r,t,s,',LHS'} \right) + \sum_{r \in uc, r_s sum} \left(NCAP_{r,t,',LHS'} + CAP_{r,t,',LHS'} \right)$$

when control parameter VAR_UC=NO: $\{\leq : = : \geq\}$ $UC_RHSTS_{uc_n,t,s,i} + \sum_{r \in uc_r, sum} UC_TIME_{uc_n,r,t} \times D_t$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the LHS is set equal to the variable VAR_UCTS. The bounds UC_RHSTS are then applied to the variable VAR_UCTS.

$$= VAR_UCTS_{uc_n,t,s} + \sum_{r \in uc_r_sum} UC_TIME_{uc_n,r,t} \times D_t$$

with

$$\begin{split} &VAR_UCTS.LO_{uc_n,t,s} = UC_RHSTS_{uc_n,t,s,t/LO'} \\ &VAR_UCTS.UP_{uc_n,t,s} = UC_RHSTS_{uc_n,t,s,t/UP'} \\ &VAR_UCTS.FX_{uc_n,t,s} = UC_RHSTS_{uc_n,t,s,t/EV'} \end{split}$$

6.4.3 Timeslice-dynamic user constraints

Mathematical formulation of timeslice-dynamic user constraints

In the mathematical description of the different variants of timeslice-dynamic user constraints, on the LHS the same placeholders can be used as for pure LHS user constraints: $ACT_{r,t,p,s,THS'}$, $FLO_{r,t,p,s,THS'}$, $IRE_{r,t,p,s,THS'}$, $COMPRD_{r,t,s,THS'}$, $COMNET_{r,t,s,THS'}$, $NCAP_{r,t,p,s,THS'}$, $CAP_{r,t,p,s,THS'}$. The LHS placeholder expressions are identical to those of LHS user constraints.

On the RHS (preceding timeslice ds = s-RS_STG(r,s)), the following placeholders can be used: $ACT_{r,t,p,ds, \mathcal{R}HS'}$, $FLO_{r,t,p,ds, \mathcal{R}HS'}$, $IRE_{r,t,p,ds, \mathcal{R}HS'}$, $COMPRD_{r,t,ds, \mathcal{R}HS'}$, $COMNET_{r,t,ds, \mathcal{R}HS'}$, $NCAP_{r,t,p,ds, \mathcal{R}HS'}$, $CAP_{r,t,p,ds, \mathcal{R}HS'}$. The RHS placeholder expressions can be written by replacing in them the timeslice index s by d(s).

Note that the timeslice-specific terms in the equation are all divided by $G_YRFR_{r,s}$ in order to make it easy to combine flow and capacity terms in the constraint.

6.4.3.1 Equation: EQ(*l*)_UCRS / EQE_UCRS

Indices: user constraint (uc_n), period (t), timeslice (s)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The timeslice-dynamic constraint $EQ(l)_UCRS$ establishes a constraint between two successive timeslices s and $d(s) = s-RS_STG(r,s)$. The constraint is generated for all regions r in the set uc_r_each, all periods t in the set uc_t_each, and for each timeslice on the timeslice level defined by uc_tsl, such that also have the corresponding $UC_RHSRT_{r,uc_n,t,s,bd}$ specified.

Equation:

$$EQ(l)_UCRS_{r,\mu c_n,t,tsl,s} \ni UC_RHSRTS_{r,\mu c_n,t,s,bd} \land uc_r_each_{r,uc_n} \land uc_t_each_{r,uc_n,t} \land (s \in \{ts \mid ts_grp_{r,tsl,ts} \land \bigcup_{slde} _tsl_{r,\mu c_n,slde,tsl}\})$$

$$\begin{pmatrix} ACT_{r,t,s,'LHS'} + FLO_{r,t,s,'LHS'} + IRE_{r,t,s,'LHS'} \\ + COMNET_{r,t,s,'LHS'} + COMPRD_{r,t,s,'LHS'} \end{pmatrix} \times \frac{1}{G_YRFR_{r,s}} + \begin{pmatrix} NCAP_{r,t,tHS'} + CAP_{r,t,tHS'} \end{pmatrix}$$

When control parameter VAR_UC=NO: {≤;=;≥}

$$\begin{array}{l} UC_RHSRTS_{r,\mu c_nt,sJ} + UC_TIME_{\mu c_n,r,t} \times D_{t} \\ + \\ \begin{pmatrix} (ACT_{rt,d,(s)'RHS'} + FLO_{rt,d,(s)'RHS'} + IRE_{rt,d,(s)'RHS'}) \\ + COMNET_{r,t,d,(s)'RHS'} + COMPRD_{r,t,d,(s)'RHS'} \end{pmatrix} \times \frac{1}{G_YRFR_{r,d,(s)}} \\ + \\ \begin{pmatrix} NCAP_{r,t,RHS'} + CAP_{r,t,RHS'} \end{pmatrix} \end{pmatrix} if \mathbf{uc_tsl}_{r,\mu c_n',RHS',s} \end{array}$$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHSRTS is replaced by the variable VAR_UCRTS. The bounds UC_RHSRTS are then applied to the variable VAR_UCRTS.

$$= VAR_UCRTS_{r,\mu c_n,l,s} + UC_TIME_{\mu c_n,r,l} \times D_{t}$$

$$+ \left(\begin{pmatrix} ACT_{r,l,d(s), RHS'} + FLO_{r,l,d(s), RHS'} + IRE_{r,l,d(s), RHS'} \\ + COMNET_{r,l,d(s), RHS'} + COMPRD_{r,l,d(s), RHS'} \end{pmatrix} \times \frac{1}{G_YRFR_{r,d(s)}} \right)$$

$$+ \left(NCAP_{r,l, RHS'} + CAP_{r,l, RHS'} \right)$$

$$with VAR_UCRTSLO_{r,\mu c_n,l,s} = UC_RHSRTS_{r,\mu c_n,l,s,'/LO'}$$

$$\begin{split} &VAR_UCRTSLO_{r,\mu c_n,l,s} = UC_RHSRTS_{r,\mu c_n,l,s,'LO'} \\ &VAR_UCRTSUP_{r,\mu c_n,l,s} = UC_RHSRTS_{r,\mu c_n,l,s,'UP'} \\ &VAR_UCRTS.FX_{r,\mu c_n,l,s} = UC_RHSRTS_{r,\mu c_n,l,s,'EX'} \end{split}$$

6.4.4 Dynamic user constraints of type (t,t+1)

Mathematical formulation of dynamic user constraints and growth constraints of type (t,t+1)

In the mathematical description of the dynamic user constraints and growth constraints of type (**t**, **t**+1), the following placeholders are used for variable terms on the LHS (period **t**): $ACT_GROW_{r,t,p,s,'LHS'}$, $FLO_GROW_{r,t,p,s,'LHS'}$, $IRE_GROW_{r,t,p,s,'LHS'}$, $COMPRD_GROW_{r,t,p,'LHS'}$, $COMNET_GROW_{r,t,p,s,'LHS'}$, $NCAP_GROW_{r,t,p,'LHS'}$, $CAP_GROW_{r,t,p,'LHS'}$, $IRE_GROW_{r,t,p,'LHS'}$, $CAP_GROW_{r,t,p,'LHS'}$, $FLO_GROW_{r,t,q,s,'LHS'}$, $IRE_GROW_{r,t,p,'LHS'}$, $CAP_GROW_{r,t,p,'LHS'}$, $FLO_GROW_{r,t+1,p,s,'RHS'}$, $IRE_GROW_{r,t+1,p,s,'RHS'}$, $COMPRD_GROW_{r,t+1,p,s,'RHS'}$, $FLO_GROW_{r,t+1,p,s,'RHS'}$, $NCAP_GROW_{r,t+1,p,'RHS'}$, $COMPRD_GROW_{r,t+1,p,'RHS'}$, $COMNET_GROW_{r,t+1,s,'RHS'}$, $NCAP_GROW_{r,t+1,p,'RHS'}$, $CAP_GROW_{r,t+1,p,'RHS'}$, $CAP_GROW_{r,t+1,p,'RHS'}$.

The expressions for the variable terms on the LHS can be written as follows:

$$ACT_GROW_{r,t,s,tHS'}$$
=
$$\sum_{(p,v)\in rtp_vintyr_{r,v,p}} \sum_{ts\in prc_ts_{r,p,s}} \left(VAR_ACT_{r,v,t,p,ts} \times UC_ACT_{uc_n,'LHS',r,t,p,ts} \times \left(RS_FR_{r,s,ts} \right) \times \left(\sum_{curerdcur_{r,cur}} OBJ_ACOST_{r,t,p,cur} \right) if UC_ATTR_{r,uc_n,'LHS',ACT',COST'} is given \times \left(UC_ACT_{uc_n,'LHS',r,t,p,ts} \right)^{M(t+1)-M(t)-1} if UC_ATTR_{r,uc_n,'LHS',ACT','GROWTH'} is given X_{ts}$$

 $FLO_GROW_{r,t,s,'LHS'}$

Ξ

$$\sum_{(p,e,t) \in \operatorname{reps}_{y \to t_{i}, p,e,t,t}} \sum_{\forall UC = FLO_{uc_n, RHS^{i}y, t, p,e,t,t}} \left\{ VAR_FLO_{r,y,t,p,e,t,t} \times UC_FLO_{uc_n, RHS^{i}y, t, p,e,t,t}} \times \left(RTCS_TSFR_{r,t,e,s,t,t} \right) \right\}$$

$$\left\{ \begin{bmatrix} OBJ_FCOST_{r,t,p,e,t,s,c,ur} & \text{if } UC_ATTR_{r,uc_n, 1LHS^{i}yFLO^{i}COST} & \text{is given} \\ + \\ OBJ_FDELV_{r,t,p,e,t,s,c,ur} & \text{if } UC_ATTR_{r,uc_n, 1LHS^{i}yFLO^{i}COST} & \text{is given} \\ + \\ OBJ_FSUB_{r,t,p,e,t,s,c,ur} & \text{if } UC_ATTR_{r,uc_n, 1LHS^{i}yFLO^{i}COST} & \text{is given} \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,c,ur} & \text{if } UC_ATTR_{r,uc_n, 1LHS^{i}yFLO^{i}COST} & \text{is given} \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,c,ur} & \text{if } UC_ATTR_{r,uc_n, 1LHS^{i}yFLO^{i}COST} & \text{is given} \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,c,ur} & \text{if } UC_ATTR_{r,uc_n, 1LHS^{i}yFLO^{i}COSOT} & \text{is given} \\ \end{bmatrix} \right\}$$

$$\left[UC_FLO_{uc_n, 1LHS^{i}y, t, p,e,t,s} & \sum_{(p,o,t,s) \in uv} \sum_{i \in uc_n, i, 1LHS^{i}y, r, i, p,e,t,s,c} \\ (UC_FLO_{uc_n, 1LHS^{i}y, t, p,e,t,s,c,ur} & \text{if } UC_ATTR_{r,uc_n, 1LHS^{i}yHE^{i}COSOT} & \text{is given} \\ \end{bmatrix} \right]$$

$$IRE_GROW_{r,t,s, 1LHS^{i}, r, t, p,e,t,s,c,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FDELV_{r,t,p,e,t,s,c,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FDELV_{r,t,p,e,t,s,c,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FSUB_{r,t,p,e,t,s,c,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FSUB_{r,t,p,e,t,s,c,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,c,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,c,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,ur} & \text{if } u_attr_{r,uc_n, 1LHS^{i}yHE^{i}COST} & \\ + \\ OBJ_FTAX_{r,t,p,e,t,s,ur} & \\$$

 $COMPRD_GROW_{r,t,s,',LHS'}$ $= \sum_{(cfz) \in rtcs_var_{c,t,s,u}} \begin{pmatrix} VAR_COMPRD_{r,t,c,t,s} \times UC_COMPRD_{uc_n,',LHS',r,t,c,s} \\ \times \left(RTCS_TSFR_{r,t,c,s,t,s}\right) \times \\ \sum_{(cfz) \in rtcs_var_{c,t,s,u}} (OBJ_COMPD_{r,t,c,t,s,u,c_cout,cur} \quad if uc_attr_{r,uc_n,',LHS','CCMPRD',uc_cost}) \\ \times \left(UC_COMPRD_{uc_n,',LHS',r,t,c,s}\right)^{M(t+1)-M(t)-1} \quad if uc_attr_{r,uc_n,',LHS','COMPRD','GROWTH'} \quad given \end{pmatrix}$

 $COMNET _GROW_{r_{t,s,'LHS'}}$

$$= \sum_{(c,t_{s})\in \mathbf{rics_var}_{\mathsf{c},t_{c},t_{s}}} \left(\begin{array}{c} VAR_COMNET_{r,t,c,t_{s}} \times UC_COMNET_{uc_n,'LHS',r,t,c,s} \\ \times \left(RTCS_TSFR_{r,t,c,s,t_{s}} \right) \times \\ \sum_{\substack{ cure \mathbf{rdeur}_{r,ur} \\ uc_cost}} \left(OBJ_COMNT_{r,t,c,t_{s},uc_cost,cur} \quad if \ \mathbf{uc_attr}_{r,uc_n,'LHS',COMNET',uc_cost} \right) \\ \times \left(UC_COMNET_{uc_n,'LHS',r,t,c,s} \right)^{\mathcal{M}(t+1)-\mathcal{M}(t)-1} \quad if \ \mathbf{uc_attr}_{r,uc_n,'LHS',COMNET',GROWTH'} \ given \right)$$

 $NCAP _ GROW_{r,t,p,'LHS'}$

=

$$\sum_{p} \begin{bmatrix} VAR _NCAP_{r,t,p} \times UC _NCAP_{uc_n,!LHS',r,t,p} \times \\ \sum_{curer d cur_{r,cur}} OBJ _ICOST_{r,t,p,cur} & \text{if } UC _ATTR_{r,uc_n,!LHS',!NCAP,!COST'} \text{ is given} \\ - \\ \sum_{curer d cur_{r,cur}} OBJ _ISUB_{r,t,p,cur} & \text{if } UC _ATTR_{r,uc_n,!LHS',!NCAP,!SUB'} \text{ is given} \\ + \\ \sum_{curer d cur_{r,cur}} OBJ _ITAX_{r,t,p,cur} & \text{if } UC _ATTR_{r,uc_n,!LHS',!NCAP,!SUB'} \text{ is given} \\ + \\ \sum_{curer d cur_{r,cur}} OBJ _ITAX_{r,t,p,cur} & \text{if } UC _ATTR_{r,uc_n,!LHS',!NCAP,!TAX'} \text{ is given} \\ \times (UC_NCAP_{uc_n,'LHS',r,t,p})^{M(l+1)-M(l)-1} & \text{if } UC _ATTR_{r,uc_n,!LHS',!NCAP,!GROWTH'} \text{ given} \end{bmatrix}$$

$$\begin{array}{l} CAP_GROW_{r,t,',LHS'} \\ = \\ & \\ \sum_{p} \begin{pmatrix} VAR_CAP_{rf,p} \times UC_CAP_{uc_n,',LHS',rf,p} \\ \times \\ PRC_ACTFLO_{r,p} \quad if \ UC_ATTR_{r,\muc_n,',LHS','CAP,',CAPACT'}, is \ given \\ \times \\ (UC_CAP_{uc_n,',LHS',rf,p})^{M(t+1)-M(t)-1} \quad if \ UC_ATTR_{r,\muc_n,',LHS','CAP,',GROWTH'}, is \ given \end{pmatrix} \end{array}$$

The expressions for the variable terms on the RHS can be written as follows:



$$\sum_{(p,v) \in r\mathbf{p}_vintyr_{r,v,t+1p}} \sum_{t = epre_4r_{r,p,t}} \sum_{t = epre_4r_{r,p,t}} \left(\sum_{\alpha, \nu = r \neq ue_n, \forall ACT_{r,v,t+1,p,ds}} \times UC_ACT_{ue_n, \forall BES', r,t+1,p,ds} \times (rG_rrR_{r,de_n, \forall BES', ACT', COST'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', r,t+1,p,ds})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_{ue_n, \forall BES', ACT', GROWTH'})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ACT_ACT, GROWTH', UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'})^{M(t)-M(t+1)-1} if UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'} is given \times (UC_ATTR_{r,de_n, \forall BES', ACT', GROWTH'})^{M(t)-M(t+1)-1} is given \times (UC_ATTR_{r,de_n, \exists BES', ACT', GROWTH'})^{M(t)-M(t+1)-1} is given \times (UC_ATTR_{r,de_n, ide_n, i$$

$$FLO_GROW_{r_{1}+1,s,;RHS'}$$
=
$$\sum_{\{p,c,ts\} \in \mathsf{rtpcs}, \mathsf{varf}_{t,s+1,p,c,ts} \neq \mathsf{vertp}_\mathsf{vintyt}_{\mathsf{r},\mathsf{v},\mathsf{s},\mathsf{s},\mathsf{p}}}} \begin{bmatrix} VAR_FLO_{r,y,t+1,p,c,ts} \times UC_FLO_{uc_n,;RHS',r,t+1,p,c,ts} \times \begin{pmatrix} RTCS_TSFR_{r,t,c,s,ts} \end{pmatrix} \\ \times \\ \begin{bmatrix} OBJ_FCOST_{r,t+1,p,c,ts,cur} & if UC_ATTR_{r,\muc_n,;RHS',FLO',COST} & is given \\ + \\ OBJ_FDELV_{r,t+1,p,c,ts,cur} & if UC_ATTR_{r,\muc_n,;RHS',FLO',UBLIV'} & is given \\ + \\ OBJ_FSUB_{r,t+1,p,c,ts,cur} & if UC_ATTR_{r,\muc_n,;RHS',FLO',UBLIV'} & is given \\ + \\ OBJ_FTAX_{r,t+1,p,c,ts,cur} & if UC_ATTR_{r,\muc_n,;RHS',FLO',UBLIV'} & is given \\ + \\ OBJ_FTAX_{r,t+1,p,c,ts,cur} & if UC_ATTR_{r,\muc_n,;RHS',FLO',UBLIV'} & is given \\ \end{bmatrix} \\ \times \\ (UC_FLO_{uc_n,;RHS',r,t+1,p,c,ts})^{M(t)-M(t+1)-1} & if UC_ATTR_{r,\muc_n,;RHS',FLO',GROWTH'} & is given \\ \end{bmatrix}$$



 $COMPRD_GROW_{r_{f}+1,s,'RHS'}$

$$= \sum_{\substack{(c,tz) \in \mathsf{rtcs_varc_{n,t+1,c,tz}}\\ w_{c} = oot}} \left(\begin{array}{c} VAR_COMPRD_{r,t+1,c,tz} \times UC_COMPRD_{uc_n',RHS',r,t+1,c,z} \\ \times \left(RTCS_TSFR_{r,t+1,c,s,tz} \right) \times \\ \sum_{\substack{(c,tz) \in \mathsf{rtcs_varc_{n,t+1,c,tz}}\\ w_{c} = oot} \\ w_{c} = oot} \\ \times \left(UC_COMPRD_{uc_n',RHS',r,t+1,c,z} \right)^{M(t)-M(t+1)-1} & if \ uc_attr_{r,uc_n',RHS',COMPRD',GROWTH'} \ given \end{array} \right)$$

$$COMNET_GROW_{r,t+1,s,'RHS'}$$

$$= \sum_{\substack{(c,ts) \in \mathsf{rtes}_\mathsf{vare}_{\mathsf{c}_{i}t+1,c,s}}} \left(\begin{array}{c} VAR_COMNET_{r,t+1,c,s} \times UC_COMNET_{uc_n,'RHS',r,t+1,c,s} \\ \times \left(RTCS_TSFR_{r,t+1,c,s,ts} \right) \times \\ \times \left(RTCS_TSFR_{r,t+1,c,s,ts} \right) \times \\ \sum_{\substack{curerdour, cur}{uc_cost}} \left(OBJ_COMNT_{r,t+1,c,ts,uc_cost,cur} & if uc_attr_{r,uc_n,'RHS','COMNET',uc_cost} \right) \\ \times \left(UC_COMNET_{uc_n,'RHS',r,t+1,c,s} \right)^{M(t)-M(t+1)-1} & if uc_attr_{r,uc_n,'RHS','COMNET',GROWTH'} given \right)$$

$NCAP _GROW_{r,t+1,'RHS'}$

=

$$\begin{pmatrix} VAR _ NCAP_{r,t+1,p} \times UC _ NCAP_{uc_n, RHS', r,t+1,p} \\ \times \\ \begin{pmatrix} \sum_{currendum_{r,our}} OBJ _ ICOST_{r,t+1,p,our} & if UC _ ATTR_{r,uc_n, RHS', NCAP, COST'} & is given \\ - \\ \sum_{p} & OBJ _ ISUB_{r,t+1,p,our} & if UC _ ATTR_{r,uc_n, RHS', NCAP', SUB'} & is given \\ + \\ & \sum_{currendum_{r,our}} OBJ _ ITAX_{r,t+1,p,our} & if UC _ ATTR_{r,uc_n, RHS', NCAP', TAX'} & is given \\ + \\ & \sum_{currendum_{r,our}} (UC _ NCAP_{uc_n, RHS', r,t+1,p})^{M(t)-M(t+1)-1} & if UC _ ATTR_{r,uc_n, RHS', NCAP', GROWTH'} & is given \\ \end{pmatrix}$$

$$\begin{split} & CAP_GROW_{r,t+1,'RHS'} \\ = \\ & \sum_{p} \begin{pmatrix} VAR_CAP_{r,t+1,p} \times UC_CAP_{uc_n,'RHS',r,t+1,p} \\ \times \\ PRC_CAPACT_{r,p} & if UC_ATTR_{r,uc_n,'RHS','CAP,'CAPACT'}, is given \\ \times \\ & (UC_CAP_{uc_n,'RHS',r,t+1,p})^{M(t)-M(t+1)-1} & if UC_ATTR_{r,uc_n,'RHS','CAP,'GROWTH'}, is given \end{pmatrix} \end{split}$$

6.4.4.1 Equation: EQ(*l*)_UCSU / EQE_UCSU

Indices: user constraint (uc_n), period (t)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The dynamic user constraint or growth constraint of type (t,t+1) EQ(t)_UCSU establishes a constraint between two successive periods t and t+1. For dynamic user constraints the period t is specified by the set uc_t_succ, growth constraints are generated for all periods bur the last. The constraint is summing over regions (uc_r_sum) and timeslices (uc_ts_sum).

Equation:

=

$$EQ(l) _UCSU_{uc_n,t} \ni UC_RHST_{uc_n,t,bd} \land uc_ts_sum_{r,uc_n,s} \land uc_r_sum_{r,uc_n} \land uc_t_succ_{r,uc_n,t}$$

$$\sim uc_t_succ_{r,uc_n,t}$$

$$\sum_{r \in uc_r_sum} \sum_{sum_(b_sum} \left(ACT_GROW_{r,t,s,'LHS'} + FLO_GROW_{r,t,s,'LHS'} + IRE_GROW_{r,t,s,'LHS'} \right) + COMNET_GROW_{r,t,s,'LHS'} + COMPRD_GROW_{r,t,s,'LHS'} \right)$$

$$+ \sum_{r \in uc_r_sum} \left((NCAP_GROW_{r,t,'LHS'} + CAP_GROW_{r,t,s,'LHS'} \right) \\ When control parameter VAR_UC=NO: {\leq :=; \geq } UC_RHST_{uc_n,t,t} + COMPRD_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + CAP_GROW_{r,t+1,s,'RHS'} + CAP_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + CAP_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + CAP_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + CAP_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} + CAP_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} +$$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHST is replaced by the variable VAR_UCT. The bounds UC_RHST are then applied to the variable VAR_UCT.

$$\begin{split} & VAR_UCT_{uc_n,t} \\ + \\ & \sum_{r \in uc_r_sum} \sum_{s \in uc_ts_sum} \begin{pmatrix} ACT_GROW_{r,t+1,s,'RHS'} + FLO_GROW_{r,t+1,s,'RHS'} + IRE_GROW_{r,t+1,s,'RHS'} \\ + COMNET_GROW_{r,t+1,s,'RHS'} + COMPRD_GROW_{r,t+1,s,'RHS'} \end{pmatrix} \\ + \\ & \sum_{r \in uc_r_sum} (NCAP_GROW_{r,t+1,'RHS'} + CAP_GROW_{r,t+1,'RHS'}) \\ & \text{with} \\ & VAR_UCT_LO_{uc_n,t} = UC_RHST_{uc_n,t,'LO'} \\ & VAR_UCT_UP_{uc_n,t} = UC_RHST_{uc_n,t,'UP'} \\ & VAR_UCT_FX_{uc_n,t} = UC_RHST_{uc_n,t,'EX'} \end{split}$$

6.4.4.2 Equation: EQ(1) UCRSU / EQE_UCRSU

Indices: region (r), user constraint (uc_n), period (t)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The dynamic user constraint or growth constraint of type (t,t+1) EQ(t)_UCSU establishes a constraint between two successive periods t and t+1. For dynamic user constraints the period t is specified by the set uc_t_succ, growth constraints are generated for all periods but the last. The constraint is generated for each region of the set uc_r_each and is summing over timeslices (uc_ts_sum).

Equation:

=

 $UC_RHSRT_{r,\muc_n,t,l} + \sum_{s \in uc_ts_sum} \left(ACT_GROW_{r,t+1,s,?RHS'} + FLO_GROW_{r,t+1,s,?RHS'} + IRE_GROW_{r,t+1,s,?RHS'} + COMPRD_GROW_{r,t+1,s,?RHS'} + COMPRD_GROW_{r,t+1,s,?RHS'} + (NCAP_GROW_{r,t+1,?RHS'} + CAP_GROW_{r,t+1,?RHS'}) \right)$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHSRT is replaced by the variable VAR_UCRT. The bounds UC_RHSRT are then applied to the variable VAR_UCRT.

$$\begin{split} & VAR_UCRT_{r,\mu\sigma_n,t} \\ + \\ & \sum_{s \in \mathbf{u} \in [\mathbf{t}_s] s \in \mathbf{u} \in [\mathbf{t}_s] s \in \mathbf{u} \in [\mathbf{t}_s] s \in \mathbf{t}_s]} \begin{pmatrix} ACT_GROW_{r,t+1,s,;RHS'} + FLO_GROW_{r,t+1,s,;RHS'} + IRE_GROW_{r,t+1,s,;RHS'} \\ + COMNET_GROW_{r,t+1,s,;RHS'} + COMPRD_GROW_{r,t+1,s,;RHS'} \end{pmatrix} \\ & + \\ & \left(NCAP_GROW_{r,t+1,;RHS'} + CAP_GROW_{r,t+1,;RHS'}\right) \\ & \text{with} \\ & VAR_UCRT\,LO_{r,\mu\sigma_n,t} = UC_RHSRT_{r,\mu\sigma_n,t,;LO'} \\ & VAR_UCRT\,UP_{r,\mu\sigma_n,t} = UC_RHSRT_{r,\mu\sigma_n,t,;EK'} \\ & VAR_UCRT\,FX_{r,\mu\sigma_n,t} = UC_RHSRT_{r,\mu\sigma_n,t,;EK'} \end{split}$$

6.4.4.3 Equation: EQ(1) UCRSUS / EQE_UCRSUS

Indices: region (r), user constraint (uc_n), period (t), timeslice (s)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The dynamic user constraint or growth constraint of type $(t,t+1) EQ(l)_UCSUS$ establishes a constraint between two successive periods t and t+1. For dynamic user constraints the period t is specified by the set uc_t_succ, growth constraints are generated for all periods but the last. The constraint is generated for each region of the set uc_r_each and each timeslice of the set uc_ts_each.

Equation:

=

 $\textit{EQ}(l)_\textit{UCRSUS}_{\textit{r,uc_n,t,s}} \ni \textit{UC_RHSRTS}_{\textit{r,uc_n,t,s,bd}} \land \textbf{uc_ts_each}_{\textit{r,uc_n,s}} \land \textbf{uc_r_each}_{\textit{r,uc_n}}$

```
 \wedge uc\_t\_succ_{r,uc\_n,t} \\ \begin{pmatrix} ACT\_GROW_{r,t,s,'LHS'} + FLO\_GROW_{r,t,s,'LHS'} + IRE\_GROW_{r,t,s,'LHS'} \\ + COMNET\_GROW_{r,t,s,'LHS'} + COMPRD\_GROW_{r,t,s,'LHS'} \end{pmatrix} \\ + \\ + \\ \end{pmatrix}
```

 $(NCAP _ GROW_{r,t,'LHS'} + CAP _ GROW_{r,t,'LHS'})$

When control parameter VAR_UC=NO:

$$\begin{split} \{\leq:=:\geq\} \\ UC_RHSRTS_{r,uc_n,l,s,l} \\ + \\ \begin{pmatrix} ACT_GROW_{r,l+1,s,;RHS'} + FLO_GROW_{r,l+1,s,;RHS'} + IRE_GROW_{r,l+1,s,;RHS'} \\ + COMNET_GROW_{r,l+1,s,;RHS'} + COMPRD_GROW_{r,l+1,s,;RHS'} \end{pmatrix} \\ + \\ \begin{pmatrix} NCAP_GROW_{r,l+1,;RHS'} + CAP_GROW_{r,l+1,;RHS'} \end{pmatrix} \end{split}$$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHSRTS is replaced by the variable VAR_UCRTS. The bounds UC_RHSRTS are then applied to the variable VAR_UCRTS.

```
\begin{split} &VAR\_UCRTS_{r,\mu\sigma\_n,t,s} \\ + \\ & \left(ACT\_GROW_{r,t+1,s,'RHS'} + FLO\_GROW_{r,t+1,s,'RHS'} + IRE\_GROW_{r,t+1,s,'RHS'} \\ + COMNET\_GROW_{r,t+1,s,'RHS'} + COMPRD\_GROW_{r,t+1,s,'RHS'} \\ + \\ & \left(NCAP\_GROW_{r,t+1,'RHS'} + CAP\_GROW_{r,t+1,'RHS'}\right) \\ & \text{with} \\ & VAR\_UCRTS.LO_{r,\mu\sigma\_n,t,s} = UC\_RHSRTS_{r,\mu\sigma\_n,t,s,'LO'} \\ & VAR\_UCRTS.UP_{r,\mu\sigma\_n,t,s} = UC\_RHSRTS_{r,\mu\sigma\_n,t,s,'UP'} \\ & VAR\_UCRTS.FX_{r,\mu\sigma\_n,t,s} = UC\_RHSRTS_{r,\mu\sigma\_n,t,s,'EFX'} \end{split}
```

6.4.4.4 Equation: EQ(1) UCSUS / EQE_UCSUS

Indices: user constraint (uc_n), period (t), timeslice (s)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The dynamic user constraint or growth constraint of type $(t,t+1) EQ(l)_UCSUS$ establishes a constraint between two successive periods t and t+1. For dynamic user constraints the period t is specified by the set uc_t_succ, growth constraints are generated for all periods but the last. The constraint generated for each timeslice uc_ts_each and is summing over regions (uc_r_sum).

Equation:

=

$$EQ(t) _UCSUS_{uc_n,t,s} \ni UC_RHSTS_{uc_n,t,s,bd} \land uc_ts_each_{r,uc_n,s} \land uc_r_sum_{r,uc_n} \land uc_t_succ_{r,uc_n,t}$$

$$\land uc_t_succ_{r,uc_n,t}$$

$$\sum_{reuc_r_sum} \begin{pmatrix} ACT_GROW_{r,t,s,t,HS'} + FLO_GROW_{r,t,s,t,HS'} + IRE_GROW_{r,t,s,t,HS'} \\ + COMNET_GROW_{r,t,s,t,HS'} + COMPRD_GROW_{r,t,s,t,HS'} \end{pmatrix}$$

$$+ \sum_{reuc_r_sum} (NCAP_GROW_{r,t,t,HS'} + CAP_GROW_{r,t,t,HS'})$$
When control parameter VAR_UC=NO:
$$\{\leq;=;\geq\} \\ UC_RHSTS_{uc_n,t,s,t} \\ + \\ \sum_{reuc_r_sum} (ACT_GROW_{r,t+1,s,t,HS'} + FLO_GROW_{r,t+1,s,t,HS'} + IRE_GROW_{r,t+1,s,t,HS'})$$

$$+ \sum_{reuv_r_sum} (ACT_GROW_{r,t+1,s,t,HS'} + FLO_GROW_{r,t+1,s,t,HS'} + IRE_GROW_{r,t+1,s,t,HS'})$$

$$+ \sum_{reuv_r_sum} (NCAP_GROW_{r,t+1,s,t,HS'} + FLO_GROW_{r,t+1,s,t,HS'} + IRE_GROW_{r,t+1,s,t,HS'})$$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHSTS is replaced by the variable VAR_UCTS. The bounds UC_RHSTS are then applied to the variable VAR_UCTS.

$$\begin{aligned} & VAR _UCTS_{uc_nl,s} \\ + \\ & \sum_{r \in uc_r_sum} \left(ACT _GROW_{r,l+1,s,'RHS'} + FLO _GROW_{r,l+1,s,'RHS'} + IRE _GROW_{r,l+1,s,'RHS'} \\ & + COMNET _GROW_{r,l+1,s,'RHS'} + COMPRD _GROW_{r,l+1,s,'RHS'} \\ & + \\ & \sum_{r \in uc_r_sum} \left(NCAP _GROW_{r,l+1,'RHS'} + CAP _GROW_{r,l+1,'RHS'} \right) \\ & \text{ with } \\ & VAR _UCTS_LO_{uc_nl,s} = UC _RHSTS_{uc_nl,s,'LO'} \\ & VAR _UCTS_UP_{uc_nl,s} = UC _RHSTS_{uc_nl,s,'UP'} \\ & VAR _UCTS_FX_{uc_nl,s} = UC _RHSTS_{uc_nl,s,'EX'} \end{aligned}$$

6.4.5 Dynamic user constraints of type (t-1,t)

Mathematical formulation of dynamic user constraints and growth constraints of type (t-1, t)

In the mathematical description of the dynamic user constraints and growth constraints of type (t-1, t), the following placeholders are used for variable terms on the RHS (period t-1): $ACT_GROW_{r,t-1,p,s,(RHS')}$, $FLO_GROW_{r,t-1,p,s,(RHS')}$, $IRE_GROW_{r,t-1,p,s,(RHS')}$, $COMPRD_GROW_{r,t-1,p,s,(RHS')}$, $COMNET_GROW_{r,t-1,s,(RHS')}$, $NCAP_GROW_{r,t-1,p,(RHS')}$, $CAP_GROW_{r,t-1,p,(RHS')}$.

For the LHS terms (period \mathbf{t}), the placeholders are the same ones as defined for the LHS user constraints.

The expressions for the variable terms on the RHS can be written as follows:

$$ACT _ GROW_{r,t-1,s,'RHS'}$$

=

$$\sum_{\substack{(p,v) \in r\mathfrak{p}_{v} \text{ vintyr}_{r,vt-1,p}}} \sum_{\substack{tz \in pre_S_{rg,ts}}} \left(\begin{array}{c} VAR_ACT_{r,v,t-1,p,tz} \times UC_ACT_{uo_n,'RHS',r,t-1,p,tz} \times \left(RS_FR_{r,s,tz}\right) \\ \times \\ \left(\sum_{\substack{(u,v) \in rdeur_{r,our}}} OBJ_ACOST_{r,t-1,p,our}\right) & if UC_ATTR_{r,uo_n,'RHS','ACT','COST'} & is given \\ \times \\ \left(UC_ACT_{uo_n,'RHS',r,t-1,p,tz}\right)^{M(t)-M(t-1)-1} & if UC_ATTR_{r,uo_n,'RHS','ACT','GROWTH'} & is given \\ \end{array} \right)$$

 $FLO_GROW_{r,t-1,s,'RHS'}$

$$\sum_{\substack{(p,c,t,s) \in rtp \in s_vart_{t,c-1,p,c,t,s} \\ (p,c,t,s) \in rtp \in s_vart_{t,c-1,p,c,t,s} \\ \times UC_FLO_{r,v,t-1,p,c,t,s} \times UC_FLO_{uc_n,'RHS',r,t-1,p,c,t,s} \times \left(RTCS_TSFR_{r,t-1,c,s,t,s}\right) \\ \times \\ \begin{bmatrix} OBJ_FCOST_{r,t-1,p,c,t,s,c,ur} & if UC_ATTR_{r,uc_n,'RHS',FLO',COST} & is given \\ + \\ OBJ_FDELV_{r,t-1,p,c,t,s,c,ur} & if UC_ATTR_{r,uc_n,'RHS',FLO',COST} & is given \\ + \\ OBJ_FSUB_{r,t-1,p,c,t,s,c,ur} & if UC_ATTR_{r,uc_n,'RHS',FLO',SUB'} & is given \\ + \\ OBJ_FTAX_{r,t-1,p,c,t,s,c,ur} & if UC_ATTR_{r,uc_n,'RHS',FLO',SUB'} & is given \\ + \\ OBJ_FTAX_{r,t-1,p,c,t,s,c,ur} & if UC_ATTR_{r,uc_n,'RHS',FLO',TAX'} & is given \\ \end{bmatrix} \\ \times \\ \begin{pmatrix} UC_FLO_{uc_n,'RHS',r,t-1,p,c,t,s} \end{pmatrix}^{M(t)-M(t-1)-1} & if UC_ATTR_{r,uc_n,'RHS',FLO',GROWTH'} & is given \\ \end{bmatrix}$$



$$COMPRD_GROW_{r,t-1,s,;RHS'}$$

$$= \sum_{(c,ts) \in rics_varc_{s,t-1c,s}} \left(\begin{array}{c} VAR_COMPRD_{r,t-1,c,ts} \times UC_COMPRD_{uc_n,;RHS',r,t-1,c,s} \\ \times \left(RTCS_TSFR_{r,t-1,c,s,ts} \right) \times \\ \sum_{currendour_{r,sw}} \left(OBJ_COMPD_{r,t-1,c,ts,uc_cost,cur} & if uc_attr_{r,uc_n,;RHS';COMPRD;uc_cost} \right) \\ \times \left(UC_COMPRD_{uc_n,;RHS',r,t-1,c,s} \right)^{M(t)-M(t-1)-1} & if uc_attr_{r,uc_n,;RHS';COMPRD;URCWTH'} & given \end{array} \right)$$

$$COMNET _GROW_{r,t-1,s,'RHS''}$$

$$= \sum_{(c,ts)\in rtes_var_{v,t-1,c,ts}} \left(\begin{array}{c} VAR_COMNET_{r,t-1,c,ts} \times UC_COMNET_{uc_n,'RHS',r,t-1,c,s} \\ \times \left(RTCS_TSFR_{r,t-1,c,s,ts} \right) \times \\ \sum_{\substack{(c,ts)\in rtes_var_{v,t-1,c,ts}}} \left(OBJ_COMNT_{r,t-1,c,ts,uc_cost,cur} & \text{if uc_attr}_{r,uc_n,'RHS','COMNET',uc_cost} \right) \\ \times \left(UC_COMNET_{uc_n,'RHS',r,t-1,c,s} \right)^{M(t)-M(t-1)-1} & \text{if uc_attr}_{r,uc_n,'RHS','COMNET','OROMET,'OROMET','OROMET,'OROMET','OROMET,'OROME$$

$$NCAP_GROW_{r,t-1,'RHS'}$$

=

$$\begin{pmatrix} VAR _ NCAP_{r,t-1,p} \times UC _ NCAP_{uc_n, RHS'; r,t-1,p} \\ \times \\ \begin{pmatrix} \sum_{curerd cur_{r,uu}} OBJ _ ICOST_{r,t-1,p,cur} & if UC _ ATTR_{r,uc_n, RHS'; NCAP; COST} & is given \\ - \\ \sum_{curerd cur_{r,uu}} OBJ _ ISUB_{r,t-1,p,cur} & if UC _ ATTR_{r,uc_n, RHS'; NCAP; SUB'} & is given \\ + \\ \sum_{curerd cur_{r,uu}} OBJ _ ITAX_{r,t-1,p,cur} & if UC _ ATTR_{r,uc_n, RHS'; NCAP; SUB'} & is given \\ + \\ \begin{pmatrix} UC _ NCAP_{uc_n, RHS'; r,t-1,p} \end{pmatrix}^{M(t)-M(t-1)-1} & if UC _ ATTR_{r,uc_n, RHS'; NCAP; GROWTH'} & is given \\ \end{pmatrix}$$

$$\begin{split} & CAP_GROW_{r,t-1,'RHS'} \\ = \\ & \sum_{p} \begin{pmatrix} VAR_CAP_{r,t-1,p} \times UC_CAP_{uc_n,'RHS',r,t-1,p} \\ \times \\ PRC_CAPACT_{r,p} & if UC_ATTR_{r,uc_n,'RHS','CAP','CAPACT'} & is given \\ \times \\ & (UC_CAP_{uc_n,'RHS',r,t-1,p})^{M(t+1)-M(t)-1} & if UC_ATTR_{r,uc_n,'RHS','CAP','GROWTH'} & is given \end{pmatrix} \end{split}$$

6.4.5.1 Equation: EQ(*l*)_UCSU / EQE_UCSU

Indices: user constraint (uc_n), period (t)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The growth constraint of type (t-1,t) EQ(l)_UCSU establishes a constraint between two successive periods t-1 and t. The growth constraint is generated for all periods t having the *UC_RHST* constant specified. The constraint is summing over regions $(uc_r.sum)$ and timeslices (uc_ts_sum) .

Equation:

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHST is replaced by the variable VAR_UCT. The bounds UC_RHST are then applied to the variable VAR_UCT.

=

$$\begin{split} & VAR_UCT_{uc_n,t} \\ + \\ & \sum_{r \in uc_r_sum} \sum_{s \in uc_t_sum} \left(\begin{matrix} ACT_GROW_{r,t-1,s;RHS'} + FLO_GROW_{r,t-1,s;RHS'} + IRE_GROW_{r,t-1,s;RHS'} \\ + COMNET_GROW_{r,t-1,s;RHS'} + COMPRD_GROW_{r,t-1,s;RHS'} \\ + \\ & \sum_{r \in uc_r_sum} \left(NCAP_GROW_{r,t-1;RHS'} + CAP_GROW_{r,t-1,s;RHS'} \right) \\ & \text{with} \\ & VAR_UCT_LO_{uc_n,t} = UC_RHST_{uc_n,t;LO'} \\ & VAR_UCT_UP_{uc_n,t} = UC_RHST_{uc_n,t;UP'} \\ & VAR_UCT_FFX_{uc_n,t} = UC_RHST_{uc_n,t;UP'} \end{split}$$

6.4.5.2 <u>Equation: EQ(*l*)_UCRSU / EQE_UCRSU</u>

Indices: region (r), user constraint (uc_n), period (t)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The growth constraint of type (t-1,t) EQ(*t*)_UCSU establishes a constraint between two successive periods t-1 and t. The growth constraint is generated for all periods t having the *UC_RHSRT* attribute defined. The constraint is generated for each region of the set uc_r_each and is summing over timeslices (uc_ts_sum).

Equation:

=

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHSRT is replaced by the variable VAR_UCRT. The bounds UC_RHSRT are then applied to the variable VAR_UCRT.

$$\begin{split} & VAR_UCRT_{r,\mu\sigma_n,t} \\ + \\ & \sum_{s \in \mathbf{u} \in [\mathbf{t}_s] s \in \mathbf{u}} \begin{pmatrix} ACT_GROW_{r,t-\mathbf{l},s,r,HE}' + FLO_GROW_{r,t-\mathbf{l},s,r,HE}' + IRE_GROW_{r,t-\mathbf{l},s,r,HES'} \\ + COMNET_GROW_{r,t-\mathbf{l},s,r,HES'} + COMPRD_GROW_{r,t-\mathbf{l},s,r,HES'} \end{pmatrix} \\ & + \\ & (NCAP_GROW_{r,t-\mathbf{l},r,HES'} + CAP_GROW_{r,t-\mathbf{l},r,HES'}) \end{pmatrix} \\ & \text{with} \\ & VAR_UCRT\,LO_{r,\mu\sigma_n,t} = UC_RHSRT_{r,\mu\sigma_n,t,r,LO'} \\ & VAR_UCRT\,UP_{r,\mu\sigma_n,t} = UC_RHSRT_{r,\mu\sigma_n,t,r,HES'} \end{pmatrix} \end{split}$$

6.4.5.3 Equation: EQ(1) UCRSUS / EQE_UCRSUS

Indices: region (r), user constraint (uc_n), period (t), timeslice (s)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The growth constraint of type (t-1,t) EQ(l)_UCSUS establishes a constraint between two successive periods t-1 and t. The growth constraint is generated for all periods t having the *UC_RHSRTS* attribute defined. The constraint is generated for each region of the set uc_r_each and each timeslice of the set uc_ts_each .

Equation:

```
\begin{split} EQ(l)\_UCRSUS_{r,uc\_n,t,s} \ni UC\_RHSRTS_{r,uc\_n,t,s,bd} \land uc\_ts\_each_{r,uc\_n,s} \land uc\_r\_each_{r,uc\_n} \\ \land uc\_t\_succ_{r,uc\_n,t} \\ & \left( ACT\_GROW_{r,t,s,LHS'} + FLO\_GROW_{r,t,s,LHS'} + IRE\_GROW_{r,t,s,LHS'} \\ + COMNET\_GROW_{r,t,s,'LHS'} + COMPRD\_GROW_{r,t,s,'LHS'} \\ + \\ & \left( NCAP\_GROW_{r,t,tHS'} + CAP\_GROW_{r,t,tHS'} \right) \\ \end{split}
When control parameter VAR_UC=NO:

 \{ \leq; =; \geq \} \\ UC\_RHSRTS_{r,uc\_n,t,s,j} \end{split}
```

+ $\begin{pmatrix}
ACT _GROW_{r,t-1,s,'RHS'} + FLO _GROW_{r,t-1,s,'RHS'} + IRE _GROW_{r,t-1,s,'RHS'} \\
+COMNET _GROW_{r,t-1,s,'RHS'} + COMPRD _GROW_{r,t-1,s,'RHS'} \\
+ \\
(NCAP _GROW_{r,t-1,'RHS'} + CAP _GROW_{r,t-1,'RHS'})
\end{pmatrix}$

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHSRTS is replaced by the variable VAR_UCRTS. The bounds UC_RHSRTS are then applied to the variable VAR_UCRTS.

=

```
\begin{split} &VAR\_UCRTS_{r,\mu\sigma\_n,l,s} \\ + \\ & \left(ACT\_GROW_{r,l-1,s,;RHS'} + FLO\_GROW_{r,l-1,s,;RHS'} + IRE\_GROW_{r,l-1,s,;RHS'} \\ & + COMNET\_GROW_{r,l-1,s,;RHS'} + COMPRD\_GROW_{r,l-1,s,;RHS'} \\ & + \\ & \left(NCAP\_GROW_{r,l-1,;RHS'} + CAP\_GROW_{r,l-1,;RHS'}\right) \\ & \text{with} \\ & VAR\_UCRTS\_LO_{r,\mu\sigma\_n,l,s} = UC\_RHSRTS_{r,\mu\sigma\_n,l,s,;UP'} \\ & VAR\_UCRTS\_UP_{r,\mu\sigma\_n,l,s} = UC\_RHSRTS_{r,\mu\sigma\_n,l,s,;UP'} \\ & VAR\_UCRTS\_FX_{r,\mu\sigma\_n,l,s} = UC\_RHSRTS_{r,\mu\sigma\_n,l,s,;UP'} \\ & VAR\_UCRTS\_FX_{r,\mu\sigma\_n,l,s,;UP'} \\ & VAR\_UCRTS\_FX_{r,\mu\sigma\_n,l,s,;UP'}
```

6.4.5.4 Equation: EQ(1) UCSUS / EQE_UCSUS

Indices: user constraint (uc_n), period (t), timeslice (s)

Related variables: VAR_FLO, VAR_IRE, VAR_NCAP, VAR_CAP, VAR_ACT, VAR_COMPRD, VAR_COMNET

Purpose: The growth constraint of type (t-1,t) EQ(l)_UCSUS establishes a constraint between two successive periods t-1 and t. The growth constraint is generated for all periods t having the *UC_RHSTS* attribute defined. The constraint generated for each timeslice **uc_ts_each** and is summing over regions (**uc_r_sum**).

Equation:

=

When control parameter VAR_UC=YES, the user constraint is created as strict equality and the RHS constant UC_RHSTS is replaced by the variable VAR_UCTS. The bounds UC_RHSTS are then applied to the variable VAR_UCTS.

$$\begin{split} & VAR_UCTS_{uc_n,t,s} \\ + \\ & \sum_{r \in uc_r_sum} \begin{pmatrix} ACT_GROW_{r,t-1,s,'RHS'} + FLO_GROW_{r,t-1,s,'RHS'} + IRE_GROW_{r,t-1,s,'RHS'} \\ + COMNET_GROW_{r,t-1,s,'RHS'} + COMPRD_GROW_{r,t-1,s,'RHS'} \\ + \\ & \sum_{r \in uc_r_sum} \begin{pmatrix} NCAP_GROW_{r,t-1,'RHS'} + CAP_GROW_{r,t-1,'RHS'} \end{pmatrix} \\ & \text{ with } \\ & VAR_UCTS_LO_{uc_n,t,s} = UC_RHSTS_{uc_n,t,s,'LO'} \\ & VAR_UCTS_UP_{uc_n,t,s} = UC_RHSTS_{uc_n,t,s,'UP'} \\ & VAR_UCTS_FX_{uc_n,t,s} = UC_RHSTS_{uc_n,t,s,'UP'} \\ & VAR_UCTS_FX_{uc_n,t,s} = UC_RHSTS_{uc_n,t,s,'EX'} \end{split}$$

6.4.6 User constraint modifiers

6.4.6.1 <u>Overview</u>

The user constraint facility in TIMES provides a very powerful tool for specifying a large variety of custom user constraints in a TIMES model. Such constraints can refer to practically any combination of individual variables. Moreover, the constraint definitions can be optionally refined by specifying additional modifier attributes that are applied to specific components (variable terms) of the constraints. The modifier attributes available in the current version are listed in Table 25. The note "DYN only" in the table means that the attribute is valid for dynamic constraints only (the constraint is in those cases automatically defined as dynamic if the attribute is used).

As indicated in Table 25, one can easily specify, for example, that the FLO coefficients of the user constraint should apply to the sum of all annual flows in each period, by using the PERIOD attribute. In addition, as cumulative user constraints (summed over periods) are typically almost always meant to be applied also to the sum of the annual flows/activities in each period, the PERIOD modifier is now by default applied to the FLO, ACT, IRE, COMPRD and COMCON components of all cumulative constraints (this can be overridden by the explicit use of the input set **uc_ts_sum** for the constraint). The specification of various kinds of cumulative constraints is thus possible quite easily.

6.4.6.2 Cost modifiers (COST, TAX, SUB, DELIV)

The cost modifiers are applied to the variable terms by multiplying them with the corresponding cost attribute, or with the sum of multiple cost attributes, if several cost modifiers are specified. The expressions for the $FLO_{r,t,p,s,TLHS}$ term above in section 6.4.2 give a detailed example of how the cost attributes are applied. The cost attributes applied are the following:

- COST: ACT_COST for ACT, FLO_COST for FLO/IRE, NCAP_COST for NCAP, COM_CSTNET for COMNET, and COM_CSTPRD for COMPRD
- TAX: FLO_TAX for FLO/IRE, NCAP_ITAX for NCAP, COM_TAXNET for COMNET, and COM_TAXPRD for COMPRD
- SUB: FLO_SUB for FLO/IRE, NCAP_ISUB for NCAP, COM_SUBNET for COMNET, and COM_SUBPRD for COMPRD
- DELIV: FLO_DELIV for FLO/IRE

6.4.6.3 Annuity modifiers (INVCOST, INVTAX, INVSUB)

The annuity modifiers are applied to the variable terms by summing the VAR_NACP variables over all vintage periods tt \leq t that have an annual investment payment in period t, and multiplying these with the annual cost coefficient. The INVCOST modifier applies the investment cost payments, the INVTAX modifier the tax payments, and the INVSUB modifier the subsidy payments (taken as negative values). By combining several of these modifiers, the payments are summed together.

Attribute	Description	Applicable UC components
COST	Multiple by primary cost attribute (summing together with other cost attributes requested)	NCAP,ACT, FLO,COMPRD, COMCON
ТАХ	Multiple by tax attribute (summing together with other cost attributes requested)	NCAP,FLO
SUB	Multiple by subsidy attribute (summing together with other cost attributes requested)	NCAP,FLO
DELIV	Multiple by delivery cost attribute (summing together with other cost attributes requested)	FLO
INVCOST	Multiply by investment cost annuities; implies CUMSUM	NCAP
INVTAX	Multiply by investment tax annuities; implies CUMSUM	NCAP
INVSUB	Multiply by investment subsidy annuities; implies CUMSUM	NCAP
CAPACT	Multiply by PRC_CAPACT	CAP
CAPFLO	Apply coefficients also to any capacity-related flows	FLO
CUMSUM	Sum over all periods up to current or previous period (DYN only)	All
EFF	Multiply by COM_IE (UC_COMPRD), divide by COM_IE (UC_COMCON)	COMPRD, COMCON
GROWTH	Interpret coefficients as annual change coefficients (DYN only)	All
NET	Apply to <i>net</i> production (UC_COMPRD) or consumption UC_COMCON)	COMPRD, COMCON
NEWFLO	Apply coefficient to the flows of the new vintage only	ACT, FLO, IRE
ONLINE	Apply coefficient to the on-line capacity only (assumed equal to the full capacity if ACT_MINLD has not been defined).	САР
PERIOD	Multiply by period length (all but NCAP) or COEF_RPTI (NCAP)	All but CAP
SYNC	Synchronize LHS and RHS sides to refer to the same period	All (RHS only)
YES	Declares the constraint to be dynamic, of type $(t-1, t)$	All (RHS only)

Table 25. User constraint modifier attributes available in TIMES.

6.4.6.4 CAPACT modifier

The CAPACT modifier is applied only to the CAP terms of user constraints, if such exist. The CAP term for each process is multiplied by the PRC_CAPACT parameter of that process when the modifier is used.

6.4.6.5 CAPFLO modifier

The CAPFLO modifier is applied only to the FLO terms of user constraints, if such exist. When the modifier is used, the FLO term for each process/commodity, which normally includes only the VAR_FLO variables, is augmented by the capacity-related flows of the same commodity, if such exists.

6.4.6.6 CUMSUM modifier

The CUMSUM attribute means that the corresponding variable term for any milestone year **t** consists of the cumulative sum of the variables in all previous periods up to (and including) the year **t**. For example, when combined with the INVCOST attribute, the resulting NCAP variable term represents the cumulative sum of capital cost annuities related to all new capacities installed up to the year **t**, which are paid in year **t**.

6.4.6.7 EFF modifier

The EFF modifier can currently only be applied to COMPRD/COMCON/COMNET, and it causes the variable terms to be either multiplied or divided by the commodity efficiency COM_IE. For COMPRD, the variable terms will refer to the net production after taking into account the commodity efficiency COM_IE. For COMCON, it will refer to the gross consumption before applying the commodity efficiency COM_IE, and with COMNET it will refer to the gross NET production (gross production minus gross consumption).

6.4.6.8 GROWTH modifier

The GROWTH modifier used for a variable term causes the user constraint coefficients specified for that term to be interpreted as annual growth/decay coefficients. For example, a coefficient of 1.1 will be interpreted as a growth coefficient corresponding to a 10% annual growth. The final effective coefficient applied to the variable term will be the growth/decay coefficient raised to the power (M(t)-M(t-1)) or (M(t+1)-M(t)), depending on the dynamic type.

6.4.6.9 <u>NET modifier</u>

The NET modifier can only be applied to COMPRD/COMCON. For COMPRD, it causes the variable terms to represent the net amount of production after consumption is subtracted (i.e. gross production in excess of gross consumption). For COMCON, it represents the net consumption in excess of net production (after applying COM_IE), which is normally a non-positive value. Using COMPRD with the NET modifier will thus result in the same variable term as when using COMNET with the EFF modifier.

6.4.6.10 NEWFLO modifier

The NEWFLO modifier can only be applied to the ACT, FLO and IRE variable terms. The modifier causes the variable terms to be restricted to the flows or activities of newly installed capacity vintages in the commissioning period only, whenever the process is

vintaged. For non-vintaged processes, the modifier has no effect. Therefore, for making consistent use of the modifier, usually all processes referred to in the variable terms should be vintaged when this modifier is used.

6.4.6.11 ONLINE modifier

The ONLINE modifier can only be used with the CAP variable term. It causes the capacity term to be referring to the on-line capacity instead of the full installed capacity. However, in TIMES, the on-line capacity of a process may differ from the full capacity only when start-ups and shut-downs have been modelled, by defining a minimum stable operating level with the parameter ACT MINLD.

6.4.6.12 PERIOD modifier

The PERIOD modifier can be used for all other components (variable terms) except for the CAP term. For the ACT, FLO, IRE, COMPRD, COMCON, and COMNET terms, the modifier adds the multiplier D(t), i.e. they are multiplied by the period length. For the NCAP term, the modifier multiplies the new capacity variable for each period by the number of repeated investments in that period.

6.4.6.13 SYNC modifier

The SYNC modifier can be used on the RHS for any component of a user component to signify that the RHS term should refer to the same period as the LHS term. It will also automatically declare the constraint to be dynamic of type (t-1,t), unless uc_t_succ is defined or the GROWTH modifier is used on the LHS side, which both force it to be of type (t,t+1).

6.4.6.14 YES modifier

The only function of the YES modifier is to declare the user constraint to be dynamic. The constraint will be of type (t-1,t), unless **uc_t_succ** is defined or the GROWTH modifier is used on the LHS side, which forces it to be of type (t,t+1). Using this modifier can thus be useful, if there are no other relevant modifiers to be used on the RHS of the constraint, which would automatically declare the constraint dynamic.

6.4.7 Non-binding user constraints

Non-binding user constraints of any type (intoduced for reporting purposes) can be defined in the same way as binding constraints, but using the 'N' lim type when specifying the UC_RHSxxx constant, with any value defined for it (-1 is recommended). Non-binding user constraints can only be defined when user constraint variables are enabled (i.e. when using the option VAR_UC==YES). The levels of the non-ninding constraints (i.e. the levels of the slack variables) are reported in the PAR_UCSL reporting attribute (see Section 3.3).
Appendix A The TIMES Climate Module

1 Introduction

This Appendix contains the updated (November 2010) documentation on the Climate Module option for the TIMES model. It provides is a streamlined version of the older version, and contains 5 sections: section 2 contains a detailed description of the theoretical approach taken, section 3 describes the parameters of the climate module, section 4 the variables and section 5 the equations. This version of the documentation does not include the complete formulations for all of the Climate equations (in GAMS form), and neither does it include the full GAMS specifications. However, it should be sufficient to gain a complete understanding of the equations in mathematical form, and should enable the user to define the parameters of the climate module.

2 Mathematical formulation

The Climate Module starts from the global emissions of CO_2 , CH_4 , and N_2O , as generated by the TIMES global model, and proceeds to compute successively:

- the changes in CO₂, CH₄, and N₂O concentrations via three separate sets of equations,
- the total change (over pre-industrial times) in atmospheric radiative forcing resulting from the three gases plus an exogenously specified additional forcing resulting from other causes (other anthropogenic and/or natural causes, as defined by the user), and
- the temperature changes (over pre-industrial times) in two reservoirs (surface and deep ocean).

The Climate Equations used to perform these calculations were initially adapted from Nordhaus and Boyer (1999), who proposed linear recursive equations for calculating concentrations and temperature changes based on the CO_2 life cycle. These linear equations give results that are good approximations of those obtained from more complex climate models (Drouet *et al.*, 2004; Nordhaus and Boyer, 1999). The non-linear radiative forcing equation used by these authors and in TIMES is the same as the one used in most models. The choice of the Nordhaus and Boyer's climate equations is motivated by the simplicity of their approach and by the fact that their climate module is well-documented and acceptably accurate. In our implementation, the forcing equation has been replaced by a linear approximation whose values closely approach the exact ones as long as the useful range is carefully selected. This was done in order to keep the entire model linear, and therefore to allow the user to set constraints on forcing and on temperature as well as on concentrations and on emissions.

Rigorously, the concentration and forcing equations used in the climate module are applicable only to CO2 emissions, since the concentration equations simulate the carbon cycle. In order to model other GHGs, one way is to use these same equations, while replacing CO2 emissions by CO2-equivalent emissions of any number of gases endogenous to the model. However, a more detailed and generally preferable approach is to model separately the life cycle of each endogenous emission separately, and this is the approach used in TIMES. The additional forcing due to the remaining (non endogenous) emissions, is accounted for via an exogenous forcing quantity directly defined by the user.

We now describe the mathematical equations used at each of the three steps of the climate module.

2.1 Concentrations (accumulation of CO2, CH4, N2O)

a) CO2 accumulation is represented as the linear three-reservoir model below⁴⁵: the atmosphere, the quickly mixing upper ocean + biosphere, and the deep ocean. CO2 flows in both directions between adjacent reservoirs. The 3-reservoir model is represented by the following 3 equations when the step of the recursion is equal to one year:

$$\begin{aligned} M_{atm}(y) &= E(y) + (1 - \varphi_{atm-up}) M_{atm}(y-1) + \varphi_{up-atm} M_{up}(y-1) & (1) \\ M_{up}(y) &= (1 - \varphi_{up-atm} - \varphi_{up-lo}) M_{up}(y-1) + \varphi_{atm-up} M_{atm}(y-1) + \varphi_{lo-up} M_{lo}(y-1) & (2) \\ M_{lo}(y) &= (1 - \varphi_{lo-up}) M_{lo}(y-1) + \varphi_{up-lo} M_{up}(y-1) & (3) \end{aligned}$$

with

- $M_{atm}(y)$, $M_{up}(y)$, $M_{lo}(y)$: masses of CO₂ in atmosphere, in a quickly mixing reservoir representing the upper level of the ocean and the biosphere, and in deep oceans (GtC), respectively, in year y (GtC)
- $E(y-1) = CO_2$ emissions in previous year (GtC)
- φ_{ij} , transport rate from reservoir *i* to reservoir *j* (*i*, *j* = *atm*, *up*, *lo*) from year *y*-*l* to *y*

b) CH4 accumulation is represented by a so-called single-box model in which the atmospheric methane concentration obeys the following equations assuming a constant annual decay rate of the anthropogenic concentrations Φ_{CH4} (whereas the natural concentration is assumed in equilibrium):

$$CH4_{atm}(y) = (1 - \Phi_{CH4}) \cdot CH4_{atm}(y - 1) + EA_{CH4}(y)$$
(1a)

$$CH4_{up}(y) = CH4_{up}(y-1)$$
 (1b)

$$CH4_{tot}(y) = CH4_{abs}(y) + CH4_{us}(y)$$
(1c)

where

 $^{^{45}}$ There exists another well-known representation of CO₂ accumulation equations, using a five-box model.

- $CH4_{atm}$, $CH4_{up}$, and EA_{CH4} are respectively: the atmospheric concentration, the natural concentration⁴⁶ (both expressed in Mt), and the anthropogenic emission of CH₄ (expressed in Mt/yr). EA_{CH4} is generated within the model, but $CH4_{up}$ is fully exogenous (see values for CH4-UP and CH4-ATM in Table A-2). All quantities are indexed by year.
- $d_{CH4} = 2.84$ (the density of *CH4*, expressed in *Mt/ppbv*) is then used to convert concentration in Mt into ppbv.
- $1 \Phi_{CH4}$ is the one-year retention rate of CH₄ in the atmosphere, see Table A-1.

c) N2O accumulation is also represented by a single-box model in which the atmospheric N2O concentration obeys the following equations:

$$N2O_{atm}(y) = (1 - \Phi_{N2O}) \cdot N2O_{atm}(y - 1) + EA_{N2O}(y)$$
(1b)

$$N2O_{up}(y) = N2O_{up}(y-1)$$
(2b)

$$N2O_{tot}(y) = N2O_{atm}(y) + N2O_{up}(y)$$
(2c)

where

- N2O_{atm}, N2O_{up}, and EA_{N2O}, are respectively: the atmospheric concentration, the
 natural concentration (both expressed in Mt), and the anthropogenic emission of
 N₂O (expressed in Mt/yr). EA_{N2O} is generated within the model, but N2O_{up} is
 fully exogenous (see values for N2O-UP and N2O-ATM in Table A-2). All
 quantities are indexed by year,
- d_{N20} = 7.81 (the density of N20, expressed in Mt/ppbv) is then used to convert concentration in Mt to ppbv units.
- $1 \Phi_{N2O}$ is the one-year retention rate of N₂O in the atmosphere, see table A-1.

Note: For both CH_4 and N_2O , the total atmospheric concentrations (UP+ATM) are used in the forcing expressions (see below) and are reported in the results.

2.2 Radiative forcing

We assume, as is routinely done in atmospheric science, that the atmospheric radiative forcing caused by the various gases are additive (IPCC, 2007). Thus:

$$\Delta F(y) = \Delta F_{CO2}(y) + \Delta F_{CH4}(y) + \Delta F_{N2O}(y) + EXOFOR(y)$$
(3)
We now explain these four terms.

a) The relationship between CO2 accumulation and increased radiative forcing, $\Delta F_{CO2}(y)$, is derived from empirical measurements and climate models (IPCC 2007).

⁴⁶ Note that the subscripts *atm* and *up*, which for the CO2 equations referred to the atmosphere and upper reservoirs, have been reused for the CH4 and N2O equations to stand for anthropogenic and natural concentrations.

$$\Delta F_{CO2}(y) = \gamma \times \frac{\ln \left(M_{atm}(y)/M_0\right)}{\ln 2}$$
(4a)

where:

- M₀ (i.e.CO2ATM_PRE_IND) is the pre-industrial (circa 1750) reference atmospheric concentration of CO2 = 596.4 GtC
- γ is the radiative forcing sensitivity to atmospheric CO₂ concentration doubling = 3.7 W/m²
- b) The radiative forcing due to atmospheric CH4 is given by the following expression (IPCC, 2001)

$$\Delta F_{CH4}(y) = 0.036 \cdot \left(\sqrt{CH4_{y}} - \sqrt{CH4_{0}} \right) - \left[f(CH4_{y}, N2O_{0}) - f(CH4_{0}, N2O_{0}) \right]$$
(4b)

 c) The radiative forcing due to atmospheric N2O is given by the following expression (IPCC, 2001)

$$\Delta F_{N2O}(y) = 0.12 \cdot \left(\sqrt{N2O_y} - \sqrt{N2O_0} \right) - \left[f(CH4_0, N2O_y) - f(CH4_0, N2O_0) \right] \quad (4c)$$

where:

$$f(x, y) = 0.47 \cdot \ln \left[1 + 2.01 \cdot 10^{-5} \cdot (xy)^{0.75} + 5.31 \cdot 10^{-15} \cdot x(xy)^{1.52} \right]$$
(4*d*)

Note that the f(x,y) function, which quantifies the cross-effects on forcing of the presence in the atmosphere of both gases (CH4 and N2O), is not quite symmetrical in the two gases. As usual, the 0 subscript indicates the pre-industrial times (1750)

d) EXOFOR(y) is the increase in total radiative forcing at period *t* relative to preindustrial level due to GHGs that are not represented explicitly in the model. Units = W/m². In Nordhaus and Boyer (1999), only emissions of CO₂ were explicitly modeled, and therefore O(y) accounted for all other GHG's. In TIMES, N₂O and CH₄ are fully accounted for, but some other substances are not (e.g. CFC's, aerosols, ozone, etc.). Therefore, our values for *EXOFOR(y)* will differ from those in Nordhaus and Boyer. It is the modeler's responsibility to include in the calculation of *EXOFOR(y)* only the forcings from those gases and other causes that are not modeled. Table A-3 shows a possible trajectory for EXOFOR.

The parameterization of the three forcing equations (4a, 4b, 4c) is not controversial and relies on the results reported by Working Group I in the IPCC. IPCC (2001, Table 6.2, p.358) provides a value of 3.7 for γ , smaller than the one used by Nordhaus and Boyer ($\gamma = 4.1$). We have adopted this lower value of 3.7 W/m² as default in TIMES. Users are free to experiment with other values of the γ parameter. The same reference provides the entire expressions for all three forcing equations.

2.3 Linear approximations

In TIMES, each of the three forcing expressions is replaced by a linear approximation, in order to preserve linearity of the entire model. All three forcing expressions (4a, 4b, 4c) happen to be concave functions. Therefore, two linear approximations are obvious candidates. The first one is an approximation from below, consisting of the chord of the graph between two selected points. The second one has the same slope as the chord and is tangent to the graph, thus approximating the function from above. The final approximation is taken to be the arithmetic average of the two approximations. These linear expressions are easily derived once a range of interest is defined by the user.

As an example, we derive below the linear approximation for the CO2 forcing expression. The other approximations are obtained in a similar manner, and the parameters of the linear approximations are shown in the next section.

Linear approximation for the CO2 forcing expression:

First, an interval of interest for the concentration M must be selected by the user. The interval should be wide enough to accommodate the anticipated values of the concentrations, but not so wide as to make the approximation inaccurate. We denote the interval (M_1, M_2) .

Next, the linear forcing equation is taken as the half sum of two linear expressions, which respectively underestimate and overestimate the exact forcing value. The underestimate consists of the chord of the logarithmic curve, whereas the overestimate consists of the tangent to the logarithmic curve that is parallel to the chord.

By denoting the pre-industrial concentration level as M_0 , the general formulas for the two estimates are as follows:

Overestimate:
$$F_1(M) = \frac{\gamma}{\ln 2} \cdot \left[\ln(\frac{\gamma}{slope \cdot \ln(2) \cdot M_0}) - 1 \right] + slope \cdot M$$
 (5)

Underestimate:
$$F_2(M) = \gamma \cdot \ln(M_1 / M_0) / \ln 2 + slope \cdot (M - M_1)$$
 (6)

Final
approximation:
$$F_3(M) = \frac{F_1(M) + F_2(M)}{2}$$
(7)
$$\ln(M - M) / \ln 2$$

where:
$$slope = \gamma \cdot \frac{\ln(M_2 / M_1) / \ln 2}{(M_2 - M_1)}$$

The linearized forcing expression implemented in TIMES is the average of the two linear estimates.

2.4 Temperature increase

In the TIMES Climate Module as in many other integrated models, climate change is represented by the global mean surface temperature. The idea behind the two-reservoir model is that a higher radiative forcing warms the atmospheric layer, which then quickly warms the upper ocean. In this model, the atmosphere and upper ocean form a single layer, which slowly warms the second layer consisting of the deep ocean.

$$\Delta T_{up}(y) = \Delta T_{up}(y-1) + \sigma_1 \{ F(y) - \lambda \Delta T_{up}(y-1) - \sigma_2 [\Delta T_{up}(y-1) - \Delta T_{low}(y-1)] \}$$
(8)
$$\Delta T_{low}(y) = \Delta T_{low}(y-1) + \sigma_3 [\Delta T_{up}(y-1) - \Delta T_{low}(y-1)]$$
(9)

with

- ΔT_{up} = globally averaged surface temperature increase above pre-industrial level,
- $\Delta T_{low} =$ deep-ocean temperature increase above pre-industrial level,
- $\sigma_1 = 1$ -year speed of adjustment parameter for atmospheric temperature (also known as the *lag* parameter),
- $\sigma_2 = \text{coefficient of heat loss from atmosphere to deep oceans},$
- $\sigma_3 = 1$ -year coefficient of heat gain by deep oceans,
- $\lambda =$ feedback parameter (climatic retroaction). It is customary to write λ as $\lambda = \gamma C_s$, C_s being the *climate sensitivity* parameter, defined as the change in equilibrium atmospheric temperature induced by a doubling of CO₂ concentration.

<u>Remark</u>: in contrast with most other parameters, the value of C_s is highly uncertain, with a possible range of values from 1°C to 10°C. This parameter is therefore a prime candidate for sensitivity analysis, or for treatment by probabilistic methods such as stochastic programming. In Table A-2, a best estimate value of 2.9 °C is shown, as per IPCC (2001, 2007).

In the next section we describe all the input parameters required to define the climate equations and those needed to define climate constraints. With few exceptions (such as the densities of the gases), all parameters are modifiable by the user, should the need arise. We also provide Table A-2 summarizing the default values of the parameters.

3 Switches and Parameters

3.1 Activating the Climate Module

The Climate Module (CLI) extension of TIMES can be activated and employed by using the Parameters and Switches described in this chapter.

Besides the basic input data parameters described in Table A-1, the user also has full control over the CLI component being activated by means of the \$SET CLI YES switch. This switch is provided by the data handling system when the user indicates that the CLI option is to be included:

\$SET CLI YES

3.2 Calibration

The calibration of the Climate Module to historical values is an important aspect of using the module. The mass balance and temperature equations can be calibrated for the first period by using three alternative calibration years B(1)-1, m(1)-1, and m(1). Whenever D(1)=1, the first two alternatives are equal. The default calibrating year is m(1)-1. The alternative calibration years can be activated by using one of the following two settings in the run-file:

\$SET CM_CALIB B	! Calibrate at the end of B(1)-1
\$SET CM CALIB M	! Calibrate at the end of m(1)

3.3 Controlling the years considered beyond EOH

The Climate Equations will be calculated beyond EOH at each of the years for which either a user-defined emission target or a temperature or concentration bound is specified. The years considered thus span between the EOH and the last year for which a CM_MAXC is specified.

In addition, by default any Climate Equations beyond EOH will be calculated only at each year having a year value divisible by 20. This default year resolution can be changed by using the Climate Module constant **'BEOHMOD'**. However, note that the years available in the model extend by default to 2200 only, and therefore one may need to adjust the year-span e.g. to 2300 by using the following switch:

\$SET EOTIME 2300

The reporting years for the climate variables are the same as the calculation years.

3.4 Input parameters

Like all other aspects of TIMES, the user defines the Climate Module components of the energy system by means of input parameters, which are described in this section. Table A-1 below describes the User Input Parameters that are associated with the Climate Module option.

Table A-1. Definition of Climate Module user input parameters.

Input Parameter	Units &	Description		
(Indexes)	Deraults			
(item)	Units: See on the right Defaults: See below	Various Climate Module constants, where item can be: PHI-UP-AT: carbon transfer coefficient UP \rightarrow ATM PHI-AT-UP: carbon transfer coefficient ATM \rightarrow UP PHI-LO-UP: carbon transfer coefficient LO \rightarrow UP PHI-UP-LO: carbon transfer coefficient UP \rightarrow LO GAMMA: radiative forcing sensitivity, in W/m2 CS: temperature sensitivity, in °C LAMBDA: $\lambda = \gamma / C_{a}$ SIGMA1: speed of adjustment, in W-yr/m ² /°C SIGMA2: thermal capacity ratio, in W/m ² /°C SIGMA3: transfer rate upper to deep ocean, in yr ⁻¹ CO2-PREIND: pre-industrial atmosph. CO2, in GtC PHI-CH4: annual decay of atmospheric CH4, fraction PHI-N2O: annual decay of atmospheric N2O, fraction EXT-EOH: activates horizon extension, \geq 0, year BEOHMOD: defines year interval for reporting, years		
CM_HISTORY (y,cm_var)	Units: See on the right Defaults: See below	Historical calibration values at years y, for cm_var: CO2-ATM: atmospheric mass of CO2, in GtC CO2-UP: mass of CO2 in biosphere, in GtC CO2-LO: mass of CO2 in lower ocean, in GtC DELTA-ATM: atmospheric temperature change, in °C DELTA-LO: oceanic temperature change, in °C CH4-ATM: anthropogenic CH4 concentration, in Mt CH4-UP: natural CH4 concentration, in Mt N2O-ATM: anthropogenic N2O concentration, in Mt N2O-ATM: anthropogenic N2O concentration, in Mt		
CM_GHGMAP (r,c,cg)	Global units: CO2: GtC CH4: Mt N2O: Mt	Conversion factors from regional GHG commodities (c) to global emissions (cg) in the Climate Module, where cg= CO2-GtC: global CO2 emissions in GtCCH4-Mt: N2O-Mt:global CH4 emissions in Mt		
CM_EXOFORC (y)	Unit: W/m2	Radiative forcing from exogenous sources (from greenhouse gases not modelled) in year <i>y</i> .		
CM_LINFOR (y,cm_var,lim)	Unit: For CO2: ppm CH4/N2O: W/m2/ppb Default:	Parameters for the linear forcing functions for cm_var: CO2-PPM: lower (LO) and upper (UP) end of the concentration range over which the forcing function for CO ₂ is linearized (in ppm) CH4-PPB: multiplier (N) for the CH ₄ concentration and		

Input Parameter (Indexes)	Units & Defaults	Description		
	none	constant term (FX) of the linear forcing function N2O-PPB: multiplier (N) for the N ₂ O concentration and constant term (FX) of the linear forcing function		
CM_MAXC (y,cm_var)	Default: none	Maximum level of climate indicator cm_var in year y.CO2-GtC:CO2 emissions in GtCCH4-Mt:CH4 emissions in MtN2O-Mt:N2O emissions in MtCO2-ATM:atm. CO2 concentration / pre-industrial ratioCO2-PPM:atm. CO2 concentration in ppmCH4-PPB:atm. CH4 concentration in ppbN2O-PPB:atm. N2O concentration in ppbDELTA-ATM:atmospheric temperature change, in °CFORCING:total radiative forcing, in W/m2		
CM_MAXCO2C (y)	Unit: GtC	Maximum level of CO2 concentration in GtC.		

3.4.1 Mapping of regional emissions to global emissions

Conversion from regional emissions to global emissions must be done by using the CM_GHGMAP(r,c,cg) parameter, in adequate units. The labels for the global emissions cg are 'CO2-GtC', 'CH4-Mt' and 'N2O-Mt'. The parameter IRE_CCVT(r,c,r,cg) can alternatively be also used, if CM_GHGMAP is not available.

Assuming here that the total regional emissions are represented by the commodities TOTCO2, TOTCH4 and TOTN2O, and are measured in kt, as is the case in TIAM models for instance, the mapping and conversion would be the following:

CM_GHGMAP(R,'TOTCH4','CH4-MT') = 1E-3; CM_GHGMAP(R,'TOTN20','N20-MT') = 1E-3; CM_GHGMAP(R,'TOTCO2','CO2-GtC') = 2.727272 E-7

3.4.2 Deterministic input parameters for CO₂

- CM_CONST({PHI_AT_UP, PHI_UP_AT, PHI_UP_LO, PHI_LO_UP}) (also denoted φ_{atm-up} , φ_{up-atm} , etc, in the equations of section 2): annual CO₂ flow coefficients between the three reservoirs (AT=Atmosphere, UP=Upper ocean layer, LO=Deep ocean layer). These are time-independent coefficients. Units: none
- CM_HISTORY(y,{CO2-ATM, CO2-UP, CO2-LO}): Values at the end of the calibration year *y* of the masses of CO₂ in the atmosphere, the upper ocean layer, and the deep ocean layer, respectively. Note that these values are time- indexed so that the model generator can pick up the correct value according to the calibration year chosen by the user. Units: GtC, Mt(CH4), Mt(N2O).
- CM CONST(CO2-PREIND): Pre-industrial atmospheric mass of CO2. Units = GtC

3.4.3 Parameters for the linear CO₂ forcing approximation

CM_LINFOR(datayear,item,lim): lower and upper limit for the concentration of CO2 in atmosphere, used in the approximation of the radiative forcing equation for CO₂ (see section 2.2 above). item may be equal to CO2-ATM (in which case the limit is expressed as a ratio of concentration over pre-industrial concentration), or to CO2-PPM (in which case the limit is expressed in ppm of CO2-equivalent). The index lim is either equal to LO or to UP, depending on whether the lower or the upper limit of the range is being specified. For example, the following specifications may be used to select a range from 375 to 550 ppm for the approximation at year 2020:

- CM_LINFOR('2020', 'CO2-PPM', 'LO') = 375; CM_LINFOR('2020', 'CO2-PPM', 'UP') = 550;
- •

Note that the values of LINFOR are systematically interpolated. The range can also be specified in a time-dependent manner taking into account the gradual increase in the expected range of possible concentration levels over time. That would further improve the accuracy of the linearization. For example, for 2005 the range could be specified to consist of only a single value, because the actual concentration in 2005 is well-known.

3.4.4 Parameters for modeling the concentrations and forcings of other greenhouse gases

Historical base year values of natural (UP) and anthropogenic (ATM) concentrations (in Mt), needed at for the base year of the model (default 2005):

CM HISTORY('2005', 'CH4-UP') = 1988; CM_HISTORY('2005', 'CH4-ATM') = 3067; CM_HISTORY('2005', 'N2O-UP') = 2109; CM HISTORY('2005', 'N2O-ATM') = 390;

In the results the total concentrations (UP+ATM) are reported for both CH₄ and N₂O.

Annual exponential decay of concentrations (PHI-xxx = 1/Life):

CM_CONST('PHI-CH4') = 0.09158; CM CONST('PHI-N2O') = 0.008803;

Here Φ_{CH4} , Φ_{N20} , are the one-year decay rates for methane and N₂O respectively

Parameters for the linear CH₄ and N₂O forcing approximations:

Note that for specifying the linear forcing functions for CH₄ and N₂O, the LO/UP bounds cannot be used, but the slope ('N') and constant ('FX') of the forcing functions must be directly defined by the user. Example:

CM LINFOR('2010','CH4-PPB','N') = 0.000340:

CM_LINFOR('2010', 'CH4-PPB', 'FX') = -0.110; CM_LINFOR('2010', 'N2O-PPB', 'N') = 0.00292; CM_LINFOR('2010', 'N2O-PPB', 'FX') = -0.769;

Parameter for the exogenous radiative forcing from non-modeled gases in each year from initial year: CM EXOFOR(y)

Units: Watts/m².

3.4.5 Parameters for the temperature equations

- CM_CONST(SIGMA1) (also denoted σ_1): speed of adjustment parameter for atmospheric temperature. $1/\sigma_1$ represents the thermal capacity of the atmospheric + upper ocean layer (W-yr/m^{2/o}C). Note however that when SIGMA1 is assumed stochastic, its multiple values are specified via the generic S_CM_CONST parameter described below.
- CM_CONST(SIGMA2) (also denoted σ₂): ratio of the thermal capacity of the deep oceans to the transfer rate from shallow to deep ocean (W/m²/^oC).
- CM_CONST(SIGMA3) (also denoted σ_3): $1/\sigma_3$ is the transfer rate (per year) from the upper level of the ocean to the deep ocean (yr ⁻¹).
- CM_CONST(GAMMA) (also denoted γ): radiative forcing sensitivity to a doubling of the atmospheric CO₂ concentration. Units: Watts/m².
- CM_CONST(CS): C_s, the temperature sensitivity to a doubling of the CO₂ concentration (^oC).
- CM_CONST(LAMBDA) (also denoted λ): a feedback parameter, representing the equilibrium impact of CO₂ concentrations doubling on climate. λ = γ / C_s. Note however that when C_s is assumed stochastic, its multiple values are specified via the generic S_CM_CONST parameter described below. If all three of λ, γ and C_s are specified, the user-specified λ is overridden by the derived value γ / C_s.
- CM_HISTORY(y,{DELTA_ATM, DELTA_LOW}): values at the end of the calibration year y of the temperature changes (wrt to pre-industrial time) in atmosphere and deep layer, respectively. Units: °C

3.4.6 Upper bounds on climate variables

The following parameters are needed if constraints on some climate variables are desired. In TIMES, several climate upper bounds may be specified at any year. These upper bounds are specified via the single generic parameter CM_MAXC(datayear,item), where *datayear* is the year at which the bound applies, and *item* may be any of the following nine choices:

- CO2-ATM: for bounding the *ratio* of GHG concentration to the preindustrial concentration (where the pre-industrial concentration is defined by CO2-PREIND);
- CO2-PPM: for bounding the CO2 concentration expressed in ppm;
- CH4_PPB: for bounding the CH4 concentration expressed in ppbv;
- N2O-PPB: for bounding the N2O concentration expressed in ppbv;

- FORCING: for bounding the total atmospheric radiative forcing expressed in W/ m². (If this bound or the next one on temperature is used, the linearized forcing equation is used rather than the exact forcing equation);
- DELTA-ATM: for bounding the change in global atmospheric temperature over pre-industrial temperature, expressed in °C;
- CO2-GTC: for bounding the global CO2 emissions expressed in GtC;
- CH4-MT: for bounding the global CH4 emissions expressed in Mt;
- N2O-MT: for bounding the global N2O emissions expressed in Mt.

In addition, the user can also bound the $\rm CO_2$ concentration expressed in GtC, by using CM MAXCO2C.

3.4.7 Incorporating climate variables in UC constraints

When using the Climate Module extension, one can also refer to the climate variables in user constraints. The UC attribute for that purpose is the following:

UC_CLI(uc_n, side, reg, y, item)

This parameter can be used to define climate variable coefficients in any period-wise user constraints. The UC_GRPTYPE (to be used in UC_ATTR) for this parameter is 'CLI'. The *item* index can be any of the following climate variables:

- CO2-GTC total global CO2 emissions (or CO2-eq. GHGs)
- CO2-ATM CO2 concentration in the atmosphere
- CO2-UP CO2 concentration in the biosphere/upper ocean
- CO2-LO CO2 concentration in the deep ocean layer
- FORCING radiative forcing
- DELTA-ATM atmospheric temperature
- DELTA-LO deep oceanic temperature

The attribute can be used for defining custom relationships by each region, between any of the climate variables and e.g. process flows, activities or capacities, or total commodity flows. However, if used in a global constraint, one should normally define the UC_CLI attribute only for one region (e.g. GLB).

3.4.8 Random climate parameters (refer to documentation on stochastic TIMES)

If the stochastic programming version of TIMES is used, several climate parameters may be assumed random. These fall into two categories: the upper bounds on climate quantities discussed in the previous section, and the two climate coefficients, Cs and SIGMA1.

Regarding the random upper bounds, their multiple values are specified via the stochastic version of the CM_MAX parameter, namely S_CM_MAX(datayear,item,stage,sow), where in addition to datayear and item already explained, stage refers to the stage of the event tree and sow refers to the state-of-the-world. Note that this single generic parameter will be specified as many times as there are

stages and sow's in the stochastic event tree. If this parameter is specified, the corresponding values of the deterministic parameter CM MAX are superseded.

Regarding the two random coefficients, their multiple values are then declared via the single generic parameter S_CM_CONST(item,stage,sow), where item may be equal to CS or to SIGMA1, stage is the stage number, and sow is the state-of-the-world. Note that this single generic parameter will be specified as many times as there are stages and sow's in the stochastic event tree. If this parameter is specified, the corresponding values of the deterministic parameter (LAMBDA and/or SIGMA1) are superseded.

The reader is referred to Chapter 8 of Part I and the documentation of the stochastic programming version of TIMES for the precise meaning of the **stage** and **sow** concepts.

Remark: in addition to the possible values of the random parameters, the user must specify the probabilities attached to each *sow*. This is also explained in the documentation on stochastic TIMES.

3.4.9 Parameters for extending the Climate Module equations beyond EOH

The main purpose of extending the climate equations beyond EOH is to be able to set climate targets beyond EOH. This is particularly useful for DeltaT targets, because there is a considerable time lag between the decline of emissions and the peak of DeltaT.

The extended climate equations must be explicitly activated by the user. The activation can be done by specifying any non-negative value for the new Climate Module constant **CM_CONST('EXT-EOH')**. Different values of the constant will have the following meaning:

Value	Meaning
-1 (default)	The feature is deactivated.
0	In this case 'EXT-EOH' will be automatically adjusted to $E(M)$, where M is the last model year m for which the end-year E(m) is specified. The adjusted parameter will then have the same meaning as in the case EXT-EOH > 0 below.
>0	The emissions at EOH will remain constant at the endogenous value in EOH=E(T) (where T=last milestone year) until the year MAX(EXT-EOH , EOH), and then develop linearly from that value to the first user-defined emission value in a subsequent year.

The setting **EXT-EOH=0** may be useful for ensuring that any user-defined target values for the emissions will only be taken into account beyond the last *model year*, even in model runs where a truncated model horizon is used. In such case, when **EXT-EOH=0** is used, the emissions are assumed to remain constant between the truncated EOH and the end of the full model horizon.

A positive value **EXT-EOH=y ≤ EOH** means that a linear development of emissions towards the first user-defined value is requested to start immediately at the EOH, regardless of the model horizon being truncated or not. Finally, a positive value **EXT-EOH=y > EOH** can be useful if the user wishes the emissions to remain constant at the EOH value until a predefined year y > EOH, before turning into the linear development towards the first user-defined value.

<u>Warning</u>: If $0 < \text{EXT-EOH} < E(M) = \text{MAX}_m(E(m))$, any user-defined global emission bounds for CO2-GTC, CH4-MT or N2O-MT, which may be inadvertently specified at years between MAX(EXT-EOH, EOH) and E(M), will also be taken into account as target values for the emission trajectories.

The global greenhouse gas emissions that can be considered by the extended climate equations are the three main input emissions to the Climate Module:

- CO2-GTC Global CO2 emissions, expressed in GtC
- CH4-MT Global CH4 emissions, expressed in Mt
- N2O-MT Global N2O emissions, expressed in Mt

The user can specify target emission values for these emissions at any year(s) beyond EOH. For simplicity, the target emission values are specified by using the **CM_MAXC** parameter, which is normally used for specifying upper bounds for the global emissions, as well as for the temperature and concentrations.

Starting from the year B = MAX(EOH, EXT-EOH), the emissions will be assumed to develop linearly from the value at EOH to the first user-specified value beyond B. If no target values are specified, the emissions will be assumed to remain constant at the EOH value. If several successive values are specified, the emissions will develop linearly also between the successive target values.

Bounds on the global atmospheric temperature, forcing or GHG concentrations can be specified at any years beyond the EOH, in the normal way. In addition, exogenous forcing can be specified and is interpolated beyond EOH.

The Climate Equations will be calculated beyond EOH at each of the years for which either a user-defined emission target or a temperature or concentration bound is specified. The years considered thus span between the EOH and the last year for which a CM_MAXC is specified. However, as described above, any emission bounds between EOH and MAX(EOH,EXT-EOH) will be ignored.

In addition, by default the Climate Equations will be calculated also at each year having a year value divisible by 20. This default year resolution can be changed by using the new Climate Module constant **'BEOHMOD'**. Accordingly, if the user wishes the Climate Equations to be calculated at 10 years' intervals (in addition to the CM_MAXC years) she can specify the following parameter:

PARAMETER CM_CONST / BEOHMOD 10 /;

The reporting years for the climate variables are the same as the calculation years.

3.5 Internal parameters

- CM_PPMcm_var: The densities of the greenhouse gases are hard coded in TIMES (via the internal parameter), with the following values:
 - density of $\, CH_4 : 2.84 \, Mt \, / \, ppbv$
 - density of N_2O : $\ 7.81 \ Mt$ / ppbv
 - density of CO_2 : 2.13 Gt / ppm.

- *CM_PHI_{cm_var,t,i,j}*: The transition matrix for climate indicator cm_var between reservoirs i and j and successive years t-1 and t;
- *CM_AA_{cm_var,t,i,j}*: The transition matrix for climate indicator cm_var between reservoirs i and j and between the milestone years of periods t–1 and t;
- *CM_BB_{cm_var,t,ij}*: The transition matrix for climate indicator cm_var from emissions in period t to reservoir contents in the same period;
- *CM_CC_{cm_var,t,i,j}*: The transition matrix for climate indicator cm_var from emissions in period t–1 to reservoir contents in the period t.

3.6 Reporting parameters

There are two reporting parameters, CM_RESULT and CM_MAXC_M, which contain the results on the levels of the climate variables (or reporting quantities) and the dual values of the constraints defined by using CM_MAXC.

CM_RESULT is indexed by year y and result type {e.g. CO2-ATM, CO2-PPM, FORCING, DELTA-ATM, DELTA_LO}. The values represent the quantities at the end of year y. The reporting years y include the milestone years plus any years beyond m(T) that either have some CM_MAXC bound defined or are modulo(BEOHMOD).

- CO2-GtC(y): the total global CO2 emissions at the end of year y.
- CO2-ATM(y): the value of the atmospheric mass of CO2-equivalent at the end of year y, obtained directly from the variable VAR_CLIBOX('CO2-ATM',y).
- CO2-PPM(y): the value of the atmospheric concentration of CO2-equivalent at the end of year **v**.
- FORCING(y): forcing value at end of year y, calculated using the linearized forcing functions as defined by the user.
- FORC+TOT(y): exact forcing value at end of year y, calculated using the logarithmic forcing equation defined in section 2.2 and the CO2-ATM(y) value.
- DELTA_ATM(y): exact atmospheric temperature value at end of year y, calculated using the forcing FORC+TOT(y).
- DELTA_LOW(y): exact lower ocean temperature value at end of year y, calculated using the forcing FORC+TOT(y).

CM_MAXC_M is indexed by year y and constraint type. The values are reported for each of the EQ_CLITOT and EQ_CLIMAX equations. The values represent directly the dual values of these constraints at year y.

3.7 Default values of the climate parameters

Table A-2 shows the default values of all parameters of the Climate Module except exogenous forcing. All defaults may be modified by the user.

- CS and SIGMA1 may be assumed random, in which case the default values are not used. The user must specify their values explicitly using the appropriate parameter names described earlier.
- The parameters highlighted blue are upper bounds on five climate variables (in this example, they are set high enough to be inoperative).
- The three parameters highlighted pink concern the extension of emissions beyond EOH, as described in the separate note on this subject.

Table A-3 shows an example of specification of the EXOFORCING time series.

Table A-2. Parameters of the climatic module (default values)

Attribute	Lim	DataYear	Item	Default value
CM_HISTORY		2005	CO2-ATM	807.27
CM_HISTORY		2005	CO2-UP	793
CM_HISTORY		2005	C02-L0	19217
CM_HISTORY		2005	DELTA-ATM	0.76
CM_HISTORY		2005	DELTA-LO	0.06
CM_HISTORY		2005	CH4-UP	1988
CM_HISTORY		2005	CH4-ATM	3067
CM_HISTORY		2005	N2O-UP	2109
CM_HISTORY		2005	N2O-ATM	390
CM_CONST			GAMMA	3.71
CM_CONST			PHI-UP-AT	0.0453
CM_CONST			PHI-AT-UP	0.0495
CM_CONST			PHI-LO-UP	0.00053
CM_CONST			PHI-UP-LO	0.0146
CM_CONST			LAMBDA	1.41
CM_CONST			CS	2.9
CM_CONST			SIGMA1	0.024
CM_CONST			SIGMA2	0.44
CM_CONST			SIGMA3	0.002
CM_CONST			CO2-PREIND	596.4
CM_CONST			PHI-CH4	0.09158
CM_CONST			PHI-N2O	0.008803
CM_LINFOR	LO	2005	CO2-PPM	375
CM_LINFOR	UP	2005	CO2-PPM	550
CM_LINFOR	Ν	2005	CH4-PPB	0.00034
CM_LINFOR	FX	2005	CH4-PPB	-0.11000

CM_LINFOR	N	2005	N2O-PPB	0.00292
CM_LINFOR	FX	2005	N2O-PPB	-0.76900
CM_MAXC		2005	CO2-PPM	500
CM_MAXC		2005	CO2-ATM	1000
CM_MAXC		2005	FORCING	10
CM_MAXC		2005	DELTA-ATM	10
CM_MAXC		2005	CO2-GTC	50
CM_CONST			EXT-EOH	2150
CM_CONST			BEOHMOD	20
CM_MAXC		2200	CO2-GTC	0

Table A-3. Example of EXOFORCING (from TIAM-WORLD, 2010 version)

Attribute	DataYear	Value
CM_EXOFORC	2005	-0.25376
CM_EXOFORC	2010	-0.20475
CM_EXOFORC	2015	-0.16055
CM_EXOFORC	2020	-0.11689
CM_EXOFORC	2025	-0.10104
CM_EXOFORC	2030	-0.0774
CM_EXOFORC	2035	-0.06398
CM_EXOFORC	2040	-0.03787
CM_EXOFORC	2045	-0.0354
CM_EXOFORC	2050	-0.04528
CM_EXOFORC	2055	-0.06434
CM_EXOFORC	2060	-0.08634
CM_EXOFORC	2065	-0.09485
CM_EXOFORC	2070	-0.09632
CM_EXOFORC	2075	-0.09254
CM_EXOFORC	2080	-0.08929
CM_EXOFORC	2085	-0.08868
CM_EXOFORC	2090	-0.08273
CM_EXOFORC	2095	-0.0796
CM_EXOFORC	2100	-0.07447

4 Variables

The variables that are used in the Climate Module in TIMES are presented in **Table A-4** below. The climate indicators represented in the Climate Module are grouped according to the following internal sets, which are referred to in the GAMS formulation, presented in Section 5:

- **cm_var**: the set of all climate indicators
- cm_tkind: aggregate total indicators (CO2-GtC, CH4-Mt, N2O-Mt, FORCING)
- cm_emis: emission indicators (CO2-GtC, CH4-Mt, N2O-Mt)
- cm_boxmaptisind,cm_var,cm_box: mapping between aggregate indicators tkind, reservoir indicators cm_var, and corresponding box labels (ATM/UP/LO);
- cm_atmaptisind,cm_var: mapping between aggregate indicators *tkind* and the corresponding boundable atmospheric indicators (CO2-PPM / CH4-PPM / N2O-PPM / DELTA_ATM);
- cm_atbox tkind,cm_box: mapping between mapping between aggregate emission indicators tkind and the corresponding reservoirs that comprise the atmospheric concentration part; contains the pairs {(CO2-GtC,ATM), (CH4-Mt,ATM),(CH4-Mt,UP),(N2O-Mt,ATM),(N2O-Mt,UP) }

Table A-4. Model variables specific to the Climate Module.

Variable (Indexes)	Variable Description
VAR_CLITOT (cm_var,y)	Represents the total amount of climate indicator <i>cm_var</i> in year y, where <i>cm_var</i> is one of {CO2-GtC, CH4-Mt, N2O-Mt, FORCING}.
VAR_CLIBOX (cm_var,y)	Represents the amount of reservoir indicator <i>cm_var</i> in a single reservoir/box in year y, where <i>cm_var</i> is one of {CO2-ATM, CO2-UP, CO2-LO, CH4-ATM, CH4-UP, N2O-ATM, N2O-UP, DELTA-ATM, DELTA-LO}.

4.1 VAR_CLITOT(cm_var,y)

Description: The total amount of aggregate climate indicator in year y.

Purpose and Occurrence:	This variable tracks the total amount of an aggregate climate indicator by period. This variable is generated for each main emission type of the Climate Module as well as for the total forcing from all greenhouse gas concentrations.
Units:	GtC (for CO2 emissions), Mt (for CH4 and N2O emissions), or W/m2 (for total radiative forcing).

Bounds: This variable can be directly bounded with the CM_MAXC attribute.

4.2 VAR_CLIBOX(cm_var,y)

- Description: The amount of climate indicator in a reservoir.
- Purpose and This variable tracks the amount of reservoir-specific climate indicator by Occurrence: period. This variable is generated for each of the reservoirs for each of the aggregate indicators: ATM/UP/LO for CO2 emissions, ATM/UP for CH4 and N2O emissions, and ATM/LO for FORCING (connected to the temperature reservoirs).
- Units: GtC (for CO2 emissions), Mt (for CH4 and N2O emissions), or °C (for temperature reservoirs).
- **Bounds:** Only the total atmospheric amounts can be bounded with the CM_MAXC attribute (CO2-ATM, CO2-PPM, CH4-PPB, N2O-PPB, DELTA-ATM).

5 Equations

There are three blocks of definitional equations: the first block of equations calculates the global emissions of GHG (either all in CO_2 eq., or separately for CO_2 , CH_4 and N_2O) as well as the total (linearized) radiative forcing, the next block calculates the concentrations of the greenhouse gases in the reservoirs, and the third block calculates the atmospheric temperature and lower ocean temperature at period t.

In addition, there is a generic block of equations expressing the upper bounding of the five climate quantities discussed in subsection 3.4.6. This generic equation is generated as many times as an upper bound on any climate variable is specified by the user, and is not generated if no upper bound is specified.

We now give the formulations of these constraints.

Reminder: the Climate Module formulation is activated at run time from the data handling system, which in turn set the \$SET CLI YES switch.

General notation:

- D(t): duration of period t, t=1 to T
- B(t): first year in period t, t=1 to T
- m(t): milestone year of period t (approximate middle year of period, defined as $m(t) = B(t) + \lfloor (D(t) 1)/2 \rfloor$
- y: designates a year, while t designates a period (ranging from 1 to T)
- *Y*: designates the calibration year, which can be chosen by the user to be either B(1)-1, m(1)-1, or m(1), see section 3.2 above.

Table A-5. Climate Module specific constraints (all in the GAMS file equ ext.cli).

Constraints (Indexes)	Constraint Description
EQ_CLITOT (cm_var,t)	Defines the amount of global greenhouse gas emissions in each period; defines the amount of total radiative forcing from the greenhouse gas concentrations in each period t .
EQ_CLICONC (cm_var, cm_box,t)	Defines the mass of each greenhouse gas cm_var in each reservoir cm_box at the end of the milestoneyr $m(t)$ of period t .
EQ_CLITEMP (cm_box,t)	Defines the temperature increase in the each reservoir cm_box (the lower atmosphere and the lower ocean layer) over its pre-industrial temperature measured at the end of milestoneyr m(t) of period <i>t</i> .
EQ_CLIMAX (y,cm_var)	Imposes an upper bound on any or all of the climate variables <i>cm_var</i> (<i>CO2-GTC, CH4-MT, N2O-MT, CO2-ATM, CO2-PPM, CH4-PPB, N2O-PPB, FORCING, DELTA-ATM</i>), at any desired year <i>y</i> , according to the user-defined input parameter CM_MAXC.

5.1 EQ_CLITOT(cm_var,t)

Description: Defines the total amount of aggregate climate indicator in period t.

Purpose:	This constraint defines the amount of global greenhouse gas emissions in each period and the amount of total radiative forcing from the greenhouse gas concentrations in each period t . This equation is generated in each time period for all indicators considered.
Units:	Global emission units (GtC, Mt) or forcing units (W/m2)
Туре:	Binding. The equation is an equality (=) constraint.

Interpretation of the results:

Primal: The level of this constraint must be zero in a feasible solution.

Dual variable: The dual variables represent the marginal prices of the global emissions / forcing (when undiscounted).

Remarks:

• For CO2, the linear forcing function parameters $CM_LINFOR_{t,cm_emis,TX'}$ and $CM_LINFOR_{t,cm_emis,W'}$ are automatically calculated by the model generator from any user-defined $CM_LINFOR_{t,cm_emis,'LO'}$ and $CM_LINFOR_{t,cm_emis,'UP'}$.

Equation:

$$EQ_CLITOT_{cm_tkind,t} \forall [(t \in milestonyr)]$$

$$\sum_{\substack{xm_tkind \in cm_emis\\(r,c,s) \in rtcs_varc_{r,cts}}} VAR_COMNET_{r,t,c,s} \times CM_GHGMAP_{r,c,cm_tkind}$$

$$\sum_{\substack{cm_tkind \in cm_emis\\(r,c,s) \in rtcs_varc_{r,cts}}} \binom{CM_LINFOR_{t,cm_emis,'N'} \times }{\begin{pmatrix} \sum_{\substack{xm_ctbox_{cm_emis,cm_tkind}\\(m_atbox_{m_emis,cm_box}\\(m_boxmaBm_emis,cm_varem_box\\CM_LINFOR_{t,cm_emis,'EX'}) \end{pmatrix}} + CM_EXOFORC_{t}$$

$$\{=\}$$

$$VAR_CLITOT_{cm_tkind,t}$$

5.2 EQ_CLICONC(cm_var,cm_box,t)

- Description: Defines the reservoir-specific amounts of concentration indicator in each period.
- **Purpose:** Defines the dynamic relationship between emissions and the concentration in the reservoirs modelled for each greenhouse gas, such that the amount of concentration in reservoir i and period t may depend on the amounts of concentrations in any reservoir k in period t-l, and on the emissions in period t.

Units: Global emission units (GtC, Mt).

Type: Binding. The equation is an equality (=) constraint.

Interpretation of the results:

Primal: The level of this constraint must be zero in a feasible solution. *Dual variable:* The dual variable of this constraint in the solution is of little interest.

Remarks:

- See expressions for the transfer matrices on next page.
- The equations beyond the last milestone year m(T) are similar, but omitted here.

Equation:

cm_boxmapm_emis, cm_var,cm_box2

$$\sum_{\text{cm_boxma} R_{\text{m_emis, cm_var,m_box}}} VAR _CLIBOX_{cm_var,m}$$

 $\begin{array}{l} CM_AA_{cm_emis,t,i,j} = \left\{ A_{ij}(t) \right\} = PHI^{n(t)} (PHI^{0} = I), \text{ where} \\ PHI \text{ is the } 3 \times 3 \text{ matrix} : \begin{bmatrix} (1-PHI_AT_UP) & PHI_UP_AT & 0 \\ PHI_AT_UP & (1-PHI_UP_AT-PHI_UP_LO) & PHI_LO_UP \\ 0 & PHI_UP_LO & (1-PHI_LO_UP) \end{bmatrix} \\ CM_BB_{cm_emis,t,i} = \left\{ BB_{i1}(t) \right\} \text{ is the first column of the matrix} : \\ BB(t) = \sum_{i=0}^{p(t)-1} PHI^{i} & if p(t) \ge 1 \\ BB(t) = 0 & if p(t) = 0 \\ CM_CC_{cm_emis,t,i} = \left\{ CC_{i1}(t) \right\} \text{ is the first column of the matrix} : \\ CC(t) = \sum_{i=p(t)}^{n(t)-1} PHI^{i} & if n(t) \ge p(t) + 1 \\ CC(t) = 0 & if n(t) = p(t) \\ p(t) = \left\lfloor \frac{D(t)+1}{2} \right\rfloor, \quad n(t) = m(t) - m(t-1) & if t \ne 1, \\ p(t) = m(t) - Y, \quad n(t) = p(t) & if t = 1 \\ D(t) \text{ is the number of y earsin period } t, and m(t) \text{ is the middle year of period } t \text{ defined as} \\ m(t) = B(t) + \left\lfloor \frac{D(t)-1}{2} \right\rfloor \\ \lfloor x \rfloor \text{ denotes the largest integer smaller than or equal to } x \end{array}$

EQ_CLITEMP(cm_var,cm_box,t) 5.3

- Description: Defines the reservoir-specific amounts of temperature indicator in each period.
- **Purpose:** Defines the dynamic relationship between forcing and the temperature increase in the reservoirs modelled, such that the amount of temperatures increase in reservoir i and period t may depend on the amounts of temperature increase in any reservoir k in period t-1, and on the radiative forcing in period t.

Units: Global temperature units (°C).

Type: Binding. The equation is an equality (=) constraint.

Interpretation of the results:

The level of this constraint must be zero in a feasible solution. Primal: Dual variable: The dual variable of this constraint in the solution is of little interest.

Remarks:

- See expressions for the transfer matrices on next page.
- The equations for years beyond m(T) are similar, but omitted here.

Equation:

$$EQ_CLITEMP_{cm_box,t} \forall [(t \in milestonyr)]$$

$$\sum_{cm_boxmap_{FORCING; d, cm_var, t-1}} \times CM _ AA_{FORCING; d, cm_boxcm_box2} + CM _ BB_{FORCING; d, cm_box} \times VAR _ CLITOT_{FORCING; d, t-1} + CM _ CC_{FORCING; d, cm_box} \times VAR _ CLITOT_{FORCING; d-1} + \sum_{miyr_1} CM _ CONST_{cm_var} \times CM _ AA_{FORCING; d, cm_box2}$$

PFORCING;cm_var,cm_box2

 $CM _AA_{i_{FORCENCY,t,i,j}} = \left\{A_{ij}(t)\right\} = PHI^{n(t)} (PHI^{0} = I), \text{ where}$ $PHI \text{ is the } 3 \times 3 \text{ matrix} : \begin{bmatrix} (1-SIGMAI \times (IAMBDA + SIGMA2) & SIGMAI \times SIGMA2 & 0 \\ SIGMA3 & (1-SIGMA3) & 0 \\ 0 & 0 & 0 \end{bmatrix}$ $CM _BB_{i_{FORCENCY,t,i}} = \left\{BB_{i1}(t)\right\} \text{ is the first column of the matrix} :$ $BB(t) = SIGMAI \times \sum_{i=0}^{n(t)-1} \frac{n(t)-i}{n(t)} \times PHI^{i}$ $CM _CC_{i_{FORCENCY,t,i}} = \left\{CC_{i1}(t)\right\} \text{ is the first column of the matrix} :$ $CC(t) = SIGMAI \times \sum_{i=0}^{n(t)-1} \frac{i}{n(t)} \times PHI^{i}$ $n(t) = m(t) - m(t-1) \qquad \text{if } t \neq 1,$ $n(t) = m(t) - Y \qquad \text{if } t = 1$ D(t) is the number of years in period t, and m(t) is the middle year of period t defined as $m(t) = B(t) + \left\lfloor \frac{D(t)-1}{2} \right\rfloor$ $\lfloor x \rfloor \text{ denotes the largest integer smaller than or equal to x}$

5.4 EQ_CLIMAX(y,cm_var)

Description: Constraint that sets an upper bound on the climate indicator in a give year.

- **Purpose:** To set an upper bound for a climate indicator variable in any desired year y. The variables that can be bounded are the total global emissions and the total radiative forcing (VAR_CLITOT), the atmospheric concentrations of greenhouse gases (sum of VAR_CLIBOX variables), and the increase in atmospheric temperature (VAR_CLIBOX). The bounds can be specified by using the *CM_MAXCy.cm_var* attribute.
- **Units:** Units of the variable(s) bounded.
- **Type:** Binding. The equation is a less than or equal to inequality (\leq) constraint.

Interpretation of the results:

- *Primal:* The level of this constraint must be less than or equal to zero in a feasible solution.
- *Dual variable:* The dual variable of this constraint in the solution may be used to derive the marginal price of the climate indicator constrained (when undiscounted; global dual values are, ex officio, reported without undiscounting, as no well-defined "global discount factors" exist, only regional ones).

Remarks:

- The CM_MAXC bounds defined on CO2-ATM are automatically converted into equivalent bounds on CO2-PPM.
- The coefficients a_y and β_y in the equations are such that $y = a_y (m(t)-y) + \beta_y (y-m(t-1))$, for all y in the range $m(t-1) < y \le m(t)$.

Equation:

$$EQ_CLIMAX_{y,cm_var} \quad \forall [\{(y,cm_var) | CM_MAXC_{y,cm_var}\}]$$

Case A. For total emissions, up to m(T)

$$\alpha_{y} \times VAR_CLITOT_{cm_emis,t-1} + \beta_{y} \times VAR_CLITOT_{cm_emis,t}$$
$$\leq CM_MAXC_{y,cm_emis}$$

Case B. For atmospheric GHG concentrations, up to m(T)

$$\sum_{\substack{\text{cm}_a\text{tbo}_{xm}_\text{ends},\text{cm}_b\text{ox}\\ \text{cm}_a\text{tbo}_{xm}_\text{ends},\text{cm}_b\text{ox}\\ \text{cm}_b\text{ox}\text{max}_{Rm}_\text{ends},\text{cm}_v\text{argm}_b\text{ox}\\ \leq CM_MAXC_{y,cm}_\text{var}$$

Case C. For total radiative forcing, up to m(T)

$$\begin{aligned} \alpha_{y} \times VAR_CLITOT_{FORC NG', t-1} + \beta_{y} \times VAR_CLITOT_{FORC NG', t} \\ \leq CM_MAXC_{y, FORC NG'} \end{aligned}$$

Case D. For increase in global atmospheric temperature, up to m(T):

$$\sum_{\substack{m_v \neq VAR \ CLIBOX_{cm_var,t-1} + \beta_y \neq VAR \ CLIBOX_{cm_var,t-1} \\ \leq CM \ MAXC_{y,cm_var}}} \beta_{y,cm_var} + \beta_y + MAR \ CLIBOX_{cm_var,t-1} + \beta_y + MAR \ CLIBOX_{cm_var,t-1} + \beta_y + \beta_y + MAR \ CLIBOX_{cm_var,t-1} + \beta_y +$$

Case E. For atmospheric GHG concentrations, beyond m(T):

 $\sum_{cm_a \text{ dbo}_{xm_a \text{ emis, cm_box}}} VAR_CLIBOX_{cm_var,y} \leq CM_MAXC_{y,cm_var}$ cm_boxma_{Rm_amis,cm_var_m_box}

Case F. For total radiative forcing, beyond m(T):

 $VAR _ CLITOT_{FORCING',y} \le CM _MAXC_{y,FORCING'}$

Case G. For increase in global atmospheric temperature, beyond m(T):

$$\sum_{cm_var,y} VAR _CLIBOX_{cm_var,y} \leq CM _MAXC_{y,cm_var}$$
cm_boxma $\mu_{FORCING;cm_var;ATM'}$

References

- Drouet L., Edwards N.R. and A. Haurie (2004). "Coupling Climate and Economic Models in a Cost-Benefit Framework: A Convex Optimization Approach". Submitted to *Environmental Modeling and Assessment*.
- IPCC. 1996. *Climatic change 1995: The Science of Climate Change*. Intergovernmental Panel on Climate Change, Second Assessment Report, Working Group I. Cambridge: Cambridge University Press. p.572.
- IPCC. 2001. Climatic change 2001: The Scientific Basis. Intergovernmental Panel on Climate Change, Third Assessment Report, Working Group I. Cambridge: Cambridge University Press. p.944.
- Nordhaus, W. D. and J. Boyer. 1999. Roll the DICE Again: Economic Models of Global Warming. Yale University, manuscript edition.
- Wigley, T.M.L., Solomon, M. and S.C.B. Raper. 1994. Model for the Assessment of Greenhouse-Gas Induced Climate Change. Version 1.2. Climate Research Unit, University of East Anglia, UK.

Appendix B Damage Cost Functions

1 Introduction

This Appendix contains the documentation on the Damage Cost Function extensions for the TIMES model. The chapter contains 6 sections: section 2 contains the mathematical formulation, section 3 describes the parameters for the Damage Cost Functions, and section 4 gives two examples. Finally, section 5 describes the variables and section 6 describes the equations.

The Damage Cost Function option of TIMES is intended for modelers who wish to evaluate the environmental externalities caused by an energy system. For instance, emissions of toxic or environmentally harmful pollutants from the energy system create social costs linked to impacts of the pollution on human health and the environment. In another example, in global studies of GHG emissions, it may be of interest to evaluate the impact of GHG emissions on concentrations and ultimately on damages created by climate change induced by increased concentration of GHGs.

Until recently, in most studies involving bottom-up models, emission externalities have been modeled in one of two ways: either by introducing an emission tax, or by imposing emission caps. In the first case, the tax is (ideally) supposed to represent the external cost created by one unit of emission. However, using a tax assumes that the cost is a linear function of emissions. In the second approach, it is assumed that such a cost is unknown but that exogenous studies (or regulations, treaties, etc.) have defined a level of acceptable emissions that should not be exceeded. However, using this approach is akin to making the implicit assumption that emissions in excess of the cap have an infinite external cost. Both of these approaches have merit and have been successfully applied to many energy system model studies.

It is however possible to extend these two approaches by introducing an option to better model the cost of damages created by emissions. The damage function option discussed in this document extends the concept of an emission tax by modeling more accurately the assumed cost of damages due to emissions of a pollutant.

2 Mathematical formulation

We now describe the mathematical formulation used for the damage cost functions. With respect to optimization, two distinct approaches to account for damage costs can be distinguished:

- 1. Environmental damages are computed ex-post, without feedback into the optimization process, and
- 2. Environmental damages are part of the objective function and therefore taken into account in the optimization process.

In both approaches, a number of assumptions are made:

- Emissions in each region may be assumed to cause damage only in the same region or, due to trans-boundary pollution, also in other regions; however, all damage costs are allocated to the polluters in the source region, in accordance with the Polluter Pays Principle, or Extended Polluter Responsibility;
- Damages in a given time period are linked to emissions in that same period only (damages are not delayed, nor are they cumulative); and
- Damages due to several pollutants are the sum of damages due to each pollutant (no cross impacts).

In a given time period, and for a given pollutant, the damage cost is modeled as follows:

$$DAM(EM) = \alpha \cdot EM^{\beta+1} \tag{1}$$

where:

- EM is the emission in the current period;
- DAM is the damage cost in the current period;
- $\beta \ge 0$ is the elasticity of marginal damage cost to amount of emissions; and
- α > 0 is a calibrating parameter, which may be obtained from dose-response studies that allow the computation of the marginal damage cost per unit of emission at some reference level of emissions.

If we denote the marginal cost at the reference level MC₀, the following holds:

$$MC_0 = \alpha \cdot (\beta + 1) \cdot EM_0^{\beta}$$
⁽²⁾

where EM_0 is the reference amount of emissions. Therefore expression (1) may be rewritten as:

$$DAM(EM) = MC_0 \cdot \frac{EM^{\beta+1}}{(\beta+1) \cdot EM_0^{\beta}}$$
(3)

The marginal damage cost is therefore given by the following expression:

$$MC(EM) = MC_0 \cdot \frac{EM^{\beta}}{EM_0^{\beta}}$$
(4)

The approach to damage costs described in this section applies more particularly to local pollutants. Extension to global emissions such GHG emissions requires the use of a global TIMES model and a reinterpretation of the equations discussed above.

The modeling of damage costs via equation (3) introduces a non-linear term in the objective function if the β parameter is strictly larger than zero. This in turn requires that the model be solved via a Non-Linear Programming (NLP) algorithm rather than a LP algorithm. However, the resulting Non-Linear Program remains convex as long as the elasticity parameter is equal to or larger than zero. For additional details on convex programming, see Nemhauser et al (1989). If linearity is desired (for instance if problem instances are very large), we can approximate expression (3) by a sequence of linear segments with increasing slopes, and thus obtain a Linear Program.

The linearization can be done by choosing a suitable range of emissions, and dividing that range into m intervals below the reference level, and n intervals above the reference level. We also assume a middle interval centered at the reference emission level. To each interval corresponds one step variable S. Thus, we have for emissions:

$$EM = \sum_{i=1}^{m} S_i^{lo} + S^{mid} + \sum_{i=1}^{n} S_i^{up}$$
(5)

The damage cost can then be written as follows:

$$DAM(EM) = \sum_{i=1}^{m} MC_i^{lo} \cdot S_i^{lo} + MC_0 \cdot S^{mid} + \sum_{i=1}^{n} MC_i^{up} \cdot S_i^{up}$$
(6)

where:

- MC_i^{lo} and MC_i^{up} are the approximate marginal costs at each step below and above the reference level as shown in (7) below; and
- S_i^{lo} , S^{mid} and S_i^{up} are the non-negative step variables for emissions. Apart from the final step, each step variable has an upper bound equal to the width of the interval. In this formulation we choose intervals of uniform width on each side of the reference level. However, the intervals below and above the reference level can have different sizes. The width of the middle interval is always the average of the widths below and above the reference level.

The approximate marginal costs at each step can be assumed to be the marginal costs at the center of each step. If all the steps intervals are of equal size, the marginal costs for the steps below the reference level are obtained by the following formula:

$$MC_{i}^{lo} = MC_{0} \cdot \left(\frac{(i-0.5)}{(m+0.5)}\right)^{\beta}$$
(7)

Formulas for the marginal costs of the other steps can be derived similarly.

The TIMES implementation basically follows the equations shown above. Both the non-linear and linearized approaches can be used. However, in order to provide some additional flexibility, the implementation supports also defining a threshold level of emissions, below which the damage costs are zero. This refinement can be taken into account in the balance equation (5) by adding one additional step variable having an upper bound equal to the threshold level, and by adjusting the widths of the other steps accordingly. The threshold level can also easily be taken into account in the formulas for the approximate marginal costs.

In addition, the implementation supports different elasticities and step sizes to be used below and above the reference level. See Section 3 for more details.

3 Switches and Parameters

3.1 Activating the Damage Cost Functions

Like all other aspects of TIMES, the user describes the Damage Cost Functions by means of a Set and the Parameters and Switches described in this chapter.

As discussed in Section 2, the TIMES Damage Cost Function facility permits the assessment of environmental externalities by means of two approaches to determine the impact or cost of damages arising from emissions: ex-post calculation and internalized damage costs. The second approach can be further divided into the non-linear and linear formulations, and therefore the following three approaches are available in Standard TIMES:

- The environmental damages are computed ex-post, without feedback into the optimization process;
- 2. The environmental damages are a linearized part of the objective function and therefore taken into account in the optimization process;
- 3. The environmental damages are a non-linear part of the objective function and therefore taken into account in the optimization process.

The user can control whether or not the damage costs are activated in the objective function by means of the switch \$SET DAMAGE LP/NLP/NO. This switch is provided by the data handling system according to how the user wishes the option to be included:

\$SET	DAMAGE	LP
\$SET	DAMAGE	NLP
\$SET	DAMAGE	NO

The setting \$SET DAMAGE LP is the default, and activates the linearized formulation of damage costs, with the costs included in the objective function. The setting \$SET DAMAGE NLP activates the non-linear damage cost option, with the costs included in the objective function. The setting \$SET DAMAGE NO causes the damage costs only to be computed ex-post, without feedback into the optimization process.

Note that owing to the non-linear nature of the modified objective function that endogenizes the damages, the NLP damage option requires non-linear solution methods that can lead to much larger resource utilization compared to LP models. In addition, the options with an augmented objective function cannot be currently activated with the nonlinear TIMES-MACRO model variant. However, the linear option LP can be used together with the decomposed MACRO MSA option.

3.2 Input parameters

All the parameters for describing damage functions are available in the VEDA-FE shell, where they may be specified. All parameters have a prefix 'DAM_' in the GAMS code of the model generator. The parameters are discussed in more detail below:

- 1. The parameter **DAM_COST** is used to specify the marginal damage cost at the reference level of emissions. The parameter has a year index, which can be utilized also for turning damage accounting on/off for an emission in a period (by specifying an EPS value for the cost). **DAM_COST** is interpolated/extrapolated by default, but unlike other cost parameters, the interpolation is sparse, and the costs are assumed to be constant within each period.
- The parameter DAM_BQTY is used to specify the reference level of emissions. If not specified or set to zero, the marginal damage costs will be assumed constant, and no emission steps are used.
- The parameter DAM_ELAST is used to specify the elasticity of marginal damage costs to emissions in the lower and upper direction. If specified in one direction only, the elasticity is assumed in both directions. If neither is specified, the marginal damage costs will be constant in both directions.
- 4. The parameter DAM_STEP can be used for specifying the number of emission steps below and above the reference level of emissions. The last step above the reference level will always have an infinite bound. If the number of steps is not provided in either direction, but the elasticity is, one step is assumed in that direction. If a non-zero DAM_STEP(r,c,'N') is specified, the damage costs for commodity c in region r are not included in the objective. If the NLP formulation is used (DAMAGE=NLP), all DAM_STEP parameters will be ignored.
- 5. The parameter DAM_VOC can be used for specifying the variation in emissions covered by the emission steps, both in the lower an upper direction. The variation in the lower direction should be less than or equal to the reference level of emissions. If the lower variation is smaller than DAM_BQTY, the damage costs.

The input parameters are listed in Table B-1.

Table B-1. Input parameters for the TIMES Damage Cost Functions.

And D.A. Apar primiters for the Thirds During Cost Cartasta						
Input parameter (Indexes) ⁴⁷	Related parameters ⁴⁸	Units / Ranges & Default values & Default inter- /extrapolation ⁴⁹	Instances ⁵⁰ (Required / Omit / Special conditions)	Description	Affected equations or variables ⁵¹	
DAM_COST (r,datayear,c,cur)	DAM_BQTY, DAM_ELAST, DAM _STEP, DAM _VOC	TIMES cost unit [0, INF); default value: none Default i/e ^{s2} : standard	Required for each commodity for which damage costs are to be accounted.	Marginal damage cost of emission c at reference emission level.	EQ_OBJDAM	
DAM_BQTY (r,c)	See above	TIMES emission unit [0, INF); default value: 0	Only taken into account if DAM_COST has been specified	Reference level of emissions c	EQ_DAMAGE EQ_OBJDAM	
DAM_ELAST (r,c,bd)	See above	Dimensionless [0, INF); default value: 0	Only taken into account if DAM_COST has been specified	Elasticity of marginal damage cost to emissions on the lower and upper side of the reference level	EQ_OBJDAM	
DAM_STEP (r,c,bd)	See above	Dimensionless [0, INF), integer; default value: 0	Only taken into account if DAM_COST is specified. Non-zero 'N' value excludes costs from the objective.	Number of emission steps for the linearized cost function in the lower/upper direction. Can also be used for excluding the costs from the objective.	EQ_DAMAGE EQ_OBJDAM	
DAM_VOC (r,c,bd)	See above	TIMES emission unit (0, INF); ≤ DAM_BQTY; default value: DAM_BQTY	Only taken into account if DAM_COST has been specified	Variation in emissions covered by the emission steps in the lower/upper direction. A threshold emission level can be defined with bd='LO'.	EQ_DAMAGE EQ_OBJDAM	

⁴⁷ The first row contains the parameter name, the second row contains in brackets the index domain over which the parameter is defined.
 ⁴⁸ This column gives references to related input parameters or sets being used in the context of this parameter as well as internal parameters/sets or result parameters being derived from the input parameter.
 ⁴⁹ This column lists the unit of the parameter, the possible range of its numeric value [in square brackets] and the inter-/extrapolation rules that apply.
 ⁴⁰ An indication of circumstances for which the parameter is to be provided or omitted.
 ⁴¹ Equations or variables that are directly affected by the parameter.
 ⁵² Abbreviation i/e = inter-/extrapolation.

3.3 Reporting parameters

There is only one reporting parameter specifically related to the Damage Cost functions. The parameter represents the undiscounted damage costs by region, period and emission commodity. The parameter has two flavours; the first one is for standard TIMES and the second one for stochastic TIMES:

- CST_DAM(r,t,c): Annual damage costs from emission c in region r,
- SCST_DAM(w,r,t,c): Annual damage costs from emission c in region r and stochastic scenario w.

However, in addition the standard reporting parameters REG_WOBJ, and REG_ACOST are augmented with damage costs results, using the label 'DAM'/'DAM-EXT' to distinguish damage costs from other cost components.

These parameters are included in the .vdd files that describe the parameters to be transferred to VEDA-BE under standard TIMES and stochastic TIMES. Therefore, the corresponding result parameter is always available in VEDA-BE whenever Damage Cost functions have been defined, even with the setting DAMAGE=NO.

The damage costs are always reported by using the accurate non-linear expressions, even if the linearized formulation is chosen for the augmented objective function.

Table B-2. Reporting parameters for the TIMES Damage cost functions.

Parameter	Description			
CST_DAM(r,t,c)	Damage costs by region, period and emission (standard TIMES)			
SCST_DAM (w,r,t,c)	Damage costs by region, period and emission (stochastic TIMES)			
4 Examples

Assume that we wish to define linearized damage costs for the emission commodity 'EM' so that the cost function has the following properties:

- The reference level of emissions is 80 units;
- The marginal cost at the reference level are 10 cost units per emission unit;
- The cost elasticity is 1 in the lower direction, and 0.7 in the upper direction;

The damage function can be specified with the following parameters:

PARAMETER DAM COST	/ REG.2000.EM.CUR 10 /:
PARAMETER DAM_BQTY	/ REG.EM 80 /;
PARAMETER DAM_ELAST	/ REG.EM.LO 1, REG.EM.UP 0.7 /;

As we did not specify the number of steps, but we did specify the elasticities in both directions, the number of steps is assumed to be 1 in both directions. The resulting damage cost function is illustrated in Figure 15. Because the damage function has a very coarse representation, the total costs have notable deviations from the accurate non-linear function. Note that the step size has been automatically determined to be DAM_BQTY/(DAM_STEP+0.5) = 80/1.5. However, the last step has no upper bound.

Assume next that we would like to refine the damage function by the following specifications:



Figure 15. Example of a linearized damage function with 1+1+1 steps (1 lower step, 1 middle step, 1 upper step).

- We want to have 5 steps below the reference, and 3 steps above it;
- The threshold level of damage costs is 20 units of emissions;
- The steps above the reference level should cover 100 units of emissions.

The damage function can be specified with the following parameters

PARAMETER	DAM_COST	/ REG.2000.EM.CUR 10 /;
PARAMETER	DAM_BQTY	/ REG.EM 80 /;
PARAMETER	DAM_ELAST	/ REG.EM.LO 1, REG.EM.UP 0.7 /;
PARAMETER	DAM_STEP	/ REG.EM.LO 5, REG.EM.UP 3 /;
PARAMETER	DAM_VOC	/ REG.EM.LO 60, REG.EM.UP 100 /;

The resulting damage cost function is illustrated in Figure 16. The cost function follows now very closely the accurate non-linear function. Note that the step sizes derived from the VOC specifications are 10 units for the lower steps, 20 for the middle step, and 30 units for the upper steps. However, the last step of course has no upper bound.



Figure B-16. Example of a linearized damage function with 1+5+1+3 steps (one zero cost step, 5 lower steps, one middle step, 3 upper steps).

5 Variables

There are only two sets of new variables in the damage cost formulation, VAR_DAM and VAR_OBJDAM, which are shown below in Table B-3. The variables VAR_DAM represent the steps in the emissions in each period. In the linearized formulation, there are DAM_STEP(...,'LO') number of step variables on the lower side and DAM_STEP(...,'UP') number of step variables on the higher side of emissions. In addition, one step variable of type 'FX' corresponds to the middle step that includes the reference level of emissions, and an optional additional step variable of type 'FX' corresponds to the zero-damage fraction of emissions, as defined by the difference between DAM_BQTY(..) and DAM_VOC(...,'LO').

The variables VAR_OBJDAM represent the total discounted damage costs by region. The undiscounted costs in each period described in Section 2 are discounted and summed over all periods and emissions in each region. As emissions are in TIMES assumed to be constant within each period, damage costs are likewise assumed to be constant within each period.

Table B-3. Model variables specific to the Damage Cost Functions.

Variable (Indexes)	Variable Description
VAR_DAM (r,t,c,bd,j)	The emission step variable for the damage function of commodity ${\bf c}$ in region ${\bf r},$ for each step ${\bf j}$ in each direction ${\bf bd}.$
VAR_OBJ (r,'OBJDAM',cur)	The variable is equal to the sum of the total discounted damage costs in each region r with currency cur .

5.1 VAR_DAMAGE(r,t,c,bd,j)

Description:	The amount of emission indicator \mathbf{c} at cost step \mathbf{j} in direction \mathbf{bd} , in period \mathbf{t} .
Purpose:	This variable tracks the amount of an emission indicator by cost step and period, in both the lower and upper direction from the reference level.
Occurrence:	The variable is generated for emission indicator that has damage costs specified, whenever the damage cost functions are included in the objective function.
Units:	Units of the emission commodity c .
Bounds:	This variable cannot be directly bounded by the user.

5.2 VAR_OBJ(r,'OBJDAM',cur)

Description:	The total present value of damage costs by region.
Purpose:	This variable is included in the objective function in order to include damage costs in the objective when requested by the user.
Occurrence:	This variable is generated for each region when damage cost functions are included in the objective function
Units:	Currency units used for damage functions.
Bounds:	This variable cannot be directly bounded by the user.

6 Equations

There are two blocks of equations generated for damage cost functions, whenever they are included in the objective function. The two equations related to the damage functions are listed and briefly described below in Table B-4. The equations include the balance of stepped emissions, the objective component for damage costs, and the augmented total objective function.

In addition, the standard TIMES objective function, **EQ_OBJ**, is augmented by the present value of the damage costs, as defined by the equation EQ_OBJDAM.

We now give the formulations of these constraints.

Reminder: the Damage Cost Functions are activated at run time from the data handling system, which in turn sets the switch SET DAMAGE LP/NLP/NO.

Table B-4. Constraints specific to	damage costs (in t	the GAMS file eqdan	1age.mod).
------------------------------------	--------------------	---------------------	------------

Constraints (Indexes)	Constraint Description		
EQ_DAMAGE (r,t,c)	The balance equation between the stepped emission variables and the total emissions in each period.		
EQ_OBJDAM (r,cur)	The total discounted damage costs by region, which will be added as a component to the objective function.		

6.1 EQ_DAMAGE(r,t,c)

Description: Allocates the total amount of emission indicator in period t to cost steps.

Purpose:	This constraint allocates the total amount of emission indicator c to the cost steps of the linearized / non-linear damage cost functions in each period t . This equation is generated in each time period for all emission indicators considered.
Units:	Units of the emission commodity c .

Type: Binding. The equation is an equality (=) constraint.

Remarks:

- The damage costs can be defined either on the net production (VAR_COMNET) or the gross production (VAR_COMPRD) of the commodity c. By default the damage costs are applied to the NET amount, unless *DAM_ELAST*_{*r,c,N*} is also specified. *DAM_ELAST*_{*r,c,N*} defines a multiplier for the Base prices to be added to the damage cost function, when it is to be applied to the gross production.
- The internal parameter *DAM_COEF*_{r,t,c,s} is set to the base prices, if *DAM_ELAST*_{r,c,N'} is specified, and otherwise to 1.

Equation:

$$EQ_DAMAGE_{r,t,c} \quad \Rightarrow \left(\mathsf{rtc}_{r,t,c} \land \exists (cur) : DAM_COST_{r,t,c,cur} \right)$$

$$\sum_{\substack{(ij,bd) \in dam_num_{r,s,jj,bd} \ j \leq ij}} \sum_{j \leq ij} VAR _DAM_{r,t,c,bd,j}$$

$$\{=\}$$

$$\sum_{\substack{com_ts_{r,e,ts}}} \begin{pmatrix} DAM _COEF_{r,t,c,ts} \times \\ (VAR _COMNET_{r,t,c,ts} & if \ DAM _ELAST_{r,c,N'} \ not \ given \\ VAR _COMPRD_{r,t,c,ts} & otherwise \end{pmatrix} \end{pmatrix}$$

6.2 EQ_OBJDAM(r,cur)

Description: Computes the present value of all damage costs by region and currency.

Purpose:	Defines the variable VAR_OBJ(r,'OBJDAM',cur), which represents the
	total present value of all damage costs in region r, having currency cur.
	This variable is included in the TIMES objective function.

Units: Currency units.

Type: Binding. The equation is an equality (=) constraint.

Remarks:

• The internal parameter *DAM_SIZE*_{*r,c,bd*} represents the sizes of cost steps of the dlinearized damage cost function, for both directions (bd=LO/UP) and for the middle step (bd=FX), as described above in Section 2.

Equation:

$$EQ_OBJDAM_{r,cur} \quad \mathbf{\mathfrak{s}}\left(\mathbf{rdcur}_{\mathbf{r},\mathbf{cur}}\right)$$

Case A: Linearized functions

$$\left\{ \begin{array}{l} \sum\limits_{\substack{j \in \operatorname{dam_num}_{r,z,j,UO'} \\ j \leq jj}} \left(\frac{VAR_DAM_{r,t,c,'LO',j}}{DAM_BQTY_{r,c}^{-DAM_ELAST_{r,z,UO'}}} \times \\ \left(\frac{DAM_BQTY_{r,c} - DAM_VOC_{r,c,'LO'} + }{DAM_ELAST_{r,z,UO'}} \right)^{DAM_ELAST_{r,z,UO'}} \right)^{+} \\ VAR_DAM_{r,t,c,'FX',1} \\ \left\{ \begin{array}{l} \sum\limits_{\substack{j \in \operatorname{dam_num}_{r,z,j,UP'} \\ j \leq ORD(jj)}} \left(\frac{VAR_DAM_{r,t,c,'UP',j}}{DAM_BQTY_{r,c}^{-DAM_ELAST_{r,z,UP'}}} \times \\ \left(\frac{DAM_BQTY_{r,c} + \frac{DAM_SIZE_{r,c,'EX'}}{2} + \\ DAM_SIZE_{r,c,'UP'} \times (j - 0.5) \end{array} \right)^{-} \right)^{-} \\ \end{array} \right)^{-} \\ \left\{ = \} \end{array} \right\}$$

 $VAR_OBJ_{r,'OBJDAM',cur}$

Case B: Non-linear functions

$$\sum_{(t,c) \in \{\mathsf{rtc}_{r,t,c} | (DAM_COST_{r,t,c,cur} \times OBJ_PVT_{r,t,cur} \times OBJ_PVT_{r,t,cur} \times (DAM_COST_{r,t,c})\} } \left[\left(\left(VAR_DAM_{r,t,c,'LO',j} + DAM_BQTY_{r,c} - DAM_VOC_{r,c,'LO'} \right)^{(DAM_ELAST_{r,c,'LO'}+1)} - DAM_BQTY_{r,c} - DAM_VOC_{r,c,'LO'} \right)^{(DAM_ELAST_{r,c,'LO'}+1)} + \left(\frac{(DAM_BQTY_{r,c} - DAM_VOC_{r,c,'LO'})^{(DAM_ELAST_{r,c,'LO'}+1)}}{DAM_BQTY_{r,c} - DAM_PDAM_ELAST_{r,c,'LO'} \times (DAM_ELAST_{r,c,'LO'}+1)} + \left(\frac{(VAR_DAM_{r,t,c,'UP',j} + DAM_BQTY_{r,c})^{(DAM_ELAST_{r,c,'UP'}+1)}}{DAM_BQTY_{r,c} - DAM_LAST_{r,c,'UP'} \times (DAM_ELAST_{r,c,'UP'}+1)} - DAM_BQTY_{r,c} \right)^{(DAM_ELAST_{r,c,'UP'}+1)} - \left(\frac{(DAM_BQTY_{r,c})^{(DAM_ELAST_{r,c,'UP'}+1)} \times (DAM_ELAST_{r,c,'UP'}+1)}{DAM_BQTY_{r,c} - DAM_ELAST_{r,c,'UP'} \times (DAM_ELAST_{r,c,'UP'}+1)} - \right) \right]$$

 $VAR _OBJ_{r,'OBJDAM',cur}$

References

- Goldstein, G., Noble, K. & Van Regemorter, D. 2001. Adaptation to MARKAL for including environmental damages. MARKAL User Information Note.
- Loulou, R., Goldstein, G. & Noble, K. 2004. Documentation for the MARKAL Family of Models. October 2004. <u>http://www.iea-etsap.org/web/Documentation.asp</u>
- Loulou, R., Remme, U., Kanudia, A., Lehtilä, A. & Goldstein, G. 2005. Documentation for the TIMES Model. Energy Technology Systems Ananlysis Programme (ETSAP), April 2005. <u>http://www.iea-etsap.org/web/Documentation.asp</u>
- Nemhauser, G.L., Rinnooy Kan, A.H.G. & Todd, M.J. (eds.) 1989. Handbooks in Operations Research and Management Science, Vol I: Optimization. North-Holland.

Appendix C Endogenous Technological Learning (ETL)

1 Introduction

As discussed in Chapter 11 of Part I, there are situations in which the rate at which a technology's unit investment cost changes over time is a function of cumulative investment in the technology. In these situations, technological learning is called endogenous.

Mixed Integer Programming (MIP) is employed in order to model Endogenous Technological Learning (ETL) in TIMES. As has already been noted in the case of Lumpy Investments, MIP problems are much more difficult to solve than standard LP problems, and so the ETL feature should be applied only where it is deemed necessary to model a limited number of technologies as candidates for Endogenous Technological Learning. This caution is especially required for large-scale TIMES instances. Another important caveat is that ETL is relevant when the modeling scope is broad e.g. when a large portion of (or perhaps the entire) world energy system is being modeled, since the technological learning phenomenon rests on global cumulative capacity of a technology, and not on the capacity implemented in a small portion of the world.

In this chapter we provide the data and modeling details associated with modeling Endogenous Technological Learning (ETL) in TIMES. The implementation of ETL in TIMES is based on the realization in the MARKAL model generator. The major part of the MARKAL code for ETL could be transferred to TIMES. Accordingly the description of ETL presented here follows the MARKAL documentation of ETL. To this end the next three sections will address the Sets, Parameters, Variables, and Equations related to the Endogenous Technological Learning option, including the special clustered learning ETL option where a component common to several technologies learns, thereby benefiting all the related (clustered) technologies.

2 Sets, Switches and Parameters

Like all other aspects of TIMES the user describes the ETL components of the energy system by means of a Set and the Parameters and Switches described in this chapter. Table C-1 and Table C-2 below describe the User Input Parameters, and the Matrix Coefficient and Internal Model Sets and Parameters, respectively, that are associated with the Endogenous Technological Learning option. Note that the special clustered learning ETL option requires one additional User Input Parameter (ETL-CLUSTER), and two additional Matrix Coefficient/Internal Model Parameters (CLUSTER and NTCHTEG).

Besides the basic data described in Table the user controls whether or not the ETL component is activated by means of the \$SET ETL 'YES' switch. This switch is provided by the data handling system when the user indicates that the ETL option is to be included in a run. This permits the easy exclusion of the feature if the user does not want to perform a MIP solve without having to remove the ETL data.

Tuble e 1.	Tube C 1. Deminuon of DTD user input parameters					
Input Parameter (Indexes)	Alias / Internal Name	Related Parameters	Units/Range & Defaults	Instance (Required/Omit/ Special Conditions)	Description	
ССАРО (г,р)	TL_CCAP0	PAT CCOSTO	 Units of capacity (e.g., GW, PJa). [open]; no default. 	 Required, along with the other ETL input parameters, for each learning technology (TEG). 	 The initial cumulative capacity (starting point on the learning curve) for a (non-resource) technology that is modeled as one for which endogenous technology learning (ETL) applies. Learning only begins once this level of installed capacity is realized. The CCAP0 parameter appears as the righthand-side of the cumulative capacity definition constraint (EQ_CUINV). Note that if the NCAP_PASTI parameter is specified for an ETL technology, then its value in the first period should match the value of CCAP0, otherwise an infeasibility will occur. 	
CCAPM (r,p)	TL_CCAPM	CCOSTM	 Units of capacity (e.g., GW, PJa). [open]; no default 	 Required, along with the other ETL input parameters, for each learning technology (TEG). 	 The maximum cumulative capacity (ending point on the learning curve) for a (non-resource) technology that is modeled as one for which endogenous technology learning (ETL) applies. The parameter CCAPM does not appear in any of the ETL constraints, but its value affects the values of a number of internal parameters that directly contribute to one or more of the ETL constraints. 	
TEG (p)	TEG	ETL-CUMCAP0 ETL- CUMCAPMAX ETL-INVCOST0 ETL-NUMSEG ETL-PROGRATIO	 Indicator. [1]; no default. 	 Required to identify the learning technologies. For each TEG the other ETL input parameters are 	An indicator (always 1) that a process is modeled as one for which endogenous technology learning (ETL) applies. The set TEG controls the generation of the ETL constraints. Each of the ETL constraints is generated only for those technologies that are in set TEG.	

Table C-1. Definition of ETL user input parameters

Input	Alias /	Related	Units/Range &	Instance	Description
Parameter	Internal	Parameters	Defaults	(Required/Omit/	
(Indexes)	Name			Conditions)	
				required.	
SC0 (r,p)	TL_SC0	PAT	 Base year monetary units per unit of capacity (e.g., 2000 M\$/GW or PJa). [open]; no default. 	 Required, along with the other ETL input parameters, for each learning technology (TEG). 	 The investment cost corresponding to the starting point on the learning curve for a technology that is modeled as one for which endogenous technology learning (ETL) applies. The parameter SC0 does not appear in any of the ETL constraints, but its value affects the values of a number of internal parameters that directly contribute to one or more of the ETL constraints.
SEG (r,p)	TL_SEG	ALPH BETA CCAPK CCOSTK	 Number of steps, [1-6]; no default. 	 Required, along with the other ETL input parameters, for each learning technology (TEG). 	 The number of segments to be used in approximating the learning curve for a technology that is modeled as one for which endogenous technology learning (ETL) applies. The SEG parameter appears in all of the ETL constraints that are related to piecewise linear approximation of the learning curve (EQ_CC, EQ_COS, EQ_EXPE1, EQ_EXPE2, EQ_LA1, EQ_LA2).
PRAT (r,p)	TL_PRAT	CCAPK CCOSTO CCOSTM PAT PBT	 Decimal fraction. [0-1]; no default. 	 Required, along with the other ETL input parameters, for each learning technology (TEG). 	The "progress ratio" for a technology that is modeled as one for which endogenous technology learning (ETL) applies. The progress ratio, which is referred to as the learning rate, is defined as the ratio of the change in unit investment cost each time cumulative investment in an ETL technology doubles. That is, if the initial unit investment cost is SCO and the progress ratio is PRAT, then after cumulative investment is doubled the unit investment cost will be PRAT * SCO. • The parameter PRAT does not appear in any of the ETL constraints, but its value affects the

Input Parameter (Indexes)	Alias / Internal Name	Related Parameters	Units/Range & Defaults	Instance (Required/Omit/ Special	Description
()				Conditions)	
					values of a number of internal parameters (ALPH, BETA, CCAPK, CCOSTO) that directly contribute to one or more of the ETL constraints.
CLUSTER (r,p,p)	TL_CLUSTER NCLUSTER	TL_MRCLUST	 Decimal fraction. [0-1]; no default. 	 Provided to model clustered endogenous technology learning. Each of the learning parameters must also be specified for the key learning technology. 	The "cluster mapping and coupling factor" for a technology that is modeled as a <u>clustered</u> technology to which endogenous technology is associated with a <u>key</u> learning technology. Schuler and technology. Schuler and technology. The first index of the CLUSTER parameter is a <u>key</u> learning technology. The first index of the CLUSTER parameter is a <u>clustered</u> technology that is associated with this <u>key</u> learning technology. The first index of the CLUSTER parameter is a <u>clustered</u> technology that is associated with this <u>key</u> learning technology, and hence there may be several <u>instances</u> of the CLUSTER parameter is a <u>clustered</u> technology. The second index of the several <u>clustered</u> technologies each of which is associated with the same <u>key</u> learning technology as its first index. The numerical value of the CLUSTER parameter indicates the extent of coupling between the <u>clustered</u> technology and the <u>key</u> learning technology technolog on the <u>key</u> learning technology the solution to the same there may be several there t
TL_MRCLUST (r,teg,reg,p)		CLUSTER	 Decimal fraction. 	 See CLUSTER 	The multi-region cluster mapping and coupling factor. Similar to CLUSTER, but may be used to
			 [0-1]; no default. 		map technologies p in multilple regions reg to key components teg in region r. See CLUSTER.

361

Table C-2. ETL-specific matrix coefficient and internal model parameters⁵³

Matrix Controls & Coefficients (indexes)	Туре	Description & Calculations
ALPH (r,k,p)	Ι	ALPH are the intercepts on the vertical axis of the line segments in the piecewise linear approximation of the cumulative cost curve. They are calculated in COEF_ETL.ETL from the starting and ending points of the cumulative cost curve, its assumed form, the number of segments used in its piecewise linear approximation, and the choice of successive interval lengths on the vertical axis to be such that each interval is twice as wide as the preceding one. The parameter ALPH occurs in the ETL equation EQ COS that defines the piecewise linear approximation to the cumulative cost curve.
BETA (r,k,p)	I	BETA are the slopes of the line segments in the piecewise linear approximation of the cumulative cost curve. They are calculated in COEF_ETLETL from the starting and ending points of the cumulative cost curve, its assumed form, the number of segments used in its piecewise linear approximation, and the choice of successive interval lengths on the vertical axis to be such that each interval is twice as wide as the preceding one. The parameter BETA occurs in the ETL equation EQ_COS that defines the piecewise linear approximation to the cumulative cost curve.
CCAP0 (r,p)	A	CCAP0 is the initial cumulative capacity (starting point on the learning curve). The parameter CCAP0 occurs in the ETL equation EQ_CUINV that defines cumulative capacity in each period.
ССАРК (k,p)	I	CCAPK are the break points on the horizontal axis in the piecewise linear approximation of the cumulative cost curve. They are calculated in COEF_ETL.ETL from the starting and ending points of the cumulative cost curve, its assumed form, the number of segments used in its piecewise linear approximation, and the choice of successive interval lengths on the vertical axis to be such that each interval is twice as wide as the preceding one. The parameter CCAPK occurs in the ETL equations EQ_LA1 and EQ_LA2 whose role is to ensure that variable R_LAMB(r,t,k,p) lies in the k^{th} interval, i.e., between CCAPK(r,k-1,p) and CCAPK(r,k,p), when its associated binary variable R_DELTA(r,t,k,p) = 1.
CCOSTO (r,p)	I	CCOST0 is the initial cumulative cost (starting point on the learning curve). It is calculated in COEF_ETLETL from the initial cumulative capacity (CCAP0) and corresponding initial investment cost (user input parameter SC0) and the progress ratio (user input parameter PRAT). The parameter CCOST0 occurs in the ETL equation EQ_IC1 that defines first period investment costs (prior to discounting).

³³ Parameters that occur in the ETL-specific equations but that also occur in non-ETL equations (e.g., TCH_LIFE) are not listed in this table.

		-
Matrix Controls & Coefficients (indexes)	Туре	Description & Calculations
SEG	A	The user input parameter SEG is the number of segments in the cumulative cost curve. The
(r.p)		parameter SEG occurs in all of those ETL equations that are related to the piecewise linear
		approximation of the cumulative cost curve.
TEG	S	TEG is the set of technologies to which endogenous technology learning (ETL) applies. Each of the ETL
(p)		equations has set TEG as an index.
CLUSTER	I	The user input parameter CLUSTER (cluster mapping and coupling factor) is only relevant when
(r.p.p)		modeling clustered endogenous technology learning. The parameter occurs in the special ETL cluster
		equation EQ_CLU that defines investment in new capacity (VAR_NCAP) in the key learning technology
		as the weighted sum of investments in new capacity of the clustered technologies that are attached to
		the key technology. (The weights used are the numeric values of the CLUSTER parameter.)
TL_MRCLUST	I	The user input parameter TL_MRCLUST is only relevant when modeling clustered endogenous
(r,teg,reg,p)		technology learning. The parameter occurs in the special ETL cluster equation EQ_MRCLU that defines
(,),),),)		investment in new capacity (VAR_NCAP) in the key learning technology as the weighted sum of
		investments in new capacity of the clustered technologies that are attached to the key technology.
NTCHTEG	I	The parameter NTCHTEG is only relevant when modeling clustered endogenous technology learning. If
(r,p)		TEG is an ETL technology, then NTCHTEG(R,TEG) is the number of clustered technologies that are
		attached to key technology TEG. NTCHTEG is calculated in COEF_ETL.ETL from the "cluster mapping"
		and coupling factor" (CLUSTER). It occurs in the special ETL cluster equation EQ_CLU.
PBT		The learning index PBT is an internal parameter calculated in COEF_ETL.ETL. It is derived from the
(r,p)		progress ratio PRAT using the formula: $PBT(r,p) = -log(PRAT(r,p))/log(2)$. PBT does not occur directly
a - a a		in the equations, but is used in the calculation of equation coefficients.
PAT		The internal parameter PAT describes the specific investment costs of the first unit. It is derived in
(r,p)		COEF_ETL.ETL using PBT, SC0 and CCAP0. PAT does not occur directly in the equations, but is used in
0.00		the calculation of equation coefficients.
K		The set K has the members '1'-'6' and is used as indicator for the kink points of the piecewise linear
		approximation of the cumulative cost curve. The number of elements can be changed in the *run file
		lif desired.
WEIG	I	The internal parameter WEIG is calculated in COEF_ETL.ETL and is used as a factor in the calculation
(r,k,prc)		of the length of the intervals being used in the piecewise linear approximation of the cumulative cost
		curve. The interval lengths on the vertical axis are chosen in such a way that each interval is twice as
		wide as the preceding one.

3 Variables

The variables that are used to model the Endogenous Technological Learning option in TIMES are presented in Table below. As is the case with the modeling of lumpy investments, the primary role of the variables and equations used to model ETL is to control the standard TIMES investment variable (VAR_NCAP) and the associated dynamic cost of these investments, so ETL is rather self-contained. That is the VAR_NCAP variable links the ETL decisions to the rest of the model, and the VAR_IC investment cost variable determines the associated contribution to the regional investment costs (VAR_OBJINV). Note that the special clustered learning ETL option does not require any additional variables, as compared with the modeling of endogenous technology learning when there are no clusters.

Table C-3. ETL-specific model variables

Variable	Variable Description
(Indexes)	
VAR_CCAP (r,t,p)	The cumulative investment in capacity for an ETL technology. This variable represents the initial cumulative capacity (CCAP0) plus investments in new capacity made up to and including the current period. This variable differs from the total installed capacity for a technology (VAR_CAP) in that it includes all investments in new capacity made up to and including the current period, whereas the latter only includes investments that are still available (i.e. whose life has not expired yet).
VAR_CCOST (r,t,p)	The cumulative cost of investment in capacity for an ETL technology. The cumulative cost is interpolated from the piecewise linear approximation of the cumulative cost curve.
VAR_DELTA (r,t,p,k)	Binary variable (takes the value 0 or 1) used for an ETL technology to indicate in which interval of the piecewise linear approximation of the cumulative cost curve the cumulative investment in capacity (VAR_CCAP) lies. A value of 1 for this variable for exactly one interval k indicates that VAR_CCAP lies in the k th interval.
VAR_IC (r,t,p)	The portion of the cumulative cost of investment in capacity for an ETL technology (VAR_CCOST) that is incurred in period t, and so subject to the same discounting that applies to other period t investment costs. This variable is calculated as the difference between the cumulative costs of investment in capacity for periods t and t-1, and enters the regional investment cost part of the objective function (EQ_OBJINV)
VAR_LAMBD (r,t,p,k)	Continuous variable used for an ETL technology to represent the portion of cumulative investment in capacity (VAR_CCAP) that lies in the k th interval of the piecewise linear approximation of the cumulative cost curve. For a given ETL technology and given time period, ETL model constraints involving this variable and the associated binary variable VAR_DELTA ensure that VAR_LAMBD is positive for exactly one interval k.

3.1 VAR_CCAP(r,t,p)

Description: The cumulative investment in capacity for an ETL technology.

Purpose and This variable tracks the cumulative investment in capacity for an ETL **Occurrence:** technology which then determines, along with the progress ratio, how much the investment cost is to be adjusted for the learning gains.

This variable is generated for each ETL technology in all time periods beginning from the period that the technology is first available. It appears in the cumulative capacity definition constraint (EQ_CUINV) that defines it as the initial cumulative capacity (CCAP0) plus investments in new capacity (VAR_NCAP) made up to and including the current period. It also appears in the cumulative capacity interpolation constraint (EQ_CC). This constraint equates VAR_CCAP(r,t,p) to the sum over k of the variables VAR_LAMBD(r,t,p,k) used to represent the cumulative investment in capacity lying in the kth interval of the piecewise linear approximation of the cumulative cost curve.

- Units: PJ/a, Gw, or Bvkm/a, or any other unit defined by the analyst to represent technology capacity.
- Bounds: This variable is not directly bounded. It may be indirectly bounded by specifying a bound (NCAP_BND) on the level of investment in new capacity (VAR_NCAP).

3.2 VAR_CCOST(r,t,p)

Description: The cumulative cost of investment in capacity for an ETL technology.

Purpose and This variable defines the interpolated cumulative cost of investment in Occurrence: capacity in terms of the continuous variables VAR_LAMBD and the binary variables VAR_DELTA, and the internal model parameters ALPH and BETA. ALPH and BETA represent the intercepts on the vertical axis and the slopes, respectively, of the line segments in the piecewise linear approximation of the cumulative cost curve.

This variable is generated for each ETL technology in all time periods beginning from the period that the technology is first available. It appears in the cumulative cost interpolation equation (EQ_COS) that defines it. It also appears in the equations EQ_IC1 and EQ_IC2 that define the

VAR_IC variables that represent the portions of the cumulative cost of investment in capacity that are incurred in period t.

Units: Million 2000 US\$, or any other unit in which costs are tracked.

Bounds: None.

3.3 VAR_DELTA(r,t,p,k)

Description: Binary variable (takes the value 0 or 1) used for an ETL technology to indicate in which interval of the piecewise linear approximation of the cumulative cost curve the cumulative investment in capacity (VAR_CCAP) lies.

Purpose andTo indicate which step on the learning curve a technology achieves. AOccurrence:value of 1 for this variable for interval k, and zero values for intervals \neq k,
imply that the cumulative investment in capacity (VAR_CCAP) lies in the
kth interval of the piecewise linear approximation of the cumulative cost
curve.

This binary variable, along with the associated continuous variable VAR_LAMBD, are generated for each ETL technology in all time periods beginning from the period that the technology is first available, and for each interval in the piecewise linear approximation. It appears in the constraint EQ_DEL, whose purpose is to ensure that, for each ETL technology in each period, it has a value of 1 for exactly one interval k (with zero values for intervals \neq k); and in the cumulative cost interpolation constraint (MR_COS). It also appears in the pair of constraints EQ_LA1 and EQ_LA2, whose purpose is to ensure that VAR_LAMBD, if positive for interval k, is between the two break points on the horizontal axis for interval k in the piecewise linear approximation. (See below under "Purpose and Occurrence" for the variable VAR_LAMBD.)

Finally, this binary variable appears in two constraints EQ_EXPE1 and EQ_EXPE2, whose purpose is to reduce the domain of feasibility of the binary variables and thereby improve solution time for the Mixed Integer Program (MIP).

Units: None. This is a binary variable that takes the value 0 or 1.

Bounds: This binary variable is not directly bounded.

3.4 VAR_IC(r,t,p)

Description: The portion of the cumulative cost of investment in capacity for an ETL technology (VAR CCOST) that is incurred in period t.

Purpose and This variable represents the portion of the cumulative cost of investment **Occurrence:** in capacity for an ETL technology that is incurred in period t, and so is subject to the same discounting in the investment cost part of the objective function (EQ OBJINV) that applies to other period t investment costs.

This variable is calculated as the difference between the cumulative costs of investment in capacity for period t and t-1, and is generated for each ETL technology in all time periods beginning from the period that the technology is first available. Apart from its appearance in the objective function, this variable appears in the constraints EQ_IC1 and EQ_IC2 that define it in the first period that the technology is available, and in subsequent periods, respectively. It also appears in the salvage of investments constraint (EQ_OBJSALV), which calculates the amount to be credited back to the objective function for learning capacity remaining past the modeling horizon.

Units: Million 2000 US\$, or any other unit in which costs are tracked.

Bounds: None.

3.5 VAR_LAMBD(r,t,p,k)

Description: Continuous variable used for an ETL technology to represent the portion of cumulative investment in capacity (VAR_CCAP) that lies in the kth interval of the piecewise linear approximation of the cumulative cost curve.

Purpose andA positive value for this variable for interval k, and zero values forOccurrence:intervals \neq k, imply that the cumulative investment in capacity
(VAR_CCAP) lies in the kth interval of the piecewise linear approximation
of the cumulative cost curve. This continuous variable, along with the
associated binary variable VAR_DELTA, are generated for each ETL
technology in all time periods beginning from the period that the
technology is first available (START), and for each interval in the
piecewise linear approximation.

Since this variable represents the portion of the cumulative investment in capacity (VAR_CCAP) that lies in the k^{th} interval of the piecewise linear approximation of the cumulative cost curve, the value of

EQ_LAMBD – if positive – is required to be between CCAPK(k-1,p) and CCAP(k,p), where the internal model parameters CCAPK are the break points on the horizontal axis in the piecewise linear approximation of the cumulative cost curve. A zero value for VAR_LAMBD is also allowed. These requirements on the value of VAR_LAMBD are imposed via the pair of constraints EQ_LA1 and EQ_LA2, in which the value for VAR_LAMBD is subject to lower and upper bounds of CCAPK(k-1,p) * VAR_DELTA and CCAP(k,p) * VAR_DELTA respectively, where VAR_DELTA = VAR_DELTA(r,t,p,k) is the binary variable associated with VAR_LAMBD = VAR_LAMBD

This variable also appears in the cumulative capacity interpolation constraint (EQ_CC), and the cumulative cost interpolation constraint (EQ_COS).

- Units: PJ/a, Gw, or Bvkm/a, or any other unit defined by the analyst to represent technology capacity.
- Bounds: The pair of constraints EQ_LA1 and EQ_LA2 that are discussed above have the effect of either bounding VAR_LAMBD between CCAPK(k-1,p) and CCAP(k,p), or forcing VAR_LAMBD to be zero.

4 Equations

The equations that are used to model the Endogenous Technological Learning option in TIMES are presented in Table C-4 below. Since the primary role of the variables and equations used to model ETL is to control the standard TIMES investment variable (VAR_NCAP) and the associated dynamic cost of these investments, ETL is rather self-contained. That is the VAR_NCAP variable links the ETL decisions to the rest of the model, and the VAR_IC investment cost variable determines the associated contribution to the regional investment cost part objective function (EQ_OBJINV). Note that the special clustered learning ETL option involves one additional equation (EQ_CLU), as compared with the modeling of endogenous technology learning where there are no clusters. IN BOX BELOW, ADD ANSWER or CHANGE TO "system"

Reminder: the ETL formulation is activated at run time from the data handling system, which in turn sets the SET ETL 'YES' switch.

Constraints (Indexes)	Constraint Description	GAMS Ref
EQ_CC (r,t,p)	The Cumulative Capacity Interpolation constraint for an ETL technology. This constraint defines the cumulative investment in capacity for a technology (VAR_CCAP) in a period as the sum over all intervals k of the continuous variables R_LAMBD(r,t,p,k) that represent cumulative investment in capacity as lying in the k th interval of the piecewise linear approximation of the cumulative cost curve.	EQU_EXT.ETL
EQ_CLU (r,t,p)	Constraint that is generated only for the special clustered learning ETL option (CLUSTER). For a key learning ETL technology it defines investment in new capacity (VAR_NCAP) as the weighted sum of investments in new capacity of the associated clustered technologies.	EQU_EXT.ETL
EQ_COS (r,t,p)	The Cumulative Cost Interpolation constraint for an ETL technology. This constraint defines the interpolated cumulative cost of investment in capacity for a technology (VAR_CCOST) in a period in terms of the binary variables VAR_DELTA and the continuous variables VAR_LAMBD, and the internal model parameters ALPH and BETA.	EQU_EXT.ETL
EQ_CUINV (r,t,p)	The Cumulative Capacity Definition constraint for an ETL technology. Defines the cumulative investment in capacity for a technology in a period as the initial cumulative capacity (CCAP0) plus the sum of investments in new capacity (VAR_NCAP) made up to and including this period.	EQU_EXT.ETL

Tahle	C-4.	ETL-9	mecific	model	constraints
LAUIC	U-4.		DECHIC	mouci	CONSULATION

Constraints (Indexes)	Constraint Description	GAMS Ref
EQ_DEL (r,t,p)	The constraint for an ETL technology that ensures that in each period there is exactly one interval k for which the binary variable R_DELTA(r,t,p,k) has value 1 (with zero values for intervals \neq k).	EQU_EXT.ETL
EQ_EXPE1 (r,t,p,k)	One of two constraints for an ETL technology to improve MIP solution time by reducing the domain of feasibility of the binary variables VAR_DELTA.	EQU_EXT.ETL
EQ_EXPE2 (r,t,p,k)	Second of two constraints for an ETL technology to improve MIP solution time by reducing the domain of feasibility of the binary variables VAR_DELTA.	EQU_EXT.ETL
EQ_IC1 (r,t,p)	The constraint for an ETL technology that defines the portion of the cumulative cost of investment in capacity (VAR_IC) that is incurred in the first period of the model horizon.	EQU_EXT.ETL
EQ_IC2 (r,t,p)	The constraint for an ETL technology that defines the portion of the cumulative cost of investment in capacity (VAR_IC) that is incurred in each period but the first one.	EQU_EXT.ETL
EQ_LA1 (r,t,p,k)	The constraint for an ETL technology that sets a lower bound on the continuous variable VAR_LAMBD(r,t,p,k).	EQU_EXT.ETL
EQ_LA2 (r,t,p,k)	The constraint for an ETL technology that sets an upper bound on the continuous variable VAR_LAMBD(r,t,p,k).	EQU_EXT.ETL
EQ_MRCLU (r,t,p)	Constraint that is generated only for the special clustered learning ETL option (TL_MRCLUST). For a key learning ETL technology it defines investment in new capacity (VAR_NCAP) as the weighted sum of investments in new capacity of the associated clustered technologies in multilple regions.	EQU_EXT.ETL
EQ_OBJSAL (r,cur)	For an ETL technology in periods appropriately close to the model horizon, part of the investment costs (VAR_IC) exceed the model horizon. This part of the investment cost is reflected in the calculation of the salvage value variable VAR_OBJSAL.	EQOBSALV.MOD
EQ_OBJINV (r,cur)	The endogenously calculated cost of investments for learning technologies (VAR_IC) needs to be discounted and included in the regional investment cost part of the objective function (EQ_OBJINV) in place of the traditional investment calculation using variable VAR_NCAP.	EQOBJINV.MOD

4.1 EQ_CC(r,t,p)

Description: The Cumulative Capacity Interpolation constraint for an ETL technology.

Purpose andThis constraint defines the cumulative investment in capacity for aOccurrence:technology in a period (VAR_CCAP) as the sum over all intervals k of the
continuous variables VAR_LAMBD(r,t,p,k) that represent cumulative
investment in capacity as lying in the kth interval of the piecewise linear
approximation of the cumulative cost curve. This constraint links the
cumulative capacity investment variable (VAR_CCAP) to the variables
VAR_LAMBD. In combination with other ETL constraints, it is
fundamental to ensuring the validity of the piecewise linear approximation
of the cumulative cost curve.

This equation is generated in each time period for which the ETL technology is available.

Units: Technology capacity units.

Type: Binding. The equation is an equality (=) constraint.

Interpretation of the results:

Primal:The level of this constraint must be zero in a feasible solution.Dual variable:The dual variables of mixed integer problems have limited usefulness, as

discussed in Section 10.3 of PART I.

Equation

$$EQ_CC_{r,t,p} \forall [(p \in teg) \land ((r,t,p) \in rtp)]$$

Cumulative investment in capacity in the current period.

$$\begin{array}{c} & V\!A\!R_CC\!A\!P_{r,t,p} \\ \{=\} \end{array}$$

Sum over all intervals k (in the piecewise linear approximation of the cumulative cost curve) of the ${\it continuous}$ variables ${\rm VAR_LAMBD}$ in the current period t.

$$\sum_{k} VAR _ LAMBD_{r,t,p,k}$$

4.2 EQ_CLU(r,t,p)

- **Description:** For a key learning ETL technology it defines investment in new capacity (VAR_NCAP) as the weighted sum of investments in new capacity of the attached clustered technologies. The weights used are the numeric values of the CLUSTER parameter.
- Purpose andDefines the relationship between investment in new capacity for a keyOccurrence:learning ETL technology and investment in new capacity for the
associated clustered technologies. This equation is generated in each time
period for which the ETL technology is available. It is a key learning
technology, that is, it has associated clustered technologies.
- Units: Money units, e.g., million 2000 US\$, or any other unit in which costs are tracked.
- **Type:** Binding. The equation is an equality (=) constraint.

Interpretation of the results:

 Primal:
 The level of this constraint must be zero in a feasible solution.

 Dual variable:
 The dual variable (DVR_CLU) of this constraint in the MIP solution is of little interest.

Remarks: Activation of the special <u>clustered</u> learning ETL option occurs automatically if data is included for the CLUSTER parameter.

Equation

{

$$EQ_CLU_{r,t,p} \forall \left[(p \in teg) \land (NTCHTEG_{r,p} > 0) \land \\ ((r,t,p) \in rtp) \right]$$

Investment in new capacity (for <u>key</u> learning technology $p \in teg$) in period t.

$$VAR _NCAP_{r,t,p}$$

The weighted sum of the investments in new capacity in period t of the clustered technologies p' attached to the <u>key</u> learning technology $p \in$ teg, and whose START period is less than or equal to t. The weights used are the numeric values of the CLUSTER parameter.

$$p' \left(\sum_{\substack{CLUSTER_{r,p,p'} > 0 \\ (r,t,p' \in rtp)}} (CLUSTER_{r,p,p'} * VAR _ NCAP_{r,t,p'}) \right)$$

4.3 EQ_COS(r,t,p)

Description: The Cumulative Cost Interpolation constraint for an ETL technology.

Purpose andThis constraint defines the interpolated cumulative cost of investment in
capacity for a technology in a period (VAR_CCOST) in terms of the
binary variables VAR_DELTA and the continuous variables
VAR_LAMBD, and the internal model parameters ALPH and BETA,
where ALPH and BETA represent the intercepts on the vertical axis and
the slopes, respectively, of the line segments in the piecewise linear
approximation of the cumulative cost curve. For a more precise definition,
see "Equation" below. In combination with other ETL constraints, it is
fundamental to ensuring the validity of the piecewise linear approximation
of the cumulative cost curve. This equation is generated in each time for
which the ETL technology is available.

- Units: Money units, e.g., million 2000 US\$, or any other unit in which costs are tracked.
- **Type:** Binding. The equation is an equality (=) constraint.

Interpretation of the results:

Primal:The level of this constraint must be zero in a feasible solution.Dual variable:The dual variables of mixed integer problems have limited usefulness, as
discussed in Section 10.3 of PART I.

Equation

$$EQ_COS_{r,t,p} \forall [(p \in teg) \land ((r,t,p) \in rtp)]$$

Interpolated cumulative cost of investment in capacity in the current period.

$$VAR_CCOST_{r,t,p}$$

Sum over all intervals k (in the piecewise linear approximation of the cumulative cost curve) of ALPH times the *binary* variable VAR_DELTA plus BETA times the *continuous* variable VAR_LAMBD, for the current period t, where ALPH and BETA represent the intercepts on the vertical axis and the slopes, respectively, of the kth interval.

$$\sum_{k} (ALPH_{k,p} *VAR_DELTA_{r,t,p,k} + BETA_{k,p} *VAR_LAMBD_{r,t,p,k})$$

4.4 EQ_CUINV(r,t,p)

Description: The Cumulative Capacity Definition constraint for an ETL technology.

Purpose and	This constraint defines the cumulative investment in capacity of a		
Occurrence:	technology in a period (VAR_CCAP) as the initial cumulative capacity		
	(CCAP0) plus the sum of investments in new capacity made up to and		
	including this period. This equation is generated in each time period for which the ETL technology is available.		

Units: Technology capacity units.

Type: Binding. The equation is an equality (=) constraint.

Interpretation of the results:

 Primal:
 The level of this constraint must be zero in a feasible solution.

 Dual variable:
 The dual variables of mixed integer problems have limited usefulness, as mentioned above.

Equation

$$EQ_CUINV_{r,t,p} \forall [(p \in teg) \land ((r,t,p) \in rtp)]$$

Cumulative investment in capacity in the current period.

 $V\!A\!R_CC\!A\!P_{r,t,p} \ \{=\}$

Cumulative investment in capacity at the start of the learning process.

 $CCAP0_{r,p} +$

u

Sum of the investments made since the technology is first available.

$$\sum_{ertp_{r,u,p} \land u \leq t} VAR _ NCAP_{r,u,p}$$

4.5 **EQ_DEL**(**r**,**t**,**p**)

Description:	The constraint for an ETL technology that ensures that in each time period there is exactly one interval k for which the <i>binary</i> variable VAR_DELTA(r,t,p,k) has value 1 (with zero values for intervals \neq k).
Purpose and Occurrence:	To ensure that only one of the <i>binary</i> variable VAR_DELTA(r,t,p,k) has value 1 for each technology. This constraint, in combination with other ETL constraints, is fundamental to ensuring the validity of the piecewise linear approximation of the cumulative cost curve. This equation is generated in each time period for which the ETL technology is available.
Units:	None.

Type: Binding. The equation is an equality (=) constraint.

Interpretation of the results:

Primal:The level of this constraint must be 1 in a feasible solution.Dual variable:The dual variables of mixed integer problems have limited usefulness, as
already mentioned.

Equation

$$EQ_DEL_{r,t,p} \forall [(p \in teg) \land ((r,t,p) \in rtp)]$$

Sum over all intervals k (in the piecewise linear approximation of the cumulative cost curve) of the *binary* variables VAR_DELTA in the current period t.

$$\sum_{k} VAR _ DELTA_{r,t,p,k}$$

$$\models \} 1$$

4.6 EQ_EXPE1(r,t,p,k)

Description:	One of two constraints for an ETL technology to improve MIP solution time by reducing the domain of feasibility of the binary variables VAR_DELTA.
Purpose and Occurrence:	To improve MIP solution time this constraint takes advantage of the observation that cumulative investment is increasing with time, thus ensuring that if the cumulative investment in period t lies in segment k, then it will not lie in segments k-1, k-2,, 1 in period t+1. This equation is generated for each ETL technology in each time period, for which the technology is available, and excluding the final period (TLAST), and for each interval k in the piecewise linear approximation of the cumulative cost curve.
Units:	None.
Туре:	<i>Binding</i> . The equation is a greater than or equal to (\geq) constraint.

Interpretation of the results:

Primal:	The level of this constraint must be greater than or equal to zero in a
	feasible solution.
Dual variable:	The dual variables of mixed integer problems have limited usefulness, as
	already mentioned.

Equation

$$EQ_EXPE1_{r,t,p,k} \forall [(p \in teg) \land ((r,t,p) \in rtp) \land (t < TLAST)]$$

Sum over intervals $j \leq k$ of binary variables $\mathrm{VAR_DELTA}(r,t,p,j),$ for the k^{th} interval, in period t.

$$\sum_{j \le k} (VAR_DELTA_{r,t,p,j})$$

{≥}

Sum over intervals $j \leq k$ of binary variables <code>VAR_DELTA(r,t,p,j)</code>, for the k^{th} interval, in period t+1.

$$\sum_{j \le k} (VAR_DELTA_{r,t+1,p,j})$$

4.7 EQ_EXPE2(r,t,p,k)

Description:	Second of two constraints for an ETL technology to improve MIP solution time by reducing the domain of feasibility of the binary variables VAR_DELTA. Both constraints rely on the observation that cumulative investment is increasing as time goes on.
Purpose and Occurrence:	To improve MIP solution times this constraint is derived from the observation that if cumulative investment in period t lies in segment k, then it must lie in segment k or $k+1$ or $k+2$ etc in period $t+1$.
	This equation is generated for each ETL technology in each time period, for which the technology is available, and excluding the final period (TLAST), and for each interval k in the piecewise linear approximation of the cumulative cost curve.
Units:	None.
Туре:	<i>Binding</i> . The equation is a less than or equal to (\leq) constraint.
Interpretatio Primal:	n of the results: The level of this constraint must be less than or equal to zero in a feasible

solution. Dual variable: The dual variables of mixed integer problems have limited usefulness, as already mentioned.

Equation

$$EQ_EXPE2_{r,t,p,k} \forall [(p \in teg) \land ((r,t,p) \in rtp) \land (t < TLAST)]$$

Sum over intervals $j \geq k$ of binary variables $\mathrm{VAR_DELTA}(r,t,p_xj),$ for the k^{th} interval, in period t.

$$\sum_{\substack{j \ge k}} (VAR _ DELTA_{r,t,p,j})$$
 { \leq }

Sum over intervals $j \geq k$ of binary variables <code>VAR_DELTA(r,t,p,j)</code>, for the k^{th} interval, in period t+1.

$$\sum_{j \ge k} (VAR _ DELTA_{r,t+1,p,j})$$

4.8 EQ_IC1(r,t,p)

Description: The constraint for an ETL technology that defines the portion of the cumulative cost of investment in capacity (VAR_IC) that is incurred in period t, where t = first period of model horizon.

Purpose andTo determine the variable VAR_IC which represents the currentOccurrence:investment cost incurred in the first period a learning technology is
available according to the cumulative investments made in that period.
VAR_IC then enters the regional investment cost part of the objective
function (EQ_OBJINV) subject to the same discounting that applies to
other period t investment costs. This equation is generated for the first
period of the model horizon.

- Units: Money units, e.g., million 2000 US\$, or any other unit in which costs are tracked.
- **Type:** Binding. The equation is an equality (=) constraint.

Interpretation of the results:

 Primal:
 The level of this constraint must be zero in a feasible solution.

 Dual variable:
 The dual variables of mixed integer problems have limited usefulness, as already mentioned.

Equation

$$EQ_ICl_{rtn} \forall (p \in teg) \land (t = MIYR_V1)$$

The portion of the cumulative cost of investment in capacity that is incurred in period t, in this case the first period the technology is available.

$$VAR_IC_{r,t,p}$$

ł

The cumulative cost of investment in new capacity in the first period t (t = $MIYR_V1).$

 $VAR_CCOST_{r,t,p}$ –

The initial cumulative cost of investment in new capacity for a learning technology.

CCOST0

4.9 EQ_IC2(r,t,p)

Description:	The constraint for an ETL technology that defines the portion of the cumulative cost of investment in capacity that is incurred in each period t other than the first period.
Purpose and Occurrence:	To determine the variable VAR_IC which represents the current investment cost incurred in period t according to the cumulative investments made thus far, where VAR_IC then enters the regional investment cost part of the objective function (EQ_OBJINV) subject to the same discounting that applies to other period t investment costs. This equation is generated in each time period other than the first period of the model horizon.
Units:	Money units, e.g., million 2000 US\$, or any other unit in which costs are tracked.
Туре:	Binding. The equation is an equality (=) constraint.

Interpretation of the results:

Primal:The level of this constraint must be zero in a feasible solution.Dual variable:The dual variables of mixed integer problems have limited usefulness, as
already mentioned.

Equation

$$EQ_IC2_{r,t,p} \forall (p \in teg) \land (t > MIYR_V1)$$

The portion of the cumulative cost of investment in capacity that is incurred in period t.

 $VAR_IC_{r,t,p}$

{=}

The cumulative cost of investment in new capacity as of period t.

 $VAR_CCOST_{r,t,p}$ –

The cumulative cost of investment in new capacity as of the previous period t-1.

 $VAR_CCOST_{r,t-1,p}$

4.10 EQ_LA1(r,t,p,k)

Description: The constraint for an ETL technology that sets a lower bound on the continuous variable VAR_LAMBD(r,t,p,k).

Purpose and
Occurrence:To set the lower bound for VAR_LAMBD(r,t,p,k) to CCAPK(r,k-1,p) *
VAR_DELTA, where CCAPK(r,k-1,p) is the left hand end of the kth
interval and VAR_DELTA = VAR_DELTA(r,t,p,k) is the binary variable
associated with VAR_LAMBD(r,t,p,k). If binary variable VAR_DELTA
= 1, the effect is to set a lower bound on variable VAR_LAMBD(r,t,p,k)
of CCAPK(r,k-1,p), whereas if VAR_DELTA = 0 the effect is to set a
lower bound of 0. This constraint, in combination with other ETL
constraints, is fundamental to ensuring the validity of the piecewise linear
approximation of the cumulative cost curve.

This equation is generated in each time period, for which the ETL technology is available, and for each interval k in the piecewise linear approximation of the cumulative cost curve.

- Units: Technology capacity units.
- **Type:** Binding. The equation is a greater than or equal to (\geq) constraint.

Interpretation of the results:

- *Primal:* The level of this constraint must be greater than or equal to zero in a feasible solution.
- *Dual variable:* The dual variables of mixed integer problems have limited usefulness, as already mentioned.

Equation

$$EQ_LA1_{r,t,p,k} \forall [(p \in teg) \land ((r,t,p) \in rtp)]$$

Portion of the cumulative investment in capacity that lies in the kth interval (of the piecewise linear approximation of the cumulative cost curve), in the current period.

$$VAR_LAMBD_{r,t,\,p,k}$$

{≥}

Left hand end of the k^{th} interval (CCAPK(r,k-1,p)) times binary variable VAR_DELTA(r,t,p,k), in the current period.

$$CCAPK_{r,k-1,p} * VAR _ DELTA_{r,t,p,k}$$

4.11 EQ_LA2(r,t,p,k)

Description: The constraint for an ETL technology that sets an upper bound on the continuous variable VAR_LAMBD(r,t,p,k).

Purpose andTo set the upper bound of VAR_LAMBD(r,t,p,k) to CCAPK(r,k,p) *Occurrence:VAR_DELTA, where CCAPK(r,k,p) is the right hand end of the kth
interval and VAR_DELTA = VAR_DELTA(r,t,p,k) is the binary variable
associated with VAR_LAMBD(r,t,p,k). If binary variable VAR_DELTA
= 1, the effect is to set an upper bound on variable VAR_LAMBD(r,t,p,k)
of CCAPK(r,k,p), whereas if VAR_DELTA = 0 the effect is to set an
upper bound of 0. This constraint, in combination with other ETL
constraints, is fundamental to ensuring the validity of the piecewise linear
approximation of the cumulative cost curve.

This equation is generated in each time period, for which the ETL technology is available, and for each interval k in the piecewise linear approximation of the cumulative cost curve.

Units: Technology capacity units.

Type: Binding. The equation is a less than or equal to (\leq) constraint.

Interpretation of the results:

Primal: The level of this constraint must be less than or equal to zero in a feasible solution.

Dual variable: The dual variable (DVR_LA2) of this constraint in the MIP solution is of little interest.

Equation

$$MR_LA2_{r,t,p,k} \forall [(p \in teg) \land ((r,t,p) \in rtp)]$$

Portion of the cumulative investment in capacity that lies in the kth interval (of the piecewise linear approximation of the cumulative cost curve), in the current period.

$$VAR_LAMBD_{r,t,p,k}$$

{≤}

Right hand end of the k^{th} interval (CCAPK(r,k,p)) times binary variable $R_DELTA(r,t,p,k),$ in the current period.

$$CCAPK_{r,k,p} * VAR _ DELTA_{r,t,p,k}$$

4.12 EQ_MRCLU(r,t,p)

Description: For a key learning ETL technology it defines investment in new capacity (VAR_NCAP) as the weighted sum of investments in new capacity of the attached clustered technologies in multiple regions. The weights used are the numeric values of the TL_MRCLUST parameter.

Purpose and Defines the relationship between investment in new capacity for a key

- Occurrence: learning ETL technology and investment in new capacity for the associated clustered technologies. This equation is generated in each time period for which the ETL technology is available. It is a key learning technology, that is, it has associated clustered technologies, possibly in multiple regions.
- Units: Money units, e.g., million 2010 US\$, or any other unit in which costs are tracked.

Type: Binding. The equation is an equality (=) constraint.

Interpretation of the results:

Primal:The level of this constraint must be zero in a feasible solution.Dual variable:The dual variable of this constraint in the MIP solution is of little interest.

Remarks: Activation of the special <u>clustered</u> learning ETL option occurs automatically if data is included for the TL_MRCLUST parameter.

Equation

$$EQ_MRCLU_{r,t,p} \forall \begin{bmatrix} (p \in teg) \land (TL_RP_KC_{r,p}) \land \\ ((r,t,p) \in rtp) \end{bmatrix}$$

Investment in new capacity (for key learning technology $p \in teg$) in period t.

$$VAR _NCAP_{r,t,p}$$

{=

The weighted sum of the investments in new capacity in period t of the clustered technologies p' attached to the key learning technology $p \in$ teg, and whose START period is less than or equal to t. The weights used are the numeric values of the CLUSTER parameter.

$$\sum_{(reg,t,pro \in rtp)} (TL_MRCLUST_{reg,p,prc} \times VAR_NCAP_{reg,t,prc})$$

4.13 EQ_OBJSAL(r,cur)

Description:	Regional salvage value part of objective function adjusted to include the salvage value of endogenously determined investments (VAR_IC) in learning technologies. A salvage value for a learning technology investment exists when the technical lifetime of the investment exceeds the model horizon.
Purpose and Ocurrence:	The objective function part calculating the salvage value is changed (for learning technologies only) by replacing the traditional calculation of the salvage value of investments with one based on the investment costs of learning technologies (VAR_IC).
Units:	Money units, e.g., million 2000 US\$, or any other unit in which costs are tracked.
Туре:	<i>Binding</i> . The equation is an equality (=) constraint.

Equation

 $EQ_OBJSAL_{r,cur}$

All the basic objective function term for calculating the salvage value (section 5.2.8)

•••

The calculated salvage value associated with the ETL technologies. The internally derived parameter coefficient OBJSIC describing the portion of the investment costs that has to be salvaged. It takes into account the discounting of the salvage value.

$$+ \sum_{t,p \in teg} \left[OBJSIC_{r,t,p} * VAR_{IC_{r,t,p}} \right]$$

4.14 EQ_OBJINV(r,cur)

- see EQ_OBJINV in section 5.2.2 for a general description without ETL

Description: Regional investment cost part of objective function adjusted to include the endogenously determined investment cost (VAR_IC) for new investments in learning technologies.

Purpose andThe objective function part calculating the investment costs is changedOccurrence:(for learning technologies only) by replacing the traditional calculation of
discounted cost of investments in new capacity with that of the
endogenously determined value. This equation is generated for each
region where the learning investment costs occur in each time period
beginning from the period, for which the ETL technology is available.

Equation

EQ_OBJINV_{r.cur}

All the basic objective function terms for investment costs (section 5.2.2)

• • •

The calculated investments costs associated with the ETL technologies.

 $+ \sum_{t,p \in teg} \left[DISC_{r,t,p} * VAR _ IC_{r,t,p} \right]$

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
Agriculture stationary combustion, coal	2010	PM10	AGR	0.117	kilotonnes	PJ	NAEI 2013
Agriculture stationary combustion, coal	2010	PM2.5	AGR	0.109	kilotonnes	PJ	NAEI 2013
Agriculture stationary combustion, coal	2010	NH3	AGR	0.000	kilotonnes	PJ	NAEI 2013
Agriculture stationary combustion, coal	2010	SO2	AGR	0.655	kilotonnes	PJ	NAEI 2013
Agriculture stationary combustion, coal	2010	NOx	AGR	0.179	kilotonnes	PJ	NAEI 2013
Agriculture stationary combustion, coal	2010	Non Methane VOC	AGR	0.002	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, natural gas	2010	PM10	AGR	0.001	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, natural gas	2010	PM2.5	AGR	0.001	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, natural gas	2010	NOx	AGR	0.044	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, natural gas	2010	Non Methane VOC	AGR	0.0	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, straw	2010	PM10	AGR	0.732	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, straw	2010	PM2.5	AGR	0.718	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, straw	2010	NH3	AGR	0.000	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, straw	2010	SO2	AGR	0.000	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, straw	2010	NOx	AGR	0.089	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, straw	2010	Non Methane VOC	AGR	0.002	kilotonnes	PJ	NAEI 2013

A.2: Emission Factors – National Scale (United Kingdom) Analysis

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Agriculture - stationary combustion, fuel oil	2010	PM10	AGR	0.0243	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, fuel oil	2010	PM2.5	AGR	0.0243	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, fuel oil	2010	SO2	AGR	0.3526	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, fuel oil	2010	NOx	AGR	0.1867	kilotonnes	PJ	NAEI 2013
Agriculture - stationary combustion, fuel oil	2010	Non Methane VOC	AGR	0.0033	kilotonnes	PJ	NAEI 2013
Agriculture - mobile machinery, gas oil	2010	PM10	AGR	0.0575	kilotonnes	PJ	NAEI 2013
Agriculture - mobile machinery, gas oil	2010	PM2.5	AGR	0.0546	kilotonnes	PJ	NAEI 2013
Agriculture - mobile machinery, gas oil	2010	NH3	AGR	0.0008	kilotonnes	PJ	NAEI 2013
Agriculture - mobile machinery, gas oil	2010	SO2	AGR	0.0335	kilotonnes	PJ	NAEI 2013
Agriculture - mobile machinery, gas oil	2010	NOx	AGR	0.5978	kilotonnes	PJ	NAEI 2013
Agriculture - mobile machinery, gas oil	2010	Non Methane VOC	AGR	0.1114	kilotonnes	PJ	NAEI 2013
Industrial off-road mobile machinery (weighted average)	2010	NOx	IND	0.755	kilotonnes	PJ	NAEI 2013
Industrial off-road mobile machinery (weighted average)	2010	Non Methane VOC	IND	0.237	kilotonnes	PJ	NAEI 2013
Industrial off-road mobile machinery	2010	PM10	IND	0.068	kilotonnes	PJ	NAEI 2013
Industrial off-road mobile machinery	2010	PM2.5	IND	0.064	kilotonnes	РЈ	NAEI 2013
Industrial off-road mobile machinery	2010	SO2	IND	0.026	kilotonnes	РЈ	NAEI 2013
Industrial off-road mobile machinery (weighted average)	2010	NH3	IND	0.001	kilotonnes	PJ	NAEI 2013
Domestic combustion, coke	2010	PM10	RES	0.060	kilotonnes	PJ	NAEI 2013
Domestic combustion, coke	2010	PM2.5	RES	0.059	kilotonnes	PJ	NAEI 2013
Domestic combustion, coke	2010	NH3	RES	0.035	kilotonnes	PJ	NAEI 2013
Domestic combustion, coke	2010	SO2	RES	0.531	kilotonnes	PJ	NAEI 2013
Domestic combustion, coke	2010	NOx	RES	0.141	kilotonnes	PJ	NAEI 2013
Domestic combustion, coke	2010	Non Methane VOC	RES	0.173	kilotonnes	PJ	NAEI 2013
Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
-------------------------------------	------	-----------------------	--------	----------	------------	----------	--------------------------
technology				Factor		Units	
Domestic combustion, coal	2010	PM10	RES	0.324	kilotonnes	PJ	NAEI 2013
Domestic combustion, coal	2010	PM2.5	RES	0.319	kilotonnes	PJ	NAEI 2013
Domestic combustion, coal	2010	NH3	RES	0.035	kilotonnes	PJ	NAEI 2013
Domestic combustion, coal	2010	SO2	RES	0.631	kilotonnes	PJ	NAEI 2013
Domestic combustion, coal	2010	NOx	RES	0.124	kilotonnes	PJ	NAEI 2013
Domestic combustion, coal	2010	Non Methane VOC	RES	0.494	kilotonnes	PJ	NAEI 2013
Domestic combustion, SSF	2010	PM10	RES	0.057	kilotonnes	PJ	NAEI 2013
Domestic combustion, SSF	2010	PM2.5	RES	0.057	kilotonnes	PJ	NAEI 2013
Domestic combustion, SSF	2010	NH3	RES	0.000	kilotonnes	PJ	NAEI 2013
Domestic combustion, SSF	2010	SO2	RES	0.565	kilotonnes	PJ	NAEI 2013
Domestic combustion, SSF	2010	NOx	RES	0.134	kilotonnes	PJ	NAEI 2013
Domestic combustion, SSF	2010	Non Methane VOC	RES	0.173	kilotonnes	PJ	NAEI 2013
Domestic combustion, burning oil	2010	PM10	RES	0.0032	kilotonnes	PJ	NAEI 2013
Domestic combustion, burning oil	2010	PM2.5	RES	0.0032	kilotonnes	PJ	NAEI 2013
Domestic combustion, burning oil	2010	SO2	RES	0.012	kilotonnes	PJ	NAEI 2013
Domestic combustion, burning oil	2010	NH3	RES	0.000	kilotonnes	PJ	NAEI 2013
Domestic combustion, burning oil	2010	NOx	RES	0.074	kilotonnes	PJ	NAEI 2013
Domestic combustion, burning oil	2010	Non Methane VOC	RES	0.001	kilotonnes	PJ	NAEI 2013
Domestic combustion, LPG	2010	PM10	RES	0.004	kilotonnes	PJ	NAEI 2013
Domestic combustion, LPG	2010	PM2.5	RES	0.004	kilotonnes	PJ	NAEI 2013
Domestic combustion, LPG	2010	SO2	RES	0.000	kilotonnes	PJ	IIR, 2011 Denmark
Domestic combustion, LPG	2010	NH3	RES	0.000	kilotonnes	PJ	IIR, 2011 Switzerland
Domestic combustion, LPG	2010	NOx	RES	0.071	kilotonnes	PJ	NAEI 2013
Domestic combustion, LPG	2010	Non Methane VOC	RES	0.004	kilotonnes	PJ	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Domestic combustion, natural gas	2010	PM10	RES	0.0006	kilotonnes	PJ	NAEI 2013
Domestic combustion, natural gas	2010	PM2.5	RES	0.0006	kilotonnes	PJ	NAEI 2013
Domestic combustion, natural gas	2010	SO2	RES	0.000	kilotonnes	PJ	IIR, 2011 Denmark
Domestic combustion, natural gas	2010	NH3	RES	0.000	kilotonnes	PJ	IIR, 2011 Switzerland
Domestic combustion, natural gas	2010	NOx	RES	0.025	kilotonnes	PJ	NAEI 2013
Domestic combustion, natural gas	2010	Non Methane VOC	RES	0.002	kilotonnes	PJ	NAEI 2013
Domestic combustion, wood	2010	PM10	RES	0.624	kilotonnes	PJ	NAEI 2013
Domestic combustion, wood	2010	PM2.5	RES	0.583	kilotonnes	PJ	NAEI 2013
Domestic combustion, wood	2010	NH3	RES	0.075	kilotonnes	PJ	NAEI 2013
Domestic combustion, wood	2010	SO2	RES	0.008	kilotonnes	PJ	NAEI 2013
Domestic combustion, wood	2010	NOx	RES	0.066	kilotonnes	PJ	NAEI 2013
Domestic combustion, wood	2010	Non Methane VOC	RES	0.536	kilotonnes	PJ	NAEI 2013
Domestic combustion, gas oil	2010	PM10	RES	0.003	kilotonnes	PJ	NAEI 2013
Domestic combustion, gas oil	2010	PM2.5	RES	0.003	kilotonnes	PJ	NAEI 2013
Domestic combustion, gas oil	2010	NH3	RES	0.000	kilotonnes	PJ	NAEI 2013
Domestic combustion, gas oil	2010	SO2	RES	0.034	kilotonnes	PJ	NAEI 2013
Domestic combustion, gas oil	2010	NOx	RES	0.074	kilotonnes	PJ	NAEI 2013
Domestic combustion, gas oil	2010	Non Methane VOC	RES	0.001	kilotonnes	PJ	NAEI 2013
Public sector combustion, coal	2010	PM10	SER	0.104	kilotonnes	PJ	NAEI 2013
Public sector combustion, coal	2010	PM2.5	SER	0.096	kilotonnes	PJ	NAEI 2013
Public sector combustion, coal	2010	NH3	SER	0.000	kilotonnes	PJ	NAEI 2013
Public sector combustion, coal	2010	SO2	SER	0.718	kilotonnes	PJ	NAEI 2013
Public sector combustion, coal	2010	NOx	SER	0.190	kilotonnes	PJ	NAEI 2013
Public sector combustion, coal	2010	Non Methane VOC	SER	0.002	kilotonnes	PJ	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Public sector combustion, sewage gas	2010	PM10	SER	0.006	kilotonnes	PJ	NAEI 2013
Public sector combustion, sewage gas	2010	PM2.5	SER	0.006	kilotonnes	PJ	NAEI 2013
Public sector combustion, sewage gas	2010	NH3	SER	NE	kilotonnes	PJ	ATSDR, 2001
Public sector	2010	SO2	SER	0.000	kilotonnes	PJ	NAEI 2013
Public sector combustion, sewage gas	2010	NOx	SER	0.267	kilotonnes	PJ	NAEI 2013
Public sector combustion, sewage gas	2010	Non Methane VOC	SER	0.003	kilotonnes	PJ	NAEI 2013
Public sector combustion, natural gas	2010	PM10	SER	0.001	kilotonnes	PJ	NAEI 2013
Public sector combustion, natural gas	2010	PM2.5	SER	0.001	kilotonnes	PJ	NAEI 2013
Public sector combustion, natural gas	2010	NH3	SER	0.000	kilotonnes	PJ	NAEI 2013
Public sector combustion, natural gas	2010	SO2	SER	0.000	kilotonnes	PJ	NAEI 2013
Public sector combustion, natural gas	2010	NOx	SER	0.061	kilotonnes	PJ	NAEI 2013
Public sector combustion, natural gas	2010	Non Methane VOC	SER		kilotonnes	PJ	NAEI 2013
Industry, coal (weighted average)	2010	PM10	IND	0.027	kilotonnes	PJ	NAEI 2013
Industry, coal (weighted average)	2010	PM2.5	IND	0.025	kilotonnes	PJ	NAEI 2013
Industry, coal (weighted average)	2010	NH3	IND	0.000	kilotonnes	PJ	NAEI 2013
Industry, coal (weighted average)	2010	SO2	IND	0.661	kilotonnes	PJ	NAEI 2013
Industry, coal (weighted average)	2010	NOx	IND	0.186	kilotonnes	PJ	NAEI 2013
Industry, coal (weighted average)	2010	Non Methane VOC	IND	0.002	kilotonnes	PJ	NAEI 2013
Industry, coke (weighted average)	2010	PM10	IND	0.027	kilotonnes	PJ	NAEI 2013
Industry, coke (weighted average)	2010	PM2.5	IND	0.025	kilotonnes	PJ	NAEI 2013
Industry, coke (weighted average)	2010	NH3	IND	0.000	kilotonnes	PJ	NAEI 2013
Industry, coke (weighted average)	2010	SO2	IND	0.671	kilotonnes	PJ	NAEI 2013
Industry, coke (weighted average)	2010	NOx	IND	0.031	kilotonnes	PJ	NAEI 2013
Industry, coke (weighted average)	2010	Non Methane VOC	IND	0.002	kilotonnes	PJ	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Industry, fuel oil (weighted average)	2010	PM10	IND	0.025	kilotonnes	PJ	NAEI 2013
Industry, fuel oil (weighted average)	2010	PM2.5	IND	0.025	kilotonnes	PJ	NAEI 2013
Industry, fuel oil (weighted average)	2010	NH3	IND	0.000	kilotonnes	PJ	IIR Switzerland, 2013
Industry, fuel oil (weighted average)	2010	SO2	IND	0.353	kilotonnes	PJ	NAEI 2013
Industry, fuel oil (weighted average)	2010	NOx	IND	0.222	kilotonnes	PJ	NAEI 2013
Industry, fuel oil (weighted average)	2010	Non Methane VOC	IND	0.001	kilotonnes	PJ	NAEI 2013
Other industrial combustion, LPG	2010	PM10	IND	0.004	kilotonnes	PJ	NAEI 2013
Other industrial combustion, LPG	2010	PM2.5	IND	0.004	kilotonnes	PJ	NAEI 2013
Other industrial combustion, LPG	2010	NH3	IND	0.000	kilotonnes	PJ	IIR, 2011 Switzerland
Other industrial combustion, LPG	2010	SO2	IND	0.000	kilotonnes	PJ	IIR, 2011 Denmark
Other industrial combustion, LPG	2010	NOx	IND	0.073	kilotonnes	PJ	NAEI 2013
Other industrial combustion, LPG	2010	Non Methane VOC	IND	0.004	kilotonnes	PJ	NAEI 2013
Industry, coke oven gas (weighted average)	2010	PM10	IND	0.001	kilotonnes	PJ	NAEI 2013
Industry, coke oven gas (weighted average)	2010	PM2.5	IND	0.001	kilotonnes	PJ	NAEI 2013
Industry, coke oven gas (weighted average)	2010	NH3	IND	NE	kilotonnes	PJ	
Industry, coke oven gas (weighted average)	2010	SO2	IND	0.416	kilotonnes	PJ	NAEI 2013
Industry, coke oven gas (weighted average)	2010	NOx	IND	0.082	kilotonnes	PJ	NAEI 2013
Industry, coke oven gas (weighted average)	2010	Non Methane VOC	IND	NE	kilotonnes	PJ	
Industrial sector combustion of natural gas (weighted average)	2010	PM10	IND	0.002	kilotonnes	PJ	NAEI 2013
Industrial sector combustion of natural gas (weighted average)	2010	PM2.5	IND	0.002	kilotonnes	PJ	NAEI 2013
Industrial sector combustion of natural gas (weighted average)	2010	SO2	IND	0.000	kilotonnes	PJ	NAEI 2013
Industrial sector combustion of natural gas (weighted average)	2010	NOx	IND	0.107	kilotonnes	PJ	NAEI 2013
Industrial sector combustion of natural gas (weighted average)	2010	Non Methane VOC	IND	0.002	kilotonnes	PJ	NAEI 2013
Biogas from anaerobic digestion for the industry sector	2010	NH3	IND	NE	kilotonnes	PJ	ATSDR, 2001

technologyFactorUnitsMiscellaneous industrial/commercial combustion, MSW2010PM10IND0.001kilotonnesPJMiscellaneous industrial/commercial2010PM2.5IND0.001kilotonnesPJ	NAEI 2013 NAEI 2013
Miscellaneous industrial/commercial combustion, MSW 2010 PM10 IND 0.001 kilotonnes PJ Miscellaneous industrial/commercial 2010 PM2.5 IND 0.001 kilotonnes PJ	NAEI 2013 NAEI 2013
Miscellaneous 2010 PM2.5 IND 0.001 kilotonnes PJ industrial/commercial <	NAEI 2013
combustion, MSW	
Miscellaneous 2010 NH3 IND 0.002 kilotonnes PJ industrial/commercial combustion, MSW	NAEI 2013
Miscellaneous 2010 SO2 IND 0.003 kilotonnes PJ industrial/commercial combustion, MSW	NAEI 2013
Miscellaneous 2010 NOx IND 0.094 kilotonnes PJ industrial/commercial combustion, MSW IND 0.094 kilotonnes PJ	NAEI 2013
Miscellaneous industrial/commercial combustion, MSW 2010 Non IND 0.001 kilotonnes PJ	NAEI 2013
Other industrial 2010 PM10 IND 0.033 kilotonnes PJ combustion, wood	NAEI 2013
Other industrial 2010 PM2.5 IND 0.033 kilotonnes PJ combustion, wood <td< td=""><td>NAEI 2013</td></td<>	NAEI 2013
Other industrial 2010 NH3 IND 0.000 kilotonnes PJ combustion, wood	NAEI 2013
Other industrial 2010 SO2 IND 0.010 kilotonnes PJ combustion, wood <	NAEI 2013
Other industrial 2010 NOx IND 0.085 kilotonnes PJ combustion, wood <	NAEI 2013
Other industrial 2010 Non IND 0.030 kilotonnes PJ combustion, wood Methane VOC VOC VOC	NAEI 2013
Other industrial combustion, burning oil 2010 PM10 IND 0.004 kilotonnes PJ	NAEI 2013
Other industrial 2010 PM2.5 IND 0.004 kilotonnes PJ combustion, burning oil	NAEI 2013
Other industrial combustion, burning oil 2010 NH3 IND NE kilotonnes PJ	NAEI 2013
Other industrial combustion, burning oil 2010 SO2 IND 0.012 kilotonnes PJ	NAEI 2013
Other industrial combustion, burning oil 2010 NOx IND 0.076 kilotonnes PJ	NAEI 2013
Other industrial combustion, burning oil 2010 Non IND 0.001 kilotonnes PJ VOC VOC </td <td>NAEI 2013</td>	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Miscellaneous industrial/commercial combustion (of landfill gas)	2010	PM10	IND	0.011	kilotonnes	PJ	NAEI 2013
Miscellaneous industrial/commercial combustion (of landfill gas)	2010	PM2.5	IND	0.011	kilotonnes	PJ	NAEI 2013
Miscellaneous industrial/commercial combustion (of landfill gas)	2010	NH3	IND	NE	kilotonnes	PJ	ATSDR, 2001
Miscellaneous industrial/commercial combustion (of landfill gas)	2010	SO2	IND	0.031	kilotonnes	PJ	NAEI 2013
Miscellaneous industrial/commercial combustion (of landfill gas)	2010	NOx	IND	0.273	kilotonnes	PJ	NAEI 2013
Miscellaneous industrial/commercial combustion (of landfill gas)	2010	Non Methane VOC	IND	0.004	kilotonnes	PJ	NAEI 2013
Stationary combustion in manufacturing industries and construction: Other (coke oven gas)	2010	PM10	IND	0.001	kilotonnes	PJ	NAEI 2013
Stationary combustion in manufacturing industries and construction: Other (coke oven gas)	2010	PM2.5	IND	0.001	kilotonnes	PJ	NAEI 2013
Stationary combustion in manufacturing industries and construction: Other (coke oven gas)	2010	NH3	IND	0.002	kilotonnes	PJ	Sowa et al. 2009
Stationary combustion in manufacturing industries and construction: Other (coke oven gas)	2010	SO2	IND	0.416	kilotonnes	PJ	NAEI 2013
Stationary combustion in manufacturing industries and construction: Other (coke oven gas)	2010	NOx	IND	0.082	kilotonnes	PJ	NAEI 2013
Stationary combustion in manufacturing industries and construction: Other (coke oven gas)	2010	Non Methane VOC	IND	0.025	kilotonnes	PJ	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Domestic combustion, petroleum coke	2010	PM10	RES	0.090	kilotonnes	PJ	NAEI 2013
Domestic combustion, petroleum coke	2010	PM2.5	RES	0.089	kilotonnes	PJ	NAEI 2013
Domestic combustion, petroleum coke	2010	NH3	RES	0.000	kilotonnes	PJ	IIR, 2011 Switzerland
Domestic combustion, petroleum coke	2010	SO2	RES	4.215	kilotonnes	PJ	NAEI 2013
Domestic combustion, petroleum coke	2010	NOx	RES	0.100	kilotonnes	PJ	NAEI 2013
Domestic combustion, petroleum coke	2010	Non Methane VOC	RES	0.144	kilotonnes	PJ	NAEI 2013
Industrial sector combustion of natural gas (weighted average)	2010	NH3	IND	0.000	kilotonnes	PJ	IIR, 2011 Switzerland
Power stations, coal	2010	PM10	ELC	0.004	kilotonnes	PJ	NAEI 2013
Power stations, coal	2010	PM2.5	ELC	0.002	kilotonnes	PJ	NAEI 2013
Power stations, coal	2010	NH3	ELC	0.000	kilotonnes	PJ	NAEI 2013
Power stations, coal	2010	SO2	ELC	0.166	kilotonnes	PJ	NAEI 2013
Power stations, coal	2010	NOx	ELC	0.175	kilotonnes	PJ	NAEI 2013
Power stations, coal	2010	Non Methane VOC	ELC	0.001	kilotonnes	PJ	NAEI 2013
Power stations, coal CCS	2010	PM10	ELC	0.004	kilotonnes	PJ	EEA CCS 2011
Power stations, coal CCS	2010	PM2.5	ELC	0.002	kilotonnes	PJ	EEA CCS 2011
Power stations, coal CCS	2010	NH3	ELC	0.000	kilotonnes	PJ	EEA CCS 2011
Power stations, coal CCS	2010	SO2	ELC	0.025	kilotonnes	PJ	EEA CCS 2011
Power stations, coal CCS	2010	NOx	ELC	0.175	kilotonnes	PJ	EEA CCS 2011
Power stations, coal CCS	2010	Non Methane VOC	ELC	0.001	kilotonnes	PJ	EEA CCS 2011
Electricity sector blast furnace gas	2010	NH3	ELC	NE	kilotonnes	PJ	Berdowski, Pacyna and Woodfield, 1999
Electricity sector blast furnace gas	2010	SO2	ELC	0.000	kilotonnes	PJ	EMEP / EEA, 2013 Energy p17
Electricity sector blast furnace gas	2010	NOx	ELC	0.089	kilotonnes	PJ	EMEP / EEA, 2013 Energy p17
Electricity sector blast furnace gas	2010	Non Methane VOC	ELC	0.003	kilotonnes	PJ	EMEP / EEA, 2013 Energy p17
Electricity sector blast furnace gas	2010	PM10	ELC	0.089	kilotonnes	PJ	EMEP / EEA, 2013 Energy p17
Electricity sector blast furnace gas	2010	PM2.5	ELC	0.089	kilotonnes	PJ	EMEP / EEA, 2013 Energy p17

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Power stations, landfill gas	2010	PM10	ELC	0.011	kilotonnes	PJ	NAEI 2013
Power stations, landfill gas	2010	PM2.5	ELC	0.011	kilotonnes	РJ	NAEI 2013
Power stations, landfill gas	2010	NH3	ELC	NE	kilotonnes	PJ	NAEI 2013
Power stations, landfill gas	2010	SO2	ELC	0.031	kilotonnes	РJ	NAEI 2013
Power stations, landfill gas	2010	NOx	ELC	0.273	kilotonnes	PJ	NAEI 2013
Power stations, landfill gas	2010	Non Methane VOC	ELC	0.004	kilotonnes	PJ	NAEI 2013
Power stations, sewage gas	2010	PM10	ELC	0.006	kilotonnes	PJ	NAEI 2013
Power stations, sewage gas	2010	PM2.5	ELC	0.006	kilotonnes	PJ	NAEI 2013
Power stations, sewage gas	2010	NH3	ELC	NE	kilotonnes	PJ	NAEI 2013
Power stations, sewage gas	2010	SO2	ELC	0.000	kilotonnes	PJ	NAEI 2013
Power stations, sewage gas	2010	NOx	ELC	0.267	kilotonnes	PJ	NAEI 2013
Power stations, sewage gas	2010	Non Methane VOC	ELC	0.003	kilotonnes	PJ	NAEI 2013
Existing small oil-fired gas turbine	2010	NH3	ELC	0.000	kilotonnes	PJ	IIR Switzerland, 2013
Power stations, MSW	2010	PM10	ELC	0.001	kilotonnes	PJ	NAEI 2013
Power stations, MSW	2010	PM2.5	ELC	0.001	kilotonnes	PJ	NAEI 2013
Power stations, MSW	2010	NH3	ELC	0.002	kilotonnes	PJ	NAEI 2013
Power stations, MSW	2010	SO2	ELC	0.003	kilotonnes	РЈ	NAEI 2013
Power stations, MSW	2010	NOx	ELC	0.003	kilotonnes	PJ	NAEI 2013
Power stations, MSW	2010	Non Methane VOC	ELC	0.001	kilotonnes	PJ	NAEI 2013
Autogenerators, biogas	2010	PM10	ELC	0.000	kilotonnes	PJ	NAEI 2013
Autogenerators, biogas	2010	PM2.5	ELC	0.000	kilotonnes	PJ	NAEI 2013
Autogenerators, biogas	2010	NH3	ELC	0.000	kilotonnes	PJ	NAEI 2013
Autogenerators, biogas	2010	SO2	ELC	0.031	kilotonnes	PJ	NAEI 2013
Autogenerators, biogas	2010	NUX	ELC	0.264	Kilotonnes	L L L	NAEI 2013
Autogenerators, biogas	2010	Non Methane VOC	ELC	0.004	kilotonnes	PJ	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Public sector combustion (sewage gas)	2010	PM10	ELC	0.006	kilotonnes	PJ	NAEI 2013
Public sector combustion (sewage gas)	2010	PM2.5	ELC	0.006	kilotonnes	PJ	NAEI 2013
Public sector combustion (sewage gas)	2010	NH3	ELC	NE	kilotonnes	PJ	ATSDR, 2001
Public sector combustion (sewage gas)	2010	SO2	ELC	0.000	kilotonnes	PJ	NAEI 2013
Public sector combustion (sewage gas)	2010	NOx	ELC	0.267	kilotonnes	PJ	NAEI 2013
Public sector combustion (sewage gas)	2010	Non Methane VOC	ELC	0.003	kilotonnes	PJ	NAEI 2013
Existing hydro reservoir plants	2010	PM10	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010
Existing hydro reservoir plants	2010	PM2.5	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing hydro reservoir plants	2010	NH3	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing hydro reservoir plants	2010	SO2	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing hydro reservoir plants	2010	NUX	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing hydro reservoir plants	2010	Non Methane VOC	ELC	NE	Kilotonnes	L L L L L L L L L L L L L L L L L L L	Black and Flarend, 2010 p121
Existing onshore wind turbines	2010	PM10	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing onshore wind turbines	2010	PM2.5	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing onshore wind turbines	2010	NH3	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing onshore wind turbines	2010	502	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
turbines	2010	NUX	ELC	NE	kilotonnes	PJ	2010 p121
Existing onshore wind turbines	2010	Non Methane VOC	ELC	NE	kilotonnes	L L L L L L L L L L L L L L L L L L L	Black and Flarend, 2010 p121
Existing offshore wind turbines	2010	PM10	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing offshore wind turbines	2010	PM2.5	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing offshore wind turbines	2010	NH3	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing offshore wind turbines	2010	SO2	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing offshore wind turbines	2010	NOx	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing offshore wind turbines	2010	Non Methane VOC	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing large-scale solar photovoltaic (PV) installations	2010	PM10	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing large-scale solar photovoltaic (PV) installations	2010	PM2.5	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing large-scale solar photovoltaic (PV) installations	2010	NH3	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing large-scale solar photovoltaic (PV) installations	2010	SO2	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing large-scale solar photovoltaic (PV) installations	2010	NOx	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing large-scale solar photovoltaic (PV) installations	2010	Non Methane VOC	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing small-scale PV installations	2010	PM10	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing small-scale PV installations	2010	PM2.5	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing small-scale PV installations	2010	NH3	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Existing small-scale PV installations	2010	SO2	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
installations	2010	NOX	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
installations	2010	Methane VOC			kilotonnes	PJ	2010 p121
Pressurized water reactor (PWR)	2010	PM10	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Pressurized water reactor (PWR)	2010	PM2.5	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Pressurized water reactor (PWR)	2010	NH3	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
reactor (PWR)	2010	502	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
reactor (PWR)	2010	NOX		INE .	kilotonnes	PJ	2010 p121
reactor (PWR)	2010	Non Methane VOC	ELC		Kilotonnes	14	2010 p121
Advanced gas-cooled reactor (AGR), newer generation	2010	PM10	ELC	NE	kilotonnes	РЈ	Black and Flarend, 2010 p121
Advanced gas-cooled reactor (AGR), newer generation	2010	PM2.5	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Advanced gas-cooled reactor (AGR), newer generation	2010	NH3	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Advanced gas-cooled reactor (AGR), newer generation	2010	SO2	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Advanced gas-cooled reactor (AGR), newer generation	2010	NOx	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Advanced gas-cooled reactor (AGR), newer generation	2010	Non Methane VOC	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
AGR, older generation	2010	PM10	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
AGR, older generation	2010	PM2.5	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
AGR, older generation	2010	NH3	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
AGR, older generation	2010	SO2	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
AGR, older generation	2010	NOx	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
AGR, older generation	2010	Non Methane VOC	ELC	NE	kilotonnes	PJ	Black and Flarend, 2010 p121
Power stations, natural gas	2010	PM10	ELC	0.001	kilotonnes	PJ	NAEI 2013
Power stations, natural gas	2010	PM2.5	ELC	0.001	kilotonnes	PJ	NAEI 2013
Power stations, natural gas	2010	NH3	ELC	NE	kilotonnes	PJ	EMEP/EEA, 2009
Power stations, natural gas	2010	SO2	ELC	0.000	kilotonnes	PJ	NAEI 2013
Power stations, natural gas	2010	NOx	ELC	0.037	kilotonnes	PJ	NAEI 2013
Power stations, natural gas	2010	Non Methane VOC	ELC	0.001	kilotonnes	PJ	NAEI 2013
Power stations, heavy fuel oil	2010	PM10	ELC	0.025	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, heavy fuel oil	2010	PM2.5	ELC	0.019	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, heavy fuel oil	2010	NH3	ELC	NE	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, heavy fuel oil	2010	SO2	ELC	0.495	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, heavy fuel oil	2010	NOx	ELC	0.142	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, heavy fuel oil	2010	Non Methane VOC	ELC	0.002	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gaseous fuels	2010	PM10	ELC	0.001	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gaseous fuels	2010	PM2.5	ELC	0.001	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gaseous fuels	2010	NH3	ELC	NE	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gaseous fuels	2010	SO2	ELC	0.000	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gaseous fuels	2010	NOx	ELC	0.089	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gaseous fuels	2010	Non Methane VOC	ELC	0.003	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, biomass	2010	PM10	ELC	0.155	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, biomass	2010	PM2.5	FLC	0.133	kilotonnes	PI PI	EMEP/EEA, 2013
Power stations, biomass	2010	SO2	ELC	0.011	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, biomass	2010	NOx	ELC	0.081	kilotonnes	PJ	EMEP/EEA, 2013

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
Power stations, gas oil	2010	PM10	ELC	0.003	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gas oil	2010	PM2.5	ELC	0.001	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gas oil	2010	NH3	ELC	NE	kilotonnes	РЈ	EMEP/EEA, 2013
Power stations, gas oil	2010	SO2	ELC	0.047	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, gas oil	2010	NOx	ELC	0.065	kilotonnes	РЈ	EMEP/EEA, 2013
Power stations, gas oil	2010	Non Methane VOC	ELC	0.001	kilotonnes	PJ	EMEP/EEA, 2013
Power stations, LPG	2010	PM10	ELC	0.004	kilotonnes	PJ	NAEI 2013
Power stations, LPG	2010	PM2.5	ELC	0.004	kilotonnes	PJ	NAEI 2013
Power stations, LPG	2010	NH3	ELC	NE	kilotonnes	PJ	Not estimated as no Efs available
Power stations, LPG	2010	SO2	ELC	0.000	kilotonnes	PJ	IIR Denmark, 2011
Power stations, LPG	2010	NOx	ELC	0.074	kilotonnes	РЈ	NAEI 2013
Power stations, LPG	2010	Non Methane VOC	ELC	0.004	kilotonnes	PJ	NAEI 2013
Public sector combustion, fuel oil	2010	PM10	SER	0.026	kilotonnes	PJ	NAEI 2013
Public sector combustion. fuel oil	2010	PM2.5	SER	0.026	kilotonnes	РЈ	NAEI 2013
Public sector combustion, fuel oil	2010	NH3	SER	0.000	kilotonnes	PJ	NAEI 2013
Public sector combustion. fuel oil	2010	SO2	SER	0.353	kilotonnes	PJ	NAEI 2013
Public sector combustion, fuel oil	2010	NOx	SER	0.175	kilotonnes	PJ	NAEI 2013
Public sector combustion, fuel oil	2010	Non Methane VOC	SER	0.003	kilotonnes	PJ	NAEI 2013
Domestic combustion, anthracite	2010	PM10	RES	0.057	kilotonnes	PJ	NAEI 2013
Domestic combustion, anthracite	2010	PM2.5	RES	0.034	kilotonnes	PJ	NAEI 2013
Domestic combustion, anthracite	2010	NH3	RES	0.030	kilotonnes	PJ	NAEI 2013
Domestic combustion, anthracite	2010	SO2	RES	0.457	kilotonnes	PJ	NAEI 2013
Domestic combustion, anthracite	2010	NOx	RES	0.134	kilotonnes	PJ	NAEI 2013
Domestic combustion, anthracite	2010	Non Methane VOC	RES	0.052	kilotonnes	PJ	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Stationary combustion in manufacturing industries and construction: Iron and steel (blast furnace gas)	2010	PM10	IND	0.001	kilotonnes	PJ	NAEI 2013
Stationary combustion in manufacturing industries and construction: Iron and steel (blast furnace gas)	2010	PM2.5	IND	0.001	kilotonnes	PJ	NAEI 2013
Stationary combustion in manufacturing industries and construction: Iron and steel (blast furnace gas)	2010	NOx	IND	0.075	kilotonnes	PJ	NAEI 2013
Stationary combustion in manufacturing industries and construction: Iron and steel (blast furnace gas)	2010	Non Methane VOC	IND	0.023	kilotonnes	PJ	NAEI 2013
Fuels combusted in the refinery sector (calculated IEF)	2010	PM10	PRC	0.000	kilotonnes	PJ	NAEI 2013
Fuels combusted in the refinery sector (calculated IEF)	2010	PM2.5	PRC	0.000	kilotonnes	PJ	NAEI 2013
Fuels combusted in the refinery sector (calculated IEF)	2010	NH3	PRC	0.000	kilotonnes	PJ	NAEI 2013
Fuels combusted in the refinery sector (calculated IEF)	2010	SO2	PRC	0.017	kilotonnes	PJ	NAEI 2013
Fuels combusted in the refinery sector (calculated IEF)	2010	NOx	PRC	0.007	kilotonnes	PJ	NAEI 2013
Fuels combusted in the refinery sector (calculated IEF)	2010	Non Methane VOC	PRC	0.007	kilotonnes	PJ	NAEI 2013
Coke production (calculated IEE)	2010	PM10	PRC	0.003	kilotonnes	PJ	NAEI 2013
Coke production (calculated IEF)	2010	PM2.5	PRC	0.002	kilotonnes	PJ	NAEI 2013
Coke production (calculated IEF)	2010	NH3	PRC	0.001	kilotonnes	PJ	NAEI 2013
Coke production (calculated IEF)	2010	SO2	PRC	0.085	kilotonnes	PJ	NAEI 2013
Coke production (calculated IEF)	2010	NOx	PRC	0.034	kilotonnes	L PJ	NAEI 2013
Coke production (calculated IEF)	2010	Non Methane VOC	PRC	0.003	kilotonnes	PJ	NAEI 2013

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
Natural gas for the	2010	DM10	DPC	I IE	kilotoppes	DI	Included
processing sector (refineries)	2010	1 1010	The		Kilotofines	15	elsewhere:
Natural gas for the	2010	PM2.5	PRC	IE	kilotonnes	PJ	Included
processing sector (refineries)							elsewhere:
Natural gas for the processing sector (refineries)	2010	NH3	PRC	IE	kilotonnes	PJ	Included elsewhere:
Natural gas for the processing sector (refineries)	2010	SO2	PRC	IE	kilotonnes	PJ	Included elsewhere:
Natural gas for the processing sector (refineries)	2010	NOx	PRC	IE	kilotonnes	PJ	Included elsewhere:
Natural gas for the processing sector (refineries)	2010	Non Methane VOC	PRC	IE	kilotonnes	PJ	Included elsewhere:
Coal for the processing sector (coke production)	2010	PM10	PRC	IE	kilotonnes	PJ	Included elsewhere:
Coal for the processing sector (coke production)	2010	PM2.5	PRC	IE	kilotonnes	PJ	Included elsewhere:
Coal for the processing sector (coke production)	2010	NH3	PRC	IE	kilotonnes	PJ	Included elsewhere:
Coal for the processing sector (coke production)	2010	SO2	PRC	IE	kilotonnes	PJ	Included elsewhere:
Coal for the processing sector (coke production)	2010	NOx	PRC	IE	kilotonnes	PJ	Included elsewhere:
Coal for the processing sector (coke production)	2010	Non Methane VOC	PRC	IE	kilotonnes	PJ	Included elsewhere:
Heavy fuel oil for the processing sector (CHP from refineries)	2010	PM10	PRC	IE	kilotonnes	PJ	Included elsewhere:
Heavy fuel oil for the processing sector (CHP from refineries)	2010	PM2.5	PRC	IE	kilotonnes	PJ	Included elsewhere:
Heavy fuel oil for the processing sector (CHP from refineries)	2010	NH3	PRC	0.000	kilotonnes	PJ	IIR Switzerland, 2013
Heavy fuel oil for the processing sector (CHP from refineries)	2010	SO2	PRC	IE	kilotonnes	PJ	Included elsewhere:
Heavy fuel oil for the processing sector (CHP from refineries)	2010	NOx	PRC	IE	kilotonnes	PJ	Included elsewhere:
Heavy fuel oil for the processing sector (CHP from refineries)	2010	Non Methane VOC	PRC	IE	kilotonnes	PJ	Included elsewhere:

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Hydrogen production for the processing sector	2010	PM10	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Hydrogen production for the processing sector	2010	PM2.5	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Hydrogen production for the processing sector	2010	NH3	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Hydrogen production for the processing sector	2010	SO2	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Hydrogen production for the processing sector	2010	NOx	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Hydrogen production for the processing sector	2010	Non Methane VOC	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Bioenergy for the processing sector	2010	PM10	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Bioenergy for the processing sector	2010	PM2.5	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
processing sector	2010	NH3	PRC	NE	Kilotonnes	PJ	no Efs available
Bioenergy for the processing sector	2010	SO2	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Bioenergy for the processing sector	2010	NOx	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Bioenergy for the processing sector	2010	Non Methane VOC	PRC	NE	kilotonnes	PJ	Not estimated as no Efs available
Existing Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	PM10	TRA	0.023	gram	kilometre	NAEI 2013
Existing Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	PM2.5	TRA	0.013	gram	kilometre	NAEI 2013
Existing Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	NH3	TRA	0.037	gram	kilometre	NAEI 2013
Existing Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	SO2	TRA	0.001	gram	kilometre	NAEI 2013
Existing Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	NOx	TRA	0.204	gram	kilometre	NAEI 2013
Existing Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	Non Methane VOC	TRA	0.070	gram	kilometre	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	PM10	TRA	0.023	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	PM2.5	TRA	0.013	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	NH3	TRA	0.037	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	SO2	TRA	0.001	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	NOx	TRA	0.204	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	Non Methane VOC	TRA	0.070	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing LPG Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	PM10	TRA	0.023	gram	kilometre	NAEI 2013
Existing LPG Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	PM2.5	TRA	0.013	gram	kilometre	NAEI 2013
Existing LPG Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	NH3	TRA	0.037	gram	kilometre	NAEI 2013
Existing LPG Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	SO2	TRA	0.001	gram	kilometre	NAEI 2013
Existing LPG Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	NOx	TRA	0.153	gram	kilometre	NAEI 2013
Existing LPG Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	Non Methane VOC	TRA	0.070	gram	kilometre	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing Hybrid Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	PM10	TRA	0.023	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	PM2.5	TRA	0.013	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	NH3	TRA	0.037	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	SO2	TRA	0.001	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	NOx	TRA	0.133	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Petrol Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	Non Methane VOC	TRA	0.070	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	PM10	TRA	0.023	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	PM2.5	TRA	0.013	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	NH3	TRA	0.037	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	SO2	TRA	0.001	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	NOx	TRA	0.133	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013
Existing Hybrid Car Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	Non Methane VOC	TRA	0.070	gram	kilometre	NAEI 2013 & NAEI alternative fuelled vehicle report, Feb 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
				[[ſ
Existing Diesel Car Emissions incl. Brake	2010	PM10	TRA	0.045	gram	kilometre	NAEI 2013
Tyre Road abrasion							
hot exhaust							
Existing Diesel Car	2010	PM2.5	TRA	0.034	gram	kilometre	NAEI 2013
Emissions incl. Brake,					0		
Tyre, Road abrasion,							
hot exhaust							
Existing Diesel Car	2010	NH3	TRA	0.001	gram	kilometre	NAEI 2013
Emissions incl. Brake,							
Tyre, Road abrasion,							
hot exhaust	2010	603	TDA	0.001		luile as et as	NAEL 2012
Existing Diesei Car	2010	502		0.001	gram	Kilometre	INAEI 2013
Ture Bood chrosien							
hot exhaust							
Fristing Diesel Car	2010	NOx	TRA	0.643	gram	kilometre	NAFI 2013
Emissions incl. Brake.	2010			0.045	Brann	kilometre	
Tyre, Road abrasion,							
hot exhaust							
Existing Diesel Car	2010	Non	TRA	0.012	gram	kilometre	NAEI 2013
Emissions incl. Brake,		Methane					
Tyre, Road abrasion,		VOC					
hot exhaust	2010	51440	75.4	0.010			
Existing Car Emissions	2010	PM10	TRA	0.042	gram	kilometre	NAEI 2013 & NAEI
Inci. Brake, Tyre, Road							alternative fuelled
(bio-diesel fuelled)							Feb 2013
Existing Car Emissions	2010	PM2 5	TRA	0.032	gram	kilometre	NΔFI 2013 & NΔFI
incl. Brake, Tyre, Road	2010				8.0		alternative fuelled
abrasion, hot exhaust							vehicle report,
(bio-diesel fuelled)							Feb 2013
Existing Car Emissions	2010	NH3	TRA	0.001	gram	kilometre	NAEI 2013 & NAEI
incl. Brake, Tyre, Road							alternative fuelled
abrasion, hot exhaust							vehicle report,
(bio-diesel fuelled)							Feb 2013
Existing Car Emissions	2010	502	IRA	0.001	gram	kilometre	NAEI 2013 & NAEI
Inci. Brake, Tyre, Road							alternative fuelled
abrasion, not exhaust (big discal fuelled)							Fob 2012
Existing Car Emissions	2010	NOx	TRA	0.707	gram	kilometre	NAFI 2013 & NAFI
incl. Brake, Tyre, Road	2010		1101	0.707	Brann	kilometre	alternative fuelled
abrasion, hot exhaust							vehicle report,
(bio-diesel fuelled)							Feb 2013
Existing Car Emissions	2010	Non	TRA	0.012	gram	kilometre	NAEI 2013 & NAEI
incl. Brake, Tyre, Road		Methane					alternative fuelled
abrasion, hot exhaust		voc					vehicle report,
(bio-diesel fuelled)							Feb 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing Diesel Buses	2010	PM10	TRA	0.186	gram	kilometre	NAEI 2013
Emissions incl. Brake,							
Tyre, Road abrasion,							
hot exhaust							
Existing Diesel Buses	2010	PM2.5	TRA	0.133	gram	kilometre	NAFI 2013
Emissions incl. Brake,	2010	1112.5		0.135	Brain	kilometre	10/12/ 2013
Tyre, Road abrasion,							
evaporative emissions,							
hot exhaust							
Existing Diesel Buses	2010	NH3	TRA	0.003	gram	kilometre	NAEI 2013
Emissions incl. Brake,							
Tyre, Road abrasion,							
evaporative emissions,							
hot exhaust			75.4				
Existing Diesel Buses	2010	502	TRA	0.004	gram	kilometre	NAEI 2013
Emissions Incl. Brake,							
avanorative emissions							
hot exhaust							
Fristing Diesel Buses	2010	ΝΟχ	TRA	6 581	gram	kilometre	NAFI 2013
Emissions incl. Brake	2010	NOX		0.581	grain	Kilometre	
Tyre. Road abrasion.							
evaporative emissions.							
hot exhaust							
Existing Diesel Buses	2010	Non	TRA	0.169	gram	kilometre	NAEI 2013
Emissions incl. Brake,		Methane					
Tyre, Road abrasion,		VOC					
evaporative emissions,							
hot exhaust							
Existing Buses Emissions	2010	PM10	TRA	0.177	gram	kilometre	NAEI 2013 & NAEI
incl. Brake, Tyre, Road							alternative fuelled
abrasion, hot exhaust							vehicle report,
(bio-diesel fuelled)	2010	DN 42 5	TDA	0.425			Feb 2013
Existing Buses Emissions	2010	PIVIZ.5	IKA	0.125	gram	Kilometre	NAEI 2013 & NAEI
inci. Brake, Tyre, Koad							alternative fuelled
abiasion, not exhaust (bio-diesel fuelled)							Feb 2013
Existing Buses Emissions	2010	NH3	ΤΡΔ	0.003	gram	kilometro	NAFI 2013 8. NAEI
incl. Brake, Tyre. Road	2010	NII S		0.003	gram	Kilometre	alternative fuelled
abrasion, hot exhaust							vehicle report,
(bio-diesel fuelled)							Feb 2013
Existing Buses Emissions	2010	SO2	TRA	0.004	gram	kilometre	NAEI 2013 & NAEI
incl. Brake, Tyre, Road							alternative fuelled
abrasion, hot exhaust							vehicle report,
(bio-diesel fuelled)							Feb 2013
Existing Buses Emissions	2010	NOx	TRA	7.240	gram	kilometre	NAEI 2013 & NAEI
incl. Brake, Tyre, Road							alternative fuelled
abrasion, hot exhaust							vehicle report,
(bio-diesel fuelled)							Feb 2013
Existing Buses Emissions	2010	Non	TRA	0.169	gram	kilometre	NAEI 2013 & NAEI
inci. Brake, Tyre, Road		Methane					alternative fuelled
aprasion, hot exhaust		VUC					vehicle report,
(bio-diesel fuelled)							Feb 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing Diesel Light	2010	PM10	TRA	0.079	gram	kilometre	NAEI 2013
Truck Emissions Incl.							
Brake, Tyre, Road							
Existing Diesel Light	2010	PM2 5	TRA	0.063	gram	kilometre	NAEL 2013
Truck Emissions incl.	2010	1 1012.5		0.005	Brann	Kilometre	NALI 2013
Brake, Tyre, Road							
abrasion, hot exhaust							
Existing Diesel Light	2010	NH3	TRA	0.001	gram	kilometre	NAEI 2013
Truck Emissions incl.							
Brake, Tyre, Road							
abrasion, hot exhaust	2010	503	TDA	0.001	gram	kilomotro	
Truck Emissions incl	2010	302	IKA	0.001	gram	Kilometre	NAEI 2015
Brake Tyre Road							
abrasion. hot exhaust							
Existing Diesel Light	2010	NOx	TRA	0.901	gram	kilometre	NAEI 2013
Truck Emissions incl.							
Brake, Tyre, Road							
abrasion, hot exhaust							
Existing Diesel Light	2010	Non	TRA	0.045	gram	kilometre	NAEI 2013
Iruck Emissions Inci.		Wethane					
abrasion, hot exhaust		VUC					
Existing Diesel Light	2010	PM10	TRA	0.075	gram	kilometre	NAEI 2013 & NAEI
Truck Emissions incl.					Ū		alternative fuelled
Brake, Tyre, Road							vehicle report,
abrasion, hot exhaust							Feb 2013
(bio-diesel fuelled)							
Existing Diesel Light	2010	PM2.5	IRA	0.059	gram	kilometre	NAEI 2013 & NAEI
Brake Tyre Boad							vehicle report
abrasion hot exhaust							Feb 2013
(bio-diesel fuelled)							
Existing Diesel Light	2010	NH3	TRA	0.001	gram	kilometre	NAEI 2013 & NAEI
Truck Emissions incl.							alternative fuelled
Brake, Tyre, Road							vehicle report,
abrasion, hot exhaust							Feb 2013
(bio-diesel fuelled)	2010	602	TDA	0.001		1.1	
Existing Diesel Light	2010	502	IRA	0.001	gram	kilometre	NAEI 2013 & NAEI
Brake, Tyre, Road							vehicle report
abrasion. hot exhaust							Feb 2013
(bio-diesel fuelled)							
Existing Diesel Light	2010	NOx	TRA	0.991	gram	kilometre	NAEI 2013 & NAEI
Truck Emissions incl.							alternative fuelled
Brake, Tyre, Road							vehicle report,
abrasion, hot exhaust							Feb 2013
(DIO-GIESEI TUEIIEG)	2010	Non	TRA	0.045	gram	kilometro	NAEL 2013 & NAEL
Truck Emissions incl.	2010	Methane	11/4	0.045	5.011	Kilomette	alternative fuelled
Brake, Tyre, Road		VOC					vehicle report,
abrasion, hot exhaust							Feb 2013
(bio-diesel fuelled)							

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing Petrol Light Truck Emissions incl.	2010	PM10	TRA	0.032	gram	kilometre	NAEI 2013
Brake, Tyre, Road abrasion, hot exhaust							
Existing Petrol Light	2010	PM2.5	TRA	0.018	gram	kilometre	NAEI 2013
Truck Emissions incl. Brake Tyre Road							
abrasion, hot exhaust							
Existing Petrol Light	2010	NH3	TRA	0.043	gram	kilometre	NAEI 2013
Brake, Tyre, Road							
abrasion, hot exhaust							
Existing Petrol Light	2010	SO2	TRA	0.001	gram	kilometre	NAEI 2013
Brake, Tyre, Road							
abrasion, hot exhaust							
Existing Petrol Light	2010	NOx	TRA	0.669	gram	kilometre	NAEI 2013
Brake, Tyre, Road							
abrasion, hot exhaust							
Existing Petrol Light Truck Emissions incl.	2010	Non Methane	TRA	0.218	gram	kilometre	NAEI 2013
Brake, Tyre, Road		VOC					
abrasion, hot exhaust	2010	DM10	ТРА	0.021	gram	kilomotro	NAEL 2012
Truck Emissions incl.	2010	PIVIIO		0.051	gram	Kilometre	NAEI 2015
Brake, Tyre, Road							
abrasion, hot exhaust (ethanol fuelled)							
Existing Petrol Light	2010	PM2.5	TRA	0.018	gram	kilometre	NAEI 2013
Truck Emissions incl.							
Brake, Tyre, Road abrasion, hot exhaust							
(ethanol fuelled)							
Existing Petrol Light	2010	NH3	TRA	0.043	gram	kilometre	NAEI 2013
Brake, Tyre, Road							
abrasion, hot exhaust							
(ethanol fuelled)	2010	502	TRA	0.001	gram	kilometre	NAFI 2013
Truck Emissions incl.	2010	502		0.001	Brain	Kilometre	
Brake, Tyre, Road							
(ethanol fuelled)							
Existing Petrol Light	2010	NOx	TRA	0.669	gram	kilometre	NAEI 2013
Truck Emissions incl. Brake Tyre Road							
abrasion, hot exhaust							
(ethanol fuelled)							
Existing Petrol Light	2010	Non Methane	TRA	0.218	gram	kilometre	NAEI 2013
Brake, Tyre, Road		VOC					
abrasion, hot exhaust							
(ethanol fuelled)	1	I	1	1	I		

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing Petrol Two- wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	PM10	TRA	0.025	gram	kilometre	NAEI 2013
Existing Petrol Two- wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	PM2.5	TRA	0.020	gram	kilometre	NAEI 2013
Existing Petrol Two- wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	NH3	TRA	0.002	gram	kilometre	NAEI 2013
Existing Petrol Two- wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	SO2	TRA	0.000	gram	kilometre	NAEI 2013
Existing Petrol Two- wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	NOx	TRA	0.223	gram	kilometre	NAEI 2013
Existing Petrol Two- wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust	2010	Non Methane VOC	TRA	1.095	gram	kilometre	NAEI 2013
Existing Two-wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	PM10	TRA	0.022	gram	kilometre	NAEI 2013
Existing Two-wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	PM2.5	TRA	0.017	gram	kilometre	NAEI 2013
Existing Two-wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	NH3	TRA	0.002	gram	kilometre	NAEI 2013
Existing Two-wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	SO2	TRA	0.000	gram	kilometre	NAEI 2013
Existing Two-wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	NOx	TRA	0.223	gram	kilometre	NAEI 2013
Existing Two-wheelers Emissions incl. Brake, Tyre, Road abrasion, hot exhaust (ethanol fuelled)	2010	Non Methane VOC	TRA	1.095	gram	kilometre	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing Diesel Heavy Truck Emissions incl.	2010	PM10	TRA	0.153	gram	kilometre	NAEI 2013
Brake, Tyre, Road abrasion, hot exhaust							
Existing Diesel Heavy	2010	PM2.5	TRA	0.110	gram	kilometre	NAEI 2013
Truck Emissions incl. Brake Tyre Boad							
abrasion, hot exhaust							
Existing Diesel Heavy	2010	NH3	TRA	0.003	gram	kilometre	NAEI 2013
Truck Emissions incl.							
abrasion, hot exhaust							
Existing Diesel Heavy	2010	SO2	TRA	0.004	gram	kilometre	NAEI 2013
Truck Emissions incl.							
abrasion, hot exhaust							
Existing Diesel Heavy	2010	NOx	TRA	4.045	gram	kilometre	NAEI 2013
Truck Emissions incl.							
abrasion, hot exhaust							
Existing Diesel Heavy	2010	Non	TRA	0.097	gram	kilometre	NAEI 2013
Truck Emissions incl.		Methane					
Brake, Tyre, Road abrasion. hot exhaust		VOC					
Existing Heavy Truck	2010	PM10	TRA	0.146	gram	kilometre	NAEI 2013
Emissions incl. Brake,							
lyre, Road abrasion, hot exhaust (bio-diesel							
fuelled)							
Existing Heavy Truck	2010	PM2.5	TRA	0.104	gram	kilometre	NAEI 2013
Tyre. Road abrasion.							
hot exhaust (bio-diesel							
fuelled)	2010	NUID	TDA	0.002		1.1	NA51 2012
Existing Heavy Truck Emissions incl. Brake.	2010	NH3	IRA	0.003	gram	Kilometre	NAEI 2013
Tyre, Road abrasion,							
hot exhaust (bio-diesel							
Tuelled)	2010	502	TRA	0.004	gram	kilometre	NAFI 2013
Emissions incl. Brake,		002			B		
Tyre, Road abrasion,							
not exhaust (bio-diesei fuelled)							
Existing Heavy Truck	2010	NOx	TRA	4.450	gram	kilometre	NAEI 2013
Emissions incl. Brake,							
hot exhaust (bio-diesel							
fuelled)							
Existing Heavy Truck	2010	Non	TRA	0.097	gram	kilometre	NAEI 2013
Emissions incl. Brake, Tyre Road abrasion		Methane					
hot exhaust (bio-diesel		100					
fuelled)							

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing diesel passenger train (bio- diesel fuelled)	2010	PM10	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- diesel fuelled)	2010	PM2.5	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- diesel fuelled)	2010	NH3	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- diesel fuelled)	2010	SO2	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- diesel fuelled)	2010	NOx	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- diesel fuelled)	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	
Railways passenger, gas oil (weighted average)	2010	PM10	TRA	0.05	kilotonnes	PJ	NAEI 2013
oil (weighted average) Railways passenger, gas	2010	NH3	TRA	NE U.05	kilotonnes	PJ	NAEI 2013
oil (weighted average) Railways passenger, gas oil (weighted average)	2010	SO2	TRA	0.03	kilotonnes	PJ	NAEI 2013
Railways passenger, gas oil (weighted average)	2010	NOx	TRA	1.01	kilotonnes	РЈ	NAEI 2013
Railways passenger, gas oil (weighted average)	2010	Non Methane VOC	TRA	0.06	kilotonnes	PJ	NAEI 2013
Railways freight, gas oil	2015	PM10	TRA	0.03	kilotonnes	PJ	NAEI 2013
Railways freight, gas oil	2015	PM2.5	TRA	0.03	kilotonnes	PJ	NAEI 2013
Railways freight, gas oil	2015	NH3		NE 0.02	kilotonnes	PJ	NAEI 2013
Railways freight gas oil	2013	NOv		2 55	kilotonnes		NAEL 2013
Railways freight, gas oil	2015	Non Methane	TRA	0.14	kilotonnes	PJ	NAEI 2013
Existing diesel passenger train (bio- light fuel oil fuelled)	2010	PM10	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- light fuel oil fuelled)	2010	PM2.5	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- light fuel oil fuelled)	2010	NH3	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- light fuel oil fuelled)	2010	\$O2	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- light fuel oil fuelled)	2010	NOx	TRA	NE	kilotonnes	PJ	
Existing diesel passenger train (bio- light fuel oil fuelled)	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing electric passenger train (electricity fuelled)	2010	PM10	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2010	PM2.5	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2010	NH3	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2010	SO2	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2010	NOx	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2010	Non Methane VOC	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Rail, coal	2010	PM10	TRA	0.127794731	kilotonnes	PJ	NAEI 2013
Rail, coal	2010	PM2.5	TRA	0.048561998	kilotonnes	PJ	NAEI 2013
Rail, coal	2010	NH3	TRA	NE	kilotonnes	РЈ	NAEI 2013
Rail, coal	2010	SO2	TRA	0.50969892	kilotonnes	PJ	NAEI 2013
Rail, coal	2010	NOx	TRA	0.187569686	kilotonnes	PJ	NAEI 2013
Rail, coal	2010	Non Methane VOC	TRA	0.206120348	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2010	PM10	TRA	0.000158612	kilotonnes	РЈ	NAEI 2013
Aircraft - international	2010	PM2.5	TRA	0.000158612	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2010	SO2	TRA	0.001194318	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2010	NOx	TRA	0.019927719	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2010	Non Methane VOC	TRA	0.001838914	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing (bio-kerosene fuelled)	2010	PM10	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2010	PM2.5	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2010	NH3	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2010	SO2	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2010	NOx	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Aircraft - domestic take off and landing	2010	PM10	TRA	0.000456766	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2010	PM2.5	TRA	0.000456766	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2010	SO2	TRA	0.003565642	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2010	NOx	TRA	0.05	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2010	Non Methane VOC	TRA	0.009690053	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2010	PM10	TRA	NE	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2010	PM2.5	TRA	NE	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2010	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2010	SO2	TRA	NE	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2010	NOx	TRA	NE	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2010	PM10	TRA	0.0415	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2010	PM2.5	TRA	0.0393	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2010	SO2	TRA	0.1626	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2010	NH3	TRA	NE	kilotonnes	PJ	EMEP / EEA, 2013 Navigation p14
Domestic shipping - Gas oil	2010	NOx	TRA	0.86	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2010	Non Methane VOC	IRA	NE	kilotonnes	14	NAEI 2013
Domestic shipping - fuel oil	2010	PM10	TRA	0.0271	kilotonnes	PJ	NAEI 2013
Domestic shipping - fuel oil	2010	PM2.5	TRA	0.0256	kilotonnes	PJ	NAEI 2013
Domestic shipping - fuel oil	2010	SO2	TRA	0.10	kilotonnes	PJ	NAEI 2013
Domestic shipping - fuel oil	2010	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
Domestic shipping - fuel oil	2010	NOx	TRA	0.33	kilotonnes	PJ	NAEI 2013
Domestic shipping - fuel oil	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	NĀEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Domestic shipping (bio- light fuel oil fuelled)	2010	PM10	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2010	PM2.5	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2010	NH3	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2010	SO2	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2010	NOx	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping - gas oil	2010	PM10	TRA	0.048673067	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2010	PM2.5	TRA	0.046112864	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2010	SO2	TRA	0.586505391	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2010	NH3	TRA	NE	kilotonnes	PJ	EMEP / EEA, 2013 Navigation p14
linternational shipping - gas oil	2010	NOx	TRA	1.825520201	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2010	PM10	TRA	0.14512424	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2010	PM2.5	TRA	0.137490705	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2010	SO2	TRA	0.74397264	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2010	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2010	NOx	TRA	1.670984436	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping (bio-oil fuelled)	2010	PM10	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2010	PM2.5	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2010	NH3	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2010	SO2	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2010	NOx	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2010	Non Methane VOC	TRA	NE	kilotonnes	PJ	European Biofuels, 2015

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing diesel	2015	PM10	TRA	NE	kilotonnes	PJ	NAEI 2013
passenger train (bio- diesel fuelled)							
Existing diesel	2015	PM2.5	TRA	NE	kilotonnes	PJ	NAEI 2013
passenger train (bio- diesel fuelled)							
Existing diesel	2015	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
diesel fuelled)							
Existing diesel	2015	SO2	TRA	NE	kilotonnes	PJ	NAEI 2013
passenger train (bio-							
diesel fuelled)						L	
Existing diesel	2015	NOx	TRA	NE	kilotonnes	PJ	NAEI 2013
diesel fuelled)							
Existing diesel	2015	Non	TRA	NE	kilotonnes	PJ	NAEI 2013
passenger train (bio-		Methane					
diesel fuelled)		VOC					
Railways passenger, gas	2015	PM10	TRA	0.05	kilotonnes	PJ	NAEI 2013
oil (weighted average)	2015	DM2 5	TDA	0.05	lilatanaa	DI	NAEL 2012
Railways passenger, gas	2015	PINIZ.5	IRA	0.05	Kilotonnes		NAEI 2013
Railways passenger, gas	2015	NH3	TRA	NF	kilotonnes	PI	NAFI 2013
oil (weighted average)	2010	11110					
Railways passenger, gas oil (weighted average)	2015	SO2	TRA	0.03	kilotonnes	PJ	NAEI 2013
Railways passenger, gas oil (weighted average)	2015	NOx	TRA	1.01	kilotonnes	PJ	NAEI 2013
Railways passenger, gas	2015	Non Methane	TRA	0.06	kilotonnes	PJ	NAEI 2013
on (weighted average)		VOC					
Existing diesel	2015	PM10	TRA	NE	kilotonnes	PJ	
passenger train (bio- light fuel oil fuelled)							
Existing diesel	2015	PM2.5	TRA	NE	kilotonnes	PJ	
passenger train (bio-							
light fuel oil fuelled)	2015	NH3	TRA	NE	kilotonnos	DI	
passenger train (bio- light fuel oil fuelled)	2015	СПИ	IKA		kilotonnes		
Existing diesel	2015	SO2	TRA	NE	kilotonnes	PJ	
passenger train (bio-							
light fuel oil fuelled)							
Existing diesel	2015	NOx	TRA	NĒ	kilotonnes	PJ	
passenger train (bio- light fuel oil fuelled)							
Existing diesel	2015	Non	TRA	NE	kilotonnes	PJ	
passenger train (bio-		Methane					
light fuel oil fuelled)		VOC				1	

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing electric passenger train (electricity fuelled)	2015	PM10	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2015	PM2.5	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2015	NH3	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2015	SO2	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2015	NOx	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Rail, coal	2015	PM10	TRA	0.127794731	kilotonnes	PJ	NAEI 2013
Rail, coal	2015	PM2.5	TRA	0.048561998	kilotonnes	РЈ	NAEI 2013
Rail, coal	2015	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
Rail, coal	2015	SO2	TRA	0.50969892	kilotonnes	PJ	NAEI 2013
Rail, coal	2015	NOx	TRA	0.187569686	kilotonnes	PJ	NAEI 2013
Rail, coal	2015	Non Methane VOC	TRA	0.206120348	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2015	PM10	TRA	0.000158612	kilotonnes	РЈ	NAEI 2013
Aircraft - international take off and landing	2015	PM2.5	TRA	0.000158612	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2015	SO2	TRA	0.001194318	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2015	NOx	TRA	0.019927719	kilotonnes	РЈ	NAEI 2013
Aircraft - international take off and landing	2015	Non Methane VOC	TRA	0.001838914	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing (bio-kerosene fuelled)	2015	PM10	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2015	PM2.5	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2015	NH3	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2015	SO2	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2015	NOx	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2015	Non Methane VOC	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Aircraft - domestic take off and landing	2015	PM10	TRA	0.000456766	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2015	PM2.5	TRA	0.000456766	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2015	SO2	TRA	0.003565642	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2015	NOx	TRA	0.047648032	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2015	Non Methane VOC	TRA	0.009690053	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2015	PM10	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2015	PM2.5	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2015	NH3	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2015	SO2	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2015	NOx	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2015	Non Methane VOC	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Domestic shipping - Gas oil	2015	PM10	TRA	0.0415	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2015	PM2.5	TRA	0.039288003	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2015	SO2	TRA	0.162595264	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2015	NH3	TRA	NE	kilotonnes	PJ	EMEP / EEA, 2013 Navigation p14
Domestic shipping - Gas oil	2015	NOx	TRA	0.86	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2015	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2015	PM10	TRA	0.0271	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2015	PM2.5	TRA	0.0256	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2015	SO2	TRA	0.0980	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2015	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2015	NÖx	TRA	0.3254	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2015	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Domestic shipping (bio- light fuel oil fuelled)	2015	PM10	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2015	PM2.5	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2015	NH3	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2015	SO2	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2015	NOx	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2015	Non Methane VOC	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping - gas oil	2015	PM10	TRA	0.048673067	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2015	PM2.5	TRA	0.046112864	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2015	SO2	TRA	0.586505391	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2015	NH3	TRA	NE	kilotonnes	PJ	EMEP / EEA, 2013 Navigation p14
linternational shipping - gas oil	2015	NOx	TRA	1.825520201	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2015	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2015	PM10	TRA	0.14512424	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2015	PM2.5	TRA	0.137490705	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2015	SO2	TRA	0.74397264	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2015	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2015	NOx	TRA	1.670984436	kilotonnes	PJ	NAEI 2013
linternational shipping - fuel oil	2015	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping (bio-oil fuelled)	2015	PM10	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2015	PM2.5	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2015	NH3	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2015	SO2	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2015	NOx	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2015	Non Methane VOC	TRA	NĒ	kilotonnes	PJ	European Biofuels, 2015

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New hydrogen fuel cell passenger train	2015	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New electric freight	2015	PM10	TRA	IE	kilotonnes	PJ	Included
train with additional							elsewhere:
track electrification							
New electric freight	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included
train with additional							elsewhere:
New electric freight	2015	NH3	ΤΡΔ	IF	kilotonnes	DI	Included
train with additional	2015	NIIS		"	Kilotofilics	''	elsewhere:
track electrification							
New electric freight	2015	SO2	TRA	IE	kilotonnes	PJ	Included
train with additional							elsewhere:
track electrification							
New electric freight	2015	NOx	TRA	IE	kilotonnes	PJ	Included
train with additional							elsewhere:
track electrification	2015	N	TDA	15	1.1.1	DI	Lead and a
New electric freight	2015	Non	IKA		Kilotonnes	L PJ	Included
track electrification		VOC					eisewiiere.
Existing diesel	2020	PM10	TRA	NE	kilotonnes	PJ	EMEP/EEA. 2013
passenger train (bio-	2020						Railways p8
diesel fuelled)							
Existing diesel	2020	PM2.5	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013
passenger train (bio-							Railways p8
diesel fuelled)							
Existing diesel	2020	NH3	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013
passenger train (bio-							Railways p8
diesel fuelled)	2020	502	ТРА	NE	kilotoppos	DI	
nassenger train (hio-	2020	302			Kilotofilles	FJ	Railways n8
diesel fuelled)							
Existing diesel	2020	NOx	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013
passenger train (bio-							Railways p8
diesel fuelled)							
Existing diesel	2020	Non	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013
passenger train (bio-		Methane					Railways p8
diesel fuelled)	2020		ТДА	0.0	E kilotoppos	DI	
nil (weighted average)	2020	FINITO	INA	0.0	5 Kilotonnes	FJ	Railways n8
Railways passenger, gas	2020	PM2.5	TRA	0.0	5 kilotonnes	PI	EMEP/EFA, 2013
oil (weighted average)	2020						Railways p8
Railways passenger, gas	2020	NH3	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013
oil (weighted average)							Railways p8
Railways passenger, gas	2020	SO2	TRA	0.0	3 kilotonnes	PJ	EMEP/EEA, 2013
oil (weighted average)							Railways p8
Railways passenger, gas	2020	NOx	TRA	1.0	1 kilotonnes	PJ	EMEP/EEA, 2013
oil (weighted average)	2020	Nee	TDA		C Hilsteine	DI	Railways p8
kanways passenger, gas	2020	Methano	IKA	0.0	^b kilotonnes	[Y	EIVIEP/EEA, 2013
on (weighten average)		VOC					Naliways po
							1

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Existing diesel passenger train (bio- light fuel oil fuelled)	2020	PM10	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing diesel passenger train (bio- light fuel oil fuelled)	2020	PM2.5	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing diesel passenger train (bio- light fuel oil fuelled)	2020	NH3	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing diesel passenger train (bio- light fuel oil fuelled)	2020	SO2	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing diesel passenger train (bio- light fuel oil fuelled)	2020	NOx	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing diesel passenger train (bio- light fuel oil fuelled)	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	EMEP/EEA, 2013 Railways p8
Existing electric passenger train (electricity fuelled)	2020	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
Existing electric passenger train (electricity fuelled)	2020	PM2.5	TRA	ΙE	kilotonnes	PJ	Included elsewhere:
Existing electric passenger train (electricity fuelled)	2020	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
Existing electric passenger train (electricity fuelled)	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
Existing electric passenger train (electricity fuelled)	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
Existing electric passenger train (electricity fuelled)	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
Rail, coal	2020	PM10	TRA	0.127794731	kilotonnes	PJ	NAEI 2013
Rail, coal	2020	PM2.5	TRA	0.048561998	kilotonnes	PJ	NAEI 2013
Rail, coal	2020	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
Rail, coal	2020	SO2	TRA	0.50969892	kilotonnes	PJ	NAEI 2013
Rail, coal	2020	NOx	TRA	0.187569686	kilotonnes	PJ	NAEI 2013
Rail, coal	2020	Non Methane VOC	TRA	0.206120348	kilotonnes	PJ	NAEI 2013
New hydrogen fuel cell	2020	PM10	TRA	IE	kilotonnes	PJ	Included
New hydrogen fuel cell passenger train	2020	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell	2020	NH3	TRA	IE	kilotonnes	РЈ	Included elsewhere:
New hydrogen fuel cell passenger train	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New electric passenger train with additional track electrification	2020	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
Railways freight, gas oil	2020	PM10	TRA	0.03	kilotonnes	PJ	NAEI 2013
Railways freight, gas oil	2020	PM2.5	TRA	0.03	kilotonnes	PJ	NAEI 2013
Railways freight, gas oil	2020	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
Railways freight, gas oil	2020	SO2	TRA	0.03	kilotonnes	PJ	NAEI 2013
Railways freight, gas oil	2020	NOx	TRA	2.55	kilotonnes	PJ	NAEI 2013
Railways freight, gas oil	2020	Non Methane VOC	TRA	0.14	kilotonnes	PJ	NAEI 2013
New electric freight train	2020	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2020	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2020	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2020	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New electric freight train with additional track electrification	2020	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	PM2.5	TRA	ΙE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	NH3	TRA	ΙE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	SO2	TRA	ΙE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	NOx	TRA	ΙE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
Aircraft - international take off and landing	2020	PM10	TRA	0.000158612	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2020	PM2.5	TRA	0.000158612	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2020	SO2	TRA	0.001194318	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2020	NOx	TRA	0.019927719	kilotonnes	PJ	NAEI 2013
Aircraft - international take off and landing	2020	Non Methane VOC	TRA	0.001838914	kilotonnes	Гd	NAEI 2013
Aircraft - international take off and landing (bio-kerosene fuelled)	2020	PM10	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2020	PM2.5	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2020	NH3	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2020	SO2	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2020	NOx	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - international take off and landing (bio-kerosene fuelled)	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing	2020	PM10	TRA	0.000456766	kilotonnes	PJ	NAEI 2013
Aircraft - domestic take off and landing	2020	PM2.5	TRA	0.000456766	kilotonnes	РЈ	NAEI 2013
Aircraft - domestic take off and landing Aircraft - domestic take	2020	SO2 NOx	TRA TRA	0.003565642	kilotonnes kilotonnes	PJ PJ	NAEI 2013 NAEI 2013
off and landing Aircraft - domestic take	2020	Non	TRA	0.009690053	kilotonnes	PJ	NAEI 2013
off and landing		Methane VOC					
Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
--	------	-----------------------	--------	-------------	------------	----------	------------------------------------
technology				Factor		Units	
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2020	PM10	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2020	PM2.5	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2020	NH3	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2020	SO2	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2020	NOx	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Aircraft - domestic take off and landing (bio- kerosene fuelled)	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	Aviation Economics, 2014
Domestic shipping - Gas oil	2020	PM10	TRA	0.0415	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2020	PM2.5	TRA	0.039288003	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2020	SO2	TRA	0.162595264	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2020	NH3	TRA	NE	kilotonnes	PJ	EMEP / EEA, 2013 Navigation p14
Domestic shipping - Gas oil	2020	NOx	TRA	0.86	kilotonnes	PJ	NAEI 2013
Domestic shipping - Gas oil	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2020	PM10	TRA	0.0271	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2020	PM2.5	TRA	0.0256	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2020	SO2	TRA	0.0980	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2020	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2020	NOx	TRA	0.3254	kilotonnes	PJ	NAEI 2013
Domestic shipping - Fuel oil	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
Domestic shipping (bio- light fuel oil fuelled)	2020	PM10	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2020	PM2.5	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2020	NH3	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2020	SO2	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2020	NOx	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Domestic shipping (bio- light fuel oil fuelled)	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	European Biofuels, 2015

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
linternational shipping - gas oil	2020	PM10	TRA	0.048673067	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2020	PM2.5	TRA	0.046112864	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2020	SO2	TRA	0.586505391	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2020	NH3	TRA	NE	kilotonnes	PJ	EMEP / EEA, 2013 Navigation p14
linternational shipping - gas oil	2020	NOx	TRA	1.825520201	kilotonnes	PJ	NAEI 2013
linternational shipping - gas oil	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping - Fuel oil	2020	PM10	TRA	0.14512424	kilotonnes	PJ	NAEI 2013
linternational shipping - Fuel oil	2020	PM2.5	TRA	0.137490705	kilotonnes	PJ	NAEI 2013
linternational shipping - Fuel oil	2020	SO2	TRA	0.74397264	kilotonnes	PJ	NAEI 2013
linternational shipping - Fuel oil	2020	NH3	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping - Fuel oil	2020	NOx	TRA	1.670984436	kilotonnes	PJ	NAEI 2013
linternational shipping - Fuel oil	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	NAEI 2013
linternational shipping (bio-oil fuelled)	2020	PM10	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2020	PM2.5	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2020	NH3	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2020	SO2	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2020	NOx	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
linternational shipping (bio-oil fuelled)	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	European Biofuels, 2015
Hydrogen for the residential sector	2010	PM10	RES	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen for the residential sector	2010	PM2.5	RES	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen for the residential sector	2010	NH3	RES	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen for the residential sector	2010	SO2	RES	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen for the residential sector	2010	NOx	RES	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen for the residential sector	2010	Non Methane VOC	RES	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
Fuel cells in the residential sector	2010	PM10	RES	NE	kilotonnes	PJ	H2FC Supergen White Paper. May 2014
Fuel cells in the residential sector	2010	PM2.5	RES	NE	kilotonnes	PJ	H2FC Supergen White Paper. May 2014
Fuel cells in the residential sector	2010	NH3	RES	NE	kilotonnes	PJ	H2FC Supergen White Paper. May 2014
Fuel cells in the residential sector	2010	SO2	RES	0.00015	kilotonnes	PJ	H2FC Supergen White Paper. May 2014
Fuel cells in the residential sector	2010	NOx	RES	0.000846878	kilotonnes	PJ	H2FC Supergen White Paper. May 2014
Fuel cells in the residential sector	2010	Non Methane VOC	RES	NE	kilotonnes	PJ	H2FC Supergen White Paper. May 2014
Hydrogen (gaseous) for the services sector	2010	PM10	SER	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the services sector	2010	PM2.5	SER	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the services sector	2010	NH3	SER	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the services sector	2010	SO2	SER	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the services sector	2010	NOx	SER	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the services sector	2010	Non Methane VOC	SER	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the industry sector (before distribution grid)	2010	PM10	IND	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the industry sector (before distribution grid)	2010	PM2.5	IND	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the industry sector (before distribution grid)	2010	NH3	IND	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the industry sector (before distribution grid)	2010	SO2	IND	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the industry sector (before distribution grid)	2010	NOx	IND	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004
Hydrogen (gaseous) for the industry sector (before distribution grid)	2010	Non Methane VOC	IND	NE	kilotonnes	PJ	Crabtree, Dresselhaus and Buchanan, 2004

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
New petrol cars	2015	PM10	TRA	0.023	gram	kilometre	UK IIR, 2015
New petrol cars	2015	PM2.5	TRA	0.013	gram	kilometre	UK IIR, 2015
New petrol cars	2015	NH3	TRA	0.024	gram	kilometre	UK IIR, 2015
New petrol cars	2015	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New petrol cars	2015	NOx	TRA	0.026	gram	kilometre	UK IIR, 2015
New petrol cars	2015	Non Methane VOC	TRA	0.012	gram	kilometre	UK IIR, 2015
New hybrid petrol cars	2015	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2015	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2015	NH3	TRA	0.024	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2015	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2015	NOx	TRA	0.017	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2015	Non Methane VOC	TRA	0.012	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2015	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2015	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2015	NH3	TRA	0.010	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2015	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2015	NOx	TRA	0.011	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2015	Non Methane VOC	TRA	0.005	gram	kilometre	DEFRA, 2013
New diesel cars	2015	PM10	TRA	0.023	gram	kilometre	UK IIR, 2015
New diesel cars	2015	PM2.5	TRA	0.013	gram	kilometre	UK IIR, 2015
New diesel cars	2015	NH3	TRA	0.001	gram	kilometre	UK IIR, 2015
New diesel cars	2015	SO2	TRA	0.001	gram	kilometre	UK IIR, 2015
New diesel cars	2015	NOx	TRA	0.634	gram	kilometre	UK IIR, 2015
New diesel cars	2015	Non Methane VOC	TRA	0.010	gram	kilometre	UK IIR, 2015
New hybrid diesel cars	2015	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2015	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2015	NH3	TRA	0.001	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2015	SO2	TRA	0.001	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2015	NOx	TRA	0.427	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2015	Non Methane VOC	TRA	0.010	gram	kilometre	DEFRA, 2013

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
New plug-in hybrid diesel cars	2015	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2015	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2015	NH3	TRA	0.001	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2015	SO2	TRA	0.001	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2015	NOx	TRA	0.427	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2015	Non Methane VOC	TRA	0.010	gram	kilometre	DEFRA, 2013
New hydrogen fuel cell cars	2015	PM10	TRA	0.022	gram	kilometre	TIMES Model Assumption
New hydrogen fuel cell cars	2015	PM2.5	TRA	0.012	gram	kilometre	TIMES Model Assumption
New hydrogen fuel cell cars	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hydrogen fuel cell cars	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hydrogen fuel cell cars	2015	NOx	TRA	0.000	gram	kilometre	Assumption
New hydrogen fuel cell cars	2015	Non Methane VOC	TRA	0.000	gram	kilometre	Assumption
New hybrid hydrogen fuel cell cars	2015	PM10	TRA	0.022	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2015	PM2.5	TRA	0.012	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New plug-in hybrid hydrogen fuel cell cars	2015	PM10	TRA	0.022	gram	kilometre	TIMES Model Assumption
New plug-in hybrid hydrogen fuel cell cars	2015	PM2.5	TRA	0.012	gram	kilometre	TIMES Model Assumption
New plug-in hybrid hydrogen fuel cell cars	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New plug-in hybrid hydrogen fuel cell cars	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New plug-in hybrid hydrogen fuel cell cars	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New plug-in hybrid hydrogen fuel cell cars	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
A new hybrid liquified hydrogen car	2015	PM10	TRA	0.022	gram	kilometre	TIMES Model Assumption
A new hybrid liquified hydrogen car	2015	PM2.5	TRA	0.012	gram	kilometre	TIMES Model Assumption
A new hybrid liquified	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model
A new hybrid liquified	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model
A new hybrid liquified	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model
nydrogen car	2015	Non		0.000	gram	kilomotro	Assumption
hydrogen car	2015	Methane VOC	INA	0.000	gram	Kilometre	Assumption
New CNG fueled cars	2015	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New CNG fueled cars	2015	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New CNG fueled cars	2015	NH3	TRA	0.024	gram	kilometre	DEFRA, 2013
New CNG fueled cars	2015	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New CNG fueled cars	2015	NOx	TRA	0.020	gram	kilometre	DEFRA, 2013
New CNG fueled cars	2015	Non Methane VOC	TRA	0.012	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2015	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2015	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2015	NH3	TRA	0.024	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2015	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2015	NOx	TRA	0.020	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2015	Non Methane VOC	TKA	0.012	gram	Kliometre	DEFRA, 2013
New flexible-fuel cars (for E85)	2015	PM10	TRA	0.023	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2015	PM2.5	TRA	0.013	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2015	NH3	TRA	0.024	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2015	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2015	NOx	TRA	0.026	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2015	Non Methane VOC	TRA	0.012	gram	kilometre	UK IIR, 2015
New electric battery	2015	PM10	TRA	0.022	gram	kilometre	TIMES Model
New electric battery	2015	PM2.5	TRA	0.012	gram	kilometre	TIMES Model
New electric battery	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model
cars New electric battery	2015	SO2	TRA	0.000	gram	kilometre	Assumption TIMES Model
cars .							Assumption
New electric battery cars	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery cars	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
New hybrid flexible-fuel car (for E85)	2015	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2015	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2015	NH3	TRA	0.024	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2015	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2015	NOx	TRA	0.017	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2015	Non Methane VOC	TRA	0.012	gram	kilometre	DEFRA, 2013
New electric battery operated two-wheelers	2015	PM10	TRA	0.011	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2015	PM2.5	TRA	0.006	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New conventionally fuelled (petrol) two- wheelers	2015	PM10	TRA	0.017	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2015	PM2.5	TRA	0.012	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2015	NH3	TRA	0.002	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2015	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2015	NOx	TRA	0.127	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2015	Non Methane VOC	TRA	0.267	gram	kilometre	UK IIR, 2015
New two-wheelers with hydrogen fuel-cell	2015	PM10	TRA	0.011	gram	kilometre	TIMES Model Assumption
New two-wheelers with hydrogen fuel-cell	2015	PM2.5	TRA	0.006	gram	kilometre	TIMES Model Assumption
New two-wheelers with hydrogen fuel-cell	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New two-wheelers with hydrogen fuel-cell	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New two-wheelers with hydrogen fuel-cell	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New two-wheelers with hydrogen fuel-cell	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
New diesel light trucks	2015	PM10	TRA	0.032	gram	kilometre	UK IIR, 2015
New diesel light trucks	2015	PM2.5	TRA	0.018	gram	kilometre	UK IIR, 2015
New diesel light trucks	2015	NH3	TRA	0.001	gram	kilometre	UK IIR, 2015
New diesel light trucks	2015	SO2	TRA	0.001	gram	kilometre	UK IIR, 2015
New diesel light trucks	2015	NOx	TRA	0.928	gram	kilometre	UK IIR, 2015
New diesel light trucks	2015	Non Methane VOC	TRA	0.025	gram	kilometre	UK IIR, 2015
New electric battery light truck	2015	PM10	TRA	0.030	gram	kilometre	TIMES Model Assumption
New electric battery light truck	2015	PM2.5	TRA	0.016	gram	kilometre	TIMES Model Assumption
New electric battery light truck	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model
New electric battery	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model
New electric battery light truck	2015	Non Methane	TRA	0.000	gram	kilometre	TIMES Model Assumption
New petrol light trucks	2015	PM10	TRA	0.031	gram	kilometre	UK IIR, 2015
New petrol light trucks	2015	PM2.5	TRA	0.017	gram	kilometre	UK IIR, 2015
New petrol light trucks	2015	NH3	TRA	0.025	gram	kilometre	UK IIR, 2015
New petrol light trucks	2015	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New petrol light trucks	2015	NOx	TRA	0.032	gram	kilometre	UK IIR, 2015
New petrol light trucks	2015	Non Methane VOC	TRA	0.010	gram	kilometre	UK IIR, 2015
New hydrogen fuel cell light trucks	2015	PM10	TRA	0.030	gram	kilometre	TIMES Model Assumption
New hydrogen fuel cell light trucks	2015	PM2.5	TRA	0.016	gram	kilometre	TIMES Model Assumption
New hydrogen fuel cell light trucks	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hydrogen fuel cell light trucks	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model
New hydrogen fuel cell	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model
New hydrogen fuel cell light trucks	2015	Non Methane	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell light trucks	2015	PM10	TRA	0.030	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell light trucks	2015	PM2.5	TRA	0.016	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell light trucks	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model
New hybrid hydrogen	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model
New hybrid hydrogen fuel cell light trucks	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New hybrid liquified hydrogen light trucks	2015	PM10	TRA	0.030	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2015	PM2.5	TRA	0.016	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid petrol light trucks	2015	PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2015	PM2.5	TRA	0.017	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2015	NH3	TRA	0.025	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2015	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2015	NOx	TRA	0.021	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2015	Non Methane VOC	TRA	0.010	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2015	PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2015	PM2.5	TRA	0.017	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2015	NH3	TRA	0.025	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2015	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2015	NOx	TRA	0.021	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2015	Non Methane VOC	TRA	0.010	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2015	PM10	TRA	0.032	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2015	PM2.5	TRA	0.018	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2015	NH3	TRA	0.001	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2015	SO2	TRA	0.001	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2015	NÖx	TRA	0.617	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2015	Non Methane VOC	TRA	0.025	gram	kilometre	DEFRA, 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New plug-in hybrid diesel light truck	2015	PM10	TRA	0.032	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2015	PM2.5	TRA	0.018	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2015	NH3	TRA	0.001	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2015	SO2	TRA	0.001	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2015	NOx	TRA	0.617	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2015	Non Methane VOC	TRA	0.025	gram	kilometre	DEFRA, 2013
New CNG fueled light trucks	2015	PM10	TRA	0.031	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2015	PM2.5	TRA	0.017	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2015	NH3	TRA	0.025	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2015	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2015	NOx	TRA	0.024	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2015	Non Methane	TRA	0.010	gram	kilometre	UK IIR, 2015
New LPG fueled light	2015	VOC PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
trucks New LPG fueled light	2015	PM2.5	TRA	0.017	gram	kilometre	DEFRA, 2013
trucks	2015	NH3	TRA	0.025	gram	kilometre	DEFRA, 2013
trucks	2015	502	TRA	0.000	gram	kilometre	DEERA 2013
trucks	2015	NOx	TRA	0.024	gram	kilometre	DEFRA 2013
trucks	2015	Non		0.024	gram	kilometre	DEERA 2013
trucks	2015	Methane VOC	INA	0.010	gram	Kiometre	DEI IIA, 2013
New flexible-fuel light trucks (for E85)	2015	PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
New flexible-fuel light trucks (for E85)	2015	PM2.5	TRA	0.017	gram	kilometre	DEFRA, 2013
New flexible-fuel light trucks (for E85)	2015	NH3	TRA	0.025	gram	kilometre	DEFRA, 2013
New flexible-fuel light trucks (for E85)	2015	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New flexible-fuel light trucks (for E85)	2015	NOx	TRA	0.032	gram	kilometre	DEFRA, 2013
New flexible-fuel light trucks (for E85)	2015	Non Methane VOC	TRA	0.010	gram	kilometre	DEFRA, 2013
New diesel heavy trucks	2015	PM10	TRA	0.113	gram	kilometre	UK IIR, 2015
New diesel heavy trucks		D1 40 5	70.4			1.0	111/110 0045
New diesel beaux trucks	2015	PM2.5	TRA	0.074	gram	kilometre	UK IIR, 2015
New diesel heavy trucks New diesel heavy trucks	2015 2015 2015	PM2.5 NH3 SO2	TRA TRA TRA	0.074 0.003 0.004	gram gram gram	kilometre kilometre kilometre	UK IIR, 2015 UK IIR, 2015 UK IIR, 2015

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New hybrid diesel heavy trucks	2015	PM10	TRA	0.113	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2015	PM2.5	TRA	0.074	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2015	NH3	TRA	0.003	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2015	SO2	TRA	0.004	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2015	NOx	TRA	1.623	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2015	Non Methane VOC	TRA	0.016	gram	kilometre	DEFRA, 2013
New hybrid hydrogen fuel cell heavy trucks	2015	PM10	TRA	0.087	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell heavy trucks	2015	PM2.5	TRA	0.048	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell heavy trucks	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell heavy trucks	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell heavy trucks	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen	2015	Non	TRA	0.000	gram	kilometre	TIMES Model
fuel cell heavy trucks		Methane VOC					Assumption
New CNG fueled heavy trucks	2015	PM10	TRA	0.091	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2015	PM2.5	TRA	0.052	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2015	NH3	TRA	0.003	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2015	SO2	TRA	0.004	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2015	NOx	TRA	1.039	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2015	Non Methane VOC	TRA	0.016	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2015	PM10	TRA	0.124	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2015	PM2.5	TRA	0.076	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2015	NH3	TRA	0.003	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2015	SO2	TRA	0.004	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2015	NOx	TRA	2.845	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2015	Non Methane VOC	TRA	0.026	gram	kilometre	UK IIR, 2015
New hybrid diesel buses	2015	PM10	TRA	0.124	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2015	PM2.5	IRA	0.076	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2015	NH3	TRA	0.003	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2015	502	IRA	0.004	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2015	NOx	TRA	2.845	gram	kilometre	DEFRA, 2013

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
New electric battery	2015	PM10	TRA	0.098	gram	kilometre	TIMES Model
New electric battery buses	2015	PM2.5	TRA	0.050	gram	kilometre	TIMES Model Assumption
New electric battery buses	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery buses	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery buses	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery buses	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New CNG fueled buses	2015	PM10	TRA	0.102	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2015	PM2.5	TRA	0.054	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2015	NH3	TRA	0.003	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2015	SO2	TRA	0.004	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2015	NOx	TRA	1.821	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2015	Non Methane VOC	TRA	0.026	gram	kilometre	UK IIR, 2015
New hybrid hydrogen fuel cell buses	2015	PM10	TRA	0.098	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2015	PM2.5	TRA	0.050	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2015	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2015	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2015	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2015	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric passenger train	2015	PM10	TRA	IE	gram	kilometre	Included elsewhere:
New electric passenger train	2015	PM2.5	TRA	IE	gram	kilometre	Included elsewhere:
New electric passenger train	2015	NH3	TRA	IE	gram	kilometre	Included elsewhere:
New electric passenger train	2015	SO2	TRA	IE	gram	kilometre	Included elsewhere:
New electric passenger train	2015	NOx	TRA	IE	gram	kilometre	Included elsewhere:
New electric passenger train	2015	Non Methane VOC	TRA	IE	gram	kilometre	Included elsewhere:

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New hydrogen fuel cell passenger train	2015	PM10	TRA	IE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell passenger train	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell freight train	2015	Non Methane VOC	TRA	^{IE}	kilotonnes	PJ	Included elsewhere:

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New electric freight train with additional track electrification	2015	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2015	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2015	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2015	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2015	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2015	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New petrol cars	2020	PM10	TRA	0.024	gram	kilometre	UK IIR, 2015
New petrol cars	2020	PM2.5	TRA	0.014	gram	kilometre	UK IIR, 2015
New petrol cars	2020	NH3	TRA	0.012	gram	kilometre	UK IIR, 2015
New petrol cars	2020	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New petrol cars	2020	NOx	TRA	0.061	gram	kilometre	UK IIR, 2015
New petrol cars	2020	Non Methane VOC	TRA	0.065	gram	kilometre	UK IIR, 2015
New hybrid petrol cars	2020	PM10	TRA	0.024	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2020	PM2.5	TRA	0.014	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2020	NH3	TRA	0.012	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2020	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2020	NOx	TRA	0.040	gram	kilometre	DEFRA, 2013
New hybrid petrol cars	2020	Non Methane VOC	TRA	0.065	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2020	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2020	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2020	NH3	TRA	0.005	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2020	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2020	NOx	TRA	0.025	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol cars	2020	Non Methane VOC	TRA	0.026	gram	kilometre	DEFRA, 2013

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
New diesel cars	2020	PM10	TRA	0.024	gram	kilometre	UK IIR, 2015
New diesel cars	2020	PM2.5	TRA	0.014	gram	kilometre	UK IIR, 2015
New diesel cars	2020	NH3	TRA	0.002	gram	kilometre	UK IIR, 2015
New diesel cars	2020	SO2	TRA	0.001	gram	kilometre	UK IIR, 2015
New diesel cars	2020	NOx	TRA	0.210	gram	kilometre	UK IIR, 2015
New diesel cars	2020	Non Methane VOC	TRA	0.008	gram	kilometre	UK IIR, 2015
New hybrid diesel cars	2020	PM10	TRA	0.024	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2020	PM2.5	TRA	0.014	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2020	NH3	TRA	0.002	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2020	SO2	TRA	0.001	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2020	NOx	TRA	0.141	gram	kilometre	DEFRA, 2013
New hybrid diesel cars	2020	Non Methane VOC	TRA	0.008	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2020	PM10	TRA	0.024	gram	kilometre	DEFRA, 2013
New plug-in hybrid	2020	PM2.5	TRA	0.014	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2020	NH3	TRA	0.002	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2020	SO2	TRA	0.001	gram	kilometre	DEFRA, 2013
New plug-in hybrid	2020	NOx	TRA	0.141	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel cars	2020	Non Methane VOC	TRA	0.008	gram	kilometre	DEFRA, 2013
New hydrogen fuel cell	2020	PM10	TRA	0.022	gram	kilometre	TIMES Model
New hydrogen fuel cell cars	2020	PM2.5	TRA	0.012	gram	kilometre	TIMES Model
New hydrogen fuel cell cars	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hydrogen fuel cell cars	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model
New hydrogen fuel cell	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model
New hydrogen fuel cell cars	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2020	PM10	TRA	0.022	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2020	PM2.5	TRA	0.012	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model
New hybrid hydrogen	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model
New hybrid hydrogen fuel cell cars	2020	NOx	TRA	0.000	gram	kilometre	Assumption TIMES Model Assumption
New hybrid hydrogen fuel cell cars	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New plug-in hybrid hydrogen fuel cell cars	2020	PM10	TRA	0.022	gram	kilometre	TIMES Model Assumption
New plug-in hybrid hydrogen fuel cell cars	2020	PM2.5	TRA	0.012	gram	kilometre	TIMES Model Assumption
New plug-in hybrid hydrogen fuel cell cars	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New plug-in hybrid	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model
New plug-in hybrid	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model
New plug-in hybrid	2020	Non	ΤΡΔ	0.000	gram	kilometre	TIMES Model
hydrogen fuel cell cars	2020	Methane VOC		0.000	Brann	Kilometre	Assumption
A new hybrid liquified	2020	PM10	TRA	0.022	gram	kilometre	TIMES Model
A new hybrid liquified	2020	PM2.5	TRA	0.012	gram	kilometre	TIMES Model
hydrogen car							Assumption
A new hybrid liquified hydrogen car	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
A new hybrid liquified hydrogen car	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
A new hybrid liquified	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model
hydrogen car	2020	Non	ТРА	0.000	gram	kilomotro	Assumption
hydrogen car	2020	Methane VOC		0.000	gran	Kilometre	Assumption
New CNG fueled cars	2020	PM10	TRA	0.024	gram	kilometre	UK IIR, 2015
New CNG fueled cars	2020	PM2.5	TRA	0.014	gram	kilometre	UK IIR, 2015
New CNG fueled cars	2020	NH3	TRA	0.012	gram	kilometre	UK IIR, 2015
New CNG fueled cars	2020	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New CNG fueled cars	2020	NOx		0.046	gram	kilometre	UK IIR, 2015
New CNG Tueled Cars	2020	Methane VOC	TKA	0.065	gram	Kilometre	OK IIK, 2015
New LPG fueled cars	2020	PM10	TRA	0.024	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2020	PM2.5	TRA	0.014	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2020	NH3	TRA	0.012	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2020	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2020	NOx	TRA	0.046	gram	kilometre	DEFRA, 2013
New LPG fueled cars	2020	Non Methane VOC	TRA	0.065	gram	kilometre	DEFRA, 2013
New flexible-fuel cars (for E85)	2020	PM10	TRA	0.023	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2020	PM2.5	TRA	0.013	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2020	NH3	TRA	0.012	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2020	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2020	NOx	TRA	0.061	gram	kilometre	UK IIR, 2015
New flexible-fuel cars (for E85)	2020	Non Methane VOC	TRA	0.065	gram	kilometre	UK IIR, 2015

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
New electric battery cars	2020	PM10	TRA	0.022	gram	kilometre	TIMES Model Assumption
New electric battery cars	2020	PM2.5	TRA	0.012	gram	kilometre	TIMES Model Assumption
New electric battery cars	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery cars	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery cars	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery cars	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid flexible-fuel car (for E85)	2020	PM10	TRA	0.023	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2020	PM2.5	TRA	0.013	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2020	NH3	TRA	0.012	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2020	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2020	NOx	TRA	0.040	gram	kilometre	DEFRA, 2013
New hybrid flexible-fuel car (for E85)	2020	Non Methane VOC	TRA	0.065	gram	kilometre	DEFRA, 2013
New electric battery operated two-wheelers	2020	PM10	TRA	0.011	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2020	PM2.5	TRA	0.006	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery operated two-wheelers	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New conventionally fuelled (petrol) two- wheelers	2020	PM10	TRA	0.017	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2020	PM2.5	TRA	0.012	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2020	NH3	TRA	0.002	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2020	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2020	NOx	TRA	0.127	gram	kilometre	UK IIR, 2015
New conventionally fuelled (petrol) two- wheelers	2020	Non Methane VOC	TRA	0.267	gram	kilometre	UK IIR, 2015

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New two-wheelers with hydrogen fuel-cell	2020	PM10	TRA	0.011	gram	kilometre	TIMES Model
New two-wheelers with	2020	PM2 5	TRA	0.006	gram	kilometre	TIMES Model
hydrogen fuel-cell	2020	11112.5		0.000	Brain	kilometre	Assumption
New two-wheelers with	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model
hydrogen fuel-cell					U		Assumption
New two-wheelers with	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model
hydrogen fuel-cell							Assumption
New two-wheelers with	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model
hydrogen fuel-cell							Assumption
New two-wheelers with	2020	Non	TRA	0.000	gram	kilometre	TIMES Model
hydrogen fuel-cell		Methane					Assumption
		VOC					
New diesel light trucks	2020	PM10	TRA	0.031	gram	kilometre	UK IIR, 2015
New diesel light trucks	2020	PM2.5	TRA	0.017	gram	kilometre	UK IIR, 2015
Now diosal light trucks	2020	NH3	ΤΡΛ	0.002	gram	kilometre	
New dieser light trucks	2020	ИПЗ		0.002	gram	Kilometre	OK IIK, 2013
New diesel light trucks	2020	SO2	TRA	0.001	gram	kilometre	UK IIR, 2015
New diesel light trucks	2020	NOx	TRA	0.221	gram	kilometre	UK IIR, 2015
No	2020	N	TDA	0.025		1.1	LUK UD 2015
New diesel light trucks	2020	Non	IRA	0.035	gram	Kilometre	UK IIR, 2015
		VOC					
New electric battery	2020	PM10	TRA	0.030	gram	kilometre	TIMES Model
light truck					0.0		Assumption
New electric battery	2020	PM2.5	TRA	0.016	gram	kilometre	TIMES Model
light truck					-		Assumption
New electric battery	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model
light truck							Assumption
New electric battery	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model
light truck	2020	NO	TDA	0.000			Assumption
light truck	2020	NUX	IKA	0.000	gram	kilometre	Assumption
New electric battery	2020	Non	TRA	0.000	gram	kilometre	TIMES Model
light truck	2020	Methane			Brann		Assumption
		VOC					
New petrol light trucks	2020	PM10	TRA	0.031	gram	kilometre	UK IIR, 2015
New petrol light trucks	2020	PM2.5	TRA	0.018	gram	kilometre	UK IIR, 2015
N	2020	AUL 12	TDA	0.012	-	1.1	
New petrol light trucks	2020	NH3	IKA	0.012	gram	Kilometre	UK IIR, 2015
New petrol light trucks	2020	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New petrol light trucks	2020	NOx	TRA	0.064	gram	kilometre	UK IIR, 2015
New petrol light trucks	2020	Non	TRA	0.096	gram	kilometre	
New petrol light tracks	2020	Methane		0.050	grann	Kilometre	0K III, 2015
		VOC					
New hydrogen fuel cell	2020	PM10	TRA	0.030	gram	kilometre	TIMES Model
light trucks							Assumption
New hydrogen fuel cell	2020	PM2.5	TRA	0.016	gram	kilometre	TIMES Model
light trucks							Assumption
New hydrogen fuel cell	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model
light trucks	2000	603	TDA			1.1.	Assumption
New hydrogen fuel cell	2020	502	IKA	0.000	gram	kilometre	Assumption
New hydrogen fuel cell	2020	ΝΟχ	TRA	0.000	gram	kilometre	
light trucks	2020		1101	0.000	5.u.i.		Assumption
New hydrogen fuel cell	2020	Non	TRA	0.000	gram	kilometre	TIMES Model
light trucks		Methane					Assumption
		VOC				1	

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New hybrid hydrogen fuel cell light trucks	2020	PM10	TRA	0.030	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell light trucks	2020	PM2.5	TRA	0.016	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell light trucks	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell light trucks	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell light trucks	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell light trucks	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2020	PM10	TRA	0.030	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2020	PM2.5	TRA	0.016	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid liquified hydrogen light trucks	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid petrol light trucks	2020	PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2020	PM2.5	TRA	0.018	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2020	NH3	TRA	0.012	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2020	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2020	NOx	TRA	0.042	gram	kilometre	DEFRA, 2013
New hybrid petrol light trucks	2020	Non Methane VOC	TRA	0.096	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2020	PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2020	PM2.5	TRA	0.018	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2020	NH3	TRA	0.012	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2020	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2020	NOx	TRA	0.042	gram	kilometre	DEFRA, 2013
New plug-in hybrid petrol light truck	2020	Non Methane VOC	TRA	0.096	gram	kilometre	DEFRA, 2013

Source sector/ technology	Year	Pollutant	Sector	Emission Factor	Units	Activity Units	Reference
New hybrid diesel light	2020	PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2020	PM2.5	TRA	0.017	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2020	NH3	TRA	0.002	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2020	SO2	TRA	0.001	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2020	NOx	TRA	0.147	gram	kilometre	DEFRA, 2013
New hybrid diesel light trucks	2020	Non Methane VOC	TRA	0.035	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2020	PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2020	PM2.5	TRA	0.017	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2020	NH3	TRA	0.002	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2020	SO2	TRA	0.001	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2020	NOx	TRA	0.147	gram	kilometre	DEFRA, 2013
New plug-in hybrid diesel light truck	2020	Non Methane VOC	TRA	0.035	gram	kilometre	DEFRA, 2013
New CNG fueled light trucks	2020	PM10	TRA	0.031	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2020	PM2.5	TRA	0.018	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2020	NH3	TRA	0.012	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2020	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2020	NOx	TRA	0.048	gram	kilometre	UK IIR, 2015
New CNG fueled light trucks	2020	Non Methane VOC	TRA	0.096	gram	kilometre	UK IIR, 2015
New LPG fueled light trucks	2020	PM10	TRA	0.031	gram	kilometre	DEFRA, 2013
New LPG fueled light trucks	2020	PM2.5	TRA	0.018	gram	kilometre	DEFRA, 2013
New LPG fueled light trucks	2020	NH3	TRA	0.012	gram	kilometre	DEFRA, 2013
New LPG fueled light trucks	2020	SO2	TRA	0.000	gram	kilometre	DEFRA, 2013
New LPG fueled light trucks	2020	NOx	TRA	0.048	gram	kilometre	DEFRA, 2013
New LPG fueled light trucks	2020	Non Methane VOC	TRA	0.096	gram	kilometre	DEFRA, 2013

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor	1	Units	
New flexible-fuel light trucks (for E85)	2020	PM10	TRA	0.031	gram	kilometre	UK IIR, 2015
New flexible-fuel light trucks (for E85)	2020	PM2.5	TRA	0.017	gram	kilometre	UK IIR, 2015
New flexible-fuel light trucks (for E85)	2020	NH3	TRA	0.012	gram	kilometre	UK IIR, 2015
New flexible-fuel light trucks (for E85)	2020	SO2	TRA	0.000	gram	kilometre	UK IIR, 2015
New flexible-fuel light trucks (for E85)	2020	NOx	TRA	0.064	gram	kilometre	UK IIR, 2015
New flexible-fuel light trucks (for E85)	2020	Non Methane VOC	TRA	0.096	gram	kilometre	UK IIR, 2015
New diesel heavy trucks	2020	PM10	TRA	0.087	gram	kilometre	UK IIR, 2015
New diesel heavy trucks	2020	PM2.5	TRA	0.048	gram	kilometre	UK IIR, 2015
New diesel heavy trucks	2020	NH3		0.011	gram	kilometre	UK IIR, 2015
New diesel heavy trucks	2020	NOx		0.004	gram	kilometre	UK IIR, 2015
	2020			0.231	8.u.ii	Kilometre	
New diesel heavy trucks	2020	Non Methane VOC	TRA	0.008	gram	kilometre	UK IIR, 2015
New hybrid diesel heavy trucks	2020	PM10	TRA	0.087	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2020	PM2.5	TRA	0.048	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2020	NH3	TRA	0.011	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2020	SO2	TRA	0.004	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2020	NOx	TRA	0.291	gram	kilometre	DEFRA, 2013
New hybrid diesel heavy trucks	2020	Non Methane VOC	TRA	0.008	gram	kilometre	DEFRA, 2013
New hybrid hydrogen fuel cell heavy trucks	2020	PM10	TRA	0.087	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell heavy trucks	2020	PM2.5	TRA	0.048	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell heavy trucks	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell heavy trucks	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell heavy trucks	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model
New hybrid hydrogen fuel cell heavy trucks	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New CNG fueled heavy trucks	2020	PM10	TRA	0.087	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2020	PM2.5	TRA	0.048	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2020	NH3	TRA	0.011	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2020	SO2	TRA	0.004	gram	kilometre	UK IIR, 2015
New CNG fueled heavy trucks	2020	NOx	TRA	0.186	gram	kilometre	UK IIR, 2015

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New conventionally fueled (diesel) buses	2020	PM10	TRA	0.101	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2020	PM2.5	TRA	0.053	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2020	NH3	TRA	0.003	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2020	SO2	TRA	0.004	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2020	NOx	TRA	0.597	gram	kilometre	UK IIR, 2015
New conventionally fueled (diesel) buses	2020	Non Methane VOC	TRA	0.022	gram	kilometre	UK IIR, 2015
New hybrid diesel buses	2020	PM10	TRA	0.101	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2020	PM2.5	TRA	0.053	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2020	NH3	TRA	0.003	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2020	SO2	TRA	0.004	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2020	NOx	TRA	0.597	gram	kilometre	DEFRA, 2013
New hybrid diesel buses	2020	Non Methane VOC	TRA	0.022	gram	kilometre	DEFRA, 2013
New electric battery buses	2020	PM10	TRA	0.098	gram	kilometre	TIMES Model Assumption
New electric battery buses	2020	PM2.5	TRA	0.050	gram	kilometre	TIMES Model Assumption
New electric battery buses	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery buses	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery buses	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New electric battery buses	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption
New CNG fueled buses	2020	PM10	TRA	0.099	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2020	PM2.5	TRA	0.051	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2020	NH3	TRA	0.003	gram	kilometre	UK IIR, 2015
New CNG tueled buses	2020	SO2	TRA	0.004	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2020	NOx	TRA	0.382	gram	kilometre	UK IIR, 2015
New CNG fueled buses	2020	Non Methane VOC	TRA	0.022	gram	kilometre	UK IIR, 2015
New hybrid hydrogen fuel cell buses	2020	PM10	TRA	0.098	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2020	PM2.5	TRA	0.050	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2020	NH3	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2020	SO2	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2020	NOx	TRA	0.000	gram	kilometre	TIMES Model Assumption
New hybrid hydrogen fuel cell buses	2020	Non Methane VOC	TRA	0.000	gram	kilometre	TIMES Model Assumption

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor		Units	
New electric passenger train	2020	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train	2020	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train	2020	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New hydrogen fuel cell passenger train	2020	PM10	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell passenger train	2020	PM2.5	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell passenger train	2020	NH3	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell passenger train	2020	SO2	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell passenger train	2020	NOx	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell passenger train	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New electric passenger train with additional track electrification	2020	PM10	TRA	IE	kilotonnes	PJ	Not estimated as no Efs available
New electric passenger train with additional track electrification	2020	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric passenger train with additional track electrification	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:

Source sector/	Year	Pollutant	Sector	Emission	Units	Activity	Reference
technology				Factor	-	Units	
New hydrogen fuel cell freight train	2020	PM10	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell freight train	2020	PM2.5	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell freight train	2020	NH3	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell freight train	2020	SO2	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell freight train	2020	NOx	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New hydrogen fuel cell freight train	2020	Non Methane VOC	TRA	NE	kilotonnes	PJ	Not estimated as no Efs available
New electric freight train with additional track electrification	2020	PM10	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	PM2.5	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	NH3	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	SO2	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	NOx	TRA	IE	kilotonnes	PJ	Included elsewhere:
New electric freight train with additional track electrification	2020	Non Methane VOC	TRA	IE	kilotonnes	PJ	Included elsewhere:
Aircraft - domestic take off and landing	2010	NH3	TRA	NE	kilotonnes	PJ	
Aircraft - domestic take off and landing	2015	NH3	TRA	NE	kilotonnes	PJ	
Aircraft - domestic take off and landing	2020	NH3	TRA	NE	kilotonnes	PJ	
Aircraft - international take off and landing	2010	NH3	TRA	NE	kilotonnes	PJ	
Aircraft - international take off and landing	2015	NH3	TRA	NE	kilotonnes	PJ	
Aircraft - international take off and landing	2020	NH3	TRA	NE	kilotonnes	PJ	

A.3: Graphs of Air Pollution by Sector for All Scenarios (National Scale, United Kingdom)













A.3.2: Graphs of Air Pollution by Sector - Ref Scenario











A.2.3: Graphs of Air Pollution by Sector - lowGHG Scenario











A.2.4: Graphs of Air Pollution by Sector - Base_DAMC Scenario











A.2.5: Graphs of Air Pollution by Sector - Ref_DAMC Scenario










A.2.6: Graphs of Air Pollution by Sector - lowGHG_DAMC Scenario











A.3: Demand Values - Urban Scale (Greater London) Analysis

A.3.1: For Scenarios without Behaviour Change

Group 1 (LT migration)	Population d	riven (LT mig	ration) - using	2010 per cap	oita rates		
	Motorcycle	Car	Buses	LGV	HGV	TOTAL	
2010	0.70	23.70	0.60	3.80	1.00	29.80	
2015	0.74	25.17	0.64	4.03	1.06	31.64	
2020	0.78	26.41	0.67	4.23	1.11	33.20	
2025	0.81	27.47	0.70	4.41	1.16	34.55	
2030	0.84	28.37	0.72	4.55	1.20	35.67	
2035	0.88	29.91	0.76	4.80	1.26	37.61	
2040	0.93	31.53	0.80	5.06	1.33	39.65	
2045	0.98	33.24	0.84	5.33	1.40	41.79	
2050	1.03	35.04	0.89	5.62	1.48	44.06	
Group 2 (ST migration)	Population d	riven (ST mig	ration) - using	2010 per cap	oita rates		
	Motorcycle	Car	Buses	LGV	HGV	TOTAL	
2010	0.70	23.70	0.60	3.80	1.00	29.800	
2015	0.75	25.31	0.64	4.06	1.07	31.820	
2020	0.79	26.88	0.68	4.31	1.13	33.801	
2025	0.83	28.26	0.72	4.53	1.19	35.530	
2030	0.87	29.42	0.74	4.72	1.24	36.994	
2035	0.93	31.39	0.79	5.03	1.32	39.475	
2040	0.99	33.50	0.85	5.37	1.41	42.122	
2045	1.06	35.75	0.90	5.73	1.51	44.946	
2050	1.13	38.14	0.97	6.12	1.61	47.960	
Group 3 (DfT)	DfT Scenario	1 (version: N	1arch 2015) th	rough to 204	0**		
	Motorcycle	Car	Buses	LGV	HGV	TOTAL	
2010	0.70	0.60	0.60	1.00	29.80	32.70	
2015	0.00	26.42	0.00	4.69	1.00	32.10	values are calculated
2020	0.00	28.24	0.00	5.32	1.03	34.59	values are directly from DfT
2025	0.00	30.08	0.00	5.94	1.06	37.08	updated to match DfT's TRA0206
2030	0.00	31.08	0.00	6.53	1.09	38.70	
2035	0.00	32.23	0.00	7.05	1.13	40.41	
2040	0.00	33.20	0.00	7.57	1.16	41.94	
2045	0.00	34.20	0.00	8.14	1.20	43.53	
2050	0.00	35.22	0.00	8.75	1.23	45.21	

**population assumptions (ONS Long-Term migration) for motorcycles and buses (because those aren't in DfT's projections); for all 2045 and 2050 projections for cars, LGV, HGV, assumed same rate of increase as from 2030 - 2040

A.3.2: For Scenarios with Behaviour Change

			Group 1 (LT migration)	80:20		-		
	2010	2015	2020	2025	2030	2035	2040	2045	2050
Car	23.70	25.17	25.40	25.47	25.37	25.91	26.52	27.23	28.03
Motorcycle	0.7	0.7	0.8	0.8	0.8	0.9	0.9	1.0	1.0
Buses	0.6	0.6	0.7	0.7	0.7	0.8	0.8	0.8	0.9
LGV	3.8	4.0	4.2	4.4	4.5	4.8	5.1	5.3	5.6
HGV	1	1.1	1.1	1.2	1.2	1.3	1.3	1.4	1.5
Mode Shifting	0	0	1.00	2.00	3.00	4.00	5.01	6.01	7.01
TOTAL	29.80	31.64	33.20	34.55	35.67	37.61	39.65	41.79	44.06
Group 1 (LT migration) 60:40									
	2010	2015	2020	2025	2030	2035	2040	2045	2050
Car	23.70	25.17	24.40	23.47	22.37	21.90	21.52	21.23	21.02
Motorcycle	0.7	0.7	0.8	0.8	0.8	0.9	0.9	1.0	1.0
Buses	0.6	0.6	0.7	0.7	0.7	0.8	0.8	0.8	0.9
LGV	3.8	4.0	4.2	4.4	4.5	4.8	5.1	5.3	5.6
HGV	1	1.1	1.1	1.2	1.2	1.3	1.3	1.4	1.5
Mode Shifting	0	0	2.00	4.00	6.01	8.01	10.01	12.01	14.02
TOTAL	29.80	31.64	33.20	34.55	35.67	37.61	39.65	41.79	44.06

			Group 2 (ST migration)	80:20				
	2010	2015	2020	2025	2030	2035	2040	2045	2050
Car	23.70	25.31	25.79	26.08	26.15	27.04	28.05	29.21	30.51
Motorcycle	0.7	0.7	0.8	0.8	0.9	0.9	1.0	1.1	1.1
Buses	0.6	0.6	0.7	0.7	0.7	0.8	0.8	0.9	1.0
LGV	3.8	4.1	4.3	4.5	4.7	5.0	5.4	5.7	6.1
HGV	1	1.1	1.1	1.2	1.2	1.3	1.4	1.5	1.6
Mode Shifting	0	0	1.09	2.18	3.27	4.36	5.45	6.54	7.63
Total	29.80	31.82	32.71	33.35	33.72	35.12	36.67	38.41	40.33
Group 2 (ST migration) 60:40									
	2010	2015	2020	2025	2030	2035	2040	2045	2050
Car	23.70	25.31	24.70	23.90	22.88	22.68	22.60	22.67	22.89
Motorcycle	0.7	0.7	0.8	0.8	0.9	0.9	1.0	1.1	1.1
Buses	0.6	0.6	0.7	0.7	0.7	0.8	0.8	0.9	1.0
LGV	3.8	4.1	4.3	4.5	4.7	5.0	5.4	5.7	6.1
HGV	1	1.1	1.1	1.2	1.2	1.3	1.4	1.5	1.6
Mode Shifting	0	0	2.18	4.36	6.54	8.72	10.90	13.08	15.26
Total	29.80	31.82	31.62	31.17	30.46	30.76	31.22	31.87	32.70

			Grou	p 3 (DfT) 8	0:20				
	2010	2015	2020	2025	2030	2035	2040	2045	2050
Car	23.70	26.42	27.15	27.90	27.81	27.87	27.75	27.66	28.18
Motorcycle	0.7	0.8	0.8	0.8	0.8	0.9	0.9	1.0	1.0
Buses	0.6	0.6	0.7	0.7	0.7	0.8	0.8	0.8	0.9
LGV	3.8	4.7	5.3	5.9	6.5	7.0	7.6	8.1	8.7
HGV	1	1.0	1.0	1.1	1.1	1.1	1.2	1.2	1.2
Mode Shifting	0	0	1.09	2.18	3.27	4.36	5.45	6.54	7.04
Total	29.80	33.52	36.04	38.59	40.26	42.05	43.66	45.36	47.13
			Grou	p 3 (DfT) 6	0:40				
	2010	2015	2020	2025	2030	2035	2040	2045	2050
Car	23.70	26.42	26.06	25.72	24.54	23.51	22.30	21.12	21.13
Motorcycle	0.7	0.8	0.8	0.8	0.8	0.9	0.9	1.0	1.0
Buses	0.6	0.6	0.7	0.7	0.7	0.8	0.8	0.8	0.9
LGV	3.8	4.7	5.3	5.9	6.5	7.0	7.6	8.1	8.7
HGV	1	1.0	1.0	1.1	1.1	1.1	1.2	1.2	1.2
Mode Shifting	0	0	2.18	4.36	6.54	8.72	10.90	13.08	14.09
Total	29.80	33.52	33.86	34.23	33.72	33.33	32.77	32.28	33.04

A.3.3: Zero-Tailpipe Emission Cars Availability versus Demand for Scenarios with

Behaviour Change







A.4: Emission Factors – Urban Scale (Greater London) Analysis for Scenarios without Behaviour Change (i.e. no mode shift) for all Demand Groups

A.4.1. PW_{10}	A	.4.	1:	PM_{10}
------------------	---	-----	----	-----------

			London (to	onnes/bvkm)					
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange									
Cars	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21
LGVs	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08
HGVs	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling						-			-
Cars	0.031	0.028	0.025	0.024	0.024	0.024	0.023	0.021	0.010
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London						•			
Cars	0.031	0.028	0.025	0.023	0.022	0.017	0.010	0.010	0.010
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time						•			
Cars	0.031	0.028	0.025	0.024	0.024	0.024	0.024	0.023	0.010
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09

A.4.2: PM_{2.5}

			London (t	onnes/bvkm)				
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange									
Cars	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.013	0.013	0.013
2W	0.018	0.016	0.013	0.014	0.014	0.015	0.015	0.012	0.007
Buses	0.147	0.073	0.071	0.060	0.045	0.044	0.042	0.041	0.040
LGVs	0.060	0.034	0.017	0.017	0.017	0.017	0.017	0.017	0.017
HGVs	0.110	0.074	0.048	0.048	0.048	0.048	0.048	0.048	0.048
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.021	0.017	0.014	0.014	0.013	0.012	0.009	0.005	0.005
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.013	0.013	0.005
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London									
Cars	0.021	0.017	0.014	0.014	0.012	0.009	0.005	0.005	0.005
2W	0.02	0.02	0.01	0.01	0.01	0.01	0.02	0.01	0.01
Buses	0.15	0.07	0.07	0.06	0.05	0.04	0.04	0.04	0.04
LGVs	0.06	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02
HGVs	0.11	0.07	0.05	0.05	0.05	0.05	0.05	0.05	0.05
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time									
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.014	0.014	0.005
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06
	0 11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11

A.4.3: NO_x

			London (tor	nnes/bvkm)					
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange									
Cars	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37
2W	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20
Buses	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27
LGVs	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89
HGVs	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.05	0.06
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.37	0.21	0.09	0.05	0.04	0.03	0.02	0.00	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.05	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London									
Cars	0.37	0.21	0.09	0.05	0.04	0.02	0.00	0.00	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time									
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.06	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04

A.5: Emission Factors – Urban Scale (Greater London) Analysis for Scenarios with 20% Mode Shift Away from Cars

A.5.1: PM₁₀

A.	5.	1.	1:	Group	1
<i>/ \</i> .	<i>.</i>	<u> </u>	- .	Group	-

			London (to	onnes/bvkm)					
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange	Emission Facto	ors Constant	from 2010						
Cars	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21
LGVs	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08
HGVs	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.01	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.01	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09

			London (to	onnes/bvkm)					
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange	Emission Fac	tors Constant	t from 2010						
Cars	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21
LGVs	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08
HGVs	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.01	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.01	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09

London (tonnes/bvkm)										
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
NoChange	Emission Fac	tors Constan	t from 2010							
Cars	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02	
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	
Buses	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21	
LGVs	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	
HGVs	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
UK										
Cars	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02	
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01	
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08	
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03	
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
50:50										
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.01	0.01	
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01	
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08	
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03	
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Doubling										
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01	
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01	
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08	
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03	
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Clean London	I									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.01	0.01	0.01	
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01	
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08	
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03	
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Just in Time										
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01	
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01	
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08	
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03	
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09	

A.5.2: PM_{2.5}

London (tonnes/bvkm)											
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
NoChange	Emission Fac	tors Constant	t from 2010								
Cars	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
UK											
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.013	0.013	0.013		
2W	0.018	0.016	0.013	0.014	0.014	0.015	0.015	0.012	0.007		
Buses	0.147	0.073	0.071	0.060	0.045	0.044	0.042	0.041	0.040		
LGVs	0.060	0.034	0.017	0.017	0.017	0.017	0.017	0.017	0.017		
HGVs	0.110	0.074	0.048	0.048	0.048	0.048	0.048	0.048	0.048		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
50:50											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Doubling											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Clean London						1					
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.01	0.01	0.01	0.01	0.02	0.01	0.01		
Buses	0.15	0.07	0.07	0.06	0.05	0.04	0.04	0.04	0.04		
LGVs	0.06	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
HGVs	0.11	0.07	0.05	0.05	0.05	0.05	0.05	0.05	0.05		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Just in Time											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		

London (tonnes/bvkm)											
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
NoChange	Emission Fac	tors Constant	from 2010								
Cars	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
UK											
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.013	0.013	0.013		
2W	0.018	0.016	0.013	0.014	0.014	0.015	0.015	0.012	0.007		
Buses	0.147	0.073	0.071	0.060	0.045	0.044	0.042	0.041	0.040		
LGVs	0.060	0.034	0.017	0.017	0.017	0.017	0.017	0.017	0.017		
HGVs	0.110	0.074	0.048	0.048	0.048	0.048	0.048	0.048	0.048		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
50:50											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Doubling											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Clean London	I										
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.01	0.01	0.01	0.01	0.02	0.01	0.01		
Buses	0.15	0.07	0.07	0.06	0.05	0.04	0.04	0.04	0.04		
LGVs	0.06	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
HGVs	0.11	0.07	0.05	0.05	0.05	0.05	0.05	0.05	0.05		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Just in Time											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		

London (tonnes/bvkm)											
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
NoChange	Emission Fac	tors Constant	t from 2010								
Cars	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
UK											
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.013	0.013	0.013		
2W	0.018	0.016	0.013	0.014	0.014	0.015	0.015	0.012	0.007		
Buses	0.147	0.073	0.071	0.060	0.045	0.044	0.042	0.041	0.040		
LGVs	0.060	0.034	0.017	0.017	0.017	0.017	0.017	0.017	0.017		
HGVs	0.110	0.074	0.048	0.048	0.048	0.048	0.048	0.048	0.048		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
50:50											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Doubling											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Clean London											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.01	0.01	0.01	0.01	0.02	0.01	0.01		
Buses	0.15	0.07	0.07	0.06	0.05	0.04	0.04	0.04	0.04		
LGVs	0.06	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
HGVs	0.11	0.07	0.05	0.05	0.05	0.05	0.05	0.05	0.05		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Just in Time											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		

A.5.3: NO_x

			London (tor	nnes/bvkm)					
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange	Emission Fact	ors Constant	from 2010						
Cars	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37
2W	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20
Buses	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27
LGVs	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89
HGVs	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.05	0.06
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.02	0.00	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.04	0.04	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London	1 <u> </u>					-			-
Cars	0.37	0.21	0.09	0.05	0.04	0.03	0.00	0.00	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time									
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.06	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04

			London (tor	nnes/bvkm)					
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange	Emission Fact	tors Constant	from 2010						
Cars	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37
2W	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20
Buses	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27
LGVs	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89
HGVs	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.05	0.06
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.02	0.00	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.04	0.04	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London									-
Cars	0.37	0.21	0.09	0.05	0.04	0.02	0.00	0.00	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time				-		r		-	1
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.06	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04

			London (tor	nes/bvkm)					
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange	Emission Fact	ors Constant	from 2010						
Cars	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37
2W	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20
Buses	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27
LGVs	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89
HGVs	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.05	0.06
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.02	0.00	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.04	0.03	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London									
Cars	0.37	0.21	0.09	0.05	0.04	0.02	0.00	0.00	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time									
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.06	0.00
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04

A.6: Emission Factors – Urban Scale (Greater London) Analysis for Scenarios with 40% Mode Shift Away from Cars

A.6.1: PM₁₀

۸	-	1 1		\sim			1
Α.	5	1.1	:	Gr	ои	р	T

			London (to	onnes/bvkm)					
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange	Emission Fact	ors Constant	from 2010						
Cars	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21
LGVs	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08
HGVs	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.01	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time									
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09

London (tonnes/bvkm)											
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
NoChange	Emission Fac	tors Constant	t from 2010								
Cars	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21		
LGVs	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08		
HGVs	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
UK											
Cars	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
50:50											
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Doubling											
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Clean London											
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Just in Time											
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		

London (tonnes/bvkm)											
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
NoChange	Emission Fac	tors Constant	t from 2010								
Cars	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21	0.21		
LGVs	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08	0.08		
HGVs	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
UK											
Cars	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
50:50											
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Doubling											
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Clean London											
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Just in Time		-					1				
Cars	0.03	0.03	0.03	0.02	0.02	0.02	0.02	0.02	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.01		
Buses	0.21	0.12	0.12	0.10	0.09	0.08	0.08	0.08	0.08		
LGVs	0.08	0.05	0.03	0.03	0.03	0.03	0.03	0.03	0.03		
HGVs	0.15	0.11	0.09	0.09	0.09	0.09	0.09	0.09	0.09		

A.6.2: PM_{2.5}

London (tonnes/bvkm)											
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
NoChange	Emission Fac	tors Constant	from 2010								
Cars	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
UK											
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.013	0.013	0.013		
2W	0.018	0.016	0.013	0.014	0.014	0.015	0.015	0.012	0.007		
Buses	0.147	0.073	0.071	0.060	0.045	0.044	0.042	0.041	0.040		
LGVs	0.060	0.034	0.017	0.017	0.017	0.017	0.017	0.017	0.017		
HGVs	0.110	0.074	0.048	0.048	0.048	0.048	0.048	0.048	0.048		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
50:50											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Doubling											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Clean London											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.01	0.01	0.01	0.01	0.02	0.01	0.01		
Buses	0.15	0.07	0.07	0.06	0.05	0.04	0.04	0.04	0.04		
LGVs	0.06	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
HGVs	0.11	0.07	0.05	0.05	0.05	0.05	0.05	0.05	0.05		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Just in Time											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		

London (tonnes/bvkm)											
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
NoChange	Emission Fac	tors Constant	from 2010								
Cars	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
UK											
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.013	0.013	0.013		
2W	0.018	0.016	0.013	0.014	0.014	0.015	0.015	0.012	0.007		
Buses	0.147	0.073	0.071	0.060	0.045	0.044	0.042	0.041	0.040		
LGVs	0.060	0.034	0.017	0.017	0.017	0.017	0.017	0.017	0.017		
HGVs	0.110	0.074	0.048	0.048	0.048	0.048	0.048	0.048	0.048		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
50:50	•										
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Doubling											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Clean London											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.01	0.01	0.01	0.01	0.02	0.01	0.01		
Buses	0.15	0.07	0.07	0.06	0.05	0.04	0.04	0.04	0.04		
LGVs	0.06	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
HGVs	0.11	0.07	0.05	0.05	0.05	0.05	0.05	0.05	0.05		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Just in Time											
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01		
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02		
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15		
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06		
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11		

London (tonnes/bvkm)									
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
NoChange	Emission Fac	tors Constant	from 2010						
Cars	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211	0.0211
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
UK									
Cars	0.021	0.017	0.014	0.014	0.014	0.014	0.013	0.013	0.013
2W	0.018	0.016	0.013	0.014	0.014	0.015	0.015	0.012	0.007
Buses	0.147	0.073	0.071	0.060	0.045	0.044	0.042	0.041	0.040
LGVs	0.060	0.034	0.017	0.017	0.017	0.017	0.017	0.017	0.017
HGVs	0.110	0.074	0.048	0.048	0.048	0.048	0.048	0.048	0.048
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
50:50									
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Doubling									
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Clean London									
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01
2W	0.02	0.02	0.01	0.01	0.01	0.01	0.02	0.01	0.01
Buses	0.15	0.07	0.07	0.06	0.05	0.04	0.04	0.04	0.04
LGVs	0.06	0.03	0.02	0.02	0.02	0.02	0.02	0.02	0.02
HGVs	0.11	0.07	0.05	0.05	0.05	0.05	0.05	0.05	0.05
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050
Just in Time									
Cars	0.02	0.02	0.01	0.01	0.01	0.01	0.01	0.01	0.01
2W	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02	0.02
Buses	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15	0.15
LGVs	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06	0.06
HGVs	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11	0.11

A.6.3: NO_x

London (tonnes/bvkm)										
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
NoChange	Emission Fact	ors Constant	from 2010							
Cars	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	
2W	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	
Buses	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	
LGVs	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	
HGVs	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
UK										
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.05	0.06	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
50:50										
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.03	0.00	0.00	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Doubling										
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.05	0.05	0.00	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Clean Londor	1									
Cars	0.37	0.21	0.09	0.05	0.04	0.03	0.00	0.00	0.00	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Just in Time										
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.06	0.00	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	

London (tonnes/bvkm)										
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
NoChange	Emission Fact	ors Constant	from 2010							
Cars	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	
2W	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	
Buses	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	
LGVs	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	
HGVs	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
UK										
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.05	0.06	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
50:50										
Cars	0.37	0.21	0.09	0.05	0.05	0.04	0.04	0.00	0.00	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Doubling										
Cars	0.37	0.21	0.09	0.05	0.05	0.04	0.04	0.04	0.00	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Clean London						1			-	
Cars	0.37	0.21	0.09	0.05	0.05	0.04	0.00	0.00	0.00	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050	
Just in Time						1			-	
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.06	0.00	
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00	
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19	
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03	
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04	

	London (tonnes/bvkm)										
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
NoChange	Emission Facto	ors Constant f	rom 2010								
Cars	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37	0.37		
2W	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20	0.20		
Buses	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27	7.27		
LGVs	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89	0.89		
HGVs	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05	4.05		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
UK											
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.05	0.06		
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00		
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19		
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03		
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
50:50											
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.03	0.00	0.00		
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00		
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19		
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03		
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Doubling											
Cars	0.37	0.21	0.09	0.05	0.04	0.05	0.05	0.05	0.00		
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00		
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19		
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03		
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Clean London	1										
Cars	0.37	0.21	0.09	0.05	0.04	0.04	0.00	0.00	0.00		
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00		
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19		
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03		
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04		
kt/bvkm	2010	2015	2020	2025	2030	2035	2040	2045	2050		
Just in Time											
Cars	0.37	0.21	0.09	0.05	0.05	0.05	0.05	0.06	0.00		
2W	0.20	0.18	0.14	0.14	0.15	0.15	0.16	0.09	0.00		
Buses	7.27	2.71	2.62	1.70	0.43	0.41	0.32	0.19	0.19		
LGVs	0.89	0.36	0.11	0.11	0.05	0.06	0.04	0.04	0.03		
HGVs	4.05	1.62	0.29	0.21	0.17	0.15	0.13	0.09	0.04		

This page was intentionally left blank.